IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re:

GRACEWAY PHARMACEUTICALS, LLC, et al..¹

Chapter 11

Case No. 11-13036 (PJW)

Jointly Administered

Debtors.

DISCLOSURE STATEMENT FOR THE FIRST AMENDED JOINT PLAN OF LIQUIDATION OF GRACEWAY PHARMACEUTICALS, LLC, et al.

David S. Heller, Esq.
Josef S. Athanas, Esq.
Matthew L. Warren, Esq.
LATHAM & WATKINS LLP
Suite 5800
233 South Wacker Drive
Chicago, Illinois 60606
(312) 876-7700

-and-

Michael R. Nestor (No. 3526)
Kara Hammond Coyle (No. 4410)
YOUNG CONAWAY STARGATT & TAYLOR, LLP
Rodney Square
1000 North King Street
Wilmington, DE 19801
(302) 571-6600

Attorneys for the Debtors and Debtors-in-Possession

Dated: March 1, 2012

The Debtors in these cases, along with the last four digits of each Debtor's federal tax identification number, are: Graceway Pharma Holding Corp., a Delaware corporation (9175), Case No. 11-13037 (PJW); Graceway Holdings, LLC, a Delaware limited liability company (2502), Case No. 11-13038 (PJW); Graceway Pharmaceuticals, LLC, a Delaware limited liability company (5385), Case No. 11-13036 (PJW); Chester Valley Holdings, LLC, a Delaware limited liability company (9457), Case No. 11-13039 (PJW); Chester Valley Pharmaceuticals, LLC, a Delaware limited liability company (3713), Case No. 11-13041 (PJW); Graceway Canada Holdings, Inc., a Delaware corporation (6663), Case No. 11-13042 (PJW); and Graceway International, Inc., a Delaware corporation (2399), Case No. 11-13043 (PJW). The mailing address for Graceway Pharmaceuticals, LLC is 340 Martin Luther King Jr. Blvd., Suite 400, Bristol, TN 37620 (Attn: John Bellamy). On October 4, 2011, Graceway Canada Company filed an application in the Ontario Superior Court of Justice (Commercial List) pursuant to the *Courts of Justice Act*, R.S.O. 1990, c. C. 43.

I. INT	RODU	CTION	1
	A.	Overview of the Plan	3
	B.	Voting Instructions	
	C.	Confirmation of the Plan by the Bankruptcy Court	
II DA	CVCD	OLDID OF THE DEDTODS	1.2
Π . BA	CKGR	OUND OF THE DEBTORS	13
	A.	The Debtors	13
	В.	Events Leading to Chapter 11	14
III. TI	HE CHA	APTER 11 CASES	16
	A.	Overview of the Chapter 11 Cases	16
	B.	Retained Professionals	17
	C.	First Day Orders	
	D.	Cash Collateral and Debtor-In-Possession Financing	
	E.	Sale of Assets	
	F.	The Committee Settlement	23
	G.	The Debtors' Current Management	25
	H.	The Claims Process	
	I.	Substantive Consolidation of the Debtors	26
IV C	НАРТБ	R 11 PLAN	27
1,, 0			
	A.	Plan Overview	28
	В.	Plan Summary	28
	C.	Unclassified and/or Unimpaired Claims	28
	D.	Treatment of Claims and Interests	30
	E.	Post-Confirmation Operations of the Debtors	36
	F.	Liquidating Trust Agreement	42
	G.	Procedures for Treatment of Disputed Claims	55
	H.	Distributions Under the Plan	56
	I.	Conditions to Confirmation	59
	J.	Conditions to the Effective Date	59
	K.	Termination of Plan for Failure To Become Effective.	
	L.	Waiver of Conditions	
	M.	Modification of the Plan	
	N.	Effect of Confirmation	
	O.	Exculpation, Injunction, and Limitation of Liability	
	P.	Retention of Jurisdiction	
	Q.	Treatment of Executory Contracts and Unexpired Leases	
	R.	No Admissions	
	S.	Preservation of Rights of Setoffs	
	T.	Defenses with Respect to Unimpaired Claims	
	/	Dissolution of Creditors' Committee	

V. CONFIR	MATION OF THE PLAN	67
A.	Confirmation Hearing	67
В.	Confirmation Standards	68
VI FUNDI	NG AND FEASIBILITY OF THE PLAN	68
VI. I ONDI		00
A.	Funding of the Plan	68
В.	Best Interests Test	
C.	Avoidance Action Analysis	70
D.	Nycomed Litigation	72
E.	The Debtors' Remaining Property	74
F.	Feasibility	76
G.	Risk Factors Associated with the Plan	76
VII. ALTEI	RNATIVES TO THE PLAN	77
VIII. CERT	AIN U.S. FEDERAL INCOME TAX CONSIDERATIONS	78
A.	In General	78
В.	U.S. Federal Income Tax Consequences to the Debtors	
C.	U.S. Federal Income Tax Treatment of the Liquidating Trust	
D.	Disputed Claims Reserve	
E.	U.S. Federal Income Tax Consequences to Holders of Claims	
F.	Backup Withholding and Information Reporting	

Exhibits To Disclosure Statement

Exhibit A	First Amended Joint Plan of Liquidation
Exhibit B	Claims Objections
Exhibit C	Curriculum Vitae of Kip Horton
Exhibit D	Liquidation Analysis
Exhibit E	Non-Insider Transfers Subject to Preference Review
Exhibit F	Insider Transfers Subject to Preference Review
Exhibit G	Distribution to Insiders On March 18, 2010

DISCLOSURE STATEMENT FOR THE FIRST AMENDED JOINT PLAN OF LIQUIDATION OF GRACEWAY PHARMACEUTICALS, LLC, et al.

I. INTRODUCTION

Graceway Pharma Holding Corp., a Delaware corporation; Graceway Holdings, LLC, a Delaware limited liability company; Graceway Pharmaceuticals, LLC, a Delaware limited liability company; Chester Valley Holdings, LLC, a Delaware limited liability company; Chester Valley Pharmaceuticals, LLC, a Delaware limited liability company; Graceway Canada Holdings, Inc., a Delaware corporation; and Graceway International, Inc., a Delaware corporation (collectively, the "Debtors" or the "Company"), provide this Disclosure Statement (the "Disclosure Statement") to certain of the Debtors' Impaired Creditors to permit such Creditors to make an informed decision in voting to accept or reject the First Amended Joint Plan of Liquidation of Graceway Pharmaceuticals, LLC, et al. (the "Plan") filed on February 28, 2012 with the Bankruptcy Court (defined below) in connection with the above-captioned cases filed pursuant to chapter 11 of the Bankruptcy Code (the "Chapter 11 Cases"). A copy of the Plan is attached to this Disclosure Statement as Exhibit A. Capitalized terms used herein but not otherwise defined have the meanings assigned to such terms in the Plan. Whenever the words "include," "includes" or "including" are used in this Disclosure Statement, they are deemed to be followed by the words "without limitation."

This Disclosure Statement is presented to certain holders of Claims against the Debtors in accordance with the requirements of Section 1125 of the United States Bankruptcy Code, 11 U.S.C. §§ 101-1532 (the "Bankruptcy Code"). Section 1125 of the Bankruptcy Code requires that a disclosure statement provide information sufficient to enable a hypothetical and reasonable investor, typical of the Debtors' Creditors and stockholders, to make an informed judgment whether to accept or reject the Plan. This Disclosure Statement may not be relied upon for any purpose other than that described above.

The Disclosure Statement is based on information publicly available in pleadings filed with the Bankruptcy Court, information provided by the Debtors' management, claims information provided by BMC Group, Inc. ("BMC"), the official claims agent, a liquidation analysis (the "Liquidation Analysis") prepared by the Debtors with the assistance of Alvarez and Marsal North America, LLC ("Alvarez"), financial advisors to the Debtors, and legal analysis by Latham & Watkins LLP ("Latham & Watkins") and Young Conaway Stargatt & Taylor, LLP ("Young Conaway"), co-counsel for the Debtors.

THIS DISCLOSURE STATEMENT AND THE PLAN ARE AN INTEGRAL PACKAGE, AND THEY MUST BE CONSIDERED TOGETHER FOR THE READER TO BE ADEQUATELY INFORMED. THIS INTRODUCTION IS QUALIFIED IN ITS ENTIRETY BY THE REMAINING PORTIONS OF THIS DISCLOSURE STATEMENT, AND THIS DISCLOSURE STATEMENT IN TURN IS QUALIFIED, IN ITS ENTIRETY, BY THE PLAN.

NO REPRESENTATIONS CONCERNING THE DEBTORS (PARTICULARLY AS TO THE VALUE OF THEIR PROPERTY) ARE AUTHORIZED BY THE DEBTORS OTHER THAN AS SET FORTH IN THIS DISCLOSURE STATEMENT AND ITS EXHIBITS. ANY REPRESENTATIONS OR INDUCEMENTS MADE TO SECURE YOUR ACCEPTANCE OF THE PLAN OTHER THAN AS CONTAINED IN THIS DISCLOSURE STATEMENT AND ITS EXHIBITS SHOULD NOT BE RELIED UPON BY YOU IN ARRIVING AT YOUR DECISION, AND SUCH ADDITIONAL REPRESENTATIONS AND INDUCEMENTS SHOULD BE REPORTED TO COUNSEL FOR THE DEBTORS, WHO WILL IN TURN DELIVER SUCH INFORMATION TO THE BANKRUPTCY COURT FOR SUCH ACTION AS MAY BE APPROPRIATE.

THE INFORMATION CONTAINED IN THIS DISCLOSURE STATEMENT, INCLUDING ANY EXHIBITS CONCERNING THE FINANCIAL CONDITION OF THE DEBTORS AND THE OTHER INFORMATION CONTAINED HEREIN, HAS NOT BEEN SUBJECT TO AN AUDIT OR INDEPENDENT REVIEW EXCEPT AS EXPRESSLY SET FORTH HEREIN. ACCORDINGLY, THE DEBTORS ARE UNABLE TO WARRANT OR REPRESENT THAT THE INFORMATION CONCERNING THE DEBTORS OR THEIR FINANCIAL CONDITION IS ACCURATE OR COMPLETE. THE PROJECTED INFORMATION CONTAINED IN THIS DISCLOSURE STATEMENT HAS BEEN PRESENTED FOR ILLUSTRATIVE PURPOSES ONLY, AND, BECAUSE OF THE UNCERTAINTY AND RISK FACTORS INVOLVED, ACTUAL RESULTS MAY NOT BE AS PROJECTED HEREIN.

ALTHOUGH AN EFFORT HAS BEEN MADE TO BE ACCURATE AND THE DEBTORS BELIEVE IN GOOD FAITH THAT THE INFORMATION HEREIN IS ACCURATE, THE DEBTORS DO NOT WARRANT OR REPRESENT THAT THE INFORMATION CONTAINED IN THIS DISCLOSURE STATEMENT AND ITS EXHIBITS IS CORRECT. THIS DISCLOSURE STATEMENT CONTAINS ONLY A SUMMARY OF THE PLAN. EACH CREDITOR IS STRONGLY URGED TO REVIEW THE PLAN PRIOR TO VOTING ON IT.

THE STATEMENTS CONTAINED IN THIS DISCLOSURE STATEMENT ARE MADE AS OF THE DATE OF THIS DISCLOSURE STATEMENT UNLESS ANOTHER TIME IS SPECIFIED. THE DELIVERY OF THIS DISCLOSURE STATEMENT WILL NOT UNDER ANY CIRCUMSTANCES CREATE AN IMPLICATION THAT THERE HAS NOT BEEN ANY CHANGE IN THE FACTS SET FORTH SINCE THE DATE OF THIS DISCLOSURE STATEMENT.

A STATEMENT OF THE ASSETS AND LIABILITIES OF THE DEBTORS AS OF THE DATE OF THE COMMENCEMENT OF THEIR CHAPTER 11 CASES IS ON FILE WITH THE CLERK OF THE BANKRUPTCY COURT AND MAY BE INSPECTED BY INTERESTED PARTIES DURING REGULAR BUSINESS HOURS.

THIS DISCLOSURE STATEMENT HAS BEEN PREPARED IN ACCORDANCE WITH SECTION 1125 OF THE BANKRUPTCY CODE AND NOT IN ACCORDANCE WITH FEDERAL OR STATE SECURITIES LAWS OR OTHER

APPLICABLE NON-BANKRUPTCY LAW. ENTITIES HOLDING OR TRADING IN OR OTHERWISE PURCHASING, SELLING OR TRANSFERRING CLAIMS AGAINST, INTERESTS IN OR SECURITIES OF, THE DEBTORS SHOULD EVALUATE THIS DISCLOSURE STATEMENT ONLY IN LIGHT OF THE PURPOSE FOR WHICH IT WAS PREPARED.

THIS DISCLOSURE STATEMENT HAS NOT BEEN APPROVED OR DISAPPROVED BY THE SECURITIES AND EXCHANGE COMMISSION NOR HAS SUCH COMMISSION PASSED UPON THE ACCURACY OR ADEQUACY OF THE STATEMENTS CONTAINED HEREIN.

THIS DISCLOSURE STATEMENT WILL NOT BE CONSTRUED TO BE ADVICE ON THE TAX, SECURITIES OR OTHER LEGAL EFFECTS OF THE PLAN. EACH CREDITOR SHOULD, THEREFORE, CONSULT WITH ITS OWN LEGAL, BUSINESS, FINANCIAL AND TAX ADVISERS AS TO ANY SUCH MATTERS CONCERNING THE SOLICITATION, THE PLAN OR THE TRANSACTIONS CONTEMPLATED THEREBY.

Pursuant to the Bankruptcy Code, this Disclosure Statement was filed on March 1, 2012 and the Plan was filed on February 28, 2012. The Bankruptcy Court will hold a hearing on confirmation of the Plan beginning at 2:00 p.m. (prevailing Eastern time) on April 11, 2012, in the United States Bankruptcy Court, 824 North Market Street, 5th Floor, Wilmington, Delaware 19801 (the "Confirmation Hearing"). At that Confirmation Hearing, the Bankruptcy Court will consider whether the Plan satisfies the requirements of the Bankruptcy Code, including whether the Plan is in the best interests of the claimants, and will review a ballot report concerning votes cast for acceptance or rejection of the Plan.

To obtain, at your cost, additional copies of this Disclosure Statement or of the Plan, please contact:

BMC Group, Inc. Attn: Graceway Pharmaceuticals Document Requests PO Box 3020 Chanhassen, MN 55317-3020 Telephone: 888-909-0100 www.bmcgroup.com/graceway

A. Overview of the Plan

THE FOLLOWING IS A BRIEF SUMMARY OF THE TREATMENT OF CLAIMS AND INTERESTS UNDER THE PLAN. THE DESCRIPTION OF THE PLAN SET FORTH BELOW CONSTITUTES A SUMMARY ONLY AND IS QUALIFIED, IN ITS ENTIRETY, BY THE PLAN, THE LIQUIDATING TRUST AGREEMENT, AND THE OTHER PLAN DOCUMENTS. CREDITORS AND OTHER PARTIES IN INTEREST ARE URGED TO REVIEW THE MORE DETAILED DESCRIPTION OF THE PLAN CONTAINED IN SECTION IV OF THIS DISCLOSURE STATEMENT AND

THE PLAN ITSELF. THE PLAN IS ATTACHED AS EXHIBIT A TO THIS DISCLOSURE STATEMENT. IN THE EVENT OF ANY INCONSISTENCY BETWEEN THIS DISCLOSURE STATEMENT AND THE PLAN, THE PLAN CONTROLS.

Chapter 11 is the chapter of the Bankruptcy Code primarily used for business reorganization. The fundamental purpose of a chapter 11 case is to formulate a plan to restructure a debtor's finances so as to maximize recoveries to its creditors. With this purpose in mind, businesses sometimes use chapter 11 as a means to conduct asset sales and other forms of liquidation. Whether the aim is reorganization or liquidation, a chapter 11 plan sets forth and governs the treatment and rights to be afforded to creditors and stockholders with respect to their claims against and equity interests in a debtor's bankruptcy estate.

As an initial matter, the Plan is premised on the substantive consolidation of all of the Debtors with respect to the treatment of all Claims and Interests, except for the Other Secured Claims in Class 4. This means that the Debtors propose to satisfy the claims of all their respective Creditors from a common pool comprised of their collective assets. The Plan divides the Claims against and Interests in the Debtors into Classes. Certain Claims – in particular, Administrative Expense Claims, Statutory Fees, Professional Claims and Priority Tax Claims – remain unclassified in accordance with Section 1123(a)(1) of the Bankruptcy Code. The Plan assigns all other Claims and Interests to Classes as described below.

Class 1 consists of all Other Priority Claims. On the Effective Date or as soon thereafter as is reasonably practicable in recognition of the applicable Claims reconciliation process set forth in the Plan, each Allowed Other Priority Claim shall receive on account of the Allowed Other Priority Claim and in full satisfaction, settlement and release of and in exchange for such Allowed Other Priority Claim, (i) Cash equal to the amount of such Allowed Other Priority Claim, or (ii) such other treatment as to which the Debtors and the holder of such Allowed Other Priority Claim have agreed upon in writing, provided such treatment is on more favorable terms to the Debtors or the Liquidating Trustee, as the case may be, than the treatment set forth in subsection (i) hereof. Class 1 is Unimpaired and will be deemed to accept the Plan.

Class 2 consists of the First Lien Facility Secured Claim, any First Lien Facility Deficiency Claim and any First Lien Facility Adequate Protection Claim. On the Effective Date or as soon thereafter as is reasonably practicable in recognition of the applicable Claims reconciliation process set forth in the Plan, Bank of America, N.A., as administrative agent for the First Lien Facility Lenders (defined below) and collateral agent for the First Lien Claimholders, or any successor thereto (the "First Lien Facility Agent") shall receive, for the benefit of the First Lien Claimholders, in full satisfaction, settlement and release of and in exchange for the First Lien Facility Claims (including the First Lien Facility Secured Claim, any First Lien Facility Adequate Protection Claim and any First Lien Facility Deficiency Claim), (i) any First Lien Available Cash that is available to be distributed on the Effective Date; (ii) the Nycomed Litigation Proceeds; and (iii) beneficial interests in the Liquidating Trust entitling each First Lien Claimholder to receive its Pro Rata share of (A) any First Lien Available Cash that becomes available to be distributed after the Effective Date, (B) seventy-five percent (75%) of any Avoidance Action Proceeds, (C) one hundred percent (100%) of any other Net Causes of Action Proceeds, including, without limitation, 549 Avoidance Action Proceeds, and (D) all

Cash from the collection of accounts receivable. Until the First Lien Facility Claims are paid in full in Cash, the Liens securing the First Lien Facility Secured Claim and any First Lien Facility Adequate Protection Claim shall remain on the Assets (including, without limitation, the Reserve Accounts and the Cash deposited therein and the Nycomed Litigation Proceeds) other than the Other Assets, provided that, except for their obligations under the Plan, the Debtors and their Estates shall have no further liability to the First Lien Facility Agent or the lenders under the First Lien Credit Agreement (defined below) (the "First Lien Facility Lenders"). Notwithstanding the full satisfaction, settlement and release of the First Lien Facility Claims under the Plan, the aggregate amount of the beneficial interests in the Liquidating Trust the First Lien Claimholders receive under Section 5.2 of the Plan in exchange for such Claims shall increase from time to time by an amount equal to the interest accrual on, and the fees incurred on account of, such Claims to and including the applicable date of determination (calculated at the rate provided for under the First Lien Credit Agreement and in accordance with the terms thereof and of the Intercreditor Agreement) as if such Claims had not been satisfied, settled and released under the Plan; it being understood that distributions on account of such interest accrual and incurred fees shall only be permitted in the event the First Lien Facility Secured Claim (had it not been satisfied, settled and released under the Plan) is oversecured. Holders of the First Lien Facility Claims shall be Beneficiaries; it being understood that distributions from the Liquidating Trust shall be exclusively governed in accordance with the terms set forth in the Plan. Class 2 is Impaired and may vote to accept or reject the Plan.

Class 3 consists of any Second Lien Facility Secured Claim and any Second Lien Facility Adequate Protection Claim. Deutsche Bank Trust Company Americas, as agent for the Second Lien Facility Lenders (defined below) under the Second Lien Credit Agreement (defined below), or any successor thereto (the "Second Lien Facility Agent") shall receive, for the benefit of the Second Lien Facility Lenders, in full satisfaction, settlement and release of and in exchange for any Allowed Second Lien Facility Secured Claim and any Allowed Second Lien Facility Adequate Protection Claim on the Effective Date or as soon thereafter as is practicable in recognition of the applicable Claims reconciliation process set forth in the Plan, beneficial interests in the Liquidating Trust entitling each Second Lien Facility Lender to receive its Pro Rata share of (a) any Other Available Cash subject to the Liens granted to the Second Lien Facility Agent pursuant to the Second Lien Credit Agreement or the DIP Order (defined below) and (b) after payment in full in Cash of the First Lien Facility Claims, (i) any First Lien Available Cash; (ii) the Nycomed Litigation Proceeds, (iii) seventy-five percent (75%) of any Avoidance Action Proceeds; (iv) one hundred percent (100%) of any other Net Causes of Action Proceeds, including, without limitation, any 549 Avoidance Action Proceeds and (v) all Cash from the collection of accounts receivable. Until the Second Lien Facility Claims are paid in full in Cash, the Liens securing any Second Lien Facility Secured Claim and any Second Lien Facility Adequate Protection Claim shall remain on the Assets, provided that, (a) except for their obligations under the Plan, the Debtors and their Estates shall have no further liability to the Second Lien Facility Agent or the Second Lien Facility Lenders and (b) such Liens shall continue to be junior and subordinate in all respects to the Liens securing the First Lien Facility Secured Claim and First Lien Adequate Protection Claim that have not been released in connection with the Committee Settlement until such Claims are paid in full in Cash. Any Second Lien Facility Deficiency Claim shall not constitute a Class 3 Second Lien Facility Claims and shall be treated as a Class 5 General Unsecured Claim under the Plan. Holders of the Allowed Second Lien Facility Claims shall be Beneficiaries; it being understood that distributions from the Liquidating Trust shall be exclusively governed in accordance with the terms set forth in the Plan. Class 3 is Impaired and may vote to accept or reject the Plan.

Class 4 consists of all Other Secured Claims. This Class is subdivided into subclasses designated by letters of the alphabet (Class 4A, Class 4B and so on), so that each holder of any Other Secured Claim is in a Class by itself, except to the extent that there are Other Secured Claims that are substantially similar to each other and may be included within a single Class, and except for a precautionary class of otherwise unclassified Other Secured Claims. On the Effective Date or as soon thereafter as is practicable in recognition of the applicable Claims reconciliation process set forth in the Plan, each holder of an Allowed Other Secured Claim that was not assumed by the Purchaser (defined below) in connection with the 363 Sale shall receive, in full satisfaction, settlement and release of and in exchange for such Allowed Other Secured Claim, (i) the collateral securing any such Allowed Other Secured Claim (to the extent such collateral does not constitute collateral securing the First Lien Facility Secured Claim, any First Lien Facility Adequate Protection Claim, the Second Lien Facility Secured Claim or any Second Lien Facility Adequate Protection Claim or, if such collateral does secure any such Claim, to the extent that the Lien securing such Allowed Other Secured Claim is senior to the Lien securing the First Lien Facility Secured Claim, any First Lien Facility Adequate Protection Claim, the Second Lien Facility Secured Claim or any Second Lien Facility Adequate Protection Claim, as applicable), (ii) Cash in an amount equal to such Allowed Other Secured Claim to the extent the collateral securing such Allowed Other Secured Claim was sold and such collateral does not constitute collateral securing the First Lien Facility Secured Claim, any First Lien Facility Adequate Protection Claim, the Second Lien Facility Secured Claim or any Second Lien Facility Adequate Protection Claim or, if such collateral does secure any such Claim, to the extent that the Lien securing such Allowed Other Secured Claim is senior to the Lien securing the First Lien Facility Secured Claim, any First Lien Facility Adequate Protection Claim, the Second Lien Facility Secured Claim or any Second Lien Facility Adequate Protection Claim, as applicable, or (iii) such other treatment that leaves such Allowed Other Secured Claim Unimpaired pursuant to the Section 1124(2) of the Bankruptcy Code. The Other Secured Claim Liens shall be released and the Debtors and their Estates shall have no further liability therefor; provided, however, that any Deficiency Claims of holders of Class 4 Other Secured Claims shall not constitute Class 4 Other Secured Claims and shall be treated as Class 5 General Unsecured Claims under the Plan.² Class 4 is Unimpaired and will be deemed to accept the Plan.

Class 5 consists of all General Unsecured Claims. On the Effective Date or as soon thereafter as is practicable in recognition of the applicable Claims reconciliation process set forth in the Plan, the holders of Allowed General Unsecured Claims shall receive, in full satisfaction, settlement and release of and in exchange for such Allowed Claims, their Pro Rata

The Debtors will deliver a Solicitation Package (as defined in the Motion of the Debtors for Entry of an Order (a) Approving the Disclosure Statement, (b) Establishing the Voting Record Date, Voting Deadline and Other Dates, (c) Approving Procedures for Soliciting, Receiving and Tabulating Votes on the Plan and for Filing Objections to the Plan and (d) Approving the Manner and Forms of Notices and Other Related Documents [Docket No. 485], filed on January 25, 2012) to each party that has asserted an Other Secured Claim and each such party shall have the right to indicate what portion of such asserted Other Secured Claim should be treated as, and counted for voting purposes as, a Class 5 General Unsecured Claim.

share of the Other Available Cash that is not subject to the Liens granted to the Second Lien Facility Agent pursuant to the Second Lien Credit Agreement or the DIP Order, after payment in full of all Allowed Other Secured Claims (to the extent not otherwise satisfied in accordance with the terms of the Plan), Allowed Administrative Expense Claims (other than Administrative Expense Claims that are Assumed Liabilities), Allowed Priority Tax Claims and Allowed Other Priority Claims; provided, however, that pursuant to and in accordance with the Mezzanine Credit Agreement (defined below), any amounts distributed on account of the Allowed Mezzanine Facility Claim (a) are subject and subordinate to payment in full in Cash of the Second Lien Obligations and (b) shall be distributed directly to the Second Lien Facility Agent for the benefit of the holders of Allowed Second Lien Facility Claims until the Second Lien Obligations are paid in full in Cash; provided further, however, that upon notice to the Liquidating Trustee from the Second Lien Facility Agent that the Second Lien Facility Agent has received \$10,000,000.00 in Cash from the Other Available Cash for distribution to holders of Allowed Second Lien Facility Claims in excess of any legal, trustee or other fees and expenses that constitute Second Lien Obligations, the Liquidating Trustee shall cease any distributions to the Second Lien Facility Agent solely on account of the Allowed Mezzanine Facility Claim until the Liquidating Trustee has distributed \$250,000.00 on account of the Allowed Mezzanine Facility Claim (which shall be paid out as if no subordination were in place), after which time the Liquidating Trustee shall resume any distributions to the Second Lien Facility Agent on account of the Allowed Mezzanine Facility Claim until the Second Lien Obligations are paid in full in Cash. Subject to the foregoing proviso, in the event any distribution on account of the Allowed Mezzanine Facility Claim is made other than to the Second Lien Facility Agent, such distribution shall be held in trust for the benefit of, and paid over and delivered to, the Second Lien Facility Agent for the benefit of the Second Lien Facility Lenders. Holders of Allowed General Unsecured Claims shall be Beneficiaries; it being understood that distributions from the Liquidating Trust shall be exclusively governed in accordance with the terms set forth in the Plan and beneficial interests in the Liquidating Trust granted to such holders will not entitle them to any distribution of more than twenty-five percent (25%) of any Avoidance Action Proceeds or any distribution of any other Causes of Action Proceeds, including, without limitation, any 549 Avoidance Action Proceeds. Class 5 is Impaired and may vote to accept or reject the Plan.

Class 6 consists of all Intercompany Claims. As a result of substantive consolidation of the Debtors for distribution purposes under the Plan as provided in Section 7.1 of the Plan, holders of Intercompany Claims will not receive any distribution of property under the Plan on account of their Intercompany Claims and, on the Effective Date, the Intercompany Claims will be cancelled; *provided*, *however*, Class 6 shall exclude any Claims of a Debtor against a non-Debtor Affiliate or a non-Debtor subsidiary. Class 6 is Impaired and will be deemed to reject the Plan.

Class 7 consists of all Interests arising from or relating to Old Equity. Holders of Old Equity Interests will not receive and will not retain any property of the Debtors under the Plan on account of such Interests and all Old Equity Interests will be cancelled as of the Effective Date. Class 7 is impaired and will be deemed to reject the Plan.

Class 8 consists of all Claims and Interests arising from or relating to Old Equity Rights. Holders of any Claim or Interest arising from or relating to Old Equity Rights will not receive and will not retain any property of the Debtors under the Plan on account of such Claim

or Interest and all Old Equity Rights and any Interests or Claims arising from or relating thereto will be cancelled as of the Effective Date. Class 8 is Impaired and will be deemed to reject the Plan.

The Debtors believe that the distributions under the Plan will provide Creditors of the Debtors at least the same recovery on account of Allowed Claims as would distributions by a chapter 7 trustee. However, distributions under the Plan to Creditors of the Debtors would be made more quickly than distributions by a chapter 7 trustee and a chapter 7 trustee would charge a substantial fee, reducing the amount available for distribution on account of Allowed Claims.

ACCORDINGLY, THE DEBTORS URGE EACH CREDITOR ENTITLED TO VOTE ON THE PLAN TO VOTE TO <u>ACCEPT</u> THE PLAN.

Following are detailed, Class-by-Class summaries of the estimated Allowed Claims against the Debtors and the estimated distribution to the Creditors of the Debtors on account of such Allowed Claims.

The Debtors calculated the total asserted claims (other than Administrative Expense Claims)³ based on filed and scheduled claims. To the extent a Creditor filed a proof of Claim that superseded a scheduled claim, the amount in the filed proof of Claim was utilized.

The Debtors then reviewed all of the proofs of Claims asserting First Lien Facility Claims, Second Lien Facility Claims, Other Secured Claims, Priority Tax Claims and Other Priority Claims, and a large sample size of the proofs of Claims asserting General Unsecured Claims.

The Debtors then determined a range of estimated Allowed Claims by reducing asserted claims amounts for duplicate claims, amended claims, misclassified claims, settled claims, repaid claims and claims that lack merit.

The differences between asserted claims and Allowed Claims estimated in the Plan are typical of large chapter 11 cases. See, e.g., In re GST Telecom Inc., Case No. 00-1982 (GMS) (Bankr. D. Del. 2000) (over \$5 billion of claims filed and \$1.25 billion of claims scheduled, approximately \$1.18 billion of claims allowed).

THE RECOVERY PERCENTAGES SET FORTH IN THE FOLLOWING TABLE ARE MERELY ESTIMATES. THE ACTUAL AMOUNTS DISTRIBUTED TO HOLDERS OF ALLOWED CLAIMS UNDER THE PLAN MAY BE HIGHER OR LOWER THAN ESTIMATED.

Asserted Administrative Expense Claims are based on filed, accrued and budgeted amounts.

Class	Description	Approximate Amount Asserted ⁴	Range of Estimated Allowed Amounts	Range of Estimated Recovery	Status
N/A	Administrative Expense Claims	\$4,675,015	\$11,438,000 — \$19,150,000	100%	Unclassified and not entitled to vote
N/A	Professional Claims	N/A	< \$2,189,000	100%	Unclassified and not entitled to vote
N/A	Statutory Fees	N/A	<\$50,000	100%	Unclassified and not entitled to vote
N/A	Priority Tax Claims	\$672,286	< \$694,000	100%	Unclassified and not entitled to vote
1	Other Priority Claims	\$443,380	<\$50,000	100%	Unimpaired, deemed to accept the Plan and not entitled to vote
2	First Lien Facility Claim	\$2,584,190,385	\$432,986,000 - \$446,446,000 (prior to partial satisfaction) ⁵	96.7 - 100% (including prior partial satisfaction)	Impaired and entitled to vote
3	Second Lien Facility Claim – Secured Amount	\$2,183,672,889	\$4,852,000 - \$6,369,000	100%6	Impaired and entitled to vote
4	Other Secured Claims	\$4,742,932	<\$350,000 (requiring cash payment)	100%	Unimpaired, deemed to accept the Plan and not entitled to vote
5	General Unsecured Claims	\$157,268,679	\$441,457,000 - \$477,576,000	0.9 - 1.4%	Impaired and entitled to vote
6	Intercompany Claims	\$1,013,757	N/A	0%	Impaired, deemed to reject the Plan and not entitled to vote
7	Old Equity	N/A	N/A	0%	Impaired, deemed to reject the Plan and not entitled to vote

Based on filed and scheduled proofs of claim as of February 6, 2012.

For purposes of this presentation, the Range of Estimated Allowed Amounts for First Lien Facility Claims includes principal, plus estimated fees incurred by First Lien professionals, plus accrued and unpaid pre- and post-petition interest, all up to the aggregate amount of proceeds the Debtors anticipate will be available for distribution to the First Lien Claimholders under the Plan. Based on these assumptions, activity to date results in a claim today ranging from \$37,900,000 - \$44,000,000 used for the purposes of this analysis; however, the aggregate amount of proceeds available for distribution to the First Lien Claimholders under the Plan may be higher than the Debtors currently anticipate and additional postpetition interest and fees may be owed on account of the First Lien Facility Claims, leading to a higher Range of Estimated Allowed Amounts for the First Lien Facility Claims and a recovery in excess of the Favorable Case amounts set forth in the Liquidation Analysis.

Estimated allowed amount and recovery based on estimated secured portion. Deficiency claim for Second Lien Facility included in Class 5.

Class	Description	Approximate Amount Asserted ⁴	Range of Estimated Allowed Amounts	Range of Estimated Recovery	Status
8	Old Equity Rights	N/A	N/A	0%	Impaired, deemed to reject the Plan and not entitled to vote

B. Voting Instructions

THE DEBTORS STRONGLY RECOMMEND THAT EACH CREDITOR ENTITLED TO VOTE ON THE PLAN VOTE TO ACCEPT THE PLAN.

The Bankruptcy Code entitles only holders of Impaired claims or equity interests who receive some distribution under a proposed plan to vote to accept or reject that plan. Holders of claims or equity interests that are Unimpaired under a proposed plan are conclusively presumed to have accepted that plan and are not entitled to vote on it. Holders of classes of claims or equity interests that will receive no distributions under a proposed plan are conclusively presumed to reject that plan and, therefore, are also not entitled to vote on it.

The holders of Claims in Classes 1 and 4 are not Impaired under the Plan. Thus, pursuant to Section 1126(f) of the Bankruptcy Code, the claimants in Classes 1 and 4 are deemed to have accepted the Plan and are not entitled to vote.

The holders of Claims in Class 2, Class 3 and Class 5 are Impaired and thus may vote to accept or reject the Plan. The Debtors have enclosed Ballots (defined below) with this Disclosure Statement to solicit the votes of all claimants in Class 2, Class 3 and Class 5.

The holders of Claims in Class 6 and Class 8 and holders of Class 7 Interests will receive no distribution under the Plan. Thus, pursuant to Section 1126(g) of the Bankruptcy Code, the holders of Claims in Class 6 and Class 8 and holders of Class 7 Interests are deemed to have rejected the Plan.

A BALLOT FOR ACCEPTANCE OR REJECTION OF THE PLAN IS BEING PROVIDED ONLY TO HOLDERS OF CLAIMS IN CLASS 2, CLASS 3 AND CLASS 5. BEFORE VOTING, SUCH HOLDERS SHOULD READ THIS DISCLOSURE STATEMENT AND ITS EXHIBITS, INCLUDING THE PLAN AND THE PLAN DOCUMENTS, IN THEIR ENTIRETY.

You should use the ballot sent to you with this Disclosure Statement (the "Ballot") to cast your vote for or against the Plan. You may not cast Ballots or votes orally or by facsimile. In order for your Ballot to be considered by the Bankruptcy Court, it must be received at the below address by 4:00 p.m. (prevailing Eastern time) on April 3, 2012. If you are a claimant in Class 2, Class 3 or Class 5, and you did not receive a Ballot with this Disclosure Statement, please contact:

BMC Group, Inc.

Attn: Graceway Pharmaceuticals Solicitation Requests

PO Box 3020

Chanhassen, MN 55317-3020 Telephone: 888-909-0100 www.bmcgroup.com/graceway

Only holders of Allowed Claims in Impaired Classes of Claims are entitled to vote on the Plan. Any Ballot executed by the holder of an Allowed Claim, but which does not indicate acceptance or rejection of the Plan, will not be counted as a vote to accept or reject the Plan. Any Ballot not executed by the holder of an Allowed Claim will not be counted as a vote to accept or reject the Plan.

An Impaired Class of Claims shall have accepted the Plan if (i) the holders (other than any holder designated under Section 1126(e) of the Bankruptcy Code) of at least two-thirds in amount of the Allowed Claims actually voting in such Class have voted to accept the Plan and (ii) the holders (other than any holder designated under Section 1126(e) of the Bankruptcy Code) of more than one-half in number of the Allowed Claims actually voting in such Class have voted to accept the Plan. Whether or not a Creditor or Interest holder votes on the Plan, such Person will be bound by the terms and treatment set forth in the Plan if the Plan is accepted by the requisite majorities of the Classes of Creditors and is confirmed by the Bankruptcy Court. Pursuant to the provisions of Section 1126(e) of the Bankruptcy Code, the Bankruptcy Court may disallow any vote accepting or rejecting the Plan if such vote is not cast in good faith.

If the voting members of an Impaired Class do not vote unanimously for the Plan but, nonetheless, vote for the Plan by at least the requisite two-thirds (2/3) in amount and one-half (1/2) in number of Allowed Claims in that Class actually voted, the Plan, at a minimum, must provide that each member of such Class will receive property of a value, as of the Effective Date, that is not less than the amount such Class members would receive or retain if the Debtors were liquidated under chapter 7 of the Bankruptcy Code.

The Debtors or other parties in interest, including the Creditors' Committee (defined below), may dispute proofs of Claims or Interests that have been filed or that the Debtors listed as disputed in the Schedules the Debtors filed with the Bankruptcy Court. A list of all Claims that the Debtors and the Creditors' Committee have disputed as of the date hereof is attached hereto as Exhibit B. Persons whose Claims are disputed may vote on or otherwise participate in distributions under the Plan only to the extent that the Bankruptcy Court allows their Claims. The Bankruptcy Court may temporarily allow a Claim for voting purposes only. Allowance of a Claim for voting purposes does not necessarily mean that all or a portion of that Claim will be allowed or disallowed for distribution purposes. The Debtors' Schedules listing Claims and whether such Claims are disputed can be inspected at the Delaware offices of Young Conaway at Rodney Square, 1000 North King Street, Wilmington, Delaware 19801, or at the offices of BMC, as set forth above.

C. Confirmation of the Plan by the Bankruptcy Court

Once it is determined which Impaired Classes have or have not accepted the Plan, the Bankruptcy Court will determine whether the Plan may be confirmed. Class 6, Class 7 and Class 8 will receive no distributions on account of their respective Claims and Interests and are therefore deemed to have rejected the Plan. The Bankruptcy Court may confirm the Plan even if all but one of the Impaired Classes do not accept the Plan if the Bankruptcy Court finds that the remaining Impaired Class of Claims (not including any acceptances by "insiders" as defined in Section 101(31) of the Bankruptcy Code) has accepted the Plan and that certain additional conditions are met. The Debtors will therefore request that the Bankruptcy Court confirm the Plan under Section 1129(b) of the Bankruptcy Code with respect to any non-accepting Class of Claims or Interests.

Section 1129(b) of the Bankruptcy Code is generally referred to as the "cramdown" provision. Pursuant to the cramdown provision, the Bankruptcy Court may confirm the Plan over the objection of a non-accepting Class of Secured Claims if the Plan satisfies one of the alternative requirements of Section 1129(b)(2)(A) of the Bankruptcy Code. Likewise, the Bankruptcy Court may confirm the Plan over the objection of a non-accepting Class of General Unsecured Claims if the non-accepting claimants will receive the full value of their Claims, or, if the non-accepting claimants receive less than full value, if no Class of junior priority will receive anything on account of their prepetition Claims or Interests.

If the Plan does not meet the cramdown requirements as set forth above with respect to all of the Debtors, (a) in the Debtors' sole discretion, the Plan may be revoked as to all of the Debtors, or (b) the Debtors may revoke the Plan as to any Debtor not satisfying the cramdown requirements (and any such Debtor's Chapter 11 Case being converted to a chapter 7 liquidation, continued or dismissed in the Debtors' sole discretion after consultation with the First Lien Facility Agent, the Second Lien Facility Agent and the Creditors' Committee) and confirmed as to the remaining Debtors.

THESE ARE COMPLEX STATUTORY PROVISIONS, AND THE PRECEDING PARAGRAPHS ARE NOT INTENDED TO BE A COMPLETE SUMMARY OF THE LAW. IF YOU DO NOT UNDERSTAND THESE PROVISIONS, PLEASE CONSULT WITH YOUR ATTORNEY. BECAUSE CLASSES 6, 7 AND 8 RECEIVE NO DISTRIBUTIONS ON ACCOUNT OF THEIR RESPECTIVE CLAIMS AND INTERESTS AND ARE THUS DEEMED TO HAVE REJECTED THE PLAN, THE DEBTORS INTEND TO RELY UPON THE "CRAMDOWN" PROVISION OF SECTION 1129(b) OF THE BANKRUPTCY CODE.

The Plan provides for the liquidation of substantially all of the property of the Debtors' estates. Pursuant to Section 1141(d)(3) of the Bankruptcy Code, confirmation of the Plan will not discharge the Debtors from any of their debts which arose prior to September 29, 2011; however, Confirmation will make the Plan binding upon the Debtors, their Creditors, holders of Claims and Interests, and other parties in interest regardless of whether they have accepted the Plan.

The Plan is premised on the substantive consolidation of the Debtors with respect to the voting and treatment of all Claims and Interests other than Other Secured Claims, as provided below. The Plan does not contemplate substantive consolidation of the Debtors with respect to Other Secured Claims against the Debtors, which Claims shall apply separately with respect to each Plan proposed by each Debtor. If the Plan cannot be confirmed as to some or all of the Debtors, in the Debtors' sole discretion after consultation with the First Lien Facility Agent, the Second Lien Facility Agent and the Creditors' Committee, (a) the Plan may be revoked as to all of the Debtors, or (b) the Debtors may revoke the Plan as to any Debtor not satisfying the cramdown requirements of Section 1129(b)(7) of the Bankruptcy Code (and any such Debtor's Chapter 11 Case being converted to a chapter 7 liquidation, continued or dismissed in the Debtors' sole discretion after consultation with the First Lien Facility Agent, the Second Lien Facility Agent and the Creditors' Committee) and confirmed as to the remaining Debtors. A list of each Debtor who is a proponent of the Plan and its corresponding Chapter 11 Case docket number is attached to the Plan as Exhibit 1. The Plan shall serve as a request by the Debtors, in lieu of a separate motion, to the Bankruptcy Court that it grant substantive consolidation with respect to the treatment of all Claims and Interests other than Class 4 Claims.

II. BACKGROUND OF THE DEBTORS

A. The Debtors

The Debtors operated as a specialty pharmaceutical company focused on bringing branded prescription products to the market through acquisition, in-licensing or product development primarily in the areas of dermatology, respiratory, and women's health. Revenues were primarily derived from sales to the three largest pharmaceutical wholesalers in the United States. Sales were generated primarily by the direct marketing of the Debtors' approved products to physicians by the Debtors' sales force. The Debtors were founded in 2006 through a partnership of industry veteran Jefferson J. Gregory and GTCR (as defined below), a leader in the private equity industry for more than 25 years.

On January 2, 2007, the Debtors announced that they had completed the acquisition of the pharmaceuticals business of 3M Company (together with its subsidiaries and affiliates, "3M") in North America, Puerto Rico, Mexico and Latin America for approximately \$875,000,000.00 consideration paid in full at closing plus amounts to be paid under a technology access and development agreement. Also, on January 2, 2007, the Debtors announced that Chester Valley Pharmaceuticals, LLC, a dermatology-focused specialty pharmaceutical company and another company sponsored by GTCR, had merged into Graceway Pharmaceuticals, LLC.

Debtor Graceway Pharma Holding Corp. is the parent company of Debtor Graceway Holdings, LLC, and Graceway Holdings, LLC is the parent company of Graceway Pharmaceuticals, LLC. Graceway Pharmaceuticals, LLC was the primary operating subsidiary of the Debtors and owns on hundred percent (100%) of the equity in Debtors Graceway Canada Holdings, Inc., Graceway International, Inc., and Chester Valley Holdings, LLC. Chester Valley Holdings, LLC is the parent company of Chester Valley Pharmaceuticals, LLC. Graceway Canada Holdings, Inc. is the parent company of non-debtor Graceway Canada Company ("Graceway Canada").

As of the Petition Date (defined below), the Debtors' products sold in the United States included, among others: Zyclara® (imiquimod) Cream 3.75% ("Zyclara"), Aldara® (imiquimod) Cream 5% ("Aldara"), Maxair® Autohaler® (pirbuterol acetate inhalation aerosol) ("Maxair"), Atopiclair® Nonsteroidal Cream, and Estrasorb® (estradiol topical emulsion). The Debtors collectively employed approximately 165 full-time employees, 89 of which were sales related employees.

B. Events Leading to Chapter 11

Aldara was the Debtors' flagship or core product at the time of the acquisition from 3M. Zyclara was not approved for marketing by the Food and Drug Administration ("FDA") until March, 2010.

In May of 2007, the Debtors completed a refinancing of their debt structure that produced the credit facilities, described more fully below, held at the time of the Debtors' filing for bankruptcy under Chapter 11. In 2007, the Debtors had total net sales of approximately \$313,580,000.00. Of the total 2007 net sales, the sales of Aldara accounted for 77.02% of all sales, Maxair accounted for 10.39%, and the Debtors' other products accounted for the remaining 12.59%. The Debtors' EBITDA for 2007 was approximately \$189,800,000.00. In 2008, the Debtors had total net sales of approximately \$340,260,000.00. Of the total 2008 net sales, the sales of Aldara accounted for 80.96% of all sales, Maxair accounted for 8.91%, and the Debtors' other products accounted for the remaining 13%. The Debtors' EBITDA for 2008 was approximately \$187,290,000.00. In 2009, the Debtors had total net sales of approximately \$354,610,000.00. Of the total 2009 net sales, the sales of Aldara accounted for 84.95% of all sales, Maxair accounted for 8.04%, and the Debtors' other products accounted for the remaining 7.01%. The Debtors' EBITDA for 2009 was approximately \$214,570,000.00.

Two competitors of the Debtors, Nycomed U.S. Inc. ("Nycomed") and Perrigo Company and Perrigo Israel Pharmaceuticals, Ltd. (collectively "Perrigo"), filed Abbreviated New Drug Applications ("ANDA") seeking FDA approval to market a generic version of Aldara using oleic acid as the solubilizing agent for imiquimod – the active ingredient in Aldara. On February 2, 2010, the U.S. Patent and Trademark Office issued to 3M IPC patent number 7,655,672 (the "672 Patent") covering imiquimod cream formulations that contain certain forms of oleic acid. At the time, there were also 27 other patent applications that were related to the same application that resulted in the '672 Patent. The Debtors held an exclusive license under the '672 Patent and the related applications, including the right to assert, defend, maintain and enforce them. Based on preliminary analyses performed, the Debtors believed that the other companies' products likely infringed the Debtors' '672 Patent.

On February 23, 2010, the Company filed a patent-infringement lawsuit against Nycomed and Perrigo in the U.S. District Court for the District of New Jersey captioned Graceway Pharmaceuticals, LLC and 3M Innovative Properties Company v. Perrigo Company, Perrigo Israel Pharmaceuticals, Ltd. and Nycomed U.S. Inc., Civil Action No. 10-937 (WJM)(MF), in the United States District Court for the District of New Jersey. The lawsuit alleged, among other things, that Nycomed and Perrigo infringed the claims of the '672 Patent. At the same time, the Company requested an opportunity to meet and confer with both

Nycomed and Perrigo in advance of possible FDA approval of their ANDAs. The Company's complaint sought, among other things, judgment that the generic 5% imiquimod topical cream products infringed the '672 Patent, damages, and other relief.

On February 25, 2010, the FDA granted approval to Nycomed's generic version of Aldara, notwithstanding the fact that Nycomed's product used an oleic acid vehicle to deliver imiquimod instead of the isostearic acid vehicle included in Aldara. The practical effect of the FDA's approval was that Nycomed's product was then an approved generic for Aldara and was frequently and automatically substituted by pharmacists when a physician wrote a prescription for Aldara. That same day, Nycomed began shipping its product notwithstanding the lawsuit and the Debtors' pending request to meet and confer.

The Debtors believed Nycomed's product infringed the claims of the '672 Patent, and therefore filed for a temporary restraining order ("TRO") to prevent Nycomed from manufacturing, selling, and shipping its product pending further legal proceedings in the short term. On March 8, 2010, the Court denied the Debtors' motion for a TRO against Nycomed. The Debtors strongly disagreed with the court's position in denying the TRO. The Debtors filed a motion for Preliminary Injunction ("PI") on March 24, 2010 to prevent Nycomed from manufacturing, selling, and shipping its product pending a full trial on the merits. This PI was denied on June 10, 2010. The litigation continues.

In addition to pursuing legal action, the Debtors sought to mitigate the harm to them by entering into an agreement to allow another company to distribute an authorized generic version of Aldara. The Debtors ultimately entered into an agreement with Perrigo. As part of the agreement with Perrigo, the Debtors and Perrigo fully settled all claims in the litigation. Thus the pending patent litigation involves only Nycomed (the "Nycomed Litigation").

In March 2010, the Debtors retained Lazard (defined below) to identify and explore strategic alternatives for them, including, but not limited options for refinancing, restructuring, or selling the Assets of the Debtors.

In 2010, the Debtors had total net sales of approximately \$219,530,000.00, consisting of product sales of \$150,890,000.00 and royalty and licensing revenues of \$68,640,000.00. Of the total net sales, net sales from Aldara accounted for 23.9%, royalty revenues of an authorized generic form of Aldara (marketed by Perrigo) accounted for 17.7%, net sales from Maxair accounted for 12.0%, net sales from Zyclara accounted for 17.8%, licensing revenues for Zyclara accounted for 13.2%, and the Debtors' other products accounted for the remaining 15.4%. The Debtors' EBITDA for 2010 was \$83,330,000.00

During the first two quarters of 2011, the Debtors' net sales were \$67,540,000.00 consisting of \$58,120,000.00 of product sales and \$9,420,000.00 of royalty revenues. The Debtors had no licensing revenues for the first two quarters of 2011. The Debtors' EBITDA for the period from January 1, 2011 through June 30, 2011 was approximately \$14,930,000.00.

The Debtors' capital structure consists of a first lien secured credit agreement (the "First Lien Credit Agreement"), a second lien secured credit agreement (the "Second Lien Credit Agreement") and an unsecured mezzanine credit agreement (the "Mezzanine Credit Agreement"). Despite the Debtors' extensive efforts to reduce operating costs following the adverse rulings in the Debtors' lawsuit against Nycomed, including, but not limited to multiple reductions in workforce, salary reductions to senior management, reduction or elimination of benefits to the workforce, and the like, the Debtors' deteriorating financial condition failed to improve sufficiently and the Debtors were unable to avoid defaults under the Debtors' credit agreements. As a result of such potential defaults, the Debtors commenced these Chapter 11 Cases.

III. THE CHAPTER 11 CASES

A. Overview of the Chapter 11 Cases

On September 27, 2011, the Debtors and Graceway Canada (the "Sellers") entered into a stalking horse asset purchase agreement (the "Stalking Horse Asset Purchase Agreement") with Galderma S.A. as the Stalking Horse Bidder (defined below), pursuant to which the Sellers agreed to sell substantially all of their assets to the Stalking Horse Bidder for \$275,000,000.00 and the Stalking Horse Bidder agreed to assume certain liabilities of the Sellers. The sale was to be conducted under the provisions of Section 363 of the Bankruptcy Code and was subject to proposed bidding procedures and the receipt of a higher and better bid at auction.

On September 29, 2011 (the "Petition Date"), the Debtors filed voluntary petitions under chapter 11 of the Bankruptcy Code. The Debtors in the Chapter 11 Cases are continuing in possession of their respective properties and are operating their respective businesses, as debtors-in-possession, pursuant to Sections 1107 and 1108 of the Bankruptcy Code. The Chapter 11 Cases are pending before the Honorable Peter J. Walsh, Bankruptcy Judge for the United States Bankruptcy Court for the District of Delaware, located at the United States Bankruptcy Court, 824 North Market Street, 6th Floor, Wilmington, Delaware 19801 (the "Bankruptcy Court").

On October 4, 2011, Graceway Canada filed an application in the Ontario Superior Court of Justice (Commercial List) (the "Canadian Court") pursuant to the Courts of Justice Act, R.S.O. 1990, c. C. 43. RSM Richter Inc. was appointed as the receiver of Graceway Canada under the Courts of Justice Act (the "Receiver"). The Canadian Court and the Bankruptcy Court, under separate Orders, approved a cross-border protocol to assist with the holding of joint hearings and coordination of proceedings between the Canadian Court and the Bankruptcy Court. On December 12, 2011, the Canadian Court approved the substitution of Duff & Phelps Canada Restructuring Inc. for RSM Richter Inc. as the acting Receiver.

Following a joint hearing on October 17, 2011, the Canadian Court and the Bankruptcy Court issued Orders approving Galderma S.A. as the stalking horse bidder (the "<u>Stalking Horse Bidder</u>") in the sale of substantially all of the Sellers' assets, authorizing certain protections for the Stalking Horse Bidder, including a break-up fee and expense reimbursement,

and authorizing an auction in the New York offices of Latham & Watkins for the sale of all of the assets of the Sellers.

On November 17, 2011, the Sellers conducted an auction pursuant to the provisions of Section 363 of the Bankruptcy Code at which Medicis Pharmaceutical Corporation ("Medicis" or the "Purchaser") was determined to be the highest and best offer. On November 18, 2011, the Sellers entered into an Asset Purchase Agreement (the "Asset Purchase Agreement") with Medics pursuant to which Medicis agreed to purchase substantially all of the assets of Sellers for \$455,000,000.00 in cash and to assume certain liabilities associated with the purchased Assets (the "Asset Sale"). The Asset Sale was approved by the Bankruptcy Court on November 22, 2011, and the Asset Sale closed on December 2, 2011.

Since the closing of the Asset Sale, the Receiver of non-debtor Graceway Canada has obtained an Order from the Canadian Court approving a claims process and is reviewing such claims to determine the amounts payable to the creditors of Graceway Canada. All claims against Graceway Canada were required to be submitted by the claims bar date of January 31, 2012 (the "Canadian Bar Date"). The Receiver, with the assistance of Graceway Canada, is continuing to review all filed claims and anticipates being in a position to begin making interim distributions within a short time.

Pursuant to the Plan, on the Effective Date, (i) the Debtors' equity interests in Graceway Canada and (ii) the Debtors' Claims against Graceway Canada arising from any intercompany claims, loans, notes, transfers, or other obligations shall be transferred by the Debtors to the Liquidating Trust, and the equity interests in Graceway Canada Company shall continue to be subject to the Liens securing the First Lien Facility Secured Claim and any First Lien Facility Adequate Protection Claim as provided in Section 5.2 of the Plan and the Liens of the Second Lien Facility Agent in favor of the lenders under the Second Lien Credit Agreement (the "Second Lien Facility Lenders") as provided in Section 5.3 of the Plan; provided, however, that the portion of the Canadian Distribution Amount (defined below) used to fund the Estate Fund Amount (defined below) shall be free and clear of the Liens of the First Lien Facility Agent in accordance with the Committee Settlement (as defined below). The Liquidating Trust shall exercise its powers as the sole shareholder to appoint the Liquidating Trustee as the sole officer and director of Graceway Canada and shall make distributions of the Canadian Distribution Amount as provided under the Plan. The Debtors expect to receive in excess of \$10,000,000.00 on account of their claims against, and equity interests in, Graceway Canada. However, because the Canadian claims process is not complete, this estimate could be higher or lower. The actual amount received may be higher or lower than this estimate depending on the total amount of other claims ultimately allowed by the Canadian Court against Graceway Canada.

B. Retained Professionals

The Bankruptcy Court authorized the Debtors to retain certain professionals to represent them and assist them in connection with the Chapter 11 Cases. Specifically, the Debtors have retained, and the Bankruptcy Court has approved the retention of, the following Professionals: (a) Latham & Watkins, as counsel for the Debtors in these Chapter 11 Cases; (b) Young Conaway as co-counsel to the Debtors; (c) Alvarez, as financial advisors to the Debtors; (d) BMC, as official noticing, claims and/or solicitation and balloting agent for the

Debtors; (e) Lazard Frères & Co. LLC ("<u>Lazard</u>"), as investment banker; and (f) PricewaterhouseCoopers LLP, as tax consultant. In addition, the Bankruptcy Court authorized the Debtors to retain, employ, compensate and reimburse the expenses of certain Professionals, primarily attorneys, who have rendered services to the Debtors unrelated to the Chapter 11 Cases to assist with the operation of the Debtors' businesses in the ordinary course.

On October 11, 2011, the Office of the United States Trustee for the District of Delaware (the "<u>U.S. Trustee</u>") appointed an official committee of unsecured Creditors pursuant to Section 1102 of the Bankruptcy Code (the "<u>Creditors' Committee</u>"). The Creditors' Committee retained Lowenstein Sandler, PC, as counsel for the Creditors' Committee, Elliot Greenleaf, as co-counsel for the Creditors' Committee and FTI Consulting, Inc. as its financial advisors. The Bankruptcy Court has approved the Creditors' Committee's retention of all of these professional firms.

Since its formation, the Debtors have consulted with the Creditors' Committee concerning the administration of the Chapter 11 Cases. The Debtors have kept the Creditors' Committee informed about their operations and have sought the concurrence of the Creditors' Committee for actions and transactions taken outside of the ordinary course of the Debtors' business. The Creditors' Committee has participated actively, together with the Debtors' management and Professionals, in connection with negotiation of the Plan. The Creditors' Committee initially consisted of the following members: Value Recovery Fund LLC, Metaphor Inc., and 3M Company. As of the date hereof, the Creditors' Committee consists of Value Recovery Fund LLC, 3M Company, and McKesson Specialty Arizona, Inc.

C. First Day Orders

On the Petition Date, the Debtors filed a number of motions seeking approval of certain so-called "first day orders." The first day orders facilitated the transition between the Debtors' prepetition and postpetition business operations by authorizing the Debtors to continue with certain regular business practices that may not be specifically authorized under the Bankruptcy Code, or for which the Bankruptcy Code requires prior court approval. The first day orders in these Chapter 11 Cases, the majority of which were signed at a hearing held on September 30, 2011, authorized, among other things, the following:

1. Motion for Joint Administration of the Chapter 11 Cases

The Bankruptcy Court authorized the joint administration of the Debtors' Chapter 11 Cases for procedural purposes only.

2. Motion to Continue Using Existing Cash Management Systems

The Bankruptcy Court authorized the Debtors to continue to utilize their existing centralized cash management systems, to continue to use their existing bank accounts and business forms, and to continue employing third-party benefits administrators to prepare and issue payroll and benefits checks on their behalf.

3. Motion for Authority to Pay Prepetition Importing and Shipping Claims

The Bankruptcy Court authorized the Debtors to pay or otherwise honor the prepetition claims in an amount not to exceed \$35,000 on account of prepetition shipping and importing expenses, including customs duties, general order penalties, ocean freight, air freight, trucking charges, brokerage fees, detention and demurrage fees, surety bond premiums, consolidation and deconsolidation charges, and the like.

4. Motion for Authority to Pay Prepetition Employee Wages, Salaries, Benefits, Miscellaneous Employee Expenses and Independent Contractor Fees and Expenses

The Bankruptcy Court initially entered an interim Order and subsequently entered a final Order authorizing the Debtors to pay or otherwise honor the prepetition wages and salaries in the approximate amount of \$840,000, reimbursable employee and executive expenses, independent contractor fees and expenses in the approximate amount of \$100,000, and prepetition benefits obligations, including severance obligations, in the approximate amount of \$2,616,007. The Bankruptcy Court further authorized the Debtors to continue forwarding deductions from employee paychecks to appropriate third parties and to continue paying any and all accrued and unpaid payroll tax obligations.

5. Motion Prohibiting Utilities From Altering, Refusing or Discontinuing Services, Deeming Utility Providers Adequately Assured of Future Performance and Establishing Procedures for Determining Requests for Additional Adequate Assurance

The Bankruptcy Court initially entered an interim Order and subsequently entered a final Order prohibiting the Debtors' utility service providers from altering, refusing or discontinuing services on account of prepetition invoices, deeming the utility providers adequately assured of future performance and establishing procedures for determining requests for additional adequate assurance.

6. Motion For Authority to Pay Taxes and Fees

The Bankruptcy Court authorized the Debtors to (a) pay and remit prepetition sales, use, income, franchise, personal property, real property, business and other taxes in operating their business to the appropriate taxing authorities and (b) authorized the Debtors to pay and remit fees including, but not limited to, permitting fees, license fees, regulatory fees, user fees and other similar charges and assessments, collectively in an aggregate amount not to exceed \$700,000 with respect to all amounts accrued or arising prior to the Petition Date.

7. Motion for Authority to Pay Prepetition Amounts Relating to Customer, Marketing and Medical Affairs Programs and to Continue Such Customer, Marketing and Medical Affairs Programs Postpetition

The Bankruptcy Court authorized the Debtors to honor and perform their obligations with respect to their customer programs in their discretion and in the ordinary course of business, consistent with past practices. The Bankruptcy Court also authorized the Debtors to pay certain prepetition obligations, subject to the aggregate dollar amounts of: (a) \$1,200,000.00 for prepetition obligations under chargeback agreements or pricing and purchase agreements;

(b) \$1,300,000.00 for prepetition obligations under channel management agreements; (c) \$10,000,000.00 for prepetition obligations on account of product returns; and (d) \$2,900,000.00 for obligations under certain "co-pay coupons" administered by a division of McKesson Corporation; provided that the Debtors were authorized to make payments on account of such prepetition obligations only to wholesale drug distributors who executed a "Wholesaler Support Agreement" in substantially the form attached to the motion as Exhibit C. As of the date hereof, the Debtors have not entered into any Wholesaler Support Agreements and have thus not made any payments on account of these prepetition obligations.

D. Cash Collateral and Debtor-In-Possession Financing

The Debtors' obligations to its First Lien Facility Lenders and Second Lien Facility Lenders (collectively, the "Prepetition Secured Lenders") are secured, respectively, by perfected first and second priority liens upon and security interests in substantially all of the assets of the Debtors (the "Prepetition Collateral"). As of the Petition Date, the Prepetition Collateral included substantially all of the Debtors' cash and cash equivalents. As a result, the Debtors did not have any unencumbered cash with which to operate their businesses. Moreover, the Prepetition Secured Lenders were unwilling to lend additional funds to the Debtors in excess of the amounts outstanding under such agreements.

In order to permit the Debtors to fund day-to-day activities pending the sale of the Debtors' assets, the First Lien Facility Agent and certain First Lien Facility Lenders (the "Majority First Lien Facility Lenders") agreed to (a) permit the Debtors to use the cash in which the Prepetition Secured Lenders had a security interest (the "Cash Collateral") and (b) to obtain senior secured debtor-in-possession financing in a principal amount of up to \$6,000,000 from the Debtors' non-debtor subsidiary Graceway Canada. This agreement, as well as, among other things, the support of the First Lien Facility Lenders for the sale of substantially all of the Debtors' Assets, was set forth in that certain Sale Support Agreement, dated as of September 28, 2011, by and among the Debtors, Graceway Canada and the Majority First Lien Facility Lenders, and attached as an exhibit to each of the Debtors' petitions.

On the Petition Date, the Debtors sought the entry of an agreed Order (the "<u>DIP Order</u>") between the Debtors and the First Lien Facility Agent authorizing the Debtors to obtain debtor-in-possession financing and use Cash Collateral pursuant to certain terms and conditions described in the agreed Order, including, but not limited to:

- (i) authorizing the Debtors to enter into certain intercompany term loans (the "<u>Intercompany Notes</u>") and related agreements with Graceway Canada and to perform its obligations thereunder;
- (ii) authorizing the Debtors to borrow and use up to \$6,000,000 in principal amount pursuant to the Intercompany Notes;
- (iii) granting Graceway Canada first priority liens on the Debtors' assets securing the Debtors' obligations under the Intercompany Notes and super-priority administrative claims with respect to such amounts;
 - (iv) granting certain adequate protection to the Prepetition Secured Lenders;

- (v) the Debtors' agreement to certain factual stipulations regarding the First Lien Credit Agreement and the Second Lien Credit Agreement (the "Debtors' Stipulations");
- (vii) the granting by the Debtors of certain releases vis-à-vis the First Lien Facility Agent, the First Lien Facility Lenders, the Second Lien Facility Agent and the Second Lien Facility Lenders;
- (viii) approval of a budget for the use of Cash Collateral, including various covenants relating thereto;
- (ix) the establishment of a "carve-out" for (a) unpaid fees required to be paid to the clerk of the Bankruptcy Court and the office of the U.S. Trustee and (b) certain professional fees and expenses allowed by the Bankruptcy Court under Sections 105(a), 330 and 331 of the Bankruptcy Code (the "Carve-Out");
- (x) the establishment of a period during which the Creditors' Committee could investigate and challenge the Debtors' Stipulations, which were otherwise binding on all parties in interest; and
- (xi) certain other terms and conditions typically found in Orders granting similar relief.

The DIP Order was entered on an interim basis on September 30, 2011, and was approved on a final basis on November 7 pursuant to the Final Order (I) Authorizing Debtors to Utilize Cash Collateral Pursuant to 11 U.S.C. § 363; (II) Granting Adequate Protection to the Prepetition Secured Parties Pursuant to 11 U.S.C. §§ 361, 362, 363, and 364 and (III) Authorizing Debtors to Obtain Postpetition Financing Pursuant to 11 U.S.C. § 364. In connection with the Closing, the Debtors and the First Lien Facility Lenders, negotiated and agreed upon an amended budget (the "Revised Budget") reflecting the revised timeline for the wind down of the Debtors' estates following the Closing Date (defined below). The agreed upon Revised Budget was attached as Exhibit A to that certain Amendment No. 1 dated as of December 2, 2011 and filed with the Court on December 2, 2011. The Revised Budget was subsequently amended pursuant to that certain Amendment No. 2 dated as of January 5, 2012 and filed with the Court on January 6, 2012.

E. Sale of Assets

In March 2010, the Debtors retained Lazard as their investment banker to assist the Debtors in addressing their financial situation and exploring potential financing and restructuring alternatives. After exploring the strategic alternatives available to them, the Debtors, after consulting with Lazard, determined that the best way to maximize the value of their assets for the benefit of their Creditors was to seek a sale of substantially all of their assets pursuant to Section 363 of the Bankruptcy Code

As noted above, on September 27, 2011, the Sellers entered into the Stalking Horse Asset Purchase Agreement with the Stalking Horse Bidder pursuant to which the Sellers agreed to sell substantially all of their assets to the Stalking Horse Bidder and the Stalking Horse Bidder agreed to assume certain liabilities of the Sellers. On the Petition Date the Debtors filed

the Debtors' Motion for Entry of (I) an Order Approving and Authorizing (A) Bidding Procedures in Connection with the Sale of Certain Assets of the Debtors, (B) Stalking Horse Bid Protections, (C) the Form and Manner of Notice of the Sale Hearing, and (d) Related Relief; and (II) an Order Authorizing (A) the Sale of Certain Assets of the Debtors Free and Clear of All Claims, Liens, Liabilities, Rights, Interests and Encumbrances; (B) the Debtors to Enter into and Perform Their Obligations Under the Asset Purchase Agreement; (C) the Debtors to Assume and Assign Certain Executory Contracts and Unexpired Leases; and (D) Granting Related Relief (the "Sale Motion").

On October 17, 2011, the Bankruptcy Court entered the Order Approving and Authorizing (A) Bidding Procedures in Connection with the Sale of Certain Assets of the Debtors, (B) Stalking Horse Bid Protections, (C) the Form and Manner of Notice of the Sale Hearing, and (D) Related Relief (the "Bid Procedures Order"). Pursuant to the Bid Procedures Order, the assets of the Debtors and Graceway Canada were offered for sale at an auction. The auction was scheduled for November 17, 2011 (the "Auction") and November 22, 2009 was set as the date of the sale hearing (the "Sale Hearing").

On November 17, 2011, the Debtors conducted and completed the Auction at which an offer submitted by Medicis was determined to be the highest and best offer. On November 18, 2011, the Debtors and Graceway Canada entered into the Asset Purchase Agreement with Medics pursuant to which Medicis agreed to purchase the Acquired Assets for \$455,000,000.00 in cash (the "Sale Proceeds") and to assume certain liabilities associated with the purchased Assets. The Asset Sale was approved by the Bankruptcy Court at the Sale Hearing on November 22, 2011 pursuant to the Order Authorizing (A) the Sale of Certain Assets of the Debtors Free and Clear of All Claims, Liens, Liabilities, Rights, Interests and Encumbrances; (B) the Debtors to Enter into and Perform Their Obligations Under the Asset Purchase Agreement; (C) the Debtors to Assume and Assign Certain Executory Contracts and Unexpired Leases; and (D) Granting Related Relief (the "Sale Order"). The Asset Sale was consummated on December 2, 2011 (the "Closing Date").

Pursuant to the Sale Order, on the Closing Date the following transactions took place:

- (a) \$15,123,36.14 was funded into a segregated account to pay the Debtors' cure costs under the Asset Purchase Agreement;
- (b) \$4,253,828.76 was distributed directly to the Receiver on account of the value allocated to the assets of Graceway Canada;
- (c) \$6,049,191.78 was distributed to Graceway Canada as payment in full in cash of the Debtors' obligations under the Intercompany Notes;
- (d) \$2,481,250 was funded into a segregated account to pay the transaction fee (the "Success Fee") due and owing to Lazard upon allowance of the Bankruptcy Court;
- (e) \$4,410,193.60 was funded into a segregated account for purposes of funding any shortfall with respect to (X) the establishment of the Carve-Out (less the amount of the Success Fee) and (Y) solely to the extent a written notice by First Lien Facility Agent following

the occurrence and during the continuation of any event of default under the DIP Order, expressly stating that the Carve-Out is invoked, has not been delivered or a plan of reorganization or liquidation in the Bankruptcy Case is confirmed, distributions on account of (I) allowed priority claims and allowed administrative claims (including, without limitation, claims arising under Section 503(b)(9) of the Bankruptcy Code, but excluding liabilities on account of chargebacks, channel management agreements, product returns and rebates) not to exceed in the aggregate \$15,852,000 (excluding any amounts approved under Section 330 of the Bankruptcy Code to the extent of the Carve-Out) and (II) allowed prepetition claims (excluding liabilities on account of chargebacks, channel management agreements, product returns and rebates) authorized to be paid by the Bankruptcy Court pursuant to the Orders approving those certain documents, motions and pleadings, filed by the Debtors with the Bankruptcy Court on the Petition Date, pursuant to which the Debtors sought authorization to pay the prepetition obligations specifically described therein or any motions filed by the Debtors after the Petition Date consented to by the Majority First Lien Facility Lenders that seek authorization to pay the prepetition obligations specifically described therein (the "Prepetition Claims Motions") and not paid as of the Closing Date, not to exceed in the aggregate \$250,000;

- (f) \$4,213,125.55 was funded into a segregated account to be maintained at, and at all times under the control of, the First Lien Facility Agent (the "1.0% Holdback Account");
- (g) \$2,593,647.29 was funded into a segregated account for purposes of funding the Debtors' severance and COBRA liabilities budgeted for under the DIP Order;
- (h) \$17,684,000 was funded into a segregated account for purposes of paying liabilities on account of chargebacks, channel management agreements, product returns and rebates constituting allowed (A) prepetition claims authorized to be paid by the Bankruptcy Court pursuant to the Orders approving the Prepetition Claims Motions or (B) administrative claims, in each case, unpaid as of the Closing Date or arising thereafter;
- (i) \$8,250,000 was distributed to Galderma S.A. on account of the break-up fee and expense reimbursement authorized by the Bid Procedures Order; and
- (j) \$389,941,456.88 was transferred to the First Lien Facility Agent for distribution to the First Lien Facility Lenders.

As set forth in the Sale Order, liens, claims, interests and encumbrances attached to Sale Proceeds in the same order and priority as they existed prior to the completion of the Asset Sale.

F. The Committee Settlement

On October 17, 2011, the Debtors filed the Debtors' Motion for Entry of an Order Determining the Value of the Assets of Graceway Canada Company Proposed to be Purchased Under the Stalking Horse Asset Purchase Agreement and on October 25, 2011 the Debtors filed the Supplement to the Debtors' Motion for Entry of an Order Determining the Value of the Assets of Graceway Canada Company Proposed to be Purchased Under the Stalking Horse Asset Purchase Agreement (collectively, the "Sale Allocation Motion"), pursuant to which the Debtors sought approval of the allocation of \$4,404,683.65 of the Sale Proceeds to the assets owned by

non-Debtor Graceway Canada and allocation of the remainder of the Sale Proceeds to the assets owned by the Debtors. On November 14, 2011, the Creditors' Committee objected to both the Sale Allocation Motion and the Sale Motion pursuant to the (x) Limited Objection of the Official Committee of Unsecured Creditors To Debtors' Motion for Order (I) Approving and Authorizing (A) Bidding Procedures in Connection with the Sale of Certain Assets of the Debtors, (B) Stalking Horse Bid Protections, (C) the Form and Manner of Notice of the Sale Hearing and (D) Related Relief; and (II) Authorizing (A) the Sale of Certain Assets of the Debtors Free and Clear of All Claims, Liens, Liabilities, Rights, Interests and Encumbrances; (B) the Debtors to Enter Into and Perform Their Obligations Under the Asset Purchase Agreement; (C) the Debtors to Assume and Assign Certain Executory Contracts and Unexpired Leases; and (D) Granting Related Relief and (y) Objection of the Official Committee of Unsecured Creditors to Debtors' Motion for Entry of an Order Determining the Value of the Assets of Graceway Canada Company Proposed to be Purchased Under the Stalking Horse Asset Purchase Agreement (together the "Committee Objections").

Prior to the Sale Hearing, the Debtors, the Creditors' Committee and the First Lien Facility Agent negotiated and reached a settlement (the "Committee Settlement") of the Committee Objections. The terms of the Committee Settlement were set forth on the record at the Sale Hearing. Pursuant to the Committee Settlement, the First Lien Facility Agent and certain First Lien Facility Lenders, subject to confirmation of a joint plan of liquidation incorporating terms of such a settlement, released their respective Liens and Claims on:

- (a) an amount equal to \$10,000,000 (the "Estate Fund Amount") to be funded as follows: (a) first, from the first \$10,000,000 of the Cash, if any, distributed by Graceway Canada to the Debtors on account of the Debtors' equity interests in Graceway Canada after payment in full of all creditors of Graceway Canada (the "Canadian Distribution Amount"); (b) second, solely to the extent the Canadian Distribution Amount is less than \$10,000,000, from the 1.0% Holdback Account; and (c) finally, solely to the extent the sum of the Canadian Distribution Amount and the funds in the 1.0% Holdback Account is less than \$10,000,000, from the Sale Proceeds distributed to the First Lien Facility Agent pursuant to the Sale Order;
- (b) \$150,000, which shall be funded into the Trust Administrative Fund (defined below) by the First Lien Facility Agent from Sale Proceeds; and
- (c) 25% of all proceeds, whether obtained through litigations, settlement or otherwise, of any avoidance or equitable subordination or recovery actions under Bankruptcy Code Sections 105, 502(d), 510, 542, 543, 544, 545, 546, 547, 548, 550, 551, and 553.

In exchange for this release, under the Committee Settlement the First Lien Facility Agent (for the benefit of the First Lien Claimholders) received:

- (i) any Canadian Distribution Amount in excess of the Estate Fund Amount;
- (ii) 100% of all proceeds, whether obtained through litigations, settlement or otherwise, of any avoidance or equitable subordination or recovery actions under Bankruptcy Code Section 549; and

(iii) 75% of all proceeds, whether obtained through litigations, settlement or otherwise, of any avoidance or equitable subordination or recovery actions under Bankruptcy Code Sections 105, 502(d), 510, 542, 543, 544, 545, 546, 547, 548, 550, 551, and 553.

The Committee Settlement is the cornerstone of the Plan. To the extent the Committee Settlement is not already deemed approved by the Sale Order, pursuant to Section 363 of the Bankruptcy Code and Bankruptcy Rule 9019 and in consideration for the distributions and other benefits provided pursuant to the Plan, the provisions of the Plan relating to and effectuating the Committee Settlement will constitute a good faith compromise. The entry of the Confirmation Order shall constitute the Bankruptcy Court's approval of the Committee Settlement, as well as a finding by the Bankruptcy Court that the Committee Settlement is in the best interests of the Debtors, their Estates and holders of Claims and Interests and is fair, equitable and reasonable.

G. The Debtors' Current Management

In connection with the consummation of the Asset Sale, substantially all of the Debtors' employees were terminated on the Closing Date. Certain of the Debtors' employees, including certain of the Debtors' officers, continued to perform in limited roles following the Closing Date. After Closing, George Sperzel and John Bellamy joined Kenneth Dunn and Xavier Yon as directors of Graceway Pharma Holding Corp., Graceway Holdings, LLC, Graceway Pharmaceuticals, LLC, Chester Valley Holdings, LLC, and Chester Valley Pharmaceuticals, LLC. All other directors of the Debtors resigned, namely Jefferson Gregory, Dean Mihas, Ned Jannotta, Ed Fiorentino and Bob Moccia. John Bellamy is acting as the sole director of Graceway International, Inc. and Graceway Canada Holdings, Inc. For all Debtor and non-Debtor entities, John Bellamy is serving as the President and Tariq Zaidi is serving as the Secretary. All of the Debtors' remaining employees will be terminated upon the Effective Date.

H. The Claims Process

The Bankruptcy Code provides a procedure for each person who believes he has a claim against a debtor to assert such claim, so that such claimant can receive distributions from the debtor's bankruptcy estate. The bankruptcy court establishes a "bar date" – a date by which Creditors must file their claims, or else such claims will not participate in the bankruptcy case or any distribution. After the filing of all claims, the debtor evaluates such claims and can, along with other parties in interest, raise objections to them. These claims objections allow the debtor to minimize claims against it, and thereby maximize the recovery to creditors with Allowed Claims.

By Order dated October 17, 2011, the Bankruptcy Court established December 30, 2011 (the "Bar Date") as the deadline for filing proofs of Claims relating to prepetition Claims against the Debtors, other than Claims of Governmental Units and Administrative Expense Claims. The Bankruptcy Court established March 27, 2012 as the date as the deadline for filing proofs of Claims by Governmental Units relating to prepetition Claims against the Debtors (the "Governmental Claims Bar Date").

Pursuant to the Plan, all unpaid Administrative Expense Claims must be filed on or before the Administrative Expense Claims Bar Date. The Administrative Expense Claims Bar Date is the first Business Day that is at least thirty-five (35) days following the Effective Date.

The Debtors have been reviewing, analyzing and resolving Claims on an ongoing basis as part of the claims reconciliation process. To date approximately 206 proofs of Claim have been asserted in the Chapter 11 Cases. Over the last several months, the Debtors have only preliminarily begun to reconcile the amount and classification of outstanding claims and to assert and prosecute objections to claims. The Debtors have also identified Claims for future resolution, as well as other existing or potential claims disputes. Nonetheless, a significant number of Claims have not yet been resolved to date, and the actual ultimate aggregate amount of Allowed Claims may differ significantly from the amounts used for the purposes of the Debtors' estimates. Additionally, the Debtors are in the process of reviewing their Schedules filed with the Bankruptcy Court and, upon the completion of such review, will likely amend such Schedules to account for certain Claims that have been satisfied or settled. Accordingly, the amount of the Pro Rata share that will ultimately be received by any particular holder of an Allowed Claim may be adversely affected by the outcome of the claims resolution process

I. Substantive Consolidation of the Debtors

As set forth in more detail in Section IV, the Plan provides for the substantive consolidation of the Debtors with respect to the voting and treatment of all Claims and Interests except Other Secured Claims. Section 105(a) of the Bankruptcy Code empowers a bankruptcy court to authorize substantive consolidation. The United States Court of Appeals for the Third Circuit has adopted a standard for granting a request for substantive consolidation similar to the standards adopted by other Circuits authorizing substantive consolidation. See In re Owens Corning, 419 F.3d 195, 211 (3d Cir. 2005); Reider v. F.D.I.C. (In re Reider), 31 F.3d 1102, 1107-1108 (11th Cir. 1994); Woburn Assoc. v. Kahn (In re Hemingway Transport Inc.), 954 F.2d 1, 11-12 (1st Cir. 1992); First Nat'l Bank of El Dorado v. Giller (In re Giller), 962 F.2d 796, 798-99 (8th Cir. 1992); Union Sav. Bank. v. Augie/Restivo Baking Co. (In re Augie/Restivo Baking Co.) 860 F.2d 515, 518 (2d Cir. 1988); Drabkin v. Midland-Ross Corp. (In re Auto-Train Corp.), 810 F.2d 270, 276 (D.C. Cir. 1987).

In the Third Circuit, debtors seeking substantive consolidation must show either "(i) prepetition they disregarded separateness so significantly their creditors relied on the breakdown of entity borders and treated them as one legal entity, or (ii) postpetition their assets and liabilities are so scrambled that separating them is prohibitive and hurts all creditors." Owens Corning, 419 F.3d at 211 (emphasis added). "A prima facie case for [the first rationale] typically exists when, based on the parties' prepetition dealings, a proponent proves corporate disregard creating contractual expectations of creditors that they were dealing with debtors as one indistinguishable entity." Id. at 212. The facts of these Chapter 11 Cases necessitate substantive consolidation, and substantive consolidation is warranted under the aforementioned tests.

There is ample evidence that prior to the Petition Date Creditors of the Debtors treated them as one legal entity. The Debtors prepared and disseminated consolidated financial reports to the public, including customers, suppliers, landlords, lenders, and credit rating

agencies. Because the Debtors disseminated financial information to the public on a consolidated basis, it is highly unlikely that Creditors relied on the separate identity of any Debtor in extending credit to such Debtor.

Because financial information disseminated to customers, suppliers, landlords, lenders, and credit rating agencies, has been prepared and presented on a consolidated basis, it is clear that creditors treated the Debtors as one legal entity when deciding whether to extend credit. Substantive consolidation would ensure that all of the Debtors' creditors, having relied on the creditworthiness of the Debtors as a unit, receive the benefit of a distribution in satisfaction of their claims from the single pool of assets. Thus, substantive consolidation is warranted in this case under the test set forth by the Third Circuit.

Several courts within the Third Circuit have acknowledged the existence and application of substantive consolidation of separate bankruptcy estates in appropriate circumstances. See In re Molnar Bros., 200 B.R. 555 (Bankr. D.N.J. 1996) (recognizing the application of substantive consolidation of two or more bankruptcy estates); In re PWS Holding Corp., Case No. 98-212-223 (SLR) (D. Del. 1998) (approving substantive consolidation of debtors pursuant to a plan of reorganization); Bracaglia v. Manzo (In re United Stairs Corp.), 176 B.R. 359, 368 (Bankr. D.N.J. 1995) (stating that it is "well established that in the appropriate circumstances the court may substantively consolidate corporate entities"); In re Buckhead Am. Corp., 1992 Bankr. LEXIS 2506 (Bankr. D. Del. Aug. 13, 1992) (substantively consolidating debtors); In re Cooper, 147 B.R. 678, 682 (Bankr. D.N.J. 1992).

For the reasons set forth above, the Debtors believe that the requirements for substantive consolidation of the Debtors with respect to voting and treatment of all Claims and Interests except for Class 4 Claims are satisfied.

The Plan and Disclosure Statement do not seek substantive consolidation with respect to Other Secured Claims. Thus, any pledge by any of the Debtors of any of its respective Assets, including, without limitation, any stock in any of its subsidiaries, is unaffected by the limited substantive consolidation proposed in the Plan.

IV. CHAPTER 11 PLAN

THE FOLLOWING IS A BRIEF SUMMARY OF CERTAIN OF THE MORE SIGNIFICANT MATTERS CONTEMPLATED BY OR IN CONNECTION WITH THE CONFIRMATION OF THE PLAN. THUS, THE FOLLOWING SUMMARY IS QUALIFIED IN ITS ENTIRETY BY THE PLAN, WHICH IS ATTACHED TO THIS DISCLOSURE STATEMENT AS EXHIBIT A. THIS SUMMARY ONLY HIGHLIGHTS CERTAIN SUBSTANTIVE PROVISIONS OF THE PLAN. CONSIDERATION OF THIS SUMMARY WILL NOT, NOR IS IT INTENDED TO, YIELD A THOROUGH UNDERSTANDING OF THE PLAN. SUCH CONSIDERATION IS NOT A SUBSTITUTE FOR A FULL AND COMPLETE READING OF THE PLAN. ALL HOLDERS OF CLAIMS AND INTERESTS ARE URGED TO REVIEW THE PLAN CAREFULLY. THE PLAN, IF CONFIRMED, WILL BE BINDING ON EACH OF THE DEBTORS AND ALL HOLDERS OF CLAIMS AND INTERESTS.

A. Plan Overview

The Plan contains four types of unclassified Claims: Administrative Expense Claims, Professional Claims, Statutory Fees and Priority Tax Claims. In addition, the Plan classifies Claims and Interests as follows: Class 1 Other Priority Claims, Class 2 First Lien Facility Claims, Class 3 Second Lien Facility Claims, Class 4 Other Secured Claims, Class 5 General Unsecured Claims, Class 6 Intercompany Claims, Class 7 Old Equity and Class 8 Old Equity Rights. Classes 1 and 4 are Unimpaired and Classes 2, 3, 5, 6, 7 and 8 are Impaired.

B. Plan Summary

The Plan is premised on the substantive consolidation of all of the Debtors with respect to the treatment of all Claims and Interests, except for the Other Secured Claims in Class 4. The Plan does not contemplate substantive consolidation of the Debtors with respect to the Class 4 Claims, which shall be deemed to apply separately with respect to the Plan proposed by each Debtor. The Plan shall serve as a request by the Debtors, in lieu of a separate motion, to the Bankruptcy Court that it grant substantive consolidation with respect to the treatment of all Claims and Interests: on the Effective Date, (a) all Class 6 Intercompany Claims will be eliminated (except to the extent such Claims are by a Debtor against a non-Debtor Affiliate); (b) no distributions shall be made under the Plan on account of any equity interest held by a Debtor in any other Debtor; (c) all Assets and liabilities of the Debtors will be merged or treated as though they were merged; (d) all guarantees of the Debtors of the obligations of any other Debtor and any joint or several liability of any of the Debtors shall be eliminated; and (e) each and every Claim or Interest against any Debtor shall be deemed Filed against the consolidated Debtors and all Claims Filed against more than one Debtor for the same liability shall be deemed one Claim against any obligation of the consolidated Debtors. In the event the Bankruptcy Court determines that such substantive consolidation of the Debtors is not appropriate, the Debtors may request that the Bankruptcy Court otherwise confirm the Plan and the treatment of the different Classes under the Plan on a Debtor by Debtor basis.

C. Unclassified and/or Unimpaired Claims

The Unclassified and/or Unimpaired Claims consist of Administrative Expense Claims, Professional Claims, Statutory Fees and Priority Tax Claims. These Claims shall be treated as follows:

Administrative Expense Claims. Subject to the allowance procedures and deadlines provided in the Plan, on the Effective Date or as soon thereafter as is practicable in recognition of the applicable Claims resolution process set forth in the Plan, the holder of an Allowed Administrative Expense Claim (other than an Administrative Expense Claim that is an Assumed Liability or a Professional Claim) shall receive on account of such Allowed Administrative Expense Claim and in full satisfaction, settlement and release of and in exchange for such Allowed Administrative Expense Claim, (a) Cash equal to the unpaid portion of such Allowed Administrative Expense Claim, or (b) such other treatment as to which the Debtors, or, if after the Effective Date, the Liquidating Trustee, and the holder of such Allowed Administrative Expense Claim have agreed upon in writing provided such treatment is on more favorable terms to the Debtors or the Liquidating Trustee, as the case may be, than the treatment

set forth in subsection (a) hereof; provided, however, that (i) notwithstanding any other provision of the Plan, Administrative Expense Claims that are Assumed Liabilities have been Assumed and shall be satisfied in accordance with the Asset Purchase Agreement and shall not receive distributions pursuant to the Plan and (ii) Professional Claims shall be paid in accordance with Section 2.3 of the Plan.

Statutory Fees. On or before the Effective Date, all fees due and payable pursuant to 28 U.S.C. § 1930, as determined by the Bankruptcy Court at the Confirmation Hearing, shall be paid in full, in Cash.

Professional Claims. Immediately prior to the Effective Date, the Debtors shall pay from the Designated Account all amounts owing to the Professionals other than Lazard and from the Lazard Account all amounts owing to Lazard for all outstanding Professional Claims relating to prior periods and for the period ending on the Effective Date; provided, however, that the amount paid to each Professional immediately prior to the Effective Date shall not exceed the lesser of (a) the excess, if any, of (i) the amount provided for in the Budget for such Professional over (ii) the sum of the aggregate amount of all payments already made to such Professional (including the application of any retainer held by such Professional) for services rendered since the Petition Date and the amount of any retainers currently held by such Professional and (b) the sum of (i) the amount billed and unpaid for such Professional and (ii) the amount estimated by such Professional as necessary to cover unbilled periods through the Effective Date, which estimated amount shall be provided to the Debtors and the First Lien Facility Agent by each Professional on or prior to the Effective Date. The Professionals shall estimate Professional Claims due for periods that have not been billed as of the Effective Date. On or prior to the Administrative Expense Claims Bar Date, each Professional shall File with the Bankruptcy Court its final fee application seeking final approval of all fees and expenses from the Petition Date through the Effective Date. Within ten (10) days after entry of a Final Order with respect to its final fee application, each Professional shall remit any overpayment to the Liquidating Trustee to be deposited in the Designated Account or the Liquidating Trustee shall pay any outstanding amounts owed to the Professional, but in no event shall any Professional receive more than the amount provided in the Budget for such Professional from the Designated Account. Payment of Allowed Professional Claims asserted by any Professional shall be funded from Other Assets (other than the Liquidating Trustee Fee Amount) to the extent that the aggregate amount of such Allowed Professional Claims of such Professional since the Petition Date exceeds the amount provided for in the Budget for such Professional.

Priority Tax Claims. Subject to the allowance procedures and deadlines provided in the Plan, on the Effective Date or as soon thereafter as is practicable in recognition of the applicable Claims resolution process set forth in the Plan, with respect to each Allowed Priority Tax Claim, at the sole option of the Debtors or, if after the Effective Date, the Liquidating Trustee, the holder of an Allowed Priority Tax Claim shall be entitled to receive on account of such Allowed Priority Tax Claim, in full satisfaction, settlement and release of and in exchange for such Allowed Priority Tax Claim, (i) Cash in an amount equal to the unpaid portions of such Allowed Priority Tax Claims; or (ii) such other treatment agreed to in writing by the holder of such Allowed Priority Tax Claim and the Debtors or, if after the Effective Date, the Liquidating Trustee, provided such treatment is on more favorable terms to the Debtors or the Liquidating Trustee, as the case may be, than the treatment set forth in subsection (i) hereof.

In accordance with the Bankruptcy Code, Administrative Expense Claims, Professional Claims, Statutory Fees and Priority Tax Claims are not classified. Therefore, the claimants holding the aforementioned Claims may not vote on the Plan.

D. Treatment of Claims and Interests

The treatment of and consideration to be received by holders of Allowed Claims and the treatment of Interests pursuant to Article 4 and 5 of the Plan will be in full satisfaction, settlement, release, extinguishment and discharge of their respective Allowed Claims against, or Interests in, the Debtors and the Debtors' Estates, except as otherwise expressly provided in the Plan or the Confirmation Order.

1. Class 1 – Other Priority Claims

a. Definition of Class 1 – Other Priority Claims

Other Priority Claims include any Claim, other than an Administrative Expense Claim or a Priority Tax Claim, of a Creditor to the extent such Claim is entitled to priority pursuant to Bankruptcy Code Section 507(a).

b. Treatment of Class 1 – Other Priority Claims

On the Effective Date or as soon thereafter as is reasonably practicable in recognition of the applicable Claims reconciliation process set forth in the Plan, the holder of an Allowed Other Priority Claim shall receive on account of the Allowed Other Priority Claim and in full satisfaction, settlement and release of and in exchange for such Allowed Other Priority Claim, (a) Cash equal to the amount of such Allowed Other Priority Claim, or (b) such other treatment as to which the Debtors and the holder of such Allowed Other Priority Claim have agreed upon in writing, provided such treatment is on more favorable terms to the Debtors or the Liquidating Trustee, as the case may be, than the treatment set forth in subsection (a) hereof.

c. Voting Status of Class 1 – Other Priority Claims

Class 1 is not Impaired and is deemed to have accepted the Plan pursuant to Section 1126(f) of the Bankruptcy Code. Thus, the claimants in Class 1 may not vote on the Plan.

2. Class 2 – First Lien Facility Claims

a. Definition of Class 2 – First Lien Facility Claims

The First Lien Facility Claims include the First Lien Facility Secured Claim, any First Lien Facility Deficiency Claim and any First Lien Facility Adequate Protection Claim.

b. Treatment of Class 2 – First Lien Facility Claims

On the Effective Date or as soon thereafter as is reasonably practicable in recognition of the applicable Claims reconciliation process set forth in the Plan, the First Lien

Facility Agent shall receive, for the benefit of the First Lien Claimholders, in full satisfaction, settlement and release of and in exchange for the First Lien Facility Claims (including the First Lien Facility Secured Claim, any First Lien Facility Adequate Protection Claim and any First Lien Facility Deficiency Claim), (i) any First Lien Available Cash that is available to be distributed on the Effective Date; (ii) the Nycomed Litigation Proceeds; and (iii) beneficial interests in the Liquidating Trust entitling each First Lien Claimholder to receive its Pro Rata share of (A) any First Lien Available Cash that becomes available to be distributed after the Effective Date, (B) seventy-five percent (75%) of any Avoidance Action Proceeds, (C) one hundred percent (100%) of any other Net Causes of Action Proceeds, including, without limitation, 549 Avoidance Action Proceeds, and (D) all Cash from the collection of accounts receivable. Until the First Lien Facility Claims are paid in full in Cash, the Liens securing the First Lien Facility Secured Claim and any First Lien Facility Adequate Protection Claim shall remain on the Assets (including, without limitation, the Reserve Accounts and the Cash deposited therein and the Nycomed Litigation Proceeds) other than the Other Assets, provided that, except for their obligations under the Plan, the Debtors and their Estates shall have no further liability to the First Lien Facility Agent or the First Lien Facility Lenders. Notwithstanding the full satisfaction, settlement and release of the First Lien Facility Claims under the Plan, the aggregate amount of the beneficial interests in the Liquidating Trust the First Lien Claimholders receive under Section 5.2 of the Plan in exchange for such Claims shall increase from time to time by an amount equal to the interest accrual on, and the fees incurred on account of, such Claims to and including the applicable date of determination (calculated at the rate provided for under the First Lien Credit Agreement and in accordance with the terms thereof and of the Intercreditor Agreement) as if such Claims had not been satisfied, settled and released under the Plan; it being understood that distributions on account of such interest accrual and incurred fees shall only be permitted in the event the First Lien Facility Secured Claim (had it not been satisfied, settled and released under the Plan) is oversecured. Holders of the First Lien Facility Claims shall be Beneficiaries; it being understood that distributions from the Liquidating Trust shall be exclusively governed in accordance with the terms set forth in the Plan. Class 2 is Impaired and may vote to accept or reject the Plan.

c. Voting Status of Class 2 – First Lien Facility Claims

Class 2 is Impaired and may vote on the Plan.

3. Class 3 – Second Lien Facility Claims

a. Definition of Class 3 – Second Lien Facility Claims

Second Lien Facility Claims include any Second Lien Facility Secured Claim and any Second Lien Facility Adequate Protection Claim.

b. Treatment of Class 3 – Second Lien Facility Claims

The Second Lien Facility Agent shall receive, for the benefit of the Second Lien Facility Lenders, in full satisfaction, settlement and release of and in exchange for any Allowed Second Lien Facility Secured Claim and any Allowed Second Lien Facility Adequate Protection Claim on the Effective Date or as soon thereafter as is practicable in recognition of the applicable

Claims reconciliation process set forth in the Plan, beneficial interests in the Liquidating Trust entitling each Second Lien Facility Lender to receive its Pro Rata share of (a) any Other Available Cash subject to the Liens granted to the Second Lien Facility Agent pursuant to the Second Lien Credit Agreement or the DIP Order and (b) after payment in full in Cash of the First Lien Facility Claims, (i) any First Lien Available Cash; (ii) the Nycomed Litigation Proceeds, (iii) seventy-five percent (75%) of any Avoidance Action Proceeds; (iv) one hundred percent (100%) of any other Net Causes of Action Proceeds, including, without limitation, any 549 Avoidance Action Proceeds and (v) all Cash from the collection of accounts receivable. Until the Second Lien Facility Claims are paid in full in Cash, the Liens securing any Second Lien Facility Secured Claim and any Second Lien Facility Adequate Protection Claim shall remain on the Assets, provided that, (a) except for their obligations under the Plan, the Debtors and their Estates shall have no further liability to the Second Lien Facility Agent or the Second Lien Facility Lenders and (b) such Liens shall continue to be junior and subordinate in all respects to the Liens securing the First Lien Facility Secured Claim and First Lien Adequate Protection Claim that have not been released in connection with the Committee Settlement until such Claims are paid in full in Cash. Any Second Lien Facility Deficiency Claim shall not constitute a Class 3 Second Lien Facility Claim and shall be treated as a Class 5 General Unsecured Claim under the Plan. Holders of the Allowed Second Lien Facility Claims shall be Beneficiaries; it being understood that distributions from the Liquidating Trust shall be exclusively governed in accordance with the terms set forth in the Plan. Class 3 is Impaired and may vote to accept or reject the Plan.

c. Voting Status of Class 3 – Second Lien Facility Claims

Class 3 is Impaired and may vote on the Plan.

4. Class 4 – Other Secured Claims

a. Definition of Class 4 – Other Secured Claims

Other Secured Claims are all Secured Claims against the Debtors other than the First Lien Facility Secured Claim and the Second Lien Facility Secured Claim; provided, however, that any Secured Claim that is an Assumed Liability has been assumed and shall be satisfied in accordance with the Asset Purchase Agreement and shall not receive distributions pursuant to the Plan.

This Other Secured Claims Class is subdivided into subclasses designated by letters of the alphabet (Class 4A, Class 4B and so on), so that each holder of any Other Secured Claim is in a Class by itself, except to the extent that there are Other Secured Claims that are substantially similar to each other and may be included within a single Class, and except for a precautionary class of otherwise unclassified Other Secured Claims.

b. Treatment of Class 4 – Other Secured Claims

On the Effective Date or as soon thereafter as is practicable in recognition of the applicable Claims reconciliation process set forth in the Plan, each holder of an Allowed Other Secured Claim that was not assumed by the Purchaser in connection with the 363 Sale shall receive, in full satisfaction, settlement and release of and in exchange for such Allowed Other

Secured Claim, (i) the collateral securing any such Allowed Other Secured Claim (to the extent such collateral does not constitute collateral securing the First Lien Facility Secured Claim, any First Lien Facility Adequate Protection Claim, the Second Lien Facility Secured Claim or any Second Lien Facility Adequate Protection Claim or, if such collateral does secure any such Claim, to the extent that the Lien securing such Allowed Other Secured Claim is senior to the Lien securing the First Lien Facility Secured Claim, any First Lien Facility Adequate Protection Claim, the Second Lien Facility Secured Claim or any Second Lien Facility Adequate Protection Claim, as applicable), (ii) Cash in an amount equal to such Allowed Other Secured Claim to the extent the collateral securing such Allowed Other Secured Claim was sold and such collateral does not constitute collateral securing the First Lien Facility Secured Claim, any First Lien Facility Adequate Protection Claim, the Second Lien Facility Secured Claim or any Second Lien Facility Adequate Protection Claim or, if such collateral does secure any such Claim, to the extent that the Lien securing such Allowed Other Secured Claim is senior to the Lien securing the First Lien Facility Secured Claim, any First Lien Facility Adequate Protection Claim, the Second Lien Facility Secured Claim or any Second Lien Facility Adequate Protection Claim, as applicable, or (iii) such other treatment that leaves such Allowed Other Secured Claim Unimpaired pursuant to the Section 1124(2) of the Bankruptcy Code. The Other Secured Claim Liens shall be released and the Debtors and their Estates shall have no further liability therefor; provided, however, that any Deficiency Claims of holders of Class 4 Other Secured Claims shall not constitute Class 4 Other Secured Claims and shall be treated as Class 5 General Unsecured Claims under the Plan.

c. Voting Status of Class 4 – Other Secured Claims

Class 4 is not Impaired and is deemed to have accepted the Plan pursuant to Section 1126(f) of the Bankruptcy Code. Thus, the claimants in Class 4 may not vote on the Plan.

5. Class 5 – General Unsecured Claims

a. Definition of Class 5 – General Unsecured Claims

General Unsecured Claims are means all Unsecured Claims against the Debtors, including, without limitation, the Mezzanine Facility Claim and any Second Lien Facility Deficiency Claim, but excluding any First Lien Facility Deficiency Claim, the Intercompany Claims and the Old Equity Rights.

b. Treatment of Class 5 – General Unsecured Claims

On the Effective Date or as soon thereafter as is practicable in recognition of the applicable Claims reconciliation process set forth in the Plan, the holders of Allowed General Unsecured Claims shall receive, in full satisfaction, settlement and release of and in exchange for such Allowed Claims, their Pro Rata share of the Other Available Cash that is not subject to the Liens granted to the Second Lien Facility Agent pursuant to the Second Lien Credit Agreement or the DIP Order, after payment in full of all Allowed Other Secured Claims (to the extent not otherwise satisfied in accordance with the terms of the Plan), Allowed Administrative Expense Claims (other than Administrative Expense Claims that are Assumed Liabilities), Allowed

Priority Tax Claims and Allowed Other Priority Claims; provided, however, that pursuant to and in accordance with the Mezzanine Credit Agreement, any amounts distributed on account of the Allowed Mezzanine Facility Claim (a) are subject and subordinate to payment in full in Cash of the Second Lien Obligations and (b) shall be distributed directly to the Second Lien Facility Agent for the benefit of the holders of Allowed Second Lien Facility Claims until the Second Lien Obligations are paid in full in Cash; provided further, however, that upon notice to the Liquidating Trustee from the Second Lien Facility Agent that the Second Lien Facility Agent has received \$10,000,000.00 in Cash from the Other Available Cash for distribution to holders of Allowed Second Lien Facility Claims in excess of any legal, trustee or other fees and expenses that constitute Second Lien Obligations, the Liquidating Trustee shall cease any distributions to the Second Lien Facility Agent solely on account of the Allowed Mezzanine Facility Claim until the Liquidating Trustee has distributed \$250,000.00 on account of the Allowed Mezzanine Facility Claim (which shall be paid out as if no subordination were in place), after which time the Liquidating Trustee shall resume any distributions to the Second Lien Facility Agent on account of the Allowed Mezzanine Facility Claim until the Second Lien Obligations are paid in full in Cash. Subject to the foregoing proviso, in the event any distribution on account of the Allowed Mezzanine Facility Claim is made other than to the Second Lien Facility Agent, such distribution shall be held in trust for the benefit of, and paid over and delivered to, the Second Lien Facility Agent for the benefit of the Second Lien Facility Lenders. Holders of Allowed General Unsecured Claims shall be Beneficiaries; it being understood that distributions from the Liquidating Trust shall be exclusively governed in accordance with the terms set forth in the Plan and beneficial interests in the Liquidating Trust granted to such holders will not entitle them to any distribution of more than twenty-five percent (25%) of any Avoidance Action Proceeds or any distribution of any other Causes of Action Proceeds, including, without limitation, any 549 Avoidance Action Proceeds. In addition, the holders of Allowed General Unsecured Claims shall be entitled to share Pro Rata in any amount remaining in or released from the Disputed Claims Reserve (defined below) after resolution of all Disputed Unsecured Claims.

c. Voting Status of Class 5 – General Unsecured Claims

Class 5 is Impaired and may vote on the Plan.

6. Class 6 – Intercompany Claims

a. Definition of Class 6– Intercompany Claims

Intercompany Claims are the Claims of a Debtor against any other Debtor, and shall exclude any Claims of a Debtor against a non-Debtor Affiliate.

b. Treatment of Class 6– Intercompany Claims

As a result of substantive consolidation of the Debtors for distribution purposes under the Plan as provided in Section 7.1 of the Plan, holders of Intercompany Claims will not receive any distribution of property under the Plan on account of their Intercompany Claims and, on the Effective Date, the Intercompany Claims will be cancelled; *provided*, *however*, Class 6 shall exclude any Claims of a Debtor against a non-Debtor Affiliate or a non-Debtor subsidiary.

c. Voting Status of Class 6 – Intercompany Claims

Class 6 is Impaired. Because the holders of Claims in Class 6 will receive no distribution under the Plan, they are deemed to have rejected the Plan pursuant to Section 1126(g) of the Bankruptcy Code. Consequently, holders of Claims in Class 6 may not vote on the Plan.

7. Class 7 – Old Equity

a. Definition of Class 7 – Old Equity

Old Equity means any outstanding or retired ownership Interests in any of the Debtors, including Interests evidenced by stock, membership Interests or their equivalents, but excluding the Old Equity Rights

b. Treatment of Class 7 – Old Equity

Holders of Old Equity Interests will not receive and will not retain any property of the Debtors under the Plan on account of such Interests and all Old Equity Interests will be cancelled as of the Effective Date.

c. Voting Status of Class 7 – Old Equity

Class 7 is Impaired. Because the holders of Interests in Class 7 will receive no distribution under the Plan, they are deemed to have rejected the Plan pursuant to Section 1126(g) of the Bankruptcy Code. Consequently, holders of Interests in Class 7 may not vote on the Plan.

8. Class 8 – Old Equity Rights

a. Definition of Class 8 – Old Equity Rights

Old Equity Rights are any calls, rights, puts, awards, commitments, repurchase rights, unvested or unexercised options, warrants, unvested common, preferred or other Interests or any other agreement of any character related to Old Equity, obligating any of the Debtors to issue, transfer, purchase, redeem, or sell any common, preferred or other Interests or securities, any rights under any equity incentive plans, voting agreements, investor agreements and registration rights agreements regarding common, preferred or other Interests or equity securities of any of the Debtors, any Claims arising from the rescission of a purchase, sale or other acquisition of any outstanding common, preferred, membership or other Interests or equity securities (or any right, Claim, or Interest in and to any common, preferred, membership or other interests or equity securities) of any of the Debtors, any Claims for the payment of any distributions with respect to any common, preferred, membership or other Interests or equity securities of any of the Debtors, and any Claims for damages or any other relief arising from the purchase, sale, or other acquisition of the outstanding common, preferred, membership or other Interests or equity securities of any of the Debtors.

b. Treatment of Class 8 – Old Equity Rights

Holders of any Claim or Interest arising from or relating to Old Equity Rights will not receive and will not retain any property of the Debtors under the Plan on account of such Claim or Interest and all Old Equity Rights and any Interests or Claims arising from or relating thereto will be cancelled as of the Effective Date.

c. Voting Status of Class 8 – Old Equity Rights

Class 8 is Impaired. Because the holders of Interests in Class 8 will receive no distribution under the Plan, they are deemed to have rejected the Plan pursuant to Section 1126(g) of the Bankruptcy Code. Consequently, holders of Interests in Class 8 may not vote on the Plan.

E. Post-Confirmation Operations of the Debtors

Pursuant to Section 7.2 of the Plan and the Confirmation Order, and effective as of the Effective Date of the Plan, the Beneficiaries and the Debtors, under the Liquidating Trust Agreement, will create the Liquidating Trust, to be formally known as the "Graceway Liquidating Trust," for the benefit of the Beneficiaries. A summary of certain significant terms of the Liquidating Trust Agreement are set forth below. The Liquidating Trust Agreement is attached to the Plan as Exhibit 2 and the Plan is attached hereto as Exhibit A.

The Liquidating Trust is created and organized for the sole purposes of collecting. holding, liquidating, and distributing the Trust Estate and administering, compromising, settling, withdrawing, objecting to, or litigating any and all Causes of Action and objections to the Claims under the Plan, with no objective to engage in the conduct of a trade or business, except to the extent reasonably necessary to, and consistent with, the liquidating purpose of the Liquidating Trust. In accordance with such express and limited purposes, as of the Effective Date, the Liquidating Trust is authorized and directed: (i) to take any and all steps necessary to maintain the Liquidating Trust (except with respect to the Disputed Claims Reserve) as one or more liquidating trusts for federal income tax purposes in accordance with Treasury Regulation § 301.7701-4(d) and as one or more "grantor trusts" subject to the provisions of Subchapter J. Subpart E of the IRC unless otherwise required; (ii) to take all reasonable and necessary actions to conserve and protect the Trust Estate; (iii) to administer, compromise, settle, and litigate any and all Causes of Action and any other Claims or causes of action belonging to the Liquidating Trust; (iv) to the extent necessary and appropriate, object to any Claims asserted against the Debtors' Estates and the Liquidating Trust; and (v) to maintain, operate or lease (for purposes of holding for sale), or sell or otherwise liquidate or dispose of the Trust Estate, in accordance with the terms of the Liquidating Trust Agreement, the Plan and the Confirmation Order, and to distribute the net proceeds of such disposition to the Beneficiaries, in as prompt, efficient and orderly a fashion as possible in accordance with the provisions of Section 5 of the Liquidating Trust Agreement.

On the Effective Date of the Plan, each of the Debtors shall transfer and assign all of its respective Assets (other than the Nycomed Litigation Proceeds) to the Liquidating Trust free and clear of all Liens, Claims, interests and encumbrances (except that the Assets other than

the Other Assets shall remain subject to the Liens securing (i) the First Lien Facility Secured Claim and any First Lien Facility Adequate Protection Claim as provided in Section 5.2 of the Plan and (ii) the Second Lien Facility Secured Claim and any Second Lien Facility Adequate Protection Claim as provided in Section 5.3 of the Plan).

As soon as possible after the Effective Date, but in no event later than ninety (90) days thereafter, the Liquidating Trustee, based upon his good faith determination after consultation with his counsel, shall inform the Beneficiaries in writing solely as to his estimate of the value of the Assets transferred to the Liquidating Trust. The valuation shall be used consistently by all parties (including, without limitation, the Debtors, the Liquidating Trustee, and the Beneficiaries) for all federal income tax purposes, and the parties shall file tax returns consistent with such valuation; provided, however, that such valuation shall not be binding on the Liquidating Trustee or any other party for any other purposes, including, without limitation, in regard to the liquidation of the Trust Assets, whether by disposition, liquidation, litigation, settlement, or otherwise.

For U.S. federal income tax purposes: (i) the Debtors, the Liquidating Trustee and the Beneficiaries will treat the transfer of Assets to the Liquidating Trust as a transfer by the Debtors of the Debtors' Assets (net of any applicable liabilities) to the Beneficiaries (to the extent of the value of their respective interests in the applicable Trust Assets), followed by a transfer of such Assets (net of any applicable liabilities) by the Beneficiaries to the Liquidating Trust (to the extent of the value of their respective interests in the applicable Trust Assets); (ii) the Liquidating Trust (except with respect to the Disputed Claims Reserve) will be treated as one or more grantor trusts, and the Beneficiaries will be treated as the grantors and deemed owners of the Trust Assets; and (iii) the income of the Liquidating Trust will be treated as subject to tax on a current basis.

In furtherance of the implementation of the Plan, except as otherwise provided in the Plan, all injunctions or stays provided for in the Chapter 11 Cases pursuant to Sections 105 or 362 of the Bankruptcy Code, or otherwise, and in existence on the Confirmation Date, shall remain in full force and effect and apply to all Beneficiaries and Creditors holding Claims against the Debtors, the Estates, the Assets, the Liquidating Trustee, the Liquidating Trust and the Trust Assets until the Final Distribution Date (defined below).

Except as otherwise provided by the Plan or the Liquidating Trust Agreement, upon the Effective Date, title to the Trust Assets shall pass to the Liquidating Trust free and clear of all Claims and Interests, in accordance with Section 1141 of the Bankruptcy Code. The Liquidating Trustee shall not be permitted to receive or retain Cash or Cash equivalents in excess of a reasonable amount to meet distributions as provided in the Liquidating Trust Agreement and the Plan or to maintain the value of the Trust Assets during liquidation.

On the Effective Date, (i) the Debtors' equity interests in Graceway Canada and (ii) the Debtors' Claims against Graceway Canada arising from any intercompany claims, loans, notes, transfers or other obligations shall be transferred by the Debtors to the Liquidating Trustee, and the equity interests in Graceway Canada shall continue to be subject to the Liens securing the First Lien Facility Secured Claim and any First Lien Adequate Protection Claim as provided in Section 5.2 of the Plan and the Liens of the Second Lien Facility Agent in favor of

the Second Lien Facility Lenders as provided in Section 5.3 of the Plan; provided, however, that the portion of the Canadian Distribution Amount used to fund the Estate Fund Amount shall be free and clear of the Liens of the First Lien Facility Agent in accordance with the Committee Settlement. The Liquidating Trust shall exercise its powers as the sole shareholder to appoint the Liquidating Trustee as the sole officer and director of Graceway Canada and shall make distributions of the Canadian Distribution Amount as provided under the Plan.

Following Confirmation and prior to the occurrence of the Effective Date, the then-current officers and directors of each of the Debtors shall continue in their respective capacities and the Debtors shall execute such documents and take such other action as is necessary to effectuate the transactions provided for in the Plan. On and after the Effective Date, all such officers and directors shall be deemed to have resigned.

Payment of Allowed Other Secured Claims (to the extent not otherwise satisfied in accordance with the terms of the Plan), Allowed Administrative Expense Claims, Allowed Priority Tax Claims and Allowed Other Priority Claims shall be made from the Reserve Accounts to the extent permitted by the Budget and/or the Sale Support Agreement. To the extent any of such Allowed Claims are not satisfied from the Reserve Accounts (and, with respect to Allowed Other Secured Claims, are not otherwise satisfied in accordance with the terms of the Plan), such Allowed Claims shall be paid from the Other Available Cash. The Debtors anticipate that the Cash in the Reserve Accounts and the Other Available Cash shall be sufficient to pay all Allowed Other Secured Claims (to the extent not otherwise satisfied in accordance with the terms of the Plan), Allowed Administrative Expense Claims, Allowed Priority Tax Claims, Allowed Other Priority Claims and cure costs for executory contracts or unexpired leases on the Assumption Schedule, if any, in full in Cash on the Effective Date or once such Claims become Allowed Claims. Other Allowed Claims shall be satisfied as set forth in Article IV.D above. Notwithstanding anything contained in the Plan or otherwise, (i) Cash in the CCR Account shall be available, to the extent permitted by the Budget and/or the Sale Support Agreement, only to pay Claims on account of chargebacks, channel management agreements, product returns and rebates constituting (a) Allowed Claims authorized to be paid by the Bankruptcy Court pursuant to the Orders approving the Prepetition Claims Motions or (b) Allowed Administrative Expense Claims, (ii) Cash in the Employee Account shall be available only to pay Claims on account of employee severance and benefits approved in the Orders approving the Prepetition Claims Motion and permitted by the Budget and/or Sale Support Agreement, (iii) Cash in the Lazard Account shall be available only to pay Allowed Professional Fees of Lazard, and (iv) Cash in the Designated Account shall be available only to pay Allowed Administrative Expense Claims, Allowed Other Secured Claims (to the extent not otherwise satisfied in accordance with the terms of the Plan), Allowed Priority Tax Claims and Allowed Other Priority Claims to the extent permitted by the Budget and/or Sale Support Agreement, other than any such Claims on account of (A) chargebacks, channel management agreements, product returns and rebates or (B) employee severance and benefits payable from the Employee Account.

From time to time as agreed between the First Lien Facility Agent and the Liquidating Trustee, the First Lien Facility Agent shall fund amounts into the First Lien Trust Administrative Fund (defined below) to pay the Liquidating Trustee First Lien Fees, incurred from the Effective Date through and including the date all First Lien Facility Claims are paid in

full in Cash, in connection with (a) the collection of accounts receivable, (b) the pursuit of Causes of Action (including 549 Avoidance Actions but excluding for the avoidance of doubt Avoidance Actions), and (c) the objection to, and settlement of, Other Secured Claims (including Possessory Lienholder Claims), Administrative Expense Claims, Priority Tax Claims and Other Priority Claims from Sale Proceeds to which the First Lien Facility Agent or First Lien Claimholders would otherwise be entitled. On the Effective Date, the Liquidating Trustee Fee Amount shall be funded into the Trust Administrative Fund by the First Lien Facility Agent from Sale Proceeds to which the First Lien Facility Agent or First Lien Claimholders would otherwise be entitled free and clear of all Liens and Claims of the First Lien Facility Agent and the First Lien Claimholders, including the First Lien Facility Secured Claim, any First Lien Facility Adequate Protection Claim and any First Lien Facility Deficiency Claim for the sole purpose of paying the Liquidating Trustee Fees. Any additional amount the Liquidating Trustee deems necessary to fund post-Effective Date operations other than fees and expenses incurred from the Effective Date through and including the date all First Lien Facility Claims are paid in full in Cash in connection with (a) the collection of accounts receivable, (b) the pursuit of Causes of Action (including 549 Avoidance Actions but excluding for the avoidance of doubt Avoidance Actions), and (c) the objection to, and settlement of, Other Secured Claims (including Possessory Lienholder Claims), Administrative Expense Claims, Priority Tax Claims and Other Priority Claims, shall be funded into the Trust Administrative Fund from the Other Assets.

As soon as practicable after the Effective Date, each of the Debtors will be dissolved for all purposes without the necessity for any other or further actions to be taken by or on behalf of the Debtors or payments to be made in connection therewith; *provided*, *however*, that pursuant to Section 1142(b) of the Bankruptcy Code, the Liquidating Trustee shall file each Debtor's final tax returns, and shall be authorized, but not directed, to file with the official public office for keeping corporate records in each Debtor's state of incorporation a certificate of dissolution or equivalent document. Such a certificate of dissolution may be executed by the Liquidating Trustee without need for any action or approval by the shareholders or Board of Directors of any Debtor. From and after the Effective Date, the Debtors (i) for all purposes shall be deemed to have withdrawn their business operations from any state in which they were previously conducting, or are registered or licensed to conduct, their business operations, and shall not be required to file any document, pay any sum or take any other action in order to effectuate such withdrawal, (ii) shall be deemed to have cancelled pursuant to the Plan all Interests and all Intercompany Claims, and (iii) shall not be liable in any manner to any taxing authority for franchise, business, license or similar taxes accruing on or after the Effective Date.

Notwithstanding anything to the contrary in the Bankruptcy Rules providing for earlier closure of the Debtors' Chapter 11 Cases, when all Assets contributed to the Liquidating Trust have been liquidated and converted into Cash (other than those Assets abandoned by the Liquidating Trust), and such Cash has been distributed in accordance with the Liquidating Trust Agreement and the Plan, the Liquidating Trustee shall seek authority from the Bankruptcy Court to close the Debtors' Chapter 11 Cases in accordance with the Bankruptcy Code and the Bankruptcy Rules; *provided*, *however*, that, in light of the substantive consolidation of the Debtors' estates for distribution purposes, the Liquidating Trustee may, in the Liquidating Trustee's sole discretion, seek authority from the Bankruptcy Court to close all of the Debtors' Chapter 11 Cases except for the Graceway Chapter 11 Case immediately upon satisfaction of all Other Secured Claims.

Subject to Section 7.11 of the Plan, on and after the Effective Date, the Liquidating Trustee shall have the exclusive authority to settle Causes of Action owned by the Liquidating Trust. On and after the Effective Date and until such time as the First Lien Facility Claims are paid in full in Cash, in order to settle Causes of Action owned by the Liquidating Trust, the Liquidating Trustee shall be required to obtain (a) the prior written consent of the Approving Majority First Lien Lenders or (b) an Order of the Bankruptcy Court approving such settlement after notice and a hearing. After the First Lien Facility Claims are paid in full in Cash and until such time as the Second Lien Facility Claims are paid in full in Cash, in order to settle Causes of Action owned by the Liquidating Trust, the Liquidating Trustee shall be required to obtain (i) the prior written consent of the Second Lien Facility Agent or (ii) an Order of the Bankruptcy Court approving such settlement after notice and a hearing. After the Second Lien Facility Claims are paid in full in Cash, except as otherwise set forth in the Plan, the Confirmation Order or the Liquidating Trust Agreement, the Liquidating Trustee shall not be required to obtain approval of the Approving Majority First Lien Lenders, the Second Lien Facility Agent or the Bankruptcy Court to settle Causes of Action owned by the Liquidating Trust.

Once each quarter, the Liquidating Trustee shall File with the Bankruptcy Court and serve on the U.S. Trustee an unaudited written report setting forth the amounts and timing of all distributions made and expenses paid by the Liquidating Trustee under the Plan in each applicable reporting period and the recipients thereof. In addition, the Liquidating Trustee shall file with the Bankruptcy Court and serve on the United States Trustee quarterly reports summarizing the Cash receipts of the Liquidating Trust for the immediately preceding three-month period. Each quarterly report shall also state the Liquidating Trust's Cash balances as of the beginning and ending of each such period. Quarterly reports shall be provided no later than the fifteenth (15th) day of each January, April, July and October until all Final Distributions under the Plan have been made.

On the Effective Date, any document, agreement, or instrument evidencing any Claim or Interest against or in the Debtors (including, but not limited to, the First Lien Credit Agreement and the Second Lien Credit Agreement) that has not been Assumed by the Purchaser shall be deemed cancelled without further act or action under any applicable agreement, law, regulation, Order, or rule and the Debtors shall have no further obligations under such documents, agreements or instruments; provided, however, that, although the Debtors shall have no further obligations or liability under Section 10.04 of the Credit Agreement, the Fee Letter (as defined in the First Lien Credit Agreement), the Intercreditor Agreement or Article XI of the Mezzanine Credit Agreement, as among parties other than the Debtors, Section 10.04 of the First Lien Credit Agreement, the Fee Letter, the Intercreditor Agreement and Article XI of the Mezzanine Credit Agreement shall remain in full force and effect as though the obligations of the Debtors continue to exist. Notwithstanding anything to the contrary set forth in the Plan, the Liens on the Assets (including the Reserve Accounts and the Cash deposited therein and the Nycomed Litigation Proceeds, but excluding the Other Assets) securing the First Lien Facility Secured Claim, any First Lien Facility Adequate Protection Claim, the Second Lien Facility Secured Claim and any Second Lien Facility Adequate Protection Claim shall remain valid and perfected Liens thereon.

On the Confirmation Date, the Creditors' Committee shall automatically be dissolved and the members of, and Professionals retained by, the Creditors' Committee shall have no further duties or obligations after the Confirmation Date.

On the Effective Date, the Debtors shall assign and transfer all of their rights to and interests in Nycomed Litigation Proceeds to the First Lien Facility Agent subject to any Liens thereon, which Liens shall remain valid, binding and perfected to the same extent as on the Petition Date on and after the Effective Date. Upon receipt of any Nycomed Litigation Proceeds, the First Lien Facility Agent will escrow an amount thereof sufficient to cover each Other Secured Claim asserted to be secured by Liens on the Nycomed Litigation Proceeds that are senior in priority to those held by the First Lien Facility Agent for the benefit of the First Lien Claimholders and the First Lien Facility Agent will distribute from such escrow funds sufficient to pay any such Other Secured Claim in full if and when any such Other Secured Claim is determined by an Order of the Bankruptcy Court to be an Allowed Other Secured Claim. In the event that the First Lien Facility Agent receives any Nycomed Litigation Proceeds, the First Lien Facility Agent shall provide written notice of the receipt thereof to the Second Lien Facility Agent. In the event that the First Lien Facility Claims are paid in full in Cash prior to the First Lien Facility Agent's receipt of any Nycomed Litigation Proceeds, the First Lien Facility Agent shall assign and transfer all of its rights to and interests in the Nycomed Litigation Proceeds to the Second Lien Facility Agent subject to any Liens thereon, which Liens shall remain valid and perfected to the same extent as on the Petition Date on and after the Effective Date. Upon receipt of any Nycomed Litigation Proceeds, the Second Lien Facility Agent will escrow an amount thereof sufficient to cover any Other Secured Claims asserted to be secured by Liens on the Nycomed Litigation Proceeds that are senior in priority to those held by the Second Lien Facility Agent for the benefit of the Second Lien Facility Lenders and the Second Lien Facility Agent will distribute from such escrow funds sufficient to pay any such Other Secured Claim in full if and when any such Other Secured Claim is determined by an Order of the Bankruptcy Court to be an Allowed Other Secured Claim.

Pursuant to Section 1123(b)(3) of the Bankruptcy Code, except as otherwise provided in the Plan or the Confirmation Order, after transfer of the Assets (other than the Nycomed Litigation Proceeds) to the Liquidating Trust pursuant to Section 7.2.3 of the Plan, the Liquidating Trustee (and to the extent retained by the Liquidating Trust to perform such work, any other Person) will have the exclusive right and standing in any court of competent jurisdiction to enforce any and all Causes of Action against any Entity and rights of the Debtors that arose before or after the Petition Date, including, but not limited to, the rights and powers of a trustee and debtor-in-possession, against any Entity whatsoever, including, but not limited to, all Avoidance Actions and 549 Avoidance Actions but excluding Released Claims.

No Entity may rely on the absence of a specific reference in the Plan or this Disclosure Statement to any Causes of Action against them as any indication that the Debtors or Liquidating Trust, as applicable, will not pursue any and all available Causes of Action against them. Except with respect to Causes of Action as to which the Debtors or Liquidating Trust have released any Person or Entity on or before the Effective Date (including pursuant to the Releases by the Debtors in the Plan or otherwise), the Debtors or Liquidating Trust, as applicable, expressly reserve all rights to prosecute any and all Causes of Action against any Entity, except as otherwise expressly provided in the Plan. Unless any Causes of Action against an Entity are

expressly waived, relinquished, exculpated, released, compromised or settled in the Plan or a Bankruptcy Court Order, the Liquidating Trust expressly reserves all Causes of Action, for later adjudication, and, therefore, no preclusion doctrine, including the doctrines of res judicata, collateral estoppel, issue preclusion, claim preclusion, estoppel (judicial, equitable or otherwise) or laches, shall apply to such Causes of Action upon, after or as a consequence of the Confirmation.

Subject to Section 8.2 of the Plan, on and after the Effective Date, Liquidating Trustee shall have the exclusive authority to abandon Causes of Action or other Assets owned by the Liquidating Trust. On and after the Effective Date and until such time as the First Lien Facility Claims are paid in full in Cash, in order to abandon Causes of Action or other Assets owned by the Liquidating Trust, the Liquidating Trustee shall be required to obtain (i) the prior written consent of the Approving Majority First Lien Lenders or (ii) an Order of the Bankruptcy Court approving such abandonment after notice and a hearing. After the First Lien Facility Claims are paid in full in Cash and until such time as the Second Lien Facility Claims are paid in full in Cash, in order to abandon Causes of Action or other Trust Assets owned by the Liquidating Trust, the Liquidating Trustee shall be required to obtain (i) the prior written consent of the Second Lien Facility Agent or (ii) an Order of the Bankruptcy Court approving such abandonment after notice and a hearing. After the Second Lien Facility Claims are paid in full in Cash, except as otherwise set forth in the Plan, the Confirmation Order or the Liquidating Trust Agreement, the Liquidating Trustee shall not be required to obtain approval of the Approving Majority First Lien Lenders, the Second Lien Facility Agent or the Bankruptcy Court to abandon Causes of Action or other Assets owned by the Liquidating Trust.

F. Liquidating Trust Agreement

Pursuant to Section 7.2 of the Plan and the Confirmation Order, and effective as of the Effective Date of the Plan, the Beneficiaries and the Debtors, under the Liquidating Trust Agreement, will create the Liquidating Trust for the benefit of the Beneficiaries. A summary of certain significant terms of the Liquidating Trust Agreement are set forth below. The Liquidating Trust Agreement is attached to the Plan as Exhibit 2 and the Plan is attached hereto as Exhibit A. The initial Liquidating Trustee shall be Kip Horton of RPA Advisors, LLC. Mr. Horton's curriculum vitae is attached hereto as Exhibit C.

1. Purpose of Liquidating Trust.

The Liquidating Trust is created and organized for the sole purposes of collecting, holding, liquidating, and distributing the Trust Estate and administering, compromising, settling, withdrawing, objecting to, or litigating any and all Causes of Action and objections to the Claims under the Plan, with no objective to engage in the conduct of a trade or business, except to the extent reasonably necessary to, and consistent with, the liquidating purpose of the Liquidating Trust. In accordance with such express and limited purposes, as of the Effective Date, the Liquidating Trust is authorized and directed: (i) to take any and all steps necessary to maintain the Liquidating Trust (except with respect to the Disputed Claims Reserve) as one or more liquidating trusts for federal income tax purposes in accordance with Treasury Regulation § 301.7701-4(d) and as one or more "grantor trusts" subject to the provisions of Subchapter J, Subpart E of the IRC unless otherwise required; (ii) to take all reasonable and necessary actions

to conserve and protect the Trust Estate; (iii) to administer, compromise, settle, and litigate any and all Causes of Action and any other claims or Causes of Action belonging to the Liquidating Trust; (iv) to the extent necessary and appropriate, object to any Claims asserted against the Debtors' Estates and the Liquidating Trust; and (v) to maintain, operate or lease (for purposes of holding for sale), or sell or otherwise liquidate or dispose of the Trust Estate, in accordance with the terms of the Liquidating Trust Agreement, the Plan and the Confirmation Order, and to distribute the net proceeds of such disposition to the Beneficiaries, in as prompt, efficient and orderly a fashion as possible in accordance with the provisions of Section 5 of the Liquidating Trust Agreement.

2. Contribution of the Debtors' Assets to the Liquidating Trust.

On the Effective Date of the Plan, each of the Debtors shall transfer and assign all of its respective Assets (other than the Nycomed Litigation Proceeds) to the Liquidating Trust free and clear of all Liens, Claims, interests and encumbrances (except that the Assets other than the Other Assets shall remain subject to the Liens securing (i) the First Lien Facility Secured Claim and any First Lien Facility Adequate Protection Claim as provided in Section 5.2 of the Plan and (ii) the Second Lien Facility Secured Claim and any Second Lien Facility Adequate Protection Claim as provided in Section 5.3 of the Plan).

3. <u>Valuation of Trust Assets.</u>

As soon as possible after the Effective Date, but in no event later than ninety (90) days thereafter, the Liquidating Trustee, based upon his good faith determination after consultation with his counsel, shall inform the Beneficiaries in writing solely as to his estimate of the value of the Assets transferred to the Liquidating Trust. The valuation shall be used consistently by all parties (including, without limitation, the Debtors, the Liquidating Trustee, and the Beneficiaries) for all federal income tax purposes, and the parties shall file tax returns consistent with such valuation; *provided*, *however*, that such valuation shall not be binding on the Liquidating Trustee or any other party for any other purposes, including, without limitation, in regard to the liquidation of the Trust Assets, whether by disposition, liquidation, litigation, settlement, or otherwise.

4. Certain Federal Income Tax Matters.

For U.S. federal income tax purposes; (i) the Debtors, the Liquidating Trustee and the Beneficiaries will treat the transfer of Assets to the Liquidating Trust as a transfer by the Debtors of the Debtors' Assets (net of any applicable liabilities) to the Beneficiaries (to the extent of the value of their respective interests in the applicable Trust Assets), followed by a transfer of such Assets (net of any applicable liabilities) by the Beneficiaries to the Liquidating Trust (to the extent of the value of their respective interests in the applicable Trust Assets); (ii) the Liquidating Trust (except with respect to the Disputed Claims Reserve) will be treated as one or more grantor trusts, and the Beneficiaries will be treated as the grantors and deemed owners of the Trust Assets; and (iii) the income of the Liquidating Trust will be treated as subject to tax on a current basis.

5. Continuation of the Automatic Stay.

In furtherance of the implementation of the Plan, except as otherwise provided in the Plan, all injunctions or stays provided for in the Chapter 11 Cases pursuant to Sections 105 or 362 of the Bankruptcy Code, or otherwise, and in existence on the Confirmation Date, shall remain in full force and effect and apply to all Beneficiaries and creditors holding Claims against the Debtors, the Estates, the Assets, the Liquidating Trustee, the Liquidating Trust and the Trust Assets until the Final Distribution Date.

6. Transfer of Trust Assets.

Except as otherwise provided by the Plan or the Liquidating Trust Agreement, upon the Effective Date, title to the Trust Assets shall pass to the Liquidating Trust free and clear of all Claims and Interests, in accordance with Section 1141 of the Bankruptcy Code. The Liquidating Trustee shall not be permitted to receive or retain Cash or Cash equivalents in excess of a reasonable amount to meet distributions as provided in the Liquidating Trust Agreement and the Plan or to maintain the value of the Trust Assets during liquidation.

On the Effective Date, (i) the Debtors' equity interests in Graceway Canada and (ii) the Debtors' Claims against Graceway Canada arising from any intercompany claims, loans, notes, transfers or other obligations shall be transferred by the Debtors to the Liquidating Trustee, and the equity interests in Graceway Canada shall continue to be subject to the Liens securing the First Lien Facility Secured Claim and any First Lien Adequate Protection Claim as provided in Section 5.2 of the Plan and the Liens of the Second Lien Facility Agent in favor of the Second Lien Facility Lenders as provided in Section 5.3 of the Plan; *provided*, *however*, that the portion of the Canadian Distribution Amount used to fund the Estate Fund Amount shall be free and clear of the Liens of the First Lien Facility Agent in accordance with the Committee Settlement. The Liquidating Trust shall exercise its powers as the sole shareholder to appoint the Liquidating Trustee as the sole officer and director of Graceway Canada and shall make distributions of the Canadian Distribution Amount as provided under the Plan.

7. Trust Administrative Fund.

The Liquidating Trustee shall establish a reserve fund (the "Trust Administrative Fund") for the payment by the Liquidating Trustee of all liquidation expenses, including, without limitation, the compensation of the Liquidating Trustee, any attorney, accountant, advisor or other professional retained by the Liquidating Trustee, and the payment of all other reasonable and reasonably anticipated expenses, debts, charges, liabilities and obligations relating to the Trust Estate and its administration. On the Effective Date, the Liquidating Trustee Fee Amount shall be funded into the Trust Administrative Fund by the First Lien Facility Agent from Sale Proceeds to which the First Lien Facility Agent would otherwise be entitled free and clear of all Liens and Claims of the First Lien Facility Agent, including the First Lien Facility Secured Claim, any First Lien Facility Adequate Protection Claim and any First Lien Facility Deficiency Claim, for the sole purpose of paying the Liquidating Trustee Fees. Any additional amount the Liquidating Trustee deems reasonable and necessary to fund post-Effective Date operations other than fees and expenses incurred from the Effective Date through and including the date all First Lien Facility Claims are paid in full in Cash in connection with (a) the collection of accounts

receivable, (b) the pursuit of Causes of Action (including 549 Avoidance Actions but excluding for the avoidance of doubt Avoidance Actions), and (c) the objection to, and settlement of, Other Secured Claims (including Possessory Lienholder Claims), Administrative Expense Claims, Priority Tax Claims and Other Priority Claims, shall be funded into the Trust Administrative Fund from the Other Assets. Any balance remaining in the Trust Administrative Fund after the payment of all expenses, debts, charges, liabilities and obligations intended to be paid therefrom shall be treated as Other Assets and distributed to Beneficiaries as provided in Section 5 of the Liquidating Trust Agreement. Any monies deposited in the Trust Administrative Fund pursuant to the terms of the Liquidating Trust Agreement shall be invested in interest-bearing deposits or investments specified in Section 4.6(b) of the Liquidating Trust Agreement, and the interest earned thereon shall be credited thereto. Any unspent Cash remaining in the Trust Administrative Fund immediately prior to the Final Distribution Date shall be distributed to the Beneficiaries as if such Cash were Other Assets.

8. First Lien Trust Administrative Fund.

The Liquidating Trustee shall establish a reserve fund (the "First Lien Trust Administrative Fund"), to be funded from time to time as agreed between the First Lien Facility Agent and the Liquidating Trustee, for the payment by the Liquidating Trustee of all reasonable fees and out-of-pocket expenses for services performed by the Liquidating Trustee, and any attorneys or other professionals hired by the Liquidating Trustee, incurred from the Effective Date through and including the date all First Lien Facility Claims are paid in full in Cash, in connection with (a) the collection of accounts receivable, (b) the pursuit of Causes of Action (including 549 Avoidance Actions but excluding for the avoidance of doubt Avoidance Actions), and (c) the objection to, and settlement of, Other Secured Claims (including Possessory Lienholder Claims), Administrative Expense Claims, Priority Tax Claims and Other Priority Claims from Sale Proceeds to which the First Lien Facility Agent or First Lien Claimholders would otherwise be entitled. Any balance remaining in the First Lien Trust Administrative Fund after the payment of all Liquidating Trustee First Lien Fees intended to be paid therefrom shall be treated as First Lien Available Cash and distributed to the First Lien Facility Agent for the benefit of the First Lien Claimholders. To the extent that the amount of funds in the First Lien Trust Administrative Fund is at any time insufficient after a request by the Liquidating Trustee to the First Lien Facility Agent to make a deposit into the First Lien Trust Administrative Fund, the Liquidating Trustee may, in the Liquidating Trustee's sole discretion, pay from any of the Reserve Accounts all reasonable expenses, charges, liabilities and obligations of the Liquidating Trust incurred from the Effective Date through hand including the date all First Lien Facility Claims are paid in full in Cash in connection with (a) the collection of accounts receivable, (b) the pursuit of Causes of Action (including 549 Avoidance Actions but excluding for the avoidance of doubt Avoidance Actions), and (c) the objection to, and settlement of, Other Secured Claims (including Possessory Lienholder Claims), Administrative Expense Claims, Priority Tax Claims and Other Priority Claims. Any monies deposited in the First Lien Trust Administrative Fund pursuant to the terms of the Liquidating Trust Agreement shall be invested in interest-bearing deposits or investments specified in Section 4.6(b) of the Liquidating Trust Agreement, and the interest earned thereon shall be credited thereto.

9. <u>Disputed Claims Reserve</u>.

On the Effective Date, the Liquidating Trustee shall establish a reserve fund solely from the Other Assets other than the Liquidating Trustee Fee Amount (the "Disputed Claims Reserve") for the payment by the Liquidating Trustee of all Disputed Claims if and when Allowed; provided, however, that funds shall be added to the Disputed Claims Reserve for Administrative Expense Claims, Priority Tax Claims, Other Priority Claims and Other Secured Claims (to the extent not otherwise to be satisfied in accordance with the terms of the Plan) that are Disputed Claims only to the extent the amounts deposited in the applicable Reserve Account for the payment of such Claims are insufficient to pay the Face Amount (defined below) (or, if estimated for distribution purposes, the Estimated Amount) of such Claims. With respect to any unliquidated Disputed Claim, the Disputed Claims Reserve shall be funded upon entry of and in accordance with an Order(s) of the Bankruptcy Court estimating the amount (the "Estimated Amount") of such unliquidated Disputed Claim. Upon ultimate determination that any Disputed Claim is not an Allowed Claim, the Liquidating Trustee shall immediately release the portion of the Disputed Claim Reserve or applicable Reserve Account, as applicable, on account of such Claim and distribute such portion in accordance with the Plan. Each time a distribution of Other Available Cash is made to any Class of Claims, the Liquidating Trustee shall deposit into the Disputed Claims Reserve an amount equal to the distribution each holder of a Disputed Claim in such Class would have received were the Face Amount (or, if estimated for distribution purposes, the Estimated Amount) of its Disputed Claim in such Class an Allowed Claim solely from proceeds of Other Assets. Any Person whose Disputed Claim is so estimated shall be entitled to an Allowed Claim only up to an amount not to exceed the Estimated Amount even if such Person's Disputed Claim, as finally allowed, would have exceeded the maximum Estimated Amount thereof. At the time a Person's Disputed Claim becomes an Allowed Claim, such Person shall receive from the Disputed Claims Reserve a distribution equal to the distributions such Person would have received on account of its Allowed Claim had it been an Allowed Claim at the time of such prior distributions, with any surplus Cash held in the Disputed Claims Reserve on account of such Disputed Claim becoming generally available for use by the Liquidating Trustee as Other Available Cash in accordance with the Plan and the Liquidating Trust Agreement and any surplus Cash held in any Reserve Account becoming available for distribution by the Liquidating Trustee as First Lien Available Cash in accordance with the Plan. For the avoidance of doubt, no payment shall be made from the Disputed Claims Reserve on account of the First Lien Facility Claims.

10. Administrative Powers of the Liquidating Trustee.

During the Liquidating Trustee's administration of the Liquidating Trust, and subject to (i) all the other provisions of the Liquidating Trust Agreement (including, but not limited to, Sections 4.6 and 4.7) and (ii) the Plan, the Liquidating Trustee may exercise the power:

• To receive and hold the Trust Assets and to have exclusive possession and control thereof as permissible under applicable law;

- To manage, sell and convert all or any portion of the Trust Assets to Cash and distribute the net distributable proceeds as specified in the Plan and the Liquidating Trust Agreement;
- To enter into, perform and exercise rights under contracts binding upon the Liquidating Trust (but not upon the Liquidating Trustee in his respective individual or corporate capacity) which are reasonably incident to the administration of the Liquidating Trust and which the Liquidating Trustee, in the exercise of his best business judgment, reasonably believes to be in the best interests of the Liquidating Trust;
- To delegate his authority under the Liquidating Trust to other persons, provided that such delegation must be made pursuant to a written agreement that either has been approved by the Bankruptcy Court in conjunction with the Confirmation of the Plan or is approved by the Liquidating Trust Committee (defined below);
- To establish and maintain accounts at banks and other financial institutions, in a clearly specified fiduciary capacity, into which the Trust Administrative Fund, First Lien Trust Administrative Fund, Disputed Claims Reserve or other Cash and property of the Liquidating Trust may be deposited, and draw checks or make withdrawals from such accounts, and to pay or distribute such amounts of the Trust Estate as permitted or required under the Plan and the Liquidating Trust Agreement;
- To employ attorneys, accountants, appraisers, expert witnesses, insurance adjusters or other persons whose services may be reasonably necessary or advisable, in the sole judgment of the Liquidating Trustee after consultation with the Liquidating Trust Committee, to advise or assist him in the discharge of his duties as Liquidating Trustee or otherwise in the exercise of any powers vested in the Liquidating Trustee, and to pay from the Trust Administrative Fund or First Lien Trust Administrative Fund, as applicable, reasonable compensation to such attorneys, accountants, appraisers, expert witnesses, insurance adjusters or other persons;
- To pay any and all reasonable and necessary expenses attributable or relating to the management, maintenance, operation, preservation or liquidation of the Trust Estate;
- To investigate, file, compromise, settle, withdraw or litigate in the Bankruptcy Court or on appeal (or pursuant to a withdrawal of the reference of jurisdiction) objections to Claims filed against the Debtors' Estates, the Trust Estate or the Liquidating Trust; provided, however, that (a) on and after the Effective Date and until such time as the First Lien Facility Claims are paid in full in Cash, in order to compromise, settle, otherwise resolve, or withdraw any objections to Disputed Claims, the Liquidating Trustee shall be required to obtain (i) the prior written consent

of the Approving Majority First Lien Lenders or (ii) an Order of the Bankruptcy Court approving such compromise, settlement, other resolution, or withdrawal of any objections to Disputed Claims and (b) once the First Lien Facility Claims are paid in full in Cash until such time as the Second Lien Facility Claims are paid in full in Cash, in order to compromise, settle, otherwise resolve, or withdraw any objections to Disputed Claims, the Liquidating Trustee shall be required to obtain (i) the prior written consent of the Second Lien Facility Agent or (ii) an Order of the Bankruptcy Court approving such compromise, settlement, other resolution, or withdrawal of any objections to Disputed Claims;

- To investigate, analyze, compromise, adjust, arbitrate, sue on or defend, pursue, prosecute, abandon, or otherwise deal with and settle, in accordance with the terms set forth in the Liquidating Trust Agreement, any and all Causes of Action and claims in favor of or against the Debtors, the Debtors' Estates, the Trust Estate or the Liquidating Trust that have not been released under the Plan or otherwise as the Liquidating Trustee shall deem advisable; provided, however, that (a) on and after the Effective Date and until such time as the First Lien Facility Claims are paid in full in Cash, in order to settle or abandon Causes of Action, the Liquidating Trustee shall be required to obtain (i) the prior written consent of the Approving Majority First Lien Lenders or (ii) an Order of the Bankruptcy Court approving such settlement or abandonment after notice and a hearing and (b) once the First Lien Facility Claims are paid in full in Cash until such time as the Second Lien Facility Claims are paid in full in Cash, in order to settle or abandon Causes of Action, the Liquidating Trustee shall be required to obtain (i) the prior written consent of the Second Lien Facility Agent or (ii) an Order of the Bankruptcy Court approving such settlement or abandonment after notice and a hearing;
- To avoid and recover transfers of the Debtors' property as may be permitted by the Bankruptcy Code or applicable state law, including, without limitation, those transfers identified in the Disclosure Statement;
- To take all appropriate action with respect to the Trust Estate, including, without limitation, the filing, prosecution, settlement or other resolution of Claims and Litigation Claims;
- To sue or be sued in connection with any matter arising from or related to the Plan or the Liquidating Trust Agreement that affects in any way the rights or obligations of the Liquidating Trust, the Liquidating Trustee or the Beneficiaries;
- To represent the interests of the Beneficiaries with respect to any matters relating to the Plan, the Liquidating Trust Agreement, or the Liquidating Trust affecting the rights of such Beneficiaries;

- If the Liquidating Trust shall become subject to federal or state income tax, the Liquidating Trustee shall have the power, exercisable at his reasonable discretion, to take any action reasonably necessary to minimize any adverse federal or state income tax consequences to the Beneficiaries resulting from any distribution made by the Liquidating Trust to such Beneficiaries;
- In general, without in any manner limiting any of the foregoing or the following, to deal with the Trust Assets or any part or parts thereof in all other ways as would be lawful for any person owning the same to deal therewith; provided, however, that the investment powers of the Liquidating Trustee, other than those reasonably necessary to maintain the value of the Trust Assets of the Liquidating Trust and to further the liquidating purpose of the Liquidating Trust, are limited by the terms thereof;
- To do any and all other things not in violation of any other terms of the Plan, the Confirmation Order and the Liquidating Trust Agreement that, in the reasonable business judgment of the Liquidating Trustee, are necessary or appropriate for the proper liquidation, management, investment and distribution of the Trust Assets in accordance with the provisions of the Liquidating Trust Agreement and the Plan;
- To file tax returns for the Liquidating Trust and the Disputed Claims Reserve, including pursuant to Treasury Regulation § 1.468B-9, and each of the Debtors; and
- At the appropriate time, to request that the Bankruptcy Court enter a final decree closing each of the Debtors' Chapter 11 Cases; provided, however, that, in light of the substantive consolidation of the Debtors' estates for distribution purposes, the Liquidating Trustee may, in the Liquidating Trustee's sole discretion, seek authority from the Bankruptcy Court to close all of the Debtors' Chapter 11 Cases except for the Graceway Chapter 11 Case immediately upon satisfaction of all Other Secured Claims.

11. <u>Limitations on Liquidating Trustee.</u>

The Liquidating Trustee shall carry out the purposes of the Liquidating Trust and the directions contained therein and shall not at any time cause the Liquidating Trust to enter into or engage in any trade or business (except as may be consistent with the limited purposes of the Liquidating Trust), including, without limitation, the purchase of any assets or property (other than such assets or property as are reasonably necessary to carry out the purposes of the Liquidating Trust Agreement, on behalf of the Liquidating Trust or the Beneficiaries). The Liquidating Trustee is directed to take all reasonable and necessary actions to dispose of the Trust Estate in as prompt, efficient and orderly a fashion as possible, to make timely distributions

of the proceeds of the Trust Estate, and to otherwise not unduly prolong the duration of the Liquidating Trust.

The Liquidating Trustee shall invest any monies held at any time as part of the Trust Estate, including, without limitation, in the Trust Administrative Fund, First Lien Trust Administrative Fund and every other reserve or escrow fund established pursuant to the terms of the Liquidating Trust Agreement, only in interest-bearing deposits or certificates of deposit issued by any federally insured banking institution or short-term obligations of, or short-term obligations unconditionally guaranteed as to payment by, the United States of America and its agencies or instrumentalities, pending the need for the disbursement thereof in payment of costs, expenses, and liabilities of the Liquidating Trust or in making distributions pursuant to Section 5 of the Liquidating Trust Agreement. The Liquidating Trustee shall be restricted to the collection and holding of such monies and any income earned on such monies and to the payment and distribution thereof (at least annually if such monies are not necessary to maintain the value of the Trust Estate or to satisfy Claims against the Trust Estate) for the purposes set forth in the Plan and the Liquidating Trust Agreement, and to the conservation and protection of the Trust Estate in accordance with the provisions of the Liquidating Trust Agreement.

Neither the Liquidating Trust, nor the Liquidating Trustee on behalf of the Liquidating Trust, shall incur any debt.

The Liquidating Trustee shall be obligated to consult with the Liquidating Trust Committee regarding all matters affecting the Trust Estate involving \$750,000 or more, including, but not limited to, Claims objections, litigation, contested matters and the sale, transfer or abandonment of other Trust Assets. Notwithstanding anything contained in the Liquidating Trust Agreement, a member of the Liquidating Trust Committee must recuse itself from the Liquidating Trust Committee regarding any contested matter, objection to Claim, or other Litigation Claim to which it is a party or otherwise has a conflict of interest.

Subject to Sections 7.11 and 8.2 of the Plan and Sections 4.4(i) and 4.7(b) of the Liquidating Trust Agreement, on and after the Effective Date, the Liquidating Trustee shall have the exclusive authority to settle or abandon Causes of Action or abandon other Trust Assets owned by the Liquidating Trust. On and after the Effective Date and until such time as the First Lien Facility Claims are paid in full in Cash, in order to settle or abandon Causes of Action or abandon other Trust Assets owned by the Liquidating Trust, the Liquidating Trustee shall be required to obtain (i) the prior written consent of the Approving Majority First Lien Lenders or (ii) an Order of the Bankruptcy Court approving such settlement or abandonment after notice and a hearing. After the First Lien Facility Claims are paid in full in Cash and until such time as the Second Lien Facility Claims are paid in full in Cash, in order to settle or abandon Causes of Action or abandon other Trust Assets owned by the Liquidating Trust, the Liquidating Trustee shall be required to obtain (i) the prior written consent of the Second Lien Facility Agent or (ii) an Order of the Bankruptcy Court approving such settlement or abandonment after notice and a hearing. After the Second Lien Facility Claims are paid in full in Cash, except as otherwise set forth in the Plan, the Confirmation Order or the Liquidating Trust Agreement, the Liquidating Trustee shall not be required to obtain approval of the Approving Majority First Lien Lenders, the Second Lien Facility Agent or the Bankruptcy Court to settle or abandon Causes of Action or abandon other Trust Assets owned by the Liquidating Trust.

In all other events, except as otherwise set forth in the Liquidating Trust Agreement, the Liquidating Trustee is authorized to compromise or settle an action without any notice or consent if the Liquidating Trustee reasonably believes such settlement or compromise to be in the best interests of the Liquidating Trust, and shall be held harmless by the Beneficiaries in taking such action.

Notwithstanding anything in the Liquidating Trust Agreement to the contrary, the Liquidating Trustee shall be authorized to pay the reasonable fees and expenses incurred by its professionals after the Effective Date without providing notice to or obtaining the approval of any party; *provided*, *however*, that such payments may be made only from the Trust Administrative Fund or First Lien Trust Administrative Fund, as applicable.

12. <u>Compensation of Liquidating Trustee.</u>

In each case, according to the terms of Schedule 4.12 of the Liquidation Trust Agreement and subject to Section 4.10 of the Liquidating Trust Agreement, (a) the Liquidating Trustee Fees will be compensated from the Liquidating Trustee Fee Amount and the Other Assets, (b) the Liquidating Trustee First Lien Fees will be compensated from the Sale Proceeds to which the First Lien Facility Agent or First Lien Claimholders would otherwise be entitled and (c) the Contingency Fee Counsel Fees and Expenses shall be paid from the Causes of Action Proceeds. As set forth on Schedule 4.12 of the Liquidating Trust Agreement, the Liquidating Trustee will receive a fee of \$640 per hour.

13. Reports.

The Liquidating Trustee shall:

Prepare and file unaudited interim financial reports as may be required by regulatory authorities, applicable laws, rules or regulations or as the Liquidating Trustee deems advisable during the fiscal year;

Prepare, file and mail, within the time required by applicable law or regulation, necessary income tax information, tax returns or reports to the Beneficiaries and applicable taxing authorities, including, on an annual basis, the manner and calculation of the Liquidating Trust's taxable income or loss which the Liquidating Trust would recognize if it were a separate taxable entity. In this connection, the Liquidating Trustee shall file returns for the Liquidating Trust, except with respect to the Disputed Claims Reserve, as one or more grantor trusts pursuant to Treasury Regulation Section 1.671-4(a);

Prepare and provide to the Liquidating Trust Committee periodic budgets acceptable to the Liquidating Trust Committee, which set forth, among other things, Liquidating Trustee Fees and Liquidating Trustee First Lien Fees expected to be incurred over the budgeted period; and

Once each quarter, the Liquidating Trustee shall File with the Bankruptcy Court and serve on the U.S. Trustee, the Liquidating Trust Committee, and any Beneficiary who requests copies of such quarterly report after the Confirmation Date, an unaudited written report setting forth the amounts and timing of all distributions made and expenses paid by the

Liquidating Trustee under the Plan in each applicable reporting period and the recipients thereof. Each quarterly report shall also state the Liquidating Trust's Cash balances as of the beginning and ending of each such period. Quarterly reports shall be provided no later than the fifteenth (15th) day of each January, April, July and October until all Final Distributions under the Plan have been made.

14. Liquidating Trust Committee.

An administrative trust committee (the "Liquidating Trust Committee"), consisting of no more than five (5) members, shall consult from time to time on various matters as set forth in the Liquidating Trust Agreement. The Liquidating Trust Committee initially shall consist of (i) three (3) members appointed by the First Lien Facility Agent, (ii) one (1) member appointed by the Second Lien Facility Agent, and (iii) one (1) member appointed by the Creditors' Committee. Each member of the Liquidating Trust Committee shall serve for a term of two (2) years. Upon the conclusion of a member's term, or in the event that a member resigns prior to the expiration of his term or is otherwise unable to serve out his term (other than due to the payment in full of the First Lien Facility Claims as discussed below), he shall appoint a successor to serve on the Liquidating Trust Committee. If, for any reason, such member is unable to appoint a successor member, the Liquidating Trustee shall appoint such member. Within ten (10) Business Days following payment in full of the First Lien Facility Claims, the three (3) members appointed by the First Lien Facility Agent (or its successors) shall resign from the Liquidating Trust Committee. Thereafter, the member appointed by the Second Lien Facility Agent (or his successor) shall appoint one (1) additional member. Unless otherwise specified in the Liquidating Trust Agreement, approval of a majority of the members of such Liquidating Trust Committee shall be required for the Liquidating Trust Committee to act. The Liquidating Trust Committee shall have the rights and powers set forth in the Liquidating Trust Agreement. In the event that a Liquidating Trust Committee shall not be formed and continuing to exist under the Liquidating Trust Agreement, all references therein to required approval or other action of such Liquidating Trust Committee shall be of no force or effect. In performance of their duties thereunder, members of the Liquidating Trust Committee shall be entitled to receive reimbursement of reasonable costs, expenses and obligations as set forth in Section 7.6e(iii) of the Liquidating Trust Agreement.

15. Removal of the Liquidating Trustee.

The Liquidating Trust Committee shall be authorized, by unanimous vote of the members, to remove the Liquidating Trustee for reasonable cause and to select a replacement or successor liquidating trustee in accordance with the provisions of the Liquidating Trust Agreement.

16. **Indemnification.**

Pursuant to the Liquidating Trust Agreement, the Liquidating Trust agrees to indemnify to the full extent of the Trust Estate any person or entity who was or is a party, or is threatened to be made a party to any threatened, pending, or completed action, suit or proceeding, whether civil, criminal, administrative, or investigative by reason of the fact that such person or entity is or was a Liquidating Trustee or an employee, attorney or agent of the

Liquidating Trust or Liquidating Trustee, from and against any and all expenses (including attorneys' fees), judgments, fines and amounts paid in settlement actually and reasonably incurred by such person or entity in connection with such action, suit or proceeding, including appeals thereof, if such person or entity acted without gross negligence and willful misconduct in the exercise and performance of any power or duties of such person or entity in accordance with the Liquidating Trust Agreement.

17. Termination Date.

The Liquidating Trust shall terminate on the date when (a) a final decree has been entered closing all of the Chapter 11 Cases, (b) all the Trust Assets have been distributed and (c) all Claims have been pursued or abandoned pursuant to, and in accordance with, the Plan and the Liquidating Trust Agreement. However, the term of the Liquidating Trust shall not exceed five (5) years from the Effective Date; provided that, upon a finding that an extension is necessary to the liquidating purpose of the Liquidating Trust and upon approval by the Bankruptcy Court, the term may be extended for a finite term based on the particular facts and circumstances. Each such extension must be approved by the Bankruptcy Court within six (6) months of the beginning of the extended term. The Liquidating Trustee shall be released of all liabilities and discharged from his or her obligations under the Plan or the Liquidating Trust Agreement once the Liquidating Trust has terminated pursuant to the terms of the Liquidating Trust Agreement.

18. <u>Limitation on Liability</u>.

No provision of the Liquidating Trust Agreement shall be construed to impart any liability upon the Liquidating Trustee unless it shall be proven in a court of competent jurisdiction that the Liquidating Trustee's actions or omissions constituted gross negligence or willful misconduct in the exercise of or failure to exercise any right or power vested in the Liquidating Trustee under the Liquidating Trust Agreement.

19. Costs and Expenses of Liquidating Trustee.

The Liquidating Trustee shall pay out of the Trust Administrative Fund all reasonable costs, expenses and obligations incurred by the Liquidating Trustee in carrying out his duties under the Liquidating Trust Agreement or in any manner connected, incidental or related to the administration of the Liquidating Trust, other than fees and expenses incurred from the Effective Date through and including the date all First Lien Facility Claims are paid in full in Cash in connection with (A) the collection of accounts receivable, (B) the pursuit of Causes of Action (including 549 Avoidance Actions but excluding for the avoidance of doubt Avoidance Actions), and (C) the objection to, and settlement of, Other Secured Claims (including Possessory Lienholder Claims), Administrative Expense Claims, Priority Tax Claims and Other Priority Claims, without application to the Bankruptcy Court, including, without limitation:

 Any reasonable, documented fees and out-of-pocket expenses of attorneys, accountants, investment advisors, expert witnesses, insurance adjusters, professionals or other persons whom the Liquidating Trustee may reasonably deem advisable to employ in connection with the Liquidating Trust, or on his own behalf in accordance with the terms of the Liquidating Trust Agreement;

- Any taxes, charges and assessments which may be owed by, or levied or assessed against, the Trust Estate or any property held in trust under the Liquidation Trust Agreement; and
- Any reasonable, documented costs, and out-of-pocket expenses incurred by the Liquidating Trust Committee in carrying out its duties under the Plan.

20. Resignation and Removal.

The Liquidating Trustee may resign and be discharged from any future obligations and liabilities under the Liquidating Trust Agreement by giving written notice thereof to the Bankruptcy Court and the Liquidating Trust Committee at least thirty (30) days prior to the effective date of such resignation. Subject to the expiration of such notice period, such resignation shall become effective on the day specified in such notice.

The Liquidating Trustee may be removed with cause at any time by Order of the Bankruptcy Court upon motion by any party in interest or in accordance with Section 4.16 of the Liquidating Trust Agreement. Upon any such removal, such removed Liquidating Trustee shall be entitled to any reimbursement and indemnification set forth in the Liquidating Trust Agreement that remain due and owing to such Liquidating Trustee at the time of such removal.

If, at any time, the Liquidating Trustee shall give notice of his intent to resign pursuant to Section 7.8 of the Liquidating Trust Agreement or be removed or become incapable of acting, counsel to the Liquidating Trustee shall provide notice thereof to the Bankruptcy Court and the Liquidating Trust Committee shall designate a successor Liquidating Trustee to act under the Liquidating Trust Agreement.

Any successor Liquidating Trustee appointed under the Liquidating Trust Agreement shall execute an instrument accepting such appointment thereunder and shall deliver counterparts thereof to the Bankruptcy Court. Thereupon, such successor Liquidating Trustee shall, without any further act, become vested with all of the estates, properties, rights, powers, trusts and duties of his predecessor in the Liquidating Trust with like effect as if originally named in the Liquidating Trust Agreement.

The death, resignation, incompetency or removal of the Liquidating Trustee shall operate neither to terminate the Liquidating Trust created by the Liquidating Trust Agreement nor to revoke any existing agency created pursuant to the terms of the Liquidating Trust Agreement or invalidate any action theretofore taken by such Liquidating Trustee. In the event of the resignation or removal of the Liquidating Trustee, such Liquidating Trustee shall (i) promptly execute and deliver any such documents, instruments, and other writing as may be necessary to effect the termination of such Liquidating Trustee's capacity under the Liquidating Trust Agreement and the conveyance of the Trust Estate then held by such Liquidating Trustee to the temporary or successor trustee; (ii) deliver to the temporary or successor trustee all

documents, instruments, records, and other writings relating to the Liquidating Trust or Trust Estate as may be in the possession of such Liquidating Trustee; and (iii) otherwise assist and cooperate in effecting the transfer and assumption of his obligations and functions by the temporary or successor trustee.

G. Procedures for Treatment of Disputed Claims

Following the Effective Date, only the Liquidating Trustee shall be entitled to object to Claims. Any objections to Administrative Expense Claims shall be filed and served on the claimant no later than the later of (x) sixty (60) days after the Administrative Expense Claims Bar Date, and (y) such later date as may be fixed by the Bankruptcy Court, after notice and a hearing, which later date may be fixed before or after the date specified in clause (x) above. Any objections to any Claim other than an Administrative Expense Claim shall be Filed and served on the claimant no later than the later of (x) sixty (60) days after the date the Claim is Filed, (y) sixty (60) days after the Effective Date or such other date as may be ordered from time to time by the Bankruptcy Court and (z) such later date as may be fixed by the Bankruptcy Court, after notice and a hearing, which later date may be fixed before or after the date specified in clause (x) above.

Subject to Section 9.2 of the Plan, after the Effective Date, the Liquidating Trustee shall have the exclusive authority to compromise, settle, otherwise resolve, or withdraw any objections to Disputed Claims. On and after the Effective Date and until such time as the First Lien Facility Claims are paid in full in Cash, in order to compromise, settle, otherwise resolve, or withdraw any objections to Disputed Claims, the Liquidating Trustee shall be required to obtain (i) the prior written consent of the Approving Majority First Lien Lenders or (ii) an Order of the Bankruptcy Court approving such compromise, settlement, other resolution, or withdrawal of any objections to Disputed Claims. After the First Lien Facility Claims are paid in full in Cash until such time as the Second Lien Facility Claims are paid in full in Cash, the Liquidating Trustee shall be required to obtain (i) the prior written consent of the Second Lien Facility Agent or (ii) an Order of the Bankruptcy Court approving such compromise. settlement, other resolution, or withdrawal of any objections to Disputed Claims. After the Second Lien Facility Claims are paid in full in Cash, except as otherwise set forth in the Plan, the Confirmation Order or the Liquidating Trust Agreement, the Liquidating Trustee shall not be required to obtain approval of the Approving Majority First Lien Lenders, the Second Lien Facility Agent or the Bankruptcy Court to compromise, settle, otherwise resolve, or withdraw any objections to Disputed Claims.

After the Confirmation Date, a proof of Claim for any Claim other than an Administrative Expense Claim may not be amended without the authorization of the Bankruptcy Court. After the Administrative Expense Claims Bar Date, a proof of Claim for an Administrative Expense Claim may not be amended without the authorization of the Bankruptcy Court. Any amendments to proofs of Claim or proofs of Claim for an Administrative Expense Claim Filed after the Confirmation Date or the Administrative Expense Claims Bar Date (as applicable) shall be deemed disallowed in full and expunged without any action by the Debtors or the Liquidating Trustee, unless the holder of the Claim or Administrative Expense Claim has obtained prior Bankruptcy Court authorization to File the amendment.

Notwithstanding any other provision in the Plan, if any portion of a Claim or Administrative Expense Claim is Disputed, no payment or distribution provided in the Plan shall be required to be made on account of such Claim or Administrative Expense Claim unless and until such Disputed Claim or Disputed Administrative Expense Claim becomes an Allowed Claim in its entirety.

H. Distributions Under the Plan

In accordance with Section 7.1 of the Plan and unless otherwise expressly provided in the Plan, to the extent more than one Debtor is liable for any Claim, such Claim shall be considered a single Claim and entitled only to the payment provided therefor under the applicable provisions of the Plan and the Liquidating Trust Agreement.

Distributions to holders of Allowed Claims shall be made: (a) at the addresses set forth in the proofs of Claim Filed by such holders; (b) at the addresses set forth in any written notices of address change filed with the Bankruptcy Court or delivered to the Liquidating Trustee after the date on which any related proof of Claim was Filed; or (c) at the addresses reflected in the Schedules relating to the applicable Allowed Claim if no proof of Claim has been Filed and the Liquidating Trustee has not received a written notice of a change of address.

Except as otherwise provided in the Confirmation Order, Cash payments to be made pursuant to the Plan shall be made by checks drawn on a domestic bank or by wire transfer from a domestic bank, at the option of the Debtors (with respect to Cash payments made on the Effective Date) and Liquidating Trustee (with respect to Cash payment made after the Effective Date).

Except as otherwise set forth in the Plan or as required by applicable bankruptcy law, postpetition interest shall not accrue or be paid on Claims, and no holder of a Claim shall be entitled to interest accruing on or after the Petition Date on any Claim. Interest shall not accrue or be paid upon any Disputed Claim in respect of the period from the Petition Date to the date a final distribution is made thereon if and after such Disputed Claim becomes an Allowed Claim. To the extent that any Allowed Claim entitled to a distribution under the Plan is composed of indebtedness and accrued but unpaid interest thereon, such distribution shall, to the extent permitted by applicable law, be allocated for federal income tax purposes to the principal amount of the Allowed Claim first and then, to the extent the consideration exceeds the principal amount of the Allowed Claim, to the portion of such Allowed Claim representing accrued but unpaid interest.

No payment of Cash in an amount of less than \$100.00 shall be required to be made on account of any Allowed Claim. Such undistributed amount may instead be made part of the Other Available Cash for use in accordance with the Plan and the Liquidating Trust Agreement. If the Other Available Cash for the Final Distribution is less than \$10,000, and the Liquidating Trustee, in his sole discretion, determines that it would cost more than \$5,000 to distribute such funds, the Liquidating Trustee may donate such funds to the charity of his choice.

Unless otherwise expressly set forth in the Plan with respect to a specific Claim or Class of Claims, for the purpose of the provisions of Article 10 of the Plan, the "Face Amount"

of a Disputed Claim means the amount set forth on the proof of Claim unless the Disputed Claim has been estimated for distribution purposes, in which case the Face Amount shall be the Estimated Amount, or, in the alternative, if no proof of Claim has been Filed or deemed Filed, zero.

If the distribution check to any holder of an Allowed Claim is not cashed within 90 days after issuance by the Debtors or Liquidating Trustee, after consultation with the Liquidating Trust Committee, the Liquidating Trustee may give a stop payment order with respect to the check and no further distributions shall be made to such holder on account of such Allowed Claim. Such Allowed Claim shall be discharged and the holder of such Allowed Claim shall be forever barred from asserting such Claim against the Liquidating Trust, the Liquidating Trustee, the Debtors, their Estates or their respective property. In such cases, any Cash held for distribution on account of such Claim shall (i) become the property of the Liquidating Trust and (ii) be distributed to other Creditors in accordance with the terms of the Plan and the Liquidating Trust Agreement.

On the Effective Date, or as soon thereafter as practicable in accordance with the Claims resolution process described in the Plan, the Liquidating Trustee shall distribute to the holders of Allowed Administrative Expense Claims (other than Administrative Expense Claims that are Assumed Liabilities), Allowed Priority Tax Claims, Allowed Class 1 Claims and Allowed Class 4 Claims (to the extent not otherwise satisfied in accordance with the terms of the Plan) the distributions for each respective Class as set forth in the Plan.

On the Effective Date, after payment in full in Cash of all severance and benefits approved in the Orders approving the Prepetition Claims Motions, any Cash remaining in the Employee Account shall be First Lien Available Cash and shall be promptly distributed to the First Lien Facility Agent or, to the extent the First Lien Facility Claims have been paid in full in Cash, as otherwise set forth in the Plan. Promptly after the Estate Fund Amount is fully funded (or, if the Estate Fund Amount is fully funded prior to the Effective Date, on the Effective Date), any Canadian Distribution Amount remaining and any Cash in the 1.0% Holdback Account remaining shall be First Lien Available Cash and shall be promptly distributed to the First Lien Facility Agent or, to the extent the First Lien Facility Claims have been paid in full in Cash, as otherwise set forth in the Plan. Promptly after the Allowed Professional Fees of Lazard are paid in full in Cash (or, if the Allowed Professional Fees of Lazard are paid in full in Cash prior to the Effective Date, on the Effective Date), any Cash remaining in the Lazard Account shall be First Lien Available Cash and shall be promptly distributed to the First Lien Facility Agent or, to the extent the First Lien Facility Claims have been paid in full in Cash, as otherwise set forth in the Plan. Promptly after the Administrative Expense Claims Bar Date, any amounts in the CCR Account in excess of the aggregate Face Amounts (or, if estimated for distribution purposes, the Estimated Amounts) of Claims Filed asserting liabilities related to chargebacks, channel management agreements, product returns and/or rebates shall be First Lien Available Cash and shall be promptly distributed to the First Lien Facility Agent or, to the extent the First Lien Facility Claims have been paid in full in Cash, as otherwise set forth in the Plan. In addition, upon the ultimate determination that any Disputed Claim to be funded from the CCR Account is not Allowed and payment of any Liquidating Trustee First Lien Fees associated with such determination, the portion of the CCR Account held on account of such Claim shall be First Lien Available Cash and shall be promptly distributed to the First Lien Facility Agent or, to the extent the First Lien Facility Claims have been paid in full in Cash, as otherwise set forth in the Plan. On the Effective Date, any amounts in the Designated Account in excess of the aggregate amount set forth in the Budget for all expenses to be paid from the Designated Account shall be First Lien Available Cash and shall be promptly distributed to the First Lien Facility Agent or, to the extent the First Lien Facility Claims have been paid in full in Cash, as otherwise set forth in the Plan. Promptly after the Administrative Expense Claims Bar Date, any amounts in the Designated Account in excess of the aggregate Face Amounts (or, if estimated for distribution purposes, the Estimated Amounts) of the Claims Filed that are to be funded from the Designated Account shall be First Lien Available Cash and shall be promptly distributed to the First Lien Facility Agent or, to the extent the First Lien Facility Claims have been paid in full in Cash, as otherwise set forth in the Plan. In addition, upon the ultimate determination that any Disputed Claim to be funded from the Designated Account is not Allowed and payment of any Liquidating Trustee First Lien Fees associated with such determination, the portion of the Designated Account held on account of such Claim shall be First Lien Available Cash and shall be promptly distributed to the First Lien Facility Agent or, to the extent the First Lien Facility Claims have been paid in full in Cash, as otherwise set forth in the Plan.

From time to time, but in no event less than once every three (3) months, the Liquidating Trustee shall distribute all Other Available Cash and any First Lien Available Cash not already distributed pursuant to Section 10.8.2 of the Plan as set forth in the Plan.

Not later than ten (10) Business Days after receipt thereof, after the liquidation into Cash of Causes of Action (other than those Causes of Action abandoned by the Liquidating Trustee in accordance with the terms of the Liquidating Trust Agreement and the Plan), the Liquidating Trustee shall distribute Net Causes of Action Proceeds to the holders of Allowed Claims in accordance with Sections 5.2, 5.3 and 5.4 of the Plan.

From and after the Effective Date, the Liquidating Trustee shall distribute on a biweekly basis all collections in respect of accounts receivable received during each such twoweek period in accordance with Sections 5.2 and 5.3 of the Plan.

After (a) the distribution of the payments described in Section 10.8.1 of the Plan, (b) the liquidation into Cash of all Assets owned by the Liquidating Trust (other than those Assets abandoned by the Liquidating Trustee in accordance with the terms of the Plan), (c) the collection of other sums due or otherwise remitted or returned to the Estates, and (d) the resolution of all Disputed Claims and (e) the payment in full in Cash of all Liquidating Trustee Fees and Liquidating Trustee First Lien Fees, the Liquidating Trustee shall distribute the remaining Assets owned by the Liquidating Trust in accordance with the Plan (the "Final Distribution"). The date of the Final Distribution shall be the "Final Distribution Date." The Liquidating Trustee shall make the Final Distribution in accordance with Section 5.6 of the Liquidating Trust Agreement.

On the Effective Date, the Liquidating Trustee shall establish the Disputed Claims Reserve for the payment of Disputed Claims in accordance with the terms of the Liquidating Trust Agreement. Upon ultimate determination that any Disputed Claim is not an Allowed Claim, the Liquidating Trustee shall immediately release the portion of the Disputed Claim

Reserve on account of such Claim and distribute such portion in accordance with the Plan and Section IV.F.8 herein.

In connection with the Plan and the distributions made in accordance therewith, to the extent applicable, the Debtors and the Liquidating Trust shall comply with all tax withholding and reporting requirements imposed by any Governmental Unit, and all distributions pursuant to the Plan shall be subject to such withholding and reporting requirements. The Liquidating Trustee shall be authorized to take any and all actions that may be necessary or appropriate to comply with such withholding and reporting requirements.

I. Conditions to Confirmation

The following are each conditions to entry of the Confirmation Order:

- a) The Bankruptcy Court shall have approved the Disclosure Statement with respect to the Plan in form and substance that is reasonably acceptable to the Debtors and the First Lien Facility Agent; and
- b) The Confirmation Order shall be in form and substance reasonably satisfactory to the Debtors and the First Lien Facility Agent

J. Conditions to the Effective Date

The Plan shall not become effective and the Effective Date shall not occur unless and until:

- a) The Bankruptcy Court shall have entered the Confirmation Order in form and substance reasonably satisfactory to the Debtors and the First Lien Facility Agent;
- b) The Confirmation Order shall not have been amended or modified without the consent of the Debtors and the First Lien Facility Agent;
- c) All documents, instruments and agreements, in form and substance reasonably satisfactory to the Debtors and the First Lien Facility Agent provided for under the Plan or necessary to implement the Plan, including, without limitation, the Liquidating Trust Agreement, shall have been executed and delivered by the parties thereto, unless such execution or delivery has been waived by the parties benefited thereby;
- d) The Debtors shall have reasonably determined that the Cash in the Reserve Accounts and the Other Available Cash is sufficient to pay in full in Cash on the Effective Date all estimated Allowed Other Secured Claims (to the extent not otherwise satisfied in accordance with the terms of the Plan), Allowed Administrative Expense Claims (other than Administrative Expense Claims that are Assumed Liabilities), Allowed Other Priority Claims and Allowed Priority Tax Claims; and
 - e) The Confirmation Order shall have become a Final Order.

K. Termination of Plan for Failure To Become Effective.

If the Effective Date shall not have occurred on or prior to the date that is forty-five (45) days after the Confirmation Date, then the Plan shall terminate and be of no further force or effect unless the provisions of Section 11.3 of the Plan, describing the termination of the Plan for failure to become effective, are waived in writing by the Debtors and First Lien Facility Agent.

L. Waiver of Conditions

The Debtors may waive any or all of the conditions set forth in Sections 11.1, describing the conditions to Confirmation, and/or 11.2, describing the conditions to the Effective Date, of the Plan (other than the conditions set forth in Sections 11.1.1 (as described in Section IV.I(a) above), 11.2.1 (as described in Section IV.J(a) above) and 11.2.4 (as described in Section IV.J(d) above) of the Plan) subject to the consent of the First Lien Facility Agent.

M. Modification of the Plan

Subject to the restrictions on Plan modifications set forth in Section 1127 of the Bankruptcy Code, the Debtors reserve the right, after consultation with the First Lien Facility Agent, the Second Lien Facility Agent and the Creditors' Committee, to alter, amend or modify the Plan before its substantial consummation.

N. Effect of Confirmation

Pursuant to Sections 105(a) and 1142 of the Bankruptcy Code, and notwithstanding entry of the Confirmation Order and occurrence of the Effective Date, the Bankruptcy Court shall retain exclusive jurisdiction over all matters arising out of, and related to, the Debtors' Chapter 11 Cases and the Plan to the fullest extent permitted by law.

Except as otherwise provided in Section 1141(d) of the Bankruptcy Code, on and after the Confirmation Date, the provisions of the Plan shall bind (i) any holder of a Claim against or Interest in the Debtors and its respective successors and assigns (including the Liquidating Trust), whether or not the Claim or Interest of such holder is Impaired under the Plan and whether or not such holder has accepted the Plan and (ii) the Liquidating Trustee.

Unless otherwise provided in a separate Order from the Bankruptcy Court, all injunctions or stays arising under or entered during the Debtors' Chapter 11 Cases in accordance with Sections 105 or 362 of the Bankruptcy Code, or otherwise, and in existence on the Confirmation Date, shall remain in full force and effect and apply to all Beneficiaries and Creditors holding Claims against the Debtors, their Estates, the Debtors' Assets, the Liquidating Trustee and the Liquidating Trust until the Final Distribution Date or, if different, the date indicated in such applicable Order.

Upon the entry of the Confirmation Order, all holders of Claims and Interests and other parties in interest, along with their respective present or former Affiliates, employees, agents, officers, directors, principals or advisors, shall be enjoined from taking any actions to interfere with the implementation or consummation of the Plan.

O. Exculpation, Injunction, and Limitation of Liability

Except as otherwise specifically provided in the Plan, none of the Purchaser (in its capacity as a purchaser under the Asset Purchase Agreement), the Debtors, the Liquidating Trustee, the First Lien Facility Agent, the First Lien Claimholders in their capacity as such, the Second Lien Facility Agent, the Second Lien Facility Lenders in their capacity as such, the Creditors' Committee (solely with respect to its conduct as a committee and not with respect to the actions of its members as individual Creditors), nor any of such parties' respective present members (with respect to members of the Creditors' Committee, solely with respect to each member's conduct in furtherance of its, his, or her duties as a member of the Creditors' Committee, and not with respect to the actions of such members as individual Creditors), officers, directors, shareholders, employees, representatives, advisors, attorneys, financial advisors, investment bankers, agents or Affiliates, or any of such parties' successors and assigns, shall have or incur, and each such Person is thereby released from, any Claim, obligation, Causes of Action in any form whatsoever or liability to one another or to any holder of a Claim or an Interest, or any other party in interest, or any of their respective officers, directors, shareholders, members and/or enrollees, employees, representatives, advisors, attorneys, financial advisors, investment bankers, agents, related professionals or Affiliates, or any of their successors or assigns, for any act or omission in connection with, relating to, or arising out of, the Chapter 11 Cases, the negotiation and execution of the Plan, the Disclosure Statement, the Asset Purchase Agreement and the 363 Sale, the solicitation of votes for and the pursuit of Confirmation of the Plan, the consummation of the Plan, or the administration of the Plan, the Asset Purchase Agreement, the 363 Sale and the property to be distributed under the Plan, including all documents ancillary thereto, all decisions, actions, inactions and alleged negligence or misconduct relating thereto and all prepetition activities leading to the promulgation and Confirmation of the Plan except fraud, willful misconduct or gross negligence as determined by a Final Order. Nothing in Section 12.5 of the Plan shall: (i) be construed as a release of any entity's fraud, gross negligence or willful misconduct with respect to matters set forth in Section 12.5 of the Plan or (ii) limit or abrogate the obligations of the Debtors or the Purchaser and any of their respective affiliates to one another under the Asset Purchase Agreement. Any of the foregoing parties in all respects shall be entitled to reasonably rely upon the advice of counsel with respect to their duties and responsibilities (if any) under the Plan.

Notwithstanding any other provision of the Plan, neither any holder of a Claim or Interest, nor other party in interest, nor any of their respective officers, directors, shareholders, members and/or enrollees, employees, representatives, advisors, attorneys, financial advisors, investment bankers, related professionals, agents or Affiliates, and no successors or assigns of the foregoing, shall have any right of action against any Debtor or the Purchaser (in its capacity as a purchaser under the Asset Purchase Agreement), the Liquidating Trustee, the First Lien Facility Agent, the First Lien Claimholders (in their capacity as such), the Second Lien Facility Agent, the Second Lien Facility Lenders (in their capacity as such), the Creditors' Committee (solely with respect to its conduct as a committee and not with respect to the actions of its members as individual Creditors), or any of such parties' respective present members (with respect to members of the Creditors' Committee, solely with respect to each member's conduct in furtherance of its, his, or her duties as a member of the Creditors' Committee, and not with respect to the actions of such members as individual Creditors), officers, directors, shareholders, employees, representatives, advisors, attorneys, financial advisors, investment bankers, related

professionals, agents or Affiliates, or any of such parties' successors and assigns, for any act or omission in connection with, relating to, or arising out of, the Chapter 11 Cases, the negotiation and execution of the Plan, the Disclosure Statement, the Asset Purchase Agreement and the 363 Sale, the solicitation of votes for and the pursuit of Confirmation of the Plan, the consummation of the Plan, or the administration of the Plan, the Asset Purchase Agreement, the 363 Sale and the property to be distributed under the Plan, including all documents ancillary thereto, all decisions, actions, inactions and alleged negligence or misconduct relating thereto and all prepetition activities leading to the promulgation and Confirmation of the Plan except fraud, willful misconduct or gross negligence as determined by a Final Order.

Except as otherwise specifically provided in the Plan or the Confirmation Order, all Entities or Persons who have held, hold or may hold Claims, rights, Causes of Action, liabilities or any equity interests based upon any act or omission, transaction or other activity of any kind or nature related to the Debtors, the Liquidating Trust or the Chapter 11 Cases that occurred prior to the Effective Date, other than as expressly provided in the Plan or the Confirmation Order, regardless of the filing, lack of filing, allowance or disallowance of such a Claim or Interest and regardless of whether such Entity has voted to accept the Plan, and any successors, assigns or representatives of such Entities shall be precluded and permanently enjoined on and after the Effective Date from (a) the enforcement, attachment, collection or recovery by any manner or means of any judgment, award, decree or Order with respect to any Claim, Interest or any other right or Claim against the Debtors, the Liquidating Trustee, the Liquidating Trust or any Assets of the Debtors or the Liquidating Trust which such Entities possessed or may possess prior to the Effective Date, (b) the creation, perfection or enforcement of any encumbrance of any kind with respect to any Claim, Interest or any other right or Claim against the Debtors, the Liquidating Trustee, the Liquidating Trust or any assets of the Debtors or the Liquidating Trust which they possessed or may possess prior to the Effective Date, and (c) the assertion of any Claims that are released thereby.

Except as expressly provided in the Plan, upon the Effective Date, each of the Debtors thereby (i) remises, acquits, waives, releases and forever discharges each of the Debtor Releasees from, and (ii) covenants and agrees never to institute or cause to be instituted any suit or other form of action or proceeding of any kind or nature whatsoever against any of the Debtor Releasees based upon, any Claims, demands, indebtedness, agreements, promises, Causes of Action, obligations, damages or liabilities of any nature whatsoever (other than rights to enforce obligations of the Debtor Releasees under any Order of the Bankruptcy Court, the Plan and all contracts, instruments, releases and other agreements delivered in connection therewith), in law or in equity, whether or not known, suspected or claimed, that the Debtors or the Estates ever had, claimed to have, has, or may have or Claim to have against the Debtor Releasees, or any of them, by reason of any matter, cause, thing, act or omission of the Debtor Releasees, or any of them, in each case related to the Debtors, the Chapter 11 Cases, the 363 Sale or the Plan, except fraud, willful misconduct or gross negligence as determined by a Final Order.

As of the Effective Date, to the fullest extent permitted under applicable law, in consideration for the obligations under the Plan and the Cash, securities, contracts, instruments, releases and other agreements or documents to be delivered in connection with the Plan, and the benefits provided by the Creditor Releasees in the Plan and in the Chapter 11 Cases, each

present and former holder of a Claim or Interest will be deemed to release forever, waive and discharge any and all Claims, obligations, suits, judgments, damages, demands, debts, rights, Causes of Action and liabilities (other than the rights to enforce the Debtors' obligations under any Order of the Bankruptcy Court, the Asset Purchase Agreement, the Plan and the securities, contracts, instruments, releases and other agreements and documents delivered thereunder), whether liquidated or unliquidated, fixed or contingent, matured or unmatured, known or unknown, foreseen or unforeseen, then existing or thereafter arising, in law, equity or otherwise that are based in whole or in part on any act, omission, transaction, event or other occurrence taking place on or prior to the Effective Date in any way relating to the Debtors, the Chapter 11 Cases, the 363 Sale, or the Plan against any Creditor Releasee, except fraud, willful misconduct or gross negligence as determined by a Final Order.

Except as expressly set forth in the Plan or the Asset Purchase Agreement, following the Effective Date, none of the Debtors, the Liquidating Trust, the Liquidating Trustee, the Creditors' Committee, the First Lien Facility Agent, the First Lien Claimholders (in their capacity as such), the Second Lien Facility Agent, the Second Lien Facility Lenders (in their capacity as such), or any of their respective members, officers, directors, employees, advisors, attorneys, professionals, agents or Affiliates shall have or incur any liability to any holder of a Claim or Interest for any act or omission in connection with, related to, or arising out of, the Chapter 11 Cases, the 363 Sale, the negotiation and pursuit of Confirmation of the Plan, the consummation of the Plan or any contract, instrument, release or other agreement or document created in connection with the 363 Sale, the Plan, or the administration of the Plan or the property to be distributed under the Plan, except for fraud, gross negligence or willful misconduct as determined by a Final Order.

P. Retention of Jurisdiction

Notwithstanding the entry of the Confirmation Order and the occurrence of the Effective Date, the Bankruptcy Court may retain jurisdiction over the Debtors' Chapter 11 Cases after the Effective Date to the fullest extent legally permissible, including jurisdiction to, among other things:

- Allow, disallow, determine, liquidate, classify, estimate or establish the priority or secured or unsecured status of any Claim or Interest, including the resolution of (x) any request for payment of any Administrative Expense Claim, (y) any Disputed Claims or Interests and (z) any and all objections to the allowance or priority of any Claim or Interest;
- To the extent not inconsistent with the Bankruptcy Code, hear and determine any and all Causes of Action against any Person and rights of the Debtors that arose before or after the Petition Date, including, but not limited to, the rights and powers of a trustee and debtor-in-possession, against any Person whatsoever, including, but not limited to, all avoidance powers granted to the Debtors under the Bankruptcy Code and all Causes of Action and remedies granted pursuant to Sections 502, 506, 510, 541, 542, 543, 544, 545, 547 through 551 and 553 of the Bankruptcy Code;

- Grant or deny any applications for allowance of compensation for Professionals authorized pursuant to the Bankruptcy Code or the Plan, for periods ending on or before the Effective Date;
- Resolve any matters relating to the assumption, assumption and assignment or rejection of any executory contract or unexpired lease to which any Debtor is a party or with respect to which any of the Debtors may be liable, including, without limitation, the determination of whether such contract is executory for the purposes of Section 365 of the Bankruptcy Code, and hear, determine and, if necessary, liquidate any Claims arising therefrom;
- Enter Orders approving the Liquidating Trustee's post-Confirmation sale or other disposition of Assets of the Liquidating Trust;
- Ensure that distributions to holders of Allowed Claims are accomplished pursuant to the provisions of the Plan and the Liquidating Trust Agreement;
- Decide or resolve any motions, adversary proceedings, contested or litigated matters and any other matters and grant or deny any applications involving any Debtor or the Liquidating Trust that may be pending in the Debtors' Chapter 11 Cases on the Effective Date;
- Hear and determine matters concerning state, local or federal taxes in accordance with Sections 346, 505 or 1146 of the Bankruptcy Code;
- Enter such Orders as may be necessary or appropriate to implement or consummate the provisions of the Liquidating Trust Agreement, the Plan and the Confirmation Order;
- Hear and determine any matters concerning the enforcement of the provisions of Article 12 of the Plan and any other exculpations, limitations of liability or injunctions contemplated by the Plan;
- Resolve any cases, controversies, suits or disputes that may arise in connection with the consummation, interpretation or enforcement of the Liquidating Trust Agreement, the Plan or the Confirmation Order;
- Permit the Debtors, to the extent authorized pursuant to Section 1127 of the Bankruptcy Code, to modify the Plan or any agreement or document created in connection with the Plan, or remedy any defect or omission or reconcile any inconsistency in the Plan or any agreement or document created in connection with the Plan;
- Issue injunctions, enter and implement other Orders or take such other actions as may be necessary or appropriate to restrain interference by any Entity with consummation, implementation or enforcement of the Liquidating Trust Agreement, the Plan or the Confirmation Order;

- Enforce any injunctions entered in connection with or relating to the Plan or the Confirmation Order;
- Enter and enforce such Orders as are necessary or appropriate if the Confirmation Order is for any reason modified, stayed, reversed, revoked or vacated, or distributions pursuant to the Plan are enjoined or stayed;
- Determine any other matters that may arise in connection with or relating to the Plan, the 363 Sale, the Asset Purchase Agreement or the Confirmation Order;
- Enter any Orders in aid of prior Orders of the Bankruptcy Court; and
- Enter a final decree closing all of the Debtors' Chapter 11 Cases.

Q. Treatment of Executory Contracts and Unexpired Leases

Pursuant to the Bid Procedures Order and the Sale Order, the Bankruptcy Court established deadlines and procedures for the assumption and rejection of executory contracts and unexpired leases by the Debtors at the direction of the Purchaser and allocated responsibility for payment of cure amounts between the Debtors and the Purchaser. Any executory contracts and unexpired leases of the Debtors, including executory contracts and unexpired leases entered into after the Petition Date, not assumed and assigned to the Purchaser or rejected prior to the Effective Date (the "Remaining Contracts") shall be rejected pursuant to Section 6.5 of the Plan unless assumed or assumed and assigned pursuant to Section 6.2 of the Plan.

As of the Effective Date, the Debtors shall assume and assign to the Liquidating Trust, pursuant to Bankruptcy Code Section 365, each of the Remaining Contracts that are identified in Exhibit 3 to the Plan (the "Assumption Schedule") that have not expired under their own terms prior to the Effective Date; provided that the Assumption Schedule to the Plan shall be acceptable to the Liquidating Trustee (or the proposed Liquidating Trustee) in his or her sole discretion. The Debtors reserve the right to amend the Assumption Schedule not later than twenty-eight (28) days prior to the Confirmation Hearing either to: (a) delete any executory contract or lease listed therein and provide for its rejection pursuant to Section 6.5 of the Plan; or (b) add any executory contract or lease to the Assumption Schedule, thus providing for its assumption and assignment pursuant to Section 6.2 of the Plan; provided, however, that any amendments to the Assumption Schedule shall be acceptable to the Liquidating Trustee (or the proposed Liquidating Trustee) in his or her sole discretion. The Debtors shall provide notice of any such amendment to the Assumption Schedule to the parties to the executory contract or lease affected thereby not later than twenty-eight (28) days prior to the Confirmation Hearing. The Confirmation Order shall constitute an Order of the Bankruptcy Court pursuant to Bankruptcy Code Section 365 approving all assumptions and assignments described in Section 6.2 of the Plan and the Assumption Schedule, as of the Effective Date.

Any monetary defaults under each Remaining Contract to be assumed and assigned under the Plan shall be satisfied, pursuant to Bankruptcy Code Section 365(b)(1) by payment of the cure amount in Cash in full on the Effective Date by the Debtors from the Designated Account to the extent permitted by the Budget and/or the Sale Support Agreement

and from the Other Assets to the extent not permitted by the Budget and/or the Sale Support Agreement. In the event of a dispute regarding (i) the amount or timing of any cure payments, (ii) the ability of the Debtors, the Liquidating Trustee, or an assignee thereof to provide adequate assurance of future performance under the Remaining Contracts to be assumed or assumed and assigned, as applicable, or (iii) any other matter pertaining to assumption or assumption and assignment of the Remaining Contracts to be assumed, the Liquidating Trustee shall pay all required cure amounts promptly following the entry of a Final Order resolving the dispute from the Designated Account to the extent permitted by the Budget and/or the Sale Support Agreement and from the Other Assets to the extent not permitted by the Budget and/or the Sale Support Agreement.

To the extent that any party to a Remaining Contract identified for assumption and assignment asserts arrearages or damages pursuant to Bankruptcy Code Section 365(b)(1), or has any other objection with respect to any proposed assumption, cure or assignment on the terms and conditions provided in the Plan, all such arrearages, damages and objections must be Filed and served no later than the earlier of (i) twenty-one (21) days after such party is served with notice of such assumption and assignment and (ii) one (1) day prior to the Confirmation Hearing.

Failure to assert such arrearages, damages or objections in the manner described above shall constitute consent to the proposed assumption, cure or assignment on the terms and conditions provided in the Plan, including an acknowledgement that the proposed assumption and/or assignment provides adequate assurance of future performance and that the amount identified for "cure" in the Assumption Schedule is the amount necessary to cover any and all outstanding defaults under the Remaining Contract to be assumed, as well as an acknowledgement and agreement that no other defaults exist under such Remaining Contract.

If any assumption of a Remaining Contract proposed in the Plan for any reason is not approved by the Bankruptcy Court, then the Debtors shall be entitled, in their sole discretion, upon written notice to the applicable non-Debtor party to such Remaining Contract, to deem such Remaining Contract to have been rejected pursuant to the provisions of Section 6.5 of the Plan.

Except for those executory contracts and unexpired leases that (a) are assumed and assigned pursuant to the Plan, (b) have been previously assumed and assigned or rejected pursuant to previous Orders of the Bankruptcy Court, or (c) are the subject of a pending motion before the Bankruptcy Court with respect to the assumption and assignment of such executory contracts and unexpired leases, as of the Effective Date, all executory contracts and unexpired leases of the Debtors shall be rejected pursuant to Section 365 of the Bankruptcy Code; provided, however, that neither the inclusion by the Debtors of a contract or lease on the Assumption Schedule nor anything contained in Article 6 of the Plan shall constitute an admission by any Debtor that such contract or lease is an executory contract or unexpired lease or that any Debtor or its successors and assigns, including, but not limited to, the Liquidating Trust, has any liability thereunder.

The Confirmation Order shall constitute an Order of the Bankruptcy Court approving the rejection of executory contracts and unexpired leases under Section 6.5 of the Plan

pursuant to Bankruptcy Code Section 365 as of the Effective Date. Any Claim for damages arising from any such rejection must be Filed within thirty (30) days after the mailing of notice of the entry of the Confirmation Order, or such Claim shall receive no distribution under the Plan or otherwise on account of such Claim. All Allowed Claims arising from the rejection of the Debtors' executory contracts or unexpired leases shall be classified as General Unsecured Claims and shall be treated in accordance with Section 5.5 of the Plan.

R. No Admissions

If Confirmation or the Effective Date does not occur, nothing contained in the Plan or Disclosure Statement shall be deemed as an admission by the Debtors with respect to any matter set forth herein or therein including, without limitation, liability on any Claim or the propriety of any Claims classification.

S. Preservation of Rights of Setoffs

The Debtors or, if after the Effective Date, the Liquidating Trustee, may, but shall not be required to, set off against any Claim, and the payments or other distributions to be made pursuant to the Plan in respect of such Claim, Claims of any nature whatsoever that the Debtors or, if after the Effective Date, the Liquidating Trustee, may have against the holder of such Claims; but neither the failure to do so nor the allowance of any Claim under the Plan shall constitute a waiver or release by the Debtors or the Liquidating Trustee of any such Claim that the Debtors or, if after the Effective Date, the Liquidating Trustee, may have against such holder.

T. Defenses with Respect to Unimpaired Claims

Except as otherwise provided in the Plan, nothing shall affect the rights and legal and equitable defenses of the Debtors or, if after the Effective Date, the Liquidating Trustee, with respect to any Unimpaired Claim, including all rights in respect of legal and equitable defenses to setoffs or recoupments against Unimpaired Claims.

U. Dissolution of Creditors' Committee

The Creditors' Committee shall be dissolved on the Effective Date, without need for a further Order of the Bankruptcy Court.

V. CONFIRMATION OF THE PLAN

A. Confirmation Hearing

The Bankruptcy Court has scheduled the Confirmation Hearing for Confirmation of the Plan for April 11, 2012 at 2:00 p.m. (prevailing Eastern time) before the Honorable Peter J. Walsh, Judge for the United States Bankruptcy Court for the District of Delaware, located at the United States Bankruptcy Court, 824 Market Street, Wilmington, Delaware 19801. The Confirmation Hearing may be adjourned from time to time without notice except as given at the Confirmation Hearing or any subsequent adjourned Confirmation Hearing. The Bankruptcy Court has directed that objections, if any, to the Confirmation of the Plan be served on or before

April 3, 2012 at 4:00 p.m. (prevailing Eastern time) in the manner described in the Notice accompanying this Disclosure Statement.

B. Confirmation Standards

For a plan to be confirmed, the Bankruptcy Code requires, among other things, that a plan be proposed in good faith and comply with the applicable provisions of chapter 11 of the Bankruptcy Code. Section 1129 of the Bankruptcy Code also imposes requirements that at least one Class of Impaired Claims accept a plan, that Confirmation of a plan is not likely to be followed by the need for further financial reorganization, that a plan be in the best interest of Creditors, and that a plan be fair and equitable with respect to each Class of Claims or Interests which is Impaired under the Plan.

The Bankruptcy Court will confirm a plan only if it finds that all of the requirements enumerated in Section 1129 of the Bankruptcy Code have been met. The Debtors and the Creditors' Committee believe that the Plan satisfies all of the requirements for Confirmation.

VI. FUNDING AND FEASIBILITY OF THE PLAN

A. Funding of the Plan

Payment of Allowed Other Secured Claims (to the extent not otherwise satisfied in accordance with the terms of the Plan), Allowed Administrative Expense Claims, Allowed Priority Tax Claims and Allowed Other Priority Claims shall be made from the Reserve Accounts to the extent permitted by the Budget and/or the Sale Support Agreement. To the extent any of such Allowed Claims are not satisfied from the Reserve Accounts (and, with respect to Allowed Other Secured Claims, are not otherwise satisfied in accordance with the terms of the Plan), such Allowed Claims shall be paid from the Other Available Cash. The Debtors anticipate that the Cash in the Reserve Accounts and the Other Available Cash shall be sufficient to pay all Allowed Other Secured Claims (to the extent not otherwise satisfied in accordance with the terms of the Plan), Allowed Administrative Expense Claims, Allowed Priority Tax Claims, Allowed Other Priority Claims and cure costs for executory contracts or unexpired leases on the Assumption Schedule, if any, in full in Cash on the Effective Date or once such Claims become Allowed Claims. Other Allowed Claims shall be satisfied as set forth in Article 5 of the Plan. Notwithstanding anything contained in the Plan or otherwise, (i) Cash in the CCR Account shall be available, to the extent permitted by the Budget and/or the Sale Support Agreement, only to pay Claims on account of chargebacks, channel management agreements, product returns and rebates constituting (a) Allowed Claims authorized to be paid by the Bankruptcy Court pursuant to the Orders approving the Prepetition Claims Motions or (b) Allowed Administrative Expense Claims, (ii) Cash in the Employee Account shall be available only to pay Claims on account of employee severance and benefits approved in the Orders approving the Prepetition Claims Motion and permitted by the Budget and/or Sale Support Agreement, (iii) Cash in the Lazard Account shall be available only to pay Allowed Professional Fees of Lazard, and (iv) Cash in the Designated Account shall be available only to pay Allowed Administrative Expense Claims, Allowed Other Secured Claims (to the extent not otherwise satisfied in accordance with the terms of the Plan), Allowed Priority Tax Claims and Allowed Other Priority Claims to the extent permitted by the Budget and/or Sale Support Agreement, other than any such Claims on account of (A) chargebacks, channel management agreements, product returns and rebates or (B) employee severance and benefits payable from the Employee Account.

B. Best Interests Test

Notwithstanding acceptance of the Plan by each Impaired Class, in order to confirm the Plan, the Bankruptcy Court must determine that the Plan is in the best interests of each holder of a Claim or Interest in any such Impaired Class who has not voted to accept the Plan. Accordingly, if an Impaired Class does not unanimously accept the Plan, the best interests test requires the Bankruptcy Court to find that the Plan provides to each member of such Impaired Class a recovery on account of the Class member's Claim or Interest that has a value, as of the Effective Date, at least equal to the value of the distribution that each such Class member would receive if the Debtors were liquidated under chapter 7 of the Bankruptcy Code on such date.

To estimate what members of each Impaired Class of unsecured Creditors and equity security holders would receive if the Debtors were liquidated under chapter 7, the Bankruptcy Court must first determine the aggregate dollar amount that would be generated from the Debtors' assets if the Chapter 11 Cases were converted to a chapter 7 case under the Bankruptcy Code and the assets were liquidated by a trustee in bankruptcy (the "Liquidation Value" of such assets). The Liquidation Value would consist of the net proceeds from the disposition of the Debtors' Assets and would be augmented by any Cash held by the Debtors.

As detailed in the Liquidation Analysis, a copy of which is attached hereto as Exhibit D, the Liquidation Value of the Debtors' Assets available to general Creditors would be reduced by the costs and expenses of the liquidation, as well as other administrative expenses of the chapter 7 cases. The Debtors' costs of liquidation under chapter 7 would include the compensation of a trustee or trustees, as well as counsel and other professionals retained by the trustee(s), disposition expenses, all unpaid expenses incurred by the Debtors during their chapter 11 proceedings (such as compensation for attorneys and accountants) which are allowed in the chapter 7 proceedings, and litigation costs and Claims against the Debtors arising from their business operations during the pendency of their Chapter 11 Cases and chapter 7 liquidation proceedings. These costs, expenses and Claims would be paid in full out of the Debtors' liquidation proceeds before the balance would be made available to pay general Creditors.

Once the percentage recoveries in liquidation of secured claimants, priority claimants, general unsecured creditors and equity security holders are ascertained, the value of the distribution available out of the Liquidation Value is compared with the value of the property offered to each of the Classes of Claims and Interests under the Plan to determine whether the Plan is in the best interests of each Class. The Debtors believe that the Plan satisfies this best interest test because the Debtors' assets will be liquidated under the terms of the Plan. If the Plan is not confirmed and, instead, the Chapter 11 Cases are converted to chapter 7, the value of the Debtors' estates would diminish because (i) the estates would need to pay fees to any chapter 7 trustee and (ii) the estates would incur increased professional fee costs associated with

supporting the Chapter 7 proceedings and associated litigation costs and Claims. Comparing the Claims in Section I hereof with the Liquidation Analysis attached hereto as Exhibit D, the Debtors believe that distributions under the Plan will provide at least the same recovery to holders of Allowed Claims against each of the Debtors on account of such Allowed Claims as would distributions by a chapter 7 trustee. Conversion would also likely delay the liquidation process and the ultimate distributions to Creditors.

C. Avoidance Action Analysis

Avoidance Actions were not sold to the Purchaser under the Asset Purchase Agreement. As discussed in Section III.F above, pursuant to the Committee Settlement the First Lien Facility Agent (for the benefit of the First Lien Claimholders) will receive (a) 100% of all proceeds, whether obtained through litigations, settlement or otherwise, of any avoidance or equitable subordination or recovery actions under Bankruptcy Code Section 549 and (b) 75% of all proceeds, whether obtained through litigations, settlement or otherwise, of any avoidance or equitable subordination or recovery actions under Bankruptcy Code Sections 105, 502(d), 510, 542, 543, 544, 545, 546, 547, 548, 550, 551, and 553. The First Lien Facility Agent will release all Liens and Claims on the remaining 25% of all proceeds, whether obtained through litigations, settlement or otherwise, of any avoidance or equitable subordination or recovery actions under Bankruptcy Code Sections 105, 502(d), 510, 542, 543, 544, 545, 546, 547, 548, 550, 551, and 553.

The Debtors transferred \$34,519,689.39 to approximately 469 Creditors other than insiders during the 90 days prior to the Petition Date. A list of such transfers is attached hereto as Exhibit E. In addition, the Debtors transferred \$4,557,352.48 to 38 Creditors who are insiders, not including intercompany transfers between Debtors, during the approximately one year prior to the Petition Date. A list of such transfers is attached hereto as Exhibit F. The Debtors define "insiders" pursuant to Section 101(31) of the Bankruptcy Code as (a) directors; (b) officers; (c) shareholders holding in excess of 5% of the voting shares; (d) relatives of directors, officers or shareholders of the Debtors (to the extent known by the Debtors); and (e) Debtor/non-Debtor affiliates. The Debtors do not believe that the employees of the Debtors with the title "vice president," including "senior vice president," constitute "insiders" pursuant to Section 101(31). However, out of an abundance of caution, the Debtors have included on Exhibit F—solely with respect to disclosures relating to bonuses and any non-ordinary course payments—all employees with a title of vice president. The Debtors define "directors" as members of the board of directors of the Debtors. The Debtors have not performed a detailed review of the transfers to determine whether legal actions for recovery of any or all such transfers as preferences under Bankruptcy Code Section 547 are warranted. However, based on the Debtors' preliminary review, it appears that many of the recipients of transfers have valid defenses to any preference action under Bankruptcy Code Section 547. The Liquidating Trustee will perform a detailed analysis of such transfers to determine whether legal actions for recovery of any or all such transfers as preferences under Bankruptcy Code Section 547 are warranted. ALTHOUGH ALL RIGHTS OF THE LIQUIDATING TRUST ARE RESERVED, SUBJECT TO THE PLAN'S RELEASES, THE DEBTORS WOULD CAUTION CREDITORS AGAINST ANY EXPECTATION OF MATERIAL PREFERENCE RECOVERIES ON ACCOUNT OF SUCH TRANSFERS.

The Liquidation Analysis attached hereto as <u>Exhibit D</u> assumes potential recoveries from the pursuit of preference actions will be between \$1,400,000 and \$2,100,000 whether pursued by a Chapter 7 Trustee in a Chapter 7 liquidation or by the Liquidating Trustee under the Plan.

As set forth in Sections IV.N and IV.O above, and as described in further detail below, the Plan provides for a release of all Causes of Action, including Avoidance Actions but excluding fraud, willful misconduct or gross negligence as determined by a Final Order, against, among others, officers, directors, shareholders, members and/or enrollees, employees, representatives, advisors, attorneys, financial advisors, investment bankers, or agents of the Debtors (including their estates and beneficiaries), in each case as of the Petition Date or that have become officers and/or directors thereafter, including GTCR (as defined below). Absent confirmation of the Plan, such Causes of Action would not be released other than as set forth below in connection with the GTCR Settlement (as defined below). Except to the extent any such Causes of Action are released under the Plan, the Debtors and the Liquidating Trustee reserve their right to conduct a further review of the transfers made within 90 days (or in the case of insiders, one year) of the Petition Date and to seek recovery of such transfers under the provisions of the Bankruptcy Code before or after the Confirmation Date.

Prior to the Petition Date, Graceway Holdings, LLC made a series of distributions in a total amount in excess of \$362,616,953.68 (the "Distributions") including, but not limited to. dividend payments and tax distributions, to the owners of the equity interests of Graceway Holdings, LLC, including GTCR, pursuant to Section 4.1 of the Graceway Holdings, LLC Amended and Restated Limited Liability Company Agreement dated as of December 29, 2006. One of such Distributions was a transfer by the Debtors on or about March 18, 2010 of approximately \$9,127,166 to approximately 17 shareholders on account of certain tax liabilities of such shareholders resulting from the Debtors' status as a limited liability company. A list of such transfers is attached hereto as Exhibit G. The Debtors may have been insolvent when such transfers were made and may not have received reasonably equivalent value for such transfers. Accordingly, the Debtors may have been able to bring a legal action against these shareholders to have this last Distribution returned as a fraudulent conveyance under Sections 544 and/or 548 of the Bankruptcy Code. However, GTCR and the other shareholders who received this last Distribution would have vigorously disputed whether the Debtors were insolvent at the time of the Distribution and whether reasonably equivalent value was given for the Distribution, because the same agreement pursuant to which the shareholders made their initial equity contributions to the Graceway Holdings, LLC provided that tax distribution would be made to the shareholders. Other than with respect to the March 18, 2010 Distribution, the Debtors, after due investigation. are unaware of any valid Causes of Action, against any officer, director, shareholder, member and/or enrollee, employee, representative, advisor, attorney, financial advisor, investment banker, or agent of the Debtors, including GTCR.

On February 14, 2012, the Debtors and (i) GTCR Golder Rauner, L.L.C.; (ii) GTCR Co-Invest II, L.P.; (iii) GTCR Co-Invest III, L.P.; (iv) GTCR Fund VIII, L.P.; (v) GTCR Fund IX/A, L.P.; (vi) Fund VIII/B Graceway Splitter, L.P.; (vii) Fund IX/B Graceway Splitter, L.P.; and (viii) GTCR Golder Rauner II, L.L.C. (each a "GTCR Entity" and, collectively, including any affiliates thereof, "GTCR"), entered into a Settlement Agreement (the "GTCR Settlement Agreement") attached as Exhibit B to the Debtors' Motion for an Order

Authorizing the Debtors to Enter Into a Settlement Agreement with GTCR filed on February 14, 2012 [Docket No. 524] (the "GTCR Settlement Motion"). Pursuant to the GTCR Settlement Agreement, the Debtors will release all Causes of Action, including Avoidance Actions, other than claims or liabilities arising out of or relating to any act or omission that constitutes fraud or willful misconduct as determined by final order of a court of competent jurisdiction, against (a) each GTCR Entity and its respective current or former officers, directors, advisors, shareholders, members and/or enrollees and employees (and each of the foregoing entity's or person's respective successors, assigns and representatives) and (b) any other person or entity not otherwise identified in the preceding sub-clause (a) that directly or indirectly has received or in the future may receive any portion of the Distributions (other than any person that is an officer or director of the Debtors as of the Settlement Effective Date (as defined below), in such person's capacity as an officer or director of the Debtors). In exchange, within one (1) business day of the Settlement Effective Date, GTCR will (a) pay \$4,500,000.00 directly to the First Lien Facility Agent for the benefit of the First Lien Claimholders; and (b) pay \$1,500,000.00 directly to the Debtors' estates, which amount shall constitute Other Assets under the Plan.

The GTCR Settlement Agreement will become effective and binding upon the first date on which the Bankruptcy Court enters the order approving the GTCR Settlement Motion and such order becomes final and non-appealable (such date, the "Settlement Effective Date"). The GTCR Settlement Motion is currently set to be heard at the hearing scheduled for March 16, 2012 at 9:30 a.m. before the Honorable Peter J. Walsh, Judge for the United States Bankruptcy Court for the District of Delaware, located at the United States Bankruptcy Court, 824 Market Street, Wilmington, Delaware 19801.

As set forth in Sections IV.N and IV.O above, as a result of the GTCR Settlement Agreement as well as such persons' or entities' material contributions to the success of the Debtors' Chapter 11 Cases, the Plan provides releases of Causes of Action against, among others, the officers, directors, shareholders, members and/or enrollees, employees, representatives, advisors, attorneys, financial advisors, investment bankers, or agents of the Debtors (including their estates and beneficiaries), in each case as of the Petition Date or that have become officers and/or directors thereafter, including GTCR. From the outset of the Chapter 11 Cases, certain First Lien Claimholders agreed that, upon confirmation of an acceptable Plan the foregoing parties other than GTCR should be entitled to a release. During Plan negotiations, representatives of both the Second Lien Facility Lenders and the Creditors' Committee agreed that in light of the significant increase in the value of the Estates brought about by the efforts of the foregoing parties during these Chapter 11 Cases, the foregoing parties other than GTCR should be entitled to a release. In addition, in light of the significant cash consideration being provided by GTCR under the GTCR Settlement Agreement, representatives of the First Lien Claimholders, the Second Lien Facility Lenders and the Creditors' Committee agreed that the release in the Plan should be expanded to include GTCR.

D. Nycomed Litigation

The Nycomed Litigation was sold to the Purchaser pursuant to the Asset Purchase Agreement and the Sale Order. However, pursuant to Section 8.10 of the Asset Purchase Agreement, if the Purchaser receives any proceeds, judgments, payments, expense reimbursements, property or rights (without regard to how any such items are characterized by a

court, arbitrator, agreement or the parties to the Nycomed Litigation and specifically including, but not limited, to rights to future royalties or revenue streams, license or cross-licensing rights, marketing rights, services, or any other rights the value of which can be reasonably ascertained or estimated) relating to or arising from the final resolution of all or any portion of the Nycomed Litigation, the value of such proceeds, judgments, payments, expense reimbursements, property or rights (the "Nycomed Award Amount"), such Nycomed Award Amount shall be distributed as follows:

- (a) first, any such proceeds up to an aggregate amount of \$3,000,000.00 shall be transferred to the Debtors in cash by wire transfer of immediately available funds;
- (b) second, the Purchaser shall retain any additional proceeds up to the amount of reasonable, documented out-of-pocket costs, fees and expenses incurred by the Purchaser and its affiliates in connection with the Nycomed Litigation;
- (c) third, the Purchaser shall transfer any additional proceeds to the Debtors, in cash by wire transfer of immediately available funds, up to an amount, when taken together with the \$3,000,000.00 of proceeds transferred to the Debtors pursuant to subsection (a) above, that equals the amount of reasonable, documented out-of-pocket costs, fees and expenses incurred by the Debtors prior to the Closing in connection with the Nycomed Litigation (which shall in no event exceed \$9,000,000); and
- (d) fourth, the Purchaser shall (i) retain 50% of any additional proceeds and (ii) transfer the remaining 50% of such proceeds to the Debtors in cash by wire transfer of immediately available funds.

Pursuant to the Asset Purchase Agreement, within ten (10) Business Days after indefeasible receipt by the Purchaser of any funds constituting the Debtors' portion of the Nycomed Award Amount, the Purchaser will pay to the Debtors in cash by wire transfer of immediately available funds an amount equal to the amount apportioned to the Debtors as described above (or the applicable portion thereof). To the extent all or any portion of the Nycomed Award Amount may be payable to the Purchaser in future increments, the Purchaser and the Debtors agreed pursuant to the Asset Purchase Agreement to negotiate in good faith appropriate arrangements for the payment to the Debtors of the Debtors' portion of such delayed payments either (x) over time as such amounts are received by the Purchaser or (y) as a single lump sum of the agreed net present value of such payment streams. For the avoidance of doubt, Purchaser retains full control over the defense of the Nycomed Litigation and any decisions related thereto, including any potential settlements, and the Debtors are obligated to cooperate with the Purchaser in connection with such defense under the Asset Purchase Agreement. The Purchaser is obligated under the Asset Purchase Agreement to keep the Debtors informed of any material developments in the Nycomed Litigation including any settlement offers and acceptances. The Purchaser is further obligated to provide the Debtors reasonably detailed information, including copies of any related settlement agreement or license agreement as well as supporting documents and materials, related to any resolution (whether or not appealable) of the matters related to the Nycomed Litigation, and the Purchaser is required to allow the Debtors reasonable access to employees and outside counsel of the Purchaser to resolve any questions or ambiguities related to such information.

The Nycomed Litigation is included in the Prepetition Collateral and is subject to the Liens of the Debtors' Prepetition Secured Lenders. As a result, any recovery on account of the Nycomed Award Amount would similarly be included in the Prepetition Collateral and be subject to the Liens of the Debtors' Prepetition Secured Lenders. Additionally, the Nycomed Award Amount would also be subject to the Allowed First Lien Facility Adequate Protection Claim and the Allowed Second Lien Facility Adequate Protection Claim, if any. As discussed in Section IV.E above, on the Effective Date the Debtors will assign and transfer all of their rights to and interests in Nycomed Litigation Proceeds to the First Lien Facility Agent subject to any Liens thereon, which Liens shall remain valid, binding and perfected to the same extent as on the Petition Date on and after the Effective Date.

Whether the Debtors will be entitled to any recovery on account of the Nycomed Award Amount is uncertain and speculative to value. ALTHOUGH ALL RIGHTS TO THE NYCOMED AWARD AMOUNT, IF ANY, ARE TRANSFERRED TO THE FIRST LIEN FACILITY AGENT PURSUANT TO THE PLAN, THE DEBTORS WOULD CAUTION CREDITORS AGAINST ANY EXPECTATION OF MATERIAL RECOVERIES ON ACCOUNT OF THE NYCOMED AWARD AMOUNT. MATERIAL RECOVERIES ON ACCOUNT OF THE NYCOMED AWARD AMOUNT ARE NOT GUARANTEED AND CREDITORS SHOULD NOT RELY ON ANY EXPECTATION OF RECOVERIES ON ACCOUNT OF THE NYCOMED AWARD AMOUNT.

E. The Debtors' Remaining Property

Certain of the Debtors' property was not included in the assets sold to the Purchaser in the Asset Sale (the "<u>Acquired Assets</u>"). The assets sold to the Purchaser pursuant to the Asset Purchase Agreement did not include, among other things (collectively, the "<u>Excluded Assets</u>"):

- (a) all cash and cash equivalents, including checks, commercial paper, treasury bills, certificates of deposit and other bank deposits, securities, securities entitlements, instruments and other investments of the Debtors and all bank accounts and securities accounts, including any cash collateral that is collateralizing any letters of credit;
- (b) any refunds, rebates or credits of the Debtors on account of any federal, state, provincial, local, municipal, foreign or other income, alternative, minimum, add-on minimum, accumulated earnings, personal holding company, franchise, capital stock, net worth, capital, profits, intangibles, windfall profits, gross receipts, value added, sales, use, goods and services, excise, customs duties, transfer, conveyance, mortgage, registration, stamp, documentary, recording, premium, severance, environmental (including taxes under Section 59A of the Internal Revenue Code of 1986, as amended), natural resources, real property, personal property, ad valorem, intangibles, rent, occupancy, license, occupational, employment, unemployment insurance, social security, disability, workers' compensation, payroll, health care, withholding, estimated or other similar tax, duty, levy or other governmental charge or assessment or deficiency thereof (including all interest and penalties thereon and additions thereto), in each case imposed by any United States or Canadian federal, provincial, state, municipal or local or any foreign government, governmental agency or authority, or regulatory or administrative

authority, or any court, tribunal or judicial body having jurisdiction, including the Bankruptcy Court and the Canadian Court;

- (c) except to the extent relating to the Acquired Assets (which included the Nycomed Litigation), the Debtors' rights in (i) all Claims, Causes of Action, choses in action, rights of recovery and rights of set-off of whatever kind or description and (ii) any action, arbitration, audit, hearing, investigation, litigation, or suit (whether civil, criminal, administrative or investigative) commenced, brought, conducted, or heard by or before, or otherwise involving, any United States or Canadian federal, provincial, state, municipal or local or any foreign government, governmental agency or authority, or regulatory or administrative authority, or any court, tribunal or judicial body having jurisdiction, including the Bankruptcy Court and the Canadian Court;
- (d) any security deposits or pre-paid expenses not associated with the Acquired Assets;
- (e) all insurance policies and binders, all Claims, refunds and credits from insurance policies or binders due or to become due with respect to such policies or binders and all rights to proceeds thereof other than those related to the Acquired Assets;
- (f) all furniture, trade fixtures, equipment, computers, servers, telephones, laptop computers, machinery, apparatus, appliances, implements, signage, office supplies and all other tangible personal property of every kind and description owned by the Debtors ("Equipment") other than Equipment physically located at the location of a party that is neither a Debtor nor an affiliate of a Debtor and that manufactures products, formulations and compounds previously owned by the Debtors, or to which the Debtors had any rights;
- (g) all trade accounts receivable and other rights to payment from customers of the Debtors to the extent arising out of the Debtors' business;
- (h) all real property owned by the Debtors and all unexpired leases or other occupancy agreements for real property under which the Debtors are a lessee (or the equivalent); and
- (i) any assets, properties and rights of any of the Debtors other than the Acquired Assets.

The Debtors have liquidated certain of the Excluded Assets since the Closing Date. The Debtors believe that the remaining unliquidated Excluded Assets have a value between approximately \$26,300,000.00 and \$42,200,000.00. Certain of the Excluded Assets are included in the Prepetition Collateral and are subject to the Liens of the Debtors' Prepetition Secured Lenders. As a result, the proceeds received in a liquidation of such Excluded Assets would similarly be included in the Prepetition Collateral and be subject to the Liens of the Debtors' Prepetition Secured Lenders. Additionally, all of the Excluded Assets are also subject to the Allowed First Lien Facility Adequate Protection Claim and the Allowed Second Lien Facility Adequate Protection Claim, if any.

F. Feasibility

Section 1129(a)(11) of the Bankruptcy Code requires that the Debtors be able to perform their obligations under the Plan. For purposes of determining whether the Plan meets this requirement, the Debtors analyzed their ability to meet their obligations under the Plan. The Debtors believe that they have adequate funding to be able to meet their obligations under the Plan.

The Debtors believe that the funds held in the Reserve Accounts will be sufficient to pay fees and expenses of Professionals retained by the Debtors and the Creditors' Committee. The Debtors also believe that the funds held in the Reserve Accounts, as well as the Other Assets, will be sufficient to pay all administrative expenses and to otherwise provide the distributions to Creditors contemplated by the Plan.

Moreover, as set forth in the Liquidation Analysis attached hereto as <u>Exhibit D</u> and described in Article VI.B. above, the Debtors believe that the recovery for all of the Debtors' creditors, including Class 5 General Unsecured Creditors, will be the same or better under the Plan than it would be pursuant to a liquidation under chapter 7 of the Bankruptcy Code.

G. Risk Factors Associated with the Plan

Holders of Claims against the Debtors should read and consider carefully the information set forth below, as well as the other information set forth in this Disclosure Statement (and the documents delivered together herewith and/or incorporated by reference), prior to voting to accept or reject the Plan. This information, however, should not be regarded as the only risks involved in connection with the Plan and its implementation.

The Class 5 General Unsecured Claims are subject to the risk of dilution if the total amount of Claims is higher than the Debtors' estimate. A number of Disputed Claims are material and the total amount of all Claims, including Disputed Claims, is materially in excess of the total amount of Allowed Claims assumed in calculating the estimated distributions set forth in Section I above. Accordingly, the amount of any distribution that will ultimately be received by any particular holder of a Class 5 General Unsecured Claim may be adversely affected by the aggregate amount of all Allowed Claims. Consequently, distributions to holders of Class 5 General Unsecured Claims will be made on an incremental basis until all Disputed Claims have been resolved.

A substantial amount of time may elapse between the Effective Date and the receipt of distributions, including, but not limited to, a Final Distribution, under the Plan for certain holders of Claims because of the time required to achieve recovery of certain assets. To the extent that distributions under the Plan are derived, in whole or in part, from recoveries on the Causes of Action prosecuted by the Liquidating Trustee, there can be no assurance that any such Causes of Action will produce recoveries that will provide sufficient funds for such distributions to be made by the Liquidating Trust.

If the Bankruptcy Court were not to grant the Debtors' request for substantive consolidation of the Debtors with respect to the voting and treatment of all Claims and Interests other than Other Secured Claims, Confirmation and consummation of the Plan, if still possible,

could be substantially more burdensome, time consuming, and costly to the Debtors' Estates. As stated above, the Debtors believe that substantive consolidation of the Debtors' Estates for purposes of the voting and treatment of all Claims and Interests other than Other Secured Claims will facilitate implementation of the Plan and foster similarity and fairness of treatment among holders of Claims. There can be no assurance, however, that the Bankruptcy Court will reach the same conclusion.

Even if all Classes entitled to vote accept the Plan, the Plan might not be confirmed by the Bankruptcy Court. Section 1129 of the Bankruptcy Code sets forth the requirements for confirmation and requires, among other things, that the confirmation of a plan of reorganization is not likely to be followed by the liquidation or the need for further financial reorganization unless, as here, such liquidation or reorganization is proposed in the Plan, and that the value of distributions to dissenting Creditors and equity security holders not be less than the value of distributions such Creditors and equity security holders would receive if the debtor was liquidated under chapter 7 of the Bankruptcy Code. The Debtors believe that the Plan satisfies all the requirements for confirmation of a plan of reorganization under the Bankruptcy Code. There can be no assurance, however, that the Bankruptcy Court will also conclude that the requirements for Confirmation of the Plan have been satisfied.

One of the requirements for confirmation of a plan under Section 1129 of the Bankruptcy Code is that all Allowed Administrative Expense Claims be paid in full in Cash. The Debtors believe that the CCR Account contains sufficient Cash to pay any and all Allowed Administrative Expense Claims relating to chargebacks, channel management agreements, product returns and rebates. However, because Claims for product returns are contingent and may be substantial, the Debtors ultimately may not have sufficient Cash to pay any and all Allowed Administrative Expense Claims relating to chargebacks, channel management agreements, product returns and rebates.

Additionally, successful Confirmation of the Plan is subject to satisfaction or waiver of the conditions to Plan effectiveness, which are discussed in detail above. Not all of these conditions are within the control of the Debtors. THUS, THERE CAN BE NO ASSURANCE THAT ALL OF THE VARIOUS CONDITIONS TO EFFECTIVENESS OF THE PLAN WILL BE TIMELY SATISFIED OR WAIVED.

VII. ALTERNATIVES TO THE PLAN

Although this Disclosure Statement is intended to provide information to assist a Claim or Interest holder in determining whether to vote for or against the Plan, a summary of the alternatives to Confirmation of the Plan may be helpful.

If the Plan is not confirmed with respect to any of the Debtors, the following alternatives are available: (a) confirmation of another chapter 11 plan; (b) conversion of the Chapter 11 Cases to cases under chapter 7 of the Bankruptcy Code; or (c) dismissal of the Chapter 11 Cases leaving Creditors and Interest holders to pursue available non-bankruptcy remedies. These alternatives to the Plan are very limited and not likely to benefit Creditors. Although the Debtors could theoretically file a new plan, the most likely result if the Plan is not

confirmed is that the Chapter 11 Cases will be converted to cases under chapter 7 of the Bankruptcy Code. The Debtors believe that conversion of the Chapter 11 Cases to chapter 7 cases would result in (i) significant delay in distributions to all Creditors who would have received a distribution under the Plan and (ii) diminished recoveries for certain Classes of Creditors. If the Chapter 11 Cases are dismissed, Creditors would be free to pursue non-bankruptcy remedies in their attempts to satisfy Claims against the Debtors. However, in that event, Creditors would be faced with the costs and difficulties of attempting, each on its own, to collect Claims from a non-operating entity. In particular, there is no guarantee that Administrative Expense Claims could be paid in such a scenario because there may be substantial contingent return Claims.

VIII. CERTAIN U.S. FEDERAL INCOME TAX CONSIDERATIONS

A. In General

The following discussion summarizes certain material U.S. federal income tax consequences expected to result from the consummation of the Plan. This discussion is based on current provisions of the Internal Revenue Code of 1986, as amended (the "IRC"), applicable Treasury Regulations, judicial authority and current administrative rulings and pronouncements of the Internal Revenue Service (the "Service"). There can be no assurance that the Service will not take a contrary view or that such a contrary view would not be sustained by a court, no ruling from the Service has been or will be sought nor will any counsel provide a legal opinion as to any of the expected tax consequences set forth below.

Legislative, judicial or administrative changes or interpretations may be forthcoming that could alter or modify the statements and conclusions set forth herein. Any such changes or interpretations may or may not be retroactive and could affect the tax consequences to the beneficial owners of Claims (each a "Holder" and collectively, the "Holders"), the Liquidating Trust, the Debtors or any entity the ownership of which is transferred to the Liquidating Trust. It cannot be predicted at this time whether any tax legislation will be enacted or, if enacted, whether any tax law changes contained therein would affect the tax consequences described herein.

The following summary is for general information only. The tax treatment of a Holder may vary depending upon such Holder's particular situation. This summary does not address all of the tax consequences that may be relevant to a Holder, including any alternative minimum tax consequences, and does not address the tax consequences to a Holder that has made an agreement to resolve its Claim in a manner not explicitly provided for in the Plan. This summary also does not address the U.S. federal income tax consequences to persons not entitled to vote on the Plan or Holders subject to special treatment under the U.S. federal income tax laws, such as brokers or dealers in securities or currencies, certain securities traders, tax-exempt entities, financial institutions, insurance companies, foreign persons, partnerships and other pass-through entities, Holders that hold Claims as a position in a "straddle" or as part of a "synthetic security," "hedging," "conversion" or other integrated transaction, Holders that have a "functional currency" other than the United States dollar and Holders that have acquired Claims in connection with the performance of services. The following summary assumes that the

Claims are held by Holders as "capital assets" within the meaning of Section 1221 of the IRC and that all Claims denominated as indebtedness are properly treated as debt for U.S. federal income tax purposes.

The tax treatment of Holders and the character, amount and timing of income, gain or loss recognized as a consequence of the Plan and the distributions provided for by the Plan may vary, depending upon, among other things: (i) whether the Claim (or portion thereof) constitutes a Claim for principal or interest; (ii) the type of consideration received by the Holder in exchange for the Claim and whether the Holder receives distributions under the Plan in more than one taxable year; (iii) whether the Holder is a citizen or resident of the United States for tax purposes, is otherwise subject to U.S. federal income tax on a net basis, or falls into any special class of taxpayers, such as those that are excluded from this discussion as noted above; (iv) the manner in which the Holder acquired the Claim; (v) the length of time that the Claim has been held; (vi) whether the Claim was acquired at a discount; (vii) whether the Holder has taken a bad debt deduction with respect to the Claim (or any portion thereof) in the current or prior years; (viii) whether the Holder has previously included in income accrued but unpaid interest with respect to the Claim; (ix) the method of tax accounting of the Holder; (x) whether the Claim is an installment obligation for U.S. federal income tax purposes; and (xi) whether the "market discount" rules are applicable to the Holder. Therefore, each Holder should consult its tax advisor for information that may be relevant to its particular situation and circumstances, and the particular tax consequences to such Holder of the transactions contemplated by the Plan.

THE FOLLOWING DISCUSSION IS INTENDED ONLY AS A SUMMARY OF CERTAIN U.S. FEDERAL TAX CONSEQUENCES OF THE PLAN AND IS NOT A SUBSTITUTE FOR CAREFUL TAX PLANNING WITH A TAX PROFESSIONAL. THE FOLLOWING DISCUSSION IS FOR INFORMATION PURPOSES ONLY AND IS NOT TAX ADVICE. THE TAX CONSEQUENCES ARE IN MANY CASES UNCERTAIN AND MAY VARY DEPENDING ON A HOLDER'S PARTICULAR CIRCUMSTANCES. ACCORDINGLY, EACH HOLDER IS STRONGLY URGED TO CONSULT ITS TAX ADVISOR REGARDING THE U.S. FEDERAL, STATE, LOCAL, AND APPLICABLE NON-U.S. INCOME AND OTHER TAX CONSEQUENCES OF THE PLAN.

TO COMPLY WITH INTERNAL REVENUE SERVICE CIRCULAR 230, TAXPAYERS ARE HEREBY NOTIFIED THAT (A) ANY DISCUSSION OF U.S. FEDERAL TAX ISSUES CONTAINED OR REFERRED TO IN THIS DISCLOSURE STATEMENT (INCLUDING ANY ATTACHMENTS) IS NOT INTENDED OR WRITTEN TO BE USED, AND CANNOT BE USED, BY ANY TAXPAYER FOR THE PURPOSE OF AVOIDING PENALTIES THAT MAY BE IMPOSED ON A TAXPAYER UNDER THE INTERNAL REVENUE CODE, (B) ANY SUCH DISCUSSION IS WRITTEN IN CONNECTION WITH THE PROMOTION OR MARKETING OF THE TRANSACTIONS OR MATTERS ADDRESSED HEREIN, AND (C) TAXPAYERS SHOULD SEEK ADVICE BASED ON THEIR PARTICULAR CIRCUMSTANCES FROM AN INDEPENDENT TAX ADVISOR.

B. U.S. Federal Income Tax Consequences to the Debtors

Because Graceway Holdings, LLC, Graceway Pharmaceuticals, LLC, Chester Valley Holdings, LLC and Chester Valley Pharmaceuticals, LLC are disregarded for federal income tax purposes, their assets and liabilities are treated as assets and liabilities of Graceway Pharma Holding Corp. ("GPHC") for federal income tax purposes. In addition, Graceway Canada Company is disregarded for federal income tax purposes and thus its assets and liabilities are treated as assets and liabilities of Graceway Canada Holdings, Inc. ("GCHI"). GPHC, GCHI and Graceway International Inc. are the members of a federal consolidated tax group (the "Graceway U.S. Tax Group") and file a consolidated federal income tax return. In this discussion, references to the Debtors and/or a Debtor generally means the Graceway U.S. Tax Group and/or a member of such Group, respectively.

If there is a discharge of a debt obligation by a debtor (in the case of indebtedness with multiple obligors, indebtedness that is allocable to such debtor) for an amount less than the adjusted issue price (in most cases, the amount the debtor received on incurring the obligation, with certain adjustments), such discharge generally would give rise to cancellation of debt ("COD") income, which may have to be included in the debtor's income. However, the Debtors should be able to utilize special tax provisions which exclude from income debts discharged in a chapter 11 case or debts discharged when a debtor is insolvent, with the insolvency exception only applying to the extent of the debtor's insolvency (the "COD Exceptions").

Under Section 108(b) of the IRC and Treasury Regulations that apply to members of a consolidated group, each Debtor that excludes COD income from gross income under the COD Exceptions will be required to reduce certain tax attributes, including consolidated attributes, such as consolidated net operating losses and net operating loss carryforwards ("NOLs") and certain other losses, credits and carryforwards, if any, attributable to such Debtor, attributes that arose in separate return limitation years of such Debtor (if any), and the Debtor's tax basis in its assets (but not below the amount of its liabilities remaining immediately after the discharge of indebtedness), in an amount generally equal to the amount of such Debtor's COD income excluded from income under the COD Exceptions. A "look-through rule" applies when asset basis reduction reduces the basis of stock of another member of the consolidated group and requires corresponding adjustments to be made to the attributes attributable to the lower-tier member.

As a result of the required attribute reduction resulting from the discharge of indebtedness, the Debtors believe that a significant portion of NOLs (and alternative minimum tax NOLs) of the Debtors will be eliminated after consummation of the Plan. Because the Debtors are liquidating rather than continuing to operate in reorganized form, and because substantially all of the Debtors' Assets will be transferred to the Liquidating Trust, any remaining NOLs allocable to the Debtors are not expected to have material value.

The 363 Sale, the transfer of the Debtors' assets to the Liquidating Trust and the liquidation of the Debtors may trigger income or gain recognition by the Debtors. However, the Debtors' existing NOLs and capital losses (prior to being reduced as a result of any attribute reduction) should generally first be available to offset any such income or gain (with any capital losses available to only offset capital gains). The Debtors currently do not anticipate owing any

material amount of regular U.S. federal income taxes with respect to taxable years ending after the Petition Date. If, however, the Service were to prevail in assessing U.S. federal income tax for any of these years or for tax years ending prior to the Petition Date, payments of such taxes could reduce the amounts otherwise available for distribution under the Plan.

A corporation or a consolidated group of corporations may incur alternative minimum tax ("AMT") liability even where a NOL is generated for regular corporate income tax purposes or where NOL carryovers and certain other tax attributes are sufficient to eliminate taxable income as computed under the regular corporate income tax. In general, the AMT is imposed on a corporation's alternative minimum taxable income at a twenty percent (20%) rate to the extent such tax exceeds the corporation's regular U.S. federal income tax. For purposes of computing taxable income for AMT purposes, certain tax deductions and other beneficial allowances allowed in computing a corporation's regular U.S. federal income tax are modified or eliminated. In particular, even though a corporation otherwise might be able to offset all of its taxable income for regular tax purposes by available NOL carryforwards, a portion of a corporation's taxable income for AMT purposes may not be offset by available NOL carryforwards (as computed for AMT purposes). Although it is possible that the Debtors could be liable for the AMT, at this time the Debtors do not expect to incur a material amount of AMT.

C. U.S. Federal Income Tax Treatment of the Liquidating Trust

Except with respect to the Disputed Claims Reserve, it is intended that the Liquidating Trust will be treated as one or more "grantor trusts" for U.S. federal income tax purposes. In general, a grantor trust is not a separate taxable entity. The Service, in Revenue Procedure 94-45, 1994-2 C.B. 684, set forth the general criteria for obtaining an advanced ruling as to the grantor trust status of a liquidating trust under a chapter 11 plan. Consistent with the requirements of Revenue Procedure 94-45, the Liquidation Trust Agreement requires all relevant parties to treat, for federal income tax purposes, the transfer of the Debtors' assets to the Liquidating Trust as (i) a transfer of such assets (net of any applicable liabilities) to the Beneficiaries of the Liquidating Trust (to the extent of the value of their respective interests in the applicable Trust Assets) followed by (ii) a transfer of such assets (net of any applicable liabilities) by such Beneficiaries to the Liquidating Trust (to the extent of the value of their respective interests in the applicable Trust Assets), with the Beneficiaries being treated as the grantors and owners of the Liquidating Trust. Each Beneficiary of the Liquidating Trust will generally recognize gain (or loss) in its taxable year that includes the Effective Date in an amount equal to the difference between the amount realized in respect of its Claim and its adjusted tax basis in such Claim. The amount realized for this purpose should generally equal the amount of cash and the fair market value of any other assets (net of any applicable liabilities) received or deemed received for U.S. federal income tax purposes under the Plan in respect of such Holder's Claim. A Holder that is deemed to receive for U.S. federal income tax purposes a non-cash asset under the Plan in respect of its Claim should generally have a tax basis in such asset in an amount equal to the fair market value of such asset on the date of its deemed receipt.

The Plan and the Liquidating Trust Agreement generally provide that the Beneficiaries of the Liquidating Trust must value the assets of the Liquidating Trust consistently with the values determined by the Liquidating Trustee for all U.S. federal, state, local and foreign income tax purposes. As soon as possible after the Effective Date, but in no event later

than ninety (90) days thereafter, the Liquidating Trustee, based upon his good faith determination after consultation with his counsel, shall inform the Beneficiaries in writing as to his estimate of the value of the Assets transferred to the Liquidating Trust and the value of such assets allocable to each Class of Beneficiaries.

Consistent with the treatment of the Liquidating Trust (except with respect to the Disputed Claims Reserve) as one or more grantor trusts, the Liquidating Trust Agreement will require each Beneficiary to report on its U.S. federal income tax return its allocable share of the Liquidating Trust's income. Therefore, a Beneficiary may incur a U.S. federal income tax liability with respect to its allocable share of the income of the Liquidating Trust whether or not the Liquidating Trust has made any distributions to such Beneficiary. The character of items of income, gain, deduction, and credit to any Beneficiary and the ability of such Beneficiary to benefit from any deduction or losses will depend on the particular situation of such Beneficiary.

A distribution of underlying assets from the Liquidating Trust to a Beneficiary (other than in respect of distributions attributable to a reduction in the Disputed Claims Reserve) will generally not be taxable to such Holder because such Holders are already regarded for U.S. federal income tax purposes as owning such assets. Holders are urged to consult their tax advisors regarding the appropriate U.S. federal income tax treatment of distributions from the Liquidating Trust.

Except with respect to the Disputed Claims Reserve, the Liquidating Trustee will file with the Service tax returns for the Liquidating Trust as one or more grantor trusts pursuant to Treasury Regulation Section 1.671-4(a) and will also send to each Holder a separate statement setting forth such Holder's share of items of Trust income, gain, loss, deduction, or credit. Each such Holder will be required to report such items on its U.S. federal income tax return. The Liquidating Trustee shall determine for tax reporting purposes whether to treat the Beneficiaries as grantors in a single grantor trust or as grantors in two distinct grantor trusts. It is anticipated that such reporting will not have a material affect on the income, gain and loss recognition, for federal income tax purposes, of the Beneficiaries.

The discussion above assumes that, except with respect to the Disputed Claims Reserve, the Liquidating Trust will be respected as one or more grantor trusts for U.S. federal income tax purposes. If the Service were to challenge successfully such classification, the U.S. federal income tax consequences to the Liquidating Trust and the Beneficiaries could differ materially from those discussed herein (including the potential for an entity level tax to be imposed on all income of the Liquidating Trust).

D. Disputed Claims Reserve

Until such time as all of the beneficial interests in the Liquidating Trust can be distributed to the Holders in accordance with the terms of the Plan, the Disputed Claims Reserve will be treated as owning a portion of the assets in the Liquidating Trust. Distributions from the Disputed Claims Reserve will be made to Holders of Disputed Claims after such Claims are subsequently Allowed and to other Beneficiaries when Disputed Claims are subsequently disallowed. The Liquidating Trust shall file all income tax returns with respect to any income attributable to the Disputed Claims Reserve and shall pay the federal, state and local income

taxes attributable to the Disputed Claims Reserve, based on the items of income, deduction, credit or loss allocable thereto.

Holders should note the tax treatment of the Disputed Claims Reserve is unclear and should consult their tax advisors as to the tax consequences to them of the establishment of, the income on, and distributions from, the Disputed Claims Reserve

E. U.S. Federal Income Tax Consequences to Holders of Claims

1. Holders of Allowed Claims in Classes 2, 3 and 5.

Although not free from doubt, Holders of Allowed Claims in Class 2, Class 3 and Class 5 as of the Effective Date should be treated as receiving from the Debtors their respective shares of the applicable assets (net of any applicable liabilities) of the Liquidating Trust remaining (other than any assets allocated to the Disputed Claims Reserve) after distributions to Administrative Expense Claims, Priority Tax Claims, Class 1 Claims and Class 4 Claims in satisfaction of their Allowed Claims, and simultaneously transferring such assets (net of any applicable liabilities) to the Liquidating Trust. Accordingly, a Holder of such Claim should generally recognize gain or loss in an amount equal to the amount deemed realized on the Effective Date (as described above) less its adjusted tax basis of its Claim. Additionally, such Holders should generally recognize their allocable share of income, gain, loss and deductions recognized by the Liquidating Trust on an annual basis.

Because a Holder's ultimate share of the assets of the Liquidating Trust based on its Allowed Class 2 Claim, Allowed Class 3 Claim or Allowed Class 5 Claim will not be determinable on the Effective Date due to, among other things, the existence of Disputed Claims and the value of the assets at the time of actual receipt not being ascertainable on the Effective Date, such Holder should recognize additional or offsetting gain or loss if, and to the extent that, the aggregate amount of cash and fair market value of the assets of the Liquidating Trust ultimately received by such Holder is greater than or less than the amount used in initially determining gain or loss in accordance with the procedures described in the preceding paragraph. It is unclear when a Holder of an Allowed Class 2 Claim, an Allowed Class 3 Claim or an Allowed Class 5 Claim should recognize, as an additional amount received for purposes of computing gain or loss, an amount attributable to the disallowance of a Disputed Claim.

The character of any gain or loss as capital gain or loss or ordinary income or loss and, in the case of capital gain or loss, as short-term or long-term, will depend on a number of factors, including: (i) the nature and origin of the Claim; (ii) the tax status of the Holder of the Claim; (iii) whether the Claim has been held for more than one year; (iv) the extent to which the Holder previously claimed a loss or bad debt deduction with respect to the Claim; and (v) whether the Claim was acquired at a market discount. A Holder that purchased its Claim from a prior Holder at a market discount may be subject to the market discount rules of the IRC. Under those rules (subject to a *de minimis* exception), assuming that such Holder has made no election to accrue the market discount and include it in income on a current basis, any gain recognized on the exchange of such Claim generally would be characterized as ordinary income to the extent of the accrued market discount on such Claim as of the date of the exchange.

It is possible that the Service may assert that any loss should not be recognizable until the Liquidating Trustee makes its final distribution of the assets of the Liquidating Trust. Holders should consult their tax advisors regarding the possibility that the recognition of gain or loss may be deferred until the final distribution of the assets of the Liquidating Trust.

2. Holders of Disputed Claims.

Although not free from doubt, Holders of Disputed Claims should generally recognize gain or loss in an amount equal to the amount deemed realized on the date such Claim becomes an Allowed Claim less the adjusted tax basis of its Claim. Additionally, after such Claims become Allowed Claims, such Holders should generally recognize their allocable share of income, gain, loss and deductions recognized by the Liquidating Trust on an annual basis. However, it is possible that such Holders may be required to recognize the fair market value of such Holder's allocable share of the Liquidating Trust's assets, as an amount received for purposes of computing gain or loss on the Effective Date.

3. <u>Interest Income with respect to Allowed Claims</u>.

Holders of Allowed Claims will be treated as receiving a payment of interest (includible in income in accordance with the Holder's method of accounting for tax purposes) to the extent that any cash or other property received (or deemed received) pursuant to the Plan is attributable to accrued but unpaid interest, if any, on such Allowed Claims. The extent to which the receipt of cash or other property should be attributable to accrued but unpaid interest is unclear. The Debtors and the Liquidating Trust intend to take the position, and the Plan provides, that such cash or property distributed pursuant to the Plan will first be allocable to the principal amount of an Allowed Claim and then, to the extent necessary, to any accrued but unpaid interest thereon. Each Holder should consult its tax advisor regarding the determination of the amount of consideration received under the Plan that is attributable to interest (if any). A Holder generally will be entitled to recognize a loss to the extent any accrued interest was previously included in its gross income and is not paid in full.

F. Backup Withholding and Information Reporting

A Holder of an Allowed Claim may be subject to backup withholding (currently at a rate of 28%) with respect to any "reportable" payments received pursuant to the Plan unless (i) such Holder comes within certain exempt categories and, when required, demonstrates this fact or (ii) provides a correct taxpayer identification number, certifies as to no loss of exemption from backup withholding and otherwise complies with applicable requirements of the backup withholding rules. A Holder who does not provide a correct taxpayer identification number may be subject to penalties imposed by the Service. Amounts withheld under the backup withholding rules may be credited against a Holder's tax liability, and a Holder may obtain a refund of any excess amounts withheld under the backup holding rules by timely filing the appropriate Claim for refund with the Service.

The Liquidating Trustee will report annually to each Holder of an Allowed Class 2 Claim, an Allowed Class 3 Claim and an Allowed Class 5 Claim, and to the Service the

Holder's share of any income, gains and losses of the Liquidating Trust during the calendar year to the extent required by law.

Treasury Regulations generally require disclosure by a taxpayer on its U.S. federal income tax return of certain types of transactions in which the taxpayer participated, including certain transactions that result in the taxpayer recognizing a loss in excess of specified thresholds. Each Holder should consult its tax advisor regarding these regulations and whether the transactions contemplated by the Plan would be subject to these regulations.

THE FOREGOING DISCUSSION OF CERTAIN U.S. FEDERAL INCOME TAX CONSIDERATIONS IS FOR GENERAL INFORMATION PURPOSES ONLY AND IS NOT TAX ADVICE. ACCORDINGLY, EACH HOLDER SHOULD CONSULT ITS TAX ADVISOR WITH RESPECT TO THE TAX CONSEQUENCES OF THE PLAN DESCRIBED HEREIN AND THE APPLICATION OF FEDERAL, STATE, LOCAL AND FOREIGN TAX LAWS. NEITHER THE DEBTORS, THE CREDITORS' COMMITTEE NOR EITHER OF THEIR PROFESSIONALS SHALL HAVE ANY LIABILITY TO ANY PERSON OR HOLDER ARISING FROM OR RELATED TO THE U.S. FEDERAL, STATE, LOCAL OR FOREIGN TAX CONSEQUENCES OF THE PLAN OR THE FOREGOING DISCUSSION.

CONCLUSION

For all of the reasons set forth in this Disclosure Statement, the Debtors believe that Confirmation of the Plan is preferable to all other alternatives. Consequently, the Debtors recommend all holders of Class 2 Claims, Class 3 Claims and Class 5 Claims to vote to **ACCEPT** the Plan, and to complete and return their Ballots so that they will be **RECEIVED** by BMC on or before 4 p.m. (prevailing Eastern time) on April 3, 2012.

Dated: March 1, 2012

Dated:	March 1, 2012			
Gracev	Graceway Pharma Holding Corp.			
Ву:	Name: John A.A. Bellamy Title: EVP			
Gracev	way Pharmaceyticals, LLC			
Ву:	Name: John H.A. Bellamy Title: EVP			
Cheste	r Valley Pharmaceuticals, LLC			
Ву:	Name: Joko A.A. Bellomy Title: EVP			
Gracev	way Canada Holdings, Inc.			
Ву:	Name: John A. A. Bellamy Title: EVP			
Gracev	way International, Inc.			
Ву:	Name: John A.A. Bellamy Title: EVP			

Graceway Holdings, LLC

Chester Valley Holdings, LLC

Name: John A.A. Bellomy Title: EVP

By:

By:

Title:

EXHIBIT A

First Amended Joint Plan of Liquidation

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re:

Chapter 11

GRACEWAY PHARMACEUTICALS, LLC, et al..1

Case No. 11-13036 (PJW)

Jointly Administered

Debtors.

FIRST AMENDED JOINT PLAN OF LIQUIDATION OF GRACEWAY PHARMACEUTICALS, LLC, et al.

David S. Heller, Esq.
Josef S. Athanas, Esq.
Matthew L. Warren, Esq.
LATHAM & WATKINS LLP
Suite 5800
233 South Wacker Drive
Chicago, Illinois 60606
(312) 876-7700

-and-

Michael R. Nestor (No. 3526)
Kara Hammond Coyle (No. 4410)
YOUNG, CONAWAY, STARGATT & TAYLOR, LLP
Rodney Square
1000 North King Street
Wilmington, DE 19801
(302) 571-6600

Attorneys for the Debtors and Debtors-in-Possession

Dated: February 28, 2012

The Debtors in these cases, along with the last four digits of each Debtor's federal tax identification number, are: Graceway Pharma Holding Corp., a Delaware corporation (9175), Case No. 11-13037 (PJW); Graceway Holdings, LLC, a Delaware limited liability company (2502), Case No. 11-13038 (PJW); Graceway Pharmaceuticals, LLC, a Delaware limited liability company (5385), Case No. 11-13036 (PJW); Chester Valley Holdings, LLC, a Delaware limited liability company (9457), Case No. 11-13039 (PJW); Chester Valley Pharmaceuticals, LLC, a Delaware limited liability company (3713), Case No. 11-13041 (PJW); Graceway Canada Holdings, Inc., a Delaware corporation (6663), Case No. 11-13042 (PJW); and Graceway International, Inc., a Delaware corporation (2399), Case No. 11-13043 (PJW). The mailing address for Graceway Pharmaceuticals, LLC is 340 Martin Luther King Jr. Blvd., Suite 400, Bristol, TN 37620 (Attn: John Bellamy). On October 4, 2011, Graceway Canada Company filed an application in the Ontario Superior Court of Justice (Commercial List) pursuant to the Courts of Justice Act, R.S.O. 1990, c. C. 43.

ARTICLE 1	DEFINI	TIONS, INTERPRETATION AND RULES OF CONSTRUCTION	3
A.	Scope	of Definitions	3
В.	Defini	tions	3
	1.1	1.0% Holdback Account	
	1.2	363 Sale	
	1.3	549 Avoidance Action	
	1.4	549 Avoidance Action Proceeds	
	1.5	Acquired Assets	
	1.6	Administrative Expense Claim	
	1.7	Administrative Expense Claims Bar Date	
	1.8	Affiliate	
	1.9	Allowed Claim or Allowed Interest	4
	1.10	Allowed [] Claim or Allowed [] Interest	
	1.11	Approving Majority First Lien Lenders	5
	1.12	Asset Purchase Agreement	5
	1.13	Assets	
	1.14	Assumed	
	1.15	Assumed Liabilities	
	1.16	Assumption Schedule	
	1.17	Avoidance Action	
	1.18	Avoidance Action Proceeds	
	1.19	Bankruptcy Code	
-	1.20	Bankruptcy Court	
	1.21	Bankruptcy Rules	
	1.22	Bar Date	
	1.23	Beneficiaries	
	1.24	Bid Procedures Order	
	1.25	Budget	
	1.26	Business Day	
	1.27 1.28	Canadian Distribution Amount	
	1.28		
	1.29	Causes of Action Proceeds.	
	1.31	CCR Account.	
•	1.32	Chapter 11 Cases	
	1.33	Claim	
	1.34	Class	
	1.35	Committee Settlement.	
	1.36	Confirmation	
	1.37	Confirmation Date	
	1.38	Confirmation Hearing	
	1.39	Confirmation Order.	
	1.40	Consenting First Lien Lenders	
	1.41	Contingency Fee Counsel Fees and Evnenses	10 10

1.42	Creditor	10
1.43	Creditor Releasees	10
1.44	Creditors' Committee	11
1.45	Debtor(s)	11
1.46	Debtor Releasees	11
1.47	Deficiency Claims	12
1.48	Designated Account	12
1.49	DIP Order	12
1.50	Disclosure Statement	
1.51	Disputed Claim or Disputed Interest	12
1.52	Disputed [] Claim or Disputed [] Interest	13
1.53	Disputed Claims Reserve	13
1.54	Effective Date	13
1.55	Employee Account	13
1.56	Entity	14
1.57	Estate	14
1.58	Estate Fund Amount	14
1.59	Face Amount	14
1.60	Federal Bankruptcy Rule(s)	15
1.61	File or Filed	15
1.62	Final Distribution	15
1.63	Final Distribution Date	
1.64	Final Order	
1.65	First Lien Available Cash	
1.66	First Lien Claimholders	
1.67	First Lien Credit Agreement	
1.68	First Lien Facility Adequate Protection Claim	
1.69	First Lien Facility Agent	
1.70	First Lien Facility Claim	
1.71	First Lien Facility Deficiency Claim	17
1.72	First Lien Facility Lenders	
1.73	First Lien Facility Secured Claim	
1.74	First Lien Trust Administrative Fund	
1.75	General Unsecured Claims	
1.76	Graceway Canada	
1.77	Governmental Claims Bar Date	
1.78	Governmental Unit	
1.79	GTCR	
1.80	Impaired	
1.81	Indemnification Claims	
1.82	Intercreditor Agreement	
1.83	Intercompany Claims	
1.84	Interest	
1.85	Lazard	
1.86	Lazard Account	
1.87	Lien	.20

1.88	Liquidating Trust	20
1.89	Liquidating Trust Agreement	20
1.90	Liquidating Trust Committee	20
1.91	Liquidating Trustee	20
1.92	Liquidating Trustee Fee Amount	20
1.93	Liquidating Trustee Fees.	21
1.94	Liquidating Trustee First Lien Fees	21
1.95	Mezzanine Credit Agreement	21
1.96	Mezzanine Facility Agent	22
1.97	Mezzanine Facility Claim	
1.98	Mezzanine Facility Lenders	
1.99	Net Causes of Action Proceeds	22
1.100	Nycomed Litigation	22
1.101	Nycomed Litigation Proceeds	22
1.102	Old Equity	22
1.103	Old Equity Rights	22
1.104	Order	23
1.105	Other Assets	23
1.106	Other Available Cash	24
1.107	Other Priority Claim	24
1.108	Other Secured Claim Liens	24
1.109	Other Secured Claims	24
1.110	Overpayment Amount	24
1.111	Person	24
1.112	Petition Date	24
1.113	Plan	24
1.114	Possessory Lienholder Claims	25
1.115	Prepetition Claims Motions	25
1.116	Priority Tax Claim	25
1.117	Pro Rata	25
1.118	Professional	25
1.119	Professional Claim	26
1.120	Purchaser	26
1.121	Released Claims	26
1.122	Remaining Contracts	. 26
1.123	Reserve Accounts	. 26
1.124	Sale Allocation Motion	. 26
1.125	Sale Motion	. 27
1.126	Sale Order	. 27
1.127		
1.128	Sale Support Agreement	. 27
1.129	Scheduled	
1.130	Schedules	
1.131	Second Lien Credit Agreement	
1.132	Second Lien Facility Adequate Protection Claim	. 28
1.133	Second Lien Facility Agent	28

	1.134	Second Lien Facility Claim	28
	1.135	Second Lien Facility Deficiency Claim	28
	1.136	Second Lien Facility Lenders	28
	1.137	Second Lien Facility Secured Claim.	29
	1.138	Second Lien Obligations	29
	1.139	Secured Claim	29
	1.140	Security	29
	1.141	Tax Claims	
	1.142	Taxes	29
	1.143	Trust Administrative Fund	
	1.144	Trust Estate or Trust Assets	30
	1.145	Unimpaired	30
		United States Trustee	
	1.147	Unsecured Claim	30
C.	Rules	of Interpretation	31
D.	Comp	utation of Time	33
ARTICLE 2	TREAT	MENT OF UNCLASSIFIED CLAIMS	33
	2.1	Administrative Expense Claims	
	2.2	Statutory Fees	34
	2.3	Professional Claims	34
	2.4	Priority Tax Claims	35
	2.5	Deadline for Filing Administrative Expense Claims	35
ARTICLE 3		IFICATION AND TREATMENT OF CLASSIFIED CLAIMS AND	
	INTE	RESTS	37
A.	Gener	al	37
В.	Classi	fication	38
	3.1	Class 1: Other Priority Claims	
	3.2	Class 2: First Lien Facility Claims	
	3.3	Class 3: Second Lien Facility Claims	38
	3.4	Class 4: Other Secured Claims	38
	3.5	Class 5: General Unsecured Claims	38
	3.6	Class 6: Intercompany Claims	38
	3.7	Class 7: Old Equity	38
	3.8	Class 8: Old Equity Rights	39
ARTICLE 4		IFICATION OF CLASSES OF CLAIMS AND INTERESTS	
	IMP A	AIRED AND NOT IMPAIRED BY THE PLAN	39
	4.1	Unimpaired Classes of Claims and Interests	
	4.2	Impaired Classes of Claims and Interests	39

ARTICLE 5 PROVI	SIONS FOR THE TREATMENT OF CLAIMS AND INTERESTS	39
5.1	Class 1 (Other Priority Claims)	39
5.2	Class 2 (First Lien Facility Claims)	
5.3	Class 3 (Second Lien Facility Claims)	
5.4	Class 4 et seq. (Other Secured Claims)	
5.5	Class 5 (General Unsecured Claims)	
5.6	Class 6 (Intercompany Claims)	
5. 7	Class 7 (Old Equity)	
5.8	Class 8 (Old Equity Rights)	45
ARTICLE 6 TREAT	TMENT OF EXECUTORY CONTRACTS AND UNEXPIRED	
LEAS	SES	45
6.1	Assumption, Assignment and Rejection Pursuant to the Sale Order	45
6.2	Assumption and Assignment of Remaining Contracts	46
6.3	Cure Payments; Assurance of Performance	46
6.4	Objections To Assumption of Remaining Contracts	47
6.5	Rejection	
6.6	Approval of Rejection; Rejection Damages Claims Bar Date	48
ARTICLE 7 MEAN	S FOR EXECUTION AND IMPLEMENTATION OF THE PLAN	49
7.1	Substantive Consolidation of Claims against Debtors for Plan	
	Purposes Only	
7.2	The Liquidating Trust	
7.3	Continuation of the Automatic Stay	
7.4	Transfer of Equity Interests in Graceway Canada	
7.5	Committee Settlement	52
7.6	Post-Confirmation Operations	
7.7	Post-Confirmation Funding of Plan	53
7.8	Post-Effective Date Funding of Operations	54
7.9	Dissolution of the Debtors	55
7.10	Closing of the Debtors' Chapter 11 Cases	56
7.11	Post-Effective Date Cause of Action Settlements	56
7.12	Post-Effective Date Reporting	57
7.13	Cancellation of Existing Securities and Agreements	58
7.14	Dissolution of Creditors' Committee	
7.15	Nycomed Litigation Proceeds	58
ARTICLE 8 POST-	CONFIRMATION LITIGATION	60
8.1	Transfer and Enforcement of Causes of Action	60
8.2	Abandonment of Causes of Action and Other Assets of the Debtors.	60
ARTICLE 9 PROC	EDURES FOR TREATMENT OF DISPUTED CLAIMS	61
9.1	Objections to Claims	61

9	9.2	Resolution of Disputed Claims	61
9	9.3	Amendments to Claims	62
Ģ	9.4	No Distributions Pending Allowance	62
ARTICLE 10 D	DISTR	IBUTIONS	63
1	10.1	No Duplicate Distributions	
	10.2	Delivery of Distributions in General	
	10.3	Cash Payments	
	10.4	Interest on Claims	
	10.5	No De Minimis Distributions	
	10.6	Face Amount	
	10.7	Undeliverable Distributions	
	10.8	Timing of Distributions	
	10.9	Disputed Claims Reserve	08
	10.10	Compliance with Tax Requirements	69
ARTICLE 11 C	COND	ITIONS PRECEDENT	70
	11.1	Conditions to Confirmation	70
-	11.2	Conditions to the Effective Date	
	11.3	Termination of Plan for Failure To Become Effective	
	11.4	Waiver of Conditions	
	11.5	Notice of Effective Date	71
ARTICLE 12 I	EFFE(CT OF CONFIRMATION	72
	12.1	Jurisdiction of Court	72
	12.2	Binding Effect	72
	12.3	Term of Pre-Confirmation Injunctions or Stays	72
	12.4	Injunction Against Interference with Plan	72
	12.5	Exculpation	
	12.6	Injunction	
	12.7	Releases by Debtors	
	12.8	Releases by Holders of Claims and Interests	
	12.9	Limitation of Liability	77
ARTICLE 13	RETE	NTION OF JURISDICTION	78
	13.1	Scope of Bankruptcy Court Jurisdiction	78
ARTICLE 14	ACCE	EPTANCE OR REJECTION OF THE PLAN	80
	14.1	Persons Entitled to Vote	80
	14.2	Acceptance by Impaired Classes	8
	14.3	Request for Non-Consensual Confirmation	
ADTICLE 15	MISC	ELL ANEOLIS DROVISIONS	Q 1

15.1	Effectuating Documents and Further Transactions	81
15.2	Corporate Action	81
15.3	Payment of Statutory Fees	82
15.4	Modification of the Plan	82
15.5	Revocation of the Plan	
15.6	Governing Law	82
15.7	No Admissions	
15.8	Severability of Plan Provisions	83
15.9	Successors and Assigns	
15.10	Exemption from Certain Transfer Taxes	
15.11	Preservation of Rights of Setoffs	
15.12		
	No Injunctive Relief	
	Saturday, Sunday or Legal Holiday	
	Entire Agreement	
15.16	Dissolution of Creditors' Committee	85
	Notices	

Exhibits

- 1. List of Debtors
- Liquidating Trust Agreement Assumption Schedule 2.
- 3.

INTRODUCTION

Graceway Pharmaceuticals, LLC ("Graceway"), a Delaware limited liability company, and its Affiliates and Subsidiaries listed on Exhibit 1 hereto (collectively with Graceway, the "Debtors") propose the First Amended Joint Plan of Liquidation of Graceway Pharmaceuticals, LLC, et al. (the "Plan") for the resolution and satisfaction of all Claims against and Interests in the Debtors. The Debtors are the proponents of this Plan within the meaning of Section 1129 of the Bankruptcy Code. All capitalized terms not defined in this introduction have the meanings ascribed to them in Article 1 of this Plan. Reference is made to the Disclosure Statement, distributed contemporaneously herewith, for a discussion of the Debtors' history, businesses, resolution of material disputes, significant asset sales, financial projections for the liquidation and distribution of the Debtors' remaining Assets and a summary and analysis of the Plan and certain related matters.

This is a liquidating Plan pursuant to which substantially all of the Debtors' Assets are to be transferred to a liquidating trust (subject to existing Liens and encumbrances, to the extent specified in the Plan) that will liquidate such Assets and make distributions to holders of Allowed Claims, whose Claims against the Debtors will be exchanged for a beneficial interest in the Liquidating Trust.

The Plan is premised on the substantive consolidation of the Debtors with respect to the voting and treatment of all Claims and Interests other than Other Secured Claims, as provided below. The Plan does not contemplate substantive consolidation of the Debtors with respect to Other Secured Claims against the Debtors, which Claims shall apply separately with respect to each Plan proposed by each Debtor. If the Plan cannot be confirmed as to some or all of the Debtors, in the Debtors' sole discretion after consultation with the First Lien Facility Agent, the Second Lien Facility Agent and the Creditors' Committee, (a) the Plan may be

revoked as to all of the Debtors, or (b) the Debtors may revoke the Plan as to any Debtor not satisfying the cramdown requirements of Section 1129(b)(7) of the Bankruptcy Code (and any such Debtor's Chapter 11 Case being converted to a chapter 7 liquidation, continued or dismissed in the Debtors' sole discretion after consultation with the First Lien Facility Agent, the Second Lien Facility Agent and the Creditors' Committee) and confirmed as to the remaining Debtors. A list of each Debtor who is a proponent of the Plan and its corresponding Chapter 11 Case docket number is attached hereto as Exhibit 1.

Under Section 1125(b) of the Bankruptcy Code, a vote to accept or reject the Plan cannot be solicited from holders of Claims and/or Interests until such time as the Disclosure Statement has been approved by the Bankruptcy Court. The Debtors urge all holders of Claims entitled to vote on the Plan to read the Plan and the Disclosure Statement in their entirety before voting to accept or reject the Plan. To the extent, if any, that the Disclosure Statement is inconsistent with the Plan, the Plan will govern. No solicitation materials other than the Disclosure Statement and any schedules and exhibits attached thereto or referenced therein, or otherwise enclosed with the Disclosure Statement served by the Debtors on interested parties, have been authorized by the Debtors or the Bankruptcy Court for use in soliciting acceptances of the Plan. Subject to certain restrictions and requirements set forth in Section 1127 of the Bankruptcy Code and Federal Bankruptcy Rule 3019, the Debtors expressly reserve the right, after consultation with the First Lien Facility Agent, the Second Lien Facility Agent and the Creditors' Committee, to alter, amend, modify, revoke, or withdraw this Plan prior to its substantial consummation.

ARTICLE 1 DEFINITIONS, INTERPRETATION AND RULES OF CONSTRUCTION

- A. Scope of Definitions. For the purposes of this Plan, except as expressly provided or unless the context otherwise requires, all capitalized terms not otherwise defined shall have the meanings ascribed to them in Article 1 of this Plan. Any term used in this Plan that is not defined herein, but is defined in the Bankruptcy Code or the Federal Bankruptcy Rules, shall have the meaning ascribed to that term in the Bankruptcy Code or the Federal Bankruptcy Rules, respectively. Whenever the context requires, capitalized terms shall include the plural as well as the singular number, the masculine gender shall include the feminine, and the feminine gender shall include the masculine.
- **B. Definitions.** In addition to such other terms as are defined in other Sections of the Plan, the following terms (which appear in the Plan as capitalized terms) shall have the meanings ascribed to them in this Article 1 of the Plan.
- 1.1 1.0% Holdback Account means the account into which \$4,213,125.55 of the Sale Proceeds was deposited on December 2, 2011 pursuant to paragraph 30(vi) of the Sale Order.
- 1.2 363 Sale means the sale approved under the Sale Order and consummated on December 2, 2011.
- 1.3 549 Avoidance Action means any avoidance or equitable subordination or recovery actions under Bankruptcy Code Section 549, but excluding Released Claims.
- 1.4 549 Avoidance Action Proceeds means the proceeds of any 549 Avoidance Actions actually received by the Debtors' Estates or the Liquidating Trust, whether obtained through litigation, settlement or otherwise, net of the fees and expenses incurred in obtaining such proceeds (including Contingency Fee Counsel Fees and Expenses, if applicable).

- 1.5 Acquired Assets means "Acquired Assets," as defined in the Asset Purchase Agreement.
- 1.6 Administrative Expense Claim means any right to payment for any cost or expense of administration of the Debtors' Chapter 11 Cases asserted or arising under Sections 503, 507(a)(2), or 507(b) of the Bankruptcy Code, including, any (i) actual and necessary cost or expense of preserving the Debtors' Estates or operating the business of the Debtors arising on or after the Petition Date, (ii) payment to be made under this Plan to cure a default on an executory contract or unexpired lease that has been assumed on or after the Petition Date (and prior to the occurrence of such default) pursuant to Section 365 of the Bankruptcy Code, (iii) cost, indebtedness or contractual obligation duly and validly incurred or assumed by the Debtors in the ordinary course of business arising on or after the Petition Date, (iv) compensation or reimbursement of expenses of Professionals arising on or after the Petition Date, to the extent allowed by the Bankruptcy Court under Section 330(a) or Section 331 of the Bankruptcy Code, including, without limitation, all Allowed Professional Claims, and (v) fees or charges assessed against the Debtors' Estates under Section 1930 of title 28 of the United States Code.
- 1.7 Administrative Expense Claims Bar Date means the first Business Day that is at least thirty-five (35) days following the Effective Date or such other date ordered by the Bankruptcy Court.
- 1.8 Affiliate shall have the meaning set forth in Section 101(2) of the Bankruptcy Code.
- 1.9 Allowed Claim or Allowed Interest means a Claim against or Interest in any of the Debtors or any portion thereof (i) that has been allowed by a Final Order, (ii) as to which, on or by the Effective Date, (a) no proof of Claim or Interest has been Filed with the

Bankruptcy Court and (b) the liquidated and noncontingent amount of which is Scheduled, other than a Claim or Interest that is Scheduled at zero, in an unknown amount, or as disputed, (iii) for which a proof of Claim or Interest in a liquidated amount has been timely Filed with the Bankruptcy Court pursuant to the Bankruptcy Code and as to which either (a) no objection to its allowance has been Filed within the applicable periods of limitation fixed by the Plan, the Bankruptcy Code, or by any Order of the Bankruptcy Court sought pursuant to Section 9.1 of the Plan or otherwise entered by the Bankruptcy Court or (b) any objection to its allowance has been settled or withdrawn, or has been denied by a Final Order, or (iv) that is expressly allowed in the Plan.

- 1.10 Allowed [] Claim or Allowed [] Interest means an Allowed Claim or Allowed Interest in the particular category or Class identified.
- 1.11 Approving Majority First Lien Lenders means at least two First Lien Facility Lenders holding in aggregate a majority of the then outstanding First Lien Facility Claims held by the Consenting First Lien Lenders.
- 1.12 Asset Purchase Agreement means the Asset Purchase Agreement dated as of November 18, 2011, between the Debtors, Purchaser and Graceway Canada, as amended, and all other documentation related thereto or contemplated thereby.
- 1.13 Assets means all legal or equitable prepetition and post-petition interests of the Debtors or, to the extent transferred and assigned by the Debtors to the Liquidating Trust pursuant to Section 7.2.3 below on the Effective Date, all legal or equitable interests of the Liquidating Trust, in any and all real or personal property of any nature, including any real estate, buildings, structures, improvements, privileges, rights, easements, leases, subleases, licenses, goods, materials, supplies, furniture, fixtures, equipment, work in process, accounts,

chattel paper, tax refunds, net operating losses, Cash, deposit accounts, reserves, deposits, equity interests, contractual rights, intellectual property rights, Claims, Causes of Action (including, without limitation, Avoidance Actions), assumed executory contracts and unexpired leases, other general intangibles, and the proceeds, products, offspring, rents or profits thereof.

- 1.14 Assumed (including the term "Assumption" and any variants and derivatives thereof) means assumed by the Purchaser pursuant to the Asset Purchase Agreement.
- 1.15 Assumed Liabilities means "Assumed Liabilities," as defined in the Asset Purchase Agreement.
- 1.16 Assumption Schedule has the meaning given to such term in Section 6.2 of the Plan.
- 1.17 Avoidance Action means any avoidance or equitable subordination or recovery actions under Bankruptcy Code Sections 105, 502(d), 510, 542, 543, 544, 545, 546, 547, 548, 550, 551, and 553 and any other avoidance or similar action under the Bankruptcy Code or similar law of any state or foreign jurisdiction, but excluding Released Claims and 549 Avoidance Actions.
- 1.18 Avoidance Action Proceeds means the proceeds of any Avoidance Actions actually received by the Debtors' Estates or the Liquidating Trust, whether obtained through litigation, settlement or otherwise, net of the fees and expenses incurred in obtaining such proceeds (including Contingency Fee Counsel Fees and Expenses, if applicable).
- 1.19 Bankruptcy Code means title 11 of the United States Code, as in effect on the Petition Date and as thereafter amended, as applicable in the Chapter 11 Cases.

- 1.20 Bankruptcy Court means the United States Bankruptcy Court for the District of Delaware or any other court of the United States having jurisdiction over the Chapter 11 Cases.
- 1.21 Bankruptcy Rules means the Federal Bankruptcy Rules and the Local Bankruptcy Rules for the District of Delaware, as in effect on the Petition Date and as thereafter amended, as applicable from time to time in the Chapter 11 Cases.
- 1.22 Bar Date means December 30, 2011, which was established by the Bankruptcy Court as the deadline for filing proofs of Claims relating to prepetition Claims against the Debtors.
- 1.23 Beneficiaries means the holders of Allowed Claims as may be determined from time to time in accordance with the Plan and the Liquidating Trust Agreement, each of whom shall receive beneficial interests in the Liquidating Trust.
- 1.24 Bid Procedures Order means the Order Approving and Authorizing (A) Bidding Procedures in Connection with the Sale of Certain Assets of the Debtors, (B) Stalking Horse Bid Protections, (C) the Form and Manner of Notice of the Sale Hearing and (D) Related Relief [Docket No. 119].
- 1.25 Budget means the "Approved Budget" as defined in the DIP Order, as amended from time to time.
- 1.26 Business Day means any day other than a Saturday, Sunday or a "legal holiday" (as such term is defined in Federal Bankruptcy Rule 9006(a)(6)).
- 1.27 Canadian Distribution Amount means the aggregate amount of Cash, if any, distributed by Graceway Canada to the Debtors on account of the Debtors' equity interests in Graceway Canada after payment in full of all creditors of Graceway Canada.

- 1.28 Cash means legal tender accepted in the United States of America for the payment of public and private debts, currently denominated in United States Dollars.
- 1.29 Causes of Action means, without limitation, any and all actions, causes of action, Avoidance Actions, 549 Avoidance Actions, controversies, liabilities, obligations, rights, suits, damages, judgments, Claims and demands whatsoever owned by any of the Debtors, whether known or unknown, reduced to judgment, liquidated or unliquidated, fixed or contingent, matured or unmatured, disputed or undisputed, secured or unsecured, whether assertable directly, indirectly, derivatively or in any representative or other capacity, existing or hereafter arising, in law, equity, or otherwise, based in whole or in part upon any act, failure to act, error, omission, transaction, occurrence or other event arising or occurring prior to the Petition Date or during the course of the Chapter 11 Cases, through and including the Effective Date, including, without limitation, the Debtors' rights under the Asset Purchase Agreement to the Nycomed Litigation Proceeds, but excluding Released Claims.
- 1.30 Causes of Action Proceeds means the proceeds of any Causes of Action actually received by the Debtors' Estates or the Liquidating Trust, whether obtained through litigation, settlement or otherwise.
- 1.31 CCR Account means the "CCR Account," as defined in the Sale Order, into which \$17,684,000 was deposited on December 2, 2011 and in which approximately \$17,289,344.44 remains as of the date hereof for the sole purpose of paying liabilities permitted by the Budget and/or the Sale Support Agreement on account of chargebacks, channel management agreements, product returns and rebates constituting (a) Allowed Claims authorized to be paid by the Bankruptcy Court pursuant to the Orders approving the Prepetition Claims Motions or (b) Allowed Administrative Expense Claims.

- 1.32 Chapter 11 Cases means the chapter 11 cases of the Debtors pending before the Bankruptcy Court as set forth in Exhibit 1 hereto and as being jointly administered with one another under Case No. 11-13036 (PJW), and as to any Debtor individually, a Chapter 11 Case.
- 1.33 Claim has the meaning ascribed to such term in Section 101(5) of the Bankruptcy Code, as construed by Section 102(2) of the Bankruptcy Code.
- 1.34 Class means a group of Claims or Interests as classified in a particular class under the Plan pursuant to Section 1122 of the Bankruptcy Code.
- Creditors' Committee Settlement means the settlement of the objections of the Creditors' Committee to the Sale Motion and the Sale Allocation Motion set forth on the record at the hearing on the Sale Motion and the Sale Allocation Motion on November 22, 2011, pursuant to which (a) the First Lien Facility Agent and certain First Lien Facility Lenders agreed, subject to confirmation of a joint plan of liquidation incorporating the terms of such settlement, to release their respective Liens and Claims, including the First Lien Facility Secured Claim, any First Lien Facility Adequate Protection Claim and any First Lien Facility Deficiency Claim, on the Estate Fund Amount, the Liquidating Trustee Fee Amount and 25% of all Avoidance Action Proceeds other than 549 Avoidance Action Proceeds, and the First Lien Facility Agent (for the benefit of the First Lien Claimholders) received in exchange therefor any Canadian Distribution Amount in excess of the Estate Fund Amount, 75% of all Avoidance Action Proceeds and 100% of all 549 Avoidance Action Proceeds, all in accordance with the terms of this Plan; and (b) the Creditors' Committee agreed not to assert any Claim or Cause of Action against the First Lien Facility Agent or the First Lien Facility Lenders.

- 1.36 Confirmation means entry of the Confirmation Order by the Bankruptcy Court.
- 1.37 Confirmation Date means the date on which the Bankruptcy Court enters the Confirmation Order on its docket, within the meaning of Federal Bankruptcy Rules 5003 and 9021.
- 1.38 Confirmation Hearing means the duly noticed hearing held by the Bankruptcy Court to consider Confirmation of the Plan pursuant to Bankruptcy Code Section 1128, including any continuances thereof.
- 1.39 Confirmation Order means the Order of the Bankruptcy Court confirming the Plan pursuant to Section 1129 of the Bankruptcy Code in form and substance reasonably acceptable to the Debtors and the First Lien Facility Agent.
- 1.40 Consenting First Lien Lenders means each of the First Lien Facility Lenders party to the Sale Support Agreement.
- 1.41 Contingency Fee Counsel Fees and Expenses means the fees and expenses incurred by counsel retained by the Liquidating Trust in pursuing the Causes of Action, which, as to fees, shall be in an amount that does not exceed fifteen percent (15%) of any Causes of Action Proceeds and on such other terms and provisions acceptable to the Liquidating Trustee after consultation with the Liquidating Trust Committee.
 - 1.42 Creditor means any Entity who holds a Claim against any of the Debtors.
- 1.43 Creditor Releasees means (i) the officers, directors, shareholders, members and/or enrollees, employees, representatives, advisors, attorneys, financial advisors, investment bankers, or agents of the Debtors (including their estates and beneficiaries), in each case as of the Petition Date or that have become officers and/or directors thereafter, including

GTCR; (ii) the members of the Creditors' Committee, solely with respect to each member's conduct in furtherance of its, his, or her duties as a member of the Creditors' Committee, and not with respect to the actions of such members as individual creditors, and its agents, attorneys and other professionals acting in conjunction with the Chapter 11 Cases, and the Creditors' Committee's agents, attorneys and other professionals; (iii) the Liquidating Trustee; (iv) the First Lien Facility Agent and its agents, attorneys and other professionals; (v) the First Lien Facility Lenders in their capacity as such; (vi) the Second Lien Facility Agent and its agents, attorneys and other professionals; (vii) the Second Lien Facility Lenders in their capacity as such; and (viii) the Debtors' Affiliates and their officers and directors, including GTCR.

- 1.44 Creditors' Committee means the statutory committee of unsecured creditors appointed in the Chapter 11 Cases pursuant to Bankruptcy Code Section 1102.
- 1.45 **Debtor(s)** means, individually, Graceway Pharmaceuticals, LLC and each other Debtor listed on <u>Exhibit 1</u> hereto and, collectively, all of them, including in their capacity as debtors-in-possession pursuant to Sections 1107 and 1108 of the Bankruptcy Code.
- 1.46 Debtor Releasees means (i) the officers, directors, shareholders, members and/or enrollees, employees, representatives, advisors, attorneys, financial advisors, investment bankers, or agents of the Debtors (including their estates and beneficiaries), in each case as of the Petition Date or that have become officers and/or directors thereafter, including GTCR,; (ii) the members of the Creditors' Committee, solely with respect to each member's conduct in furtherance of its, his, or her duties as a member of the Creditors' Committee, and not with respect to the actions of such members as individual creditors, and its agents, attorneys and other professionals acting in conjunction with the Chapter 11 Cases, and the Creditors' Committee's agents, attorneys and other professionals; (iii) the Liquidating Trustee; (iv) the First Lien Facility

Agent and its agents, attorneys and other professionals; (v) the First Lien Facility Lenders in their capacity as such; (vi) the Second Lien Facility Agent and its agents, attorneys and other professionals; (vii) the Second Lien Facility Lenders in their capacity as such; and (viii) the Debtors' Affiliates and their officers and directors, including GTCR.

- 1.47 **Deficiency Claims** means with respect to any Claim secured by a Lien in any property of any Debtor having a value of less than the amount of such Claim (after taking into account other Liens of higher priority in such property), the portion of such Claim equal to the difference between (a) the allowed amount of the Claim and (b) the allowed amount of the secured portion of such Claim (which allowed secured amount may be set pursuant to this Plan).
- 1.48 Designated Account means the "Designated Account," as defined in the Sale Order, into which approximately \$4,410,193.60 was deposited on December 2, 2011 and in which an aggregate of approximately \$18,002,063.72 remains as of the date hereof.
- 1.49 DIP Order means the Final Order (i) Authorizing Debtors to Utilize Cash Collateral Pursuant to 11 U.S.C. § 363; (ii) Granting Adequate Protection to Prepetition Secured Parties Pursuant to 11 U.S.C. §§ 361, 362, 363 and 364; and (iii) Authorizing Debtors to Obtain Postpetition Financing Pursuant to 11 U.S.C. § 364, as supplemented or amended from time to time [Docket No. 219].
- 1.50 Disclosure Statement means that certain written disclosure statement that relates to this Plan as Filed in the Chapter 11 Cases by the Debtors, including the schedules and exhibits attached thereto, as may be amended, modified or supplemented from time to time, as approved by the Bankruptcy Court pursuant to Section 1125 of the Bankruptcy Code.
- 1.51 Disputed Claim or Disputed Interest means a Claim or Interest, respectively, that is not an Allowed Claim or Allowed Interest, as applicable, and that either (i)

the Debtors have Scheduled as "disputed," "contingent" or "unliquidated," or (ii) as to which a proof of Claim or Interest has been Filed or deemed Filed as contingent and as to which an objection has been or may be timely Filed by the Debtors, the Liquidating Trustee or any other party in interest entitled to do so, which objection, if timely Filed, has not been withdrawn and has not been overruled or denied by a Final Order. For purposes of this definition, Disputed Claims shall include any actual or estimated Administrative Expense Claims, Professional Claims, Priority Tax Claims or Other Priority Claims that have not been paid in full in Cash as of the Effective Date and any actual or estimated Other Secured Claims that have not been treated in accordance with Section 5.4 of this Plan as of the Effective Date.

- 1.52 Disputed [] Claim or Disputed [] Interest means a Disputed Claim or Disputed Interest in the particular category or Class identified
- 1.53 Disputed Claims Reserve has the meaning given to such term in Section 10.9 of the Plan.
- 1.54 Effective Date means the date that is the first Business Day following occurrence of the Confirmation Date on which all conditions to the Effective Date set forth in Article 11 of the Plan have been satisfied or, if waivable, waived pursuant to Section 11.4 of the Plan.
- 1.55 Employee Account means the "Employee Account," as defined in the Sale Order, into which \$2,593,647.29 was deposited on December 2, 2011 and in which approximately \$965,687.11 remains as of the date hereof for purposes of funding the Debtors' employee severance and benefits liabilities to the extent approved by the Orders approving the Prepetition Claims Motions and permitted by the Budget and/or the Sale Support Agreement.

- 1.56 Entity means an entity as defined in Section 101(15) of the Bankruptcy Code.
- 1.57 Estate means, with regard to each Debtor, the estate that was created by the commencement by such Debtor of its Chapter 11 Case pursuant to Section 541 of the Bankruptcy Code, and shall be deemed to include, without limitation, any and all rights, powers, and privileges of such Debtor and any and all Assets and Interests in property, whether real, personal or mixed, rights, Causes of Action, avoidance powers or extensions of time that such Debtor or such estate shall have had as of the Petition Date, or which such Estate acquired after the commencement of the Chapter 11 Case, whether by virtue of Sections 541, 544, 545, 546, 547, 548, 549 or 550 of the Bankruptcy Code, or otherwise. "Estates" has a correlative meaning.
- 1.58 Estate Fund Amount means an amount equal to \$10,000,000 to be available to the Debtors' estates free and clear of all Liens and Claims of the First Lien Facility Agent, including First Lien Facility Secured Claim, any First Lien Facility Adequate Protection Claim and any First Lien Facility Deficiency Claim. The Estate Fund Amount shall be funded as follows: (i) first, from the first \$10,000,000 of the Canadian Distribution Amount received by the Debtors; (ii) second, solely to the extent the Canadian Distribution Amount is less than \$10,000,000, from the 1.0% Holdback Account; and (iii) finally, solely to the extent the sum of the Canadian Distribution Amount and the funds in the 1.0% Holdback Account is less than \$10,000,000, from the Sale Proceeds distributed to the First Lien Facility Agent pursuant to the Sale Order.
- 1.59 Face Amount has the meaning given to such term in Section 10.6 of the Plan.

- 1.60 Federal Bankruptcy Rule(s) means, collectively, the Federal Rules of Bankruptcy Procedure, as in effect on the Petition Date and as thereafter amended, as applicable in the Chapter 11 Cases, and individually, a particular Federal Rule of Bankruptcy Procedure.
- 1.61 File or Filed means file or filed with the Bankruptcy Court or its authorized designee in the Chapter 11 Cases.
- 1.62 Final Distribution has the meaning given to such term in Section 10.8.3 of the Plan.
- 1.63 Final Distribution Date has the meaning given to such term in Section 10.8.3 of the Plan.
- 1.64 Final Order means an Order or judgment of the Bankruptcy Court, or other court of competent jurisdiction, as entered on the docket in any Chapter 11 Case or the docket of any other court of competent jurisdiction, (i) that has not been reversed, stayed, modified or amended, (ii) as to which the time to appeal or seek certiorari or move for a new trial, reargument or rehearing has expired, and no appeal or petition for certiorari or other proceedings for a new trial, reargument or rehearing has been timely taken, or (iii) as to which any appeal that has been taken or any petition for certiorari that has been timely filed has been withdrawn or resolved by the highest court to which the Order or judgment was appealed or from which certiorari was sought or the new trial, reargument or rehearing has been denied or resulted in no modification of such Order; provided, however, that the possibility that a motion under Rule 59 or Rule 60 of the Federal Rules of Civil Procedure, or any analogous rule under the Bankruptcy Rules or applicable state court rules of civil procedure, may be Filed with respect to such Order, shall not cause such Order not to be a Final Order.

First Lien Available Cash means (i) any Canadian Distribution Amount 1.65 remaining after the Estate Fund Amount is fully funded, (ii) any Cash in the 1.0% Holdback Account after the Estate Fund Amount is fully funded, (iii) any Cash in the CCR Account after payment, to the extent permitted by the Budget and/or the Sale Support Agreement, of all liabilities on account of chargebacks, channel management agreements, product returns and rebates constituting (a) Allowed Claims authorized to be paid by the Bankruptcy Court pursuant to the Orders approving the Prepetition Claims Motions or (b) Allowed Administrative Expense Claims, (iv) any Cash in the Employee Account after payment of severance and benefits approved in the Orders approving the Prepetition Claims Motion and permitted by the Budget and/or Sale Support Agreement, (v) any Cash in the Lazard Account after paying the Allowed Professional Fees of Lazard, (vi) any other Cash of the Debtors (whether deposited in the Designated Account or otherwise and including the Overpayment Amount) in excess of the amounts required to be set aside for payment of all Allowed Administrative Expense Claims, Allowed Other Secured Claims (to the extent not otherwise satisfied in accordance with the terms of this Plan), Allowed Priority Tax Claims, Allowed Other Priority Claims and any cure costs for executory contracts or unexpired leases on the Assumption Schedule, in each case, pursuant to the Budget and/or the Sale Support Agreement, other than any such Claims on account of (A) chargebacks, channel management agreements, product returns and rebates or (B) employee severance and benefits payable from the Employee Account, and (vii) any other Cash of the Debtors described in clause (vi) of this definition remaining after payment in full of the Allowed Claims and cure costs described in such clause. For the avoidance of doubt, First Lien Available Cash shall exclude the Other Assets.

- 1.66 First Lien Claimholders means the "First Lien Claimholders" as defined in the DIP Order.
- 1.67 First Lien Credit Agreement means that certain \$680,000,000 First Lien Credit Agreement dated as of May 3, 2007, among, *inter alia*, Graceway Pharmaceuticals, LLC, as borrower, Graceway Holdings, LLC, the First Lien Facility Lenders, and the First Lien Facility Agent.
- 1.68 First Lien Facility Adequate Protection Claim means the Claims of the First Lien Claimholders under Section 507(b) of the Bankruptcy Code and Section 4(b) of the DIP Order, if any, including, without limitation, any adequate protection Claim arising from the payment of the fees and expenses of the Professionals and the expenses of the members of the Creditors' Committee, in each case, in connection with the Chapter 11 Cases.
- 1.69 First Lien Facility Agent means Bank of America, N.A., as administrative agent for the First Lien Facility Lenders and collateral agent for the First Lien Claimholders, or any successor thereto.
- 1.70 First Lien Facility Claim means the First Lien Facility Secured Claim, any First Lien Facility Deficiency Claim and any First Lien Facility Adequate Protection Claim.
- 1.71 First Lien Facility Deficiency Claim means any Deficiency Claim with respect to the First Lien Facility Secured Claim.
- 1.72 First Lien Facility Lenders means the "Lenders" as defined in the First Lien Credit Agreement.
- 1.73 First Lien Facility Secured Claim means the Secured Claims of the First Lien Claimholders under the First Lien Credit Agreement and the DIP Order. For the avoidance of doubt, to the extent the value of the Assets that secure the First Lien Facility Secured Claim

exceed the amount of the First Lien Facility Secured Claim before taking into account postpetition interest and fees, the First Lien Facility Secured Claim shall include post-petition interest and fees.

- 1.74 First Lien Trust Administrative Fund shall have the meaning specified in Section 4.1 of the Liquidating Trust Agreement and shall be funded solely in accordance with Section 7.8 below.
- 1.75 General Unsecured Claims means all Unsecured Claims against the Debtors, including, without limitation, the Mezzanine Facility Claim and any Second Lien Facility Deficiency Claim, but excluding any First Lien Facility Deficiency Claim, the Intercompany Claims and the Old Equity Rights.
- 1.76 Graceway Canada means Graceway Canada Company, a non-Debtor Affiliate of the Debtors, which filed an application in the Ontario Superior Court of Justice (Commercial List) pursuant to the *Courts of Justice Act*, R.S.O. 1990, c. C. 43 on October 4, 2011.
- 1.77 Governmental Claims Bar Date means March 27, 2012, which was established by the Bankruptcy Court as the deadline for filing proofs of Claims relating to prepetition Claims against the Debtors.
- 1.78 Governmental Unit shall have the meaning ascribed to it in Section101(27) of the Bankruptcy Code.
- 1.79 GTCR means GTCR Partners VIII, L.P., GTCR Golder Rauner II, L.L.C., GTCR Fund VIII, L.P., Fund VIII/B Graceway Splitter, L.P., GTCR Co-Invest II, L.P., GTCR Fund IX/A, L.P., Fund IX/B Graceway Splitter, L.P., GTCR Co-Invest III, L.P., and any affiliates of the foregoing.

- 1.80 Impaired shall have the meaning ascribed to it in Section 1124 of the Bankruptcy Code when used with reference to a Claim or an Interest.
- 1.81 Indemnification Claims means the obligations of the Debtors, or any one of them, pursuant to their certificates of incorporation or other charter documents, bylaws, applicable law, any employment agreement or other express agreement operational as of the Petition Date to indemnify any of their current and former officers and directors, on the terms and subject to the limitations described therein.
- 1.82 Intercreditor Agreement means the Intercreditor Agreement dated as of May 3, 2007, among Graceway Holdings, LLC, Graceway Pharmaceuticals, LLC, the First Lien Facility Agent, the Second Lien Facility Agent and Goldman Sachs Credit Partners, L.P. as Control Agent.
- 1.83 Intercompany Claims means the Claims of a Debtor against any other Debtor, and shall exclude any Claims of a Debtor against a non-Debtor Affiliate.
- 1.84 Interest means the legal, equitable, contractual and other rights of the holders of Old Equity and Old Equity Rights.
 - 1.85 Lazard means Lazard Frères & Co. LLC.
- 1.86 Lazard Account means the "Lazard Account," as defined in the Sale Order, into which \$2,481,250.00 was deposited on December 2, 2011 for the payment of the sale transaction fee due and owing to Lazard pursuant to the terms of that certain engagement letter, dated March 12, 2010, between Lazard and Graceway Pharmaceuticals, LLC in connection with the 363 Sale.

- 1.87 Lien means with respect to any Asset, any mortgage, deed of trust, deed to secure debt, lien (statutory or otherwise), pledge, hypothecation, encumbrance, collateral assignments, charge or security interest in, on or of such Asset.
- 1.88 Liquidating Trust means the liquidating trust established pursuant to the Liquidating Trust Agreement further described in Section 7.2 hereof, which shall act through the Liquidating Trustee.
- 1.89 Liquidating Trust Agreement means the Liquidating Trust Agreement dated as of the Effective Date among the Debtors and the Liquidating Trustee, a copy of which is attached hereto as Exhibit 2 and incorporated herein by reference, and all supplements and amendments thereto.
- 1.90 Liquidating Trust Committee shall have the meaning ascribed to it in the Liquidating Trust Agreement.
- 1.91 Liquidating Trustee means the Person appointed and serving from time to time as Liquidating Trustee under the Liquidating Trust Agreement, acting in his capacity as such on behalf of the Liquidating Trust. The initial Liquidating Trustee shall be Kip Horton of RPA Advisors, LLC.
- 1.92 Liquidating Trustee Fee Amount means \$150,000, which shall be funded into the Trust Administrative Fund by the First Lien Facility Agent from Sale Proceeds to which the First Lien Facility Agent or First Lien Claimholders would otherwise be entitled and shall be free and clear of all Liens and Claims of the First Lien Facility Agent or the First Lien Claimholders, including the First Lien Facility Secured Claim, any First Lien Facility Adequate Protection Claim and any First Lien Facility Deficiency Claim.

- 1.93 Liquidating Trustee Fees means the reasonable fees and out-of-pocket expenses of the Liquidating Trustee and any attorneys or other professionals hired by the Liquidating Trustee other than Liquidating Trustee First Lien Fees and Contingency Fee Counsel Fees and Expenses, in each case, payable in accordance with the terms and conditions of the Liquidating Trust Agreement solely from the Liquidating Trustee Fee Amount and the Other Assets.
- of-pocket expenses of the Liquidating Trustee and any attorneys or other professionals hired by the Liquidating Trustee other than Contingency Fee Counsel Fees and Expenses, in each case, payable in accordance with the terms and conditions of the Liquidating Trust Agreement, incurred from the Effective Date through and including the date all First Lien Facility Claims are paid in full in Cash, in connection with (a) the collection of accounts receivable, (b) the pursuit of Causes of Action (including 549 Avoidance Actions but excluding for the avoidance of doubt Avoidance Actions), and (c) the objection to, and settlement of, Other Secured Claims (including Possessory Lienholder Claims), Administrative Expense Claims, Priority Tax Claims and Other Priority Claims, which shall be funded into the First Lien Trust Administrative Fund by the First Lien Facility Agent from Sale Proceeds to which the First Lien Facility Agent or First Lien Claimholders would otherwise be entitled.
- 1.95 Mezzanine Credit Agreement means that certain \$70,000,000 Mezzanine Credit Agreement dated as of May 3, 2007, among, *inter alia*, Graceway Pharmaceuticals, LLC, Graceway Holdings, LLC, the Mezzanine Facility Lenders, Goldman Sachs Credit Partners L.P., as administrative agent, joint lead arranger, joint lead book manager and syndication agent, Deutsche Bank Trust Company Americas, as joint lead arranger, and

Deutsche Bank Securities Inc. and Banc of America Securities LLC, as joint lead book managers.

- 1.96 Mezzanine Facility Agent means Goldman Sachs Credit Partners L.P., as administrative agent for the Mezzanine Facility Lenders, or any successor thereto.
- 1.97 Mezzanine Facility Claim means the Claim of the Mezzanine Facility
 Agent on behalf of the Mezzanine Facility Lenders under the Mezzanine Credit Agreement.
- 1.98 Mezzanine Facility Lenders means the "Lenders" as defined in the Mezzanine Credit Agreement.
- 1.99 Net Causes of Action Proceeds means the Causes of Action Proceeds, net of the fees and expenses incurred in obtaining such proceeds (including Contingency Fee Counsel Fees and Expenses, if applicable).
- 1.100 Nycomed Litigation means Graceway Pharmaceuticals, LLC and 3M Innovative Properties Company v. Perrigo Company, Perrigo Israel Pharmaceuticals, Ltd. and Nycomed U.S. Inc., Civil Action No. 10-937 (WJM)(MF) in the United States District Court for the District of New Jersey.
- 1.101 Nycomed Litigation Proceeds means the portion of the proceeds of the Nycomed Litigation to which the Debtors are entitled pursuant to the Asset Purchase Agreement.
- 1.102 Old Equity means any outstanding or retired ownership interests in any of the Debtors, including interests evidenced by stock, membership interests or their equivalents, but excluding the Old Equity Rights.
- 1.103 Old Equity Rights means any calls, rights, puts, awards, commitments, repurchase rights, unvested or unexercised options, warrants, unvested common, preferred or other interests or any other agreement of any character related to Old Equity, obligating any of

the Debtors to issue, transfer, purchase, redeem, or sell any common, preferred or other interests or securities, any rights under any equity incentive plans, voting agreements, investor agreements and registration rights agreements regarding common, preferred or other interests or equity securities of any of the Debtors, any Claims arising from the rescission of a purchase, sale or other acquisition of any outstanding common, preferred, membership or other interests or equity securities (or any right, Claim, or interest in and to any common, preferred, membership or other interests or equity securities) of any of the Debtors, any Claims for the payment of any distributions with respect to any common, preferred, membership or other interests or equity securities of any of the Debtors, and any Claims for damages or any other relief arising from the purchase, sale, or other acquisition of the outstanding common, preferred, membership or other interests or equity securities of any of the Debtors.

- 1.104 Order means an order of judgment of the Bankruptcy Court, or other court of competent jurisdiction, as entered on the docket in any Chapter 11 Case or the docket of any other court of competent jurisdiction.
- 1.105 Other Assets means the following Assets, which Assets shall be provided to the Debtors' Estates free and clear of the Liens and Claims of the First Lien Facility Agent, including the First Lien Facility Secured Claim, any First Lien Facility Adequate Protection Claim and any First Lien Facility Deficiency Claim:
 - (i) the Estate Fund Amount;
 - (ii) the Liquidating Trustee Fee Amount;
 - (iii) 25% of Avoidance Action Proceeds; and

- (iv) after payment in full of any First Lien Facility Adequate Protection Claim and any Second Lien Facility Adequate Protection Claim, any other Assets upon which there are no Liens.
- 1.106 Other Available Cash means the Other Assets, less the amount of Other Assets deposited in the Trust Administrative Fund and the Disputed Claims Reserve.
- 1.107 Other Priority Claim means any Claim, other than an Administrative Expense Claim or a Priority Tax Claim, of a Creditor to the extent such Claim is entitled to priority pursuant to Bankruptcy Code Section 507(a).
- 1.108 Other Secured Claim Liens means the valid, perfected and enforceable Liens relating to the Allowed Other Secured Claims.
- 1.109 Other Secured Claims means all Secured Claims against the Debtors other than the First Lien Facility Secured Claim and the Second Lien Facility Secured Claim; provided, however, that any Secured Claim that is an Assumed Liability has been Assumed and shall be satisfied in accordance with the Asset Purchase Agreement and shall not receive distributions pursuant to this Plan.
- 1.110 Overpayment Amount has the meaning given to such term in Section 2.3 of the Plan.
- 1.111 Person means an individual, a corporation, a limited liability company, a partnership, an association, a joint stock company, a joint venture, an unincorporated organization, or a Governmental Unit.
 - 1.112 Petition Date means September 29, 2011.
- 1.113 Plan means this first amended joint plan of liquidation of the Debtors and all exhibits annexed hereto or referenced herein, as may be amended, modified or supplemented

from time to time in accordance with the provisions of the Plan or the Bankruptcy Code and Bankruptcy Rules.

- 1.114 Possessory Lienholder Claims means all Claims held by Persons to the extent such Claims are deemed to be secured, through a possessory Lien, in property in which any Estate has an interest, but only to the extent of the value of the possessory lienholders' interest in the Estate's interest in such property as of the Effective Date.
- 1.115 Prepetition Claims Motions means those certain documents, motions and pleadings, filed by the Debtors with the Bankruptcy Court on the Petition Date, pursuant to which the Debtors sought authorization to pay the prepetition obligations specifically described therein or any motions filed by the Debtors after the Petition Date consented to by the Approving Majority First Lien Lenders that seek authorization to pay the prepetition obligations specifically described therein.
- 1.116 Priority Tax Claim means any Claim entitled to priority pursuant to Bankruptcy Code Section 507(a)(8).
- 1.117 Pro Rata means proportionately so that the ratio of the amount of consideration distributed on account of a particular Allowed Claim to the amount of the Allowed Claim is the same as the ratio of the amount of consideration distributed on account of all Allowed Claims of the Class in which the particular Allowed Claim is included to the amount of all Allowed Claims of that Class, but in any event the amount of consideration distributed on account of an Allowed Claim shall not exceed 100% of the amount of the Allowed Claim.
- 1.118 Professional means a Person employed in the Debtors' Chapter 11 Cases pursuant to a Final Order in accordance with Sections 327 and 1103 of the Bankruptcy Code and

to be compensated for services rendered prior to the Effective Date, pursuant to Sections 327, 328, 329, 330, 331 and 363 of the Bankruptcy Code.

- 1.119 Professional Claim means a Claim of a Professional retained in the Debtors' Chapter 11 Cases by the Debtors or the Creditors' Committee pursuant to a Final Order in accordance with Sections 327 or 1103 of the Bankruptcy Code or otherwise, for compensation or reimbursement of actual and necessary costs and expenses relating to services incurred after the Petition Date and prior to and including the Effective Date pursuant to Sections 327, 328, 329, 330, 331 and 363 of the Bankruptcy Code.
- 1.120 Purchaser means the purchaser of all or substantially all of the Debtors' Assets pursuant to the Sale Order.
- 1.121 Released Claims means any Claims, rights or Causes of Action of any of the Debtors or their Estates released pursuant to the DIP Order, the Asset Purchase Agreement, the Sale Order, the Plan or the Confirmation Order.
- 1.122 Remaining Contracts has the meaning given to such term in Section 6.1 of the Plan.
- 1.123 Reserve Accounts means the CCR Account, the Designated Account, the Employee Account, and the Lazard Account.
- 1.124 Sale Allocation Motion means the Debtors' Motion for Entry of an Order Determining the Value of the Assets of Graceway Canada Company Proposed to be Purchased Under the Stalking Horse Asset Purchase Agreement [Docket No. 134] dated October 17, 2011 and the Supplement to the Debtors' Motion for Entry of an Order Determining the Value of the Assets of Graceway Canada Company Proposed to be Purchased Under the Stalking Horse Asset Purchase Agreement [Docket No. 153] dated October 25, 2011, pursuant to which the Debtors

sought approval of the allocation of \$4,404,683.65 of the Sale Proceeds to the assets owned by non-Debtor Graceway Canada and allocation of the remainder of the Sale Proceeds to the Assets owned by the Debtors. For the avoidance of doubt, after adjustments, \$4,253,828.76 of the Sale Proceeds were ultimately allocated to the assets owned by non-Debtor Graceway Canada.

- 1.125 Sale Motion means Debtors' Motion for Entry of (I) An Order Approving and Authorizing (A) Bidding Procedures in Connection with the Sale of Certain Assets of the Debtors, (B) Stalking Horse Bid Protections, (C) The Form and Manner of Notice of the Sale Hearing and (D) Related Relief; and (II) An Order Authorizing (A) The Sale of Certain Assets of the Debtors Free and Clear of all Claims, Liens, Liabilities, Rights, Interests and Encumbrances; (B) The Debtors to Enter into and Perform their Obligations under the Asset Purchase Agreement; (C) The Debtors to Assume and Assign Certain Executory Contracts and Unexpired Leases; and (D) Granting Related Relief [Docket No. 12] dated September 29, 2011, pursuant to which the Debtors sought approval of the 363 Sale.
- 1.126 Sale Order means the Order Authorizing (A) The Sale of Certain Assets of the Debtors Free and Clear of All Claims, Liens, Liabilities, Rights, Interests and Encumbrances; (B) The Debtors to Enter into and Perform Their Obligations Under the Asset Purchase Agreement; (C) The Debtors to Assume and Assign Certain Executory Contracts and Unexpired Leases; and (D) Granting Related Relief [Docket No. 306] dated November 22, 2011.
 - 1.127 Sale Proceeds means the aggregate Cash proceeds of the 363 Sale.
- 1.128 Sale Support Agreement means the Sale Support Agreement, dated as of September 28, 2011, entered by and among the Debtors, Graceway Canada and the First Lien Facility Lenders from time to time party thereto, as it may be amended, supplemented or otherwise modified from time to time.

- 1.129 Scheduled means as set forth on the Schedules.
- 1.130 Schedules means the Schedules of Assets and Liabilities filed by the Debtors in accordance with Bankruptcy Code Section 521 and Federal Bankruptcy Rule 1007, as the same may be amended from time to time prior to the Effective Date in accordance with Federal Bankruptcy Rule 1009.
- 1.131 Second Lien Credit Agreement means that certain \$330,000,000 Second Lien Credit Agreement dated as of May 3, 2007, among, *inter alia*, Graceway Pharmaceuticals, LLC, as borrower, Graceway Holdings, LLC, the Second Lien Facility Lenders, and the Second Lien Facility Agent, as amended.
- 1.132 Second Lien Facility Adequate Protection Claim means the Claims of the Second Lien Facility Agent and the Second Lien Facility Lenders under Section 507(b) of the Bankruptcy Code and Section 4(b) of the DIP Order, if any.
- 1.133 Second Lien Facility Agent means Deutsche Bank Trust Company Americas, as agent for the Second Lien Facility Lenders under the Second Lien Credit Agreement, or any successor thereto.
- 1.134 Second Lien Facility Claim means any Second Lien Facility Secured
 Claim and any Second Lien Facility Adequate Protection Claim.
- 1.135 Second Lien Facility Deficiency Claim means any Deficiency Claim with respect to the Second Lien Facility Secured Claim.
- 1.136 Second Lien Facility Lenders means the "Lenders" as defined in the Second Lien Credit Agreement.

- 1.137 Second Lien Facility Secured Claim means the Secured Claim of the Second Lien Facility Agent on behalf of the Second Lien Facility Lenders under the Second Lien Credit Agreement and the DIP Order, if any.
- 1.138 Second Lien Obligations means all "Second Lien Obligations" as defined in the Second Lien Credit Agreement (including, without limitation, interest accruing after the Petition Date).
- 1.139 Secured Claim means any Claim of a Creditor, including for principal, interest and any other amounts, that is secured by a valid, perfected, and enforceable Lien that is not subject to avoidance under applicable bankruptcy or non-bankruptcy law, against property of any of the Debtors' Estates (or the proceeds of such property), that is a Possessory Lienholder Claim or that is otherwise subject to setoff under Bankruptcy Code Section 553, to the extent of the value of such Creditor's interest in that Estate's interest in such property or to the extent of the amount subject to setoff, as applicable, as determined pursuant to Bankruptcy Code Section 506(a).
- 1.140 Security means any instrument issued by, or interest in, any of the Debtors of the type described in Bankruptcy Code Section 101(49).
- 1.141 Tax Claims has the meaning given to such term in Section 2.5.2 of the Plan.
- 1.142 Taxes means all income, franchise, excise, sales, use, employment, withholding, property, payroll or other taxes, assessments, or governmental charges, together with any interest, penalties, additions to tax, fines, and similar amounts relating thereto, imposed or collected by any federal, state, local or foreign governmental authority on or from any of the Debtors.

- 1.143 Trust Administrative Fund shall have the meaning specified in Section
 4.1 of the Liquidating Trust Agreement and shall be funded solely in accordance with Section
 7.8 below.
- 1.144 Trust Estate or Trust Assets means the Assets (other than the Nycomed Litigation Proceeds), including, without limitation, the Other Assets, all of which shall be irrevocably assigned, transferred and conveyed to the Liquidating Trust as of the Effective Date of the Plan, plus any and all net income earned on the foregoing.
 - 1.145 Unimpaired means any Claim that is not Impaired.
- 1.146 United States Trustee means the United States Trustee appointed under Section 581(a)(3) of title 28 of the United States Code to serve in the District of Delaware.
- 1.147 Unsecured Claim means any Claim against the Debtors, excluding Administrative Expense Claims, Priority Tax Claims, Other Priority Claims, the First Lien Facility Secured Claim, any First Lien Facility Adequate Protection Claim, the Second Lien Facility Secured Claim, any Second Lien Facility Adequate Protection Claim and Other Secured Claims. Unsecured Claims shall include (i) any Indemnification Claims arising from or attributable to actions prior to the Petition Date but only to the extent not covered by applicable directors' and/or officers' insurance coverage and (ii) all Deficiency Claims. Any Claim that is an Assumed Liability has been Assumed and shall be satisfied in accordance with the Asset Purchase Agreement and shall not receive distributions pursuant to this Plan.

C. Rules of Interpretation

- 1. In the event of an inconsistency, (a) the provisions of the Plan shall control over the contents of the Disclosure Statement and Liquidating Trust Agreement and (b) the provisions of the Confirmation Order shall control over the contents of the Plan.
 - 2. For the purposes of the Plan:
 - (a) any reference in the Plan to a contract, instrument, release or other agreement or document being in a particular form or on particular terms and conditions means that such document shall be substantially in such form or substantially on such terms and conditions; *provided*, *however*, that any change to such form, terms or conditions that is material to a party to such document shall not be modified without such party's consent unless such document expressly provides otherwise;
 - (b) any reference in the Plan to an existing document, exhibit or schedule Filed or to be Filed means such document, exhibit or Plan schedule, as it may have been or may be amended, modified or supplemented as of the Effective Date;
 - (c) unless otherwise specified, all references in the Plan to "Sections," "Articles," "Exhibits" and "Plan Schedules" are references to Sections, Articles, Exhibits and Plan Schedules of or to the Plan;
 - (d) the words "herein," "hereof," "hereto," "hereunder" and others of similar import refer to the Plan in its entirety rather than to only a particular portion of the Plan;
 - (e) captions and headings to Articles and Sections are inserted for convenience of reference only and are not intended to be part of or to affect interpretations of the Plan; and

- (f) the word "including" means "including without limitation."
- 3. Whenever a distribution of property is required to be made on a particular date, the distribution shall be made on such date or as soon as reasonably practicable thereafter.
- 4. Whenever a reference is made in Article II hereof to a limitation on payment of Allowed Claims by an "amount provided for in the Budget" or words to that effect, it shall be understood that such amounts provided for in the Budget for any specific category of Claims shall be augmented, as to all categories of Claims other than Claims payable from the CCR Account (regardless whether such Claims are actually paid from the CCR Account), Claims payable from the Employee Account (regardless whether such Claims are actually paid from the Employee Account) and Professional Claims, by the remaining amount provided for in the Budget for "Post-Closing Other Miscellaneous," which, as of the date hereof, is \$1,250,000.00.
- 5. All Exhibits to the Plan are incorporated into the Plan and shall be deemed to be included in the Plan, regardless of when they are Filed.
- 6. Subject to the provisions of any contract, certificate, bylaws, instrument, release or other agreement or document entered into in connection with the Plan, the rights and obligations arising under the Plan shall be governed by, and construed and enforced in accordance with, federal law, including the Bankruptcy Code and Bankruptcy Rules.
- 7. This Plan is the product of extensive discussions and negotiations between and among, *inter alia*, the Debtors, the First Lien Facility Agent, the Second Lien Facility Agent, the Creditors' Committee and certain other creditors and constituencies. Each of the foregoing was represented by counsel who either (a) participated in the formulation and documentation of or (b) was afforded the opportunity to review and provide comments on, the Plan, the Disclosure Statement, and the documents ancillary thereto. Accordingly, unless explicitly indicated

otherwise, the general rule of contract construction known as "contra proferentum" shall not apply to the construction or interpretation of any provision of this Plan, the Disclosure Statement, any of the Plan Exhibits, or any contract, instrument, release, indenture, or other agreement or document generated in connection herewith.

D. Computation of Time. In computing any period of time prescribed or allowed by the Plan, unless otherwise expressly provided, the provisions of Federal Bankruptcy Rule 9006(a) shall apply.

ARTICLE 2 TREATMENT OF UNCLASSIFIED CLAIMS

Administrative Expense Claims. Subject to the allowance procedures 2.1 and deadlines provided herein, on the Effective Date or as soon thereafter as is practicable in recognition of the applicable Claims resolution process set forth herein, the holder of an Allowed Administrative Expense Claim (other than an Administrative Expense Claim that is an Assumed Liability or a Professional Claim) shall receive on account of such Allowed Administrative Expense Claim and in full satisfaction, settlement and release of and in exchange for such Allowed Administrative Expense Claim, (i) Cash equal to the unpaid portion of such Allowed Administrative Expense Claim, or (ii) such other treatment as to which the Debtors, or, if after the Effective Date, the Liquidating Trustee, and the holder of such Allowed Administrative Expense Claim have agreed upon in writing provided such treatment is on more favorable terms to the Debtors or the Liquidating Trustee, as the case may be, than the treatment set forth in subsection (i) hereof; provided, however, that (a) notwithstanding any other provision of this Plan, Administrative Expense Claims that are Assumed Liabilities have been Assumed and shall be satisfied in accordance with the Asset Purchase Agreement and shall not receive distributions pursuant to this Plan and (b) Professional Claims shall be paid in accordance with Section 2.3.

- 2.2 Statutory Fees. On or before the Effective Date, all fees due and payable pursuant to 28 U.S.C. § 1930, as determined by the Bankruptcy Court at the Confirmation Hearing, shall be paid in full, in Cash.
- 2.3 Immediately prior to the Effective Date, the Professional Claims. Debtors shall pay from the Designated Account all amounts owing to the Professionals other than Lazard and from the Lazard Account all amounts owing to Lazard for all outstanding Professional Claims relating to prior periods and for the period ending on the Effective Date; provided, however, that the amount paid to each Professional immediately prior to the Effective Date shall not exceed the lesser of (a) the excess, if any, of (i) the amount provided for in the Budget for such Professional over (ii) the sum of the aggregate amount of all payments already made to such Professional (including the application of any retainer held by such Professional) for services rendered since the Petition Date and the amount of any retainers currently held by such Professional and (b) the sum of (i) the amount billed and unpaid for such Professional and (ii) the amount estimated by such Professional as necessary to cover unbilled periods through the Effective Date, which estimated amount shall be provided to the Debtors and the First Lien Facility Agent by each Professional on or prior to the Effective Date. The Professionals shall estimate Professional Claims due for periods that have not been billed as of the Effective Date. On or prior to the Administrative Expense Claims Bar Date, each Professional shall File with the Bankruptcy Court its final fee application seeking final approval of all fees and expenses from the Petition Date through the Effective Date. Within ten (10) days after entry of a Final Order with respect to its final fee application, each Professional shall remit any overpayment (the "Overpayment Amount") to the Liquidating Trustee to be deposited in the Designated Account or the Liquidating Trustee shall pay any outstanding amounts owed to the Professional, but in no

event shall any Professional receive more than the amount provided in the Budget for such Professional from the Designated Account. Payment of Allowed Professional Claims asserted by any Professional shall be funded from Other Assets (other than the Liquidating Trustee Fee Amount) to the extent that the aggregate amount of such Allowed Professional Claims of such Professional since the Petition Date exceeds the amount provided for in the Budget for such Professional.

2.4 Priority Tax Claims. Subject to the allowance procedures and deadlines provided herein, on the Effective Date or as soon thereafter as is practicable in recognition of the applicable Claims resolution process set forth herein, with respect to each Allowed Priority Tax Claim, at the sole option of the Debtors or, if after the Effective Date, the Liquidating Trustee, the holder of an Allowed Priority Tax Claim shall be entitled to receive on account of such Allowed Priority Tax Claim, in full satisfaction, settlement and release of and in exchange for such Allowed Priority Tax Claim, (i) Cash in an amount equal to the unpaid portions of such Allowed Priority Tax Claims; or (ii) such other treatment agreed to in writing by the holder of such Allowed Priority Tax Claim and the Debtors or, if after the Effective Date, the Liquidating Trustee, provided such treatment is on more favorable terms to the Debtors or the Liquidating Trustee, as the case may be, than the treatment set forth in subsection (i) hereof.

2.5 Deadline for Filing Administrative Expense Claims.

2.5.1 <u>Administrative Expense Claims Other Than Tax Claims.</u>
Other than with respect to (i) Administrative Expense Claims for which the Bankruptcy Court previously has established a Bar Date, and (ii) Tax Claims addressed in Section 2.5.2 below, any and all requests for payment or proofs of Administrative Expense Claims, including Claims of all Professionals or other Entities requesting compensation or reimbursement of expenses pursuant

to Bankruptcy Code Sections 327, 328, 330, 331, 503(b) or 1103 for services rendered on or before the Effective Date (including any compensation requested by any Professional or any other Entity for making a substantial contribution in the Chapter 11 Cases), must be Filed and served on the Liquidating Trustee and its counsel no later than the Administrative Expense Claims Bar Date. Objections to any such Administrative Expense Claims must be Filed and served on the claimant no later than sixty (60) days after the Administrative Expense Claims Bar Date, which date may be extended by application to the Bankruptcy Court. The Liquidating Trustee shall use reasonable efforts to promptly and diligently pursue resolution of any and all Disputed Administrative Expense Claims. Holders of Administrative Expense Claims, including all Professionals or other Entities requesting compensation or reimbursement of expenses pursuant to Bankruptcy Code Sections 327, 328, 330, 331, 503(b) or 1103 for services rendered on or before the Effective Date (including any compensation requested by any Professional or any other Entity for making a substantial contribution in the Chapter 11 Cases), that are required to File a request for payment or proof of such Claims and that do not File such requests or proofs of Claim on or before the Administrative Expense Claims Bar Date shall be forever barred from asserting such Claims against the any of the Debtors, their Estates, or the Liquidating Trust.

2.5.2 <u>Tax Claims</u>. All requests for payment of Claims pursuant to Section 503 of the Bankruptcy Code by a Governmental Unit for Taxes (and for interest and/or penalties or other amounts related to such Taxes) for any tax year or period, all or any portion of which occurs or falls within the period from and including the Petition Date through and including the Effective Date, and for which no bar date has otherwise been previously established ("<u>Tax Claims</u>"), must be Filed on or before the later of: (i) thirty (30) days following

the Effective Date; or (ii) to the extent applicable, sixty (60) days following the filing of a tax return for such Taxes (if such Taxes are assessed based on a tax return) for such tax year or period with the applicable Governmental Unit. Objections to any Tax Claims must be Filed within thirty (30) days after the later of the Administrative Expense Claims Bar Date or sixty (60) days following the filing of a tax return for such Taxes, which objection dates may be extended by application to the Bankruptcy Court. Any holder of a Tax Claim that is required to File a request for payment of such Tax Claim and other amounts due related to such Taxes and which does not File such a Claim by the applicable bar date shall be forever barred from asserting any such Claim against any of the Debtors or any non-Debtor member of the Debtors' consolidated tax group, the Debtors' Estates, the Liquidating Trust or any other Entity, or their respective property, whether any such Claim is deemed to arise prior to, on, or subsequent to the Effective Date, and shall receive no distribution under the Plan or otherwise on account of such Claim.

ARTICLE 3 CLASSIFICATION AND TREATMENT OF CLASSIFIED CLAIMS AND INTERESTS

A. General. Pursuant to Section 1122 of the Bankruptcy Code, set forth below is a designation of the Classes of Claims against and Interests in the Debtors. A Claim or Interest is placed in a particular Class only to the extent that such Claim or Interest falls within the description of that Class. A Claim or Interest is also placed in a particular Class for purposes of receiving a distribution under the Plan, but only to the extent such Claim or Interest is an Allowed Claim or Interest and has not been paid, released, or otherwise settled prior to the Effective Date. Except as otherwise expressly set forth in this Plan, a Claim or Interest which is not an Allowed Claim or Allowed Interest shall not receive any payments, rights or distributions under this Plan. In accordance with Section 1123(a)(1) of the Bankruptcy Code, Administrative

Expense Claims of the kinds specified in Section 507(a)(2) and (3) and Priority Tax Claims of the kinds specified in Section 507(a)(8) of the Bankruptcy Code have not been classified and are treated as set forth in Article 2 above.

- B. Classification. As stated above, the Plan is premised on the substantive consolidation of the Debtors with respect to the voting of all Claims and Interests and the treatment of all Claims and Interests other than Other Secured Claims, as provided below. The following summary is for the convenience of all interested parties and is superseded for all purposes by the classification, description and treatment of Claims and Interests in Articles 4 and 5 of the Plan.
- 3.1 Class 1: Other Priority Claims. Class 1 consists of all Other Priority Claims.
- 3.2 Class 2: First Lien Facility Claims. Class 2 consists of the First Lien Facility Secured Claim, any First Lien Facility Deficiency Claim and any First Lien Facility Adequate Protection Claim.
- 3.3 Class 3: Second Lien Facility Claims. Class 3 consists of any Second Lien Facility Secured Claim and any Second Lien Facility Adequate Protection Claim.
- 3.4 Class 4: Other Secured Claims. Class 4 consists of all Other Secured Claims.
- 3.5 Class 5: General Unsecured Claims. Class 5 consists of all General Unsecured Claims.
- 3.6 Class 6: Intercompany Claims. Class 6 consists of all Intercompany Claims.
- 3.7 Class 7: Old Equity. Class 7 consists of all Interests arising from or relating to Old Equity.

3.8 Class 8: Old Equity Rights. Class 8 consists of all Claims and Interests arising from or relating to Old Equity Rights.

ARTICLE 4 <u>IDENTIFICATION OF CLASSES OF CLAIMS AND INTERESTS IMPAIRED AND</u> NOT IMPAIRED BY THE PLAN

- 4.1 Unimpaired Classes of Claims and Interests. Class 1 Other Priority

 Claims and Class 4 Other Secured Claims are not Impaired by the Plan.
- 4.2 Impaired Classes of Claims and Interests. Class 2 First Lien Facility Claim, Class 3 Second Lien Facility Claim, Class 5 General Unsecured Claims, Class 6 Intercompany Claims, Class 7 Old Equity, and Class 8 Old Equity Rights are Impaired by the Plan.

ARTICLE 5 PROVISIONS FOR THE TREATMENT OF CLAIMS AND INTERESTS

- thereafter as is reasonably practicable in recognition of the applicable Claims reconciliation process set forth herein, the holder of an Allowed Other Priority Claim shall receive on account of the Allowed Other Priority Claim and in full satisfaction, settlement and release of and in exchange for such Allowed Other Priority Claim, (i) Cash equal to the amount of such Allowed Other Priority Claim, or (ii) such other treatment as to which the Debtors and the holder of such Allowed Other Priority Claim have agreed upon in writing, provided such treatment is on more favorable terms to the Debtors or the Liquidating Trustee, as the case may be, than the treatment set forth in subsection (i) hereof.
- 5.2 Class 2 (First Lien Facility Claims). On the Effective Date or as soon thereafter as is reasonably practicable in recognition of the applicable Claims reconciliation

process set forth herein, the First Lien Facility Agent shall receive, for the benefit of the First Lien Claimholders, in full satisfaction, settlement and release of and in exchange for the First Lien Facility Claims (including the First Lien Facility Secured Claim, any First Lien Facility Adequate Protection Claim and any First Lien Facility Deficiency Claim), (i) any First Lien Available Cash that is available to be distributed on the Effective Date; (ii) the Nycomed Litigation Proceeds; and (iii) beneficial interests in the Liquidating Trust entitling each First Lien Claimholder to receive its Pro Rata share of (A) any First Lien Available Cash that becomes available to be distributed after the Effective Date, (B) seventy-five percent (75%) of any Avoidance Action Proceeds, (C) one hundred percent (100%) of any other Net Causes of Action Proceeds, including, without limitation, 549 Avoidance Action Proceeds, and (D) all Cash from the collection of accounts receivable. Until the First Lien Facility Claims are paid in full in Cash, the Liens securing the First Lien Facility Secured Claim and any First Lien Facility Adequate Protection Claim shall remain on the Assets (including, without limitation, the Reserve Accounts and the Cash deposited therein and the Nycomed Litigation Proceeds) other than the Other Assets, provided that, except for their obligations under this Plan, the Debtors and their Estates shall have no further liability to the First Lien Facility Agent or the First Lien Facility Lenders. Notwithstanding the full satisfaction, settlement and release of the First Lien Facility Claims hereunder, the aggregate amount of the beneficial interests in the Liquidating Trust the First Lien Claimholders receive under this Section 5.2 in exchange for such Claims shall increase from time to time by an amount equal to the interest accrual on, and the fees incurred on account of, such Claims to and including the applicable date of determination (calculated at the rate provided for under the First Lien Credit Agreement and in accordance with the terms thereof and of the Intercreditor Agreement) as if such Claims had not been satisfied, settled and released hereunder; it being understood that distributions on account of such interest accrual and incurred fees shall only be permitted in the event the First Lien Facility Secured Claim (had it not been satisfied, settled and released hereunder) is oversecured. Holders of the First Lien Facility Claims shall be Beneficiaries; it being understood that distributions from the Liquidating Trust shall be exclusively governed in accordance with the terms set forth in this Plan.

Class 3 (Second Lien Facility Claims). The Second Lien Facility Agent shall receive, for the benefit of the Second Lien Facility Lenders, in full satisfaction, settlement and release of and in exchange for any Allowed Second Lien Facility Secured Claim and any Allowed Second Lien Facility Adequate Protection Claim on the Effective Date or as soon thereafter as is practicable in recognition of the applicable Claims reconciliation process set forth herein, beneficial interests in the Liquidating Trust entitling each Second Lien Facility Lender to receive its Pro Rata share of (i) any Other Available Cash subject to the Liens granted to the Second Lien Facility Agent pursuant to the Second Lien Credit Agreement or the DIP Order and (ii) after payment in full in Cash of the First Lien Facility Claim, (a) any First Lien Available Cash; (b) the Nycomed Litigation Proceeds, (c) seventy-five percent (75%) of any Avoidance Action Proceeds; (d) one hundred percent (100%) of any other Net Causes of Action Proceeds, including, without limitation, any 549 Avoidance Action Proceeds; and (e) all Cash from the collection of accounts receivable. Until the Second Lien Facility Claims are paid in full in Cash, the Liens securing any Second Lien Facility Secured Claim and any Second Lien Facility Adequate Protection Claim shall remain on the Assets, provided that, (i) except for their obligations under this Plan, the Debtors and their Estates shall have no further liability to the Second Lien Facility Agent or the Second Lien Facility Lenders and (ii) such Liens shall continue to be junior and subordinate in all respects to the Liens securing the First Lien Facility Secured Claim and First Lien Adequate Protection Claim that have not been released in connection with the Committee Settlement until such Claims are paid in full in Cash. Any Second Lien Facility Deficiency Claim shall not constitute Class 3 Second Lien Facility Claims and shall be treated as a Class 5 General Unsecured Claim hereunder. Holders of the Allowed Second Lien Facility Claims shall be Beneficiaries; it being understood that distributions from the Liquidating Trust shall be exclusively governed in accordance with the terms set forth in this Plan.

Class 4 et seq. (Other Secured Claims). This Class is subdivided into 5.4 subclasses designated by letters of the alphabet (Class 4A, Class 4B and so on), so that each holder of any Other Secured Claim is in a Class by itself, except to the extent that there are Other Secured Claims that are substantially similar to each other and may be included within a single Class, and except for a precautionary class of otherwise unclassified Other Secured Claims. On the Effective Date or as soon thereafter as is practicable in recognition of the applicable Claims reconciliation process set forth herein, each holder of an Allowed Other Secured Claim that was not assumed by the Purchaser in connection with the 363 Sale shall receive, in full satisfaction, settlement and release of and in exchange for such Allowed Other Secured Claim, (i) the collateral securing any such Allowed Other Secured Claim (to the extent such collateral does not constitute collateral securing the First Lien Facility Secured Claim, any First Lien Facility Adequate Protection Claim, the Second Lien Facility Secured Claim or any Second Lien Facility Adequate Protection Claim or, if such collateral does secure any such Claim, to the extent that the Lien securing such Allowed Other Secured Claim is senior to the Lien securing the First Lien Facility Secured Claim, any First Lien Facility Adequate Protection Claim, the Second Lien Facility Secured Claim or any Second Lien Facility Adequate Protection Claim, as applicable),

- (ii) Cash in an amount equal to such Allowed Other Secured Claim to the extent the collateral securing such Allowed Other Secured Claim was sold and such collateral does not constitute collateral securing the First Lien Facility Secured Claim, any First Lien Facility Adequate Protection Claim, the Second Lien Facility Secured Claim or any Second Lien Facility Adequate Protection Claim or, if such collateral does secure any such Claim, to the extent that the Lien securing such Allowed Other Secured Claim is senior to the Lien securing the First Lien Facility Secured Claim, any First Lien Facility Adequate Protection Claim, the Second Lien Facility Secured Claim or any Second Lien Facility Adequate Protection Claim, as applicable, or (iii) such other treatment that leaves such Allowed Other Secured Claim Unimpaired pursuant to the Section 1124(2) of the Bankruptcy Code. The Other Secured Claim Liens shall be released and the Debtors and their Estates shall have no further liability therefor; provided, however, that any Deficiency Claims of holders of Class 4 Other Secured Claims shall not constitute Class 4 Other Secured Claims and shall be treated as Class 5 General Unsecured Claims hereunder.
- thereafter as is practicable in recognition of the applicable Claims reconciliation process set forth herein, the holders of Allowed General Unsecured Claims shall receive, in full satisfaction, settlement and release of and in exchange for such Allowed Claims, their Pro Rata share of the Other Available Cash that is not subject to the Liens granted to the Second Lien Facility Agent pursuant to the Second Lien Credit Agreement or the DIP Order, after payment in full of all Allowed Other Secured Claims (to the extent not otherwise satisfied in accordance with the terms of this Plan), Allowed Administrative Expense Claims (other than Administrative Expense Claims that are Assumed Liabilities), Allowed Priority Tax Claims and Allowed Other Priority Claims; provided, however, that pursuant to and in accordance with the Mezzanine Credit

Agreement, any amounts distributed on account of the Allowed Mezzanine Facility Claim (a) are subject and subordinate to payment in full in Cash of the Second Lien Obligations and (b) shall be distributed directly to the Second Lien Facility Agent for the benefit of the holders of Allowed Second Lien Facility Claims until the Second Lien Obligations are paid in full in Cash; provided further, however, that upon notice to the Liquidating Trustee from the Second Lien Facility Agent that the Second Lien Facility Agent has received \$10,000,000.00 in Cash from the Other Available Cash for distribution to holders of Allowed Second Lien Facility Claims in excess of any legal, trustee or other fees and expenses that constitute Second Lien Obligations, the Liquidating Trustee shall cease any distributions to the Second Lien Facility Agent solely on account of the Allowed Mezzanine Facility Claim until the Liquidating Trustee has distributed \$250,000.00 on account of the Allowed Mezzanine Facility Claim (which shall be paid out as if no subordination were in place), after which time the Liquidating Trustee shall resume any distributions to the Second Lien Facility Agent on account of the Allowed Mezzanine Facility Claim until the Second Lien Obligations are paid in full in Cash. Subject to the foregoing proviso, in the event any distribution on account of the Allowed Mezzanine Facility Claim is made other than to the Second Lien Facility Agent, such distribution shall be held in trust for the benefit of, and paid over and delivered to, the Second Lien Facility Agent for the benefit of the Second Lien Facility Lenders. Holders of Allowed General Unsecured Claims shall be Beneficiaries; it being understood that distributions from the Liquidating Trust shall be exclusively governed in accordance with the terms set forth in this Plan and beneficial interests in the Liquidating Trust granted to such holders will not entitle them to any distribution of more than twenty-five percent (25%) of any Avoidance Action Proceeds or any distribution of any other Causes of Action Proceeds, including, without limitation, any 549 Avoidance Action Proceeds. In addition, the holders of Allowed General Unsecured Claims shall be entitled to share Pro Rata in any amount remaining in or released from the Disputed Claims Reserve after resolution of all Disputed Unsecured Claims.

- 5.6 Class 6 (Intercompany Claims). As a result of substantive consolidation of the Debtors for distribution purposes under the Plan as provided in Section 7.1 hereof, holders of Intercompany Claims will not receive any distribution of property under the Plan on account of their Intercompany Claims and, on the Effective Date, the Intercompany Claims will be cancelled; *provided*, *however*, Class 6 shall exclude any Claims of a Debtor against a non-Debtor Affiliate.
- 5.7 Class 7 (Old Equity). Holders of Old Equity Interests will not receive and will not retain any property of the Debtors under the Plan on account of such Interests and all Old Equity Interests will be cancelled as of the Effective Date.
- 5.8 Class 8 (Old Equity Rights). Holders of any Claim or Interest arising from or relating to Old Equity Rights will not receive and will not retain any property of the Debtors under the Plan on account of such Claim or Interest and all Old Equity Rights and any Interests or Claims arising from or relating thereto will be cancelled as of the Effective Date.

ARTICLE 6 TREATMENT OF EXECUTORY CONTRACTS AND UNEXPIRED LEASES

6.1 Assumption, Assignment and Rejection Pursuant to the Sale Order. Pursuant to the Bid Procedures Order and the Sale Order, the Bankruptcy Court established deadlines and procedures for the assumption and rejection of executory contracts and unexpired leases by the Debtors at the direction of the Purchaser and allocated responsibility for payment of cure amounts between the Debtors and the Purchaser. Any executory contracts and unexpired leases of the Debtors, including executory contracts and unexpired leases entered into after the

Petition Date, not assumed and assigned to the Purchaser or rejected prior to the Effective Date (the "<u>Remaining Contracts</u>") shall be rejected pursuant to Section 6.5 of this Plan unless assumed or assumed and assigned pursuant to Section 6.2 of this Plan.

- 6.2 Assumption and Assignment of Remaining Contracts. Effective Date, the Debtors shall assume and assign to the Liquidating Trust, pursuant to Bankruptcy Code Section 365, each of the Remaining Contracts that are identified in Exhibit 3 to this Plan (the "Assumption Schedule") that have not expired under their own terms prior to the Effective Date; provided that the Assumption Schedule to this Plan shall be acceptable to the Liquidating Trustee (or the proposed Liquidating Trustee) in his or her sole discretion. The Debtors reserve the right to amend the Assumption Schedule not later than twenty-eight (28) days prior to the Confirmation Hearing either to: (i) delete any executory contract or lease listed therein and provide for its rejection pursuant to Section 6.5 hereof; or (ii) add any executory contract or lease to the Assumption Schedule, thus providing for its assumption and assignment pursuant to this Section 6.2; provided, however, that any amendments to the Assumption Schedule shall be acceptable to the Liquidating Trustee (or the proposed Liquidating Trustee) in his or her sole discretion. The Debtors shall provide notice of any such amendment to the Assumption Schedule to the parties to the executory contract or lease affected thereby not later than twenty-eight (28) days prior to the Confirmation Hearing. The Confirmation Order shall constitute an Order of the Bankruptcy Court pursuant to Bankruptcy Code Section 365 approving all assumptions and assignments described in this Section 6.2 and the Assumption Schedule, as of the Effective Date.
- 6.3 Cure Payments; Assurance of Performance. Any monetary defaults under each Remaining Contract to be assumed and assigned under the Plan shall be satisfied,

pursuant to Bankruptcy Code Section 365(b)(1) by payment of the cure amount in Cash in full on the Effective Date by the Debtors from the Designated Account to the extent permitted by the Budget and/or the Sale Support Agreement and from the Other Assets to the extent not permitted by the Budget and/or the Sale Support Agreement. In the event of a dispute regarding (i) the amount or timing of any cure payments, (ii) the ability of the Debtors, the Liquidating Trustee, or an assignee thereof to provide adequate assurance of future performance under the Remaining Contracts to be assumed or assumed and assigned, as applicable, or (iii) any other matter pertaining to assumption or assumption and assignment of the Remaining Contracts to be assumed, the Liquidating Trustee shall pay all required cure amounts promptly following the entry of a Final Order resolving the dispute from the Designated Account to the extent permitted by the Budget and/or the Sale Support Agreement and from the Other Assets to the extent not permitted by the Budget and/or the Sale Support Agreement.

6.4 Objections To Assumption of Remaining Contracts. To the extent that any party to a Remaining Contract identified for assumption and assignment asserts arrearages or damages pursuant to Bankruptcy Code Section 365(b)(1), or has any other objection with respect to any proposed assumption, cure or assignment on the terms and conditions provided herein, all such arrearages, damages and objections must be Filed and served no later than the earlier of (i) twenty-one (21) days after such party is served with notice of such assumption and assignment and (ii) one (1) day prior to the Confirmation Hearing.

Failure to assert such arrearages, damages or objections in the manner described above shall constitute consent to the proposed assumption, cure or assignment on the terms and conditions provided herein, including an acknowledgement that the proposed assumption and/or assignment provides adequate assurance of future performance and that the

amount identified for "cure" in the Assumption Schedule is the amount necessary to cover any and all outstanding defaults under the Remaining Contract to be assumed, as well as an acknowledgement and agreement that no other defaults exist under such Remaining Contract.

If any assumption of a Remaining Contract proposed herein for any reason is not approved by the Bankruptcy Court, then the Debtors shall be entitled, in their sole discretion, upon written notice to the applicable non-Debtor party to such Remaining Contract, to deem such Remaining Contract to have been rejected pursuant to the provisions of Section 6.5 below.

- 6.5 Rejection. Except for those executory contracts and unexpired leases that (i) are assumed and assigned pursuant to this Plan, (ii) have been previously assumed and assigned or rejected pursuant to previous Orders of the Bankruptcy Court, or (iii) are the subject of a pending motion before the Bankruptcy Court with respect to the assumption and assignment of such executory contracts and unexpired leases, as of the Effective Date, all executory contracts and unexpired leases of the Debtors shall be rejected pursuant to Section 365 of the Bankruptcy Code; provided, however, that neither the inclusion by the Debtors of a contract or lease on the Assumption Schedule nor anything contained in this Article 6 shall constitute an admission by any Debtor that such contract or lease is an executory contract or unexpired lease or that any Debtor or its successors and assigns, including, but not limited to, the Liquidating Trust, has any liability thereunder.
- 6.6 Approval of Rejection; Rejection Damages Claims Bar Date. The Confirmation Order shall constitute an Order of the Bankruptcy Court approving the rejection of executory contracts and unexpired leases under Section 6.5 above pursuant to Bankruptcy Code Section 365 as of the Effective Date. Any Claim for damages arising from any such rejection

must be Filed within thirty (30) days after the mailing of notice of the entry of the Confirmation Order, or such Claim shall receive no distribution under the Plan or otherwise on account of such Claim. All Allowed Claims arising from the rejection of the Debtors' executory contracts or unexpired leases shall be classified as General Unsecured Claims and shall be treated in accordance with Section 5.5 of the Plan.

ARTICLE 7 MEANS FOR EXECUTION AND IMPLEMENTATION OF THE PLAN

7.1 Substantive Consolidation of Claims against Debtors for Plan Purposes Only. The Plan is premised on the substantive consolidation of all of the Debtors with respect to the treatment of all Claims and Interests, except for the Other Secured Claims in Class 4. The Plan does not contemplate substantive consolidation of the Debtors with respect to the Class 4 Claims, which shall be deemed to apply separately with respect to the Plan proposed by each Debtor. This Plan shall serve as a request by the Debtors, in lieu of a separate motion, to the Bankruptcy Court that it grant substantive consolidation with respect to the treatment of all Claims and Interests: on the Effective Date, (i) all Class 6 Intercompany Claims will be eliminated (except to the extent such Claims are by a Debtor against a non-Debtor Affiliate); (ii) no distributions shall be made under the Plan on account of any equity interest held by a Debtor in any other Debtor; (iii) all Assets and liabilities of the Debtors will be merged or treated as though they were merged; (iv) all guarantees of the Debtors of the obligations of any other Debtor and any joint or several liability of any of the Debtors shall be eliminated; and (v) each and every Claim or Interest against any Debtor shall be deemed Filed against the consolidated Debtors and all Claims Filed against more than one Debtor for the same liability shall be deemed one Claim against any obligation of the consolidated Debtors. In the event the Bankruptcy Court determines that such substantive consolidation of the Debtors is not appropriate, the Debtors may

request that the Bankruptcy Court otherwise confirm the Plan and the treatment of the different Classes under the Plan on a Debtor by Debtor basis.

7.2 The Liquidating Trust.

7.2.1 Establishment of the Liquidating Trust. On the Effective Date, the Debtors, on their own behalf and on behalf of the Beneficiaries, shall execute the Liquidating Trust Agreement and take all steps necessary to establish the Liquidating Trust. The initial Liquidating Trustee shall be Kip Horton of RPA Advisors, LLC.

7.2.2 Purpose of Liquidating Trust. The Liquidating Trust is being established for the sole purpose of liquidating the Debtors' Assets (other than the Nycomed Litigation Proceeds) and distributing the proceeds thereof to certain holders of Allowed Claims, in each case, as identified in and prescribed by this Plan. The Liquidating Trust shall not continue or engage in any trade or business, except to the extent reasonably necessary to, and consistent with, the liquidating purpose of the Liquidating Trust. Unless otherwise required by law, all parties shall treat the Liquidating Trust as a liquidating trust for all federal income tax purposes.

7.2.3 Contribution of the Debtors' Assets to the Liquidating Trust.

On the Effective Date of the Plan, each of the Debtors shall transfer and assign all of its respective Assets (other than the Nycomed Litigation Proceeds) to the Liquidating Trust free and clear of all Liens, Claims, interests and encumbrances (except that the Assets other than the Other Assets shall remain subject to the Liens securing (i) the First Lien Facility Secured Claim and any First Lien Facility Adequate Protection Claim as provided in Section 5.2 of the Plan and (ii) the Second Lien Facility Secured Claim and any Second Lien Facility Adequate Protection Claim as provided in Section 5.3 of the Plan).

- Date, but in no event later than ninety (90) days thereafter, the Liquidating Trustee, based upon his good faith determination after consultation with his counsel, shall inform the Beneficiaries in writing solely as to his estimate of the value of the Assets transferred to the Liquidating Trust. The valuation shall be used consistently by all parties (including, without limitation, the Debtors, the Liquidating Trustee, and the Beneficiaries) for all federal income tax purposes, and the parties shall file tax returns consistent with such valuation; *provided*, *however*, that such valuation shall not be binding on the Liquidating Trustee or any other party for any other purposes, including, without limitation, in regard to the liquidation of the Trust Assets, whether by disposition, liquidation, litigation, settlement, or otherwise.
- 7.2.5 <u>Certain Federal Income Tax Matters</u>. For U.S. federal income tax purposes: (i) the Debtors, the Liquidating Trustee and the Beneficiaries will treat the transfer of Assets to the Liquidating Trust as a transfer by the Debtors of the Debtors' Assets (net of any applicable liabilities) to the Beneficiaries (to the extent of the value of their respective interests in the applicable Trust Assets), followed by a transfer of such Assets (net of any applicable liabilities) by the Beneficiaries to the Liquidating Trust (to the extent of the value of their respective interests in the applicable Trust Assets); (ii) the Liquidating Trust (except with respect to the Disputed Claims Reserve) will be treated as one or more grantor trusts, and the Beneficiaries will be treated as the grantors and deemed owners of the Trust Assets; and (iii) the income of the Liquidating Trust will be treated as subject to tax on a current basis.
- 7.3 Continuation of the Automatic Stay. In furtherance of the implementation of the Plan, except as otherwise provided herein, all injunctions or stays provided for in the Chapter 11 Cases pursuant to sections 105 or 362 of the Bankruptcy Code, or

otherwise, and in existence on the Confirmation Date, shall remain in full force and effect and apply to all Beneficiaries and creditors holding Claims against the Debtors, the Estates, the Assets, the Liquidating Trustee, the Liquidating Trust and the Trust Assets until the Final Distribution Date.

- Date, (i) the Debtors' equity interests in Graceway Canada. On the Effective Date, (i) the Debtors' equity interests in Graceway Canada Company and (ii) the Debtors' Claims against Graceway Canada Company arising from any intercompany claims, loans, notes, transfers or other obligations shall be transferred by the Debtors to the Liquidating Trust, and the equity interests in Graceway Canada Company shall continue to be subject to the Liens securing the First Lien Facility Secured Claim and any First Lien Facility Adequate Protection Claim as provided in Section 5.2 of the Plan and the Liens of the Second Lien Facility Agent in favor of the Second Lien Facility Lenders as provided in Section 5.3 of the Plan; provided, however, that the portion of the Canadian Distribution Amount used to fund the Estate Fund Amount shall be free and clear of the Liens of the First Lien Facility Agent in accordance with the Committee Settlement. The Liquidating Trust shall exercise its powers as the sole shareholder to appoint the Liquidating Trustee as the sole officer and director of Graceway Canada Company and shall make distributions of the Canadian Distribution Amount as provided under the Plan.
- 7.5 Committee Settlement. The Committee Settlement is the cornerstone of the Plan. To the extent the Committee Settlement is not already deemed approved by the Sale Order, pursuant to section 363 of the Bankruptcy Code and Bankruptcy Rule 9019 and in consideration for the distributions and other benefits provided pursuant to the Plan, the provisions of the Plan relating to and effectuating the Committee Settlement shall constitute a good faith compromise. The entry of the Confirmation Order shall constitute the

Bankruptcy Court's approval of the Committee Settlement, as well as a finding by the Bankruptcy Court that the Committee Settlement is in the best interests of the Debtors, their Estates and holders of Claims and Interests and is fair, equitable and reasonable.

- 7.6 Post-Confirmation Operations. Following Confirmation and prior to the occurrence of the Effective Date, the then-current officers and directors of each of the Debtors shall continue in their respective capacities and the Debtors shall execute such documents and take such other actions as are necessary to effectuate the transactions provided for in this Plan. On and after the Effective Date, all such officers and directors shall be deemed to have resigned.
- 7.7 Post-Confirmation Funding of Plan. Payment of Allowed Other Secured Claims (to the extent not otherwise satisfied in accordance with the terms of this Plan), Allowed Administrative Expense Claims, Allowed Priority Tax Claims and Allowed Other Priority Claims shall be made from the Reserve Accounts to the extent permitted by the Budget and/or the Sale Support Agreement. To the extent any of such Allowed Claims are not satisfied from the Reserve Accounts (and, with respect to Allowed Other Secured Claims, are not otherwise satisfied in accordance with the terms of this Plan), such Allowed Claims shall be paid from the Other Available Cash. The Debtors anticipate that the Cash in the Reserve Accounts and the Other Available Cash shall be sufficient to pay all Allowed Other Secured Claims (to the extent not otherwise satisfied in accordance with the terms of this Plan), Allowed Administrative Expense Claims, Allowed Priority Tax Claims, Allowed Other Priority Claims and cure costs for executory contracts or unexpired leases on the Assumption Schedule, if any, in full in Cash on the Effective Date or once such Claims become Allowed Claims. Other Allowed Claims shall be satisfied as set forth in Article 5. Notwithstanding anything contained herein or otherwise, (i) Cash in the CCR Account shall be available, to the extent permitted by the Budget and/or the

Sale Support Agreement, only to pay Claims on account of chargebacks, channel management agreements, product returns and rebates constituting (a) Allowed Claims authorized to be paid by the Bankruptcy Court pursuant to the Orders approving the Prepetition Claims Motions or (b) Allowed Administrative Expense Claims, (ii) Cash in the Employee Account shall be available only to pay Claims on account of employee severance and benefits approved in the Orders approving the Prepetition Claims Motion and permitted by the Budget and/or Sale Support Agreement, (iii) Cash in the Lazard Account shall be available only to pay Allowed Professional Fees of Lazard, and (iv) Cash in the Designated Account shall be available only to pay Allowed Administrative Expense Claims, Allowed Other Secured Claims (to the extent not otherwise satisfied in accordance with the terms of this Plan), Allowed Priority Tax Claims and Allowed Other Priority Claims to the extent permitted by the Budget and/or Sale Support Agreement, other than any such Claims on account of (A) chargebacks, channel management agreements, product returns and rebates or (B) employee severance and benefits payable from the Employee Account.

7.8 Post-Effective Date Funding of Operations. From time to time as agreed between the First Lien Facility Agent and the Liquidating Trustee, the First Lien Facility Agent shall fund amounts into the First Lien Trust Administrative Fund to pay the Liquidating Trustee First Lien Fees incurred from the Effective Date through and including the date all First Lien Facility Claims are paid in full in Cash, in connection with (a) the collection of accounts receivable, (b) the pursuit of Causes of Action (including 549 Avoidance Actions but excluding for the avoidance of doubt Avoidance Actions), and (c) the objection to, and settlement of, Other Secured Claims (including Possessory Lienholder Claims), Administrative Expense Claims, Priority Tax Claims and Other Priority Claims from Sale Proceeds to which the First Lien

Facility Agent or First Lien Claimholders would otherwise be entitled. On the Effective Date, the Liquidating Trustee Fee Amount shall be funded into the Trust Administrative Fund by the First Lien Facility Agent from Sale Proceeds to which the First Lien Facility Agent or First Lien Claimholders would otherwise be entitled free and clear of all Liens and Claims of the First Lien Facility Agent and the First Lien Claimholders, including the First Lien Facility Secured Claim, any First Lien Facility Adequate Protection Claim and any First Lien Facility Deficiency Claim for the sole purpose of paying the Liquidating Trustee Fees. Any additional amount the Liquidating Trustee deems necessary to fund post-Effective Date operations other than fees and expenses incurred from the Effective Date through and including the date all First Lien Facility Claims are paid in full in Cash in connection with (a) the collection of accounts receivable, (b) the pursuit of Causes of Action (including 549 Avoidance Actions but excluding for the avoidance of doubt Avoidance Actions), and (c) the objection to, and settlement of, Other Secured Claims (including Possessory Lienholder Claims), Administrative Expense Claims, Priority Tax Claims and Other Priority Claims, shall be funded into the Trust Administrative Fund from the Other Assets.

Date, each of the Debtors will be dissolved for all purposes without the necessity for any other or further actions to be taken by or on behalf of the Debtors or payments to be made in connection therewith; *provided*, *however*, that pursuant to Section 1142(b) of the Bankruptcy Code, the Liquidating Trustee shall file each Debtor's final tax returns, and shall be authorized, but not directed, to file with the official public office for keeping corporate records in each Debtor's state of incorporation a certificate of dissolution or equivalent document. Such a certificate of dissolution may be executed by the Liquidating Trustee without need for any action or approval

by the shareholders or Board of Directors of any Debtor. From and after the Effective Date, the Debtors (i) for all purposes shall be deemed to have withdrawn their business operations from any state in which they were previously conducting, or are registered or licensed to conduct, their business operations, and shall not be required to file any document, pay any sum or take any other action in order to effectuate such withdrawal, (ii) shall be deemed to have cancelled pursuant to this Plan all Interests and all Intercompany Claims, and (iii) shall not be liable in any manner to any taxing authority for franchise, business, license or similar taxes accruing on or after the Effective Date.

Tases, when all Assets contributed to the Liquidating Trust have been liquidated and converted into Cash (other than those Assets abandoned by the Liquidating Trust), and such Cash has been distributed in accordance with the Liquidating Trust Agreement and this Plan, the Liquidating Trustee shall seek authority from the Bankruptcy Court to close the Debtors' Chapter 11 Cases in accordance with the Bankruptcy Code and the Bankruptcy Rules; provided, however, that, in light of the substantive consolidation of the Debtors' estates for distribution purposes, the Liquidating Trustee may, in the Liquidating Trustee's sole discretion, seek authority from the Bankruptcy Court to close all of the Debtors' Chapter 11 Cases except for the Graceway Chapter 11 Case immediately upon satisfaction of all Other Secured Claims.

7.11 Post-Effective Date Cause of Action Settlements. Subject to this Section 7.11, on and after the Effective Date, the Liquidating Trustee shall have the exclusive authority to settle Causes of Action owned by the Liquidating Trust. On and after the Effective Date and until such time as the First Lien Facility Claims are paid in full in Cash, in order to

settle Causes of Action owned by the Liquidating Trust, the Liquidating Trustee shall be required to obtain (i) the prior written consent of the Approving Majority First Lien Lenders or (ii) an Order of the Bankruptcy Court approving such settlement after notice and a hearing. After the First Lien Facility Claims are paid in full in Cash and until such time as the Second Lien Facility Claims are paid in full in Cash, in order to settle Causes of Action owned by the Liquidating Trust, the Liquidating Trustee shall be required to obtain (i) the prior written consent of the Second Lien Facility Agent or (ii) an Order of the Bankruptcy Court approving such settlement after notice and a hearing. After the Second Lien Facility Claims are paid in full in Cash, except as otherwise set forth in this Plan, the Confirmation Order or the Liquidating Trust Agreement, the Liquidating Trustee shall not be required to obtain approval of the Approving Majority First Lien Lenders, the Second Lien Facility Agent or the Bankruptcy Court to settle Causes of Action owned by the Liquidating Trust.

7.12 Post-Effective Date Reporting. Once each quarter, the Liquidating Trustee shall File with the Bankruptcy Court and serve on the United States Trustee an unaudited written report setting forth the amounts and timing of all distributions made and expenses paid by the Liquidating Trustee under the Plan in each applicable reporting period and the recipients thereof. In addition, the Liquidating Trustee shall file with the Bankruptcy Court and serve on the United States Trustee quarterly reports summarizing the Cash receipts of the Liquidating Trust for the immediately preceding three-month period. Each quarterly report shall also state the Liquidating Trust's Cash balances as of the beginning and ending of each such period. Quarterly reports shall be provided no later than the fifteenth (15th) day of each January, April, July and October until all Final Distributions under this Plan have been made.

- 7.13 Cancellation of Existing Securities and Agreements. On the Effective Date, any document, agreement, or instrument evidencing any Claim or Interest against or in the Debtors (including, but not limited to, the First Lien Credit Agreement and the Second Lien Credit Agreement) that has not been Assumed by the Purchaser shall be deemed cancelled without further act or action under any applicable agreement, law, regulation, Order, or rule and the Debtors shall have no further obligations under such documents, agreements or instruments; provided, however, that, although the Debtors shall have no further obligations or liability under Section 10.04 of the Credit Agreement, the Fee Letter (as defined in the First Lien Credit Agreement), the Intercreditor Agreement or Article XI of the Mezzanine Credit Agreement, as among parties other than the Debtors, Section 10.04 of the First Lien Credit Agreement, the Fee Letter, the Intercreditor Agreement and Article XI of the Mezzanine Credit Agreement shall remain in full force and effect as though the obligations of the Debtors continue to exist. Notwithstanding anything to the contrary set forth herein, the Liens on the Assets (including the Reserve Accounts and the Cash deposited therein and the Nycomed Litigation Proceeds, but excluding the Other Assets) securing the First Lien Facility Secured Claim, any First Lien Facility Adequate Protection Claim, the Second Lien Facility Secured Claim and any Second Lien Facility Adequate Protection Claim shall remain valid and perfected Liens thereon.
- 7.14 Dissolution of Creditors' Committee. On the Confirmation Date, the Creditors' Committee shall automatically be dissolved and the members of, and Professionals retained by, the Creditors' Committee shall have no further duties or obligations after the Confirmation Date.
- 7.15 Nycomed Litigation Proceeds. On the Effective Date, the Debtors shall assign and transfer all of their rights to and interests in Nycomed Litigation Proceeds to the First

Lien Facility Agent subject to any Liens thereon, which Liens shall remain valid, binding and perfected to the same extent as on the Petition Date on and after the Effective Date. Upon receipt of any Nycomed Litigation Proceeds, the First Lien Facility Agent will escrow an amount thereof sufficient to cover each Other Secured Claim asserted to be secured by Liens on the Nycomed Litigation Proceeds that are senior in priority to those held by the First Lien Facility Agent for the benefit of the First Lien Claimholders and the First Lien Facility Agent will distribute from such escrow funds sufficient to pay any such Other Secured Claim in full if and when any such Other Secured Claim is determined by an Order of the Bankruptcy Court to be an Allowed Other Secured Claim. In the event that the First Lien Facility Agent receives any Nycomed Litigation Proceeds, the First Lien Facility Agent shall provide written notice of the receipt thereof to the Second Lien Facility Agent. In the event that the First Lien Facility Claims are paid in full in Cash prior to the First Lien Facility Agent's receipt of any Nycomed Litigation Proceeds, the First Lien Facility Agent shall assign and transfer all of its rights to and interests in the Nycomed Litigation Proceeds to the Second Lien Facility Agent subject to any Liens thereon, which Liens shall remain valid and perfected to the same extent as on the Petition Date on and after the Effective Date. Upon receipt of any Nycomed Litigation Proceeds, the Second Lien Facility Agent will escrow an amount thereof sufficient to cover any Other Secured Claims asserted to be secured by Liens on the Nycomed Litigation Proceeds that are senior in priority to those held by the Second Lien Facility Agent for the benefit of the Second Lien Facility Lenders and the Second Lien Facility Agent will distribute from such escrow funds sufficient to pay any such Other Secured Claim in full if and when any such Other Secured Claim is determined by an Order of the Bankruptcy Court to be an Allowed Other Secured Claim.

ARTICLE 8 POST-CONFIRMATION LITIGATION

- 8.1 Transfer and Enforcement of Causes of Action. Pursuant to Section 1123(b)(3) of the Bankruptcy Code, except as otherwise provided in this Plan or the Confirmation Order, after transfer of the Assets (other than the Nycomed Litigation Proceeds) to the Liquidating Trust pursuant to Section 7.2.3 hereof, the Liquidating Trustee (and to the extent retained by the Liquidating Trust to perform such work, any other Person) will have the exclusive right and standing in any court of competent jurisdiction to enforce any and all Causes of Action against any Entity and rights of the Debtors that arose before or after the Petition Date, including, but not limited to, the rights and powers of a trustee and debtor-in-possession, against any Entity whatsoever, including, but not limited to, all Avoidance Actions and 549 Avoidance Actions, but excluding Released Claims.
- 8.2 Abandonment of Causes of Action and Other Assets of the Debtors. Subject to this Section 8.2, on and after the Effective Date, Liquidating Trustee shall have the exclusive authority to abandon Causes of Action or other Assets owned by the Liquidating Trust. On and after the Effective Date and until such time as the First Lien Facility Claims are paid in full in Cash, in order to abandon Causes of Action or other Assets owned by the Liquidating Trust, the Liquidating Trustee shall be required to obtain (i) the prior written consent of the Approving Majority First Lien Lenders or (ii) an Order of the Bankruptcy Court approving such abandonment after notice and a hearing. After the First Lien Facility Claims are paid in full in Cash and until such time as the Second Lien Facility Claims are paid in full in Cash, in order to abandon Causes of Action or other Assets owned by the Liquidating Trust, the Liquidating Trustee shall be required to obtain (i) the prior written consent of the Second Lien Facility Agent or (ii) an Order of the Bankruptcy Court approving such abandonment after notice and a hearing.

After the Second Lien Facility Claims are paid in full in Cash, except as otherwise set forth in this Plan, the Confirmation Order or the Liquidating Trust Agreement, the Liquidating Trustee shall not be required to obtain approval of the Approving Majority First Lien Lenders, the Second Lien Facility Agent or the Bankruptcy Court to abandon Causes of Action or other Assets owned by the Liquidating Trust.

ARTICLE 9 PROCEDURES FOR TREATMENT OF DISPUTED CLAIMS

- 9.1 Objections to Claims. Following the Effective Date, only the Liquidating Trustee shall be entitled to object to Claims. Any objections to Administrative Expense Claims shall be filed and served on the claimant no later than the later of (x) sixty (60) days after the Administrative Expense Claims Bar Date, and (y) such later date as may be fixed by the Bankruptcy Court, after notice and a hearing, which later date may be fixed before or after the date specified in clause (x) above. Any objections to any Claim other than an Administrative Expense Claim shall be Filed and served on the claimant no later than the later of (x) sixty (60) days after the date the Claim is Filed, (y) sixty (60) days after the Effective Date or such other date as may be ordered from time to time by the Bankruptcy Court and (z) such later date as may be fixed by the Bankruptcy Court, after notice and a hearing, which later date may be fixed before or after the date specified in clause (x) above.
- 9.2 Resolution of Disputed Claims. Subject to this Section 9.2, after the Effective Date, the Liquidating Trustee shall have the exclusive authority to compromise, settle, otherwise resolve, or withdraw any objections to Disputed Claims. On and after the Effective Date and until such time as the First Lien Facility Claims are paid in full in Cash, in order to compromise, settle, otherwise resolve, or withdraw any objections to Disputed Claims, the Liquidating Trustee shall be required to obtain (i) the prior written consent of the Approving

Majority First Lien Lenders or (ii) an Order of the Bankruptcy Court approving such compromise, settlement, other resolution, or withdrawal of any objections to Disputed Claims. After the First Lien Facility Claims are paid in full in Cash until such time as the Second Lien Facility Claims are paid in full in Cash, the Liquidating Trustee shall be required to obtain (i) the prior written consent of the Second Lien Facility Agent or (ii) an Order of the Bankruptcy Court approving such compromise, settlement, other resolution, or withdrawal of any objections to Disputed Claims. After the Second Lien Facility Claims are paid in full in Cash, except as otherwise set forth in this Plan, the Confirmation Order or the Liquidating Trust Agreement, the Liquidating Trustee shall not be required to obtain approval of the Approving Majority First Lien Lenders, the Second Lien Facility Agent or the Bankruptcy Court to compromise, settle, otherwise resolve, or withdraw any objections to Disputed Claims.

- 9.3 Amendments to Claims. After the Confirmation Date, a proof of Claim for any Claim other than an Administrative Expense Claim may not be amended without the authorization of the Bankruptcy Court. After the Administrative Expense Claims Bar Date, a proof of Claim for an Administrative Expense Claim may not be amended without the authorization of the Bankruptcy Court. Any such amendments to proofs of Claim or proofs of Claim for an Administrative Expense Claim Filed after the Confirmation Date or the Administrative Expense Claims Bar Date (as applicable) shall be deemed disallowed in full and expunged without any action by the Debtors or the Liquidating Trustee, unless the holder of the Claim or Administrative Expense Claim has obtained prior Bankruptcy Court authorization to File the amendment.
- 9.4 No Distributions Pending Allowance. Notwithstanding any other provision hereof, if any portion of a Claim or Administrative Expense Claim is Disputed, no

payment or distribution provided hereunder shall be required to be made on account of such Claim or Administrative Expense Claim unless and until such Disputed Claim or Disputed Administrative Expense Claim becomes an Allowed Claim in its entirety.

ARTICLE 10 DISTRIBUTIONS

- 10.1 No Duplicate Distributions. In accordance with Section 7.1 of the Plan and unless otherwise expressly provided herein, to the extent more than one Debtor is liable for any Claim, such Claim shall be considered a single Claim and entitled only to the payment provided therefor under the applicable provisions of the Plan.
- Allowed Claims shall be made: (i) at the addresses set forth in the proofs of Claim Filed by such holders; (ii) at the addresses set forth in any written notices of address change Filed with the Bankruptcy Court or delivered to the Liquidating Trustee after the date on which any related proof of Claim was Filed; or (iii) at the addresses reflected in the Schedules relating to the applicable Allowed Claim if no proof of Claim has been Filed and the Liquidating Trustee has not received a written notice of a change of address.
- Order, Cash payments to be made pursuant to the Plan shall be made by checks drawn on a domestic bank or by wire transfer from a domestic bank, at the option of the Debtors (with respect to Cash payments made on the Effective Date) and Liquidating Trustee (with respect to Cash payment made after the Effective Date).
- 10.4 Interest on Claims. Except as otherwise set forth herein or as required by applicable bankruptcy law, postpetition interest shall not accrue or be paid on Claims, and no holder of a Claim shall be entitled to interest accruing on or after the Petition Date on any Claim.

Interest shall not accrue or be paid upon any Disputed Claim in respect of the period from the Petition Date to the date a final distribution is made thereon if and after such Disputed Claim becomes an Allowed Claim. To the extent that any Allowed Claim entitled to a distribution under the Plan is composed of indebtedness and accrued but unpaid interest thereon, such distribution shall, to the extent permitted by applicable law, be allocated for federal income tax purposes to the principal amount of the Allowed Claim first and then, to the extent the consideration exceeds the principal amount of the Allowed Claim, to the portion of such Allowed Claim representing accrued but unpaid interest.

- than \$100.00 shall be required to be made on account of any Allowed Claim. Such undistributed amount may instead be made part of the Other Available Cash for use in accordance with this Plan. If the Other Available Cash for the Final Distribution is less than \$10,000, and the Liquidating Trustee, in his sole discretion, determines that it would cost more than \$5,000 to distribute such funds, the Liquidating Trustee may donate such funds to the charity of his choice.
- a specific Claim or Class of Claims, for the purpose of the provisions of this Article, the "Face Amount" of a Disputed Claim means the amount set forth on the proof of Claim unless the Disputed Claim has been estimated for distribution purposes, in which case the Face Amount shall be the Estimated Amount, or, in the alternative, if no proof of Claim has been Filed or deemed Filed, zero.
- 10.7 Undeliverable Distributions. If the distribution check to any holder of an Allowed Claim is not cashed within 90 days after issuance by the Liquidating Trustee, the Liquidating Trustee may give a stop payment order with respect to the check and no further

distributions shall be made to such holder on account of such Allowed Claim. Such Allowed Claim shall be discharged and the holder of such Allowed Claim shall be forever barred from asserting such Claim against the Liquidating Trustee, the Debtors, or the Liquidating Trust. In such cases, any Cash held for distribution on account of such Claim shall (i) become the property of the Liquidating Trust, and (ii) be distributed to other Creditors in accordance with the terms of this Plan.

10.8 Timing of Distributions.

10.8.1 On the Effective Date, or as soon thereafter as practicable in accordance with the Claims resolution process described herein, the Liquidating Trustee shall distribute to the holders of Allowed Administrative Expense Claims (other than Administrative Expense Claims that are Assumed Liabilities), Allowed Priority Tax Claims, Allowed Class 1 Claims and Allowed Class 4 Claims (to the extent not otherwise satisfied in accordance with the terms of this Plan) the distributions for each respective Class as set forth in the Plan.

10.8.2 On the Effective Date, after payment in full in Cash of all severance and benefits approved in the Orders approving the Prepetition Claims Motions, any Cash remaining in the Employee Account shall be First Lien Available Cash and shall be promptly distributed to the First Lien Facility Agent or, to the extent the First Lien Facility Claims have been paid in full in Cash, as otherwise set forth in the Plan. Promptly after the Estate Fund Amount is fully funded (or, if the Estate Fund Amount is fully funded prior to the Effective Date, on the Effective Date), any Canadian Distribution Amount remaining and any Cash in the 1.0% Holdback Account remaining shall be First Lien Available Cash and shall be promptly distributed to the First Lien Facility Agent or, to the extent the First Lien Facility Claims have been paid in full in Cash, as otherwise set forth in the Plan. Promptly after the

Allowed Professional Fees of Lazard are paid in full in Cash (or, if the Allowed Professional Fees of Lazard are paid in full in Cash prior to the Effective Date, on the Effective Date), any Cash remaining in the Lazard Account shall be First Lien Available Cash and shall be promptly distributed to the First Lien Facility Agent or, to the extent the First Lien Facility Claims have been paid in full in Cash, as otherwise set forth in the Plan. Promptly after the Administrative Expense Claims Bar Date, any amounts in the CCR Account in excess of the aggregate Face Amounts (or, if estimated for distribution purposes, the Estimated Amounts) of Claims Filed asserting liabilities related to chargebacks, channel management agreements, product returns and/or rebates shall be First Lien Available Cash and shall be promptly distributed to the First Lien Facility Agent or, to the extent the First Lien Facility Claims have been paid in full in Cash, as otherwise set forth in the Plan. In addition, upon the ultimate determination that any Disputed Claim to be funded from the CCR Account is not Allowed and payment of any Liquidating Trustee First Lien Fees associated with such determination, the portion of the CCR Account held on account of such Claim shall be First Lien Available Cash and shall be promptly distributed to the First Lien Facility Agent or, to the extent the First Lien Facility Claims have been paid in full in Cash, as otherwise set forth in the Plan. On the Effective Date, any amounts in the Designated Account in excess of the aggregate amount set forth in the Budget for all expenses to be paid from the Designated Account shall be First Lien Available Cash and shall be promptly distributed to the First Lien Facility Agent or, to the extent the First Lien Facility Claims have been paid in full in Cash, as otherwise set forth in the Plan. Promptly after the Administrative Expense Claims Bar Date, any amounts in the Designated Account in excess of the aggregate Face Amounts (or, if estimated for distribution purposes, the Estimated Amounts) of the Claims Filed that are to be funded from the Designated Account shall be First Lien Available Cash and shall be promptly distributed to the First Lien Facility Agent or, to the extent the First Lien Facility Claims have been paid in full in Cash, as otherwise set forth in the Plan. In addition, upon the ultimate determination that any Disputed Claim to be funded from the Designated Account is not Allowed and payment of any Liquidating Trustee First Lien Fees associated with such determination, the portion of the Designated Account held on account of such Claim shall be First Lien Available Cash and shall be promptly distributed to the First Lien Facility Agent or, to the extent the First Lien Facility Claims have been paid in full in Cash, as otherwise set forth in the Plan.

10.8.3 From time to time, but in no event less than once every three (3) months, the Liquidating Trustee shall distribute all Other Available Cash and any First Lien Available Cash not already distributed pursuant to Section 10.8.2 as set forth in this Plan.

10.8.4 Not later than five (5) Business Days after receipt thereof, after the liquidation into Cash of Causes of Action (other than those Causes of Action abandoned by the Liquidating Trustee in accordance with the terms hereof), the Liquidating Trustee shall distribute Net Causes of Action Proceeds to the holders of Allowed Claims in accordance with Sections 5.2, 5.3 and 5.4 of this Plan.

10.8.5 From and after the Effective Date, the Liquidating Trustee shall distribute on a bi-weekly basis all collections in respect of accounts receivable received during each such two-week period in accordance with Sections 5.2 and 5.3 of this Plan.

10.8.6 After (i) the distribution of the payments described in Section 10.8.1 hereof, (ii) the liquidation into Cash of all Assets owned by the Liquidating Trust (other than those Assets abandoned by the Liquidating Trustee in accordance with the terms hereof), (iii) the collection of other sums due or otherwise remitted or returned to the Estates, (iv) the

resolution of all Disputed Claims and (v) the payment in full in Cash of all Liquidating Trustee Fees and Liquidating Trustee First Lien Fees, the Liquidating Trustee shall distribute the remaining Assets owned by the Liquidating Trust in accordance with this Plan (the "Final Distribution"). The date of the Final Distribution shall be the "Final Distribution Date."

10.9 Disputed Claims Reserve. On the Effective Date, the Liquidating Trustee shall establish a reserve fund solely from the Other Assets other than the Liquidating Trustee Fee Amount (the "Disputed Claims Reserve") for the payment by the Liquidating Trustee of all Disputed Claims if and when Allowed; provided, however, that funds shall be added to the Disputed Claims Reserve for Administrative Expense Claims, Priority Tax Claims, Other Priority Claims and Other Secured Claims (to the extent not otherwise to be satisfied in accordance with the terms of this Plan) that are Disputed Claims only to the extent the amounts deposited in the applicable Reserve Account for the payment of such Claims are insufficient to pay the Face Amount (or, if estimated for distribution purposes, the Estimated Amount) of such Claims. With respect to any unliquidated Disputed Claim, the Disputed Claims Reserve shall be funded upon entry of and in accordance with an Order(s) of the Bankruptcy Court estimating the amount (the "Estimated Amount") of such unliquidated Disputed Claim. Upon ultimate determination that any Disputed Claim is not Allowed, the Liquidating Trustee shall immediately release the portion of the Disputed Claim Reserve or Reserve Account, as applicable, on account of such Claim and distribute such portion in accordance with the Plan. Each time a distribution of Other Available Cash is made to any Class of Claims, the Liquidating Trustee shall deposit into the Disputed Claims Reserve an amount equal to the distribution each holder of a Disputed Claim in such Class would have received were the Face Amount (or, if estimated for distribution purposes, the Estimated Amount) of its Disputed Claim in such Class an Allowed Claim solely from proceeds of Other Assets. Any Person whose Disputed Claim is so estimated shall be entitled to an Allowed Claim only up to an amount not to exceed the Estimated Amount even if such Person's Disputed Claim, as finally allowed, would have exceeded the maximum Estimated Amount thereof. At the time a Person's Disputed Claim becomes an Allowed Claim, such Person shall receive from the applicable Reserve Account and/or the Disputed Claims Reserve a distribution equal to the distributions such Person would have received on account of its Allowed Claim had it been an Allowed Claim at the time of such prior distributions, with any surplus Cash held in the Disputed Claims Reserve on account of such Disputed Claim becoming generally available for use by the Liquidating Trustee as Other Available Cash in accordance with the Plan and the Liquidating Trust Agreement and any surplus Cash held in any Reserve Account becoming available for distribution by the Liquidating Trustee as First Lien Available Cash in accordance with the Plan. For the avoidance of doubt, no payment shall be made from the Disputed Claims Reserve on account of the First Lien Facility Claims.

10.10 Compliance with Tax Requirements. In connection with the Plan and the distributions made in accordance therewith, to the extent applicable, the Liquidating Trustee shall comply with all tax withholding and reporting requirements imposed by any Governmental Unit, and all distributions pursuant to the Plan shall be subject to such withholding and reporting requirements. The Liquidating Trustee shall be authorized to take any and all actions that may be necessary or appropriate to comply with such withholding and reporting requirements.

ARTICLE 11 CONDITIONS PRECEDENT

- 11.1 Conditions to Confirmation. The following are each conditions to entry of the Confirmation Order:
- 11.1.1 The Bankruptcy Court shall have approved the Disclosure Statement with respect to this Plan in form and substance that is reasonably acceptable to the Debtors and the First Lien Facility Agent; and
- 11.1.2 The Confirmation Order shall be in form and substance reasonably satisfactory to the Debtors and the First Lien Facility Agent.
- 11.2 Conditions to the Effective Date. The Plan shall not become effective and the Effective Date shall not occur unless and until:
- 11.2.1 The Bankruptcy Court shall have entered the Confirmation Order in form and substance reasonably satisfactory to the Debtors and the First Lien Facility Agent;
- 11.2.2 The Confirmation Order shall not have been amended or modified without the consent of the Debtors and the First Lien Facility Agent;
- 11.2.3 All documents, instruments and agreements, in form and substance reasonably satisfactory to the Debtors and the First Lien Facility Agent provided for under this Plan or necessary to implement this Plan, including, without limitation, the Liquidating Trust Agreement, shall have been executed and delivered by the parties thereto, unless such execution or delivery has been waived by the parties benefited thereby;
- 11.2.4 The Debtors shall have reasonably determined that the Cash in the Reserve Accounts and the Other Available Cash is sufficient to pay in full in Cash on the Effective Date all estimated Allowed Other Secured Claims (to the extent not otherwise satisfied in accordance with the terms of this Plan), Allowed Administrative Expense Claims (other than

Administrative Expense Claims that are Assumed Liabilities), Allowed Other Priority Claims and Allowed Priority Tax Claims; and

11.2.5 The Confirmation Order shall have become a Final Order.

- 11.3 Termination of Plan for Failure To Become Effective. If the Effective Date shall not have occurred on or prior to the date that is forty-five (45) days after the Confirmation Date, then this Plan shall terminate and be of no further force or effect unless the provisions of this Section 11.3 are waived in writing by the Debtors and the First Lien Facility Agent.
- 11.4 Waiver of Conditions. The Debtors may waive any or all of the conditions set forth in Sections 11.1 and/or 11.2 (other than the conditions set forth in Sections 11.1.1, 11.2.1, and 11.2.4) of this Plan subject to the consent of the First Lien Facility Agent.
- 11.5 Notice of Effective Date. On the Effective Date, or as soon thereafter as is reasonably practicable, the Debtors shall File with the Bankruptcy Court a "Notice of Effective Date" in a form reasonably acceptable to the Debtors in their sole discretion, which notice shall constitute appropriate and adequate notice that this Plan has become effective; *provided*, *however*, that the Debtors shall have no obligation to notify any Person of such fact. The Plan shall be deemed to be effective as of 12:01 a.m., prevailing Eastern time, on the Effective Date specified in such filing. A courtesy copy of the Notice of Effective Date may be sent by United States mail, postage prepaid (or at the Debtors' option, by courier or facsimile) to those Persons who have Filed with the Bankruptcy Court requests for notices pursuant to Federal Bankruptcy Rule 2002 and the Master Service List established in the Chapter 11 Cases.

ARTICLE 12 EFFECT OF CONFIRMATION

- 12.1 Jurisdiction of Court. Pursuant to Sections 105(a) and 1142 of the Bankruptcy Code, and notwithstanding entry of the Confirmation Order and occurrence of the Effective Date, the Bankruptcy Court shall retain exclusive jurisdiction over all matters arising out of, and related to, the Debtors' Chapter 11 Cases and the Plan to the fullest extent permitted by law.
- Bankruptcy Code, on and after the Confirmation Date, the provisions of this Plan shall bind (i) any holder of a Claim against or Interest in the Debtors and its respective successors and assigns (including the Liquidating Trust), whether or not the Claim or Interest of such holder is Impaired under this Plan and whether or not such holder has accepted the Plan and (ii) the Liquidating Trustee.
- 12.3 Term of Pre-Confirmation Injunctions or Stays. Unless otherwise provided in a separate Order from the Bankruptcy Court, all injunctions or stays arising under or entered during the Debtors' Chapter 11 Cases in accordance with Sections 105 or 362 of the Bankruptcy Code, or otherwise, and in existence on the Confirmation Date, shall remain in full force and effect and apply to all Beneficiaries and creditors holding Claims against the Debtors, their Estates, the Debtors' Assets, the Liquidating Trustee and the Liquidating Trust until the Final Distribution Date or, if different, the date indicated in such applicable Order.
- 12.4 Injunction Against Interference with Plan. Upon the entry of the Confirmation Order, all holders of Claims and Interests and other parties in interest, along with their respective present or former Affiliates, employees, agents, officers, directors, principals or

advisors, shall be enjoined from taking any actions to interfere with the implementation or consummation of this Plan.

Exculpation. Except as otherwise specifically provided in this Plan, none of the Purchaser (in its capacity as a purchaser under the Asset Purchase Agreement), the Debtors, the Liquidating Trustee, the First Lien Facility Agent, the First Lien Claimholders in their capacity as such, the Second Lien Facility Agent, the Second Lien Facility Lenders in their capacity as such, the Creditors' Committee (solely with respect to its conduct as a committee and not with respect to the actions of its members as individual creditors), nor any of such parties' respective present members (with respect to members of the Creditors' Committee, solely with respect to each member's conduct in furtherance of its, his, or her duties as a member of the Creditors' Committee, and not with respect to the actions of such members as individual creditors), officers, directors, shareholders, employees, representatives, advisors, attorneys, financial advisors, investment bankers, agents or Affiliates, or any of such parties' successors and assigns, shall have or incur, and each such Person is hereby released from, any Claim, obligation, cause of action in any form whatsoever or liability to one another or to any holder of a Claim or an Interest, or any other party in interest, or any of their respective officers, directors, shareholders, members and/or enrollees, employees, representatives, advisors, attorneys, financial advisors, investment bankers, agents, related professionals or Affiliates, or any of their successors or assigns, for any act or omission in connection with, relating to, or arising out of, the Chapter 11 Cases, the negotiation and execution of this Plan, the Disclosure Statement, the Asset Purchase Agreement and the 363 Sale, the solicitation of votes for and the pursuit of Confirmation of this Plan, the consummation of this Plan, or the administration of this Plan, the Asset Purchase Agreement, the 363 Sale and the property to be distributed under this Plan, including all documents ancillary thereto, all decisions, actions, inactions and alleged negligence or misconduct relating thereto and all prepetition activities leading to the promulgation and Confirmation of this Plan except fraud, willful misconduct or gross negligence as determined by a Final Order. Nothing in this Section 12.5 shall: (i) be construed as a release of any entity's fraud, gross negligence or willful misconduct with respect to matters set forth in this Section 12.5 or (ii) limit or abrogate the obligations of the Debtors or the Purchaser and any of their respective affiliates to one another under the Asset Purchase Agreement. Any of the foregoing parties in all respects shall be entitled to reasonably rely upon the advice of counsel with respect to their duties and responsibilities (if any) under this Plan.

Notwithstanding any other provision of this Plan, neither any holder of a Claim or Interest, nor other party in interest, nor any of their respective officers, directors, shareholders, members and/or enrollees, employees, representatives, advisors, attorneys, financial advisors, investment bankers, related professionals, agents or Affiliates, and no successors or assigns of the foregoing, shall have any right of action against any Debtor or the Purchaser (in its capacity as a purchaser under the Asset Purchase Agreement), the Liquidating Trustee, the First Lien Facility Agent, the First Lien Claimholders (in their capacity as such), the Second Lien Facility Agent, the Second Lien Facility Lenders (in their capacity as such), the Creditors' Committee (solely with respect to its conduct as a committee and not with respect to the actions of its members as individual creditors), or any of such parties' respective present members (with respect to members of the Creditors' Committee, solely with respect to each member's conduct in furtherance of its, his, or her duties as a member of the Creditors' Committee, and not with respect to the actions of such members as individual creditors), officers, directors, shareholders, employees, representatives, advisors, attorneys, financial advisors,

investment bankers, related professionals, agents or Affiliates, or any of such parties' successors and assigns, for any act or omission in connection with, relating to, or arising out of, the Chapter 11 Cases, the negotiation and execution of this Plan, the Disclosure Statement, the Asset Purchase Agreement and the 363 Sale, the solicitation of votes for and the pursuit of Confirmation of this Plan, the consummation of this Plan, or the administration of this Plan, the Asset Purchase Agreement, the 363 Sale and the property to be distributed under this Plan, including all documents ancillary thereto, all decisions, actions, inactions and alleged negligence or misconduct relating thereto and all prepetition activities leading to the promulgation and Confirmation of this Plan except fraud, willful misconduct or gross negligence as determined by a Final Order.

12.6 Injunction. Except as otherwise specifically provided in the Plan or the Confirmation Order, all Entities or Persons who have held, hold or may hold Claims, rights, causes of action, liabilities or any equity interests based upon any act or omission, transaction or other activity of any kind or nature related to the Debtors, the Liquidating Trust or the Chapter 11 Cases that occurred prior to the Effective Date, other than as expressly provided in this Plan or the Confirmation Order, regardless of the filing, lack of filing, allowance or disallowance of such a Claim or Interest and regardless of whether such Entity has voted to accept the Plan, and any successors, assigns or representatives of such Entities shall be precluded and permanently enjoined on and after the Effective Date from (a) the enforcement, attachment, collection or recovery by any manner or means of any judgment, award, decree or order with respect to any Claim, Interest or any other right or claim against the Debtors, the Liquidating Trustee, the Liquidating Trust or any Assets of the Debtors or the Liquidating Trust which such Entities possessed or may possess prior to the Effective Date, (b) the creation,

perfection or enforcement of any encumbrance of any kind with respect to any Claim, Interest or any other right or claim against the Debtors, the Liquidating Trustee, the Liquidating Trust or any assets of the Debtors or the Liquidating Trust which they possessed or may possess prior to the Effective Date, and (c) the assertion of any Claims that are released hereby.

Effective Date, each of the Debtors hereby (i) remises, acquits, waives, releases and forever discharges each of the Debtor Releasees from, and (ii) covenants and agrees never to institute or cause to be instituted any suit or other form of action or proceeding of any kind or nature whatsoever against any of the Debtor Releasees based upon, any claims, demands, indebtedness, agreements, promises, Causes of Action, obligations, damages or liabilities of any nature whatsoever (other than rights to enforce obligations of the Debtor Releasees under any Order of the Bankruptcy Court, the Plan and all contracts, instruments, releases and other agreements delivered in connection therewith), in law or in equity, whether or not known, suspected or claimed, that the Debtors or the Estates ever had, claimed to have, has, or may have or claim to have against the Debtor Releasees, or any of them, by reason of any matter, cause, thing, act or omission of the Debtor Releasees, or any of them, in each case related to the Debtors, the Chapter 11 Cases, the 363 Sale or the Plan, except fraud, willful misconduct or gross negligence as determined by a Final Order.

12.8 Releases by Holders of Claims and Interests. As of the Effective Date, to the fullest extent permitted under applicable law, in consideration for the obligations under the Plan and the Cash, securities, contracts, instruments, releases and other agreements or documents to be delivered in connection with the Plan, and the benefits provided by the Creditor Releasees in the Plan and in the Chapter 11 Cases, each present and former holder of a Claim

or Interest will be deemed to release forever, waive and discharge any and all Claims, obligations, suits, judgments, damages, demands, debts, rights, causes of action and liabilities (other than the rights to enforce the Debtors' obligations under any Order of the Bankruptcy Court, the Asset Purchase Agreement, the Plan and the securities, contracts, instruments, releases and other agreements and documents delivered thereunder), whether liquidated or unliquidated, fixed or contingent, matured or unmatured, known or unknown, foreseen or unforeseen, then existing or thereafter arising, in law, equity or otherwise that are based in whole or in part on any act, omission, transaction, event or other occurrence taking place on or prior to the Effective Date in any way relating to the Debtors, the Chapter 11 Cases, the 363 Sale, or the Plan against any Creditor Releasee, except fraud, willful misconduct or gross negligence as determined by a Final Order.

Asset Purchase Agreement, following the Effective Date, none of the Debtors, the Liquidating Trust, the Liquidating Trustee, the Creditors' Committee, the First Lien Facility Agent, the First Lien Claimholders (in their capacity as such), the Second Lien Facility Agent, the Second Lien Facility Lenders (in their capacity as such), or any of their respective members, officers, directors, employees, advisors, attorneys, professionals, agents or Affiliates shall have or incur any liability to any holder of a Claim or Interest for any act or omission in connection with, related to, or arising out of, the Chapter 11 Cases, the 363 Sale, the negotiation and pursuit of Confirmation of the Plan, the consummation of the Plan or any contract, instrument, release or other agreement or document created in connection with the 363 Sale, this Plan, or the administration of the Plan or the property to be distributed under the Plan, except for fraud, gross negligence or willful misconduct as determined by a Final Order.

ARTICLE 13 RETENTION OF JURISDICTION

- 13.1 Scope of Bankruptcy Court Jurisdiction. Notwithstanding the entry of the Confirmation Order and the occurrence of the Effective Date, the Bankruptcy Court may retain jurisdiction over the Debtors' Chapter 11 Cases after the Effective Date to the fullest extent legally permissible, including jurisdiction to, among other things:
- (a) Allow, disallow, determine, liquidate, classify, estimate or establish the priority or secured or unsecured status of any Claim or Interest, including the resolution of (x) any request for payment of any Administrative Expense Claim, (y) any Disputed Claims or Interests and (z) any and all objections to the allowance or priority of any Claim or Interest;
- (b) To the extent not inconsistent with the Bankruptcy Code, hear and determine any and all Causes of Action against any Person and rights of the Debtors that arose before or after the Petition Date, including, but not limited to, the rights and powers of a trustee and debtor-in-possession, against any Person whatsoever, including, but not limited to, all avoidance powers granted to the Debtors under the Bankruptcy Code and all Causes of Action and remedies granted pursuant to Sections 502, 506, 510, 541, 542, 543, 544, 545, 547 through 551 and 553 of the Bankruptcy Code;
- (c) Grant or deny any applications for allowance of compensation for Professionals authorized pursuant to the Bankruptcy Code or the Plan, for periods ending on or before the Effective Date;
- (d) Resolve any matters relating to the assumption, assumption and assignment or rejection of any executory contract or unexpired lease to which any Debtor is a party or with respect to which any of the Debtors may be liable, including, without limitation, the

determination of whether such contract is executory for the purposes of Section 365 of the Bankruptcy Code, and hear, determine and, if necessary, liquidate any Claims arising therefrom;

- (e) Enter Orders approving the Liquidating Trustee's post-Confirmation sale or other disposition of Assets of the Liquidating Trust;
- (f) Ensure that distributions to holders of Allowed Claims are accomplished pursuant to the provisions of the Plan and the Liquidating Trust Agreement;
- (g) Decide or resolve any motions, adversary proceedings, contested or litigated matters and any other matters and grant or deny any applications involving any Debtor or the Liquidating Trust that may be pending in the Debtors' Chapter 11 Cases on the Effective Date;
- (h) Hear and determine matters concerning state, local or federal taxes in accordance with Sections 346, 505 or 1146 of the Bankruptcy Code;
- (i) Enter such Orders as may be necessary or appropriate to implement or consummate the provisions of the Liquidating Trust Agreement, the Plan and the Confirmation Order;
- (j) Hear and determine any matters concerning the enforcement of the provisions of Article 12 of this Plan and any other exculpations, limitations of liability or injunctions contemplated by this Plan;
- (k) Resolve any cases, controversies, suits or disputes that may arise in connection with the consummation, interpretation or enforcement of the Liquidating Trust Agreement, the Plan or the Confirmation Order;
- (I) Permit the Debtors, to the extent authorized pursuant to Section 1127 of the Bankruptcy Code, to modify the Plan or any agreement or document created in connection

with the Plan, or remedy any defect or omission or reconcile any inconsistency in the Plan or any agreement or document created in connection with the Plan;

- (m) Issue injunctions, enter and implement other Orders or take such other actions as may be necessary or appropriate to restrain interference by any Entity with consummation, implementation or enforcement of the Liquidating Trust Agreement, the Plan or the Confirmation Order;
- (n) Enforce any injunctions entered in connection with or relating to the Plan or the Confirmation Order;
- (o) Enter and enforce such Orders as are necessary or appropriate if the Confirmation Order is for any reason modified, stayed, reversed, revoked or vacated, or distributions pursuant to the Plan are enjoined or stayed;
- (p) Determine any other matters that may arise in connection with or relating to the Plan, the 363 Sale, the Asset Purchase Agreement or the Confirmation Order;
 - (q) Enter any Orders in aid of prior Orders of the Bankruptcy Court; and
 - (r) Enter a final decree closing the Debtors' Chapter 11 Cases.

ARTICLE 14 ACCEPTANCE OR REJECTION OF THE PLAN

14.1 Persons Entitled to Vote. Class 1 and Class 4 are not Impaired, and holders of Class 1 Claims and Class 4 Claims are deemed pursuant to Section 1126(f) of the Bankruptcy Code to have accepted the Plan. Votes from holders of Class 1 Claims and Class 4 Claims will not be solicited. Class 2, Class 3, and Class 5 are Impaired and are expected to receive a distribution under the Plan. Votes from holders of the Class 2 Claims, the Class 3 Claims, and the Class 5 Claims will be solicited. Class 6 Claims, Class 7 Interests, and Class 8 Claims and Interests are Impaired, are not expected to receive any distributions pursuant to the

Plan, and are therefore deemed, pursuant to Section 1126(g) of the Bankruptcy Code, to have rejected the Plan. Votes from holders of Class 6 Claims, Class 7 Interests and Class 8 Claims and Interests will not be solicited.

- 14.2 Acceptance by Impaired Classes. An Impaired Class of Claims shall have accepted the Plan if (i) the holders (other than any holder designated under Section 1126(e) of the Bankruptcy Code) of at least two-thirds in amount of the Allowed Claims actually voting in such Class have voted to accept the Plan and (ii) the holders (other than any holder designated under Section 1126(e) of the Bankruptcy Code) of more than one-half in number of the Allowed Claims actually voting in such Class have voted to accept the Plan.
- 14.3 Request for Non-Consensual Confirmation. Class 6, Class 7, and Class 8 are not expected to receive any distribution on account of their Claims and Interests and are therefore deemed to have rejected the Plan. The Debtors therefore request that the Court confirm the Plan under the cramdown provisions of Section 1129(b) of the Bankruptcy Code with respect to Class 6, Class 7, and Class 8 and any Class that does not vote in favor of the Plan.

ARTICLE 15 MISCELLANEOUS PROVISIONS

- 15.1 Effectuating Documents and Further Transactions. Each of the Debtors and the Liquidating Trustee is authorized to execute, deliver, file, or record such contracts, instruments, releases, indentures, and other agreements or documents and take such actions as may be necessary or appropriate to effectuate and further evidence the terms and conditions of the Plan and any Securities issued pursuant to this Plan.
- 15.2 Corporate Action. On the Effective Date, all matters provided for under this Plan that would otherwise require approval of the stockholders, directors, members, managers or partners of one or more of the Debtors shall be deemed to have occurred and shall

be in effect from and after the Effective Date pursuant to Section 303 of the Delaware General Corporation Law or other applicable law of the states in which the Debtors are organized, without any requirement of further action by the stockholders, directors, members, managers, or partners of the Debtors.

- 15.3 Payment of Statutory Fees. After the Effective Date, the Liquidating Trustee shall pay all fees payable pursuant to section 1930 of chapter 123 of title 28 of the United States Code through the entry of a final decree closing the Debtors' cases.
- 15.4 Modification of the Plan. Subject to the restrictions on Plan modifications set forth in Section 1127 of the Bankruptcy Code, the Debtors reserve the right, after consultation with the First Lien Facility Agent, the Second Lien Facility Agent and the Creditors' Committee, to alter, amend or modify the Plan before its substantial consummation.
- withdraw the Plan prior to the Confirmation Date. If the Debtors revoke or withdraw the Plan, or if Confirmation does not occur or if the Plan does not become effective, then the Plan shall be null and void, and nothing contained in the Plan or Disclosure Statement shall: (a) constitute a waiver or release of any Claims by or against, or any Interests in, the Debtors; (b) constitute an admission of any fact or legal conclusion by the Debtors or any other Entity; or (c) prejudice in any manner the rights of the Debtors in any further proceedings involving the Debtors.
- 15.6 Governing Law. Unless a rule of law or procedure is supplied by (i) federal law (including the Bankruptcy Code and Bankruptcy Rules), or (ii) an express choice of law provision in any agreement, contract, instrument or document provided for, or executed in connection with, the Plan, the rights and obligations arising under the Plan and any agreements, contracts, documents and instruments executed in connection with the Plan shall be governed by,

and construed and enforced in accordance with, the laws of the State of New York without giving effect to the principles of conflict of laws thereof.

- 15.7 No Admissions. If Confirmation or the Effective Date does not occur, nothing contained in the Plan or Disclosure Statement shall be deemed as an admission by the Debtors with respect to any matter set forth herein or therein including, without limitation, liability on any Claim or the propriety of any Claims classification.
- provision of the Plan that does not govern the treatment of Claims or Interests is held by the Bankruptcy Court to be invalid, void or unenforceable, at the request of the Debtors the Bankruptcy Court shall have the power to alter and interpret such term or provision to make it valid or enforceable to the maximum extent practicable, consistent with the original purpose of the term or provision held to be invalid, void or unenforceable, and such term or provision shall then be applicable as altered or interpreted. Notwithstanding any such holding, alteration or interpretation, the remainder of the terms and provisions of the Plan shall remain in full force and effect and shall in no way be affected, Impaired or invalidated by such holding, alteration or interpretation. The Confirmation Order shall constitute a judicial determination and shall provide that each term and provision of the Plan, as it may have been altered or interpreted in accordance with the foregoing, is valid and enforceable pursuant to its terms.
- 15.9 Successors and Assigns. The rights, benefits and obligations of any Entity named or referred to in the Plan shall be binding on, and shall inure to the benefit of, any heir, executor, administrator, successor or assign of such Entity.
- 15.10 Exemption from Certain Transfer Taxes. Pursuant to Section 1146(a) of the Bankruptcy Code, the issuance, transfer or exchange of any Security or the making or

delivery of any instrument of transfer under this Plan may not be taxed under any law imposing a stamp tax, use tax, sales tax or similar tax. Any sale of any Asset of the Debtors or the Liquidating Trust occurring after or upon the Effective Date shall be deemed to be in furtherance of this Plan.

- Date, the Liquidating Trustee, may, but shall not be required to, set off against any Claim, and the payments or other distributions to be made pursuant to this Plan in respect of such Claim, Claims of any nature whatsoever that the Debtors or, if after the Effective Date, the Liquidating Trustee, may have against the holder of such Claims; but neither the failure to do so nor the allowance of any Claim hereunder shall constitute a waiver or release by the Debtors or the Liquidating Trustee of any such Claim that the Debtors or, if after the Effective Date, the Liquidating Trustee, may have against such holder.
- 15.12 Defenses with Respect to Unimpaired Claims. Except as otherwise provided in this Plan, nothing shall affect the rights and legal and equitable defenses of the Debtors or, if after the Effective Date, the Liquidating Trustee, with respect to any Unimpaired Claim, including all rights in respect of legal and equitable defenses to setoffs or recoupments against Unimpaired Claims.
- 15.13 No Injunctive Relief. Except as otherwise provided in the Plan or Confirmation Order, no Claim or Interest shall under any circumstances be entitled to specific performance or other injunctive, equitable, or other prospective relief.
- 15.14 Saturday, Sunday or Legal Holiday. If any payment or act under the Plan is required to be made or performed on a date that is not a Business Day, then the making of

such payment or the performance of such act may be completed on the next succeeding Business

Day, but shall be deemed to have been completed as of the required date.

15.15 Entire Agreement. This Plan (together with the Liquidating Trust Agreement) sets forth the entire agreement and undertaking relating to the subject matter hereof and supersedes all prior discussions and documents. The Debtors' Estates shall not be bound by any terms, conditions, definitions, warranties, understandings, or representations with respect to the subject matter hereof, other than as expressly provided for herein.

15.16 Dissolution of Creditors' Committee. The Creditors' Committee shall be dissolved on the Effective Date, without need for a further Order of the Bankruptcy Court.

15.17 Notices. Any notice required or permitted to be provided under this Plan shall be in writing and served by either (a) certified mail, return receipt requested, postage prepaid, (b) hand delivery, or (c) reputable overnight delivery service, freight prepaid, to be addressed as follows:

Debtors	Counsel to Debtors
Graceway Pharmaceuticals, LLC 340 Martin Luther King Jr. Blvd. Suite 400 Bristol, TN 37620 Fax: 422-274-5520 Attn: John Bellamy (john.bellamy@gracewaypharma.com)	Latham & Watkins LLP 233 South Wacker Drive Chicago, IL 60606 Fax: 312-993-9767 Attn: Josef S. Athanas, Esq. and Matthew L. Warren, Esq. (josef.athanas@lw.com and matthew.warren@lw.com)
	Young Conaway Stargatt & Taylor, LLP Rodney Square 1000 North King Street Wilmington, DE 19801 Fax: 302-576-3472 Attn: Michael R. Nestor, Esq. and Kara Hammond Coyle, Esq. (mnestor@ycst.com and kcoyle@ycst.com)

Special Bankruptcy and Restructuring Counsel to the Agent for the First Lien Facility Lenders	United States Trustee
Wachtell, Lipton, Rosen & Katz 51 West 52 nd Street New York, NY 10019 Fax: 212-403-2158 Attn: Scott K. Charles, Esq. and Michael S. Benn, Esq. (SKCharles@wlrk.com and MSBenn@wlrk.com) DLA Piper LLP 919 North Market Street, 15 th Floor Suite 1500 Wilmington, DE 19801 Fax: 302-778-7913 Attn: Stuart M. Brown, Esq. (stuart.brown@dlapiper.com)	Office of the United States Trustee for the District of Delaware 844 King Street J. Caleb Boggs Federal Building Room 2207, Lockbox 35 Wilmington, DE 19801 Fax: 302-573-6497 Attn: Juliet Sarkessian, Esq. (Juliet.M.Sarkessian@usdoj.gov)
Financing Counsel to the Agent for the First Lien Facility Lenders	Counsel to the Creditors' Committee
Morgan Lewis 225 Franklin Street, 16 th Floor Boston, MA 02110 Fax: 617-341-7701 Attn: Sula Fiszman, Esq. (sfiszman@morganlewis.com)	Lowenstein Sandler PC 65 Livingston Avenue Roseland, NJ 07068 Fax: 973-597-2347 Attn: S. Jason Teele, Esq. (steele@lowenstein.com) Elliott Greenleaf 1105 Market Street, Suite 1700 Wilmington, DE 19801 Fax: 302-384-9399 Attn: Rafael X. Zahralddin-Aravena (rxza@elliottgreenleaf.com)
Counsel to the Agent for the Second Lien Facility Lenders	Agent for the Mezzanine Lenders
Sidley Austin LLP One South Dearborn, Chicago, IL 60603 Fax: 213-896-6600 Attn: Larry Nyhan, Esq. (lnyhan@sidley.com) Jeffrey E. Bjork, Esq. (jbjork@sidley.com)	Goldman Sachs Credit Partners L.P. 30 Hudson Street, 5 th Floor Jersey City, NJ 07302 Fax: 212-357-4597 Attn: Michelle Latzoni (gsd.link@gs.com)
Receiver for Graceway Canada Company	Counsel to the Receiver for Graceway Canada Company
Duff & Phelps Canada Restructuring Inc. 200 King Street West, Suite 1002 Toronto, ON M5H 3T4 Fax: 647-497-9490 Attn: Robert Kofman and David Sieradzki (bobby.kofman@duffandphelps.com and david.sieradzki@duffandphelps.com)	Davies Ward Phillips & Vineberg LLP 1 First Canadian Place, Suite 4400 P.O. Box 63 Toronto, ON M5X 1B1 Fax: 416-863-0871 Attn: Jay Swartz (jswartz@dwpv.com)

Liquidating Trustee	Counsel to Graceway Canada Company
Kip Horton RPA Advisors, LLC 45 Eisenhower Drive Paramus, NJ 07652 Fax: 201-527-6620 Attn: Kip Horton (khorton@rpaadvisors.com)	Goodmans LLP Bay Adelaide Centre 333 Bay Street, Suite 3400 Toronto, ON M5H 2S7 Fax: 416-979-1234 Attn: Fred Myers, Joe Latham, and Caroline Descours (fmyers@goodmans.ca, jlatham@goodmans.ca, and cdescours@goodmans.ca)

Dated:	February 28, 2012
Gracev	vay Pharma Holding Corp.
Ву:	Name: John A.A. Bellomy Title: EVP
Gracev	vay Pharmaceuticals, LLC
Ву:	Name: John A. A. Ballomez Title: EVP
Cheste	r Valley Pharmaceuticals, LLC
Ву:	Name: John A.A.Bellamy Title: EVP
Gracev	way Canada Holdings, Inc.
Ву:	Name: John A. A. Ballaning Title: EVP
Graceway International, Inc.	
Ву:	Name: John A. A. Bellany Title: EVP

Graceway Holdings, LLC

Chester Valley Holdings, LLC

By:

By:

Name: John 17. A. Bellomy Title: EVP

Name: John A. A. Bellomy Title: EVP

Exhibit 1

List of Debtors

The Debtors, along with the last four digits of each Debtor's federal tax identification number and each Debtor's Chapter 11 Case Number, include:

- 1. Graceway Pharma Holding Corp. (9175): Case No. 11-13037 (PJW)
- 2. Graceway Holdings, LLC (2502): Case No. 11-13038 (PJW)
- 3. Graceway Pharmaceuticals, LLC (5385): Case No. 11-13036 (PJW)
- 4. Chester Valley Holdings, LLC (9457): Case No. 11-13039 (PJW)
- 5. Chester Valley Pharmaceuticals, LLC (3713): Case No. 11-13041 (PJW)
- 6. Graceway Canada Holdings, Inc. (6663): Case No. 11-13042 (PJW)
- 7. Graceway International, Inc. (2399): Case No. 11-13043 (PJW)

Exhibit 2

Liquidating Trust Agreement

LIQUIDATING TRUST AGREEMENT

for the

Graceway Liquidating Trust

TABLE OF CONTENTS

Section I Defi	nitions	5
1.1.	Terms Defined Above.	5
1.2.	Terms Defined in the Plan.	5
Section II Au	thority of and Certain Directions to Liquidating Trustee: Declaration of Trust	6
2.1.	Creation of Liquidating Trust.	6
2.2.	Purpose of Liquidating Trust.	6
2.3.	Title to Assets of the Debtors	6
2.4.	Segregated Funds.	7
2.5.	Tax Treatment of Transfer of Trust Assets to the Liquidating Trust	8
2.6.	Transfer Free and Clear of Claims.	8
2.7.	Transfer of Trust Assets.	8
2.8.	Property in the Liquidating Trust.	9
2.9.	Valuation of Trust Assets	9
2.10.	Continuation of the Automatic Stay.	9
Section III Be	eneficial Interests	9
3.1.	No Transfer or Exchange.	9
3.2.	No Certification.	. 10
3.3.	Absolute Owners	. 10
3.4.	Means of Payment	. 10
3.5.	Amount of Payment.	. 10
3.6.	Acceptance of Conveyance.	. 10
3.7.	Title.	. 11
Section IV A	dministration of Trust Estate	.11
4.1.	Trust Administrative Fund.	.11
4.2.	First Lien Trust Administrative Fund.	. 12
4.3.	Disputed Claims Reserve.	. 12
4.4.	Administrative Powers of the Liquidating Trustee.	.13
4.5.	Objections to Claims.	. 15
4.6.	Limitations on Liquidating Trustee; Investments; Debt.	. 16
4.7.	Limitations on Liquidating Trustee - Causes of Action and Abandonment	. 16

	4.8.	Transferee Liabilities	17
	4.9.	Administration of Trust	18
	4.10.	Payment of Expenses and Other Liabilities	18
	4.11.	Payment of United States Trustee's Fees	19
	4.12.	Liquidating Trustee Fees and Liquidating Trustee First Lien Fees	19
	4.13.	Fiscal Year.	19
	4.14.	Reports	19
	4.15.	Liquidating Trust Committee	20
	4.16.	Removal of the Liquidating Trustee	20
Sectio	n V Pay	ments and Distributions.	21
	5.1.	Distributions to Beneficiaries	21
	5.2.	Frequency and Amounts of Cash Payments	22
	5.3.	Establishment of the Claim Accounts	22
	5.4.	Distributions from Trust Estate	23
	5.5.	Fractional Distributions	23
	5.6.	Final Distribution	23
	5.7.	Termination Date	23
Section	n VI Ot	her Duties of the Trustee	24
	6.1.	Management of Trust Estate	24
	6.2.	Tax and Related Matters	24
	6.3.	No Implied Duties	24
Section	on VII C	Concerning the Liquidating Trustee.	25
	7.1.	Acceptance by Liquidating Trustee	25
	7.2.	Discretionary Submission of Questions	25
	7.3.	Liability of the Liquidating Trustee	25
	7.4.	Reliance by Liquidating Trustee	26
	7.5.	Reliance on Liquidating Trustee	26
	7.6.	Indemnification	27
	7.7.	Compensation of Liquidating Trustee	28
	7.8.	Resignation and Removal	28
Section	on VIII S	Supplements and Amendments to this Liquidating Trust Agreement	29
	8.1.	Supplements and Amendments	29

8.2.	Declining to Execute Documents	30
8.3.	Notice of Form of Supplements and Amendments Requiring Vote or Consent	30
8.4.	Notice and Effect of Executed Amendment	30
Section IX Mi	scellaneous.	30
9.1.	Title to Trust Estate	30
9.2.	Sales of Trust Assets	30
9.3.	Notices	30
9.4.	Severability	31
9.5.	Counterparts	31
9.6.	Binding Agreement	31
9.7.	No Personal Liability of Beneficiaries	31
9.8.	Headings	32
9.9.	Construction	32
9.10.	Governing Law	32
9.11.	Construction with the Plan	32
9.12.	Subject to Bankruptcy Court's Jurisdiction	32
9.13.	Intention of the Parties	32

LIQUIDATING TRUST AGREEMENT

This Liquidating Trust Agreement (the "Liquidating Trust Agreement"), dated as of [], 2012, by and among Graceway Pharmaceuticals, LLC ("Graceway") and each of the other debtors listed on the signature pages hereto (collectively with Graceway, the "Debtors") and Kip Horton, as Liquidating Trustee ("Liquidating Trustee"), is made and executed in connection with the First Amended Joint Plan of Liquidation of Graceway Pharmaceuticals, LLC, et al. under chapter 11 of title 11 of the United States Code, 11 U.S.C. §§ 101-1532 (the "Bankruptcy Code"), dated February 28, 2012, (the "Plan"), filed by the Debtors in the United States Bankruptcy Court for the District of Delaware (the "Bankruptcy Court"), which Plan was confirmed by Order of the Bankruptcy Court dated [], 2012. The Plan provides for the establishment of the liquidating trust evidenced hereby (which liquidating
trust shall formally be known as the "Graceway Liquidating Trust") to liquidate certain assets and property of the Debtors in accordance with the terms and conditions of the Plan and to resolve and realize upon certain of the Debtors' rights, claims and causes of action through enforcement by the Liquidating Trustee.
RECITALS
WHEREAS, on September 29, 2011 (the "Petition Date"), the Debtors each filed a voluntary petition for relief under chapter 11 of the Bankruptcy Code; and,
WHEREAS, the Debtors, thereafter, continued in the operation of their businesses and the management of their properties as debtors-in-possession pursuant to Sections 1107(a) and 1108 of the Bankruptcy Code; and,
WHEREAS, on [], 2012, the Debtors filed their Plan with the Bankruptcy Court; and,
WHEREAS, the Bankruptcy Court entered its Order confirming the Plan (the "Confirmation Order") on [], 2012; and,
WHEREAS, the Plan provides for, among other things, (a) liquidation of the Trust Assets, and (b) distributions from the Reserve Accounts and, to the extent necessary, the

WHEREAS, the Plan provides for, among other things, (a) liquidation of the Trust Assets, and (b) distributions from the Reserve Accounts and, to the extent necessary, the Other Available Cash, to the holders of Allowed Administrative Expense Claims, Allowed Priority Tax Claims, Allowed Other Priority Claims and Allowed Other Secured Claims to pay such Allowed Claims in full in accordance with the terms of the Plan, (c) distribution of the First Lien Available Cash to the holder of the Allowed First Lien Facility Claims as provided herein and in the Plan, and (d) distribution of the Other Available Cash to the holder of any Allowed Second Lien Facility Claims and to the holders of Allowed General Unsecured Claims as provided herein and in the Plan; and,

WHEREAS, the Plan provides for the creation of a liquidating trust to hold the Trust Assets in trust for the benefit of all Beneficiaries pursuant to the terms of this Liquidating Trust Agreement and the Plan; and,

WHEREAS, this Liquidating Trust Agreement is executed to establish the Liquidating Trust (as defined in Section 2 hereof) and to facilitate implementation of the Plan; and,

WHEREAS, the primary purpose of the Liquidating Trust is to liquidate the Trust Assets for the benefit of the Beneficiaries in accordance with Treasury Regulation Section 301.7701-4(d) and the Liquidating Trust will not be operated with the objective of continuing or engaging in the conduct of a trade or business, except to the extent reasonably necessary to, and consistent with, the liquidating purpose of the Liquidating Trust; and,

WHEREAS, except with respect to the Disputed Claims Reserve (defined below), the Liquidating Trust is intended to qualify as one or more liquidating trusts that are treated as "grantor trusts" for federal income tax purposes and the Liquidating Trustee shall operate and maintain the Liquidating Trust in compliance with Internal Revenue Service Revenue Procedure 94-45, 1994-2 C.B. 684, and Treasury Regulation Sections 1.671-4(a) and 301.7701-4(d) and all subsequent guidelines regarding liquidating trusts issued by the Internal Revenue Service; and,

WHEREAS, as contemplated by the Plan, the Beneficiaries hereby exchange their Allowed Claims under the Plan for Beneficial Interests (as hereinafter defined) in the Liquidating Trust;

AGREEMENTS

NOW THEREFORE, for and in consideration of the premises, and the mutual promises and agreements contained herein and in the Plan, the receipt and sufficiency of which are hereby expressly acknowledged, the Debtors and Liquidating Trustee hereby agree as follows:

SECTION I DEFINITIONS

1.1. Terms Defined Above.

As used in this Liquidating Trust Agreement, each of the terms "Liquidating Trust Agreement," "Graceway," "Debtors," "Liquidating Trustee," "Bankruptcy Code," "Plan," "Bankruptcy Court," "Petition Date," and "Confirmation Order" shall have the meanings hereinabove set forth.

1.2. Terms Defined in the Plan.

Capitalized terms used in this Liquidating Trust Agreement without definition shall have the meanings assigned to them in the Plan. Terms defined in the Bankruptcy Code and not otherwise specifically defined in the Plan or herein shall, when used herein, have the meanings attributed to them in the Bankruptcy Code.

SECTION II AUTHORITY OF AND CERTAIN DIRECTIONS TO LIQUIDATING TRUSTEE: DECLARATION OF TRUST

2.1. Creation of Liquidating Trust.

Pursuant to Section 7.2 of the Plan and the Confirmation Order, and effective as of the Effective Date of the Plan, the Beneficiaries and the Debtors hereby create the Liquidating Trust, to be formally be known as the Graceway Liquidating Trust, for the benefit of the Beneficiaries. Pursuant to the terms of the Plan, the Debtors execute this Liquidating Trust Agreement and, subject to the provisions of Section 2.3 hereof, absolutely and irrevocably transfer, absolutely assign, convey, set over, and deliver to the Liquidating Trust, and its successors and assigns, all right, title and interest of the Debtors in and to the Trust Estate, in trust, to and for the benefit of the Beneficiaries for the uses and purposes stated herein and in the Plan, except as may otherwise be specifically provided by the Plan. The Debtors shall execute and deliver such other instruments of sale, transfer, conveyance, assignment and confirmation, and will cooperate and take such other actions as the Liquidating Trustee may deem reasonably necessary or desirable in order to more effectively transfer, convey and assign all rights, title and interests in and to the Trust Estate to the Liquidating Trust.

2.2. Purpose of Liquidating Trust.

This Liquidating Trust is created and organized for the sole purposes of collecting, holding, liquidating, and distributing the Trust Estate and administering, compromising, settling, withdrawing, objecting to, or litigating any and all Causes of Action and objections to the Claims under the Plan, with no objective to engage in the conduct of a trade or business, except to the extent reasonably necessary to, and consistent with, the liquidating purpose of the Liquidating Trust. In accordance with such express and limited purposes, as of the Effective Date, the Liquidating Trust is hereby authorized and directed: (a) to take any and all steps necessary to maintain the Liquidating Trust (except with respect to the Disputed Claims Reserve) as one or more liquidating trusts for federal income tax purposes in accordance with Treasury Regulation § 301.7701-4(d) and as one or more "grantor trusts" subject to the provisions of Subchapter J, Subpart E of the IRC unless otherwise required; (b) to take all reasonable and necessary actions to conserve and protect the Trust Estate; (c) to administer, compromise, settle, and litigate any and all Causes of Action and any other Claims or causes of action belonging to the Liquidating Trust; (d) to the extent necessary and appropriate, object to any Claims asserted against the Debtors' Estates and the Liquidating Trust; and (e) to maintain, operate or lease (for purposes of holding for sale), or sell or otherwise liquidate or dispose of the Trust Estate, in accordance with the terms of this Liquidating Trust Agreement, the Plan and the Confirmation Order, and to distribute the net proceeds of such disposition to the Beneficiaries, in as prompt, efficient and orderly a fashion as possible in accordance with the provisions of Section 5 hereof.

2.3. Title to Assets of the Debtors.

Upon the transfer of the Trust Assets to the Liquidating Trust, the Liquidating Trustee shall succeed to all of the Debtors' right, title and interest in the Trust Assets and the

Debtors will have no further interest in or with respect to the Trust Assets or the Liquidating Trust.

2.4. Segregated Funds.

- On the Effective Date, the Liquidating Trustee shall establish a segregated deposit account to be used solely for the purpose of funding of the Trust Administrative Fund (defined below). The Liquidating Trustee Fee Amount shall be funded into the account by the First Lien Facility Agent from Sale Proceeds to which the First Lien Facility Agent or First Lien Claimholders would otherwise be entitled and shall be free and clear of all Liens and Claims of the First Lien Facility Agent or the First Lien Claimholders, including the First Lien Facility Secured Claim, any First Lien Facility Adequate Protection Claim and any First Lien Facility Deficiency Claim. In addition, the Liquidating Trustee shall fund into the account any additional amount the Liquidating Trustee deems necessary to fund post-Effective Date operations other than fees and expenses incurred from the Effective Date through and including the date all First Lien Facility Claims are paid in full in Cash in connection with (i) the collection of accounts receivable, (ii) the pursuit of Causes of Action (including 549 Avoidance Actions but excluding for the avoidance of doubt Avoidance Actions), and (iii) the objection to, and settlement of, Other Secured Claims (including Possessory Lienholder Claims), Administrative Expense Claims, Priority Tax Claims and Other Priority Claims, from the Other Assets. Any unspent Cash remaining in the Trust Administrative Fund immediately prior to the Final Distribution Date shall be distributed to the Beneficiaries as if such Cash were Other Assets.
- Ъ. The Liquidating Trustee shall establish a segregated deposit account to be used solely for the purpose of funding the First Lien Trust Administrative Fund (defined below). From time to time as agreed between the First Lien Facility Agent and the Liquidating Trustee, the First Lien Facility Agent shall fund amounts into the account to pay the Liquidating Trustee First Lien Fees incurred from the Effective Date through and including the date all First Lien Facility Claims are paid in full in Cash, in connection with (i) the collection of accounts receivable, (ii) the pursuit of Causes of Action (including 549 Avoidance Actions but excluding for the avoidance of doubt Avoidance Actions), and (iii) the objection to, and settlement of, Other Secured Claims (including Possessory Lienholder Claims), Administrative Expense Claims, Priority Tax Claims and Other Priority Claims from Sale Proceeds to which the First Lien Facility Agent or First Lien Claimholders would otherwise be entitled. Any unspent Cash remaining in the First Lien Trust Administrative Fund after the payment of all Liquidating Trustee First Lien Fees intended to be paid therefrom shall be treated as First Lien Available Cash and distributed to the First Lien Facility Agent for the benefit of the First Lien Claimholders.
- c. On the Effective Date, to the extent the Liquidating Trustee determines that the amounts in the Reserve Accounts are insufficient to pay the Allowed Claims permitted to be paid from each Reserve Account in accordance with the Plan in full, the Liquidating Trustee may establish a segregated deposit account to be used solely for the payment of Other Available Cash to holders of Allowed Other Secured Claims (to the extent not otherwise satisfied in accordance with the terms of the Plan), Allowed Administrative Expense Claims, Allowed Priority Tax Claims and Allowed Other Priority Claims. The account shall be funded solely

from Other Available Cash. To the extent Cash remains in such account following the payment in full of all Allowed Other Secured Claims (to the extent not otherwise satisfied in accordance with the terms of the Plan), Allowed Administrative Expense Claims, Allowed Priority Tax Claims and Allowed Other Priority Claims, such Cash shall be distributed to the Beneficiaries as if such Cash were Other Assets.

2.5. Tax Treatment of Transfer of Trust Assets to the Liquidating Trust.

For all federal income tax purposes, all parties (including, without limitation, the Debtors, the Liquidating Trustee, and the Beneficiaries) shall treat the transfer of Trust Assets to the Liquidating Trust, as set forth in Sections 2.1, 2.3 and 2.4 of this Liquidating Trust Agreement and in accordance with the Plan, as a transfer of such assets to the Beneficiaries and a transfer by the Beneficiaries of such assets to the Liquidating Trust. In all events, the Beneficiaries of the Liquidating Trust shall be treated as the grantors and deemed owners of the Trust Assets.

2.6. Transfer Free and Clear of Claims.

In accordance with Section 2.1 hereof, except as set forth in the Plan, the Debtors hereby transfer and assign the Trust Assets to the Liquidating Trust free and clear of any Liens, Claims, interests, encumbrances or any liability of any kind (except that the Trust Assets other than the Other Assets shall remain subject to the Liens securing (i) the First Lien Facility Secured Claim and any First Lien Facility Adequate Protection Claim as provided in Section 5.2 of the Plan and (ii) the Second Lien Facility Secured Claim and any Second Lien Facility Adequate Protection Claim as provided in Section 5.3 of the Plan), and the Liquidating Trustee on behalf of the Liquidating Trust hereby assumes and agrees that, except as set forth in the Plan, all such Trust Assets will be transferred to the Liquidating Trust free and clear of any Liens, Claims, interests, encumbrances or any liability of any kind.

2.7. Transfer of Trust Assets.

Except as otherwise provided by the Plan or this Liquidating Trust Agreement, upon the Effective Date, title to the Trust Assets shall pass to the Liquidating Trust free and clear of all claims and Interests, in accordance with Section 1141 of the Bankruptcy Code. The Liquidating Trustee shall not be permitted to receive or retain Cash or Cash equivalents in excess of a reasonable amount to meet distributions as provided herein and the Plan or to maintain the value of the Trust Assets during liquidation.

On the Effective Date, (a) the Debtors' equity interests in Graceway Canada and (b) the Debtors' Claims against Graceway Canada arising from any Intercompany Claims, loans, notes, transfers or other obligations shall be transferred by the Debtors to the Liquidating Trust, and the equity interests in Graceway Canada shall continue to be subject to the Liens securing the First Lien Facility Secured Claim and any First Lien Facility Adequate Protection Claim as provided in Section 5.2 of the Plan and the Liens of the Second Lien Facility Agent in favor of the Second Lien Facility Lenders as provided in Section 5.3 of the Plan; provided, however, that the portion of the Canadian Distribution Amount used to fund the Estate Fund Amount shall be free and clear of the Liens of the First Lien Facility Agent in accordance with the Committee

Settlement. The Liquidating Trust shall exercise its powers as the sole shareholder to appoint the Liquidating Trustee as the sole officer and director of Graceway Canada and shall make distributions of the Canadian Distribution Amount as provided under the Plan.

2.8. Property in the Liquidating Trust.

The Liquidating Trust shall hold the legal title to all property at any time constituting a part of the Trust Estate and shall hold such property in trust to be administered and disposed of by it pursuant to the terms of this Liquidating Trust Agreement, the Plan and the Confirmation Order for the benefit of the Beneficiaries. The Liquidating Trustee is authorized to make disbursements and payments from the Trust Estate in accordance with the provisions of Sections 5 and 6 of this Liquidating Trust Agreement and pursuant to the Plan.

2.9. Valuation of Trust Assets.

As soon as possible after the Effective Date, but in no event later than ninety (90) days thereafter, the Liquidating Trustee, based upon his good faith determination after consultation with his counsel, shall inform the Beneficiaries in writing solely as to his estimate of the value of the Trust Assets. The valuation shall be used consistently by all parties (including, without limitation, the Debtors, the Liquidating Trustee, and the Beneficiaries) for all federal income tax purposes, and the parties shall file tax returns consistent with such valuation; provided, however, that such valuation shall not be binding on the Liquidating Trustee or any other party for any other purposes, including without limitation in regard to the liquidation of the Trust Assets, whether by disposition, liquidation, litigation, settlement, or otherwise.

2.10. Continuation of the Automatic Stay.

In furtherance of the implementation of the Plan, except as otherwise provided in the Plan, all injunctions or stays provided for in the Chapter 11 Cases pursuant to Sections 105 or 362 of the Bankruptcy Code, or otherwise, and in existence on the Confirmation Date, shall remain in full force and effect and apply to all creditors and Beneficiaries holding Claims against the Debtors, the Estates, the Assets, the Liquidating Trustee, the Liquidating Trust and the Trust Assets until the Final Distribution Date.

SECTION III BENEFICIAL INTERESTS.

3.1. No Transfer or Exchange.

Unless the Liquidating Trustee determines otherwise, beneficial interests in the Liquidating Trust (the "Beneficial Interests") shall not be transferable. In the event the Liquidating Trustee does authorize the transfer of Beneficial Interests, the Liquidating Trustee, based upon his good faith determination after consultation with his counsel, shall establish procedures to govern the registration and transfer of Beneficial Interests ("Permitted Transfer"). Once such procedures have been established, if ever, the Liquidating Trustee shall notify all holders of Beneficial Interests of such procedures. Notwithstanding the foregoing, a transfer of a Beneficial Interest shall be not be permitted by the Liquidating Trustee if such transfer would be contrary to maintaining the Liquidating Trust as a liquidating trust for federal income tax

purposes in accordance with Treasury Regulation § 301.7701-4(d) and as one or more "grantor trusts" subject to the provisions of Subchapter J, Subpart E of the IRC.

3.2. No Certification.

Unless the Liquidating Trustee determines otherwise, the Beneficial Interests will not be certificated and no security of any sort will be distributed to the Beneficiaries with respect to their interest in the Liquidating Trust. In the event the Liquidating Trustee does permit the certification of the Beneficial Interests, the Liquidating Trustee, based upon his good faith determination after consultation with his counsel, shall establish procedures to govern such certification. Once such procedures have been established, if ever, the Liquidating Trustee shall notify all Beneficiaries of such procedures.

3.3. Absolute Owners.

The Liquidating Trustee may deem and treat the persons who are Beneficiaries (as determined in accordance with the Plan) as the absolute owners of the Beneficial Interests in the Liquidating Trust for the purpose of receiving distributions and payments thereof, or on account thereof, and for all other purposes whatsoever. Unless the Liquidating Trustee receives actual written notice of a Permitted Transfer from the duly authorized transferee not less than thirty (30) days prior to a distribution made pursuant to the terms of this Liquidating Trust Agreement, and subject to the applicable provisions of Bankruptcy Rule 3001(e), the Liquidating Trustee shall have no duty or obligation to make or direct any distributions or payments to such transferee of a Permitted Transfer.

3.4. Means of Payment.

Cash payable to Beneficiaries pursuant to Section 5 hereof will be paid by checks drawn on a domestic bank account maintained by the Liquidating Trust or by wire transfer from a domestic bank account maintained by the Liquidating Trust at the sole option of the Liquidating Trustee.

3.5. Amount of Payment.

The amount of Cash payments and distributions to Beneficiaries shall be made and calculated in accordance with the Plan.

3.6. Acceptance of Conveyance.

The Liquidating Trustee is hereby directed to, and the Liquidating Trustee agrees that he will: (a) accept delivery of the Trust Estate on behalf of the Liquidating Trust; (b) accept all bills of sale, deeds, assumptions and assignments, and all other instruments of conveyance required to be delivered by the Debtors with respect to the Trust Estate transferred to the Liquidating Trustee on behalf of the Liquidating Trust pursuant to or in connection with the Plan, the Confirmation Order, or this Liquidating Trust Agreement; and (c) take such other action as may be required of the Liquidating Trust hereunder, including the receipt and acceptance as part of the Trust Estate of any property or rights, including, without limitation, notes, other negotiable

instruments, Claims, Causes of Action, and other choses-in-action belonging to the Debtors or their Estates.

3.7. Title.

On the Effective Date, legal title to all Trust Assets shall be vested in the Liquidating Trust in accordance with and pursuant to the terms of the Plan and this Liquidating Trust Agreement. Without limiting the foregoing, on the Effective Date, the Liquidating Trustee, on behalf of the Liquidating Trust, shall be: (a) authorized to act and have standing in any court of competent jurisdiction as representative of the Debtors' Estates in respect of any and all claims or causes of action that constitute Causes of Action; and (b) substituted as successor to the Debtors and/or the Creditors' Committee (i) in all actions and proceedings pending or thereafter commenced in the Bankruptcy Court or elsewhere in regard to the Causes of Action, (ii) in all actions and proceedings pending or thereafter commenced in the Bankruptcy Court or elsewhere in connection with or regarding all Claims, and (iii) in any agreement respecting the Trust Assets to which any of the Debtors is a party, including, without limitation, the Committee Settlement.

SECTION IV ADMINISTRATION OF TRUST ESTATE.

4.1. Trust Administrative Fund.

The Liquidating Trustee shall establish a reserve fund (the "Trust Administrative Fund") for the payment by the Liquidating Trustee of all liquidation expenses, including, without limitation, the compensation of the Liquidating Trustee, any attorney, accountant, advisor or other professional retained by the Liquidating Trustee, and the payment of all other reasonable and reasonably anticipated expenses, debts, charges, liabilities and obligations relating to the Trust Estate and its administration. On the Effective Date, the Liquidating Trustee Fee Amount shall be funded into the Trust Administrative Fund by the First Lien Facility Agent from Sale Proceeds to which the First Lien Facility Agent or First Lien Claimholders would otherwise be entitled and shall be free and clear of all Liens and Claims of the First Lien Facility Agent or the First Lien Claimholders, including the First Lien Facility Secured Claim, any First Lien Facility Adequate Protection Claim and any First Lien Facility Deficiency Claim, for the sole purpose of paying the Liquidating Trustee Fees. Any additional amount the Liquidating Trustee deems reasonable and necessary to fund post-Effective Date operations other than fees and expenses incurred from the Effective Date through and including the date all First Lien Facility Claims are paid in full in Cash in connection with (a) the collection of accounts receivable, (b) the pursuit of Causes of Action (including 549 Avoidance Actions but excluding for the avoidance of doubt Avoidance Actions), and (c) the objection to, and settlement of, Other Secured Claims (including Possessory Lienholder Claims), Administrative Expense Claims, Priority Tax Claims and Other Priority Claims, shall be funded into the Trust Administrative Fund from the Other Assets. Any balance remaining in the Trust Administrative Fund after the payment of all expenses, debts, charges, liabilities and obligations intended to be paid therefrom shall be treated as Other Assets and distributed to Beneficiaries as provided in Section 5 hereof. Any monies deposited in the Trust Administrative Fund pursuant to the terms of this Liquidating Trust Agreement shall be invested in interest-bearing deposits or investments specified in Section 4.6(b) of this

Agreement, and the interest earned thereon shall be credited thereto. Any unspent Cash remaining in the Trust Administrative Fund immediately prior to the Final Distribution Date shall be distributed to the Beneficiaries as if such Cash were Other Assets.

4.2. First Lien Trust Administrative Fund.

The Liquidating Trustee shall establish a reserve fund (the "First Lien Trust Administrative Fund"), to be funded from time to time as agreed between the First Lien Facility Agent and the Liquidating Trustee, for the payment by the Liquidating Trustee of all reasonable fees and out-of-pocket expenses for services performed by the Liquidating Trustee, and any attorneys or other professionals hired by the Liquidating Trustee, incurred from the Effective Date through and including the date all First Lien Facility Claims are paid in full in Cash, in connection with (a) the collection of accounts receivable, (b) the pursuit of Causes of Action (including 549 Avoidance Actions but excluding for the avoidance of doubt Avoidance Actions), and (c) the objection to, and settlement of, Other Secured Claims (including Possessory Lienholder Claims), Administrative Expense Claims, Priority Tax Claims and Other Priority Claims from Sale Proceeds to which the First Lien Facility Agent or First Lien Claimholders would otherwise be entitled. Any balance remaining in the First Lien Trust Administrative Fund after the payment of all Liquidating Trustee First Lien Fees intended to be paid therefrom shall be treated as First Lien Available Cash and distributed to the First Lien Facility Agent for the benefit of the First Lien Claimholders. Any monies deposited in the First Lien Trust Administrative Fund pursuant to the terms of this Liquidating Trust Agreement shall be invested in interest-bearing deposits or investments specified in Section 4.6(b) of this Agreement, and the interest earned thereon shall be credited thereto.

4.3. Disputed Claims Reserve.

On the Effective Date, the Liquidating Trustee shall establish a reserve fund solely from the Other Assets other than the Liquidating Trustee Fee Amount (the "Disputed Claims Reserve") for the payment by the Liquidating Trustee of all Disputed Claims if and when Allowed; provided, however, that funds shall be added to the Disputed Claims Reserve for Other Secured Claims (to the extent not otherwise satisfied in accordance with the terms of the Plan), Administrative Expense Claims, Priority Tax Claims and Other Priority Claims that are Disputed Claims only to the extent the amounts deposited in the applicable Reserve Account for the payment of such Claims are insufficient to pay the Face Amount (or, if estimated for distribution purposes, the Estimated Amount) of such Claims. With respect to any unliquidated Disputed Claim, the Disputed Claims Reserve shall be funded upon entry of and in accordance with an Order(s) of the Bankruptcy Court estimating the amount (the "Estimated Amount") of such unliquidated Disputed Claim. Upon ultimate determination that any Disputed Claim is not Allowed, the Liquidating Trustee shall immediately release the portion of the Disputed Claim Reserve or applicable Reserve Account, as applicable, on account of such Claim and distribute such portion in accordance with the Plan. Each time a distribution of Other Available Cash is made to any Class of Claims, the Liquidating Trustee shall deposit into the Disputed Claims Reserve an amount equal to the distribution each holder of a Disputed Claim in such Class would have received were the Face Amount (or, if estimated for distribution purposes, the Estimated Amount) of its Disputed Claim in such Class an Allowed Claim solely from proceeds of Other

Assets. Any Person whose Disputed Claim is so estimated shall be entitled to an Allowed Claim only up to an amount not to exceed the Estimated Amount even if such Person's Disputed Claim, as finally allowed, would have exceeded the maximum Estimated Amount thereof. At the time a Person's Disputed Claim becomes an Allowed Claim, such Person shall receive from the applicable Reserve Account and/or the Disputed Claims Reserve a distribution equal to the distributions such Person would have received on account of its Allowed Claim had it been an Allowed Claim at the time of such prior distributions, with any surplus Cash held in the Disputed Claims Reserve on account of such Disputed Claim becoming generally available for use by the Liquidating Trustee as Other Available Cash in accordance with the Plan and this Liquidating Trust Agreement and any surplus Cash held in any Reserve Account becoming available for distribution by the Liquidating Trustee as First Lien Available Cash in accordance with the Plan. For the avoidance of doubt, no payment shall be made from the Disputed Claims Reserve on account of the First Lien Facility Claims.

4.4. Administrative Powers of the Liquidating Trustee.

During the Liquidating Trustee's administration of the Liquidating Trust, and subject to (i) all the other provisions of this Liquidating Trust Agreement (including, but not limited to, Sections 4.6 and 4.7) and (ii) the Plan, the Liquidating Trustee may exercise the power:

- a. To receive and hold the Trust Assets and to have exclusive possession and control thereof as permissible under applicable law;
- b. To manage, sell and convert all or any portion of the Trust Assets to Cash and distribute the net distributable proceeds as specified in the Plan and this Liquidating Trust Agreement;
- c. To enter into, perform and exercise rights under contracts binding upon the Liquidating Trust (but not upon the Liquidating Trustee in his respective individual or corporate capacity) which are reasonably incident to the administration of the Liquidating Trust and which the Liquidating Trustee, in the exercise of his best business judgment, reasonably believes to be in the best interests of the Liquidating Trust;
- d. To delegate his authority under this Liquidating Trust to other persons, provided that such delegation must be made pursuant to a written agreement that either has been approved by the Bankruptcy Court in conjunction with the Confirmation of the Plan or is approved by the Liquidating Trust Committee (defined below);
- e. To establish and maintain accounts at banks and other financial institutions, in a clearly specified fiduciary capacity, into which the Trust Administrative Fund, First Lien Trust Administrative Fund, Disputed Claims Reserve or other Cash and property of the Liquidating Trust may be deposited, and draw checks or make withdrawals from such accounts, and to pay or distribute such amounts of the Trust Estate as permitted or required under the Plan and this Liquidating Trust Agreement;

- f. To employ attorneys, accountants, appraisers, expert witnesses, insurance adjusters or other persons whose services may be reasonably necessary or advisable, in the sole judgment of the Liquidating Trustee after consultation with the Liquidating Trust Committee, to advise or assist him in the discharge of his duties as Liquidating Trustee or otherwise in the exercise of any powers vested in the Liquidating Trustee, and to pay from the Trust Administrative Fund or First Lien Trust Administrative Fund, as applicable, reasonable compensation to such attorneys, accountants, appraisers, expert witnesses, insurance adjusters or other persons;
- g. To pay any and all reasonable and necessary expenses attributable or relating to the management, maintenance, operation, preservation or liquidation of the Trust Estate:
- h. To investigate, file, compromise, settle, withdraw or litigate in the Bankruptcy Court or on appeal (or pursuant to a withdrawal of the reference of jurisdiction) objections to Claims filed against the Debtors' Estates, the Trust Estate or the Liquidating Trust; provided, however, that (A) on and after the Effective Date and until such time as the First Lien Facility Claims are paid in full in Cash, in order to compromise, settle, otherwise resolve, or withdraw any objections to Disputed Claims, the Liquidating Trustee shall be required to obtain (i) the prior written consent of the Approving Majority First Lien Lenders or (ii) an Order of the Bankruptcy Court approving such compromise, settlement, other resolution, or withdrawal of any objections to Disputed Claims and (B) once the First Lien Facility Claims are paid in full in Cash until such time as the Second Lien Facility Claims are paid in full in Cash, in order to compromise, settle, otherwise resolve, or withdraw any objections to Disputed Claims, the Liquidating Trustee shall be required to obtain (i) the prior written consent of the Second Lien Facility Agent or (ii) an Order of the Bankruptcy Court approving such compromise, settlement, other resolution, or withdrawal of any objections to Disputed Claims;
- i. To investigate, analyze, compromise, adjust, arbitrate, sue on or defend, pursue, prosecute, abandon, or otherwise deal with and settle, in accordance with the terms set forth in this Liquidating Trust Agreement, any and all Causes of Action and claims in favor of or against the Debtors, the Debtors' Estates, the Trust Estate or the Liquidating Trust that have not been released under the Plan or otherwise as the Liquidating Trustee shall deem advisable; provided, however, that (A) on and after the Effective Date and until such time as the First Lien Facility Claims are paid in full in Cash, in order to settle or abandon Causes of Action, the Liquidating Trustee shall be required to obtain (i) the prior written consent of the Approving Majority First Lien Lenders or (ii) an Order of the Bankruptcy Court approving such settlement or abandonment after notice and a hearing and (B) once the First Lien Facility Claims are paid in full in Cash, in order to settle or abandon Causes of Action, the Liquidating Trustee shall be required to obtain (i) the prior written consent of the Second Lien Facility Agent or (ii) an Order of the Bankruptcy Court approving such settlement or abandonment after notice and a hearing;
- j. To avoid and recover transfers of the Debtors' property as may be permitted by the Bankruptcy Code or applicable state law, including, without limitation, those transfers identified in the Disclosure Statement;

- k. To take all appropriate action with respect to the Trust Estate, including, without limitation, the filing, prosecution, settlement or other resolution of Claims and Causes of Action, subject to the limitations set forth in subjections (i) and (j) above;
- 1. To sue or be sued in connection with any matter arising from or related to the Plan or this Liquidating Trust Agreement that affects in any way the rights or obligations of the Liquidating Trust, the Liquidating Trustee or the Beneficiaries;
- m. To represent the interests of the Beneficiaries with respect to any matters relating to the Plan, this Liquidating Trust Agreement, or the Liquidating Trust affecting the rights of such Beneficiaries;
- n. If the Liquidating Trust shall become subject to federal or state income tax, the Liquidating Trustee shall have the power, exercisable at his reasonable discretion, to take any action reasonably necessary to minimize any adverse federal or state income tax consequences to the Beneficiaries resulting from any distribution made by the Liquidating Trust to such Beneficiaries;
- o. In general, without in any manner limiting any of the foregoing or the following, to deal with the Trust Assets or any part or parts thereof in all other ways as would be lawful for any person owning the same to deal therewith; *provided*, *however*, that the investment powers of the Liquidating Trustee, other than those reasonably necessary to maintain the value of the Trust Assets of the Liquidating Trust and to further the liquidating purpose of the Liquidating Trust, are limited by the terms herein;
- p. To do any and all other things not in violation of any other terms of the Plan, the Confirmation Order and this Liquidating Trust Agreement that, in the reasonable business judgment of the Liquidating Trustee, are necessary or appropriate for the proper liquidation, management, investment and distribution of the Trust Assets in accordance with the provisions of this Liquidating Trust Agreement and the Plan;
- q. To file tax returns for the Liquidating Trust and the Disputed Claims Reserve, including pursuant to Treasury Regulation § 1.468B-9, and for each of the Debtors; and
- r. At the appropriate time, to request that the Bankruptcy Court enter a final decree closing each of the Debtors' Chapter 11 Cases; *provided*, *however*, that, in light of the substantive consolidation of the Debtors' estates for distribution purposes, the Liquidating Trustee may, in the Liquidating Trustee's sole discretion, seek authority from the Bankruptcy Court to close all of the Debtors' Chapter 11 Cases except for the Graceway Chapter 11 Case immediately upon satisfaction of all Other Secured Claims.

4.5. Objections to Claims.

From and after the Effective Date of the Plan, the Liquidating Trust, through the Liquidating Trustee, shall be the sole representative of the Debtors' Estates for all purposes, including, without limitation, investigating, settling, compromising, objecting to, and litigating in

the Bankruptcy Court or on appeal (or pursuant to a withdrawal of the reference of jurisdiction) objections to Claims.

- 4.6. Limitations on Liquidating Trustee; Investments; Debt.
- a. No Trade or Business. The Liquidating Trustee shall carry out the purposes of the Liquidating Trust and the directions contained herein and shall not at any time cause the Liquidating Trust to enter into or engage in any trade or business (except as may be consistent with the limited purposes of the Liquidating Trust), including, without limitation, the purchase of any assets or property (other than such assets or property as are reasonably necessary to carry out the purposes of the Liquidating Trust Agreement, on behalf of the Liquidating Trust or the Beneficiaries). The Liquidating Trustee is directed to take all reasonable and necessary actions to dispose of the Trust Estate in as prompt, efficient and orderly a fashion as possible, to make timely distributions of the proceeds of the Trust Estate, and to otherwise not unduly prolong the duration of the Liquidating Trust.
- b. Investments. The Liquidating Trustee shall invest any monies held at any time as part of this Trust Estate, including, without limitation, in the Trust Administrative Fund, First Lien Trust Administrative Fund and every other reserve or escrow fund established pursuant to the terms of this Liquidating Trust Agreement, only in interest-bearing deposits or certificates of deposit issued by any federally insured banking institution or short-term obligations of, or short- term obligations unconditionally guaranteed as to payment by, the United States of America and its agencies or instrumentalities, pending the need for the disbursement thereof in payment of costs, expenses, and liabilities of the Liquidating Trust or in making distributions pursuant to Section 5 of this Liquidating Trust Agreement. The Liquidating Trustee shall be restricted to the collection and holding of such monies and any income earned on such monies and to the payment and distribution thereof (at least annually if such monies are not necessary to maintain the value of the Trust Estate or to satisfy Claims against the Trust Estate) for the purposes set forth in the Plan and this Liquidating Trust Agreement, and to the conservation and protection of the Trust Estate in accordance with the provisions hereof.
- c. Debt. Neither the Liquidating Trust, nor the Liquidating Trustee on behalf of the Liquidating Trust, shall incur any debt.
 - 4.7. Limitations on Liquidating Trustee Causes of Action and Abandonment.
- a. The Liquidating Trustee shall be obligated to consult with the Liquidating Trust Committee regarding all matters affecting the Trust Estate involving \$750,000 or more, including, but not limited to, Claims objections, litigation, contested matters and the sale, transfer or abandonment of other Trust Assets. Notwithstanding anything contained herein, a member of the Liquidating Trust Committee must recuse itself from the Liquidating Trust Committee regarding any contested matter, objection to Claim, or other Cause of Action to which it is a party or otherwise has a conflict of interest.
- b. Subject to Sections 7.11 and 8.2 of the Plan, Section 4.4(i) herein and this clause b, on and after the Effective Date, the Liquidating Trustee shall have the exclusive authority to settle or abandon Causes of Action or abandon other Trust Assets owned by the

Liquidating Trust. On and after the Effective Date and until such time as the First Lien Facility Claims are paid in full in Cash, in order to settle or abandon Causes of Action or abandon other Trust Assets owned by the Liquidating Trust, the Liquidating Trustee shall be required to obtain (i) the prior written consent of the Approving Majority First Lien Lenders or (ii) an Order of the Bankruptcy Court approving such settlement or abandonment after notice and a hearing. After the First Lien Facility Claims are paid in full in Cash and until such time as the Second Lien Facility Claims are paid in full in Cash, in order to settle or abandon Causes of Action or abandon other Trust Assets owned by the Liquidating Trust, the Liquidating Trustee shall be required to obtain (i) the prior written consent of the Second Lien Facility Agent or (ii) an Order of the Bankruptcy Court approving such settlement or abandonment after notice and a hearing. After the Second Lien Facility Claims are paid in full in Cash, except as otherwise set forth in the Plan, the Confirmation Order or the Liquidating Trust Agreement, the Liquidating Trustee shall not be required to obtain approval of the Approving Majority First Lien Lenders, the Second Lien Facility Agent or the Bankruptcy Court to settle or abandon Causes of Action or abandon other Trust Assets owned by the Liquidating Trust.

- c. In all other events, except as otherwise set forth herein, the Liquidating Trustee is authorized to compromise or settle an action without any notice or consent if the Liquidating Trustee reasonably believes such settlement or compromise to be in the best interests of the Liquidating Trust, and shall be held harmless by the Beneficiaries in taking such action.
- d. Notwithstanding anything herein to the contrary, the Liquidating Trustee shall be authorized to pay the reasonable fees and expenses incurred by its professionals after the Effective Date without providing notice to or obtaining the approval of any party; *provided*, *however*, that such payments may be made only from the Trust Administrative Fund or First Lien Trust Administrative Fund, as applicable.

4.8. Transferee Liabilities.

If any liability shall be asserted against the Liquidating Trust as transferee of the Trust Estate on account of any claimed liability of or through the Debtors, the Liquidating Trustee may use such part of the Trust Administrative Fund as may be necessary in contesting any such claimed liability and in payment, compromise, settlement and discharge thereof on terms reasonably satisfactory to the Liquidating Trustee; provided, however, that if any such claimed liability arises on account of the Liquidating Trustee's (a) collection of accounts receivable, (b) pursuit of Causes of Action (including 549 Avoidance Actions but excluding for the avoidance of doubt Avoidance Actions), and (c) objection to, and settlement of, Other Priority Tax Claims and Other Priority Claims, the Liquidating Trustee may use such part of the First Lien Administrative Trust Fund as may be necessary in contesting any such claimed liability and in payment, compromise, settlement and discharge thereof on terms reasonably satisfactory to the Liquidating Trustee. In no event shall the Liquidating Trustee be required or obligated to use his own property, funds or assets for any such purposes.

4.9. Administration of Trust.

In administering the Liquidating Trust, the Liquidating Trustee, subject to the express limitations contained herein, is authorized and directed to do and perform all such acts, to execute and deliver such deeds, bills of sale, instruments of conveyance, and other documents as he may deem reasonably necessary or advisable to carry out the purposes of the Liquidating Trust.

4.10. Payment of Expenses and Other Liabilities.

To the extent that the amount of funds in the Trust Administrative Fund is at any time insufficient, the Liquidating Trustee shall pay solely from the Other Assets all reasonable expenses, charges, liabilities and obligations of the Liquidating Trust other than fees and expenses incurred from the Effective Date through and including the date all First Lien Facility Claims are paid in full in Cash in connection with (a) the collection of accounts receivable. (b) the pursuit of Causes of Action (including 549 Avoidance Actions but excluding for the avoidance of doubt Avoidance Actions), and (c) the objection to, and settlement of, Other Secured Claims (including Possessory Lienholder Claims), Administrative Expense Claims, Priority Tax Claims and Other Priority Claims, but including, without limiting the generality of the foregoing, such debts, liabilities, or obligations as may be payable from the Trust Estate, interest, taxes, assessments, and public charges of every kind and nature, and the costs, charges and expenses in connection with or arising out of the execution or administration of the Liquidating Trust and the Trust Estate, and such other payments and disbursements as are provided for in this Liquidating Trust Agreement or which may be reasonably determined by the Liquidating Trustee to be proper charges against the Liquidating Trust and the Trust Estate, and which the Liquidating Trustee, in his reasonable discretion and business judgment, may determine to be necessary or advisable to meet or satisfy unascertained, unliquidated or contingent liabilities of the Liquidating Trust. To the extent that the amount of funds in the First Lien Trust Administrative Fund is at any time insufficient after a request by the Liquidating Trustee to the First Lien Facility Agent to make a deposit into the First Lien Trust Administrative Fund, the Liquidating Trustee may, in the Liquidating Trustee's sole discretion, pay from any of the Reserve Accounts all reasonable expenses, charges, liabilities and obligations of the Liquidating Trust incurred from the Effective Date through and including the date all First Lien Facility Claims are paid in full in Cash in connection with (a) the collection of accounts receivable, (b) the pursuit of Causes of Action (including 549 Avoidance Actions but excluding for the avoidance of doubt Avoidance Actions), and (c) the objection to, and settlement of, Other Secured Claims (including Possessory Lienholder Claims), Administrative Expense Claims, Priority Tax Claims and Other Priority Claims. For the avoidance of doubt, fees and expenses incurred in pursuing any Avoidance Actions shall ultimately reduce any proceeds received by the Liquidating Trust on account of such Avoidance Actions prior to distribution of such proceeds in accordance with the Plan. The Liquidating Trustee shall make such payments without application to or Order of the Bankruptcy Court, except as otherwise herein provided.

4.11. Payment of United States Trustee's Fees.

After the occurrence of the Effective Date, fees payable to the Office of the United States Trustee during the administration of the Plan and until the case is converted, dismissed or closed shall be paid by the Liquidating Trustee from the Trust Administrative Fund.

4.12. Liquidating Trustee Fees and Liquidating Trustee First Lien Fees.

The Liquidating Trustee is entitled to reasonable fees and out-of-pocket expenses for services performed by the Liquidating Trustee and any attorneys or other professionals hired by the Liquidating Trustee. The Liquidating Trustee Fees and Liquidating Trustee First Lien Fees will be paid in accordance with Schedule 4.12 hereto, provided that, subject to Section 4.10 hereof, (a) Liquidating Trustee Fees shall be paid solely from the Liquidating Trustee Fee Amount and the Other Assets, (b) Liquidating Trustee First Lien Fees shall be paid solely from the Sale Proceeds to which the First Lien Facility Agent or First Lien Claimholders would otherwise be entitled and (c) Contingency Fee Counsel Fees and Expenses shall be paid from Causes of Action Proceeds.

4.13. Fiscal Year.

The Liquidating Trust's fiscal year shall end on December 31 of each year, unless the Liquidating Trustee deems it advisable to establish some other date on which the fiscal year of the Liquidating Trust shall end.

4.14. Reports.

The Liquidating Trustee shall:

- a. Prepare and file unaudited interim financial reports as may be required by regulatory authorities, applicable laws, rules or regulations or as the Liquidating Trustee deems advisable during the fiscal year;
- b. Prepare, file and mail, within the time required by applicable law or regulation, necessary income tax information, tax returns or reports to the Beneficiaries and applicable taxing authorities, including, on an annual basis, the manner and calculation of the Liquidating Trust's taxable income or loss which the Liquidating Trust would recognize if it were a separate taxable entity. In this connection, the Liquidating Trustee shall file returns for the Liquidating Trust, except with respect to the Disputed Claims Reserve, as one or more grantor trusts pursuant to Treasury Regulation Section 1.671-4(a);
- c. Prepare and provide to the Liquidating Trust Committee periodic budgets acceptable to the Liquidating Trust Committee, which set forth, among other things, Liquidating Trustee Fees and Liquidating Trustee First Lien Fees expected to be incurred over the budgeted period; and
- d. Once each quarter, the Liquidating Trustee shall File with the Bankruptcy Court and serve on the United States Trustee, the Liquidating Trust Committee, and any

Beneficiary who requests copies of such quarterly report after the Confirmation Date, an unaudited written report setting forth the amounts and timing of all distributions made and expenses paid by the Liquidating Trustee under the Plan in each applicable reporting period and the recipients thereof. Each quarterly report shall also state the Liquidating Trust's Cash balances as of the beginning and ending of each such period. Quarterly reports shall be provided no later than the fifteenth (15th) day of each January, April, July and October until all Final Distributions under the Plan have been made.

4.15. Liquidating Trust Committee.

An administrative trust committee (the "Liquidating Trust Committee"), consisting of no more than five (5) members, shall consult from time to time on various matters as set forth in this Liquidating Trust Agreement. The Liquidating Trust Committee initially shall consist of (a) three (3) members appointed by the First Lien Facility Agent, (b) one (1) member appointed by the Second Lien Facility Agent, and (c) one (1) member appointed by the Creditors' Committee. Each member of the Liquidating Trust Committee shall serve for a term of two (2) years. Upon the conclusion of a member's term, or in the event that a member resigns prior to the expiration of his term or is otherwise unable to serve out his term (other than due to the payment in full of the First Lien Facility Claims as discussed below), he shall appoint a successor to serve on the Liquidating Trust Committee. If, for any reason, such member is unable to appoint a successor member, the Liquidating Trustee shall appoint such member. Within ten (10) Business Days following payment in full of the First Lien Facility Claims, the three (3) members appointed by the First Lien Facility Agent (or its successors) shall resign from the Liquidating Trust Committee. Thereafter, the member appointed by the Second Lien Facility Agent (or his successor) shall appoint one (1) additional member. Unless otherwise specified herein, approval of a majority of the members of such Liquidating Trust Committee shall be required for the Liquidating Trust Committee to act. The Liquidating Trust Committee shall have the rights and powers set forth in this Liquidating Trust Agreement. In the event that a Liquidating Trust Committee shall not be formed and continuing to exist under this Liquidating Trust Agreement, all references herein to required approval or other action of such Liquidating Trust Committee shall be of no force or effect. In performance of their duties hereunder, members of the Liquidating Trust Committee shall be entitled to receive reimbursement of reasonable costs, expenses and obligations as set forth in Section 7.6e(iii) of the Liquidating Trust Agreement.

4.16. Removal of the Liquidating Trustee.

The Liquidating Trust Committee shall be authorized, by unanimous vote of the members, to remove the Liquidating Trustee for reasonable cause and to select a replacement or successor liquidating trustee in accordance with the provisions of this Liquidating Trust Agreement.

SECTION V PAYMENTS AND DISTRIBUTIONS.

5.1. Distributions to Beneficiaries.

- a. Distributions to holders of Allowed Other Secured Claims (to the extent not otherwise satisfied in accordance with the terms of the Plan), Allowed Administrative Expense Claims (other than Administrative Expense Claims that are Assumed Liabilities), Allowed Priority Tax Claims and Allowed Other Priority Claims shall be made from the Reserve Accounts to the extent permitted by the Budget and/or the Sale Support Agreement and in accordance with the Plan. To the extent any of such Allowed Claims are not satisfied from the Reserve Accounts (and, with respect to Allowed Other Secured Claims, are not otherwise satisfied in accordance with the terms of the Plan), such Allowed Claims shall be paid from the Other Available Cash in accordance with the Plan.
- b. Distributions to holders of Allowed First Lien Facility Claims shall be made in accordance with the Plan.
- c. Distributions to holders of any Allowed Second Lien Facility Secured Claims or Allowed Second Lien Facility Adequate Protection Claims shall be made on account of such Claims solely from any Other Available Cash upon which the Second Lien Facility Agent has a Lien in accordance with the Plan unless and until the Allowed First Lien Facility Claims are paid in full in Cash, and thereafter, in accordance with the Plan.
- d. Distributions to holders of Allowed General Unsecured Claims shall be made solely from the Other Available Cash that is not subject to the Liens granted to the Second Lien Facility Agent pursuant to the Second Lien Credit Agreement or the DIP Order after payment in full in Cash of the Allowed Administrative Expense Claims, Allowed Priority Tax Claims, Allowed Other Priority Claims, and Allowed Other Secured Claims (to the extent not otherwise satisfied in accordance with the terms of the Plan), all in accordance with the Plan.
- e. Not later than ten (10) Business Days after receipt thereof, after the liquidation into Cash of Causes of Action (other than those Causes of Action abandoned by the Liquidating Trustee in accordance with the terms hereof and the Plan), the Liquidating Trustee shall distribute Net Causes of Action Proceeds to the holders of Allowed Claims in accordance with Sections 5.2, 5.3 and 5.4 of the Plan.
- f. From and after the Effective Date, the Liquidating Trustee shall distribute on a bi-weekly basis all collections in respect of accounts receivable during each such two-week period in accordance with Sections 5.2 and 5.3 of the Plan.
- g. Payments to the Beneficiaries hereunder shall be made only from the proceeds of the Trust Assets, and only to the extent that the Liquidating Trustee shall have sufficient proceeds from the Trust Assets to make such payments in accordance with this Section 5 and the Plan. Each Beneficiary shall be entitled to distributions solely from those Trust Assets available for distribution to such Beneficiary as expressly provided in the Plan.

5.2. Frequency and Amounts of Cash Payments.

- a. The Liquidating Trustee shall make distributions to Beneficiaries holding Allowed Other Secured Claims (to the extent not otherwise satisfied in accordance with the terms of the Plan), Administrative Expense Claims, Priority Tax Claims and Other Priority Claims in accordance with the Plan as soon as such Claims become Allowed Claims.
- b. From time to time in accordance with Sections 10.8.2, 10.8.3, 10.8.4 and 10.8.5 of the Plan, the Liquidating Trustee shall make distributions to the First Lien Facility Agent for the benefit of the First Lien Claimholders.
- c. From time to time in accordance with Section 10.8.3 of the Plan, the Liquidating Trustee shall make distributions to the Second Lien Facility Agent for the benefit of the Second Lien Facility Lenders from the Other Available Cash upon which the Second Lien Facility Agent has a Lien, unless and until the First Lien Facility Claims are paid in full in Cash, at which point such distributions shall be from the Other Available Cash upon which the Second Lien Facility Agent has a Lien in accordance with Section 10.8.3 of the Plan and the First Lien Available Cash in accordance with Sections 10.8.2, 10.8.3, 10.8.4 and 10.8.5 of the Plan.
- As often as in the reasonable discretion and judgment of the Liquidating Trustee there shall be Other Available Cash in an amount sufficient to render feasible a distribution of Cash to the holders of Allowed General Unsecured Claims, the Liquidating Trustee shall transfer and pay, or cause to be transferred and paid, to the holders of Allowed General Unsecured Claims such aggregate amount of Other Available Cash, if any, as shall then be held in the Liquidating Trust, excluding reasonable amounts of Cash held in the Trust Administrative Fund pursuant to Section 4.1 hereof or otherwise needed to pay the expenses, debts, charges, liabilities and obligations of the Liquidating Trust (the "Distribution Amount"). The aggregate amounts required to be distributed to the holders of Allowed General Unsecured Claims shall be determined by the Liquidating Trustee pursuant to and in accordance with the terms of the Plan and this Liquidating Trust Agreement. The Distribution Amount(s) shall be paid to the holders of Allowed General Unsecured Claims at least once every three (3) months and shall be determined by the Liquidating Trustee in his reasonable discretion, and his determination shall be final and conclusive on all persons, in the absence of gross negligence or willful misconduct on the part of the Liquidating Trustee, and shall not be reviewed by the Bankruptcy Court. In determining the amount of any such distribution, the Liquidating Trust may rely and shall be fully protected in relying upon the advice and opinion of independent public accountants or of counsel to the Liquidating Trust.

5.3. Establishment of the Claim Accounts.

The Liquidating Trustee will establish on the Liquidating Trust's books and records an account representing each Allowed or Disputed Claim as set forth on the official claims register maintained by BMC Group, Inc. as claims agent or its successor (each, a "Claim Account"). It is expressly understood that the establishment of the Claim Accounts by the Liquidating Trustee or his agents is solely for administrative convenience, and that amounts allocable to such Claim Accounts need not be segregated and may be commingled for investment

purposes as specified herein. The Liquidating Trustee may rely on the official claims register as correct.

5.4. Distributions from Trust Estate.

Distributions will be made to the Beneficiaries in accordance with the terms of the Plan and the Claim Accounts will be adjusted by the Liquidating Trustee to reflect such distributions.

5.5. Fractional Distributions.

No distribution in fractions of cents shall be issued. If the Distribution Amount allocated to an Allowed Claim at the time of a distribution hereunder would include fractions of cents, the amount to be distributed shall be rounded down to the highest integral number of cents in the applicable Claim Account, but such rounding down shall not affect such allocation. The aggregate amount of the retained fractional distributions from the Distribution Amount shall be retained in the Liquidating Trust by the Liquidating Trustee and shall remain part of the Trust Estate.

5.6. Final Distribution.

After (a) the distribution of the payments described in Section 10.8.1 of the Plan, (b) the liquidation into Cash of all Trust Assets owned by the Liquidating Trust (other than those Trust Assets abandoned by the Liquidating Trustee in accordance with the terms of the Plan), (c) the collection of other sums due or otherwise remitted or returned to the Estates, (d) the resolution of all Disputed Claims and (e) the payment in full in Cash of all Liquidating Trustee Fees and Liquidating Trustee First Lien Fees, the Liquidating Trustee shall prepare a final accounting of any and all monies remaining in any accounts maintained by the Liquidating Trustee on behalf of the Liquidating Trust, including the Trust Administrative Fund, First Lien Trust Administrative Fund and any escrow funds, or otherwise remaining in the Trust Estate (the "Final Cash"). Once the amount of the Final Cash has been determined, the Liquidating Trustee shall make the Final Distribution, in accordance with the Plan and this Liquidating Trust Agreement; provided, however, that if the Final Cash consisting of Other Available Cash is less than \$10,000, and the Liquidating Trustee, in his sole discretion, determines that it would cost more than \$5,000 to distribute such funds, the Liquidating Trustee may donate such funds to a charity of his choice.

5.7. Termination Date.

The Liquidating Trust shall terminate on such date when (a) a final decree has been entered closing all of the Chapter 11 Cases, (b) all Trust Assets have been distributed and (c) all claims have been pursued or abandoned pursuant to, and in accordance with, the Plan and this Liquidating Trust Agreement (the "Termination Date"). However, the term of the Liquidating Trust shall not exceed five (5) years from the Effective Date; provided that, upon a finding that an extension is necessary to the liquidating purpose of the Liquidating Trust and upon approval by the Bankruptcy Court, the term may be extended for a finite term based on the particular facts and circumstances. Each such extension must be approved by the Bankruptcy

Court within six (6) months of the beginning of the extended term. The Liquidating Trustee shall be released of all liabilities and discharged from his obligations under the Plan or the Liquidating Trust Agreement once the Liquidating Trust has terminated pursuant to this section.

SECTION VI OTHER DUTIES OF THE TRUSTEE.

6.1. Management of Trust Estate.

With respect to the Trust Assets, the Liquidating Trustee may, if sufficient funds are available from the Trust Administrative Fund or Other Available Cash, purchase and maintain in existence such insurance as the Liquidating Trustee deems reasonable and necessary or appropriate from time to time to protect the Liquidating Trust, the Trust Assets, the Liquidating Trustee, the Liquidating Trust Committee, and the Beneficiaries' interests in the Liquidating Trust or from any potential claims or liabilities relating thereto or the distribution thereof.

6.2. Tax and Related Matters.

Pursuant to and in accordance with the Plan, the Liquidating Trustee shall be responsible for all tax matters of the Trust Estate, including, but not limited to, the filing of all tax returns and other filings with governmental authorities on behalf of the Trust Estate, the Debtors' Estates and any subsidiaries (whether organized as a corporation, limited liability company or partnership and whether owned in whole or in part) for time periods ending on or before the last day in the taxable year of the Liquidating Trust including the Termination Date, including the filing of tax returns for the Liquidating Trust, except with respect to the Disputed Claims Reserve, as one or more grantor trusts pursuant to § 1.671-4(a) of the United States Income Tax Regulations, the filing of determination requests under Section 505(b) of the Bankruptcy Code, and responding to any tax audits of the Trust Estate. The Liquidating Trustee shall provide such information to the Beneficiaries as will enable them to properly file their separate tax returns and withhold and pay over any amounts required by tax law. The Liquidating Trustee is authorized to act as agent for the Trust Estate in withholding or paying over any amounts required by law (including tax law) to be withheld or paid with respect to the Trust Estate. Except as otherwise set forth in this Liquidating Trust Agreement or the Plan, any items of income, deduction, credit, or loss of the Liquidating Trust not allocable to the Disputed Claims Reserve shall be allocated for federal income tax purposes among the Beneficiaries in accordance with each Beneficiary's entitlement to the Trust Assets under the Plan for the relevant taxable period. The Liquidating Trust shall file all income tax returns with respect to any income attributable to the Disputed Claims Reserve and shall pay the federal, state and local income taxes attributable to the Disputed Claims Reserve, based on the items of income, deduction, credit or loss allocable thereto. The Liquidating Trustee shall be entitled to deduct any federal or state withholding taxes from any payments made with respect to Allowed Claims, as appropriate, and shall otherwise comply with Section 346 of the Bankruptcy Code.

6.3. No Implied Duties.

The Liquidating Trustee shall not manage, control, use, sell, dispose, collect or otherwise deal with the Trust Estate or otherwise take any action hereunder except as expressly

provided herein, and no implied duties or obligations whatsoever of the Liquidating Trustee shall be read into this Liquidating Trust Agreement.

SECTION VII CONCERNING THE LIQUIDATING TRUSTEE.

7.1. Acceptance by Liquidating Trustee.

The Liquidating Trustee accepts the Liquidating Trust hereby created for the benefit of the Beneficiaries and agrees to act as Liquidating Trustee of the Liquidating Trust pursuant to the terms of this Liquidating Trust Agreement and the Plan. The Liquidating Trustee shall have and exercise the rights and powers herein granted and shall be charged solely with the performance of the duties herein declared on the part of Liquidating Trustee. The Liquidating Trustee also agrees to receive and disburse all monies actually received by him constituting part of the Trust Estate pursuant to the terms of this Liquidating Trust Agreement and the Plan.

7.2. Discretionary Submission of Questions.

Subject to the provisions of this Section VII, the Liquidating Trustee, in his sole discretion, may, but shall not be required to, submit to the Bankruptcy Court, from time to time, any question or questions with respect to which the Liquidating Trustee may desire to have explicit approval of the Bankruptcy Court for the taking of any specific action proposed to be taken by the Liquidating Trustee with respect to the Trust Estate, or any part thereof, and the administration and distribution of the Trust Estate. The written authorization of the Bankruptcy Court set forth in a Final Order shall constitute approval by the Bankruptcy Court of the proposed action to be taken by the Liquidating Trustee. All costs and expenses incurred by the Liquidating Trust in the exercise of any right, power or authority conferred by this Section 7.2 shall be costs and reasonable expenses of the Trust Estate, payable solely from the Trust Administrative Fund or Other Available Cash.

7.3. Liability of the Liquidating Trustee.

- a. Limitation on Liability. No provision of this Liquidating Trust Agreement shall be construed to impart any liability upon the Liquidating Trustee unless it shall be proven in a court of competent jurisdiction that the Liquidating Trustee's actions or omissions constituted gross negligence or willful misconduct in the exercise of or failure to exercise any right or power vested in the Liquidating Trustee under this Liquidating Trust Agreement.
- b. Reliance on Certificates or Opinions. In the absence of gross negligence or willful misconduct on the part of the Liquidating Trustee, the Liquidating Trust may conclusively rely on the truth of the statements and correctness of the opinions expressed upon any certificates or opinions furnished to the Liquidating Trustee and conforming to the requirements of this Liquidating Trust Agreement.
- c. **Discretion of Liquidating Trustee**. The Liquidating Trustee, within the limitations and restrictions expressed and imposed by this Liquidating Trust Agreement and the Plan, may act freely under all or any of the rights, powers and authority conferred hereby, in all matters concerning the Trust Estate, after forming his best reasonable business judgment based

upon the circumstances of any particular question or situation as to the best course to pursue, without the necessity of obtaining the consent or permission or authorization of the Beneficiaries, the Bankruptcy Court, or of any official or officer; and the rights, powers and authority conferred on the Liquidating Trustee by this Liquidating Trust Agreement are conferred in contemplation of such freedom of reasonable business judgment and action within the limitations and restrictions so expressed and imposed; *provided*, *however*, that the Liquidating Trustee shall not be liable for any error or exercise of judgment, unless it shall be proved in court of competent jurisdiction that such Liquidating Trustee was grossly negligent or acted in a manner that constituted willful misconduct.

7.4. Reliance by Liquidating Trustee.

- a. **Genuineness of Documents**. The Liquidating Trustee may rely and shall be protected in acting upon any resolution, certificate, statement, instrument, opinion, report, notice, request, consent, objection, Order, judgment, decree, or other paper or document reasonably believed by him to be genuine and to have been signed, made, entered or presented by the proper party, parties, official, officials, entity or entities.
- b. Retention of Counsel. The Liquidating Trustee, after consultation with the Liquidating Trust Committee, may retain and consult with legal counsel, independent public accountants and other experts. Legal counsel so retained by the Liquidating Trustee to pursue Causes of Action for the benefit of the Beneficiaries may be retained on a contingent fee basis with such legal counsel's fees not to exceed the percentage of Causes of Action set forth in the definition of Contingency Fee Counsel Fees and Expenses in the Plan and on such other terms and provision acceptable to the Liquidating Trustee after consultation with the Liquidating Trust Committee. The Liquidating Trustee shall not be liable for any action taken or suffered by him or omitted to be taken by him without gross negligence or willful misconduct in reliance on any opinion or certification of such accountants or in accordance with the advice of such counsel or experts, provided that such accountants, counsel and experts were selected and retained with reasonable care. It shall not be a conflict of interest for the Liquidating Trustee to be represented by counsel to the Liquidating Trust Committee.

7.5. Reliance on Liquidating Trustee.

No person dealing with the Liquidating Trustee shall be obligated to see to the application of any monies, securities, or other property paid or delivered to them or to inquire into the expediency or propriety of any transaction or the right, power, or authority of the Liquidating Trustee to enter into or consummate the same upon such terms as the Liquidating Trustee may deem advisable. Persons dealing with the Liquidating Trustee shall look only to the Trust Estate to satisfy any liability incurred by the Liquidating Trustee to such persons in carrying out the terms of this Liquidating Trust Agreement, and, except as otherwise expressly provided herein, the Liquidating Trustee shall have no personal, individual or corporate obligation to satisfy any such liability.

7.6. Indemnification.

- a. Indemnification of Liquidating Trustee and Agents. The Liquidating Trust hereby agrees to indemnify to the full extent of the Trust Estate any person or entity who was or is a party, or is threatened to be made a party to any threatened, pending, or completed action, suit or proceeding, whether civil, criminal, administrative, or investigative by reason of the fact that such person or entity is or was a Liquidating Trustee or an employee, attorney or agent of the Liquidating Trust or Liquidating Trustee, from and against any and all expenses (including attorneys' fees), judgments, fines and amounts paid in settlement actually and reasonably incurred by such person or entity in connection with such action, suit or proceeding, including appeals thereof, if such person or entity acted without gross negligence and willful misconduct in the exercise and performance of any power or duties of such person or entity in accordance with this Liquidating Trust Agreement.
- b. **Payment of Expenses**. Expenses (including reasonable attorneys' fees) incurred in defending any action, suit or proceeding referred to in this Section 7.6 may be paid by the Liquidating Trust from the Trust Administrative Fund or Other Available Cash in advance of the final disposition of such action, suit or proceeding, upon an undertaking by the Liquidating Trustee or an employee or agent of the Liquidating Trust entitled to be indemnified.
- c. **Insurance**. The Liquidating Trust may maintain insurance during its existence and after its termination, at its expense payable from the Trust Administrative Fund or Other Available Cash, to protect itself, the Liquidating Trustee, and officers, employees or agents of the Liquidating Trust of and from any liability, whether or not the Liquidating Trust would have the legal power to directly indemnify the Liquidating Trustee, or officers, employees or agents of the Liquidating Trust against such liability. The terms "Liquidating Trustee," "officers," "employees" or "agents" as used herein, where applicable, include the heirs, successors, executors, administrators, personal representatives, or estates of such persons or entities.
- d. **Bond**. As a condition to serving as Liquidating Trustee hereunder, the Liquidating Trustee and any successor trustee is required to and shall post a bond in favor of the Liquidating Trust in an amount not less than one million dollars (\$1,000,000), which bond shall be in substantially the form as that required by the United States Trustee's Office for trustees serving in bankruptcy cases within the District of Delaware, or such alternative form of financial surety or financial insurance policy as the Liquidating Trust Committee deems appropriate. The costs associated with the procurement and maintenance of the bond shall be funded out of the Trust Administrative Fund.
- e. Costs and Expenses of Liquidating Trustee. The Liquidating Trustee shall pay out of the Trust Administrative Fund all reasonable costs, expenses and obligations incurred by the Liquidating Trustee in carrying out his duties under this Liquidating Trust Agreement or in any manner connected, incidental or related to the administration of the Liquidating Trust, other than fees and expenses incurred from the Effective Date through and including the date all First Lien Facility Claims are paid in full in Cash in connection with (A) the collection of accounts receivable, (B) the pursuit of Causes of Action (including 549)

Avoidance Actions but excluding for the avoidance of doubt Avoidance Actions), and (C) the objection to, and settlement of, Other Secured Claims (including Possessory Lienholder Claims), Administrative Expense Claims, Priority Tax Claims and Other Priority Claims, without application to the Bankruptcy Court, including, without limitation:

- (i) Any reasonable, documented fees and out-of-pocket expenses of attorneys, accountants, investment advisors, expert witnesses, insurance adjusters, professionals or other persons whom the Liquidating Trustee may reasonably deem advisable to employ in connection with the Liquidating Trust;
- (ii) Any taxes, charges and assessments which may be owed by, or levied or assessed against, the Trust Estate or any property held in trust hereunder; and
- (iii) Any reasonable, documented costs, and out-of-pocket expenses incurred by the Liquidating Trust Committee in carrying out its duties under the Plan.

7.7. Compensation of Liquidating Trustee.

The Liquidating Trustee will be compensated from the Trust Administrative Fund according to the terms of Schedule 4.12.

7.8. Resignation and Removal.

- a. **Resignation**. The Liquidating Trustee may resign and be discharged from any future obligations and liabilities hereunder by giving written notice thereof to the Bankruptcy Court and the Liquidating Trust Committee at least thirty (30) days prior to the effective date of such resignation. Subject to the expiration of such notice period, such resignation shall become effective on the day specified in such notice.
- b. **Removal.** The Liquidating Trustee may be removed with cause at any time by Order of the Bankruptcy Court upon motion by any party in interest or in accordance with Section 4.16 hereof. Upon any such removal, such removed Liquidating Trustee shall be entitled to any reimbursement and indemnification set forth in this Liquidating Trust Agreement that remain due and owing to such Liquidating Trustee at the time of such removal.
- c. Appointment of a Successor Trustee. If, at any time, the Liquidating Trustee shall give notice of his intent to resign pursuant to Section 7.8 hereof or be removed or become incapable of acting, counsel to the Liquidating Trustee shall provide notice thereof to the Bankruptcy Court and the Liquidating Trust Committee shall designate a successor Liquidating Trustee to act under this Liquidating Trust Agreement.
- d. Acceptance of Appointment by Successor Liquidating Trustee. Any successor Liquidating Trustee appointed hereunder shall execute an instrument accepting such appointment hereunder and shall deliver counterparts thereof to the Bankruptcy Court.

Thereupon, such successor Liquidating Trustee shall, without any further act, become vested with all of the estates, properties, rights, powers, trusts and duties of his predecessor in the Liquidating Trust hereunder with like effect as if originally named herein.

e. **Trust Continuance**. The death, resignation, incompetency or removal of the Liquidating Trustee shall operate neither to terminate the Liquidating Trust created by this Liquidating Trust Agreement nor to revoke any existing agency created pursuant to the terms of this Liquidating Trust Agreement or invalidate any action theretofore taken by such Liquidating Trustee. In the event of the resignation or removal of the Liquidating Trustee, such Liquidating Trustee shall (i) promptly execute and deliver any such documents, instruments, and other writing as may be necessary to effect the termination of such Liquidating Trustee's capacity under this Liquidating Trust Agreement and the conveyance of the Trust Estate then held by such Liquidating Trustee to the temporary or successor trustee; (ii) deliver to the temporary or successor trustee all documents, instruments, records, and other writings relating to the Liquidating Trust or Trust Estate as may be in the possession of such Liquidating Trustee; and (iii) otherwise assist and cooperate in effecting the transfer and assumption of his obligations and functions by the temporary or successor trustee.

SECTION VIII SUPPLEMENTS AND AMENDMENTS TO THIS LIQUIDATING TRUST AGREEMENT.

8.1. Supplements and Amendments.

Subject to the provisions of Sections 2, 8.2 and 8.3 hereof, at any time and from time to time, and subject to approval by the Bankruptcy Court if sought by the Liquidating Trustee pursuant to Section 7.2 of this Liquidating Trust Agreement, the Liquidating Trust Committee may direct the Liquidating Trustee to execute a supplement or amendment hereto for the purpose of adding provisions to, or changing or eliminating provisions of, this Liquidating Trust Agreement, or amendments thereto, as specified in such vote or consent; provided, however, that no such supplement or amendment shall (a) require any Beneficiary to furnish or advance funds to the Liquidating Trustee or shall entail any additional personal liability or the surrender of any individual right on the part of any Beneficiary except with the written consent of such Beneficiary, (b) without the prior written consent of the holder of an Allowed Administrative Expense Claim, Allowed Priority Tax Claim, Allowed Other Priority Claim, Allowed Other Secured Claim, Allowed First Lien Facility Claim, or Allowed Second Lien Facility Claim, change or modify the Liquidating Trustee's obligations with respect to such Claim, including, without limitation, provisions for payment of such Claim, or (c) without the consent of those holders of General Unsecured Claims holding not less than seventy-five percent (in aggregate dollar amount) of the General Unsecured Claims, change or modify any other provisions for distribution of the Trust Estate. In no event shall this Liquidating Trust Agreement be amended so as to change the purposes of the Liquidating Trust as set forth in Section 2 or the treatment of Claims under the Plan and this Liquidating Trust Agreement.

8.2. Declining to Execute Documents.

If, in the reasonable opinion of the Liquidating Trustee, any document required to be executed pursuant to the terms of Section 8.1 hereof materially and adversely affects any immunity or indemnity in favor of the Liquidating Trustee under this Liquidating Trust Agreement, the Liquidating Trustee may in his discretion decline to execute such document.

8.3. Notice of Form of Supplements and Amendments Requiring Vote or Consent.

A copy of each amendment or supplement (or a fair summary thereof) shall be furnished to the Beneficiaries promptly after the execution thereof, except that with respect to any proposed amendment or supplement for which the consent of certain Beneficiaries is required pursuant to Section 8.1 hereof, the form of such proposed supplement or amendment (or a fair summary thereof) shall be furnished to the applicable Beneficiaries prior to the Liquidating Trustee's seeking the approval thereof by vote or consent of such necessary parties.

8.4. Notice and Effect of Executed Amendment.

Upon the execution of any declaration of amendment or supplement, this Liquidating Trust Agreement shall be deemed to be modified and amended in accordance therewith and the respective rights, limitations of rights, obligations, duties and immunities under this Liquidating Trust Agreement of the Liquidating Trustee and the Beneficiaries shall thereafter be determined, exercised and enforced hereunder subject in all respects to such modification and amendment, and all the terms and conditions of any such amendment or supplement shall be thereby deemed to be part of the terms and conditions of this Liquidating Trust Agreement for any and all purposes.

SECTION IX MISCELLANEOUS.

9.1. Title to Trust Estate.

No Beneficiary or any other party other than the Liquidating Trust shall have title to any part of the Trust Estate.

9.2. Sales of Trust Assets.

Any sale or other conveyance of any Trust Assets, or part thereof, by the Liquidating Trustee made in accordance with the terms of this Liquidating Trust Agreement shall bind the Beneficiaries and shall be effective to transfer or convey all right, title and interest of the Liquidating Trustee and the Beneficiaries in and to such Trust Asset.

9.3. Notices.

Unless otherwise expressly specified or permitted by the terms of the Plan or this Liquidating Trust Agreement, all notices shall be in writing and delivered by registered or certified mail, return receipt requested, or by a hand or facsimile transmission (and confirmed by mail), in any such case addressed as follows:

If to the Liquidating Trustee:

Kip Horton RPA Advisors, LLC 45 Eisenhower Drive Paramus, New Jersey 07652

Phone: (201) 527-6653

Email: khorton@rpaadvisors.com

and if to any Beneficiary, addressed to its latest mailing address reflected on the official claims register maintained by BMC Group, Inc..

9.4. Severability.

Any provision of this Liquidating Trust Agreement which is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions thereof, and any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.

9.5. Counterparts.

This Liquidating Trust Agreement may be executed in multiple counterparts, each of which shall constitute an original, but all of which together shall constitute one and the same instrument.

9.6. Binding Agreement.

All covenants and agreements contained herein shall be binding upon, and inure to the benefit of, the Liquidating Trustee and his respective successors and assigns and any successor Liquidating Trustee provided for in Section VII, his respective successors and assigns, and the Beneficiaries, and their respective personal representatives, successors and assigns. Any request, notice, direction, consent, waiver or other instrument or action by any party hereto or any Beneficiary shall bind their respective heirs, personal representatives, successors and assigns.

9.7. No Personal Liability of Beneficiaries.

The Beneficiaries will not incur any personal liability through their ownership or possession of their Beneficial Interests, except for taxes imposed on the Beneficiaries pursuant to applicable provisions of federal, state or local law with respect to the receipt of such Beneficial Interests or distributions from or transactions of the Liquidating Trust and other charges specified herein. Liabilities of the Liquidating Trust are to be satisfied in all events (including the exhaustion of the Trust Estate) exclusively from the Trust Estate and such liabilities are not to attach to or be paid from any amounts distributed to the Beneficiaries, regardless of the time at which such distribution took place, or from the assets of the Beneficiaries.

9.8. Headings.

The headings of the various Sections herein are for convenience of reference only and shall not define or limit any of the terms or provisions hereof.

9.9. Construction.

Except where the context otherwise requires, words importing the masculine gender shall include the feminine and the neuter, if appropriate; words importing the singular number shall include the plural number and vice versa; and words importing persons shall include partnerships, associations, and corporations.

9.10. Governing Law.

This Liquidating Trust Agreement, including all matters of construction, validity and performance hereof, shall in all respects be governed by, and construed and interpreted in accordance with, the internal laws of the State of Delaware.

9.11. Construction with the Plan.

The Plan is hereby incorporated fully by reference and is made a part hereof for all purposes. In the event of any inconsistency or conflict between the terms, conditions, definitions and provisions of this Liquidating Trust Agreement and the terms, conditions, definitions and provisions of the Plan, the terms, conditions, definitions and provisions of the Plan shall control.

9.12. Subject to Bankruptcy Court's Jurisdiction.

The Bankruptcy Court shall retain jurisdiction over this Liquidating Trust Agreement, the Liquidating Trust, the Trust Estate, the Liquidating Trustee and the Debtors to issue any and all Orders and to take other actions necessary to the implementation of this Liquidating Trust Agreement, such jurisdiction to include, without limitation, the jurisdiction contemplated by Section 1142 of the Bankruptcy Code.

9.13. Intention of the Parties.

The Debtors, the Beneficiaries and the Liquidating Trustee hereby express their intent to create and maintain the Liquidating Trust, except with respect to the Disputed Claims Reserve, as a liquidating trust for Federal income tax purposes in accordance with Treasury Regulation §301.7701-4(d) and as one or more "grantor trusts" subject to the provisions of Subchapter J, Subpart E of the IRC, and the Liquidating Trustee further represents that the Liquidating Trust shall not: (a) receive transfers of listed stocks or securities, any readily-marketable assets (other than those constituting the trust estate), or any assets of a going business; or (b) receive and will not retain Cash or cash equivalents in excess of a reasonable amount to meet claims and contingent liabilities, determined in the reasonable discretion of the Liquidating Trustee in accordance with the provisions of Sections IV and V hereof.

Dated:	
Graceway Pharma Holding Corp.	Graceway Holdings, LLC
By: Name: Title:	By: Name: Title:
Graceway Pharmaceuticals, LLC	Chester Valley Holdings, LLC
By: Name: Title:	By:
Chester Valley Pharmaceuticals, LLC	Graceway Canada Holdings, Inc.
By:	By:Name:Title:
Graceway International, Inc.	Liquidating Trustee
By: Name: Title:	By: Name: Kip Horton

Schedule 4.12

Compensation of Liquidating Trustee

The Liquidating Trustee will receive a fee of \$640 per hour.

Exhibit 3

Assumption Schedule

[to come]

	Contract Counterparty	Type of Contract	Contract Date	Notice Address	Brief Summary	Cure Amount
1.						
2.	· · · · · · · · · · · · · · · · · · ·					

EXHIBIT B

Claims Objections

EXHIBIT B

NAMH OF CLAIMANT	DATEFILED	CASE NUMBER	CLAIM#	TOTAL CLAIM DOLLARS	OBJECTION DOCKET NO.
1 ADVANSTAR COMMUNICATIONS INC TODD ROURKE 131 W 1ST ST DUIJTH, MN 55802- 2065	10/26/2011	11-13036 (PJW)	9	\$42,491.76	535
2 AMERICAN MEDICAL ASSOCIATION RICHARD ARMSTRONG, MGR, CREDIT & COLLECTIONS 515 N STATE ST CHICAGO, IL 60654	11/21/2011	11-13036 (PJW)	56	\$12,173.70	467
4 BERMAN, BRIAN 1541 WEST 22ND STREET MIAMI BEACH, FL 33140	11/14/2011	11-13036 (PJW)	22	\$5,972.36	535
5 BHATIA, NEAL MD 400 N BROADWAY #801 MILWAUKEE, WI 53202	11/18/2011	11-13036 (PJW)	45	\$11,072.00	535
6 BRANCACCIO, RONALD MD 7901 4TH AVENUE BROOKLYN, NY 11209	11/14/2011	11-13036 (PJW)	21	\$1,200.00	535
7 BURLINGTON DRUG COMPANY ATTN: PRESIDENT OR GENERAL COUNSEL 91 CATAMOUNT DR MILTON, VT 05468	11/28/2011	11-13036 (PJW)	61	\$2,431.17	525
8 CAREY INTERNATIONAL 5300 SPECTRUM DR, STE D FREDERICK, MD 21703	11/18/2011	11-13036 (PJW)	47	\$628.71	535
9 CHAPMAN, M SHANE 36 OCCOM RIDGE HANOVER, NH 03755	12/19/2011	11-13036 (PJW)	107	\$5,600.00	535
10 CHOW, MD, MAY J 333 DIXIE HWY CHICAGO HEIGHTS, IL 60411-1748	11/17/2011	11-13036 (PJW)	36	Undetermined	535

* - Indicates claim contains unliquidated and/or undetermined amounts

Page 1 of 6

EXHIBIT B

NAME OF CLAIMANT	DATE FILED	CASE NUMBER	CLAIM#	TOTAL CLAIM DOLLARS	OBJECTION DOCKET NO.
11 CVS CAREMARK CORPORATION GEOFFREY S GOODMAN, ESQ FOLEY & LARDNER LIP 321 NORTH CLARK ST., SUITE 2800 CHICAGO, IL 60654	12/30/2011	11-13036 (PJW)	192	\$22,658,639.01*	538
12 CVS CAREMARK PART D SERVICES LLC GEOFFREY S GOODMAN, ESQ FOLEY & LARDNER LLP 321 NORTH CLARK ST., SUITE 2800 CHICAGO, IL 60654	12/30/2011	11-13036 (PJW)	189	\$61,699.44*	535
13 ERNEST GRUBB 205 FAIRACRES DR BRISTOL, TN 37620	11/17/2011	11-13036 (PJW)	38	Undetermined	535
14 FISHER CLINICAL SERVICES FISHER SCIENTIFIC CO, ATTN: GARY BARNES 300 INDUSTRY DR PITTSBURGH, PA 15275	11/10/2011	11-13036 (PJW)	43	\$3,785.00	467
15 FRANKLIN PHARMA SERVICES, LLC ATTN: PRESIDENT OR GENERAL COUNSEL PO BOX 415191 BOSTON, MA 02241-5191	12/05/2011	11-13036 (PJW)	81	\$12,556.26	467
16 FRANKLIN PHARMA SERVICES, LLC C/O STEPHANIE P UNION KEGLER BROWN HILL & RITTER 65 EAST STATE ST, STE 1800 COLUMBUS, OH 43215	12/19/2011	11-13036 (PJW)	. 113	\$19,086.53	467
17 GASPARI, ANTHONY (TONY) MD 13027 BEAVER DAM ROAD COCKEYSVILLE, MD 21030	12/05/2011	11-13036 (PJW)	77	\$361.39	535
18 GASPARI, MD, ANTHONY 13027 BEAVER DAM ROAD COCKEYSVILLE, MD 21030	12/05/2011	11-13036 (PJW)	78	\$636.75	535

* - Indicates claim contains unliquidated and/or undetermined amounts

Page 2 of 6

EXHIBIT B

NAME OF CLAIMANT	DATE FILED	CASE NUMBER	CLAIM#	TOTAL CLAIM DOLLARS	OBJECTION DOCKET NO.
19 GLASHOFER, MARC MD 604 EAST PARK AVE LONG BEACH, NY 11561	11/28/2011	11-13036 (PJW)	29	\$1,000.00	535
20 GLASHOFER, MARC MD 604 EAST PARK AVE LONG BEACH, NY 11561	01/03/2012	11-13036 (PJW)	194	\$1,000.00	525, 535
21 GORELICK, NP , JOSEPH A 5339 PROSPECT RD #127 SAN JOSE, CA 95129	11/14/2011	11-13036 (PJW)	24	\$2,988.50	535
22 GSW ADVERTISING, LLC C/O KEGLER BROWN HILL & RITTER ATTN: STEPHANIE P UNION 65 E STATE ST, STE 1800 COLUMBUS, OH 43215	11/21/2011	11-13036 (PJW)	58	\$37,021.63	467
23 GSW ADVERTISING, LLC STEPHANIE P UNION ESQ KEGLER BROWN HILL & RITTER 65 EAST STATE ST, STE 1800 COLUMBUS, OH 43215	12/19/2011	11-13036 (PJW)	114	\$49,109.13	467
24 HIMMELSBACH NP, ALAN J 501 RANDOLPH DR LITITZ, PA 17543	12/22/2011	11-13036 (PJW)	141	\$750.00	535
25 HOGAN LOVELLS US LLP ATTN: EDWARD C DOLAN 555 THIRTEENTH STREET, NW WASHINGTON, DC 20004-1109	12/29/2011	11-13037 (PJW)	178	\$2,214.07	535
26 HOGAN LOVELLS US LLP ATTN: EDWARD C DOLAN 555 THIRTEENTH STREET, NW WASHINGTON, DC 20004-1109	12/29/2011	11-13038 (PJW)	179	\$2,214.07	535
27 HOGAN LOVELLS US LLP ATTN: EDWARD C DOLAN 555 THIRTEENTH STREET, NW WASHINGTON, DC 20004-1109	12/29/2011	11-13039 (PJW)	180	\$2,214.07	535

^{* -} Indicates claim contains unliquidated and/or undetermined amounts

Page 3 of 6

EXHIBIT B

NAME OF CLAIMANT	DATE FILED	CASE NUMBER	CLAIM#	TOTAL CLAIM DOLLARS	OBJECTION DOCKET NO.
HOGAN LOVELLS US LLP ATTN: EDWARD C DOLAN 555 THIRTEENTH STREET, NW WASHINGTON, DC 20004-1109	12/29/2011	11-13041 (PJW)	181	\$2,214.07	535
29 HOGAN LOVELLS US LLP ATTN: EDWARD C DOLAN 555 THIRTEENTH STREET, NW WASHINGTON. DC 20004-1109	12/29/2011	11-13042 (PJW)	182	\$2,214.07	535
30 HOGAN LOVELLS US LLP ATTN: EDWARD C DOLAN 555 THIRTEENTH STREET, NW WASHINGTON, DC 20004-1109	12/29/2011	11-13043 (PJW)	183	\$2,214.07	535
31 HUMANA INC C/O GREGORY O KADEN GOULSTON & STORRS PC 400 ATLANTIC AVE BOSTON, MA 02110-3333	12/22/2011	11-13036 (PJW)	143	\$59,504.15*	525
32 IMBERG, RANDALL 611 A FISH ST. WAUNAKEE, WI 53597	01/05/2012	11-13036 (PJW)	202	Undetermined	535
33 INTERNAL REVENUE SERVICE DEPARTMENT OF THE TREASURY PO BOX 7346 PHILADELPHIA, PA 19101-7346	10/21/2011	11-13036 (PJW)	ဇ	\$448,982.84	467
34 MARCUS, LINDA 271 GODWIN AVENUE WYCKOFF, NJ 07481	11/21/2011	11-13036 (PJW)	57	\$3,124.00	535
35 MCHUGH, JOHN MD 700 S TUSTIN ORANGE, CA 92866	11/14/2011	11-13036 (PJW)	18	\$1,500.00	525
36 MERCER ISLAND DERMATOLOGY ALISON HUGHES MD, PHD 2835 82ND AVE SE, SUITE 210 MERCER ISLAND, WA 98040	12/23/2011	11-13036 (PJW)	160	\$1,000.00	535

^{* -} Indicates claim contains unliquidated and/or undetermined amounts

Page 4 of 6

EXHIBIT B

	NAME OF CLAIMANT	DATE FILED	CASE NUMBER	CLAIM#	TOTAL CLAIM DOLLARS	OBJECTION DOCKET NO.
<i>8</i> .	37 MIAMI-LUKEN INC ATTN: PRESIDENT OR GENERAL COUNSEL 265 S PIONEER BLVD SPRINGBORO, OH 45066	12/12/2011	11-13036 (PJW)	95	Undetermined	535
ਲ	38 MI DEPT OF COMM HEALTH MED SVCS ADMIN ATTN: DIRECTOR OR GENERAL COUNSEL DIR BUR OF MEDICAID PGM OPS & QUAL ASSU 400 S PINE STREET LANSING, MI 48933	11/21/2011	11-13036 (PJW)	ις.	\$8,979.67	525
39	1	12/22/2011	11-13036 (PJW)	142	\$1,500.00	535
<u>4</u>	40 MONTANA DPHHS PO BOX 202951 HELENA, MT 59620-2951	12/22/2011	11-13036 (PJW)	140	\$1,354.53	525
41		12/06/2011	11-13036 (PJW)	82	\$12,188.58	535
42	2 NYBERG, ELIAS 25 BROOKMILL RD STOW, MA 01775	11/21/2011	11-13036 (PJW)	51	\$2,900.00	525
43	3 PENNSYLVANIA DEPARTMENT OF REVENUE BANKRUPTCY DIVISION PO BOX 280946 HARRISBURG, PA 17128-0946	12/13/2011	11-13036 (PJW)	46	\$1,972.66*	467
44		12/13/2011	11-13039 (PJW)	66	\$24,532.66*	525

* - Indicates claim contains unliquidated and/or undetermined amounts

Page 5 of 6

EXHIBIT B

NAME OF CLAIMANT	DATE FILED	CASE NUMBER	CLAIM#	TOTAL CLAIM DOLLARS	OBJECTION DOCKET NO.
45 SADDLER'S PHOTOGRAPHY ATTN: PRESIDENT OR GENERAL COUNSEL 1724 EDGEMONT AVE. BRISTOL, TN 37620	11/14/2011	11-13036 (PJW)	20	Undetermined	535
46 SH MEDIA, LLC ATTN: PRESIDENT OR GENERAL COUNSEL 11 ROSE MEADOW WAY AQUINNAH, MA 02535	11/17/2011	11-13036 (PJW)	37	\$906.25	535
47 STRASSWIMMER, JOHN MD PHD 2605 WEST ATLANTIC AVENUE, SUITE D-204 DELRAY BEACH, FL 33445	11/17/2011	11-13036 (PJW)	35	\$6,000.00	. 535
48 SUPERVALU INC LEGAL DEPARTMENT 11840 VALLEY VIEW ROAD EDEN PRAIRIE, MN 55344-3691	12/29/2011	11-13036 (PJW)	165	\$142,371.12*	525
49 THE ANGIOGENESIS FOUNDATION VINCENT LI ONE BROADWAY, 14TH FLR CAMBRIDGE, MA 02242	11/14/2011	11-13036 (PJW)	31	\$514.21	535
50 VINCENT W LI, LLC 57 FAIRVIEW ST HOLLISTON, MA 01746	11/14/2011	11-13036 (PJW)	32	\$2,000.00	535
51 WEST VIRGINIA DEPT OF HEALTH & HUMAN RESOURCES BUREAU FOR MEDICAL SERVICES CHARLENE A VAUGHAN, DEPUTY ATTORNEY GENERAL 812 QUARRIER ST, 2ND FL CHARLESTON, WY 25301	10/25/2011	11-13036 (PJW)	ហ	\$49,530.20	467

*- Indicates claim contains unliquidated and/or undetermined amounts

EXHIBIT C

Curriculum Vitae of Kip Horton

Kip Horton

Member

Industry Experience

Infrastructure and Energy Real Estate and Construction Retail and Wholesale Distribution

Selected Public Cases

Boston Generating
Enron
Hechingers
Heilig-Meyers
NEGT
PG&E National Energy Group
SemGroup
South Bay Expressway
US Industries/Jacuzzi

Experience

and has consulted on strategic planning, business plan analysis, cash management, enterprise valuation, cost reduction, restructuring, divestiture analysis, mergers and acquisitions and risk management. Within the merchant power arena, Mr. Horton has extensive experience in risk Kip Horton specializes in financial and turnaround advisory services for both companies and creditors, nanagement, trading-related activities and cash management.

Mr. Horton currently serves as the Chairman of the Board of South Bay Expressway, a private toll Mr. Horton previously sat on the Board of Directors of EBG Holdings, LLC, a portfolio of merchant generation plants in and around Boston and subsequently, on the Board of US PowerGen Company. road in Southern California. Prior to founding RPA, Mr. Horton was an Executive Director at Capstone Advisory Group, a restructuring and turnaround consulting firm. Previously, he was with Prudential Securities.

Education and Affiliations

Mr. Horton holds a B.S. in Business Administration from the University of Richmond with a concentration in Finance. He is also a member of the Turnaround Management Association.



EXHIBIT D

Liquidation Analysis

LIQUIDATION ANALYSIS

Introduction

Often called the "best interests" test, section 1129(a)(7) of the Bankruptcy Code requires that the Bankruptcy Court find, as a condition to Confirmation, that each holder of a Claim or Interest in each Impaired Class: (i) has accepted the Plan; or (ii) will receive or retain under the Plan property of a value, as of the Effective Date, that is not less than the amount that such Person would receive if the Debtors were liquidated under chapter 7 of the Bankruptcy Code. To make these findings, the Bankruptcy Court must: (1) estimate the cash proceeds (the "Liquidation Proceeds") that a chapter 7 trustee would generate if each Debtor's Chapter 11 Case were converted to a chapter 7 case on the Effective Date and the assets of such Debtor's Estate were liquidated; (2) determine the distribution (the "Liquidation Distribution") that each non-accepting holder of a Claim or Interest would receive from the Liquidation Proceeds under the priority scheme dictated in chapter 7; and (3) compare each holder's Liquidation Distribution to the distribution under the Plan that such holder would receive if the Plan were confirmed and consummated.

Accordingly, asset values discussed herein may be different than amounts referred to in the Plan. The Liquidation Analysis is based upon certain assumptions discussed herein and in the Disclosure Statement.

Significant Assumptions

Hypothetical recoveries to stakeholders of the Debtors in chapter 7 were determined through multiple steps, as set forth below. The basis of the Liquidation Analysis is the Debtors' projected cash balance and assets as of April 30, 2012 and assumes that the Debtors would commence a chapter 7 liquidation on or about May 1, 2012 (the "Conversion Date") under the supervision of a court appointed chapter 7 trustee. The Liquidation Analysis assumes that certain events would take place prior to the Conversion Date as more fully described in footnote (a) below. The Liquidation Analysis reflects the wind-down and liquidation of substantially all of the Debtors' remaining assets; the reconciliation, settlement or prosecution of remaining litigation; and distribution of available proceeds to holders of Allowed Claims during a twelve month period after the Conversion Date (the "Wind-Down Period").

The Debtors believe that a chapter 7 trustee, if appointed, would elect to substantively consolidate the Debtors for the reasons set forth in the Plan and Disclosure Statement. As such, the Liquidation Analysis assumes that all Debtors will be substantively consolidated.

The statements in the Liquidation Analysis, including estimates of Allowed Claims, were prepared solely to assist the Bankruptcy Court in making the findings required under section 1129(a)(7) and they may not be used or relied upon for any other purpose.

THE DEBTORS BELIEVE THAT ANY ANALYSIS OF A HYPOTHETICAL LIQUIDATION IS NECESSARILY SPECULATIVE. THERE ARE A NUMBER OF ESTIMATES AND ASSUMPTIONS UNDERLYING THE LIQUIDATION ANALYSIS THAT ARE INHERENTLY SUBJECT TO SIGNIFICANT ECONOMIC, COMPETITIVE, AND OPERATIONAL UNCERTAINTIES AND CONTINGENCIES BEYOND THE CONTROL OF THE DEBTORS OR A CHAPTER 7 TRUSTEE. NEITHER THE LIQUIDATION ANALYSIS, NOR THE FINANCIAL INFORMATION ON WHICH IT IS BASED, HAS BEEN EXAMINED OR REVIEWED BY INDEPENDENT ACCOUNTANTS IN ACCORDANCE WITH STANDARDS PROMULGATED BY THE AMERICAN INSTITUTE OF CERTIFIED PUBLIC

ACCOUNTANTS. THERE CAN BE NO ASSURANCE THAT ACTUAL RESULTS WILL NOT VARY MATERIALLY FROM THE HYPOTHETICAL RESULTS PRESENTED IN THE LIQUIDATION ANALYSIS.

Estimate of Proceeds from Chapter 11 Liquidation

Estimates were made of cash proceeds that might be received from the liquidation of the Debtors' assets after consideration of the effects that a chapter 7 liquidation would have on the ultimate proceeds available for distribution, including (i) the costs and expenses of a liquidation under chapter 7 arising from fees payable to the trustee and professional advisors to such trustee; and, (ii) the potential erosion in value of assets in a chapter 7 case as a result of retaining new professionals and replacing existing professionals with particular knowledge of various matters.

The Debtors' three largest former customers (Cardinal Health, McKesson, AmerisourceBergen) owe the vast majority of the remaining outstanding amounts, which are all now past due. The Debtors have been actively engaged in efforts to collect all past due balances and continue processing credits and reviewing collectability of remaining receivables. Furthermore, customers may attempt to set-off certain pre and postpetition claims against receivables, which would have an impact on net proceeds collected by the trustee in a chapter 7 liquidation. The Debtors believe an adequate reserve has been provided for such costs as administrative expenses as more fully described below.

Other assets include recoveries that might be realized by the chapter 7 trustee's potential pursuit of any preference actions, as these recoveries could be highly speculative in light of, among other things, the various defenses that would likely be asserted.

Estimate of Costs

Proceeds from a chapter 7 liquidation would be reduced by administrative costs incurred during the Wind-Down Period for the recovery of assets and the reconciliation of claims. These costs include professional (including attorneys, and other tax and financial advisors) and trustee fees, insurance expenses, subcontract labor, document retention and other expenses. Additionally these costs include administrative claims incurred prior to conversion that were unpaid at the time of the conversion, for which a reserve has been included in the chapter 7 liquidation analysis. The largest portion of the reserve covers the category of claims including returns, rebates, chargebacks and other customer-related fees that are postpetition administrative claims. As of today, many of these claims are contingent, unliquidated and thus the final amount paid by the trustee in a chapter 7 liquidation is highly speculative. Furthermore, historically the Debtors issued credit memos to customers for certain such claims and customers may attempt to assert rights of set-off against receivables owed to the Debtors for such claims.

After consideration of the effects that a chapter 7 liquidation would have on the ultimate proceeds available for distribution to creditors, the Debtors have determined, as summarized in the following charts and the "Best Interest Test" section of the Disclosure Statement, that the Plan will provide creditors with a recovery that is greater than creditors would receive pursuant to a liquidation of the Debtors' assets under chapter 7.

The following liquidation analysis should be reviewed with the accompanying notes.

Graceway Pharmaceuticals, LLC Debtors' Liquidation Analysis Exhibit (\$ in 000's)

		Cha	apter 11 Plan	of Liquidation	Chapter 7	Conversion
	Notes	Bas	e	Favorable	Base	Favorable
Liquidation Proceeds:						
Cash & cash equivalents	(a)		\$ 32,500	\$ 40,02	\$ 32,500	\$ 40,022
Accounts & royalties receivable	(b)		\$ 12,668	\$ 14,78	\$ \$ 12,668	\$ 14,788
Other assets	(c)	_	\$ 13,611	\$ 16,66	\$ 13,611	\$ 16,664
Gross Liquidation Proceeds		_	\$ 58,779	\$ 71,47	\$ 58,779	\$ 71,475
Unclassified Claims:						
Administrative Expenses & Statutory Fees	(d)		\$ (19,200)	\$ (11,48	3) \$ (19,200)	\$ (11,488)
Professional Claims	(e)		\$ (2,189)	\$ (2,18)		\$ (2,189)
Priority Tax Claims	(f)		\$ (694)	\$ (2,18)		\$ (2,189)
Liquidating Trust / Chapter 7 Trust Expenses	(r) (g)		\$ (1,950)	\$ (1,95		
Equidating Trust / Chapter / Trust Expenses	(8)		\$ (1,330)	\$ (1 ,55	0) \$ (2,613)	\$ (2,768)
Net Distributable Value for Classified Claims		=	\$ 34,746	\$ 55,15	\$ 34,083	\$ 54,336
		Estimated		Estimated	Estimated	Estimated
		Claim	Net Distrib.	Claim Net Distri		Claim Net Distrib.
		Amount	Value	Amount Value	Amount Value	Amount Value
Estimated Classified Claims:						
Class 1: Other Priority		\$ 50	\$ 50	\$ - \$	- \$ 50 \$ 50	\$ - \$ -
Class 2: First Lien Facility	(h)	\$ 37,926	\$ 23,546	\$ 44,025 \$ 44,02		\$ 44,025 \$ 44,025
Class 3: Second Lien Facility - Secured	(i)		\$ 6,369	\$ 4,852 \$ 4,85		\$ 4,852 \$ 4,852
Class 4: Other Secured	(i)		\$ 350	\$ - \$	- \$ 350 \$ 350	\$ - \$ -
Class 5: General Unsecured	(k)	\$ 477,576	\$ 4,431	\$ 441,457 \$ 6,27		\$ 441,457 \$ 5,460
Class 6: Intercompany	(1)		\$ -	N/A \$	- N/A \$ -	N/A \$ -
Class 7: Old Equity	(m)	N/A	\$ -	N/A \$	- N/A \$ -	N/A \$ -
Class 8: Old Equity Rights	(n)	,	\$ -	N/A \$	- N/A \$ -	N/A \$ -
1,	.,		\$ 34,746	\$ 55,15		\$ 54,336
		-				
Estimated % Recovery - Classified Claims:						
Class 1: Other Priority			100.0%	100,0		
Class 2: First Lien Facility			62.1%	100.0	% 62.1%	100.0%
Class 3: Second Lien Facility - Secured			100.0%	100.0	% 100.0%	100.0%
Class 4: Other Secured			100.0%	100.0	% 100,0%	100.0%
Class 5: General Unsecured			0.9%	1.4	% 0.8%	1.2%
Class 6: Intercompany			0.0%	0.0	0.0%	0.0%
Class 7: Old Equity			0.0%	0.0	0,0%	0.0%
Class 8: Old Equity Rights			0.0%	0.0	0.0%	. 0.0%

See accompanying notes, which are an integral part of the Chapter 7 liquidation analysis

Notes to Liquidation Analysis

- (a) Cash & cash equivalents represent estimated cash balances available on the Conversion Date after consideration of the costs to administer the Chapter 11 Cases through the Conversion Date and any payments on account of valid postpetition claims or Court-authorized prepetition claims.
- (b) Amount represents the estimated accounts receivable and royalties receivable on the Conversion Date. The book value on November 30, 2011 was approximately \$21.2 million, of which the Debtors had collected approximately \$3.8 million as of January 13, 2012. The remaining values include a discount to remaining outstanding balances. The chapter 7 liquidation analysis assumes the recovery of accounts receivable and royalties receivable are the same as under the Plan; however, it is possible that value could be impaired under a chapter 7 liquidation.
- (c) "Other assets" include estimated recoveries from the Debtors' affiliated foreign subsidiary Graceway Canada Company, proceeds from sale of personal and real property, potential tax refunds, miscellaneous deposits and escrowed amounts, and Avoidance Action and 549 Avoidance Action Proceeds, net of fees to pursue such claims. Additionally, "Other assets" includes potential proceeds from the Nycomed Award Amount. Whether the Debtors will be entitled to any recovery on account of the Nycomed Award Amount is uncertain and speculative to value. Therefore, no value has been ascribed under the chapter 7 or Plan scenarios; however, the Debtors expect there would be no difference in the ultimate amount of proceeds under either scenario.
- (d) Administrative expenses include estimated allowed 503(b)(9) claims, postpetition accounts payable, employee-related claim payments, and other claims that were not yet paid as of the Conversion Date. Also included are estimated unpaid postpetition chargeback, returns, channel management fees, and rebate claims. Also included are Statutory Fees which will become due and payable pursuant to 28 U.S.C. § 1930.
- (e) Includes legal, tax, financial advisory and other professional fees that were incurred prior to the Conversion Date but not yet paid.
- (f) Primarily reflects unpaid tax claims as of the Conversion Date.
- (g) Liquidating Trust expenses reflect the Debtors' estimated legal, tax, financial advisory and claims management fees for the Wind-Down Period. The chapter 7 scenario has been modified to include the estimated expenses of a chapter 7 trustee in lieu of the trustee compensation proposed under the Plan. The expense of the professionals that are retained by the chapter 7 trustee is assumed to be the same as the estimate under the Plan. Pursuant to section 326(a) of the Bankruptcy Code, the expenses of the chapter 7 trustee have been estimated based on a sliding scale to be approximately 3.1% of the gross Liquidation Proceeds (excluding cash on hand on the Conversion Date). Under both scenarios it is assumed that costs of professionals, support staff, and other costs (exclusive of Liquidating Trustee or chapter 7 trustee) total \$1.8 million and are borne 50% by the holders of Class 2 First Lien Facility Claims and 50% by the holders of Class 5 General Unsecured Claims.

- (h) Estimated unpaid claims under the First Lien Credit Agreement as of Conversion Date. Base Case includes principal (\$430.3 million) plus prepetition accrued interest (\$2.6 million) less payments made to First Lien Facility Agent postpetition through January 31, 2012 (\$395.1 million), including sale proceeds on December 2, 2011 and distributions made in accordance with the DIP, Sale and other Orders. Base Case excludes potential claims for an undrawn Letter of Credit (\$350k), professional fees and expenses, postpetition interest, funding certain of the Liquidating Trustee's fees and expenses, and other amounts that may be owed to the First Lien Claimholders. Favorable Case includes estimates for all such amounts, plus postpetition interest, up to an aggregate amount of proceeds available for distribution to the First Lien Claimholders. Based on these assumptions, activity to date results in a claim today ranging from \$37.9 - \$44.0 million used for the purposes of this analysis; however, the aggregate amount of proceeds available for distribution to the First Lien Claimholders may be higher than the Debtors currently anticipate and additional postpetition interest and fees may be owed on account of the First Lien Facility Claims, leading to a higher Range of Estimated Allowed Amounts for First Lien Facility Claims and a recovery in excess of the Favorable Case amounts. The estimated % recovery for the First Lien Claimholders on account of the First Lien Facility Claim (as calculated based on the assumptions above and including amounts already distributed on account of the First Lien Facility Claim) is 96.7% and 100.0% under the Base Case and Favorable Case, respectively.
- (i) Assumes Second Lien Facility Claims entitled to secured priority recovery from 65% of proceeds of Canadian Distribution Amount (less the amount paid to First Lien Claimholders from such proceeds under the Committee Settlement). The remaining Second Lien Facility Deficiency Claim is included in Class 5 General Unsecured Claims.
- (j) Represents the estimated range of creditor claims, distributable value and recoveries for Class 4 Other Secured Claims pursuant to the Plan. Estimated claim amount represents only the amount estimated to require cash payments. Additional allowed claims for this class are expected to be 100% satisfied via receipt of collateral securing such claim.
- (k) Represents the estimated range of creditor claims, distributable value and recoveries for Class 5 General Unsecured Claims pursuant to the Plan, including the Second Lien Facility Deficiency Claim.
- (I) Represents the estimated range of creditor claims, distributable value and recoveries for Class 6 Intercompany Claims pursuant to the Plan.
- (m) Represents the estimated range of creditor claims, distributable value and recoveries for Class 7 Old Equity Interests pursuant to the Plan.
- (n) Represents the estimated range of creditor claims, distributable value and recoveries for Class 8 Old Equity Rights pursuant to the Plan.

EXHIBIT E

Non-Insider Transfers Subject to Preference Review

Name of creditor	Payment date	Amount paid
3M	7/13/2011	\$ 13,151.20
3M	8/9/2011	
3M	8/11/2011	·
3M	9/22/2011	
3M DRUG DELIVERY SYSTEMS	7/1/2011	•
3M DRUG DELIVERY SYSTEMS	7/8/2011	
3M DRUG DELIVERY SYSTEMS	7/25/2011	
3M DRUG DELIVERY SYSTEMS	8/17/2011	
3M INNOVATION SINGAPORE PTE LTD.	8/12/2011	•
3M INNOVATION SINGAPORE PTE LTD.	9/23/2011	•
ABF FREIGHT SYSTEM	7/5/2011	
ABF FREIGHT SYSTEM	7/14/2011	·
ABF FREIGHT SYSTEM	7/21/2011	
ABF FREIGHT SYSTEM	8/4/2011	
ABF FREIGHT SYSTEM	8/11/2011	
ABF FREIGHT SYSTEM	8/18/2011	
ABF FREIGHT SYSTEM	8/25/2011	
ABF FREIGHT SYSTEM	9/1/2011	
ABF FREIGHT SYSTEM	9/8/2011	
ABF FREIGHT SYSTEM	9/21/2011	
ADAMS, SOLANGE	8/5/2011	· ·
ADAMS, SOLANGE	9/2/2011	
ADAMS, SOLANGE	9/6/2011	
ADAMS, SOLANGE	9/16/2011	=
ADAMS, SOLANGE	9/30/2011	"
ADP	7/1/2011	
ADP	7/5/2011	
ADP	7/5/2011	
ADP	7/7/2011	· ·
ADP	7/11/2011	
ADP	7/11/2011	
ADP	7/15/2011	
ADP	7/18/2011	
ADP		
ADP	7/18/2011	
ADP	7/18/2011	
ADP	7/21/2011	
ADP	7/22/2011	
	7/25/2011	
ADP	7/26/2011	
ADP	7/27/2011	
ADP	7/29/2011	
ADP	8/2/2011	
ADP	8/4/2011	
ADP	8/5/2011	
ADP	8/5/2011	•
ADP	8/8/2011	
ADP	8/8/2011	
ADP	8/11/2011	
ADP	8/12/2011	
ADP	8/12/2011	
ADP	8/18/2011	
ADP	8/19/2011	
ADP	8/19/2011	
ADP	8/22/2011	\$ 76.43

Name of creditor	Payment date	Am	ount paid
ADP	8/22/2011	\$	192.30
ADP	8/22/2011		· 242.50
ADP	8/22/2011	\$	242.50
ADP	8/25/2011		47.39
ADP	8/26/2011	\$	600.79
ADP	8/29/2011		355.00
ADP	9/1/2011		164,925.94
ADP	9/2/2011		16.80
ADP	9/6/2011		192.30
ADP	9/6/2011		9 35.1 7
ADP	9/9/2011		478.25
ADP	9/15/2011	\$	410.00
ADP	9/15/2011		163,509.73
ADP	9/16/2011		38.85
ADP	9/19/2011		192.30
ADP	9/19/2011		295.00
ADP	9/19/2011		295.00
ADP	9/22/2011		238.82
ADP	9/23/2011		969.15
ADP	9/26/2011		355.00
ADP	9/26/2011		240,663.42
ADP	9/27/2011		192.30
ADP, INC	7/14/2011		167.92
ADP, INC	8/18/2011		167.92
ADP, INC	9/1/2011		169.67
ADT SECURITY SERVICES, INC.	7/5/2011		365.75
ADT SECURITY SERVICES, INC.	8/4/2011		365.75
ADT SECURITY SERVICES, INC.	9/1/2011		365.75
ADVANSTAR COMMUNICATIONS	7/5/2011		6,062.88
ADVANSTAR COMMUNICATIONS	8/18/2011	-	11,505.90
ADVANSTAR COMMUNICATIONS	8/25/2011		7,354.20
AFCO	7/7/2011		45,119.60
AFCO	8/8/2011		45,119.60
AFCO	9/1/2011		45,119.60
ALABAMA MEDICAID AGENCY	7/5/2011		22,525.10
ALVAREZ & MARSAL	8/5/2011	"	162,379.38
ALVAREZ & MARSAL	8/10/2011		55,339.00
ALVAREZ & MARSAL	8/18/2011		150,749.81
ALVAREZ & MARSAL	8/31/2011		71,285.69
ALVAREZ & MARSAL	9/7/2011		71,123.11
ALVAREZ & MARSAL	9/16/2011		94,125.22
ALVAREZ & MARSAL	9/21/2011		108,277.63
ALVAREZ & MARSAL	9/27/2011		305,042.05
AMERICAN CONGRESS OF OBSTETRICIANS & GYNECOLOGISTS	7/14/2011		1,300.00
AMERICAN CONGRESS OF OBSTETRICIANS & GYNECOLOGISTS	7/14/2011		1,300.00
AMERICAN EXPRESS	7/5/2011		8,821.94
AMERICAN EXPRESS	7/5/2011		62,713.44
AMERICAN EXPRESS	7/8/2011		200,000.00
AMERICAN EXPRESS	7/8/2011		300,000.00
AMERICAN EXPRESS	7/8/2011		66,339.35
AMERICAN EXPRESS	7/11/2011 7/14/2011		23.85
AMERICAN EXPRESS	7/14/2011		23.83 80,419.50
AMERICAN EXPRESS	7/18/2011		83,499.15
A MATERIAL NAME AND A TOTAL TOTAL OF THE PARTY OF THE PAR	. //25/2011	40	03,477.13

AMERICAN EXPRESS 8/16/2011 \$ AMERICAN EXPRESS 8/23/2011 \$ AMERICAN EXPRESS 8/30/2011 \$ AMERICAN EXPRESS 8/30/2011 \$ AMERICAN EXPRESS 9/7/2011 \$ AMERICAN EXPRESS 9/13/2011 \$	t paid
AMERICAN EXPRESS 8/23/2011 \$ AMERICAN EXPRESS 8/25/2011 \$ AMERICAN EXPRESS 8/25/2011 \$ AMERICAN EXPRESS 8/25/2011 \$ AMERICAN EXPRESS 9/17/2011 \$ AMERICAN EXPRESS 9/17/2011 \$ AMERICAN EXPRESS 9/17/2011 \$ AMERICAN EXPRESS 9/17/2011 \$ AMERICAN EXPRESS 9/26/2011 \$ AMERICAN EXPRESS 9/26/2011 \$ AMERICAN EXPRESS 9/26/2011 \$ AMERICAN EXPRESS 9/26/2011 \$ AMERICAN MEDICAL ASSOCIATION 9/1/2011 \$ ARAMARK REFRESHMENT SVCS 7/12/2011 \$ ARCLESI, JAMES 8/15/2011 \$ ARCLESI, JAMES 8/15/2011 \$ ARROWOOD, MARK 9/15/2011 \$ ARROWOOD, MARK W 7/25/2011 \$ ARROWOOD, MARK W 7/25/2011 \$ ARROWOOD, MARK W 8/15/2011 \$ ARROWOOD, MARK W 9/2/2011 \$ ARROWOOD, MARK W 9/2/2011 \$ ARROWOOD, MARK W 9/16/2011 \$ ARROWOOD, MARK W 8/15/2011 \$ ARROWOOD, MARK W 9/16/2011 \$ ARROW	67,659.44
AMERICAN EXPRESS 8/30/2011 \$ AMERICAN EXPRESS 8/30/2011 \$ AMERICAN EXPRESS 9/13/2011 \$ AMERICAN EXPRESS 9/12/2011 \$ AMERICAN EXPRESS 9/12/2011 \$ AMERICAN MEDICAL ASSOCIATION 9/1/2011 \$ ARRICAS MEDICAL ASSOCIATION 9/1/2011 \$ ARRICLESI, JAMES 8/15/2011 \$ ARRICLESI, JAMES 8/15/2011 \$ ARRICLESI, JAMES 8/15/2011 \$ ARRICLESI, JAMES 7/5/2011 \$ ARROWOOD, MARK 8/15/2011 \$ ARROWOOD, MARK W 7/8/2011 \$ ARROWOOD, MARK W 7/8/2011 \$ ARROWOOD, MARK W 8/15/2011 \$ ARROWOOD, MARK W 9/16/2011 \$ ABBLON, ROBERT 9/2/2011 \$ BABLION, ROBERT 9/12/2011 \$ BABLION, ROBERT	85,352.85
AMERICAN EXPRESS 9/17/2011 \$ AMERICAN EXPRESS 9/17/2011 \$ AMERICAN EXPRESS 9/17/2011 \$ AMERICAN EXPRESS 9/13/2011 \$ AMERICAN EXPRESS 9/21/2011 \$ AMERICAN EXPRESS 9/21/2011 \$ AMERICAN EXPRESS 9/26/2011 \$ AMERICAN MEDICAL ASSOCIATION 9/17/2011 \$ ARAMARK REFRESHMENT SVCS 7/21/2011 \$ ARAMARK REFRESHMENT SVCS 7/21/2011 \$ ARACLESI, JAMES 8/15/2011 \$ ARKZONA HEALTH CARE COST 7/5/2011 \$ ARKOWOOD, MARK W 7/25/2011 \$ ARROWOOD, MARK W 7/25/2011 \$ ARROWOOD, MARK W 7/22/2011 \$ ARROWOOD, MARK W 8/15/2011 \$ ARROWOOD, MARK W 9/16/2011 \$ ARROWOOD, MARK W 8/15/2011 \$ ARROWOOD, MARK W 9/16/2011 \$ ASSOC OF REPRODUCTIVE HEALTH PROF 9/8/2011 \$ BABLLON, ROBERT 9/16/2011 \$ BABLLEY, GREGORY K 9/16/2011 \$ BALLEY, GREGORY K 9/16/2011 \$ BALLEY, G	59,655.49
AMERICAN EXPRESS 9/17/2011 \$ AMERICAN EXPRESS 9/13/2011 \$ AMERICAN EXPRESS 9/13/2011 \$ AMERICAN EXPRESS 9/12/2011 \$ AMERICAN EXPRESS 9/21/2011 \$ AMERICAN EXPRESS 9/21/2011 \$ AMERICAN MEDICAL ASSOCIATION 9/1/2011 \$ ARCILESI, JAMES 8/15/2011 \$ ARCOWOOD, MARK COST 7/5/2011 \$ ARROWOOD, MARK W 7/22/2011 \$ ARROWOOD, MARK W 7/22/2011 \$ ARROWOOD, MARK W 8/15/2011 \$ ARROWOOD, MARK W 8/15/2011 \$ ARROWOOD, MARK W 8/15/2011 \$ ARROWOOD, MARK W 9/16/2011 \$ ARROWOOD, MARK W 8/10/2011 \$ ARROWOOD, MARK W 8/10/2011 \$ ARROWOOD, MARK W 9/16/2011 \$ ASSOC OF REPPODUCTIVE HEALTH PROF 9/16/2011 \$ ALTHOS ENERGY 7/7/2011 \$ AUTONOMY, INC 8/11/2011 \$ AUTONOMY, INC 8/11/2011 \$ BABILON, ROBERT 7/22/2011 \$ BABILON, ROBERT 8/19/2011 \$ BABILON, ROBERT 9/16/2011 \$ BABILON,	1,161.09
AMERICAN EXPRIESS 9/17/2011 \$ AMERICAN EXPRESS 9/12/2011 \$ AMERICAN EXPRESS 9/21/2011 \$ AMERICAN EXPRESS 9/22/2011 \$ AMERICAN EXPRESS 9/22/2011 \$ AMERICAN EXPRESS 9/22/2011 \$ AMERICAN EXPRESS 9/22/2011 \$ AMERICAN MEDICAL ASSOCIATION 9/11/2011 \$ ARAMARK REFRESHMENT SVCS 7/21/2011 \$ ARCILESI, JAMES 8/15/2011 \$ ARCILESI, JAMES 8/15/2011 \$ ARIZONA HEALTH CARE COST 7/5/2011 \$ ARROWOOD, MARK SOLIS 7/5/2011 \$ ARROWOOD, MARK W 7/22/2011 \$ ARROWOOD, MARK W 7/22/2011 \$ ARROWOOD, MARK W 8/15/2011 \$ ARROWOOD, MARK W 8/15/2011 \$ ARROWOOD, MARK W 8/15/2011 \$ ARROWOOD, MARK W 9/2/2011 \$ ARROWOOD, MARK W 9/16/2011 \$ ARROWOOD, MARK W 8/15/2011 \$ ARROWOOD, MARK W 8/15/2011 \$ ARROWOOD, MARK W 8/15/2011 \$ ARROWOOD, MARK W 9/16/2011 \$ BABLION, ROBERT 9/16/2011	53,436.26
AMERICAN EXPRESS 9/13/2011 \$ AMERICAN EXPRESS 9/26/2011 \$ AMERICAN EXPRESS 9/26/2011 \$ AMERICAN EXPRESS 9/26/2011 \$ AMERICAN MEDICAL ASSOCIATION 9/1/2011 \$ ARAMARK REFRESHMENT SVCS 7/21/2011 \$ ARCILESI, JAMES 8/15/2011 \$ ARIZONA HEALTH CARE COST 7/5/2011 \$ ARROWOOD, MARK W 7/25/2011 \$ ARROWOOD, MARK W 7/25/2011 \$ ARROWOOD, MARK W 7/22/2011 \$ ARROWOOD, MARK W 8/5/2011 \$ ARROWOOD, MARK W 9/2/2011 \$ ARROWOOD, MARK W 9/2/2011 \$ ARROWOOD, MARK W 8/5/2011 \$ ARROWOOD, MARK W 9/16/2011 \$ BABLLON, ROBERT 9/16/2011 \$	77,769.37
AMERICAN EXPRIESS 9/21/2011 \$ AMERICAN MEDICAL ASSOCIATION 9/1/2011 \$ ARAMARK REFRESHMENT SVCS 7/21/2011 \$ ARAMARK REFRESHMENT SVCS 7/21/2011 \$ ARCILESI, JAMES 8/15/2011 \$ ARCILESI, JAMES 8/15/2011 \$ ARIZONA HEALTH CARE COST 7/5/2011 \$ ARROWOOD, MARK 7/25/2011 \$ ARROWOOD, MARK W 7/25/2011 \$ ARROWOOD, MARK W 7/25/2011 \$ ARROWOOD, MARK W 8/15/2011 \$ ARROWOOD, MARK W 9/10/11 \$ BABLON, ROBERT 9/8/2011 \$ BABLON, ROBERT 7/22/2011 \$ BABLON, ROBERT 9/2/2011 \$	49,826.16
AMERICAN EXPRESS 9/26/2011 \$ AMERICAN MEDICAL ASSOCIATION 9/1/2011 \$ ARAMARK REFRESHMENT SVCS 7/21/2011 \$ ARCLESI, JAMES 8/15/2011 \$ ARIZONA HEALTH CARE COST 7/5/2011 \$ ARROWOOD, MARK 7/25/2011 \$ ARROWOOD, MARK W 7/8/2011 \$ ARROWOOD, MARK W 7/22/2011 \$ ARROWOOD, MARK W 8/15/2011 \$ ARROWOOD, MARK W 8/15/2011 \$ ARROWOOD, MARK W 8/15/2011 \$ ARROWOOD, MARK W 9/2/2011 \$ ARROWOOD, MARK W 9/16/2011 \$ ARROWOOD, MARK W 9/30/2011 \$ ARROWOOD, MARK W 9/30/2011 \$ ARROWOOD, MARK W 9/30/2011 \$ ASSOC OF REPRODUCTIVE HEALTH PROF 9/8/2011 \$ ATMOS ENERGY 7/7/2011 \$ AUTONOMY, INC 8/25/2011 \$ BABILON, ROBERT 7/8/2011 \$ BABILON, ROBERT 9/2/2011 \$ BABILON, ROBERT 9/2/2011 \$ BABILON, ROBERT 9/30/2011 \$ BABILON, ROBERT 9/30/2011 \$ BABILON, ROBERT 9/30/2011 \$ BABILEY, GREGORY K 7/22/2011 \$<	31,684.85
ARAMARK REFRESHMENT SVCS 7/21/2011 \$ ARCLESI, JAMES 8/15/2011 \$ ARCLESI, JAMES 8/15/2011 \$ ARCLESI, JAMES 8/15/2011 \$ ARKANSAS DHS 7/5/2011 \$ ARRAWANSAS DHS 7/5/2011 \$ ARROWOOD, MARK 7/5/2011 \$ ARROWOOD, MARK W 7/8/2011 \$ ARROWOOD, MARK W 7/8/2011 \$ ARROWOOD, MARK W 8/5/2011 \$ ARROWOOD, MARK W 8/15/2011 \$ ARROWOOD, MARK W 9/2/2011 \$ ARROWOOD, MARK W 9/2/2011 \$ ARROWOOD, MARK W 9/16/2011 \$ ASSOC OF REPRODUCTIVE HEALTH PROF 9/8/2011 \$ ASSOC OF REPRODUCTIVE HEALTH PROF 9/8/2011 \$ AUTONOMY, INC 8/11/2011 \$ AUTONOMY, INC 8/12/2011 \$ BABILON, ROBERT 7/8/2011 \$ BABILON, ROBERT 9/12/2011 \$ BABILON, ROBERT 9/2/2011 \$ BABILON, ROBERT 9/2/2011 \$ BABILON, ROBERT 9/16/2011 \$ BABILON, ROBERT 9/16	76,523.84
ARAMARK REFRESHMENT SVCS ARCLESI, JAMES ARCLESI, JAMES ARCLESI, JAMES ARCLESI, JAMES ARIZONA HEALTH CARE COST 7/5/2011 \$ ARKANSAS DHS 7/5/2011 \$ ARROWOOD, MARK 7/5/2011 \$ ARROWOOD, MARK W 7/82/2011 \$ ARROWOOD, MARK W 7/22/2011 \$ ARROWOOD, MARK W 8/5/2011 \$ ARROWOOD, MARK W 8/15/2011 \$ ARROWOOD, MARK W 9/16/2011 \$ ASSOC OF REPRODUCTIVE HEALTH PROF 9/8/2011 \$ ASSOC OF REPRODUCTIVE HEALTH PROF 9/8/2011 \$ AUTONOMY, INC 8/11/2011 \$ AUTONOMY, INC 8/11/2011 \$ BABLION, ROBERT 9/16/2011 \$ BABLION, ROBERT 9/16/2011 \$ BABLION, ROBERT 9/2/2011 \$ BABLION, ROBERT 9/16/2011 \$ BABLION,	5,649.10
ARCILESI, JAMES ARIZONA HEALTH CARE COST ARIZONA HEALTH CARE COST ARROWOOD, MARK ARRANSAS DHS ARROWOOD, MARK ARROWOOD, MARK W	358.86
ARIZONA HEALTH CARE COST ARKANSAS DHS 7/5/2011 \$ ARRANSAS DHS 7/5/2011 \$ ARROWOOD, MARK 7/25/2011 \$ ARROWOOD, MARK W 7/22/2011 \$ ARROWOOD, MARK W 7/22/2011 \$ ARROWOOD, MARK W 8/5/2011 \$ ARROWOOD, MARK W 8/5/2011 \$ ARROWOOD, MARK W 8/5/2011 \$ ARROWOOD, MARK W 8/15/2011 \$ ARROWOOD, MARK W 8/15/2011 \$ ARROWOOD, MARK W 9/2/2011 \$ ARROWOOD, MARK W 9/16/2011 \$ ASSOC OF REPRODUCTIVE HEALTH PROF 4/5/2011 \$ ASSOC OF REPRODUCTIVE HEALTH PROF 4/17/2011 \$ AUTONOMY, INC 8/11/2011 \$ BABILON, ROBERT 7/7/2011 \$ BABILON, ROBERT 7/8/2011 \$ BABILON, ROBERT 8/5/2011 \$ BABILON, ROBERT 9/16/2011 \$ B	1,708.59
ARKANSAS DHS ARROWOOD, MARK ARROWOOD, MARK W ARROWOD, MARK W ARROWOOD, MARK W ARROWOOD, MARK W ARROWOOD, MARK W ARROWOD, MARK W ARROWOOD, MARK W ARROWOOD, MARK W ARROWOD, MARK W AROWOD, MARK W ARROWOD, MARK W ARROWOD, MARK W ARROWOD, MARK W ARO	2,727.13
ARROWOOD, MARK W 7/25/2011 \$ ARROWOOD, MARK W 7/22/2011 \$ ARROWOOD, MARK W 7/22/2011 \$ ARROWOOD, MARK W 8./5/2011 \$ ARROWOOD, MARK W 9/16/2011 \$ ARROWOOD, MARK W 9/30/2011 \$ ASSOC OF REPRODUCTIVE HEALTH PROF 9/8/2011 \$ AUTONOMY, INC 8/25/2011 \$ AUTONOMY, INC 8/11/2011 \$ AUTONOMY, INC 8/25/2011 \$ BABILON, ROBERT 7/8/2011 \$ BABILON, ROBERT 8/5/2011 \$ BABILON, ROBERT 8/5/2011 \$ BABILON, ROBERT 8/5/2011 \$ BABILON, ROBERT 9/16/2011 \$ BABILON, ROBERT 9/30/2011 \$ BABILON, ROBERT 9/16/2011 \$ BABIL	5,601.52
ARROWOOD, MARK W ASCEND MEDIA ASSOC OF REPRODUCTIVE HEALTH PROF AUTONOMY, INC BABILON, ROBERT T/8/2011 \$ BABILON, ROBERT BABILON, ROB	15.00
ARROWOOD, MARK W 8/5/2011 \$ ARROWOOD, MARK W 8/5/2011 \$ ARROWOOD, MARK W 8/15/2011 \$ ARROWOOD, MARK W 8/15/2011 \$ ARROWOOD, MARK W 8/19/2011 \$ ARROWOOD, MARK W 9/2/2011 \$ ARROWOOD, MARK W 9/16/2011 \$ ARROWOOD, MARK W 9/16/2011 \$ ARROWOOD, MARK W 9/30/2011 \$ ARROWOOD, MARK W 9/2/2011 \$ ARROWOOD, MARK W 9/30/2011 \$ ARROWOOD	2,525.10
ARROWOOD, MARK W 8/5/2011 \$ ARROWOOD, MARK W 8/15/2011 \$ ARROWOOD, MARK W 8/19/2011 \$ ARROWOOD, MARK W 9/2/2011 \$ ARROWOOD, MARK W 9/2/2011 \$ ARROWOOD, MARK W 9/30/2011 \$ AUTONOME, W 8/2011 \$ AUTONOMY, INC 8/11/2011 \$ BABLON, ROBERGY 7/8/2011 \$ BABLON, ROBERT 8/5/2011 \$ BABLON, ROBERT 8/5/2011 \$ BABLON, ROBERT 8/19/2011 \$ BABLON, ROBERT 9/16/2011 \$ BABLEY, GREGORY K 7/8/2011 \$ BALLEY, GREGORY K 8/19/2011 \$ BALLEY, GREGORY K 8/19/2011 \$ BALLEY, GREGORY K 9/16/2011 \$ BALLEY, GR	2,525.10
ARROWOOD, MARK W 9/2/2011 \$ ARROWOOD, MARK W 9/30/2011 \$ ARROWOOD, MARK W 9/30/2011 \$ ARROWOOD, MARK W 9/30/2011 \$ ASSCEND MEDIA 7/5/2011 \$ ASSCEND MEDIA 7/5/2011 \$ ASSCE OF REPRODUCTIVE HEALTH PROF ASSOC OF REPRODUCTIVE HEALTH PROF ATMOS ENERGY 7/7/2011 \$ AUTONOMY, INC 8/11/2011 \$ AUTONOMY, INC 8/25/2011 \$ BABILON, ROBERT 7/8/2011 \$ BABILON, ROBERT 8/5/2011 \$ BABILON, ROBERT 8/5/2011 \$ BABILON, ROBERT 8/5/2011 \$ BABILON, ROBERT 8/19/2011 \$ BABILON, ROBERT 9/16/2011 \$ BABILEY, GREGORY K 7/22/2011 \$ BAILEY, GREGORY K 8/5/2011 \$ BAILEY, GREGORY K 8/5/2011 \$ BAILEY, GREGORY K 9/16/2011 \$ BAILEY, GREGORY K	2,525.10
ARROWOOD, MARK W ARROWOOD, MARK W 9/2/2011 \$ ARROWOOD, MARK W 9/16/2011 \$ ARROWOOD, MARK W 9/16/2011 \$ ARROWOOD, MARK W 9/30/2011 \$ ASCEND MEDIA 7/5/2011 \$ ASSOC OF REPRODUCTIVE HEALTH PROF ATMOS ENERGY 7/7/2011 \$ AUTONOMY, INC 8/11/2011 \$ AUTONOMY, INC 8/25/2011 \$ BABILON, ROBERT 7/8/2011 \$ BABILON, ROBERT 8/5/2011 \$ BABILON, ROBERT 8/5/2011 \$ BABILON, ROBERT 8/5/2011 \$ BABILON, ROBERT 9/2/2011 \$ BABILON, ROBERT 9/2/2011 \$ BABILON, ROBERT 9/16/2011 \$ BABILON, ROBERT 9/16/2011 \$ BABILON, ROBERT 9/16/2011 \$ BABILON, ROBERT 9/30/2011 \$ BALLEY, GREGORY K 8/5/2011 \$ BALLEY, GREGORY K 8/5/2011 \$ BALLEY, GREGORY K 8/19/2011 \$ BALLEY, GREGORY K 9/2/2011 \$ BALLEY, GREGORY K 9/2/2011 \$ BALLEY, GREGORY K 9/2/2011 \$ BALLEY, GREGORY K 9/30/2011 \$ BALLEY, GREGORY K 9/3	2,399.51
ARROWOOD, MARK W 9/16/2011 \$ ARROWOOD, MARK W 9/16/2011 \$ ARROWOOD, MARK W 9/30/2011 \$ ARROWOOD, MARK W 9/30/2011 \$ ASCEND MEDIA 7/5/2011 \$ ASSOC OF REPRODUCTIVE HEALTH PROF 9/8/2011 \$ ATMOS ENERGY 7/7/2011 \$ AUTONOMY, INC 8/11/2011 \$ AUTONOMY, INC 8/25/2011 \$ BABILON, ROBERT 7/8/2011 \$ BABILON, ROBERT 8/5/2011 \$ BABILON, ROBERT 8/5/2011 \$ BABILON, ROBERT 8/5/2011 \$ BABILON, ROBERT 8/5/2011 \$ BABILON, ROBERT 9/2/2011 \$ BABILON, ROBERT 9/2/2011 \$ BABILON, ROBERT 9/2/2011 \$ BABILON, ROBERT 9/16/2011 \$ BAILEY, GREGORY K 7/22/2011 \$ BAILEY, GREGORY K 9/16/2011 \$	2,525.11
ARROWOOD, MARK W 9/16/2011 \$ ARROWOOD, MARK W 9/30/2011 \$ ASCEND MEDIA 7/5/2011 \$ ASSOC OF REPRODUCTIVE HEALTH PROF 9/8/2011 \$ ATMOS ENERGY 7/7/2011 \$ AUTONOMY, INC 8/11/2011 \$ AUTONOMY, INC 8/25/2011 \$ BABILON, ROBERT 7/8/2011 \$ BABILON, ROBERT 8/5/2011 \$ BABILON, ROBERT 8/5/2011 \$ BABILON, ROBERT 8/5/2011 \$ BABILON, ROBERT 8/19/2011 \$ BABILON, ROBERT 8/19/2011 \$ BABILON, ROBERT 9/16/2011 \$ BABILON, ROBERT 9/2/2011 \$ BABILON, ROBERT 9/16/2011 \$ BALLEY, GREGORY K 8/5/2011 \$ BALLEY, GREGORY K 9/16/2011 \$ BALLEY, GREGORY K 9/16/	2,525.11
ARROWOOD, MARK W ASCEND MEDIA ASCEND MEDIA ASSOC OF REPRODUCTIVE HEALTH PROF ATMOS ENERGY AUTONOMY, INC 8/11/2011 BABILON, ROBERT ABBILON, ROBERT BABILON, ROBERT BAILEY, GREGORY K BAILEY,	2,525.09
ASCEND MEDIA 7/5/2011 \$ ASSOC OF REPRODUCTIVE HEALTH PROF 9/8/2011 \$ ATMOS ENERGY 7/7/2011 \$ AUTONOMY, INC 8/11/2011 \$ AUTONOMY, INC 8/25/2011 \$ BABILON, ROBERT 7/8/2011 \$ BABILON, ROBERT 7/8/2011 \$ BABILON, ROBERT 8/5/2011 \$ BABILON, ROBERT 8/19/2011 \$ BABILON, ROBERT 8/19/2011 \$ BABILON, ROBERT 9/2/2011 \$ BABILON, ROBERT 9/2/2011 \$ BABILON, ROBERT 9/16/2011 \$ BABILON, ROBERT 9/16/2011 \$ BABILON, ROBERT 9/10/2011 \$ BABILON, ROBERT 9/10/2011 \$ BABILON, ROBERT 9/10/2011 \$ BABILON, ROBERT 9/10/2011 \$ BALLEY COMPUTING TECHNOLOGIES 7/21/2011 \$ BAILEY, GREGORY K 7/22/2011 \$ BAILEY, GREGORY K 8/5/2011 \$ BAILEY, GREGORY K 8/19/2011 \$ BAILEY, GREGORY K 8/19/2011 \$ BAILEY, GREGORY K 9/16/2011 \$ BAILEY GREGORY K 9/16/2011 \$ BA	3,111.82
ASSOC OF REPRODUCTIVE HEALTH PROF ATMOS ENERGY 7/7/2011 \$ AUTONOMY, INC 8/11/2011 \$ AUTONOMY, INC 8/25/2011 \$ BABILON, ROBERT 7/8/2011 \$ BABILON, ROBERT 7/22/2011 \$ BABILON, ROBERT 8/5/2011 \$ BABILON, ROBERT 8/5/2011 \$ BABILON, ROBERT 8/19/2011 \$ BABILON, ROBERT 8/19/2011 \$ BABILON, ROBERT 9/2/2011 \$ BABILON, ROBERT 9/2/2011 \$ BABILON, ROBERT 9/16/2011 \$ BABILON, ROBERT 9/30/2011 \$ BABILON, ROBERT 9/30/2011 \$ BABILON, ROBERT 9/30/2011 \$ BABILEY, GREGORY K 7/22/2011 \$ BAILEY, GREGORY K 8/5/2011 \$ BAILEY, GREGORY K 8/5/2011 \$ BAILEY, GREGORY K 8/19/2011 \$ BAILEY, GREGORY K 9/16/2011 \$ BAILEY, GREGORY	3,537.50
ATMOS ENERGY AUTONOMY, INC 8/11/2011 \$ AUTONOMY, INC 8/25/2011 \$ BABILON, ROBERT 7/8/2011 \$ BABILON, ROBERT 7/22/2011 \$ BABILON, ROBERT 8/5/2011 \$ BABILON, ROBERT 8/5/2011 \$ BABILON, ROBERT 8/5/2011 \$ BABILON, ROBERT 8/19/2011 \$ BABILON, ROBERT 9/16/2011 \$ BABILON, ROBERT 9/16/2011 \$ BABILON, ROBERT 9/16/2011 \$ BABILON, ROBERT 9/30/2011 \$ BABILY COMPUTING TECHNOLOGIES 7/21/2011 \$ BAILEY, GREGORY K 7/22/2011 \$ BAILEY, GREGORY K 8/5/2011 \$ BAILEY, GREGORY K 8/19/2011 \$ BAILEY, GREGORY K 9/16/2011 \$ BAILEY, GREGOR	5,000.00
AUTONOMY, INC AUTONOMY, INC BABILON, ROBERT BABILEY, COMPUTING TECHNOLOGIES BAILEY, GREGORY K BAILEY,	33.33
AUTONOMY, INC BABILON, ROBERT BABILEY COMPUTING TECHNOLOGIES T/21/2011 \$ BAILEY, GREGORY K T/8/2011 \$ BAILEY, GREGORY K	8,399.37
BABILON, ROBERT 7/8/2011 \$ BABILON, ROBERT 7/22/2011 \$ BABILON, ROBERT 8/5/2011 \$ BABILON, ROBERT 8/19/2011 \$ BABILON, ROBERT 9/16/2011 \$ BABILON, ROBERT 9/30/2011 \$ BABILEY COMPUTING TECHNOLOGIES 7/21/2011 \$ BAILEY, GREGORY K 7/8/2011 \$ BAILEY, GREGORY K 8/5/2011 \$ BAILEY, GREGORY K 8/5/2011 \$ BAILEY, GREGORY K 8/19/2011 \$ BAILEY, GREGORY K 9/16/2011 \$ BAILEY, GREGORY K 9/16/2011 \$ BAILEY, GREGORY K 9/16/2011 \$ BAILEY, GREGORY K 9/30/2011 \$ BAILEY, GREGORY K 9/30/2011 \$ BAKER, DAVID 7/21/2011 \$ BAKER, KENNETH P. 7/8/2011 \$ BAKER, KENNETH P. 7/22/2011 \$ BAKER, KENNETH P. 8/5/2011 \$	7,899.37
BABILON, ROBERT 7/22/2011 \$ BABILON, ROBERT 8/5/2011 \$ BABILON, ROBERT 9/2/2011 \$ BABILON, ROBERT 9/16/2011 \$ BABILON, ROBERT 9/30/2011 \$ BABILEY COMPUTING TECHNOLOGIES 7/21/2011 \$ BAILEY, GREGORY K 7/8/2011 \$ BAILEY, GREGORY K 7/22/2011 \$ BAILEY, GREGORY K 8/5/2011 \$ BAILEY, GREGORY K 9/2/2011 \$ BAILEY, GREGORY K 9/2/2011 \$ BAILEY, GREGORY K 9/16/2011 \$ BAILEY, GREGORY K 9/30/2011 \$ BAILEY, GREGORY K 9/30/2011 \$ BAILEY, GREGORY K 9/30/2011 \$ BAKER, DAVID 7/21/2011 \$ BAKER, KENNETH P. 7/8/2011 \$ BAKER, KENNETH P. 8/5/2011 \$	4,664.41
BABILON, ROBERT 8/5/2011 \$ BABILON, ROBERT 8/19/2011 \$ BABILON, ROBERT 9/2/2011 \$ BABILON, ROBERT 9/30/2011 \$ BABILON, ROBERT 9/30/2011 \$ BAILEY COMPUTING TECHNOLOGIES 7/21/2011 \$ BAILEY, GREGORY K 7/8/2011 \$ BAILEY, GREGORY K 8/5/2011 \$ BAILEY, GREGORY K 8/19/2011 \$ BAILEY, GREGORY K 9/2/2011 \$ BAILEY, GREGORY K 9/16/2011 \$ BAILEY, GREGORY K 9/30/2011 \$ BAILEY, GREGORY K 9/30/2011 \$ BAILEY, GREGORY K 9/30/2011 \$ BAKER, DAVID 7/21/2011 \$ BAKER, KENNETH P. 7/8/2011 \$ BAKER, KENNETH P. 8/5/2011 \$ BAKER, KENNETH P. 8/5/2011 \$	4,664.40
BABILON, ROBERT 8/19/2011 \$ BABILON, ROBERT 9/2/2011 \$ BABILON, ROBERT 9/16/2011 \$ BABILON, ROBERT 9/30/2011 \$ BAILEY COMPUTING TECHNOLOGIES 7/21/2011 \$ BAILEY, GREGORY K 7/8/2011 \$ BAILEY, GREGORY K 8/5/2011 \$ BAILEY, GREGORY K 8/19/2011 \$ BAILEY, GREGORY K 9/2/2011 \$ BAILEY, GREGORY K 9/16/2011 \$ BAILEY, GREGORY K 9/30/2011 \$ BAKER, DAVID 7/21/2011 \$ BAKER, KENNETH P. 7/8/2011 \$ BAKER, KENNETH P. 8/5/2011 \$ BAKER, KENNETH P. 8/5/2011 \$	4,664.42
BABILON, ROBERT 9/2/2011 \$ BABILON, ROBERT 9/16/2011 \$ BABILON, ROBERT 9/30/2011 \$ BAILEY COMPUTING TECHNOLOGIES 7/21/2011 \$ BAILEY, GREGORY K 7/8/2011 \$ BAILEY, GREGORY K 8/5/2011 \$ BAILEY, GREGORY K 8/19/2011 \$ BAILEY, GREGORY K 9/2/2011 \$ BAILEY, GREGORY K 9/16/2011 \$ BAILEY, GREGORY K 9/30/2011 \$ BAKER, DAVID 7/21/2011 \$ BAKER, KENNETH P. 7/8/2011 \$ BAKER, KENNETH P. 8/5/2011 \$ BAKER, KENNETH P. 8/5/2011 \$	4,664.41
BABILON, ROBERT 9/16/2011 \$ BABILON, ROBERT 9/30/2011 \$ BAILEY COMPUTING TECHNOLOGIES 7/21/2011 \$ BAILEY, GREGORY K 7/8/2011 \$ BAILEY, GREGORY K 8/5/2011 \$ BAILEY, GREGORY K 8/19/2011 \$ BAILEY, GREGORY K 9/2/2011 \$ BAILEY, GREGORY K 9/16/2011 \$ BAILEY, GREGORY K 9/30/2011 \$ BAKER, DAVID 7/21/2011 \$ BAKER, KENNETH P. 7/8/2011 \$ BAKER, KENNETH P. 8/5/2011 \$	5,493.06
BABILON, ROBERT 9/30/2011 \$ BAILEY COMPUTING TECHNOLOGIES 7/21/2011 \$ BAILEY, GREGORY K 7/8/2011 \$ BAILEY, GREGORY K 8/5/2011 \$ BAILEY, GREGORY K 8/19/2011 \$ BAILEY, GREGORY K 9/2/2011 \$ BAILEY, GREGORY K 9/16/2011 \$ BAILEY, GREGORY K 9/30/2011 \$ BAKER, DAVID 7/21/2011 \$ BAKER, KENNETH P. 7/8/2011 \$ BAKER, KENNETH P. 8/5/2011 \$	5,133.05
BAILEY COMPUTING TECHNOLOGIES 7/21/2011 \$ BAILEY, GREGORY K 7/8/2011 \$ BAILEY, GREGORY K 8/5/2011 \$ BAILEY, GREGORY K 8/19/2011 \$ BAILEY, GREGORY K 9/2/2011 \$ BAILEY, GREGORY K 9/16/2011 \$ BAILEY, GREGORY K 9/30/2011 \$ BAKER, DAVID 7/21/2011 \$ BAKER, KENNETH P. 7/8/2011 \$ BAKER, KENNETH P. 8/5/2011 \$	5,133.07
BAILEY, GREGORY K BAKER, DAVID BAKER, LENNETH P. BAKER, KENNETH P. BAKER, SENNETH P.	20,107.49
BAILEY, GREGORY K 7/22/2011 \$ BAILEY, GREGORY K 8/5/2011 \$ BAILEY, GREGORY K 8/19/2011 \$ BAILEY, GREGORY K 9/2/2011 \$ BAILEY, GREGORY K 9/16/2011 \$ BAILEY, GREGORY K 9/30/2011 \$ BAKER, DAVID 7/21/2011 \$ BAKER, KENNETH P. 7/8/2011 \$ BAKER, KENNETH P. 8/5/2011 \$	2,037.39
BAILEY, GREGORY K 8/5/2011 \$ BAILEY, GREGORY K 8/19/2011 \$ BAILEY, GREGORY K 9/2/2011 \$ BAILEY, GREGORY K 9/16/2011 \$ BAILEY, GREGORY K 9/30/2011 \$ BAKER, DAVID 7/21/2011 \$ BAKER, KENNETH P. 7/8/2011 \$ BAKER, KENNETH P. 8/5/2011 \$	2,037.39
BAILEY, GREGORY K 8/19/2011 \$ BAILEY, GREGORY K 9/2/2011 \$ BAILEY, GREGORY K 9/16/2011 \$ BAILEY, GREGORY K 9/30/2011 \$ BAKER, DAVID 7/21/2011 \$ BAKER, KENNETH P. 7/8/2011 \$ BAKER, KENNETH P. 8/5/2011 \$	2,037.39
BAILEY, GREGORY K 9/2/2011 \$ BAILEY, GREGORY K 9/16/2011 \$ BAILEY, GREGORY K 9/30/2011 \$ BAKER, DAVID 7/21/2011 \$ BAKER, KENNETH P. 7/8/2011 \$ BAKER, KENNETH P. 7/22/2011 \$ BAKER, KENNETH P. 8/5/2011 \$	2,037.40
BAILEY, GREGORY K 9/16/2011 \$ BAILEY, GREGORY K 9/30/2011 \$ BAKER, DAVID 7/21/2011 \$ BAKER, KENNETH P. 7/8/2011 \$ BAKER, KENNETH P. 7/22/2011 \$ BAKER, KENNETH P. 8/5/2011 \$	2,037.39
BAILEY, GREGORY K 9/30/2011 \$ BAKER, DAVID 7/21/2011 \$ BAKER, KENNETH P. 7/8/2011 \$ BAKER, KENNETH P. 8/5/2011 \$	
BAKER, DAVID 7/21/2011 \$ BAKER, KENNETH P. 7/8/2011 \$ BAKER, KENNETH P. 7/22/2011 \$ BAKER, KENNETH P. 8/5/2011 \$	2,037.39 2,037.39
BAKER, KENNETH P. 7/8/2011 \$ BAKER, KENNETH P. 7/22/2011 \$ BAKER, KENNETH P. 8/5/2011 \$	1,200.00
BAKER, KENNETH P. 7/22/2011 \$ BAKER, KENNETH P. 8/5/2011 \$	=
BAKER, KENNETH P. 8/5/2011 \$	5,168.13
-,-,	5,168.13
//21/2011 15	2,903.05
, , , , , , , , , , , , , , , , , , , ,	1,610.59
· · · · · · · · · · · · · · · ·	119.85
	29,975.54
	86,294.30
	624,710.50
	49,115.40 924,722.69

Name of creditor	Payment date	Amount paid
BARB, PAM	7/8/2011	1,143.43
BARB, PAM	7/22/2011	1, 147.45
BARB, PAM	8/5/2011	1,182.44
BARB, PAM	8/19/2011	
BARB, PAM	9/2/2011	
BARB, PAM	9/16/2011	
BARB, PAM	9/30/2011	1,182.45
BARNES, ELIZABETH	7/8/2011	
BARNES, ELIZABETH	7/22/2011	·
BARNES, ELIZABETH	8/5/2011	
BARNES, ELIZABETH	8/15/2011	
BARNES, ELIZABETH	8/19/2011	
BARNES, ELIZABETH	9/2/2011	
BARNES, ELIZABETH	9/16/2011	
BARNES, ELIZABETH	9/30/2011	
BARNES, MRS ELIZABETH	7/7/2011	·
BARNES, MRS ELIZABETH	7/25/2011	
BARNES, MRS ELIZABETH	8/25/2011	
BARRIER ADVISORS, INC.	9/7/2011	
BARRIER ADVISORS, INC.	9/16/2011	
BARRIER ADVISORS, INC.	9/27/2011	
BARRON, DOUGLAS	7/8/2011	
BARRON, DOUGLAS	7/11/2011	• ,
BARRON, DOUGLAS	7/22/2011	
BARRON, DOUGLAS	8/5/2011	
BARRON, DOUGLAS	8/19/2011	•
BARRON, DOUGLAS	9/2/2011	
BARTICK, MICHELLE	7/8/2011	•
BARTICK, MICHELLE	7/11/2011	
BARTICK, MICHELLE	7/22/2011	-
BARTICK, MICHELLE	7/25/2011	
BARTICK, MICHELLE	8/5/2011	
BARTICK, MICHELLE	8/15/2011	·
BARTICK, MICHELLE	8/19/2011	
BARTICK, MICHELLE	8/25/2011	
BARTICK, MICHELLE	9/2/2011	
BARTICK, MICHELLE	9/8/2011	
BARTICK, MICHELLE	9/16/2011	
BARTICK, MICHELLE	9/26/2011	
BARTICK, MICHELLE	9/30/2011	
BEITEL, LAURIE	7/8/2011	
BEITEL, LAURIE	7/22/2011	
BEITEL, LAURIE	8/5/2011	
BEITEL, LAURIE	8/19/2011	
BEITEL, LAURIE	9/2/2011	
BEITEL, LAURIE	9/16/2011	
BEITEL, LAURIE	9/30/2011	
BENTLEY, TIMOTHY	7/5/2011	
BENTLEY, TIMOTHY	7/8/2011	
BENTLEY, TIMOTHY	7/8/2011	
BENTLEY, TIMOTHY	7/25/2011	
BENTLEY, TIMOTHY		-
	8/5/2011 8/19/2011	
BENTLEY, TIMOTHY BENTLEY, TIMOTHY	8/19/2011	
BENTLEY, TIMOTHY	9/2/2011	\$ 4,665.21

Name of creditor	Payment date	Amount paid
BENTLEY, TIMOTHY	9/15/2011	\$ 192.25
BENTLEY, TIMOTHY	9/16/2011	
BENTLEY, TIMOTHY	9/26/2011	•
BENTLEY, TIMOTHY	9/30/2011	
BILCARE, INC.	7/14/2011	
BILCARE, INC.	8/11/2011	
BILCARE, INC.	9/1/2011	
BILCARE, INC.	9/8/2011	
BIOKOSMES S.R.L.	9/20/2011	•
BLANDIN, ANTHONY	8/25/2011	
BLANDIN, ANTHONY M.	7/8/2011	
BLANDIN, ANTHONY M.	7/22/2011	·
BLANDIN, ANTHONY M.	8/5/2011	· ·
BLANDIN, ANTHONY M.	8/19/2011	
BLANDIN, ANTHONY M.	9/2/2011	
BLANDIN, ANTHONY M.	9/16/2011	· ·
BLANDIN, ANTHONY M.	9/30/2011	· ·
BLEILER, LYNN	7/5/2011	
BLEILER, LYNN	7/25/2011	
BLEILER, LYNN	8/8/2011	
BLEILER, LYNN	8/15/2011	
BLEILER, LYNN	9/8/2011	·
BLEILER, LYNN		\$ 10.00
BLEILER, LYNN N.	7/8/2011	
BLEILER, LYNN N.	7/22/2011	
BLEILER, LYNN N.	8/5/2011	
BLEILER, LYNN N.	8/15/2011	-
BLEILER, LYNN N.	8/19/2011	
BLEILER, LYNN N.	9/2/2011	
BLEILER, LYNN N.	9/16/2011	·
BLEILER, LYNN N.	9/30/2011	
BLISS, JOHN	7/8/2011	•
BLISS, JOHN	7/22/2011	
BLISS, JOHN	8/5/2011	
BLISS, JOHN	8/19/2011	
BLISS, JOHN	9/2/2011	
BLISS, JOHN	9/16/2011	
BLISS, JOHN	9/26/2011	
BLISS, JOHN	9/30/2011	•
BLUE CROSS BLUE SHIELD OF TN	7/14/2011	·
BLUE CROSS BLUE SHIELD OF TN	7/28/2011	·
BLUE CROSS BLUE SHIELD OF TN	8/11/2011	·
BLUE CROSS BLUE SHIELD OF TN	8/25/2011	•
BLUE CROSS BLUE SHIELD OF TN	9/8/2011	'
BLUE CROSS BLUE SHIELD OF TN	9/22/2011	
BLUE SHIELD OF CALIFORNIA SERVICES	8/25/2011	· ·
BLUE SHIELD OF CALIFORNIA SERVICES	8/25/2011	" ,
BLUECROSS & BLUESHIELD OF TENN	7/21/2011	•
BLUECROSS & BLUESHIELD OF TENN	8/11/2011	•
BLUECROSS & BLUESHIELD OF TENN	9/15/2011	
BMC GROUP VDR, LLC	8/26/2011	·
BMC GROUP VDR, LLC	8/26/2011	- ,
BOLGER	7/14/2011	
BORSOS, LYNNE M.	7/8/2011	
	7/0/2011	¥ 1,575.22

Name of creditor	Payment date	Amount paid
BORSOS, LYNNE M.	7/22/2011	\$ 1,545.23
BORSOS, LYNNE M.	8/5/2011	\$ 1,545.22
BORSOS, LYNNE M.	8/15/2011	\$ 5,484.25
BORSOS, LYNNE M.	8/19/2011	\$ 1,545.23
BORSOS, LYNNE M.	9/2/2011	1,545.23
BORSOS, LYNNE M.	9/16/2011	\$ 1,545.24
BORSOS, LYNNE M.	9/30/2011	\$ 3,176.18
BOTTA, JERRY	7/8/2011	\$ 2,193.34
BOTTA, JERRY	7/22/2011	\$ 2,193.35
BOTTA, JERRY	8/5/2011	\$ 2,193.33
BOTTA, JERRY	8/15/2011	\$ 3,108.93
BOTTA, JERRY	8/19/2011	
BOTTA, JERRY	9/2/2011	
BOTTA, JERRY	9/16/2011	\$ 2,193.34
BOTTA, JERRY	9/30/2011	
BOTTA, MR JERRY	7/25/2011	•
BOTTA, MR JERRY	8/8/2011	
BOTTA, MR JERRY	8/25/2011	
BOTTA, MR JERRY	9/15/2011	
BOWERS, JENNIFER J	8/15/2011	
BOWLES, JOHN	7/8/2011	•
BOWLES, JOHN	7/22/2011	
BOWLES, JOHN	8/5/2011	
BOWLES, JOHN	8/19/2011	
BOWLES, JOHN	9/2/2011	
BOWLES, JOHN	9/16/2011	
BOWLES, JOHN	9/30/2011	
BRANSON, KATHY W.	7/8/2011	•
BRANSON, KATHY W.	7/22/2011	·
BRANSON, KATHY W.	8/5/2011	
BRANSON, KATHY W.	8/19/2011	
BRANSON, KATHY W.	9/2/2011	
BRANSON, KATHY W.	9/16/2011	
BRANSON, KATHY W.	9/30/2011	
BRENNAN, SEAN	7/8/2011	
BRENNAN, SEAN	7/22/2011	
BRENNAN, SEAN	8/5/2011	"
BRENNAN, SEAN	8/19/2011	•
BRENNAN, SEAN	·	
BRENNAN, SEAN	9/2/2011	
BRENNAN, SEAN	9/16/2011	
BRISTOL OFFICE SUPPLY	9/30/2011	•
BRISTOL OFFICE SUPPLY	7/5/2011	*
BRISTOL OFFICE SUPPLY	7/14/2011	
	7/21/2011	
BRISTOL OFFICE SUPPLY	8/4/2011	
BRISTOL OFFICE SUPPLY BRISTOL OFFICE SUPPLY	8/11/2011	
BRISTOL OFFICE SUPPLY BRISTOL OFFICE SUPPLY	8/25/2011	
	9/1/2011	
BRISTOL OFFICE SUPPLY	9/21/2011	
BROWN, EDWARDS & COMPANY, L.L.P.	7/21/2011	
BROWN, EDWARDS & COMPANY, L.L.P.	8/25/2011	•
BTES	7/14/2011	
BTES	8/18/2011	-
BTES	9/21/2011	\$ 6,516.20

BUDGET CONFIRENCING, INC. 7/14/2011 \$ 2,697.70 BULLETIN NEWS, LLC 8/18/2011 \$ 29,540.00 BUSINESS WIRE 8/11/2011 \$ 1,500.00 CALIFORNIA SOCIETY OF DERMATOLOGY 7/7/2011 \$ 1,580.00 CANTU, MARC 7/2/2011 \$ 1,580.50 CANTU, MARC 8/35/2011 \$ 1,580.51 CANTU, MARC 8/35/2011 \$ 1,580.35 CANTU, MARC 8/15/2011 \$ 1,580.36 CANTU, MARC 8/19/2011 \$ 1,580.36 CANTU, MARC 9/16/2011 \$ 1,580.36 CAREMARKYCS HEALTH LP 8/4/2011 \$ 6,99.22 CAREMARKYCS HEALTH LP 8/4/2011 \$ 1,590.34 CAREMARKYCS HEALTH LP 8/4/2011 \$ 1,590.34 CARED WILLIAM 9/16/2011 \$ 1,590.34<	Name of creditor	Payment date	Amount paid
BULLETIN NEWS, LLC 8/18/2011 \$ 29,40.0 BULLETIN NEWS, LLC 8/18/2011 \$ 29,40.0 BULSETIN NEWS, LLC 8/11/2011 \$ 277.03 BUS STATE INSURANCE FUND 7/14/2011 \$ 277.03 C.S.C. FORCE MEASUREMENT, INC. 8/42011 \$ 1,880.52 CALLFORNIA SOCIETY OF DERMATOLOGY 7/7/2011 \$ 1,886.52 CANTU, MARC 7/2/2011 \$ 1,880.36 CANTU, MARC 8/15/2011 \$ 1,880.36 CANTU, MARC 9/16/2011 \$ 1,880.36 CAREMARKINGS HEALTH LP 8/4/2011 \$ 1,880.36 CAREMARKINGS HEALTH LP 8/4/2011 \$ 1,400.00 CAREW INTERNATIONAL, INC. 7/5/2011 \$ 1,400.00 CAREW INTERNATIONAL PROPERTY 7/5/2011 \$ 1,400.00 CAREW WILLIAM 9/16/2011 \$ 1,400.00 CASH, WILLIAM </td <td>BUDGET CONFERENCING, INC.</td> <td>7/14/2011</td> <td>2,697.70</td>	BUDGET CONFERENCING, INC.	7/14/2011	2,697.70
BULLTEN NEWS, LLC 8/18/2011 \$ 05.00 BUSINESS WIRE 8/11/2011 \$ 05.00 BUSINESS WIRE 8/11/2011 \$ 05.00 BUSINESS WIRE 7/14/2011 \$ 05.00 CS.C. FORCE MEASUREMENT, INC. 8/4/2011 \$ 1,880.00 CANTU, MARC 7/8/2011 \$ 1,580.50 CANTU, MARC 8/5/2011 \$ 1,580.51 CANTU, MARC 8/5/2011 \$ 1,580.35 CANTU, MARC 8/19/2011 \$ 1,580.36 CANTU, MARC 8/19/2011 \$ 1,580.36 CANTU, MARC 9/2/2011 \$ 1,580.36 CANTU, MARC 9/9/2/2011 \$ 1,580.36 CANTU, MARC 9/9/2/2011 \$ 1,580.36 CANTU, MARC 9/9/2/2011 \$ 1,580.36 CANTU, MARC 9/8/2011 \$ 1,580.36 CANTU, MARC 9/8/2011 \$ 1,580.36 CAREMARKPCS HEALTH LP 8/4/2011 \$ 1,580.36	BULLETIN NEWS, LLC		
BUSINESS WIRE 8/11/2011 \$ 050.00 BWC STATE INSURANCE FUND 7/14/2011 \$ 8,451.94 CALIFORNIA SOCIETY OF DERMATOLOGY 7/7/2011 \$ 1,588.52 CANTU, MARC 7/8/2011 \$ 1,588.52 CANTU, MARC 8/15/2011 \$ 1,588.52 CANTU, MARC 8/15/2011 \$ 1,588.52 CANTU, MARC 8/15/2011 \$ 1,580.36 CANTU, MARC 8/15/2011 \$ 1,580.36 CANTU, MARC 9/16/2011 \$ 1,580.36 CAREMARKPCS HEALTH LP 8/4/2011 \$ 1,590.34 CAREMARKPCS HEALTH LP 8/4/2011 \$ 1,590.34 CAREY INTERNATI	BULLETIN NEWS, LLC	8/18/2011	
BWC STATE INSURANCE FUND	BUSINESS WIRE		•
CS.C. FORCE MEASUREMENT, INC. 8/4/2011 \$ 1,885.00 CALIFORNIA SOCIETY OF DERMATOLOGY 7/7/2011 \$ 1,885.00 CANTU, MARC 7/8/2011 \$ 1,886.36 CANTU, MARC 8/15/2011 \$ 3,885.81 CANTU, MARC 8/15/2011 \$ 3,885.81 CANTU, MARC 8/15/2011 \$ 1,580.36 CANTU, MARC 8/19/2011 \$ 1,580.36 CANTU, MARC 9/2/2011 \$ 1,580.36 CANTU, MARC 9/2/2011 \$ 1,580.36 CANTU, MARC 9/16/2011 \$ 1,580.36 CAREMARKPCS HEALTH LP 8/4/2011 \$ 1,590.34 CAREMARKPCS HEALTH LP 8/4/2011 \$ 1,590.34 CAREY INTERNATIONAL, INC. 7/5/2011 \$ 3,500.36 CAREY INTERNATIONAL, INC. 7/5/2011 \$ 3,500.36 CAREJ WILLIAM 9/8/2011 \$ 472.77 CASH, WILLIAM 9/6/2011 \$ 472.77 CASH, WILLIAM 9/6/2011 \$ 472.77 CASH, WILLIAM 9/30/2011 \$ 2,006.39 CASH, WILLIAM 9/30/2011 \$ 2,006.39 CASH, WILLIAM 9/30/2011 \$ 2,006.39	BWC STATE INSURANCE FUND		
CALIFORNIA SOCIETY OF DERMATOLOGY	C.S.C. FORCE MEASUREMENT, INC.		
CANTU, MARC 7/8/2011 \$ 1,588.51 CANTU, MARC 7/2/2011 \$ 1,580.51 CANTU, MARC 8/5/2011 \$ 3,335.74 CANTU, MARC 8/15/2011 \$ 1,580.36 CANTU, MARC 9/16/2011 \$ 1,580.36 CAREMARKPCS HEALTH LP 8/4/2011 \$ 60.292 CAREMARKPCS HEALTH LP 8/4/2011 \$ 61,490.42 CAREY INTERNATIONAL, INC. 7/5/2011 \$ 72.31 CARGO DATA 9/8/2011 \$ 1,410.00 CARISHEMAN DERMATOLOGY SYMPOSIUM 7/7/2011 \$ 300.00 CASH, WILLIAM 9/16/2011 \$ 472.77 CASH, WILLIAM 9/16/2011 \$ 300.00 CASH, WILLIAM 9/16/2011 \$ 70.00 CENTURYLINK 7/14/2011 \$ 70.00 CENTURYLINK </td <td>CALIFORNIA SOCIETY OF DERMATOLOGY</td> <td></td> <td>•</td>	CALIFORNIA SOCIETY OF DERMATOLOGY		•
CANTU, MARC 8/5/2011 \$ 1,580.36 CANTU, MARC 8/15/2011 \$ 1,580.36 CANTU, MARC 8/19/2011 \$ 1,580.36 CANTU, MARC 9/2/2011 \$ 1,580.36 CANTU, MARC 9/30/2011 \$ 1,580.36 CANTU, MARC 9/30/2011 \$ 1,580.36 CANTU, MARC 9/30/2011 \$ 1,580.36 CAREMARKPCS HEALTH LP 8/4/2011 \$ 15.90.34 CAREMARKPCS HEALTH LP 8/4/2011 \$ 732.31 CAREG NATERNATIONAL, INC. 7/5/2011 \$ 732.31 CARGO DATA 7/5/2011 \$ 732.31 CASH, WILLIAM 9/8/2011 \$ 472.77 CASH, WILLIAM 9/6/2011 \$ 472.77 CASH, WILLIAM 9/6/2011 \$ 72.76 CASH, WILLIAM 9/5/2011 \$ 72.76 CENTURYLINK 7/14/2011 \$ 2.796.79 CENTURYLINK 8/4/2011 \$ 2.796.79 CENTURYLINK 8/4/2011 \$ 2.796.79 CENTURYLINK 8/25/2011 \$ 2.796.79 CENTURYLINK 8/25/2011 \$ 2.296.79 CHAMBERLAIN COMMUNICATIONS, LLC	CANTU, MARC	7/8/2011	
CANTU, MARC 8/15/2011 \$ 1,580.36 CANTU, MARC 8/19/2011 \$ 1,580.36 CANTU, MARC 9/16/2011 \$ 1,580.36 CANTU, MARC 9/16/2011 \$ 2,646.98 CANTU, MARC 9/16/2011 \$ 62,646.98 CAREMARKPCS HEALTH LP 8/4/2011 \$ 61,459.34 CAREMARKPCS HEALTH LP 8/4/2011 \$ 61,459.34 CAREMARKPCS HEALTH LP 8/4/2011 \$ 61,459.34 CAREMARKPCS HEALTH LP 8/8/2011 \$ 61,459.34 CAREMARKPCS HEALTH LP 8/8/2011 \$ 732.31 CARGO DATA 9/8/2011 \$ 732.31 CARGO DATA 9/8/2011 \$ 72.35 CARIBBEAN DERNATIONAL, INC. 7/5/2011 \$ 3500.00 CASH, WILLIAM 9/2/2011 \$ 472.77 CASH, WILLIAM 9/16/2011 \$ 383.14 CASH, WILLIAM 9/16/2011 \$ 383.14 CASH, WILLIAM 9/16/2011 \$ 2279.67 CENTURYLINK 8/4/2011 \$ 2,795.67 CENTURYLINK 8/4/2011 \$ 2,795.67 CENTURYLINK 8/12/2011 \$ 72.90 CENTURYLINK 8/12/2011 \$ 72.90 CHAMBERLAIN COMMUNICATIONS, LLC 7/5/2011 \$ 30,220 CHAMBERLAIN COMMUNICATIONS 1/14/2011 \$ 1,332.35	CANTU, MARC	7/22/2011	1,588.51
CANTU, MARC 8/19/2011 \$ 1,580.36 CANTU, MARC 9/2/2011 \$ 1,580.36 CANTU, MARC 9/16/2011 \$ 1,580.36 CANTU, MARC 9/30/2011 \$ 2,646.98 CAREMARKPGS HEALTH LP 8/4/2011 \$ 11,590.34 CAREMARKPCS HEALTH LP 8/4/2011 \$ 61,459.42 CAREFY INTERNATIONAL, INC. 7/5/2011 \$ 732.31 CARGO DATA 9/8/2011 \$ 472.77 CASH, WILLIAM 9/6/2011 \$ 472.77 CASH, WILLIAM 9/6/2011 \$ 472.77 CASH, WILLIAM 9/6/2011 \$ 900.93 CENTURYLINK 7/14/2011 \$ 2,793.61 CENTURYLINK 8/4/2011 \$ 902.58 CENTURYLINK 8/18/2011 \$ 902.58 CENTURYLINK 8/18/2011 \$ 902.58 CENTURYLINK 7/14/2011 \$ 2,793.61 CENTURYLINK 8/18/2011 \$ 902.58 CENTURYLINK 8/18/2011 \$ 9	CANTU, MARC	8/5/2011	1,580.36
CANTU, MARC 9/2/2011 \$ 1,580.38 CANTU, MARC 9/16/2011 \$ 1,580.36 CANTU, MARC 9/30/2011 \$ 2,646.98 CAREMARKPCS HEALTH LP 8/4/2011 \$ 622.92 CAREMARKPCS HEALTH LP 8/4/2011 \$ 615.94 CARED INTERNATIONAL, INC. 7/5/2011 \$ 732.31 CARGO DATA 9/8/2011 \$ 1410.00 CASH, WILLIAM 9/2/2011 \$ 472.77 CASH, WILLIAM 9/16/2011 \$ 383.14 CASH, WILLIAM 9/16/2011 \$ 23.31 CENTURYLINK 9/30/2011 \$ 2793.61 CENTURYLINK 8/4/2011 \$ 2,793.61 CENTURYLINK 8/4/2011 \$ 96.26 CENTURYLINK 8/25/2011 \$ 729.60 CHAMBERLAIN COMMUNICATIONS, LLC 7/5/2011 \$ 30,200.00 CHAMBERLAIN COMMUNICATIONS, LLC 9/8/2011 \$ 21,277.1 CHARTER COMMUNICATIONS 1,142.2011 \$ 1,322.35 CHEN, MIN 7/9/2011 \$ 1,322.35 CHEN, MIN 9/8/2011 \$ 1,332.35 CHEN, MIN<	CANTU, MARC	8/15/2011	3,335.74
CANTU, MARC 9/16/2011 \$ 1,580.36 CANTU, MARC 9/30/2011 \$ 2,646.98 CAREMARKPCS HEALTH LP 8/4/2011 \$ 11,509.34 CAREMARKPCS HEALTH LP 8/4/2011 \$ 11,509.34 CAREMARKPCS HEALTH LP 8/4/2011 \$ 1,509.34 CAREY INTERNATIONAL, INC. 7/5/2011 \$ 722.31 CARGO DATA 9/8/2011 \$ 1,410.00 CARIBBEAN DERMATOLOGY SYMPOSIUM 7/7/2011 \$ 472.77 CASH, WILLIAM 9/6/2011 \$ 472.77 CASH, WILLIAM 9/16/2011 \$ 383.14 CASH, WILLIAM 9/30/2011 \$ 2796.79 CENTURYLINK 7/14/2011 \$ 2796.79 CENTURYLINK 8/18/2011 \$ 2793.61 CENTURYLINK 8/18/2011 \$ 729.60 CENTURYLINK 8/18/2011 \$ 729.60 CENTURYLINK 8/18/2011 \$ 729.60 CHAMBERLAIN COMMUNICATIONS, LLC 7/5/2011 \$ 729.60 CHAMBERLAIN COMMUNICATIONS 7/14/2011 \$ 2,809.41 CHARTER COMMUNICATIONS 7/14/2011 \$ 1,332.35 CHEN, MIN 7/8/2011 \$ 1,332.35 CHEN, MIN 9/2/2011 \$ 1,332.35 CHEN, MIN 9/2/2011 \$ 1,332.35 CHEN, MIN 9/30/2011 \$ 1,953.42 <td>CANTU, MARC</td> <td>8/19/2011</td> <td>\$ 1,580.36</td>	CANTU, MARC	8/19/2011	\$ 1,580.36
CANTU, MARC 9/30/2011 \$ 629.92 CAREMARKPCS HEALTH LP 8/4/2011 \$ 15,09.34 CAREMARKPCS HEALTH LP 8/4/2011 \$ 61,459.42 CAREMARKPCS HEALTH LP 8/4/2011 \$ 61,459.42 CAREY INTERNATIONAL, INC. 7/5/2011 \$ 73231 CARGO DATA 9/8/2011 \$ 1,410.00 CASH, WILLIAM 9/2/2011 \$ 472.77 CASH, WILLIAM 9/16/2011 \$ 383.14 CASH, WILLIAM 9/16/2011 \$ 30.90 CENTURYLINK 8/1/2011 \$ 2796.79 CENTURYLINK 8/1/2011 \$ 2796.79 CENTURYLINK 8/1/2011 \$ 30.020.00 CHAMBERLAIN COMMUNICATIONS, LLC 7/5/2011 \$ 30.020.00 CHAMBERLAIN COMMUNICATIONS, LLC 7/5/2011 \$ 30.020.00 CHARTER COMMUNICATIONS 8/25/2011 \$ 2,806.41 CHEN, MIN 7/8/2011 \$ 1,332.35 CHEN, MIN 7/8/2011 \$ 1,332.35 CHEN, MIN 9/10/2011 \$ 1,332.35 CHEN, MIN 9/10/2011 \$ 1,352.35	CANTU, MARC	9/2/2011	\$ 1,580.38
CANTU, MARC 9/30/2011 \$ 2,646.98 CAREMARKPCS HEALTH LP 8/4/2011 \$ 11,509.34 CAREMARKPCS HEALTH LP 8/4/2011 \$ 61,459.42 CAREY INTERNATIONAL, INC. 7/5/2011 \$ 732.31 CARGO DATA 9/8/2011 \$ 1,410.00 CASH, WILLIAM 9/2/2011 \$ 35,00.00 CASH, WILLIAM 9/16/2011 \$ 472.77 CASH, WILLIAM 9/16/2011 \$ 383.14 CASH, WILLIAM 9/30/2011 \$ 2,796.79 CENTURYLINK 8/1/2011 \$ 30,020.00 CHAMBERLAIN COMMUNICATIONS, LLC 7/5/2011 \$ 30,020.00 CHAMBERLAIN COMMUNICATIONS 8/25/2011 \$ 2,806.48 CHEN, MIN 7/2/2011 \$ 1,332.35 CHEN, MI	CANTU, MARC	9/16/2011	\$ 1,580.36
CAREMARKPCS HEALTH LP 8/4/2011 \$ 11,509.34 CAREW INTERNATIONAL, INC. 7/5/2011 \$ 732.31 CARGO DATA 9/8/2011 \$ 1,410.00 CASH, WILLIAM 9/8/2011 \$ 472.77 CASH, WILLIAM 9/6/2011 \$ 472.77 CASH, WILLIAM 9/16/2011 \$ 383.14 CASH, WILLIAM 9/16/2011 \$ 2,796.79 CASH, WILLIAM 9/16/2011 \$ 2,796.79 CASH, WILLIAM 9/16/2011 \$ 2,796.79 CENTURYLINK 7/14/2011 \$ 2,796.79 CENTURYLINK 8/4/2011 \$ 962.58 CENTURYLINK 8/4/2011 \$ 962.58 CENTURYLINK 8/25/2011 \$ 729.60 CHAMBERLAIN COMMUNICATIONS, LLC 7/5/2011 \$ 30,200.00 CHAMBERLAIN COMMUNICATIONS, LLC 9/8/2011 \$ 2,809.41 CHARTER COMMUNICATIONS 7/14/2011 \$ 2,809.41 CHEN, MIN 7/14/2011 \$ 1,332.35 CHEN, MIN 8/5/2011 \$ 1,332.35 CHEN, MIN 9/16/2011 \$ 1,953.42 CHEN, MI	CANTU, MARC	9/30/2011	
CAREMARKPCS HEALTH LIP 8/4/2011 \$ (1,459,42) CAREY INTERNATIONAL, INC. 7/5/2011 \$ 732,31 CARGO DATA 9/8/2011 \$ 1,410,00 CARIBBEAN DERMATOLOGY SYMPOSIUM 7/7/2011 \$ 3,500,00 CASH, WILLIAM 9/2/2011 \$ 472,77 CASH, WILLIAM 9/6/2011 \$ 472,77 CASH, WILLIAM 9/30/2011 \$ 900,93 CENTURYLINK 7/14/2011 \$ 2,796,67 CENTURYLINK 8/18/2011 \$ 2,796,67 CENTURYLINK 8/18/2011 \$ 962,58 CENTURYLINK 8/18/2011 \$ 90,020 CENTURYLINK 8/18/2011 \$ 90,020 CENTURYLINK 8/18/2011 \$ 2,796,67 CENTURYLINK 8/18/2011 \$ 20,020 CENTURYLINK 8/18/2011 \$ 20,020 CENTURYLINK 8/18/2011 \$ 20,020 CHAMBERLAIN COMMUNICATIONS, LLC 7/5/2011 \$ 30,020,00 CHAMBERLAIN COMMUNICATIONS 8/25/2011 \$ 2,808,48 CHEN, MIN 7/8/2011 \$ 1,332,35 CHEN, MI	CAREMARKPCS HEALTH LP	8/4/2011	
CAREMARKPCS HEALTH LP 8/4/2011 \$ 61,459.42 CARSCY INTERNATIONAL, INC. 775/2011 \$ 72.32 CARGO DATA 9/8/2011 \$ 1,410.00 CARIBBEAN DERMATOLOGY SYMPOSIUM 7/7/2011 \$ 3,500.00 CASH, WILLIAM 9/2/2011 \$ 472.77 CASH, WILLIAM 9/16/2011 \$ 383.14 CASH, WILLIAM 9/30/2011 \$ 00.93 CENTURYLINK 7/14/2011 \$ 2,793.67 CENTURYLINK 8/18/2011 \$ 00.200.00 CENTURYLINK 8/18/2011 \$ 962.58 CENTURYLINK 8/25/2011 \$ 729.60 CHAMBERLAIN COMMUNICATIONS, LLC 7/5/2011 \$ 30,020.00 CHAMBERLAIN COMMUNICATIONS, LLC 9/8/2011 \$ 2,808.48 CHEN, MIN 7/14/2011 \$ 1,332.35 CHEN, MIN 7/8/2011 \$ 1,332.35 CHEN, MIN 8/19/2011 \$ 1,953.41 CHEN, MIN 9/16/2011 \$ 1,953.41 CHEN, MIN 9/16/2011 \$ 1,953.41 CHEN, MIN 9/16/2011 \$ 1,953.42 CHEN, M	CAREMARKPCS HEALTH LP	8/4/2011	\$ 11,509.34
CAREY INTERNATIONAL, INC. 7/5/2011 \$ 1,410.00 CARGO DATA 9/8/2011 \$ 1,410.00 CARIBBEAN DERMATOLOGY SYMPOSIUM 7/7/2011 \$ 3,500.00 CASH, WILLIAM 9/2/2011 \$ 472.77 CASH, WILLIAM 9/6/2011 \$ 383.14 CASH, WILLIAM 9/16/2011 \$ 383.14 CASH, WILLIAM 9/30/2011 \$ 900.93 CENTURYLINK 7/14/2011 \$ 2,796.79 CENTURYLINK 8/18/2011 \$ 962.58 CENTURYLINK 8/18/2011 \$ 962.58 CENTURYLINK 8/25/2011 \$ 722.60 CENTURYLINK 8/25/2011 \$ 300.20 CHAMBERLAIN COMMUNICATIONS, LLC 7/5/2011 \$ 300.20 CHAMBERLAIN COMMUNICATIONS 7/14/2011 \$ 1,280.41 CHEN, MIN 7/18/2011 \$ 1,332.35 CHEN, MIN 8/2/2011 \$ 1,332.35 CHEN, MIN 9/2/2011 \$ 1,353.42 CHEN, MIN <td>CAREMARKPCS HEALTH LP</td> <td></td> <td></td>	CAREMARKPCS HEALTH LP		
CARGO DATA 9/8/2011 \$ 1,410.00 CARIBBEAN DERMATOLOGY SYMPOSIUM 7/7/2011 \$ 3,500.00 CASH, WILLIAM 9/2/2011 \$ 472.77 CASH, WILLIAM 9/16/2011 \$ 383.14 CASH, WILLIAM 9/16/2011 \$ 900.93 CENTURYLINK 7/14/2011 \$ 2,796.79 CENTURYLINK 8/4/2011 \$ 2,793.61 CENTURYLINK 8/18/2011 \$ 729.60 CENTURYLINK 8/25/2011 \$ 729.60 CHAMBERLAIN COMMUNICATIONS, LLC 7/5/2011 \$ 30,020.00 CHAMBERLAIN COMMUNICATIONS, LLC 7/5/2011 \$ 2,808.48 CHEN, MIN 7/8/2011 \$ 1,332.35 CHEN, MIN 7/8/2011 \$ 1,332.35 CHEN, MIN 7/8/2011 \$ 1,332.35 CHEN, MIN 9/2011 \$ 1,953.42 CHEN, MIN 9/16/2011 \$ 1,953.42 CHEN, MIN <t< td=""><td>CAREY INTERNATIONAL, INC.</td><td></td><td>·</td></t<>	CAREY INTERNATIONAL, INC.		·
CARIBBEAN DERMATOLOGY SYMPOSIUM 7/7/2011 \$ 3,500.00 CASH, WILLIAM 9/2/2011 \$ 472.77 CASH, WILLIAM 9/16/2011 \$ 383.14 CASH, WILLIAM 9/16/2011 \$ 900.93 CENTURYLINK 7/14/2011 \$ 2,796.79 CENTURYLINK 8/18/2011 \$ 962.58 CENTURYLINK 8/18/2011 \$ 962.58 CENTURYLINK 8/25/2011 \$ 729.60 CENTURYLINK 8/25/2011 \$ 30,020.00 CHAMBERLAIN COMMUNICATIONS, LLC 7/5/2011 \$ 30,020.00 CHAMBERLAIN COMMUNICATIONS 7/14/2011 \$ 2,809.41 CHARTER COMMUNICATIONS 7/14/2011 \$ 2,809.41 CHARTER COMMUNICATIONS 8/25/2011 \$ 1,332.35 CHEN, MIN 7/8/2011 \$ 1,332.35 CHEN, MIN 8/5/2011 \$ 1,332.35 CHEN, MIN 8/19/2011 \$ 1,953.42 CHEN, MIN 9/16/2011 \$ 1,953.42 CHEN, MIN 9/16/2011 \$ 1,953.42 CHEN, MIN 9/16/2011 \$ 1,953.42 CHEN, MIN<	CARGO DATA		
CASH, WILLIAM 9/2/2011 \$ 472.77 CASH, WILLIAM 9/6/2011 \$ 472.77 CASH, WILLIAM 9/16/2011 \$ 383.14 CASH, WILLIAM 9/30/2011 \$ 900.93 CENTURYLINK 7/14/2011 \$ 2,796.79 CENTURYLINK 8/4/2011 \$ 2,793.61 CENTURYLINK 8/18/2011 \$ 729.60 CHAMBERLAIN COMMUNICATIONS, LLC 7/5/2011 \$ 30,020.00 CHAMBERLAIN COMMUNICATIONS, LLC 7/5/2011 \$ 2,809.41 CHARTER COMMUNICATIONS 7/14/2011 \$ 2,809.41 CHARTER COMMUNICATIONS 8/25/2011 \$ 2,809.41 CHEN, MIN 7/8/2011 \$ 1,332.35 CHEN, MIN 7/2/2011 \$ 1,332.35 CHEN, MIN 8/15/2011 \$ 1,953.42 CHEN, MIN 9/16/2011 \$ 1,953.42 CHEN, MIN 9	CARIBBEAN DERMATOLOGY SYMPOSIUM	· ·	•
CASH, WILLIAM 9/16/2011 \$ 383.14 CASH, WILLIAM 9/30/2011 \$ 000.93 CENTURYLINK 7/14/2011 \$ 2,796.79 CENTURYLINK 8/4/2011 \$ 2,793.61 CENTURYLINK 8/18/2011 \$ 729.60 CENTURYLINK 8/25/2011 \$ 729.60 CHAMBERLAIN COMMUNICATIONS, LLC 7/8/2011 \$ 2,809.40 CHAMBERLAIN COMMUNICATIONS, LLC 9/8/2011 \$ 2,809.41 CHARTER COMMUNICATIONS 7/14/2011 \$ 2,809.41 CHARTER COMMUNICATIONS 7/14/2011 \$ 2,809.41 CHEN, MIN 7/8/2011 \$ 1,332.35 CHEN, MIN 7/8/2011 \$ 1,332.35 CHEN, MIN 8/5/2011 \$ 1,352.35 CHEN, MIN 9/2/2011 \$ 1,953.41 CHEN, MIN 9/2/2011 \$ 1,953.42 CHEN, MIN 9/16/2011 \$ 1,953.42 CHEN, MIN 9/30/2011 \$ 1,953.42 CHEN, MIN 9/30/2011 \$ 1,953.42 CHEN, MIN 9/30/2011 \$ 1,953.42 CHEN, MIN 9/30/20	CASH, WILLIAM		•
CASH, WILLIAM 9/16/2011 \$ 383.14 CASH, WILLIAM 9/30/2011 \$ 000.93 CENTURYLINK 7/14/2011 \$ 2,796.79 CENTURYLINK 8/4/2011 \$ 962.58 CENTURYLINK 8/18/2011 \$ 729.60 CENTURYLINK 8/25/2011 \$ 729.60 CHAMBERLAIN COMMUNICATIONS, LLC 7/5/2011 \$ 30,020.00 CHAMBERLAIN COMMUNICATIONS, LLC 9/8/2011 \$ 21,127.71 CHARTER COMMUNICATIONS 7/14/2011 \$ 2,809.41 CHARTER COMMUNICATIONS 7/14/2011 \$ 2,809.41 CHARTER COMMUNICATIONS 7/14/2011 \$ 2,809.41 CHARTER COMMUNICATIONS 7/14/2011 \$ 1,332.35 CHEN, MIN 7/22/2011 \$ 1,332.35 CHEN, MIN 8/5/2011 \$ 1,332.35 CHEN, MIN 9/2/2011 \$ 1,953.42 CHEN, MIN 9/2/2011 \$ 1,953.42 CHEN, MIN 9/16/2011 \$ 1,953.42 CHEN, MIN 9/30/2011 \$ 1,953.42 CHEN, MIN 9/30/2011 \$ 1,953.42 CHEN,	CASH, WILLIAM	· •	-
CASH, WILLIAM 9/30/2011 \$ 900.93 CENTURYLINK 7/14/2011 \$ 2,796.79 CENTURYLINK 8/4/2011 \$ 0,62.58 CENTURYLINK 8/18/2011 \$ 0,62.58 CENTURYLINK 8/15/2011 \$ 729.60 CHAMBERLAIN COMMUNICATIONS, LLC 7/5/2011 \$ 30,020.00 CHARBERLAIN COMMUNICATIONS, LLC 9/8/2011 \$ 2,809.41 CHARTER COMMUNICATIONS 7/14/2011 \$ 2,809.48 CHEN, MIN 7/8/2011 \$ 1,332.35 CHEN, MIN 7/8/2011 \$ 1,332.35 CHEN, MIN 8/5/2011 \$ 1,332.35 CHEN, MIN 8/19/2011 \$ 1,953.42 CHEN, MIN 9/2/2011 \$ 1,953.42 CHEN, MIN 9/2/2011 \$ 1,953.42 CHEN, MIN 9/2/2011 \$ 1,953.42 CHEN, MIN 9/30/2011	CASH, WILLIAM		
CENTURYLINK 7/14/2011 \$ 2,796.79 CENTURYLINK 8/4/2011 \$ 2,793.61 CENTURYLINK 8/18/2011 \$ 962.58 CENTURYLINK 8/25/2011 7.75/2010 CHAMBERLAIN COMMUNICATIONS, LLC 7/5/2011 \$ 30,020.00 CHAMBERLAIN COMMUNICATIONS, LLC 9/8/2011 \$ 2,809.41 CHARTER COMMUNICATIONS 7/14/2011 \$ 2,809.41 CHARTER COMMUNICATIONS 8/25/2011 \$ 2,808.48 CHEN, MIN 7/8/2011 \$ 1,332.35 CHEN, MIN 7/22/2011 \$ 1,332.35 CHEN, MIN 8/5/2011 \$ 1,332.35 CHEN, MIN 9/2/2011 \$ 1,352.35 CHEN, MIN 9/16/2011 \$ 1,953.42 CHEN, MIN 9/16/20	CASH, WILLIAM		
CENTURYLINK 8/4/2011 \$ 2,793.61 CENTURYLINK 8/18/2011 \$ 962.58 CENTURYLINK 8/25/2011 \$ 729.60 CHAMBERLAIN COMMUNICATIONS, LLC 7/5/2011 \$ 30,020.00 CHAMBERLAIN COMMUNICATIONS, LLC 9/8/2011 \$ 21,127.71 CHARTER COMMUNICATIONS 7/14/2011 \$ 2,809.41 CHEN, MIR 7/8/2011 \$ 1,332.35 CHEN, MIN 7/8/2011 \$ 1,332.35 CHEN, MIN 7/22/2011 \$ 1,332.35 CHEN, MIN 8/19/2011 \$ 1,953.42 CHEN, MIN 9/2/2011 \$ 1,953.42 CHEN, MIN 9/2/2011 \$ 1,953.42 CHEN, MIN 9/16/2011 \$ 1,953.42 CHEN, MIN 9/10/2011 \$ 1,953.41 CIGNA HEALTHCARE, INC. 8/11/2011<			
CENTURYLINK 8/18/2011 \$ 962.58 CENTURYLINK 8/25/2011 \$ 729.60 CHAMBERLAIN COMMUNICATIONS, LLC 7/5/2011 \$ 30,020.00 CHAMBERLAIN COMMUNICATIONS, LLC 9/8/2011 \$ 21,127.71 CHARTER COMMUNICATIONS 7/14/2011 \$ 2,809.41 CHARTER COMMUNICATIONS 8/25/2011 \$ 2,809.48 CHEN, MIN 7/8/2011 \$ 1,332.35 CHEN, MIN 8/5/2011 \$ 1,332.35 CHEN, MIN 8/19/2011 \$ 1,332.35 CHEN, MIN 8/19/2011 \$ 1,953.41 CHEN, MIN 9/2/2011 \$ 1,953.42 CHEN, MIN 9/16/2011 \$ 1,953.42 CHEN, MIN 9/16/2011<	CENTURYLINK		
CENTURYLINK 8/25/2011 \$ 729.60 CHAMBERLAIN COMMUNICATIONS, LLC 7/5/2011 \$ 30,020.00 CHAMBERLAIN COMMUNICATIONS, LLC 9/8/2011 \$ 21,127.71 CHARTER COMMUNICATIONS 7/14/2011 \$ 2,809.41 CHARTER COMMUNICATIONS 8/25/2011 \$ 1,332.35 CHEN, MIN 7/8/2011 \$ 1,332.35 CHEN, MIN 8/5/2011 \$ 1,332.35 CHEN, MIN 8/19/2011 \$ 1,332.35 CHEN, MIN 9/2/2011 \$ 1,953.41 CHEN, MIN 9/16/2011 \$ 1,953.42 CIGNA HEALTHCARE, INC. <	CENTURYLINK		· ·
CHAMBERLAIN COMMUNICATIONS, LLC 7/5/2011 \$ 30,020.00 CHAMBERLAIN COMMUNICATIONS, LLC 9/8/2011 \$ 21,127.71 CHARTER COMMUNICATIONS 7/14/2011 \$ 2,809.41 CHARTER COMMUNICATIONS 8/25/2011 \$ 2,808.48 CHEN, MIN 7/8/2011 \$ 1,332.35 CHEN, MIN 8/5/2011 \$ 1,332.35 CHEN, MIN 8/5/2011 \$ 1,332.35 CHEN, MIN 9/2/2011 \$ 1,953.41 CHEN, MIN 9/16/2011 \$ 1,953.42 CHEN, MIN 9/30/2011 \$ 1,953.42 CHEN, MIN 9/16/2011 \$ 1,953.42 CHEN, MIN 9/16/2011 \$ 1,953.41 CIGNA HEALTHCARE, INC. 8/11/2011 \$ 15,476.58 CIGNA HEALTHCARE, INC. <	CENTURYLINK		
CHAMBERLAIN COMMUNICATIONS, LLC 9/8/2011 \$ 21,127.71 CHARTER COMMUNICATIONS 7/14/2011 \$ 2,809.41 CHARTER COMMUNICATIONS 8/25/2011 \$ 2,808.48 CHEN, MIN 7/8/2011 \$ 1,332.35 CHEN, MIN 7/22/2011 \$ 1,332.35 CHEN, MIN 8/5/2011 \$ 1,953.41 CHEN, MIN 9/2/2011 \$ 1,953.41 CHEN, MIN 9/2/2011 \$ 1,953.42 CHEN, MIN 9/16/2011 \$ 1,953.42 CHEN, MIN 9/16/2011 \$ 1,953.42 CHEN, MIN 9/16/2011 \$ 1,953.41 CIGNA HEALTHCARE, INC. 8/11/2011 \$ 15,765.58 CIGNA HEALTHCARE, INC. 8/11/2011 \$ 17,948.50 CINGULAR 7/14/2011 \$ 33,621.26 CINGULAR 9/15/2011 \$ 34,018.35 CINTAS CORPORATION 8/11/2011 \$ 489.68 CINTAS CORPORATION 8/11/2011 \$ 244.84 CINTAS DOCUMENT MANAGEMENT 7/14/2011 \$ 128.95 CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00 CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00 <td>CHAMBERLAIN COMMUNICATIONS, LLC</td> <td></td> <td></td>	CHAMBERLAIN COMMUNICATIONS, LLC		
CHARTER COMMUNICATIONS 7/14/2011 \$ 2,809.41 CHARTER COMMUNICATIONS 8/25/2011 \$ 2,808.48 CHEN, MIN 7/8/2011 \$ 1,332.35 CHEN, MIN 7/22/2011 \$ 1,332.35 CHEN, MIN 8/5/2011 \$ 1,332.35 CHEN, MIN 8/19/2011 \$ 1,953.41 CHEN, MIN 9/2/2011 \$ 1,953.42 CHEN, MIN 9/16/2011 \$ 1,953.42 CHEN, MIN 9/30/2011 \$ 1,953.42 CHEN, MIN 9/30/2011 \$ 1,953.41 CIGNA HEALTHCARE, INC. 8/11/2011 \$ 15,476.58 CIGNA HEALTHCARE, INC. 8/11/2011 \$ 15,476.58 CIGNA HEALTHCARE, INC. 8/11/2011 \$ 17,948.50 CINGULAR 7/14/2011 \$ 33,629.47 CINGULAR 7/14/2011 \$ 34,636.19 CINGULAR 9/15/2011 \$ 34,018.35 CINTAS CORPORATION 8/11/2011 \$ 489.68 CINTAS CORPORATION 9/8/2011 \$ 244.84 CINTAS DOCUMENT MANAGEMENT 7/14/2011 \$ 128.95 CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00 CINTAS D	CHAMBERLAIN COMMUNICATIONS, LLC		
CHARTER COMMUNICATIONS 8/25/2011 \$ 2,808.48 CHEN, MIN 7/8/2011 \$ 1,332.35 CHEN, MIN 7/22/2011 \$ 1,332.35 CHEN, MIN 8/5/2011 \$ 1,953.41 CHEN, MIN 9/2/2011 \$ 1,953.42 CHEN, MIN 9/2/2011 \$ 1,953.42 CHEN, MIN 9/30/2011 \$ 1,953.42 CHEN, MIN 9/30/2011 \$ 1,953.42 CHEN, MIN 9/30/2011 \$ 1,953.42 CIGNA HEALITHCARE, INC. 8/11/2011 \$ 15,476.58 CIGNA HEALITHCARE, INC. 8/11/2011 \$ 17,948.50 CINGULAR 7/14/2011 \$ 33,629.47 CINGULAR 7/14/2011 \$ 33,621.26 CINGULAR 9/15/2011 \$ 34,036.19 CINTAS CORPORATION 8/11/2011 \$ 489.68 CINTAS CORPORATION 8/11/2011 \$ 244.84 CINTAS DOCUMENT MANAGEMENT 7/14/2011 \$ 128.95 CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00 CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00 CINTAS DOCUMENT MANAGEMENT 9/8/2011 \$ 55.00	CHARTER COMMUNICATIONS	7/14/2011	· ·
CHEN, MIN 7/8/2011 \$ 1,332.35 CHEN, MIN 7/22/2011 \$ 1,332.35 CHEN, MIN 8/5/2011 \$ 1,332.35 CHEN, MIN 8/19/2011 \$ 1,953.41 CHEN, MIN 9/2/2011 \$ 1,953.42 CHEN, MIN 9/16/2011 \$ 1,953.42 CHEN, MIN 9/30/2011 \$ 1,953.42 CHEN, MIN 9/30/2011 \$ 1,953.41 CIGNA HEALTHCARE, INC. 8/11/2011 \$ 15,476.58 CIGNA HEALTHCARE, INC. 8/11/2011 \$ 17,948.50 CINGULAR 7/14/2011 \$ 33,621.26 CINGULAR 7/14/2011 \$ 33,621.26 CINGULAR 9/15/2011 \$ 33,621.26 CINTAS CORPORATION 9/15/2011 \$ 34,018.35 CINTAS CORPORATION 8/11/2011 \$ 244.84 CINTAS DOCUMENT MANAGEMENT 7/14/2011 \$ 128.95 CINTAS DOCUMENT MANAGEMENT 8/4/2011 \$ 157.95 CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00 CINTAS DOCUMENT MANAGEMENT 9/8/2011 \$ 55.00	CHARTER COMMUNICATIONS	•	•
CHEN, MIN 7/22/2011 \$ 1,332.35 CHEN, MIN 8/5/2011 \$ 1,332.35 CHEN, MIN 8/19/2011 \$ 1,953.41 CHEN, MIN 9/2/2011 \$ 1,953.42 CHEN, MIN 9/16/2011 \$ 1,953.42 CHEN, MIN 9/30/2011 \$ 1,953.41 CIGNA HEALTHCARE, INC. 8/11/2011 \$ 15,476.58 CIGNA HEALTHCARE, INC. 8/11/2011 \$ 17,948.50 CINGULAR 7/14/2011 \$ 33,621.26 CINGULAR 9/15/2011 \$ 33,621.26 CINGULAR 9/15/2011 \$ 34,018.35 CINTAS CORPORATION 9/15/2011 \$ 34,018.35 CINTAS CORPORATION 9/8/2011 \$ 244.84 CINTAS DOCUMENT MANAGEMENT 7/14/2011 \$ 128.95 CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00 CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00 CINTAS DOCUMENT MANAGEMENT 9/8/2011 \$ 55.00 CINTAS DOCUMENT MANAGEMENT 9/8/2011 \$ 55.00 CINTAS DOCUMENT MANAGEMENT 9/8/2011 \$ 55.00	CHEN, MIN		
CHEN, MIN 8/5/2011 \$ 1,332.35 CHEN, MIN 8/19/2011 \$ 1,953.41 CHEN, MIN 9/2/2011 \$ 1,953.42 CHEN, MIN 9/16/2011 \$ 1,953.42 CHEN, MIN 9/30/2011 \$ 1,953.41 CIGNA HEALTHCARE, INC. 8/11/2011 \$ 15,476.58 CIGNA HEALTHCARE, INC. 8/11/2011 \$ 17,948.50 CINGULAR 7/14/2011 \$ 33,629.47 CINGULAR 7/14/2011 \$ 33,621.26 CINGULAR 9/15/2011 \$ 33,621.26 CINGULAR 9/15/2011 \$ 34,018.35 CINTAS CORPORATION 7/5/2011 \$ 489.68 CINTAS CORPORATION 8/11/2011 \$ 244.84 CINTAS DOCUMENT MANAGEMENT 7/14/2011 \$ 128.95 CINTAS DOCUMENT MANAGEMENT 8/4/2011 \$ 157.95 CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00 CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00 CINTAS DOCUMENT MANAGEMENT 9/8/2011 \$ 55.00	CHEN, MIN		=
CHEN, MIN 8/19/2011 \$ 1,953.41 CHEN, MIN 9/2/2011 \$ 1,953.42 CHEN, MIN 9/16/2011 \$ 1,953.42 CHEN, MIN 9/30/2011 \$ 1,953.41 CIGNA HEALTHCARE, INC. 8/11/2011 \$ 15,476.58 CIGNA HEALTHCARE, INC. 8/11/2011 \$ 17,948.50 CINGULAR 7/14/2011 \$ 33,629.47 CINGULAR 7/14/2011 \$ 34,636.19 CINGULAR 9/15/2011 \$ 33,621.26 CINGULAR 9/15/2011 \$ 34,018.35 CINTAS CORPORATION 7/5/2011 \$ 489.68 CINTAS CORPORATION 8/11/2011 \$ 244.84 CINTAS DOCUMENT MANAGEMENT 7/14/2011 \$ 128.95 CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00 CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00 CINTAS DOCUMENT MANAGEMENT 9/8/2011 \$ 55.00	CHEN, MIN		·
CHEN, MIN 9/2/2011 \$ 1,953.42 CHEN, MIN 9/16/2011 \$ 1,953.42 CHEN, MIN 9/30/2011 \$ 1,953.41 CIGNA HEALTHCARE, INC. 8/11/2011 \$ 15,476.58 CIGNA HEALTHCARE, INC. 8/11/2011 \$ 17,948.50 CINGULAR 7/14/2011 \$ 33,629.47 CINGULAR 7/14/2011 \$ 34,636.19 CINGULAR 9/15/2011 \$ 33,621.26 CINGULAR 9/15/2011 \$ 34,018.35 CINTAS CORPORATION 7/5/2011 \$ 489.68 CINTAS CORPORATION 8/11/2011 \$ 244.84 CINTAS CORPORATION 9/8/2011 \$ 244.84 CINTAS DOCUMENT MANAGEMENT 7/14/2011 \$ 128.95 CINTAS DOCUMENT MANAGEMENT 8/4/2011 \$ 55.00 CINTAS DOCUMENT MANAGEMENT 9/8/2011 \$ 55.00			
CHEN, MIN 9/16/2011 \$ 1,953.42 CHEN, MIN 9/30/2011 \$ 1,953.41 CIGNA HEALTHCARE, INC. 8/11/2011 \$ 15,476.58 CIGNA HEALTHCARE, INC. 8/11/2011 \$ 17,948.50 CINGULAR 7/14/2011 \$ 33,629.47 CINGULAR 7/14/2011 \$ 34,636.19 CINGULAR 9/15/2011 \$ 33,621.26 CINGULAR 9/15/2011 \$ 34,018.35 CINTAS CORPORATION 7/5/2011 \$ 489.68 CINTAS CORPORATION 8/11/2011 \$ 244.84 CINTAS CORPORATION 9/8/2011 \$ 244.84 CINTAS DOCUMENT MANAGEMENT 7/14/2011 \$ 128.95 CINTAS DOCUMENT MANAGEMENT 8/4/2011 \$ 157.95 CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00 CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00	CHEN, MIN		•
CHEN, MIN 9/30/2011 \$ 1,953.41 CIGNA HEALTHCARE, INC. 8/11/2011 \$ 15,476.58 CIGNA HEALTHCARE, INC. 8/11/2011 \$ 17,948.50 CINGULAR 7/14/2011 \$ 33,629.47 CINGULAR 7/14/2011 \$ 34,636.19 CINGULAR 9/15/2011 \$ 33,621.26 CINTAS CORPORATION 7/5/2011 \$ 489.68 CINTAS CORPORATION 8/11/2011 \$ 244.84 CINTAS CORPORATION 9/8/2011 \$ 244.84 CINTAS DOCUMENT MANAGEMENT 7/14/2011 \$ 128.95 CINTAS DOCUMENT MANAGEMENT 8/4/2011 \$ 157.95 CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00 CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00	CHEN, MIN	9/16/2011	
CIGNA HEALTHCARE, INC. 8/11/2011 \$ 15,476.58 CIGNA HEALTHCARE, INC. 8/11/2011 \$ 17,948.50 CINGULAR 7/14/2011 \$ 33,629.47 CINGULAR 7/14/2011 \$ 34,636.19 CINGULAR 9/15/2011 \$ 33,621.26 CINGULAR 9/15/2011 \$ 34,018.35 CINTAS CORPORATION 7/5/2011 \$ 489.68 CINTAS CORPORATION 8/11/2011 \$ 244.84 CINTAS CORPORATION 9/8/2011 \$ 244.84 CINTAS DOCUMENT MANAGEMENT 7/14/2011 \$ 128.95 CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00 CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00 CINTAS DOCUMENT MANAGEMENT 9/8/2011 \$ 55.00			
CIGNA HEALTHCARE, INC. 8/11/2011 \$ 17,948.50 CINGULAR 7/14/2011 \$ 33,629.47 CINGULAR 7/14/2011 \$ 34,636.19 CINGULAR 9/15/2011 \$ 33,621.26 CINGULAR 9/15/2011 \$ 34,018.35 CINTAS CORPORATION 7/5/2011 \$ 489.68 CINTAS CORPORATION 8/11/2011 \$ 244.84 CINTAS CORPORATION 9/8/2011 \$ 244.84 CINTAS DOCUMENT MANAGEMENT 7/14/2011 \$ 128.95 CINTAS DOCUMENT MANAGEMENT 8/4/2011 \$ 157.95 CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00 CINTAS DOCUMENT MANAGEMENT 9/8/2011 \$ 55.00	CIGNA HEALTHCARE, INC.		
CINGULAR 7/14/2011 \$ 33,629.47 CINGULAR 7/14/2011 \$ 34,636.19 CINGULAR 9/15/2011 \$ 33,621.26 CINGULAR 9/15/2011 \$ 34,018.35 CINTAS CORPORATION 7/5/2011 \$ 489.68 CINTAS CORPORATION 8/11/2011 \$ 244.84 CINTAS CORPORATION 9/8/2011 \$ 244.84 CINTAS DOCUMENT MANAGEMENT 7/14/2011 \$ 128.95 CINTAS DOCUMENT MANAGEMENT 8/4/2011 \$ 157.95 CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00 CINTAS DOCUMENT MANAGEMENT 9/8/2011 \$ 55.00	CIGNA HEALTHCARE, INC.	8/11/2011	
CINGULAR 7/14/2011 \$ 34,636.19 CINGULAR 9/15/2011 \$ 33,621.26 CINGULAR 9/15/2011 \$ 34,018.35 CINTAS CORPORATION 7/5/2011 \$ 489.68 CINTAS CORPORATION 8/11/2011 \$ 244.84 CINTAS CORPORATION 9/8/2011 \$ 244.84 CINTAS DOCUMENT MANAGEMENT 7/14/2011 \$ 128.95 CINTAS DOCUMENT MANAGEMENT 8/4/2011 \$ 157.95 CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00 CINTAS DOCUMENT MANAGEMENT 9/8/2011 \$ 55.00	CINGULAR		
CINGULAR 9/15/2011 \$ 33,621.26 CINGULAR 9/15/2011 \$ 34,018.35 CINTAS CORPORATION 7/5/2011 \$ 489.68 CINTAS CORPORATION 8/11/2011 \$ 244.84 CINTAS CORPORATION 9/8/2011 \$ 244.84 CINTAS DOCUMENT MANAGEMENT 7/14/2011 \$ 128.95 CINTAS DOCUMENT MANAGEMENT 8/4/2011 \$ 157.95 CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00 CINTAS DOCUMENT MANAGEMENT 9/8/2011 \$ 55.00	CINGULAR	7/14/2011	·
CINGULAR 9/15/2011 \$ 34,018.35 CINTAS CORPORATION 7/5/2011 \$ 489.68 CINTAS CORPORATION 8/11/2011 \$ 244.84 CINTAS CORPORATION 9/8/2011 \$ 244.84 CINTAS DOCUMENT MANAGEMENT 7/14/2011 \$ 128.95 CINTAS DOCUMENT MANAGEMENT 8/4/2011 \$ 157.95 CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00 CINTAS DOCUMENT MANAGEMENT 9/8/2011 \$ 55.00	CINGULAR		-
CINTAS CORPORATION 7/5/2011 \$ 489.68 CINTAS CORPORATION 8/11/2011 \$ 244.84 CINTAS CORPORATION 9/8/2011 \$ 244.84 CINTAS DOCUMENT MANAGEMENT 7/14/2011 \$ 128.95 CINTAS DOCUMENT MANAGEMENT 8/4/2011 \$ 157.95 CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00 CINTAS DOCUMENT MANAGEMENT 9/8/2011 \$ 55.00	CINGULAR		•
CINTAS CORPORATION 8/11/2011 \$ 244.84 CINTAS CORPORATION 9/8/2011 \$ 244.84 CINTAS DOCUMENT MANAGEMENT 7/14/2011 \$ 128.95 CINTAS DOCUMENT MANAGEMENT 8/4/2011 \$ 157.95 CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00 CINTAS DOCUMENT MANAGEMENT 9/8/2011 \$ 55.00	CINTAS CORPORATION		
CINTAS CORPORATION 9/8/2011 \$ 244.84 CINTAS DOCUMENT MANAGEMENT 7/14/2011 \$ 128.95 CINTAS DOCUMENT MANAGEMENT 8/4/2011 \$ 157.95 CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00 CINTAS DOCUMENT MANAGEMENT 9/8/2011 \$ 55.00	CINTAS CORPORATION	• •	
CINTAS DOCUMENT MANAGEMENT 7/14/2011 \$ 128.95 CINTAS DOCUMENT MANAGEMENT 8/4/2011 \$ 157.95 CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00 CINTAS DOCUMENT MANAGEMENT 9/8/2011 \$ 55.00	CINTAS CORPORATION		
CINTAS DOCUMENT MANAGEMENT 8/4/2011 \$ 157.95 CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00 CINTAS DOCUMENT MANAGEMENT 9/8/2011 \$ 55.00	CINTAS DOCUMENT MANAGEMENT		
CINTAS DOCUMENT MANAGEMENT 8/11/2011 \$ 55.00 CINTAS DOCUMENT MANAGEMENT 9/8/2011 \$ 55.00			
CINTAS DOCUMENT MANAGEMENT 9/8/2011 \$ 55.00	CINTAS DOCUMENT MANAGEMENT		
	CINTAS DOCUMENT MANAGEMENT		
	CINTAS DOCUMENT MANAGEMENT		

Name of creditor	Payment date	Amount paid
CISION US, INC.	7/14/2011 \$	4,932.93
CISION US, INC.	8/18/2011 \$	
CISTONE, ROSEMARIE	7/5/2011	
CISTONE, ROSEMARIE	7/8/2011	
CISTONE, ROSEMARIE	7/22/2011	
CISTONE, ROSEMARIE	7/25/2011	•
CISTONE, ROSEMARIE	8/5/2011	
CISTONE, ROSEMARIE	8/15/2011	
CISTONE, ROSEMARIE	8/19/2011	•
CISTONE, ROSEMARIE	9/2/2011	'
CISTONE, ROSEMARIE	9/16/2011	•
CISTONE, ROSEMARIE	9/30/2011	
CLEVELAND CLINIC FLORIDA-WPSA	9/1/2011	,
COLON, EVA	7/22/2011	
COLON, EVA	8/5/2011	•
COLON, EVA	8/19/2011	•
COLON, EVA	8/25/2011	
COLON, EVA	9/2/2011	
COLON, EVA	9/16/2011	•
COLON, EVA	9/30/2011	•
COMCAST CABLE	7/14/2011	•
COMCAST CABLE	8/18/2011	
COMM. OF PA	9/15/2011	
COMMISSIONER OF SOCIAL SERVICES	9/15/2011	
COMPREHENSIVE NEUROSCIENCE	7/21/2011	•
COMPREHENSIVE NEUROSCIENCE	8/11/2011	•
COMPREHENSIVE NEUROSCIENCE	9/8/2011	•
CONRAD, MICHAEL ROGER	7/8/2011	·
CONRAD, MICHAEL ROGER	7/22/2011	•
CONRAD, MICHAEL ROGER	8/5/2011	·
CONRAD, MICHAEL ROGER	8/19/2011	•
CONRAD, MICHAEL ROGER	9/2/2011	•
CONRAD, MICHAEL ROGER	9/16/2011	· ,
CONRAD, MR MICHAEL	7/11/2011	•
COOPER, TAMI	9/2/2011	
COOPER, TAMI	9/6/2011	
COOPER, TAMI	9/16/2011	
COOPER, TAMI	9/30/2011	
CORPORACION SANPUE S. A	9/21/2011	
COSMO PAGANO	7/5/2011	·
COUPE, MS VALERIE	7/25/2011	
COUPE, MS VALERIE	8/8/2011	
COUPE, VALERIE A	7/8/2011	
COUPE, VALERIE A	7/22/2011	
COUPE, VALERIE A	8/5/2011	
COUPE, VALERIE A	8/15/2011	
COUPE, VALERIE A	8/19/2011	
COUPE, VALERIE A	9/2/2011	
COUPE, VALERIE A	9/16/2011	
COUPE, VALERIE A	9/30/2011	
COURINGTON, JULIE E	7/8/2011	
COURINGTON, JULIE E	7/22/2011	
COURINGTON, JULIE E	8/5/2011	
COURINGTON, JULIE E	8/15/2011	
<i>'</i> .	5, 15, 2511	¥ 3,723.14

Name of creditor	Payment date	Amount paid
COURINGTON, JULIE E	8/19/2011	2,036.01
COURINGTON, JULIE E	9/2/2011	•
COURINGTON, JULIE E	9/16/2011	•
COURINGTON, JULIE E	9/30/2011	•
COURINGTON, MS JULIE	7/25/2011	•
COURINGTON, MS JULIE	8/15/2011	
COURINGTON, MS JULIE	9/26/2011	
COWGILL, BRENT	7/8/2011	
COWGILL, BRENT	7/22/2011	·
COWGILL, BRENT	8/5/2011	
COWGILL, BRENT	8/19/2011	·
COWGILL, BRENT	9/2/2011	•
COWGILL, BRENT	9/16/2011	
COWGILL, BRENT	9/30/2011	
COWIN, STEPHEN	7/5/2011	·
COWIN, STEPHEN	8/8/2011	
COWIN, STEPHEN A.	7/8/2011	
COWIN, STEPHEN A.	7/22/2011	
COWIN, STEPHEN A.	8/15/2011	•
CROCKER, JOEL	7/5/2011	
CROCKER, JOEL	7/25/2011	
CROCKER, JOEL	8/8/2011	
CROCKER, JOEL	8/15/2011	
CROCKER, JOEL	8/31/2011	
CROCKER, JOEL	9/15/2011	
CROCKER, JOEL E	7/8/2011	
CROCKER, JOEL E	7/22/2011	
CROCKER, JOEL E	8/5/2011	•
CROCKER, JOEL E	8/15/2011	
CROCKER, JOEL E	8/19/2011	•
CROCKER, JOEL E	9/2/2011	
CROCKER, JOEL E	9/16/2011	
CROCKER, JOEL E	9/30/2011	
CRYSTAL SPRINGS WATER CO.	7/5/2011	
CRYSTAL SPRINGS WATER CO.	7/14/2011	
CRYSTAL SPRINGS WATER CO.	7/21/2011	
CRYSTAL SPRINGS WATER CO.	8/18/2011	•
CRYSTAL SPRINGS WATER CO.	8/25/2011	
CT CORPORATION	8/11/2011	
CT CORPORATION	9/1/2011	
CURI, ANGELA	7/8/2011	
CURI, ANGELA	7/22/2011	
CURI, ANGELA	7/25/2011	
CURI, ANGELA	8/5/2011	
CURI, ANGELA	8/15/2011	·
CURI, ANGELA	8/15/2011	
CURI, ANGELA	8/19/2011	
CURI, ANGELA	9/2/2011	•
CURI, ANGELA	9/8/2011	•
CURI, ANGELA	9/16/2011	
CURI, ANGELA	9/30/2011	•
CVETKOVSKI, DEJAN	7/8/2011	
CVETKOVSKI, DEJAN	7/22/2011	•
CVETKOVSKI, DEJAN	8/5/2011	•
CYLLINO VOIM, DEJIMY	0/ 3/ 2011	9 1,004.02

Name of creditor	Payment date	Amount paid
CVETKOVSKI, DEJAN	8/19/2011	\$ 1,664.62
CVETKOVSKI, DEJAN	8/25/2011	
CVETKOVSKI, DEJAN	9/2/2011	\$ 1,664.62
CVETKOVSKI, DEJAN	9/16/2011	•
CVETKOVSKI, DEJAN	9/30/2011	
D'AIELLO, NICHOLAS	7/5/2011	
D'AIELLO, NICHOLAS	7/25/2011	
D'AIELLO, NICHOLAS	8/8/2011	
D'AIELLO, NICHOLAS	8/25/2011	
D'AIELLO, NICHOLAS	9/8/2011	
D'AIELLO, NICHOLAS V.	7/8/2011	
D'AIELLO, NICHOLAS V.	7/22/2011	· ·
D'AIELLO, NICHOLAS V.	8/5/2011	•
D'AIELLO, NICHOLAS V.	8/15/2011	
D'AIELLO, NICHOLAS V.	8/19/2011	
D'AIELLO, NICHOLAS V.	9/2/2011	
D'AIELLO, NICHOLAS V.	9/16/2011	
D'AIELLO, NICHOLAS V.	9/30/2011	-
DALIA, BELAL	7/8/2011	-
DALIA, BELAL	7/22/2011	•
DALIA, BELAL	8/5/2011	•
DALIA, BELAL	8/15/2011	•
DALIA, BELAL	8/19/2011	-
DALIA, BELAL	9/2/2011	
DALIA, BELAL	9/16/2011	•
DALIA, BELAL	9/30/2011	
DALY, AMBER	9/16/2011	
DALY, AMBER	9/30/2011	
DATA ARCHIVES RECORDS MANAGEMENT LLC	7/5/2011	-
DATA ARCHIVES RECORDS MANAGEMENT LLC	8/4/2011	
DAVIES, THOMAS J	7/21/2011	
DAVIES, THOMAS J	9/8/2011	
DAVIES, THOMAS J	9/21/2011	•
DAVIS, DENISE	7/8/2011	
DAVIS, DENISE	7/22/2011	·
DAVIS, DENISE	8/5/2011	
DAVIS, DENISE	8/19/2011	
DAVIS, DENISE	9/2/2011	•
DAVIS, DENISE	9/16/2011	
DAVIS, DENISE	9/30/2011	•
DAWSON, JEFFREY	7/8/2011	•
DAWSON, JEFFREY	7/22/2011	·
DAWSON, JEFFREY	8/5/2011	
DAWSON, JEFFREY	8/15/2011	•
DAWSON, JEFFREY	8/19/2011	
DAWSON, JEFFREY	9/2/2011	
DAWSON, JEFFREY	9/16/2011	
DAWSON, JEFFREY		
DELLACONA JR, MICHAEL	9/30/20 1 1 7/8/2011	,
DELLACONA JR, MICHAEL	• •	•
DELLACONA JR, MICHAEL DELLACONA JR, MICHAEL	7/22/2011	•
DELLACONA JR, MICHAEL DELLACONA JR, MICHAEL	8/5/2011 8/15/2011	
DELLACONA JR, MICHAEL	8/15/2011	•
- · · · · · · · · · · · · · · · · · · ·	8/19/2011	
DELLACONA JR, MICHAEL	9/2/2011	\$ 2,802.45

Name of creditor	Payment date	Amount	paid
DELLACONA JR, MICHAEL	9/16/2011	\$	2,802.44
DELLACONA JR, MICHAEL	9/30/2011		6,312.13
DELLACONA, MICHAEL	7/5/2011	\$	707.01
DELLACONA, MICHAEL	8/8/2011		606.31
DELLACONA, MICHAEL	8/15/2011		331.03
DELLACONA, MICHAEL	8/25/2011		466.15
DELLACONA, MICHAEL	9/21/2011		533.50
DELLACONA, MICHAEL	9/26/2011		617.84
DENGLER, THERESA I	7/8/2011		1,432.79
DENGLER, THERESA I	7/22/2011		1,432.77
DENGLER, THERESA I	8/5/2011	\$	1,432.78
DENGLER, THERESA I	8/19/2011	\$	1,432.78
DENGLER, THERESA I	9/2/2011		1,432.79
DENGLER, THERESA I	9/16/2011		1,432.78
DENGLER, THERESA I	9/30/2011		1,432.78
DEPT OF LABOR & INDUSTRIES	7/14/2011	\$	78.42
DEPT, OF SOCIAL & HEALTH SERVICES	7/21/2011		3,553.20
DEPT. OF VETERANS AFFAIRS	8/19/2011		1,692.46
DER, TOMMY WAYNE	7/8/2011		2,586.99
DER, TOMMY WAYNE	7/22/2011		2,587.01
DER, TOMMY WAYNE	8/5/2011		2,586.99
DER, TOMMY WAYNE	8/19/2011		2,587.01
DER, TOMMY WAYNE	9/2/2011		2,586.99
DER, TOMMY WAYNE	9/16/2011		2,587.01
DER, TOMMY WAYNE	9/30/2011		2,586.99
DERBAUM, TODD	7/5/2011		168.60
DERBAUM, TODD	7/8/2011		2,167.52
DERBAUM, TODD	7/22/2011		2,167.52
DERBAUM, TODD	7/25/2011		93.00
DERBAUM, TODD	8/5/2011		2,167.52
DERBAUM, TODD	8/15/2011		173.94
DERBAUM, TODD	8/15/2011		2,545.11
DERBAUM, TODD	8/19/2011		2,167.52
DERBAUM, TODD	9/2/2011		2,167.52
DERBAUM, TODD	9/8/2011		41.24
DERBAUM, TODD	9/16/2011		2,167.52
DERBAUM, TODD	9/30/2011		2,582.77
DESANTIS, JOSEPH	7/8/2011		2,277.64
DESANTIS, JOSEPH	7/22/2011		2,277.66
DESANTIS, JOSEPH	8/5/2011		2,277.66
DESANTIS, JOSEPH	8/15/2011		2,105.39
DESANTIS, JOSEPH	8/19/2011		2,277.65
DESANTIS, JOSEPH	9/2/2011		2,277.66
DESANTIS, JOSEPH	9/16/2011	-	2,277.65
DESANTIS, JOSEPH	9/30/2011		3,855.85
DESANTIS, JOSEPH J.	8/31/2011		6.50
DESANTIS, JOSEPH J.	9/8/2011		71.17
DETTBARN, MARK	7/22/2011		2,207.84
DETTBARN, MARK	8/5/2011		1,940.64
DETTBARN, MARK	8/15/2011		475.37
DETTBARN, MARK	8/19/2011		1,940.65
DETTBARN, MARK DETTBARN, MARK	9/2/2011		1,940.65
DETTBARN, MARK DETTBARN, MARK	9/16/2011		1,940.65
DETIBARN, MARK DETIBARN, MARK	9/30/2013		2,782.36
DELIDIMA, MUKA	9/30/201	ιφ	4,104.30

DETTHARN, MARK F	Name of creditor	Payment date	Amount paid
DETTIBARN, MARK F	DETTBARN, MARK F.	7/25/2011	2.00
DETTIBARN, MARK F	DETTBARN, MARK F.	8/15/2011	39.50
DETTOR ALICE	DETTBARN, MARK F.		
DETTIOR ALICE	DETTOR, ALICE		
DETTOR, ALICE 9/2/2011 \$ 448.45 DETTOR, ALICE 9/6/2011 \$ 749.24 DETTOR, ALICE 9/30/2011 \$ 732.06 DEYTON, MICAH M 7/8/2011 \$ 2334.45 DEYTON, MICAH M 7/22/2011 \$ 2334.59 DEYTON, MICAH M 8/5/2011 \$ 2334.59 DEYTON, MICAH M 8/5/2011 \$ 2334.59 DEYTON, MICAH M 9/12/2011 \$ 2334.59 DEYTON, MICAH M 9/12/2011 \$ 2334.59 DEYTON, MICAH M 9/16/2011 \$ 2334.59 DEYTON, MICAH M 9/16/2011 \$ 2334.59 DEYTON, MICAH M 9/30/2011 \$ 2334.59 DEYTON, MICAH M 9/16/2011 \$ 2334.59 DEYTON, MICAH M 9/30/2011 \$ 2334.59 DEYTON, MICAH M 9/30/2011 \$ 2334.59 DEYTON, MICAH M 9/30/2011 \$ 2235.59	DETTOR, ALICE		
DETTOR, ALICE 9/2/2011 \$ 448.45 DETTOR, ALICE 9/6/2011 \$ 749.57 DETTOR, ALICE 9/16/2011 \$ 749.57 DETTOR, ALICE 9/30/2011 \$ 752.06 DEYTON, MICAH M 7/8/2011 \$ 2,334.59 DEYTON, MICAH M 1/22/2011 \$ 2,334.59 DEYTON, MICAH M 8/5/2011 \$ 2,334.59 DEYTON, MICAH M 9/16/2011 \$ 2,334.59 DEYTON, MICAH M 9/16/2011 \$ 2,334.59 DEYTON, MICAH M 9/16/2011 \$ 2,334.59 DEYTON, MICAH M 9/30/2011 \$ 2,334.59	DETTOR, ALICE	8/5/2011	\$ 763.82
DETTOR, ALICE 9/16/2011 \$ 472.94 DETTOR, ALICE 9/16/2011 \$ 735.06 DETTOR, ALICE 9/30/2011 \$ 752.06 DEYTON, MICAH M 7/8/2011 \$ 2,334.60 DEYTON, MICAH M 8/19/2011 \$ 2,334.59 DEYTON, MICAH M 8/19/2011 \$ 2,334.59 DEYTON, MICAH M 9/12/2011 \$ 2,334.59 DEYTON, MICAH M 9/16/2011 \$ 2,355.69 DILLTKANICH, KENNETH 7/12/2011 \$ 1,355.	DETTOR, ALICE		
DETTOR, ALICE	DETTOR, ALICE		
DEYTON, MICAH M	DETTOR, ALICE		
DEYTON, MICAH M 7/22/2011 \$ 2,334.59 DEYTON, MICAH M 8/5/2011 \$ 2,334.59 DEYTON, MICAH M 9/2/2011 \$ 2,334.59 DEYTON, MICAH M 9/2/2011 \$ 2,334.59 DEYTON, MICAH M 9/30/2011 \$ 2,334.59 DEYTON, MICAH M 9/30/2011 \$ 2,334.59 DEYTON, MICAH M 9/30/2011 \$ 2,334.59 DHS MD DEPT. OF HEALTH SERVICES 7/14/2011 \$ 13,250.29 DILITKANICH, KENNETH 7/8/2011 \$ 2,155.49 DILITKANICH, KENNETH 7/11/2011 \$ 29.34 DILITKANICH, KENNETH 7/22/2011 \$ 2,155.49 DILITKANICH, KENNETH 7/22/2011 \$ 2,155.49 DILITKANICH, KENNETH 8/5/2011 \$ 2,155.49 DILITKANICH, KENNETH 8/5/2011 \$ 2,155.49 DILITKANICH, KENNETH 8/5/2011 \$ 2,155.49 DILITKANICH, KENNETH 8/2/2011 \$ 2,155.49 DILITKANICH, KENNETH 9/2/2011 \$ 2,155.49 DILITKANICH, KENNETH 9/3/2011 \$ 2,155.49 DILLITKANICH, KENNETH 9/3/2011 \$ 2,155.49 DILLITKANICH, KENNETH 9/3/2011 \$ 2,155.49 DILLITKANICH, KENNETH 9/3/2011 \$ 2,155.49 DILLION, STEWART 7/5/2011 \$ 1,955.60 <	DETTOR, ALICE	9/30/2011	\$ 752.06
DEYTON, MICAH M 7/22/2011 \$ 2,334.59 DEYTON, MICAH M 8/5/2011 \$ 2,334.59 DEYTON, MICAH M 9/2/2011 \$ 2,334.59 DEYTON, MICAH M 9/2/2011 \$ 2,334.59 DEYTON, MICAH M 9/30/2011 \$ 2,334.59 DEYTON, MICAH M 9/30/2011 \$ 2,334.59 DHS MN DEPT. OF HEALTH SERVICES 7/14/2011 \$ 68.39 DHS MN DEPT. OF HEALTH SERVICES 7/14/2011 \$ 13,250.29 DILITKANICH, KENNETH 7/8/2011 \$ 2,155.49 DILITKANICH, KENNETH 7/12/2011 \$ 2,155.49 DILITKANICH, KENNETH 7/22/2011 \$ 2,155.49 DILITKANICH, KENNETH 7/22/2011 \$ 2,155.49 DILITKANICH, KENNETH 8/5/2011 \$ 2,155.49 DILITKANICH, KENNETH 8/19/2011 \$ 2,155.49 DILITKANICH, KENNETH 9/2/2011 \$ 2,155.49 DILLITKANICH, KENNETH 9/30/2011 \$ 2,155.49 DILLITKANICH, KENNETH 9/30/2011 \$ 2,155.49 DILLON, STEWART 7/5/2011 \$ 1,955.65 DILLON, STEWART	DEYTON, MICAH M	7/8/2011	\$ 2,334.60
DEYTON, MICAH M 8/19/2011 \$ 2,334.59 DEYTON, MICAH M 9/2/2011 \$ 2,334.59 DEYTON, MICAH M 9/16/2011 \$ 2,334.59 DEYTON, MICAH M 9/30/2011 \$ 2,334.59 DHS MN DEPT. OF HEALTH SERVICES 7/14/2011 \$ 16.839 DHS MN DEPT. OF HEALTH SERVICES 7/14/2011 \$ 12.50.29 DILLTKANICH, KENNETH 7/8/2011 \$ 2,135.49 DILLTKANICH, KENNETH 7/12/2011 \$ 2,135.49 DILLTKANICH, KENNETH 7/22/2011 \$ 453.27 DILLTKANICH, KENNETH 8/5/2011 \$ 2,135.49 DILLTKANICH, KENNETH 8/19/2011 \$ 2,135.49 DILLTKANICH, KENNETH 8/25/2011 \$ 70.64 DILLTKANICH, KENNETH 8/25/2011 \$ 70.64 DILLTKANICH, KENNETH 9/8/2011 \$ 2,135.49 DILLTKANICH, KENNETH 9/8/2011 \$ 2,155.49 DILLTKANICH, KENNETH <td>DEYTON, MICAH M</td> <td>7/22/2011</td> <td>\$ 2,334.59</td>	DEYTON, MICAH M	7/22/2011	\$ 2,334.59
DEYTON, MICAH M 9/12/2011 \$ 2,334.59 DEYTON, MICAH M 9/16/2011 \$ 2,334.59 DEYTON, MICAH M 9/30/2011 \$ 2,334.59 DEYTON, MICAH M 9/30/2011 \$ 2,334.59 DHS MN DEPT. OF HEALTH SERVICES 7/14/2011 \$ 13,250.29 DILITKANICH, KENNETH 7/14/2011 \$ 2,135.49 DILITKANICH, KENNETH 7/12/2011 \$ 2,135.49 DILITKANICH, KENNETH 7/22/2011 \$ 2,135.49 DILITKANICH, KENNETH 7/25/2011 \$ 2,135.49 DILITKANICH, KENNETH 8/5/2011 \$ 2,135.49 DILITKANICH, KENNETH 8/5/2011 \$ 2,135.49 DILITKANICH, KENNETH 8/5/2011 \$ 70.64 DILITKANICH, KENNETH 9/2/2011 \$ 70.64 DILITKANICH, KENNETH 9/2/2011 \$ 70.64 DILITKANICH, KENNETH 9/8/2010 \$ 328.95 DILITKANICH, KENNETH 9/8/2011 \$ 2,135.49 DILLON, STEWART 9/16/2011 \$ 2,135.49 DILLON, STEWART 9/16/2011 \$ 2,135.49 DILLON, STEWART 9/16/2011 \$ 2,135.49 DILLON, STEWART 7/5/2011 \$ 1,935.64 DILLON, STEWART 7/5/2011 \$ 1,935.64 DILLON, STEWART 8/8/2011 \$ 1,935.64	DEYTON, MICAH M	8/5/2011	\$ 2,334.59
DEYTON, MICAH M 9/16/2011 \$ 2,334.59 DEYTON, MICAH M 9/30/2011 \$ 2,334.59 DIHS MN DEPT. OF HEALTH SERVICES 7/14/2011 \$ 68.39 DIHS MN DEPT. OF HEALTH SERVICES 7/14/2011 \$ 13,250.29 DILITKANICH, KENNETH 7/18/2011 \$ 2,135.49 DILITKANICH, KENNETH 7/12/2011 \$ 2,135.49 DILITKANICH, KENNETH 7/22/2011 \$ 2,135.54 DILITKANICH, KENNETH 7/25/2011 \$ 453.27 DILITKANICH, KENNETH 8/19/2011 \$ 2,135.48 DILITKANICH, KENNETH 8/19/2011 \$ 2,135.48 DILITKANICH, KENNETH 8/19/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/8/2011 \$ 2,135.49 DILLON, STEWART 7/5/2011 \$ 1,935.69 DILLON, STEWART 7/5/2011 \$ 1,935.69 DILLON, STEWART 7/5/2011 \$ 1,935.69 DILLON, STEWART 8/8/2011 \$ 167.05 DILLON, STEWART 8/8/2011 \$ 167.05 DILLON, STEWART 8/15/2011 \$ 1,935.65 DILLON, STEWART 9/2/2011 \$ 1,935.65	DEYTON, MICAH M	8/19/2011	\$ 2,334.59
DEYTON, MICAH M 9/30/2011 \$ 2,334.59 DHS MN DEPT. OF HEALTH SERVICES 7/14/2011 \$ 13,250.29 DILSTKANICH, KENNETH 7/8/2011 \$ 2,135.49 DILITKANICH, KENNETH 7/8/2011 \$ 2,135.49 DILITKANICH, KENNETH 7/11/2011 \$ 29.34 DILITKANICH, KENNETH 7/25/2011 \$ 453.27 DILITKANICH, KENNETH 7/25/2011 \$ 453.27 DILITKANICH, KENNETH 8/5/2011 \$ 2,135.50 DILITKANICH, KENNETH 8/9/2011 \$ 2,135.49 DILITKANICH, KENNETH 8/9/2/2011 \$ 70.64 DILITKANICH, KENNETH 9/2/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/16/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/16/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/16/2011 \$ 2,135.49 DILLON, STEWART 7/8/2011 \$ 1,935.65 DILLON, STEWART 7/8/2011 \$ 1,935.65 DILLON, STEWART 7/8/2011 \$ 1,935.65 DILLON, STEWART 8/8/2011 \$ 1,935.65 DILLON, STEWART 8/8/2011 \$ 1,935.65 DILLON, STEWART 8/8/2011 \$ 1,935.65 DILLON, STEWART 8/16/2011 \$ 1,935.65 DILLON, STEWART 9/8/2011 \$ 1,935.65	DEYTON, MICAH M	9/2/2011	
DEYTON, MICAH M 9/30/2011 \$ 2,334.59 DHS MN DEPT. OF HEALTH SERVICES 7/14/2011 \$ 13,250.29 DLIS MN DEPT. OF HEALTH SERVICES 7/4/2011 \$ 2,135.49 DLIJTKANICH, KENNETH 7/8/2011 \$ 2,135.49 DILITKANICH, KENNETH 7/11/2011 \$ 2,93.4 DILITKANICH, KENNETH 7/25/2011 \$ 2,135.48 DILITKANICH, KENNETH 7/25/2011 \$ 453.27 DILITKANICH, KENNETH 8/5/2011 \$ 2,135.50 DILITKANICH, KENNETH 8/9/2011 \$ 2,135.49 DILITKANICH, KENNETH 8/9/2/2011 \$ 70.64 DILITKANICH, KENNETH 9/2/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/2/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/16/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/30/2011 \$ 2,135.49 DILLON, STEWART 7/5/2011 \$ 19.56 DILLON, STEWART 7/5/2011 \$ 19.56 DILLON, STEWART 7/8/2011 \$ 1,935.65 DILLON, STEWART 8/8/2011 \$ 1,935.65 DILLON, STEWART 8/8/2011 \$ 1,935.65 DILLON, STEWART 8/8/2011 \$ 1,935.65 DILLON, STEWART 8/16/2011 \$ 1,935.66 DILLON, STEWART 9/8/2011 \$ 1,935.66	DEYTON, MICAH M	9/16/2011	\$ 2,334.59
DHS MN DEPT. OF HEALTH SERVICES	DEYTON, MICAH M	9/30/2011	
DILITKANICH, KENNETH 7/8/2011 \$ 2,135.49 DILITKANICH, KENNETH 7/11/2011 \$ 29.34 DILITKANICH, KENNETH 7/22/2011 \$ 453.27 DILITKANICH, KENNETH 7/25/2011 \$ 453.27 DILITKANICH, KENNETH 8/5/2011 \$ 2,135.50 DILITKANICH, KENNETH 8/19/2011 \$ 70.64 DILITKANICH, KENNETH 9/2/2011 \$ 70.64 DILITKANICH, KENNETH 9/2/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/8/2011 \$ 328.95 DILITKANICH, KENNETH 9/16/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/8/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/30/2011 \$ 2,135.49 DILION, STEWART 7/5/2011 \$ 91.36 DILLON, STEWART 7/5/2011 \$ 1935.65 DILLON, STEWART 7/5/2011 \$ 1935.65 DILLON, STEWART 7/22/2011 \$ 1935.65 DILLON, STEWART 8/8/2011 \$ 167.05 DILLON, STEWART 8/8/2011 \$ 167.05 DILLON, STEWART 8/15/2011 \$ 1935.65 DILLON, STEWART 8/15/2011 \$ 1935.65 DILLON, STEWART 9/2/2011 \$ 1,935.65 DILLON, STEWART 9/2/2011 \$ 1,935.65 DILLON, STEWART	DHS MN DEPT. OF HEALTH SERVICES	7/14/2011	\$ 68.39
DILITKANICH, KENNETH 7/8/2011 \$ 2,135.49 DILITKANICH, KENNETH 7/11/2011 \$ 29.34 DILITKANICH, KENNETH 7/22/2011 \$ 2,135.48 DILITKANICH, KENNETH 7/25/2011 \$ 453.27 DILITKANICH, KENNETH 8/5/2011 \$ 2,135.48 DILITKANICH, KENNETH 8/19/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/2/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/2/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/8/2011 \$ 328.95 DILITKANICH, KENNETH 9/16/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/16/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/30/2011 \$ 2,135.49 DILLON, STEWART 7/5/2011 \$ 91.36 DILLON, STEWART 7/5/2011 \$ 1,935.65 DILLON, STEWART 7/8/2011 \$ 1,935.65 DILLON, STEWART 8/5/2011 \$ 1,935.65 DILLON, STEWART 8/5/2011 \$ 1,935.65 DILLON, STEWART 8/5/2011 \$ 1,935.65 DILLON, STEWART 8/15/2011 \$ 1,935.65 DILLON, STEWART 8/15/2011 \$ 1,935.65 DILLON, STEWART 9/2/2011 \$ 1,935.65 DILLON, STEWART 9/2/2011 \$ 1,935.65 DILLON	DHS MN DEPT, OF HEALTH SERVICES	7/14/2011	\$ 13,250.29
DILITKANICH, KENNETH 7/22/2011 \$ 2,135.48 DILITKANICH, KENNETH 7/25/2011 \$ 453.27 DILITKANICH, KENNETH 8/5/2011 \$ 2,135.50 DILITKANICH, KENNETH 8/19/2011 \$ 2,135.48 DILITKANICH, KENNETH 8/25/2011 \$ 70.64 DILITKANICH, KENNETH 9/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/8/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/16/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/30/2011 \$ 2,135.49 DILLON, STEWART 7/5/2011 \$ 91.36 DILLON, STEWART 7/8/2011 \$ 1,935.65 DILLON, STEWART 7/8/2011 \$ 1,935.65 DILLON, STEWART 7/22/2011 \$ 1,935.65 DILLON, STEWART 8/5/2011 \$ 1,935.65 DILLON, STEWART 8/8/2011 \$ 167.05 DILLON, STEWART 8/19/2011 \$ 1,935.65 DILLON, STEWART 8/19/2011 \$ 1,935.65 DILLON, STEWART 8/19/2011 \$ 1,935.65 DILLON, STEWART 9/2/2011 \$ 1,935.65 DILLON, STEWART 9/8/2011 \$ 1,935.65 DILLON, STEWART 9/8/2011 \$ 1,935.65 DILLON, STEWART 9/8/2011 \$ 1,935.65 DILLON, STEWART	DILITKANICH, KENNETH	7/8/2011	
DILITKANICH, KENNETH 7/25/2011 \$ 453.27 DILITKANICH, KENNETH 8/5/2011 \$ 2,135.58 DILITKANICH, KENNETH 8/19/2011 \$ 2,135.48 DILITKANICH, KENNETH 9/2/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/2/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/16/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/30/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/30/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/30/2011 \$ 91.36 DILLON, SIEWART 7/5/2011 \$ 91.36 DILLON, SIEWART 7/5/2011 \$ 1,935.65 DILLON, SIEWART 7/22/2011 \$ 1,935.65 DILLON, SIEWART 8/5/2011 \$ 1,935.65 DILLON, SIEWART 8/5/2011 \$ 1,935.65 DILLON, SIEWART 8/15/2011 \$ 1,935.65 DILLON, SIEWART 8/15/2011 \$ 1,935.65 DILLON, SIEWART 9/2/2011 \$ 1,935.65 DILLON, SIEWART 9/2/2011 \$ 1,935.65 DILLON, SIEWART 9/8/2011 \$ 1,935.65	DILITKANICH, KENNETH	7/11/2011	\$ 29.34
DILITKANICH, KENNETH 7/25/2011 \$ 453.27 DILITKANICH, KENNETH 8/5/2011 \$ 2,135.58 DILITKANICH, KENNETH 8/19/2011 \$ 2,135.48 DILITKANICH, KENNETH 9/2/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/2/2011 \$ 22,135.49 DILITKANICH, KENNETH 9/16/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/30/2011 \$ 2,135.49 DILION, STEWART 7/5/2011 \$ 91.36 DILLON, STEWART 7/8/2011 \$ 1935.65 DILLON, STEWART 7/8/2011 \$ 1935.65 DILLON, STEWART 7/22/2011 \$ 189.05 DILLON, STEWART 8/5/2011 \$ 1,935.65 DILLON, STEWART 8/15/2011 \$ 1,935.65 DILLON, STEWART 9/2/2011 \$ 1,935.65 DILLON, STEWART 9/8/2011 \$ 1,935.65 DILLON, STEWART 9/8/2011 \$ 1,935.65 <td>DILITKANICH, KENNETH</td> <td>7/22/2011</td> <td>\$ 2,135.48</td>	DILITKANICH, KENNETH	7/22/2011	\$ 2,135.48
DILITKANICH, KENNETH 8/19/2011 \$ 2,135.48 DILITKANICH, KENNETH 8/25/2011 \$ 70.64 DILITKANICH, KENNETH 9/2/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/8/2011 \$ 328.95 DILITKANICH, KENNETH 9/16/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/30/2011 \$ 2,135.49 DILLON, STEWART 7/5/2011 \$ 91.36 DILLON, STEWART 7/8/2011 \$ 1935.65 DILLON, STEWART 7/22/2011 \$ 1,935.64 DILLON, STEWART 7/22/2011 \$ 1,935.65 DILLON, STEWART 8/5/2011 \$ 1,935.65 DILLON, STEWART 8/8/2011 \$ 167.05 DILLON, STEWART 8/15/2011 \$ 1,935.65 DILLON, STEWART 8/15/2011 \$ 1,935.65 DILLON, STEWART 8/15/2011 \$ 1,935.65 DILLON, STEWART 9/2/2011 \$ 1,935.65 DILLON, STEWART 9/2/2011 \$ 1,935.65 DILLON, STEWART 9/8/2011 \$ 115.00 DILLON, STEWART 9/8/2011 \$ 1,935.65 DILLON, STEWART 9/8/20	DILITKANICH, KENNETH	7/25/2011	
DILITKANICH, KENNETH 8/25/2011 \$ 70.64 DILITKANICH, KENNETH 9/2/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/8/2011 \$ 328.95 DILITKANICH, KENNETH 9/16/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/30/2011 \$ 2,135.49 DILLON, STEWART 7/5/2011 \$ 1935.65 DILLON, STEWART 7/8/2011 \$ 1,935.65 DILLON, STEWART 7/22/2011 \$ 189.05 DILLON, STEWART 8/5/2011 \$ 167.05 DILLON, STEWART 8/8/2011 \$ 167.05 DILLON, STEWART 8/15/2011 \$ 15.06 DILLON, STEWART 8/15/2011 \$ 1935.64 DILLON, STEWART 8/15/2011 \$ 1935.64 DILLON, STEWART 8/15/2011 \$ 1935.64 DILLON, STEWART 9/2/2011 \$ 1935.65 DILLON, STEWART 9/8/2011 \$ 2,106.40	DILITKANICH, KENNETH	8/5/2011	\$ 2,135.50
DILITKANICH, KENNETH 8/25/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/2/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/8/2011 \$ 328.95 DILITKANICH, KENNETH 9/16/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/30/2011 \$ 2,135.49 DILLON, STEWART 7/5/2011 \$ 193.66 DILLON, STEWART 7/8/2011 \$ 1,935.65 DILLON, STEWART 7/22/2011 \$ 189.05 DILLON, STEWART 8/5/2011 \$ 167.05 DILLON, STEWART 8/8/2011 \$ 167.05 DILLON, STEWART 8/15/2011 \$ 4553.63 DILLON, STEWART 8/15/2011 \$ 1935.64 DILLON, STEWART 9/8/2011 \$ 1935.65 DILLON, STEWART 9/8/2011 \$ 1935.65 DILLON, STEWART 9/8/2011 \$ 1935.66 DILLON, STEWART 9/8/2011 \$ 1935.65 DILLON, STEWART 9/8/2011 \$ 1935.65 DILLON, STEWART 9/8/2011 \$ 1935.65 DILLON, STEWART 9/8/2011 \$ 2,106.40 DIPPOLITO, DARLENE 9/3/2011 \$ 2,106		8/19/2011	\$ 2,135.48
DILITKANICH, KENNETH 9/8/2011 \$ 328.95 DILITKANICH, KENNETH 9/16/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/30/2011 \$ 2,135.49 DILLON, STEWART 7/5/2011 \$ 91.36 DILLON, STEWART 7/8/2011 \$ 1,935.65 DILLON, STEWART 7/22/2011 \$ 1,935.64 DILLON, STEWART 7/25/2011 \$ 189.05 DILLON, STEWART 8/5/2011 \$ 1,935.65 DILLON, STEWART 8/8/2011 \$ 167.05 DILLON, STEWART 8/15/2011 \$ 1,935.63 DILLON, STEWART 8/19/2011 \$ 1,935.64 DILLON, STEWART 8/19/2011 \$ 1,935.65 DILLON, STEWART 9/2/2011 \$ 1,935.65 DILLON, STEWART 9/2/2011 \$ 1,935.65 DILLON, STEWART 9/8/2011 \$ 63.75 DILLON, STEWART 9/8/2011 \$ 63.75 DILLON, STEWART 9/8/2011 \$ 2,106.49 DIPPOLITO, DARLENE 7/8/2011 \$ 2,106.49 DIPPOLITO, DARLENE 8/59/2011 \$ 2,106.49	DILITKANICH, KENNETH	8/25/2011	
DILITKANICH, KENNETH 9/16/2011 \$ 2,135.49 DILITKANICH, KENNETH 9/30/2011 \$ 2,135.49 DILLON, STEWART 7/5/2011 \$ 19.36 DILLON, STEWART 7/8/2011 \$ 1,935.65 DILLON, STEWART 7/22/2011 \$ 189.05 DILLON, STEWART 8/5/2011 \$ 1,935.65 DILLON, STEWART 8/8/2011 \$ 1,935.65 DILLON, STEWART 8/15/2011 \$ 1,935.65 DILLON, STEWART 8/15/2011 \$ 1,935.65 DILLON, STEWART 8/19/2011 \$ 1,935.65 DILLON, STEWART 8/25/2011 \$ 1,935.65 DILLON, STEWART 9/2/2011 \$ 1,935.65 DILLON, STEWART 9/8/2011 \$ 63.75 DILLON, STEWART 9/8/2011 \$ 1,935.65 DILLON, STEWART 9/8/2011 \$ 2,106.42	DILITKANICH, KENNETH	9/2/2011	\$ 2,135.49
DILITKANICH, KENNETH 9/30/2011 \$ 2,135.49 DILLON, STEWART 7/5/2011 \$ 91.36 DILLON, STEWART 7/8/2011 \$ 1,935.65 DILLON, STEWART 7/22/2011 \$ 1,935.65 DILLON, STEWART 7/25/2011 \$ 189.05 DILLON, STEWART 8/5/2011 \$ 1,935.65 DILLON, STEWART 8/8/2011 \$ 167.05 DILLON, STEWART 8/15/2011 \$ 1,935.63 DILLON, STEWART 8/15/2011 \$ 1,935.64 DILLON, STEWART 8/25/2011 \$ 115.60 DILLON, STEWART 9/2/2011 \$ 1,935.65 DILLON, STEWART 9/8/2011 \$ 2,106.40 DIPPOLITO, DARLENE 7/8/2011 \$ 2,106.40 DIPPOLITO, DARLENE 9/2/2011 \$ 2,106.41 DIPPOLITO, DARLENE 9/1/2011 \$ 2,900.43 </td <td>DILITKANICH, KENNETH</td> <td>9/8/2011</td> <td>\$ 328.95</td>	DILITKANICH, KENNETH	9/8/2011	\$ 328.95
DILLON, STEWART 7/5/2011 \$ 1,935.65 DILLON, STEWART 7/8/2011 \$ 1,935.65 DILLON, STEWART 7/22/2011 \$ 1,935.64 DILLON, STEWART 7/25/2011 \$ 189.05 DILLON, STEWART 8/5/2011 \$ 1,935.65 DILLON, STEWART 8/8/2011 \$ 167.05 DILLON, STEWART 8/15/2011 \$ 4,553.63 DILLON, STEWART 8/19/2011 \$ 1,935.64 DILLON, STEWART 8/25/2011 \$ 115.60 DILLON, STEWART 9/2/2011 \$ 1,935.65 DILLON, STEWART 9/8/2011 \$ 1,935.65 DILLON, STEWART 9/16/2011 \$ 2,106.40 DIPPOLITO, DARLENE 7/8/2011 \$ 2,106.40 DIPPOLITO, DARLENE 8/5/2011 \$ 2,106.41 DIPPOLITO, DARLENE 8/19/2011 \$ 2,106.39 DIPPOLITO, DARLENE 9/16/2011 \$ 2,106.39 DIPPOLITO, DARLENE 9/16/2011 \$ 2,106.39 DIPPOLITO, DARLENE 9/16/2011 \$ 2,200.43 DIVISION OF MEDICAL ASSISTANCE 9/15/2011 \$ 11,550.75 DOCUMENT SOLUTION	DILITKANICH, KENNETH	9/16/2011	\$ 2,135.49
DILLON, STEWART 7/8/2011 \$ 1,935.65 DILLON, STEWART 7/22/2011 \$ 1,935.64 DILLON, STEWART 7/25/2011 \$ 189.05 DILLON, STEWART 8/5/2011 \$ 1,935.65 DILLON, STEWART 8/8/2011 \$ 167.05 DILLON, STEWART 8/15/2011 \$ 4,553.63 DILLON, STEWART 8/15/2011 \$ 1,935.64 DILLON, STEWART 8/25/2011 \$ 115.60 DILLON, STEWART 9/2/2011 \$ 1,935.65 DILLON, STEWART 9/8/2011 \$ 63.75 DILLON, STEWART 9/8/2011 \$ 1,935.65 DILLON, STEWART 9/16/2011 \$ 1,935.65 DILLON, STEWART 9/30/2011 \$ 3,457.66 DIPPOLITO, DARLENE 7/8/2011 \$ 2,106.40 DIPPOLITO, DARLENE 7/8/2011 \$ 2,106.40 DIPPOLITO, DARLENE 8/5/2011 \$ 2,106.41 DIPPOLITO, DARLENE 8/19/2011 \$ 2,106.41 DIPPOLITO, DARLENE 9/16/2011 \$ 2,000.43 DIVISION OF MEDICAL ASSISTANCE 7/5/2011 \$ 11,550.75 DOCUMENT SOLUTIONS GRO	DILITKANICH, KENNETH	9/30/2011	\$ 2,135.49
DILLON, STEWART 7/22/2011 \$ 1,935.64 DILLON, STEWART 7/25/2011 \$ 189.05 DILLON, STEWART 8/5/2011 \$ 1,935.65 DILLON, STEWART 8/8/2011 \$ 167.05 DILLON, STEWART 8/15/2011 \$ 4,553.63 DILLON, STEWART 8/19/2011 \$ 1,935.64 DILLON, STEWART 8/25/2011 \$ 115.60 DILLON, STEWART 9/2/2011 \$ 1,935.65 DILLON, STEWART 9/8/2011 \$ 63.75 DILLON, STEWART 9/16/2011 \$ 1,935.65 DILLON, STEWART 9/30/2011 \$ 2,106.40 DIPPOLITO, DARLENE 7/8/2011 \$ 2,106.40 DIPPOLITO, DARLENE 8/5/2011 \$ 2,106.41 DIPPOLITO, DARLENE 9/2/2011 \$ 2,106.41 DIPPOLITO, DARLENE 9/2/2011 \$ 2,106.41 DIPPOLITO, DARLENE 9/2/2011 \$ 2,106.41 DIPPOLITO, DARLENE 9/16/2011 \$ 2,900.43 DIVISION OF MEDICAL ASSISTANCE 7/5/2011 \$ 11,550.75 DOCUMENT SOLUTIONS GROUP, INC 9/1/2011 \$ 252.24 DODICK, MICHAEL </td <td>DILLON, STEWART</td> <td>7/5/2011</td> <td>\$ 91.36</td>	DILLON, STEWART	7/5/2011	\$ 91.36
DILLON, STEWART 7/25/2011 \$ 1,935.65 DILLON, STEWART 8/5/2011 \$ 167.05 DILLON, STEWART 8/8/2011 \$ 167.05 DILLON, STEWART 8/15/2011 \$ 1,935.63 DILLON, STEWART 8/19/2011 \$ 1,935.64 DILLON, STEWART 8/25/2011 \$ 115.60 DILLON, STEWART 9/2/2011 \$ 1,935.65 DILLON, STEWART 9/8/2011 \$ 63.75 DILLON, STEWART 9/8/2011 \$ 1,935.65 DILLON, STEWART 9/16/2011 \$ 1,935.65 DILLON, STEWART 9/30/2011 \$ 3,457.66 DIPPOLITO, DARLENE 7/8/2011 \$ 2,106.40 DIPPOLITO, DARLENE 7/8/2011 \$ 2,106.41 DIPPOLITO, DARLENE 8/19/2011 \$ 2,106.41 DIPPOLITO, DARLENE 9/2/2011 \$ 2,106.41 DIPPOLITO, DARLENE 9/2/2011 \$ 2,106.41 DIPPOLITO, DARLENE 9/2/2011 \$ 2,006.43 DIPPOLITO, DARLENE 9/16/2011 \$ 2,900.43 DIVISION OF MEDICAL ASSISTANCE 9/1/2011 \$ 252.24 DOCUMENT SOLUTIONS GROUP, INC 9/1/2011 \$ 397.49 DODICK, MICHAEL 7/5/2011 \$ 397.49 DODICK, MICHAEL 7/8/2011 \$ 2,302.74	DILLON, STEWART	7/8/2011	\$ 1,935.65
DILLON, STEWART 8/5/2011 \$ 1,935.65 DILLON, STEWART 8/8/2011 \$ 167.05 DILLON, STEWART 8/15/2011 \$ 4,553.63 DILLON, STEWART 8/19/2011 \$ 1,935.64 DILLON, STEWART 8/25/2011 \$ 115.60 DILLON, STEWART 9/2/2011 \$ 1,935.65 DILLON, STEWART 9/8/2011 \$ 63.75 DILLON, STEWART 9/16/2011 \$ 1,935.65 DILLON, STEWART 9/30/2011 \$ 3,457.66 DILLON, STEWART 9/30/2011 \$ 2,106.40 DIPPOLITO, DARLENE 8/5/2011 \$ 2,106.40 DIPPOLITO, DARLENE 9/2/2011 \$ 2,106.41 DIPPOLITO, DARLENE 9/16/2011 \$ 2,900.43 DIVISION OF MEDICAL ASSISTANCE 7/5/2011 \$ 11,550.75 DOCUMENT SOLUTIONS GROUP, INC 9/1/2011 \$ 2,302.74	DILLON, STEWART	7/22/2011	\$ 1,935.64
DILLON, STEWART 8/8/2011 \$ 167.05 DILLON, STEWART 8/15/2011 \$ 4,553.63 DILLON, STEWART 8/19/2011 \$ 1,935.64 DILLON, STEWART 8/25/2011 \$ 115.60 DILLON, STEWART 9/2/2011 \$ 1,935.65 DILLON, STEWART 9/8/2011 \$ 63.75 DILLON, STEWART 9/16/2011 \$ 1,935.65 DILLON, STEWART 9/30/2011 \$ 3,457.66 DIPPOLITO, DARLENE 7/8/2011 \$ 2,106.40 DIPPOLITO, DARLENE 7/22/2011 \$ 2,106.41 DIPPOLITO, DARLENE 8/5/2011 \$ 2,106.41 DIPPOLITO, DARLENE 8/19/2011 \$ 2,106.41 DIPPOLITO, DARLENE 9/2/2011 \$ 2,106.41 DIPPOLITO, DARLENE 9/16/2011 \$ 2,900.43 DIPPOLITO, DARLENE 9/16/2011 \$ 2,900.43 DIVISION OF MEDICAL ASSISTANCE 7/5/2011 \$ 11,550.75 DOCUMENT SOLUTIONS GROUP, INC 9/1/2011 \$ 252.24 DODICK, MICHAEL 7/5/2011 \$ 397.49 DODICK, MICHAEL 7/8/2011 \$ 2,302.74	DILLON, STEWART	7/25/2011	\$ 189.05
DILLON, STEWART 8/15/2011 \$4,553.63 DILLON, STEWART 8/19/2011 \$1,935.64 DILLON, STEWART 8/25/2011 \$115.60 DILLON, STEWART 9/2/2011 \$1,935.65 DILLON, STEWART 9/8/2011 63.75 DILLON, STEWART 9/16/2011 \$1,935.65 DILLON, STEWART 9/30/2011 \$3,457.66 DIPPOLITO, DARLENE 7/8/2011 \$2,106.40 DIPPOLITO, DARLENE 7/22/2011 \$2,106.41 DIPPOLITO, DARLENE 8/5/2011 \$2,106.42 DIPPOLITO, DARLENE 8/19/2011 \$2,106.41 DIPPOLITO, DARLENE 9/2/2011 \$2,106.39 DIPPOLITO, DARLENE 9/16/2011 \$2,900.43 DIVISION OF MEDICAL ASSISTANCE 7/5/2011 \$11,550.75 DOCUMENT SOLUTIONS GROUP, INC 9/1/2011 \$252.24 DODICK, MICHAEL 7/5/2011 \$397.49 DODICK, MICHAEL 7/8/2011 \$2,302.74	DILLON, STEWART	8/5/2011	\$ 1,935.65
DILLON, STEWART 8/19/2011 \$ 1,935.64 DILLON, STEWART 8/25/2011 \$ 115.60 DILLON, STEWART 9/2/2011 \$ 1,935.65 DILLON, STEWART 9/8/2011 \$ 63.75 DILLON, STEWART 9/16/2011 \$ 1,935.65 DILLON, STEWART 9/30/2011 \$ 3,457.66 DIPPOLITO, DARLENE 7/8/2011 \$ 2,106.40 DIPPOLITO, DARLENE 7/22/2011 \$ 2,106.41 DIPPOLITO, DARLENE 8/5/2011 \$ 2,106.42 DIPPOLITO, DARLENE 8/19/2011 \$ 2,106.41 DIPPOLITO, DARLENE 9/2/2011 \$ 2,106.39 DIPPOLITO, DARLENE 9/16/2011 \$ 2,900.43 DIVISION OF MEDICAL ASSISTANCE 7/5/2011 \$ 11,550.75 DOCUMENT SOLUTIONS GROUP, INC 9/1/2011 \$ 252.24 DODICK, MICHAEL 7/5/2011 \$ 397.49 DODICK, MICHAEL 7/8/2011 \$ 2,302.74	DILLON, STEWART	8/8/2011	\$ 167.05
DILLON, STEWART 8/25/2011 \$ 115.60 DILLON, STEWART 9/2/2011 \$ 1,935.65 DILLON, STEWART 9/8/2011 \$ 63.75 DILLON, STEWART 9/16/2011 \$ 1,935.65 DILLON, STEWART 9/30/2011 \$ 3,457.66 DIPPOLITO, DARLENE 7/8/2011 \$ 2,106.40 DIPPOLITO, DARLENE 7/22/2011 \$ 2,106.41 DIPPOLITO, DARLENE 8/5/2011 \$ 2,106.42 DIPPOLITO, DARLENE 8/19/2011 \$ 2,106.41 DIPPOLITO, DARLENE 9/2/2011 \$ 2,106.39 DIPPOLITO, DARLENE 9/16/2011 \$ 2,900.43 DIVISION OF MEDICAL ASSISTANCE 7/5/2011 \$ 11,550.75 DOCUMENT SOLUTIONS GROUP, INC 9/1/2011 \$ 252.24 DODICK, MICHAEL 7/5/2011 \$ 397.49 DODICK, MICHAEL 7/8/2011 \$ 2,302.74	DILLON, STEWART	8/15/2011	\$ 4,553.63
DILLON, STEWART 9/2/2011 \$ 1,935.65 DILLON, STEWART 9/8/2011 \$ 1,935.65 DILLON, STEWART 9/16/2011 \$ 1,935.65 DILLON, STEWART 9/30/2011 \$ 3,457.66 DIPPOLITO, DARLENE 7/8/2011 \$ 2,106.40 DIPPOLITO, DARLENE 7/22/2011 \$ 2,106.41 DIPPOLITO, DARLENE 8/5/2011 \$ 2,106.42 DIPPOLITO, DARLENE 8/19/2011 \$ 2,106.41 DIPPOLITO, DARLENE 9/2/2011 \$ 2,106.39 DIPPOLITO, DARLENE 9/16/2011 \$ 2,900.43 DIVISION OF MEDICAL ASSISTANCE 7/5/2011 \$ 11,550.75 DOCUMENT SOLUTIONS GROUP, INC 9/1/2011 \$ 252.24 DODICK, MICHAEL 7/5/2011 \$ 397.49 DODICK, MICHAEL 7/8/2011 \$ 2,302.74	DILLON, STEWART	8/19/2011	\$ 1,935.64
DILLON, STEWART 9/8/2011 \$ 63.75 DILLON, STEWART 9/16/2011 \$ 1,935.65 DILLON, STEWART 9/30/2011 \$ 3,457.66 DIPPOLITO, DARLENE 7/8/2011 \$ 2,106.40 DIPPOLITO, DARLENE 7/22/2011 \$ 2,106.41 DIPPOLITO, DARLENE 8/5/2011 \$ 2,106.42 DIPPOLITO, DARLENE 8/19/2011 \$ 2,106.41 DIPPOLITO, DARLENE 9/2/2011 \$ 2,106.39 DIPPOLITO, DARLENE 9/16/2011 \$ 2,900.43 DIVISION OF MEDICAL ASSISTANCE 7/5/2011 \$ 11,550.75 DOCUMENT SOLUTIONS GROUP, INC 9/1/2011 \$ 252.24 DODICK, MICHAEL 7/5/2011 \$ 397.49 DODICK, MICHAEL 7/8/2011 \$ 2,302.74	DILLON, STEWART	8/25/2011	\$ 115.60
DILLON, STEWART 9/16/2011 \$ 1,935.65 DILLON, STEWART 9/30/2011 \$ 3,457.66 DIPPOLITO, DARLENE 7/8/2011 \$ 2,106.40 DIPPOLITO, DARLENE 7/22/2011 \$ 2,106.41 DIPPOLITO, DARLENE 8/5/2011 \$ 2,106.42 DIPPOLITO, DARLENE 8/19/2011 \$ 2,106.41 DIPPOLITO, DARLENE 9/2/2011 \$ 2,106.39 DIPPOLITO, DARLENE 9/16/2011 \$ 2,900.43 DIVISION OF MEDICAL ASSISTANCE 7/5/2011 \$ 11,550.75 DOCUMENT SOLUTIONS GROUP, INC 9/1/2011 \$ 252.24 DODICK, MICHAEL 7/5/2011 \$ 397.49 DODICK, MICHAEL 7/8/2011 \$ 2,302.74	DILLON, STEWART	9/2/2011	\$ 1,935.65
DILLON, STEWART 9/30/2011 \$ 3,457.66 DIPPOLITO, DARLENE 7/8/2011 \$ 2,106.40 DIPPOLITO, DARLENE 7/22/2011 \$ 2,106.41 DIPPOLITO, DARLENE 8/5/2011 \$ 2,106.42 DIPPOLITO, DARLENE 8/19/2011 \$ 2,106.41 DIPPOLITO, DARLENE 9/2/2011 \$ 2,106.39 DIPPOLITO, DARLENE 9/16/2011 \$ 2,900.43 DIVISION OF MEDICAL ASSISTANCE 7/5/2011 \$ 11,550.75 DOCUMENT SOLUTIONS GROUP, INC 9/1/2011 \$ 252.24 DODICK, MICHAEL 7/5/2011 \$ 397.49 DODICK, MICHAEL 7/8/2011 \$ 2,302.74	DILLON, STEWART	9/8/2011	\$ 63.75
DIPPOLITO, DARLENE 7/8/2011 \$ 2,106.40 DIPPOLITO, DARLENE 7/22/2011 \$ 2,106.41 DIPPOLITO, DARLENE 8/5/2011 \$ 2,106.42 DIPPOLITO, DARLENE 8/19/2011 \$ 2,106.41 DIPPOLITO, DARLENE 9/2/2011 \$ 2,106.39 DIPPOLITO, DARLENE 9/16/2011 \$ 2,900.43 DIVISION OF MEDICAL ASSISTANCE 7/5/2011 \$ 11,550.75 DOCUMENT SOLUTIONS GROUP, INC 9/1/2011 \$ 252.24 DODICK, MICHAEL 7/5/2011 \$ 397.49 DODICK, MICHAEL 7/8/2011 \$ 2,302.74	DILLON, STEWART	9/16/2011	\$ 1,935.65
DIPPOLITO, DARLENE 7/22/2011 \$ 2,106.41 DIPPOLITO, DARLENE 8/5/2011 \$ 2,106.42 DIPPOLITO, DARLENE 8/19/2011 \$ 2,106.41 DIPPOLITO, DARLENE 9/2/2011 \$ 2,106.39 DIPPOLITO, DARLENE 9/16/2011 \$ 2,900.43 DIVISION OF MEDICAL ASSISTANCE 7/5/2011 \$ 11,550.75 DOCUMENT SOLUTIONS GROUP, INC 9/1/2011 \$ 252.24 DODICK, MICHAEL 7/5/2011 \$ 397.49 DODICK, MICHAEL 7/8/2011 \$ 2,302.74	DILLON, STEWART	9/30/2011	\$ 3,457.66
DIPPOLITO, DARLENE 8/5/2011 \$ 2,106.42 DIPPOLITO, DARLENE 8/19/2011 \$ 2,106.41 DIPPOLITO, DARLENE 9/2/2011 \$ 2,106.39 DIPPOLITO, DARLENE 9/16/2011 \$ 2,900.43 DIVISION OF MEDICAL ASSISTANCE 7/5/2011 \$ 11,550.75 DOCUMENT SOLUTIONS GROUP, INC 9/1/2011 \$ 252.24 DODICK, MICHAEL 7/5/2011 \$ 397.49 DODICK, MICHAEL 7/8/2011 \$ 2,302.74	DIPPOLITO, DARLENE	7/8/2011	\$ 2,106.40
DIPPOLITO, DARLENE 8/19/2011 \$ 2,106.49 DIPPOLITO, DARLENE 9/2/2011 \$ 2,106.39 DIPPOLITO, DARLENE 9/16/2011 \$ 2,900.43 DIVISION OF MEDICAL ASSISTANCE 7/5/2011 \$ 11,550.75 DOCUMENT SOLUTIONS GROUP, INC 9/1/2011 \$ 252.24 DODICK, MICHAEL 7/5/2011 \$ 397.49 DODICK, MICHAEL 7/8/2011 \$ 2,302.74	DIPPOLITO, DARLENE	7/22/2011	\$ 2,106.41
DIPPOLITO, DARLENE 9/2/2011 \$ 2,106.39 DIPPOLITO, DARLENE 9/16/2011 \$ 2,900.43 DIVISION OF MEDICAL ASSISTANCE 7/5/2011 \$ 11,550.75 DOCUMENT SOLUTIONS GROUP, INC 9/1/2011 \$ 252.24 DODICK, MICHAEL 7/5/2011 \$ 397.49 DODICK, MICHAEL 7/8/2011 \$ 2,302.74	DIPPOLITO, DARLENE	8/5/2011	\$ 2,106.42
DIPPOLITO, DARLENE 9/16/2011 \$ 2,900.43 DIVISION OF MEDICAL ASSISTANCE 7/5/2011 \$ 11,550.75 DOCUMENT SOLUTIONS GROUP, INC 9/1/2011 \$ 252.24 DODICK, MICHAEL 7/5/2011 \$ 397.49 DODICK, MICHAEL 7/8/2011 \$ 2,302.74	DIPPOLITO, DARLENE	8/19/2011	\$ 2,106.41
DIVISION OF MEDICAL ASSISTANCE 7/5/2011 \$ 11,550.75 DOCUMENT SOLUTIONS GROUP, INC 9/1/2011 \$ 252.24 DODICK, MICHAEL 7/5/2011 \$ 397.49 DODICK, MICHAEL 7/8/2011 \$ 2,302.74	DIPPOLITO, DARLENE	9/2/2011	\$ 2,106.39
DOCUMENT SOLUTIONS GROUP, INC 9/1/2011 \$ 252.24 DODICK, MICHAEL 7/5/2011 \$ 397.49 DODICK, MICHAEL 7/8/2011 \$ 2,302.74	DIPPOLITO, DARLENE	9/16/2011	\$ 2,900.43
DODICK, MICHAEL 7/5/2011 \$ 397.49 DODICK, MICHAEL 7/8/2011 \$ 2,302.74		7/5/2011	\$ 11,550.75
DODICK, MICHAEL 7/8/2011 \$ 2,302.74	DOCUMENT SOLUTIONS GROUP, INC	9/1/2011	\$ 252.24
	DODICK, MICHAEL	7/5/2011	\$ 397.49
DODICK, MICHAEL 7/22/2011 \$ 2,302.73	DODICK, MICHAEL	7/8/2011	\$ 2,302.74
	DODICK, MICHAEL	7/22/2011	\$ 2,302.73

DODICK, MICHAEL	Name of creditor	Payment date	Amount paid
DODICK, MICHAEL	DODICK, MICHAEL	7/25/2011 \$	216.12
DODICK, MICHAEL			
DODICK, MICHAEL	DODICK, MICHAEL		
DODICK, MICHAEL			
DODICK, MICHAEL 8/32/2011 \$ 24.55 DODICK, MICHAEL 8/31/2011 \$ 145.00 DODICK, MICHAEL 9/2/2011 \$ 2,384.50 DODICK, MICHAEL 9/8/2011 \$ 2,343.25 DODICK, MICHAEL 9/16/2011 \$ 132.25 DODICK, MICHAEL 9/24/2011 \$ 152.25 DODICK, MICHAEL 9/26/2011 \$ 6,022.35 DONOVAN, JAN 7/8/2011 \$ 1,346.81 DONOVAN, JAN 7/8/2011 \$ 1,346.82 DONOVAN, JAN 8/19/2011 \$ 1,346.82 DONOVAN, JAN 8/19/2011 \$ 1,346.82 DONOVAN, JAN 9/16/2011 \$ 1,346.82 DONOVAN, JAN </td <td></td> <td></td> <td></td>			
DODICK, MICHAEL			
DODICK, MICHAEL			
DODICK, MICHAEL 9/8/2011 \$ 206.93 DODICK, MICHAEL 9/16/2011 \$ 132.95 DODICK, MICHAEL 9/21/2011 \$ 132.95 DODICK, MICHAEL 9/26/2011 \$ 457.44 DODICK, MICHAEL 9/30/2011 \$ 60.22.35 DONOVAN, JAN 7/8/2011 \$ 1,346.81 DONOVAN, JAN 7/2/2011 \$ 1,346.82 DONOVAN, JAN 8/19/2011 \$ 1,346.82 DONOVAN, JAN 9/2/2011 \$ 1,346.82 DONOVAN, JAN 9/16/2011 \$ 117,734.40 DPT LABORATORIES, LTD 7/2/2011 \$ 117,734.40 DPT LABORATORIES, LTD 7/2/2011 \$ 118,917.12 DPT LABORATORIES, LTD 9/1/2011 \$ 129.174.50 UBS RFODUCTIONS 7/5/2011 \$ 100.00 DUBOIS, BERTRAND 7/1/2011 \$ 100.00 <tr< td=""><td></td><td></td><td></td></tr<>			
DODICK, MICHAEL 9/16/2011 \$ 132.95 DODICK, MICHAEL 9/26/2011 \$ 457.44 DODICK, MICHAEL 9/26/2011 \$ 457.44 DODICK, MICHAEL 9/36/2011 \$ 6,022.35 DONOVAN, JAN 7/8/2011 \$ 1,346.81 DONOVAN, JAN 7/22/2011 \$ 1,346.82 DONOVAN, JAN 8/5/2011 \$ 1,346.81 DONOVAN, JAN 9/26/2011 \$ 1,346.82 DONOVAN, JAN 9/26/2011 \$ 1,346.82 DONOVAN, JAN 9/16/2011 \$ 1,346.82 DONOVAN, JAN 9/16/2011 \$ 1,346.82 DONOVAN, JAN 9/36/2011 \$ 1,346.82 DONOVAN, JAN			
DODICK, MICHAEL			
DODICK, MICHAEL 9/36/2011 \$ 6,022.35 DODICK, MICHAEL 9/30/2011 \$ 6,022.35 DONOVAN, JAN 7/8/2011 \$ 1,346.81 DONOVAN, JAN 7/22/2011 \$ 1,346.82 DONOVAN, JAN 8/15/2011 \$ 1,346.82 DONOVAN, JAN 9/16/2011 \$ 1,346.82 DONOVAN, JAN 9/16/2011 \$ 1,346.82 DONOVAN, JAN 9/30/2011 \$ 1,346.82 DONOVAN, JAN 9/30/2011 \$ 11,734.40 DPT LABORATORIES, LTD 7/5/2011 \$ 117,734.40 DPT LABORATORIES, LTD 8/4/2011 \$ 118,917.12 DPT LABORATORIES, LTD 8/4/2011 \$ 118,917.12 DPT LABORATORIES, LTD 8/4/2011 \$ 117,734.40 DPT LABORATORIES, LTD 8/4/2011 \$ 118,917.12 DB SPRODUCTIONS 7/5/2011 \$ 125,174.56 DB SPRODUCTIONS 7/5/2011 \$ 17,925.00 DUBOIS, BERTRAND 7/12/2011 \$ 160.00 DUBOIS, BERTRAND 7/12/2011 \$ 160.00 DUBOIS, BERTRAND L 7/8/2011 \$ 160.00 DUBOIS, BERTRAND L 7/8/2011 \$ 160.00 DUBOIS, BERTRAND L 8/15/2011 \$ 160.00 DUBOIS, BERTRAND L 8/15/2011 \$ 1,871.12 DUBOIS, BERTRAND L	DODICK, MICHAEL		
DODICK, MICHAEL 9/30/2011 \$ 1,346.81 DONOVAN, JAN 77/8/2011 \$ 1,346.82 DONOVAN, JAN 8/5/2011 \$ 1,346.82 DONOVAN, JAN 8/5/2011 \$ 1,346.82 DONOVAN, JAN 8/19/2011 \$ 1,346.81 DONOVAN, JAN 9/2/2011 \$ 1,346.82 DONOVAN, JAN 9/16/2011 \$ 1,346.82 DONOVAN, JAN 9/16/2011 \$ 1,346.82 DONOVAN, JAN 9/30/2011 \$ 1,346.82 DONOVAN, JAN 9/16/2011 \$ 117,34.49 DPT LABORATORIES, LTD 7/5/2011 \$ 117,734.40 DPT LABORATORIES, LTD 7/21/2011 \$ 2,815.45 DPT LABORATORIES, LTD 9/1/2011 \$ 125,174.56 DR3 PRODUCTIONS 7/5/2011 \$ 17,925.00 DUBOIS, BERTRAND 7/5/2011 \$ 17,925.00 DUBOIS, BERTRAND 7/25/2011 \$ 160.00 DUBOIS, BERTRAND 7/25/2011 \$ 160.00 DUBOIS, BERTRAND B 7/25/2011 \$ 1,871.12 DUBOIS, BERTRAND L 7/8/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/5/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/5/2011 \$ 1,871.12 DUBOIS, BERTRAND L 9/16/2011 \$ 1,871.32 DUBOIS, BERTRAND L 9/2/2			
DONOVAN, JAN 7/8/2011 \$ 1,346.82 DONOVAN, JAN 7/22/2011 \$ 1,346.82 DONOVAN, JAN 8/5/2011 \$ 1,346.82 DONOVAN, JAN 9/2/2011 \$ 1,346.82 DONOVAN, JAN 9/2/2011 \$ 1,346.82 DONOVAN, JAN 9/16/2011 \$ 1,346.82 DONOVAN, JAN 9/30/2011 \$ 18,24.54 DPT LABORATORIES, LTD 7/21/2011 \$ 18,24.54 DPT LABORATORIES, LTD 9/1/2011 \$ 18,29.14.55 DPT LABORATORIES, LTD 9/1/2011 \$ 16,000.00 DUBOIS, BERTRAND 7/1/2011 \$ 1,000.00	DODICK, MICHAEL	9/30/2011 \$	6,022.35
DONOVAN, JAN 7/22/2011 \$ 1,346.82 DONOVAN, JAN 8/5/2011 \$ 1,346.82 DONOVAN, JAN 9/2/2011 \$ 1,346.81 DONOVAN, JAN 9/16/2011 \$ 1,346.82 DDT LABORATORIES, LTD 7/5/2011 \$ 117,734.40 DPT LABORATORIES, LTD 8/4/2011 \$ 118,917.12 DPT LABORATORIES, LTD 9/1/2011 \$ 125,174.56 DBR PRODUCTIONS 7/5/2011 \$ 17,925.00 DBR PRODUCTIONS 7/12/2011 \$ 100.09 DUBOIS, BERTRAND 7/11/2011 \$ 100.09 DUBOIS, BERTRAND 7/12/2011 \$ 100.09 DUBOIS, BERTRAND 7/15/2011 \$ 100.09 DUBOIS, BERTRAND L 7/15/2011 \$ 100.09 DUBOIS, BERTRAND L 7/15/2011 \$ 100.09 DUBOIS, BERTRAND L 7/22/2011 \$ 1,871.13 DUBOIS, BERTRAND L 8/5/2011 \$ 1,871.13 DUBOIS, BERTRAND L 8/5/2011 \$ 1,871.13 DUBOIS, BERTRAND L 9/2/2011 \$ 1,871.13 DUBOIS, BERTRAND L 9/2/2011 \$ 1,871.13 DUBOIS, BERTRAND L 9			
DONOVAN, JAN 8/5/2011 \$ 1,346.82 DONOVAN, JAN 9/19/2011 \$ 1,346.81 DONOVAN, JAN 9/16/2011 \$ 1,346.82 DONOVAN, JAN 9/30/2011 \$ 1,346.82 DONOVAN, JAN 9/30/2011 \$ 1,346.82 DONOVAN, JAN 9/30/2011 \$ 117,734.46 DPT LABORATORIES, LITD 7/21/2011 \$ 117,734.40 DPT LABORATORIES, LITD 9/1/2011 \$ 128,154.51 DPT LABORATORIES, LITD 9/1/2011 \$ 128,174.56 DRS PRODUCTIONS 7/5/2011 \$ 17,925.00 DUBOIS, BERTRAND 7/11/2011 \$ 160.09 DUBOIS, BERTRAND 7/11/2011 \$ 160.09 DUBOIS, BERTRAND 7/5/2011 \$ 474.08 DUBOIS, BERTRAND 9/15/2011 \$ 50.00 DUBOIS, BERTRAND 9/15/2011 \$ 50.00 DUBOIS, BERTRAND L 7/8/2011 \$ 1,871.13 DUBOIS, BERTRAND L 7/8/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/5/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/5/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/5/2011 \$ 1,873.23 DUBOIS, BERTRAND L 9/2/2011 \$ 1,873.23 DUBOIS, BERTRAND L 9/2/2011 \$ 1,873.23 DUBOIS, BERTRAND L	· · · · · · · · · · · · · · · · · · ·		
DONOVAN, JAN 9/2/2011 \$ 1,346.81 DONOVAN, JAN 9/2/2011 \$ 1,346.82 DONOVAN, JAN 9/16/2011 \$ 1,346.82 DONOVAN, JAN 9/30/2011 \$ 117,734.40 DPT LABORATORIES, LTD 7/5/2011 \$ 117,734.40 DPT LABORATORIES, LTD 7/12/2011 \$ 2,815.45 DPT LABORATORIES, LTD 8/4/2011 \$ 118,917.12 DPT LABORATORIES, LTD 9/1/2011 \$ 125,174.56 DR3 PRODUCTIONS 7/5/2011 \$ 17,925.00 DUBOIS, BERTRAND 7/11/2011 \$ 160.09 DUBOIS, BERTRAND 7/11/2011 \$ 160.09 DUBOIS, BERTRAND 7/15/2011 \$ 50.00 DUBOIS, BERTRAND 7/15/2011 \$ 50.00 DUBOIS, BERTRAND 7/15/2011 \$ 50.00 DUBOIS, BERTRAND L 7/8/2011 \$ 1,871.13 DUBOIS, BERTRAND L 8/5/2011 \$ 1,871.12 DUBOIS, BERTRAND L 9/2/2011 \$ 1,873.23 DUBOIS, BERTRAND L 9/2/2011 \$ 1,873.23 DUBOIS, BERTRAND L 9/2/2011 \$ 1,873.23 DUBOIS, BERTRAND L </td <td>•</td> <td></td> <td></td>	•		
DONOVAN, JAN 9/2/2011 \$ 1,346.82 DONOVAN, JAN 9/16/2011 \$ 1,346.82 DONOVAN, JAN 9/30/2011 \$ 1,346.82 DPT LABORATORIES, LITD 7/5/2011 \$ 117,734.40 DPT LABORATORIES, LITD 7/21/2011 \$ 18,917.12 DPT LABORATORIES, LITD 8/4/2011 \$ 118,917.12 DPT LABORATORIES, LITD 9/1/2011 \$ 125,174.56 DR3 PRODUCTIONS 7/5/2011 \$ 17,925.00 DUBOIS, BERTRAND 7/1/2011 \$ 160.09 DUBOIS, BERTRAND 8/15/2011 \$ 50.00 DUBOIS, BERTRAND 8/15/2011 \$ 50.00 DUBOIS, BERTRAND 9/15/2011 \$ 50.00 DUBOIS, BERTRAND 9/15/2011 \$ 50.00 DUBOIS, BERTRAND 9/15/2011 \$ 50.00 DUBOIS, BERTRAND L 7/8/2011 \$ 1,871.13 DUBOIS, BERTRAND L 7/8/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/15/2011 \$ 3,065.61 DUBOIS, BERTRAND L 8/15/2011 \$ 1,871.22 DUBOIS, BERTRAND L 9/2/2011 \$ 1,873.23 DUBOIS, BERTRAND L 9/16/2011 \$ 1,905.75 DUBOIS, BERTRAND L 9/16/2011 \$ 1,905.75 DUBOIS, BERTRAND L 9/16/2011 \$ 1,905.75 DUBOIS, BERT			•
DONOVAN, JAN 9/16/2011 \$ 1,346.82 DONOVAN, JAN 9/30/2011 \$ 1,346.82 DOT LABORATORIES, LTD 7/5/2011 \$ 117,734.40 DPT LABORATORIES, LTD 7/21/2011 \$ 2,815.45 DPT LABORATORIES, LTD 8/4/2011 \$ 118,917.12 DPT LABORATORIES, LTD 9/1/2011 \$ 125,174.56 DR3 PRODUCTIONS 7/5/2011 \$ 17,925.00 DUBOIS, BERTRAND 7/11/2011 \$ 160.09 DUBOIS, BERTRAND 7/25/2011 \$ 474.08 DUBOIS, BERTRAND 9/15/2011 \$ 50.00 DUBOIS, BERTRAND 9/15/2011 \$ 50.00 DUBOIS, BERTRAND 9/15/2011 \$ 50.00 DUBOIS, BERTRAND L 7/8/2011 \$ 1,871.13 DUBOIS, BERTRAND L 7/8/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/5/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/15/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/15/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/15/2011 \$ 1,871.32 DUBOIS, BERTRAND L 9/16/2011 \$ 1,871.32 DUBOIS, BERTRAND L 9/16/2011 \$ 1,873.23 DUBOIS, BERTRAND L 9/16/2011 \$ 1,873.23 DUBOIS, BERTRAND L 9/16/2011 \$ 1,905.74 DUB			
DONOVAN, JAN 9/30/2011 \$ 1,346.80 DPT LABORATORIES, LITD 7/5/2011 \$ 117,734.40 DPT LABORATORIES, LITD 7/21/2011 \$ 2,815.45 DPT LABORATORIES, LITD 8/4/2011 \$ 118,917.12 DPT LABORATORIES, LITD 9/1/2011 \$ 125,174.56 DR3 PRODUCTIONS 7/5/2011 \$ 17,925.00 DUBOIS, BERTRAND 7/11/2011 \$ 160.09 DUBOIS, BERTRAND 7/25/2011 \$ 160.09 DUBOIS, BERTRAND 9/15/2011 \$ 50.00 DUBOIS, BERTRAND 9/15/2011 \$ 50.00 DUBOIS, BERTRAND 9/15/2011 \$ 50.00 DUBOIS, BERTRAND L 7/82/2011 \$ 1,871.13 DUBOIS, BERTRAND L 7/22/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/15/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/15/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/15/2011 \$ 1,873.23 DUBOIS, BERTRAND L 9/2/2011 \$ 1,905.75 DUBOIS, BERTRAND L 9/16/2011 \$ 1,905.75 DUBOIS, BERTRAND L 9/16/2011 \$ 1,905.75 DUBOIS, BERTRAND L 9/16/2011 \$ 1,255.83 DULANEY, AMY D 7/8/2011 \$ 1,255.83 DULANEY, AMY D 9/2/2011 \$ 1,255.83 DULANEY, AMY D 9/16/2011 \$ 2,014.92 DULANE	· ·		· · · · · · · · · · · · · · · · · · ·
DPT LABORATORIES, LITD 7/5/2011 \$ 117,734.40 DPT LABORATORIES, LITD 7/21/2011 \$ 2,815.45 DPT LABORATORIES, LITD 8/4/2011 \$ 118,917.12 DPT LABORATORIES, LITD 9/1/2011 \$ 125,174.56 DR3 PRODUCTIONS 7/5/2011 \$ 17,925.00 DUBOIS, BERTRAND 7/11/2011 \$ 160.09 DUBOIS, BERTRAND 7/25/2011 \$ 474.08 DUBOIS, BERTRAND 8/15/2011 \$ 50.00 DUBOIS, BERTRAND 9/15/2011 \$ 50.00 DUBOIS, BERTRAND L 7/8/2011 \$ 1,871.13 DUBOIS, BERTRAND L 7/8/2011 \$ 1,871.13 DUBOIS, BERTRAND L 8/5/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/5/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/15/2011 \$ 1,873.23 DUBOIS, BERTRAND L 9/2/2011 \$ 1,873.23 DUBOIS, BERTRAND L 9/2/2011 \$ 1,905.75 DUBOIS, BERTRAND L 9/2/2011 \$ 1,905.75 DUBOIS, BERTRAND L 9/16/2011 \$ 1,905.75 DUBOIS, BERTRAND L 9/30/2011 \$ 2,442.64 DULANEY, AMY D 7/8/2011 \$ 1,255.83 DULANEY, AMY D 9/30/2011 \$ 2,014.92 DULANEY, AMY D 9/2/2011 \$ 2,014.92	· · · · · · · · · · · · · · · · · · ·		
DPT LABORATORIES, LTD 7/21/2011 \$ 2,815.45 DPT LABORATORIES, LTD 8/4/2011 \$ 118,917.12 DPT LABORATORIES, LTD 9/1/2011 \$ 125,174.56 DR3 PRODUCTIONS 7/5/2011 \$ 17,925.00 DUBOIS, BERTRAND 7/11/2011 \$ 160.09 DUBOIS, BERTRAND 7/12/2011 \$ 50.00 DUBOIS, BERTRAND 9/15/2011 \$ 50.00 DUBOIS, BERTRAND L 7/8/2011 \$ 1,871.13 DUBOIS, BERTRAND L 7/8/2011 \$ 1,871.12 DUBOIS, BERTRAND L 7/8/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/15/2011 \$ 3,065.61 DUBOIS, BERTRAND L 8/19/2011 \$ 1,905.74 DUBOIS, BERTRAND L 9/2/2011 \$ 1,905.74 DUBOIS, BERTRAND L 9/16/2011 \$ 1,905.74 DUBOIS, BERTRAND L 9/16/2011 \$ 1,905.74 DUBOIS, BERTRAND L 9/16/2011 \$ 1,905.75 DUBOIS, BERTRAND L 9/16/2011 \$ 1,905.75 DUBOIS, BERTRAND L 9/16/2011 \$ 2,014.92 DULANEY, AMY D 7/8/2011	•		· ·
DPT LABORATORIES, LTD 8/4/2011 \$ 125,174,56 DP3 PRODUCTIONS 7/5/2011 \$ 125,174,56 DUBOIS, BERTRAND 7/1/1/2011 \$ 160.09 DUBOIS, BERTRAND 7/25/2011 \$ 474.08 DUBOIS, BERTRAND 8/15/2011 \$ 50.00 DUBOIS, BERTRAND 9/15/2011 \$ 50.00 DUBOIS, BERTRAND 9/15/2011 \$ 1,871.13 DUBOIS, BERTRAND L 7/8/2011 \$ 1,871.13 DUBOIS, BERTRAND L 7/22/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/15/2011 \$ 3,065.61 DUBOIS, BERTRAND L 8/15/2011 \$ 3,065.61 DUBOIS, BERTRAND L 9/2/2011 \$ 1,873.23 DUBOIS, BERTRAND L 9/2/2011 \$ 1,905.74 DUBOIS, BERTRAND L 9/2/2011 \$ 1,905.74 DUBOIS, BERTRAND L 9/2/2011 \$ 1,905.75 DUBOIS, BERTRAND L 9/30/2011 \$ 1,255.83 DUBOIS, BERTRAND L 9/30/2011 \$ 1,255.83 DULANEY, AMY D 7/8/2011 \$ 1,255.83 DULANEY, AMY D 8/19/2011 \$ 2,014.92 DULANEY, AMY D 9/2/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 DULANEY, AMY D 9/16/2011 \$ 2,014.92 DULANEY, AMY D	•		
DPT LABORATORIES, LTD 9/1/2011 \$ 125,174.56 DR3 PRODUCTIONS 7/5/2011 \$ 17,925.00 DUBOIS, BERTRAND 7/11/2011 \$ 160.09 DUBOIS, BERTRAND 7/25/2011 \$ 474.08 DUBOIS, BERTRAND 8/15/2011 \$ 50.00 DUBOIS, BERTRAND 9/15/2011 \$ 1,871.13 DUBOIS, BERTRAND L 7/8/2011 \$ 1,871.12 DUBOIS, BERTRAND L 7/22/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/15/2011 \$ 3,065.61 DUBOIS, BERTRAND L 8/15/2011 \$ 3,065.61 DUBOIS, BERTRAND L 8/19/2011 \$ 1,873.23 DUBOIS, BERTRAND L 9/2/2011 \$ 1,905.74 DUBOIS, BERTRAND L 9/2/2011 \$ 1,905.74 DUBOIS, BERTRAND L 9/16/2011 \$ 1,905.75 DUBOIS, BERTRAND L 9/16/2011 \$ 1,255.83 DUBOIS, BERTRAND L 9/30/2011 \$ 2,442.64 DULANEY, AMY D 7/8/2011 \$ 1,255.83 DULANEY, AMY D 8/5/2011 \$ 2,014.92 DULANEY, AMY D 9/2/2011 \$ 2,014.92 DULANEY, AMY D 9/2/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 DULANEY, AMY D			•
DR3 PRODUCTIONS 7/5/2011 \$ 17,925.00 DUBOIS, BERTRAND 7/11/2011 \$ 160.09 DUBOIS, BERTRAND 7/25/2011 \$ 474.08 DUBOIS, BERTRAND 8/15/2011 \$ 50.00 DUBOIS, BERTRAND 9/15/2011 \$ 1,871.13 DUBOIS, BERTRAND L 7/8/2011 \$ 1,871.13 DUBOIS, BERTRAND L 8/5/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/5/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/15/2011 \$ 1,873.23 DUBOIS, BERTRAND L 8/19/2011 \$ 1,873.23 DUBOIS, BERTRAND L 9/2/2011 \$ 1,805.75 DUBOIS, BERTRAND L 9/16/2011 \$ 1,905.75 DUBOIS, BERTRAND L 9/30/2011 \$ 2,442.64 DULANEY, AMY D 7/8/2011 \$ 1,255.83 DULANEY, AMY D 8/5/2011 \$ 2,014.92 DULANEY, AMY D 8/5/2011 \$ 2,014.92 DULANEY, AMY D 9/16/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 EDWARDS ANGELL PALMER & DODGE <td>•</td> <td></td> <td></td>	•		
DUBOIS, BERTRAND 7/11/2011 \$ 474.08 DUBOIS, BERTRAND 7/25/2011 \$ 50.00 DUBOIS, BERTRAND 8/15/2011 \$ 50.00 DUBOIS, BERTRAND 9/15/2011 \$ 50.00 DUBOIS, BERTRAND L 7/8/2011 \$ 1,871.13 DUBOIS, BERTRAND L 7/22/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/5/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/15/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/15/2011 \$ 1,873.23 DUBOIS, BERTRAND L 8/19/2011 \$ 1,905.74 DUBOIS, BERTRAND L 9/2/2011 \$ 1,905.74 DUBOIS, BERTRAND L 9/16/2011 \$ 1,905.75 DUBOIS, BERTRAND L 9/30/2011 \$ 2,442.64 DULANEY, AMY D 7/8/2011 \$ 1,255.83 DULANEY, AMY D 7/22/2011 \$ 1,255.83 DULANEY, AMY D 8/5/2011 \$ 2,014.92 DULANEY, AMY D 8/5/2011 \$ 2,014.92 DULANEY, AMY D 9/2/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 DUBANEY, AMY D 9/30	•		•
DUBOIS, BERTRAND 7/25/2011 \$ 474.08 DUBOIS, BERTRAND 8/15/2011 \$ 50.00 DUBOIS, BERTRAND 9/15/2011 \$ 50.00 DUBOIS, BERTRAND L 7/8/2011 \$ 1,871.13 DUBOIS, BERTRAND L 7/22/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/5/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/15/2011 \$ 1,873.23 DUBOIS, BERTRAND L 8/19/2011 \$ 1,905.74 DUBOIS, BERTRAND L 9/2/2011 \$ 1,905.74 DUBOIS, BERTRAND L 9/16/2011 \$ 1,905.75 DUBOIS, BERTRAND L 9/16/2011 \$ 1,905.75 DUBOIS, BERTRAND L 9/30/2011 \$ 2,442.64 DULANEY, AMY D 7/8/2011 \$ 1,255.83 DULANEY, AMY D 7/22/2011 \$ 1,825.15 DULANEY, AMY D 8/5/2011 \$ 2,014.92 DULANEY, AMY D 8/5/2011 \$ 2,014.92 DULANEY, AMY D 9/16/2011 \$ 2,014.92 DULANEY, AMY D 9/16/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 EDWARDS ANGELL PALMER & DODGE 9/12/2011 \$ 1,274,709.77 EDWARDS ANGELL PALMER & DODGE 9/2/2011 \$ 319,418.69 EDWARDS ANGELL PALMER & DODGE 9/28/2011 \$ 319,418.69			
DUBOIS, BERTRAND 8/15/2011 \$ 50.00 DUBOIS, BERTRAND 9/15/2011 \$ 50.00 DUBOIS, BERTRAND L 7/8/2011 \$ 1,871.13 DUBOIS, BERTRAND L 7/22/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/5/2011 \$ 3,065.61 DUBOIS, BERTRAND L 8/15/2011 \$ 1,873.23 DUBOIS, BERTRAND L 8/19/2011 \$ 1,873.23 DUBOIS, BERTRAND L 9/2/2011 \$ 1,905.75 DUBOIS, BERTRAND L 9/30/2011 \$ 2,442.64 DUBOIS, BERTRAND L 9/30/2011 \$ 1,255.83 DUBOIS, BERTRAND L 9/30/2011 \$ 1,255.83 DULANEY, AMY D 7/8/2011 \$ 1,825.15 DULANEY, AMY D 8/5/2011 \$ 2,014.92 DULANEY, AMY D 9/2/2011 \$ 2,014.92 DULANEY, AMY D 9/2/2011 \$ 2,014.92 DULANEY, AMY D 9/16/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 EDWARDS ANGELL PALMER & DODGE 9/1/2011 \$ 10,376.00 EDWARDS ANGELL PAL	DUBOIS, BERTRAND		474.08
DUBOIS, BERTRAND 9/15/2011 \$ 50.00 DUBOIS, BERTRAND L 7/8/2011 \$ 1,871.13 DUBOIS, BERTRAND L 7/22/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/5/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/15/2011 \$ 3,065.61 DUBOIS, BERTRAND L 8/19/2011 \$ 1,873.23 DUBOIS, BERTRAND L 9/2/2011 \$ 1,905.74 DUBOIS, BERTRAND L 9/16/2011 \$ 1,905.75 DUBOIS, BERTRAND L 9/30/2011 \$ 2,442.64 DULANEY, AMY D 7/8/2011 \$ 1,255.83 DULANEY, AMY D 7/8/2011 \$ 2,014.92 DULANEY, AMY D 8/19/2011 \$ 2,014.92 DULANEY, AMY D 9/2/2011 \$ 2,014.92 DULANEY, AMY D 9/2/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 EAST TENNESSEE SCALE WORKS 7/21/2011 \$ 234.35 EDS 7/5/2011 \$ 10,376.00			50.00
DUBOIS, BERTRAND L 7/8/2011 \$ 1,871.13 DUBOIS, BERTRAND L 7/22/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/5/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/15/2011 \$ 3,065.61 DUBOIS, BERTRAND L 8/19/2011 \$ 1,873.23 DUBOIS, BERTRAND L 9/2/2011 \$ 1,905.74 DUBOIS, BERTRAND L 9/16/2011 \$ 1,905.75 DUBOIS, BERTRAND L 9/30/2011 \$ 2,442.64 DULANEY, AMY D 7/8/2011 \$ 1,255.83 DULANEY, AMY D 7/22/2011 \$ 1,825.15 DULANEY, AMY D 8/5/2011 \$ 2,014.92 DULANEY, AMY D 9/2/2011 \$ 2,014.92 DULANEY, AMY D 9/16/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92			
DUBOIS, BERTRAND L 7/22/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/5/2011 \$ 3,065.61 DUBOIS, BERTRAND L 8/15/2011 \$ 3,065.61 DUBOIS, BERTRAND L 8/19/2011 \$ 1,873.23 DUBOIS, BERTRAND L 9/2/2011 \$ 1,905.74 DUBOIS, BERTRAND L 9/16/2011 \$ 1,905.75 DUBOIS, BERTRAND L 9/30/2011 \$ 2,442.64 DULANEY, AMY D 7/8/2011 \$ 1,255.83 DULANEY, AMY D 7/22/2011 \$ 1,825.15 DULANEY, AMY D 8/5/2011 \$ 2,014.92 DULANEY, AMY D 9/16/2011 \$ 2,014.92 DULANEY, AMY D 9/16/2011 \$ 2,014.92 DULANEY, AMY D 9/16/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 EAST TENNESSEE SCALE WORKS 7/21/2011 \$ 234.35 EDS 7/5/2011 \$ 10,376.00 EDWARDS ANGELL PALMER & DODGE 9/1/2011 \$ 131,418.69 EDWARDS ANGELL PALMER & DODGE 9/28/2011 \$ 319,418.69 EDWARDS ANGELL PALMER & DODGE 9/28/2011 \$ 319,418.69 EDWARDS ANGELL PALMER & DODGE 9/28/2011 \$ 310,000.00 ELSEVIER INC. 8/25/2011 \$ 13,034.54	·		
DUBOIS, BERTRAND L 8/5/2011 \$ 1,871.12 DUBOIS, BERTRAND L 8/15/2011 \$ 3,065.61 DUBOIS, BERTRAND L 8/19/2011 \$ 1,873.23 DUBOIS, BERTRAND L 9/2/2011 \$ 1,905.75 DUBOIS, BERTRAND L 9/30/2011 \$ 1,905.75 DUBOIS, BERTRAND L 9/30/2011 \$ 2,442.64 DULANEY, AMY D 7/8/2011 \$ 1,255.83 DULANEY, AMY D 7/22/2011 \$ 1,825.15 DULANEY, AMY D 8/5/2011 \$ 2,014.92 DULANEY, AMY D 9/2/2011 \$ 2,014.92 DULANEY, AMY D 9/16/2011 \$ 2,014.92 DULANEY, AMY D 9/16/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 EAST TENNESSEE SCALE WORKS 7/21/2011 \$ 2,014.92 EDS 7/5/2011 \$ 10,376.00 EDWARDS ANGELL PALMER & DODGE 9/1/2011 \$ 1,274,709.77 EDWARDS ANGELL PALMER & DODGE 9/2/2011 \$ 319,418.69 EDWARDS ANGELL PALMER & DODGE 9/2/2011	DUBOIS, BERTRAND L		
DUBOIS, BERTRAND L 8/19/2011 \$ 1,873.23 DUBOIS, BERTRAND L 9/2/2011 \$ 1,905.75 DUBOIS, BERTRAND L 9/16/2011 \$ 1,905.75 DUBOIS, BERTRAND L 9/30/2011 \$ 2,442.64 DULANEY, AMY D 7/8/2011 \$ 1,255.83 DULANEY, AMY D 7/22/2011 \$ 1,825.15 DULANEY, AMY D 8/5/2011 \$ 2,014.92 DULANEY, AMY D 8/19/2011 \$ 2,014.92 DULANEY, AMY D 9/2/2011 \$ 2,014.92 DULANEY, AMY D 9/16/2011 \$ 2,014.92 DULANEY, AMY D 9/16/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 EAST TENNESSEE SCALE WORKS 7/21/2011 \$ 234.35 EDS 7/5/2011 \$ 10,376.00 EDWARDS ANGELL PALMER & DODGE 9/1/2011 \$ 1,274,709.77 EDWARDS ANGELL PALMER & DODGE 9/2/2011 \$ 319,418.69 EDWARDS ANGELL PALMER & DODGE 9/28/2011 \$ 300,000.00 ELSEVIER INC. 8/11/2011 \$ 13,034.54 ELSEVIER INC. 8/25/2011 \$ 18,297.94 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,236.72 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,236.72		•	
DUBOIS, BERTRAND L 9/2/2011 \$ 1,905.75 DUBOIS, BERTRAND L 9/16/2011 \$ 1,905.75 DUBOIS, BERTRAND L 9/30/2011 \$ 2,442.64 DULANEY, AMY D 7/8/2011 \$ 1,255.83 DULANEY, AMY D 8/5/2011 \$ 2,014.92 DULANEY, AMY D 8/19/2011 \$ 2,014.92 DULANEY, AMY D 9/2/2011 \$ 2,014.92 DULANEY, AMY D 9/16/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 EAST TENNESSEE SCALE WORKS 7/21/2011 \$ 2,014.92 EAST TENNESSEE SCALE WORKS 7/21/2011 \$ 234.35 EDS 7/5/2011 \$ 10,376.00 EDWARDS ANGELL PALMER & DODGE 9/1/2011 \$ 1,274,709.77 EDWARDS ANGELL PALMER & DODGE 9/2/2011 \$ 300,000.00 ELSEVIER INC. 8/11/2011 \$ 13,034.54 ELSEVIER INC. 8/25/2011 \$ 18,297.94 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,236.72 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,628.15	DUBOIS, BERTRAND L	8/15/2011 \$	3,065.61
DUBOIS, BERTRAND L 9/16/2011 \$ 1,905.75 DUBOIS, BERTRAND L 9/30/2011 \$ 2,442.64 DULANEY, AMY D 7/8/2011 \$ 1,255.83 DULANEY, AMY D 7/22/2011 \$ 1,825.15 DULANEY, AMY D 8/5/2011 \$ 2,014.92 DULANEY, AMY D 9/2/2011 \$ 2,014.92 DULANEY, AMY D 9/16/2011 \$ 2,014.92 DULANEY, AMY D 9/16/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 EAST TENNESSEE SCALE WORKS 7/21/2011 \$ 234.35 EDS 7/5/2011 \$ 10,376.00 EDWARDS ANGELL PALMER & DODGE 9/1/2011 \$ 1,274,709.77 EDWARDS ANGELL PALMER & DODGE 9/2/2011 \$ 319,418.69 EDWARDS ANGELL PALMER & DODGE 9/28/2011 \$ 300,000.00 ELSEVIER INC. 8/11/2011 \$ 13,034.54 ELSEVIER INC. 8/25/2011 \$ 18,297.94 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,236.72 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,236.72	DUBOIS, BERTRAND L	8/19/2011 \$	1,873.23
DUBOIS, BERTRAND L 9/30/2011 \$ 2,442.64 DULANEY, AMY D 7/8/2011 \$ 1,255.83 DULANEY, AMY D 7/22/2011 \$ 1,825.15 DULANEY, AMY D 8/5/2011 \$ 2,014.92 DULANEY, AMY D 9/2/2011 \$ 2,014.92 DULANEY, AMY D 9/16/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 EAST TENNESSEE SCALE WORKS 7/21/2011 \$ 234.35 EDS 7/5/2011 \$ 10,376.00 EDWARDS ANGELL PALMER & DODGE 9/1/2011 \$ 1,274,709.77 EDWARDS ANGELL PALMER & DODGE 9/28/2011 \$ 319,418.69 EDWARDS ANGELL PALMER & DODGE 9/28/2011 \$ 300,000.00 ELSEVIER INC. 8/11/2011 \$ 13,034.54 ELSEVIER INC. 8/25/2011 \$ 18,297.94 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,236.72 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,628.15	DUBOIS, BERTRAND L	9/2/2011 \$	1,905.74
DULANEY, AMY D 7/8/2011 \$ 1,255.83 DULANEY, AMY D 7/22/2011 \$ 1,825.15 DULANEY, AMY D 8/5/2011 \$ 2,014.92 DULANEY, AMY D 8/19/2011 \$ 2,014.92 DULANEY, AMY D 9/2/2011 \$ 2,014.92 DULANEY, AMY D 9/16/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 EAST TENNESSEE SCALE WORKS 7/21/2011 \$ 234.35 EDS 7/5/2011 \$ 10,376.00 EDWARDS ANGELL PALMER & DODGE 9/1/2011 \$ 1,274,709.77 EDWARDS ANGELL PALMER & DODGE 9/2/2011 \$ 319,418.69 EDWARDS ANGELL PALMER & DODGE 9/28/2011 \$ 300,000.00 ELSEVIER INC. 8/11/2011 \$ 13,034.54 ELSEVIER INC. 8/25/2011 \$ 18,297.94 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,236.72 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,628.15	DUBOIS, BERTRAND L	9/16/2011 \$	1,905.75
DULANEY, AMY D 7/22/2011 \$ 1,825.15 DULANEY, AMY D 8/5/2011 \$ 2,014.92 DULANEY, AMY D 8/19/2011 \$ 2,014.92 DULANEY, AMY D 9/16/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 EAST TENNESSEE SCALE WORKS 7/21/2011 \$ 234.35 EDS 7/5/2011 \$ 10,376.00 EDWARDS ANGELL PALMER & DODGE 9/1/2011 \$ 1,274,709.77 EDWARDS ANGELL PALMER & DODGE 9/2/2011 \$ 319,418.69 EDWARDS ANGELL PALMER & DODGE 9/28/2011 \$ 300,000.00 ELSEVIER INC. 8/11/2011 \$ 13,034.54 ELSEVIER INC. 8/25/2011 \$ 18,297.94 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,236.72 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,628.15	DUBOIS, BERTRAND L	9/30/2011 \$	2,442.64
DULANEY, AMY D 8/5/2011 \$ 2,014.92 DULANEY, AMY D 8/19/2011 \$ 2,014.92 DULANEY, AMY D 9/2/2011 \$ 2,014.92 DULANEY, AMY D 9/16/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 EAST TENNESSEE SCALE WORKS 7/21/2011 \$ 234.35 EDS 7/5/2011 \$ 10,376.00 EDWARDS ANGELL PALMER & DODGE 9/1/2011 \$ 1,274,709.77 EDWARDS ANGELL PALMER & DODGE 9/2/2011 \$ 319,418.69 EDWARDS ANGELL PALMER & DODGE 9/28/2011 \$ 300,000.00 ELSEVIER INC. 8/11/2011 \$ 13,034.54 ELSEVIER INC. 8/25/2011 \$ 18,297.94 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,236.72 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,628.15	DULANEY, AMY D	7/8/2011 \$	1,255.83
DULANEY, AMY D 8/19/2011 \$ 2,014.92 DULANEY, AMY D 9/2/2011 \$ 2,014.92 DULANEY, AMY D 9/16/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 EAST TENNESSEE SCALE WORKS 7/21/2011 \$ 234.35 EDS 7/5/2011 \$ 10,376.00 EDWARDS ANGELL PALMER & DODGE 9/1/2011 \$ 1,274,709.77 EDWARDS ANGELL PALMER & DODGE 9/2/2011 \$ 319,418.69 EDWARDS ANGELL PALMER & DODGE 9/28/2011 \$ 300,000.00 ELSEVIER INC. 8/11/2011 \$ 13,034.54 ELSEVIER INC. 8/25/2011 \$ 18,297.94 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,236.72 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,628.15	DULANEY, AMY D	7/22/2011 \$	1,825.15
DULANEY, AMY D 9/2/2011 \$ 2,014.92 DULANEY, AMY D 9/16/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 EAST TENNESSEE SCALE WORKS 7/21/2011 \$ 234.35 EDS 7/5/2011 \$ 10,376.00 EDWARDS ANGELL PALMER & DODGE 9/1/2011 \$ 1,274,709.77 EDWARDS ANGELL PALMER & DODGE 9/2/2011 \$ 319,418.69 EDWARDS ANGELL PALMER & DODGE 9/28/2011 \$ 300,000.00 ELSEVIER INC. 8/11/2011 \$ 13,034.54 ELSEVIER INC. 8/25/2011 \$ 18,297.94 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,236.72 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,628.15	DULANEY, AMY D	8/5/2011 \$	2,014.92
DULANEY, AMY D 9/16/2011 \$ 2,014.92 DULANEY, AMY D 9/30/2011 \$ 2,014.92 EAST TENNESSEE SCALE WORKS 7/21/2011 \$ 234.35 EDS 7/5/2011 \$ 10,376.00 EDWARDS ANGELL PALMER & DODGE 9/1/2011 \$ 1,274,709.77 EDWARDS ANGELL PALMER & DODGE 9/2/2011 \$ 319,418.69 EDWARDS ANGELL PALMER & DODGE 9/28/2011 \$ 300,000.00 ELSEVIER INC. 8/11/2011 \$ 13,034.54 ELSEVIER INC. 8/25/2011 \$ 18,297.94 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,236.72 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,628.15	DULANEY, AMY D	8/19/2011	2,014.92
DULANEY, AMY D 9/30/2011 \$ 2,014.92 EAST TENNESSEE SCALE WORKS 7/21/2011 \$ 234.35 EDS 7/5/2011 \$ 10,376.00 EDWARDS ANGELL PALMER & DODGE 9/1/2011 \$ 1,274,709.77 EDWARDS ANGELL PALMER & DODGE 9/2/2011 \$ 319,418.69 EDWARDS ANGELL PALMER & DODGE 9/28/2011 \$ 300,000.00 ELSEVIER INC. 8/11/2011 \$ 13,034.54 ELSEVIER INC. 8/25/2011 \$ 18,297.94 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,236.72 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,628.15	DULANEY, AMY D	9/2/2011	2,014.92
EAST TENNESSEE SCALE WORKS 7/21/2011 \$ 234.35 EDS 7/5/2011 \$ 10,376.00 EDWARDS ANGELL PALMER & DODGE 9/1/2011 \$ 1,274,709.77 EDWARDS ANGELL PALMER & DODGE 9/2/2011 \$ 319,418.69 EDWARDS ANGELL PALMER & DODGE 9/28/2011 \$ 300,000.00 ELSEVIER INC. 8/11/2011 \$ 13,034.54 ELSEVIER INC. 8/25/2011 \$ 18,297.94 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,236.72 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,628.15	DULANEY, AMY D	9/16/2011	2,014.92
EDS 7/5/2011 \$ 10,376.00 EDWARDS ANGELL PALMER & DODGE 9/1/2011 \$ 1,274,709.77 EDWARDS ANGELL PALMER & DODGE 9/2/2011 \$ 319,418.69 EDWARDS ANGELL PALMER & DODGE 9/28/2011 \$ 300,000.00 ELSEVIER INC. 8/11/2011 \$ 13,034.54 ELSEVIER INC. 8/25/2011 \$ 18,297.94 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,236.72 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,628.15	DULANEY, AMY D	9/30/2011	2,014.92
EDWARDS ANGELL PALMER & DODGE 9/1/2011 \$ 1,274,709.77 EDWARDS ANGELL PALMER & DODGE 9/2/2011 \$ 319,418.69 EDWARDS ANGELL PALMER & DODGE 9/28/2011 \$ 300,000.00 ELSEVIER INC. 8/11/2011 \$ 13,034.54 ELSEVIER INC. 8/25/2011 \$ 18,297.94 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,236.72 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,628.15	EAST TENNESSEE SCALE WORKS	7/21/2011	234.35
EDWARDS ANGELL PALMER & DODGE 9/2/2011 \$ 319,418.69 EDWARDS ANGELL PALMER & DODGE 9/28/2011 \$ 300,000.00 ELSEVIER INC. 8/11/2011 \$ 13,034.54 ELSEVIER INC. 8/25/2011 \$ 18,297.94 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,236.72 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,628.15	EDS	7/5/2011 \$	10,376.00
EDWARDS ANGELL PALMER & DODGE 9/28/2011 \$ 300,000.00 ELSEVIER INC. 8/11/2011 \$ 13,034.54 ELSEVIER INC. 8/25/2011 \$ 18,297.94 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,236.72 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,628.15	EDWARDS ANGELL PALMER & DODGE	9/1/2011	1,274,709.77
ELSEVIER INC. 8/11/2011 \$ 13,034.54 ELSEVIER INC. 8/25/2011 \$ 18,297.94 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,236.72 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,628.15	EDWARDS ANGELL PALMER & DODGE	9/2/2011	319,418.69
ELSEVIER INC. 8/25/2011 \$ 18,297.94 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,236.72 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,628.15	EDWARDS ANGELL PALMER & DODGE	9/28/2011 \$	300,000.00
EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,236.72 EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,628.15	ELSEVIER INC.	8/11/2011	13,034.54
EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 2,628.15	ELSEVIER INC.	8/25/2011	18,297.94
	EMBLEM HEALTH SERVICES COMPANY LLC	8/25/2011	2,236.72
EMBLEM HEALTH SERVICES COMPANY LLC 8/25/2011 \$ 3,839.69	EMBLEM HEALTH SERVICES COMPANY LLC		•
	EMBLEM HEALTH SERVICES COMPANY LLC	8/25/2011	3,839.69

Name of creditor	Payment date	Amount paid
EMBLEM HEALTH SERVICES COMPANY LLC	8/25/2011	\$ 3,859.66
ENTERPRISE FLEET SERVICES	7/20/2011	\$ 84,066.86
ENTERPRISE FLEET SERVICES	8/23/2011	\$ 84,450.94
ENTERPRISE FLEET SERVICES	9/20/2011	\$ 81,268.16
EPL PATHOLOGY ARCHIVES, INC	7/14/2011	\$ 11,240.80
ESTEP 11, JAMES W.	7/21/2011	\$ 132.00
ETHOS HEALTH COMMUNICATIONS, INC.	7/5/2011	\$ 376,332.78
ETHOS HEALTH COMMUNICATIONS, INC.	8/11/2011	\$ 18,820.34
ETHOS HEALTH COMMUNICATIONS, INC.	8/25/2011	\$ 962.25
ETSU FOUNDATION	8/11/2011	\$ 235.00
ETSU FOUNDATION	8/11/2011	\$ 235.00
EVANS, TERRY	7/25/2011	\$ 75.00
EVANS, TERRY L.	7/8/2011	\$ 4,462.90
EVANS, TERRY L.	7/22/2011	\$ 5,952.12
EVINCE COMMUNICATIONS, LLC	7/5/2011	\$ 59,760.00
EVINCE COMMUNICATIONS, LLC	7/14/2011	\$ 45,500.00
EVINCE COMMUNICATIONS, LLC	8/18/2011	\$ 121,090.00
EVINCE COMMUNICATIONS, LLC	9/21/2011	\$ 62,780.00
EXPRESS SCRIPTS/DPS, INC.	7/14/2011	\$ 8,842.28
EXPRESS SCRIPTS/DPS, INC.	7/14/2011	\$ 139,767.40
EXPRESS SCRIPTS/DPS, INC.	8/18/2011	\$ 6,860.93
EXPRESS SCRIPTS/DPS, INC.	8/18/2011	\$ 121,159.27
FABRY, JOEL	7/8/2011	\$ 5,354.38
FABRY, JOEL	7/22/2011	\$ 5,354.40
FABRY, JOEL	8/5/2011	\$ 5,354.38
FABRY, JOEL	8/19/2011	\$ 5,354.38
FABRY, JOEL	9/2/2011	\$ 5,354.39
FABRY, JOEL	9/16/2011	\$ 5,354.39
FABRY, JOEL	9/30/2011	\$ 5,354.38
FARNAN LLP	8/25/2011	\$ 3,172.25
FARRELL, SUSAN	8/11/2011	\$ 50.00
FAUNBROOK CATERING	8/25/2011	\$ 685.20
FDA EXPORT CERTIFICATE PROGRAM	7/28/2011	\$ 5,950.00
FEDEX	7/5/2011	\$ 2,931.43
FEDEX	7/14/2011	\$ 7,012.86
FEDEX	7/21/2011	\$ 1,816.22
FEDEX	7/21/2011	\$ 3,590.66
FEDEX	8/4/2011	\$ 4,196.12
FEDEX	8/5/2011	\$ 614.82
FEDEX	8/11/2011	\$ 5,450.92
FEDEX	8/18/2011	\$ 5,815.53
FEDEX	8/25/2011	\$ 6,331.57
FEDEX	9/1/2011	\$ 1,466.68
FEDEX	9/8/2011	\$ 8,524.43
FEDEX	9/15/2011	\$ 185.01
FEDEX	9/21/2011	\$ 1,738.71
FEDEX FREIGHT	7/5/2011	
FEDEX FREIGHT	7/14/2011	\$ 109.74
FEDEX FREIGHT	7/21/2011	
FEDEX FREIGHT	9/8/2011	
FINNEGAN, APRIL	7/8/2011	
FINNEGAN, APRIL	7/22/2011	
FINNEGAN, APRIL	8/5/2011	
FINNEGAN, APRIL	8/19/2011	

Name of creditor	Payment date	Amount paid
FINNEGAN, APRIL	9/2/2011	\$ 870.01
FINNEGAN, APRIL	9/16/2011	
FINNEGAN, APRIL	9/30/2011	
FISHER CLINICAL SERVICES	7/14/2011	
FISHER CLINICAL SERVICES	7/21/2011	•
FISHER CLINICAL SERVICES	9/1/2011	
FISHER, GEORGE	7/5/2011	·
FISHER, GEORGE	7/8/2011	
FISHER, GEORGE	7/11/2011	-
FISHER, GEORGE	7/22/2011	
FISHER, GEORGE	7/25/2011	
FISHER, GEORGE	8/5/2011	
FISHER, GEORGE	8/8/2011	•
FISHER, GEORGE	8/15/2011	
FISHER, GEORGE	8/19/2011	
FISHER, GEORGE	9/2/2011	•
FISHER, GEORGE	9/8/2011	·
FISHER, GEORGE	9/15/2011	
FISHER, GEORGE	9/16/2011	
FISHER, GEORGE	9/21/2011	
FISHER, GEORGE	9/26/2011	
FISHER, GEORGE	9/30/2011	"
FISHER, LAURA	7/7/2011	" ,
FISHER, LAURA	7/11/2011	
FISHER, LAURA D	7/8/2011	
FISHER, LAURA D	7/22/2011	
FISHER, LAURA D	8/15/2011	
FLEET RESPONSE	7/14/2011	
FLEET RESPONSE	8/4/2011	
FLEET RESPONSE	8/11/2011	=
FLEET RESPONSE	8/25/2011	
FLEET RESPONSE	9/8/2011	
FLEET RESPONSE	9/21/2011	
FLETCHER/CSI	8/18/2011	
FLORIDA OB/GYN SOCIETY	7/7/2011	
FOOD AND DRUG ADMINISTRATION	9/8/2011	
FOOD AND DRUG ADMINISTRATION	9/15/2011	• •
FOUNDATION FOR RESEARCH & EDUCATION IN DERM	9/1/2011	
FRANKLIN PHARMA SERVICES, LLC	7/21/2011	
FRANKLIN PHARMA SERVICES, LLC	8/11/2011	
FRANKLIN PHARMA SERVICES, LLC	9/21/2011	•
GENCO SUPPLY CHAIN SOLUTIONS	7/14/2011	
GENCO SUPPLY CHAIN SOLUTIONS	8/11/2011	
GILLINGHAM, PAMELA A	7/8/2011	
GILLINGHAM, PAMELA A	7/22/2011	* *
GILLINGHAM, PAMELA A	8/5/2011	
GILLINGHAM, PAMELA A	8/15/2011	•
GILLINGHAM, PAMELA A	8/19/2011	•
GILLINGHAM, PAMELA A	9/2/2011	•
GILLINGHAM, PAMELA A	9/2/2011	
GILLINGHAM, PAMELA A	9/30/2011	
GLOBAL KNOWLEDGE	8/8/2011	
GOLDEN, LARRY J		
GORDON, KEILONE	8/15/2011	· ·
GORDON, REILOINE	7/8/2011	\$ 1,903.03

GORDON, KELLONE	Name of creditor		Payment date	Amount	paid
GORDON, KEILONE GREEN, TERESA L. (7/8/2011 \$ 1,908.15 GREEN, TERESA L. (8/19/2011 \$ 1,008.15 GREEN, TERESA L. (8/19/2011 \$ 1,008.15 GREEN, TERESA L. (9/2/2011 \$ 1,008.15 GREEN, TERESA L. (9/16/2011 \$ 1,306.33 GREEN, TERESA L. (9/16/2011 \$ 1,306.33 GREEN, TERESA L. (9/16/2011 \$ 1,306.33 GREEN, TERESA L. (9/16/2011 \$ 1,506.33 GREEN, TERESA	GORDON, KEILONE		7/22/2011	\$	1,903.03
GORDON, KEHLONE					-
GORDON, KEILONE					
GORDON, KEILONE 9/1/2011 \$ 1,903.03 GORDON, KEILONE 9/30/2011 \$ 1,166.85 GREEN, TERESA L 7/8/2011 \$ 1,166.85 GREEN, TERESA L 1/8/2011 \$ 1,008.16 GREEN, TERESA L 8/19/2011 \$ 1,008.16 GREEN, TERESA L 8/19/2011 \$ 1,008.16 GREEN, TERESA L 9/16/2011 \$ 1,008.16 GREEN, TERESA L 9/16/2011 \$ 1,008.15 GREEN, TERESA L 9/16/2011 \$ 1,008.16 GROSS, DIANE 7/14/2011 \$ 150.00 GROSS, DIANE 8/11/2011 \$ 150.00 GROSS, DIANE 9/18/2011 \$ 150.00 GROSS, DIANE 9/14/2011 \$ 1,009.00 GSW 7/14/2011 \$ 1,009.00 GSW 7/14/2011 \$ 1,009.00 GSW 7/14/2011 \$ 1,009.00 GTB RESEARCH, INC 7/12/2011 \$ 1,009.00 GTB RESE					
GORDON, KEILONE 9/16/2011 \$ 4,703.36 GORDON, KEILONE 9/30/2011 \$ 1,166.85 GREEN, TERESA L 7/22/2011 \$ 1,008.16 GREEN, TERESA L 8/5/2011 \$ 1,008.15 GREEN, TERESA L 8/19/2011 \$ 1,008.15 GREEN, TERESA L 9/2/2011 \$ 1,008.15 GREEN, TERESA L 9/16/2011 \$ 1,008.15 GROCHOW, RYAN 8/18/2011 \$ 1,008.25 GROSS, DIANE 8/11/2011 \$ 5,000 GROSS, DIANE 9/18/2011 \$ 5,000 GROSS, DIANE 9/18/2011 \$ 5,000 GROSS, DIANE 9/18/2011 \$ 5,000 GROSS, DIANE 9/12/2011 \$ 5,000 GROSS, DIANE 9/12/2011 \$ 2,600 GSW 7/14/2011 \$ 2,600 GSW 7/14/2011 \$ 2,600 GSW 7/14/2011 \$ 2,000 GRUER, TERESARCH, INC. 9/21/2011 \$ 2,600 GTB RESEARCH, INC. 9/21/2011 \$ 2,600 GTB RESEARCH, INC. 9/21/2011 \$ 2,000 GTB RESEARCH, INC. <t< td=""><td></td><td></td><td></td><td></td><td>-</td></t<>					-
GORDON, KEILONE 9/30/2011 \$ 4703.36 GREEN, TERESA L 7/8/2011 \$ 1,068.15 GREEN, TERESA L 8/5/2011 \$ 1,068.15 GREEN, TERESA L 8/19/2011 \$ 1,068.15 GREEN, TERESA L 8/19/2011 \$ 1,068.15 GREEN, TERESA L 9/16/2011 \$ 1,306.33 GREEN, TERESA L 9/16/2011 \$ 1,306.33 GREEN, TERESA L 9/16/2011 \$ 1,306.33 GROSE, DIANE 7/14/2011 \$ 750.00 GROSS, DIANE 8/18/2011 \$ 750.00 GROSS, DIANE 9/15/2011 \$ 1,315.00 GSW 7/14/2011 \$ 750.00 GSW 7/14/2011 \$ 55.015.00 GSW 7/14/2011 \$ 50.00 GTB RESEARCH, INC. 9/15/2011 \$ 50.00 GTB RESEARCH, INC. 9/12/2011 \$ 50.00 GTB RESEARCH, INC. 9/12/2011 \$ 50.00 HALE, KATHERINE 7/8/2011 \$ 50.00 HALE, KATHERINE 8/5/2011 \$ 50.00 HA					,
GREEN, TERESA L 7/8/2011 \$ 1,068.16 GREEN, TERESA L 7/22/2011 \$ 1,068.16 GREEN, TERESA L 8/5/2011 \$ 1,068.16 GREEN, TERESA L 8/19/2011 \$ 1,068.16 GREEN, TERESA L 9/16/2011 \$ 1,068.16 GREEN, TERESA L 9/16/2011 \$ 1,306.33 GREEN, TERESA L 9/30/2011 \$ 1,306.33 GREEN, TERESA L 9/30/2011 \$ 150.03 GROCHOW, RYAN 8/18/2011 \$ 150.00 GROSS, DIANE 7/14/2011 \$ 150.00 GROSS, DIANE 9/8/2011 \$ 570.00 GROSS, DIANE 9/15/2011 \$ 150.00 GSW 7/14/2011 \$ 550.01 GSW 7/14/2011 \$ 550.05 GSW 7/21/2011 \$ 550.05 GSW 7/21/2011 \$ 550.05 GTB RESEARCH, INC. 9/21/2011 \$ 50.03 HALE, KATHERINE 7/8/2011 \$ 50.03 HALE, KATHERINE 8/5/2011 \$ 50.03 HALE, KATHERINE 8/19/2011 \$ 50.03 HALE, KATHERINE 9/16/2011 \$ 50.03	•				•
GREEN, TERESA L 7/22/2011 \$ 1,008.15 GREEN, TERESA L 8/5/2011 \$ 1,008.15 GREEN, TERESA L 9/12/2011 \$ 1,008.15 GREEN, TERESA L 9/12/2011 \$ 1,306.33 GREEN, TERESA L 9/30/2011 \$ 1,306.33 GREEN, TERESA L 9/30/2011 \$ 1,306.33 GROCHOW, RYAN 8/18/2011 \$ 750.00 GROSS, DIANE 8/11/2011 \$ 750.00 GROSS, DIANE 9/18/2011 \$ 750.00 GROSS, DIANE 9/18/2011 \$ 750.00 GROSS, DIANE 9/18/2011 \$ 750.00 GSW 9/14/2011 \$ 750.00 GSW 7/14/2011 \$ 750.00 GSW 7/14/2011 \$ 72.00 GSW 7/21/2011 \$ 72.00 GTB RESEARCH, INC. 7/21/2011 \$ 70.00 GTB RESEARCH, INC. 9/21/2011 \$ 70.00 GTB RESEARCH, INC. 9/21/2011 \$ 70.00 HALE, KATHERINE 7/8/2011 \$ 70.00 HALE, KATHERINE 7/8/2011 \$ 70.00 HALE, KATHERINE 8/15/2011 \$ 70.00 HALE, KATHERINE 9/16/2011 \$ 70.00 HALE, KATHERINE 9/30/2011 \$ 70.00 HALE, KATHERINE 9/30/2011 \$ 70.00 <td>•</td> <td></td> <td></td> <td></td> <td></td>	•				
GREEN, TERESA L 8/19/2011 \$ 1,008.15 GREEN, TERESA L 9/2/2011 \$ 1,008.15 GREEN, TERESA L 9/16/2011 \$ 1,306.32 GREEN, TERESA L 9/16/2011 \$ 1,306.32 GREEN, TERESA L 9/16/2011 \$ 1,306.32 GROEN, TERESA L 9/16/2011 \$ 226.86 GROSK, DIANE 7/14/2011 \$ 735.00 GROSS, DIANE 9/16/2011 \$ 131.50 GROW 7/14/2011 \$ 25.50 GRW 7/14/2011 \$ 25.50 GSW 7/21/2011 \$ 25.50 GTB RESEARCH, INC. 9/21/2011 \$ 25.50 GTB RESEARCH, INC. 9/21/2011 \$ 50.34 HALE, KATHERINE 8/5/2011 \$ 50.34 HALE, KATHERINE 8/5/2011 \$ 50.34 HALE, KATHERINE 8/15/2011 \$ 50.34 HALE, KATHERINE 8/16/2011 \$ 50.34					•
GREEN, TERESA L \$9/2/2011 \$ 1,008.15 GREEN, TERESA L \$9/2/2011 \$ 1,306.35 GREEN, TERESA L \$9/6/2011 \$ 1,306.32 GREEN, TERESA L \$9/0/2011 \$ 1,306.32 GROCHOW, RYAN \$8/18/2011 \$ 226.86 GROSS, DIANE 7/14/2011 \$ 150.00 GROSS, DIANE \$9/15/2011 \$ 570.00 GROSS, DIANE \$9/15/2011 \$ 570.00 GSUS \$1/15/2011 \$ 1,315.00 GSW 7/14/2011 \$ 55,015.00 GSW 7/14/2011 \$ 2,687.50 GSW 7/12/2011 \$ 2,505.00 GTB RESEARCH, INC. 7/21/2011 \$ 2,687.50 GTB RESEARCH, INC. 7/21/2011 \$ 2,687.50 GTB RESEARCH, INC. 7/21/2011 \$ 2,687.50 HALE, KATHERINE 7/8/2011 \$ 590.34 HALE, KATHERINE 8/15/2011 \$ 590.34 HALE, KATHERINE 8/15/2011 \$ 590.34 HALE, KATHERINE 8/15/2011 \$ 590.34 HALE, KATHERINE 9/2/2011 \$ 590.34 HALLIDAY, SEAN B					· · - ·
GREEN, TERESA L 9/2/2011 \$ 1,306.35 GREEN, TERESA L 9/16/2011 \$ 1,306.32 GREEN, TERESA L 9/30/2011 \$ 1,306.32 GROCHOW, RYAN 8/18/2011 \$ 226.86 GROSS, DIANE 7/14/2011 \$ 753.00 GROSS, DIANE 9/8/2011 \$ 757.00 GROSS, DIANE 9/15/2011 \$ 1,315.00 GROSS, DIANE 9/16/2011 \$ 1,315.00 GSW 7/14/2011 \$ 2,250.00 GSW 7/14/2011 \$ 2,250.00 GSW 7/21/2011 \$ 2,250.00 GTB RESEARCH, INC. 9/21/2011 \$ 2,267.50 GTB RESEARCH, INC. 9/21/2011 \$ 590.35 HALE, KATHERINE 7/82/2011 \$ 590.35 HALE, KATHERINE 8/5/2011 \$ 590.35 HALE, KATHERINE 8/5/2011 \$ 590.34 HALE, KATHERINE 9/2/2011 \$ 590.35 HALE, KATHERINE 9/2/2011 \$ 590.34 HALE, KATHERINE 9/2/2011 \$ 590.34 HALE, KATHERINE 9/2/2011 \$ 590.35 HALE, KATHERINE 9/2/2011 \$ 590.35 HALE, KATHERINE 9/2/2011 \$ 590.35 HALIDAY, SEAN B 7/8/2011 \$ 590.35 HALLIDAY, SEAN B 7/	· · · · · · · · · · · · · · · · · · ·				
GREEN, TERESA L 9/16/2011 \$ 1,306.33 GREEN, TERESA L 9/30/2011 \$ 1,306.32 GROCHOW, RYAN 8/18/2011 \$ 150.00 GROSS, DIANE 7/14/2011 \$ 150.00 GROSS, DIANE 8/11/2011 \$ 750.00 GROSS, DIANE 9/8/2011 \$ 570.00 GROSS, DIANE 9/8/2011 \$ 150.00 GROSS, DIANE 9/8/2011 \$ 455.00 GSW 7/14/2011 \$ 455.00 GSW 7/21/2011 \$ 55,015.00 GTB RESEARCH, INC. 7/21/2011 \$ 2,6687.50 GTB RESEARCH, INC. 7/21/2011 \$ 2,6887.50 HALE, KATHERINE 7/8/2011 \$ 590.34 HALE, KATHERINE 7/8/2011 \$ 590.34 HALE, KATHERINE 8/15/2011 \$ 590.34 HALE, KATHERINE 8/15/2011 \$ 590.34 HALE, KATHERINE 8/15/2011 \$ 590.34 HALE, KATHERINE 9/2/2011 \$ 590.34 HALIDAY, SEAN B 7/8/2011 \$ 590.34 HALIDAY, SEAN B 7/8/2011 \$ 590.34 HALLIDAY, SEAN B <t< td=""><td></td><td></td><td></td><td></td><td>•</td></t<>					•
GREICN, TERRESA L 9/30/2011 \$ 1,306.32 GROCHOW, RYAN 8/18/2011 \$ 226.88 GROSS, DIANE 7/14/2011 \$ 755.00 GROSS, DIANE 9/18/2011 \$ 755.00 GROSS, DIANE 9/18/2011 \$ 755.00 GROSS, DIANE 9/18/2011 \$ 1,315.00 GSW 7/14/2011 \$ 495.00 GSW 7/21/2011 \$ 5,015.00 GTB RESEARCH, INC. 9/21/2011 \$ 2,687.50 GTB RESEARCH, INC. 9/21/2011 \$ 500.34 HALE, KATHERINE 7/22/2011 \$ 500.34 HALE, KATHERINE 8/5/2011 \$ 500.34 HALE, KATHERINE 8/5/2011 \$ 500.34 HALE, KATHERINE 8/15/2011 \$ 500.34 HALE, KATHERINE 9/2/2011 \$ 500.34 HALE, KATHERINE 9/2/2011 \$ 500.34 HALE, KATHERINE 9/30/2011 \$ 500.34 HALLIDAY, SEAN B 7/22/2011 \$ 500.34 HALLIDAY, SEAN B 8/5/2011 \$ 2,001.59 HALLIDAY, SEAN B 9/30/2011 \$ 2,001.60 HAMPTON, KENNETH WA					•
GROCHOW, RYAN 8/18/2011 \$ 226.86 GROSS, DIANE 7/14/2011 \$ 735.00 GROSS, DIANE 8/11/2011 \$ 570.00 GROSS, DIANE 9/8/2011 \$ 570.00 GROSS, DIANE 9/8/2011 \$ 570.00 GSI US 9/15/2011 \$ 550.01 GSW 7/14/2011 \$ 55.015.00 GSW 7/21/2011 \$ 2250.00 GTB RESEARCH, INC. 7/21/2011 \$ 2680.50 GTB RESEARCH, INC. 7/8/2011 \$ 590.34 HALE, KATHERINE 7/8/2011 \$ 590.35 HALE, KATHERINE 8/5/2011 \$ 590.34 HALE, KATHERINE 8/15/2011 \$ 590.34 HALE, KATHERINE 8/15/2011 \$ 590.34 HALE, KATHERINE 9/16/2011 \$ 590.35 HALLIDAY, SEAN B 7/8/2011 \$ 2901.60 HALLIDAY, SEAN B 7/8/2011 \$ 2901.60 HALLIDAY, SEAN B 9/2/2011 \$ 2901.60 HALLIDAY, SEAN B 9/2/2011 \$ 2901.60 HALLIDAY, SEAN B 9/	· · · · · · · · · · · · · · · · · · ·				
GROSS, DIANE 7/14/2011 \$ 755.00 GROSS, DIANE 8/11/2011 \$ 755.00 GROSS, DIANE 9/8/2011 \$ 1,315.00 GSU 9/15/2011 \$ 1,315.00 GSW 7/14/2011 \$ 55,015.00 GSW 7/21/2011 \$ 2,250.00 GTB RESEARCH, INC. 9/21/2011 \$ 2,687.50 GTB RESEARCH, INC. 9/21/2011 \$ 590.34 HALE, KATHERINE 7/8/2011 \$ 590.34 HALE, KATHERINE 8/5/2011 \$ 590.34 HALE, KATHERINE 8/15/2011 \$ 590.34 HALE, KATHERINE 8/15/2011 \$ 590.34 HALE, KATHERINE 9/2/2011 \$ 590.34 HALLDAY, SEAN B 8/5/2011 \$ 2,901.59 HALLIDAY, SEAN B 8/5/2011 \$ 2,901.59 HALLIDAY, SEAN B 9/1/2011 \$ 2,901.59 HALLIDAY, SEAN B 9/1/2011 \$ 2,901.59 HAMPTON, KENNETH WADE	•				
GROSS, DIANE 8/11/2011 \$ 735.00 GROSS, DIANE 9/8/2011 \$ 1750.00 GSI US 9/8/2011 \$ 1,315.00 GSW 7/14/2011 \$ 495.00 GSW 7/21/2011 \$ 2,250.00 GTB RESEARCH, INC. 9/21/2011 \$ 2,687.50 HALE, KATHERINE 7/8/2011 \$ 500.34 HALE, KATHERINE 7/8/2011 \$ 500.34 HALE, KATHERINE 8/15/2011 \$ 500.34 HALE, KATHERINE 8/15/2011 \$ 43.75 HALE, KATHERINE 8/15/2011 \$ 500.34 HALE, KATHERINE 8/15/2011 \$ 500.34 HALE, KATHERINE 8/15/2011 \$ 500.34 HALE, KATHERINE 9/16/2011 \$ 500.34 HALE, KATHERINE 9/16/2011 \$ 500.34 HALIDAY, SEAN B 7/16/2011 \$ 2,001.59 HALLIDAY, SEAN B 8/5/2011 \$ 2,001.59 HALLIDAY, SEAN B 8/5/2011 \$ 2,001.60 HALLIDAY, SEAN B 9/1/22/2011 \$ 2,001.60 HALLIDAY, SEAN B 9/1/22/2011	GROSS, DIANE				
GROSS, DIANE 9/8/2011 \$ 570.00 GSI US 9/15/2011 \$ 1,315.00 GSW 7/14/2011 \$ 495.00 GSW 7/21/2011 \$ 55,015.00 GTB RESEARCH, INC. 9/21/2011 \$ 2,250.00 GTB RESEARCH, INC. 9/21/2011 \$ 50,34 HALE, KATHERINE 7/8/2011 \$ 590.34 HALE, KATHERINE 8/5/2011 \$ 590.34 HALE, KATHERINE 8/15/2011 \$ 590.34 HALE, KATHERINE 8/19/2011 \$ 590.34 HALE, KATHERINE 8/19/2011 \$ 590.34 HALE, KATHERINE 9/2/2011 \$ 590.35 HALE, KATHERINE 9/2/2011 \$ 590.35 HALE, KATHERINE 9/30/2011 \$ 590.35 HALLIDAY, SEAN B 7/8/2011 \$ 590.35 HALLIDAY, SEAN B 7/8/2011 \$ 590.35 HALLIDAY, SEAN B 8/19/2011 \$ 2,901.59 HALLIDAY, SEAN B 8/5/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.59 HALLIDAY, SEAN B 9/30/2011					
GSI US 9/15/2011 \$ 1,315.00 GSW 7/14/2011 \$ 495.00 GSW 7/21/2011 \$ 55,015.00 GTB RESEARCH, INC. 7/21/2011 \$ 2,250.00 GTB RESEARCH, INC. 9/21/2011 \$ 500.35 HALE, KATHERINE 7/8/2011 \$ 590.35 HALE, KATHERINE 8/5/2011 \$ 590.35 HALE, KATHERINE 8/5/2011 \$ 43.75 HALE, KATHERINE 8/19/2011 \$ 590.34 HALE, KATHERINE 8/19/2011 \$ 590.34 HALE, KATHERINE 8/19/2011 \$ 590.34 HALE, KATHERINE 9/2/2011 \$ 590.34 HALE, KATHERINE 9/16/2011 \$ 590.34 HALLDAY, SEAN B 7/8/2011 \$ 2,901.60 HALLIDAY, SEAN B 7/8/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.60 HAMPTON,					
GSW 7/14/2011 \$ 55,015.00 GSW 7/21/2011 \$ 55,015.00 GTB RESEARCH, INC. 7/21/2011 \$ 2,250.00 GTB RESEARCH, INC. 9/21/2011 \$ 2,687.50 HALE, KATHERINE 7/8/2011 \$ 590.34 HALE, KATHERINE 7/22/2011 \$ 590.34 HALE, KATHERINE 8/5/2011 \$ 590.34 HALE, KATHERINE 8/15/2011 \$ 590.34 HALE, KATHERINE 8/15/2011 \$ 590.34 HALE, KATHERINE 8/19/2011 \$ 590.34 HALE, KATHERINE 9/16/2011 \$ 590.33 HALE, KATHERINE 9/16/2011 \$ 590.34 HALL, KATHERINE 9/16/2011 \$ 590.34 HALLDAY, SEAN B 7/8/2011 \$ 2,901.59 HALLIDAY, SEAN B 7/22/2011 \$ 2,901.60 HALLIDAY, SEAN B 8/5/2011 \$ 2,901.60 HALLIDAY, SEAN B 8/5/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.60					
GSW 7/21/2011 \$ 55,015.00 GTB RESEARCH, INC. 7/21/2011 \$ 2,250.00 GTB RESEARCH, INC. 9/21/2011 \$ 2,687.50 HALE, KATHERINE 7/8/2011 \$ 590.34 HALE, KATHERINE 7/22/2011 \$ 590.35 HALE, KATHERINE 8/15/2011 \$ 590.35 HALE, KATHERINE 8/15/2011 \$ 590.34 HALE, KATHERINE 9/2/2011 \$ 590.34 HALE, KATHERINE 9/2/2011 \$ 590.35 HALE, KATHERINE 9/16/2011 \$ 590.35 HALE, KATHERINE 9/16/2011 \$ 590.35 HALLIDAY, SEAN B 7/8/2011 \$ 2,901.59 HALLIDAY, SEAN B 7/8/2011 \$ 2,901.59 HALLIDAY, SEAN B 7/2/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/2/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.60 HAMPTON, KENNETH WADE 7/8/2011 \$ 1,409.71 HAMPTO			· ·	-	•
GTB RESEARCH, INC. 7/21/2011 \$ 2,250.00 GTB RESEARCH, INC. 9/21/2011 \$ 2,687.50 HALE, KATHERINE 7/8/2011 \$ 590.34 HALE, KATHERINE 7/22/2011 \$ 590.35 HALE, KATHERINE 8/15/2011 \$ 590.34 HALE, KATHERINE 8/15/2011 \$ 590.34 HALE, KATHERINE 8/19/2011 \$ 590.34 HALE, KATHERINE 9/2/2011 \$ 590.34 HALE, KATHERINE 9/2/2011 \$ 590.35 HALE, KATHERINE 9/30/2011 \$ 590.35 HALE, KATHERINE 9/30/2011 \$ 590.35 HALLIDAY, SEAN B 9/30/2011 \$ 590.35 HALLIDAY, SEAN B 7/8/2011 \$ 2,901.59 HALLIDAY, SEAN B 8/5/2011 \$ 2,901.59 HALLIDAY, SEAN B 8/5/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/2/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/2/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/30/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.60 HAMPTON, KENNETH WADE 7/8/2011 \$ 1,409.71 HAMPTON, KENNETH WADE 8/5/2011 \$ 1,409.71 HAMPTON, KENNETH WADE 9/30/2011 \$ 1,409.71 HAMPTON, KENNETH WADE 9/					
GTB RESEARCH, INC. 9/21/2011 \$ 2,687.50 HALE, KATHERINE 7/8/22/2011 \$ 590.34 HALE, KATHERINE 7/22/2011 \$ 590.35 HALE, KATHERINE 8/5/2011 \$ 590.34 HALE, KATHERINE 8/15/2011 \$ 43.75 HALE, KATHERINE 8/15/2011 \$ 590.34 HALE, KATHERINE 9/2/2011 \$ 590.33 HALE, KATHERINE 9/30/2011 \$ 590.35 HALLIDAY, SEAN B 7/8/2011 \$ 2,901.59 HALLIDAY, SEAN B 7/8/2011 \$ 2,901.59 HALLIDAY, SEAN B 8/19/2011 \$ 2,901.60 HALLIDAY, SEAN B 8/19/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/2/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 1,409.71 HALLIDAY, SEAN B 9/16/2011 \$ 1,200.30 HAMPTON, KENNETH WADE 7/8/2011 \$ 1,409.71 HAMPTON, KENNETH WADE 8/5/2011 \$ 1,520.03 <tr< td=""><td>GTB RESEARCH, INC.</td><td></td><td></td><td></td><td>•</td></tr<>	GTB RESEARCH, INC.				•
HALE, KATHERINE 7/8/2011 \$ 590.35 HALE, KATHERINE 8/5/2011 \$ 590.35 HALE, KATHERINE 8/5/2011 \$ 43.75 HALE, KATHERINE 8/15/2011 \$ 590.34 HALE, KATHERINE 8/19/2011 \$ 590.34 HALE, KATHERINE 9/2/2011 \$ 590.33 HALE, KATHERINE 9/30/2011 \$ 590.35 HALE, KATHERINE 9/30/2011 \$ 590.35 HALIDAY, SEAN B 7/8/2011 \$ 2,901.59 HALLIDAY, SEAN B 7/8/2011 \$ 2,901.59 HALLIDAY, SEAN B 8/5/2011 \$ 2,901.60 HALLIDAY, SEAN B 8/19/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 1,402.11 HAMP					-
HALE, KATHERINE				•	
HALE, KATHERINE 8/5/2011 \$ 90.34 HALE, KATHERINE 8/15/2011 \$ 43.75 HALE, KATHERINE 8/19/2011 \$ 590.34 HALE, KATHERINE 9/2/2011 \$ 590.33 HALE, KATHERINE 9/16/2011 \$ 590.34 HALE, KATHERINE 9/30/2011 \$ 590.35 HALLIDAY, SEAN B 7/8/2011 \$ 2,901.59 HALLIDAY, SEAN B 8/5/2011 \$ 2,901.60 HALLIDAY, SEAN B 8/5/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/30/2011 \$ 1,400.71 HALLIDAY, SEAN B 9/30/2011 \$ 1,400.71	·				
HALE, KATHERINE 8/15/2011 \$ 43.75 HALE, KATHERINE 8/19/2011 \$ 590.34 HALE, KATHERINE 9/2/2011 \$ 590.34 HALE, KATHERINE 9/16/2011 \$ 590.35 HALE, KATHERINE 9/30/2011 \$ 590.35 HALLIDAY, SEAN B 7/8/2011 \$ 2,901.59 HALLIDAY, SEAN B 8/5/2011 \$ 2,901.69 HALLIDAY, SEAN B 8/5/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/30/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/30/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/30/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.60 HAMPTON, KENNETH WADE 7/8/2011 \$ 1,400.71 HAMPTON, KENNETH WADE 8/5/2011 \$ 1,422.19 HAMPTON, KENNETH WADE 9/16/2011 \$ 1,454.80	·				
HALE, KATHERINE 9/2/2011 \$ 590.34 HALE, KATHERINE 9/2/2011 \$ 590.33 HALE, KATHERINE 9/16/2011 \$ 590.33 HALE, KATHERINE 9/30/2011 \$ 590.35 HALLE, KATHERINE 9/30/2011 \$ 590.35 HALLIDAY, SEAN B 7/8/2011 \$ 2,901.59 HALLIDAY, SEAN B 7/2/2011 \$ 2,901.60 HALLIDAY, SEAN B 8/5/2011 \$ 2,901.60 HALLIDAY, SEAN B 8/5/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/2/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/2/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.60 HAMPTON, KENNETH WADE 9/30/2011 \$ 1,409.71 HAMPTON, KENNETH WADE 8/5/2011 \$ 1,520.03 HAMPTON, KENNETH WADE 8/5/2011 \$ 1,422.19 HAMPTON, KENNETH WADE 9/2/2011 \$ 1,451.69 HAMPTON, KENNETH WADE 9/16/2011 \$ 1,451.69 HAMPTON, KENNETH WADE 9/16/2011 \$ 1,454.80 HAMPTON, KENNETH WADE 9/30/2011 \$ 1,612.44 HANNA'S PHARMACEUTICAL SUPPLY CO. 7/14/2011 \$ 705.55 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/25/2011 \$ 25.69 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/25/2011 \$ 25.69 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/25/2011 \$ 25.69 HANNA'S PHARMACEUTICAL SUPPLY CO. 9/21/2011 \$ 400.80 HARMON'S LAWN CARE 8/4/2011 \$ 400.00 HARMON'S LAWN CARE 8/18/2011 \$ 220.00 HARMON'S LAWN CARE 8/18/2011 \$ 220.00					
HALE, KATHERINE 9/2/2011 \$ 590.33 HALE, KATHERINE 9/16/2011 \$ 590.34 HALE, KATHERINE 9/30/2011 \$ 590.35 HALLIDAY, SEAN B 7/8/2011 \$ 2,901.59 HALLIDAY, SEAN B 7/22/2011 \$ 2,901.60 HALLIDAY, SEAN B 8/5/2011 \$ 2,901.60 HALLIDAY, SEAN B 8/19/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/2/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/30/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/30/2011 \$ 1,409.71 HAMPTON, KENNETH WADE 8/5/2011 \$ 1,499.71 HAMPTON, KENNETH WADE 8/5/2011 \$ 1,193.86 HAMPTON, KENNETH WADE 9/30/2011 \$ 1,151.62	HALE, KATHERINE				
HALE, KATHERINE 9/16/2011 \$ 590.34 HALE, KATHERINE 9/30/2011 \$ 590.35 HALLIDAY, SEAN B 7/8/2011 \$ 2,901.59 HALLIDAY, SEAN B 7/22/2011 \$ 2,901.60 HALLIDAY, SEAN B 8/5/2011 \$ 2,901.60 HALLIDAY, SEAN B 8/19/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/2/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/30/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/30/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/30/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/30/2011 \$ 1,409.71 HAMPTON, KENNETH WADE 7/8/2011 \$ 1,409.71 HAMPTON, KENNETH WADE 8/5/2011 \$ 1,422.19 HAMPTON, KENNETH WADE 8/5/2011 \$ 1,193.86 HAMPTON, KENNETH WADE 9/30/2011 \$ 1,161.24 HAMPTON, KENNETH WADE 9/30/2011 \$ 1,61.62 HAMPTON, KENNETH WADE 9/30/2011 \$ 1					
HALE, KATHERINE 9/30/2011 \$ 590.35 HALLIDAY, SEAN B 7/8/2011 \$ 2,901.59 HALLIDAY, SEAN B 7/22/2011 \$ 2,901.60 HALLIDAY, SEAN B 8/5/2011 \$ 2,901.60 HALLIDAY, SEAN B 8/19/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/2/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/2/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/30/2011 \$ 2,901.60 HAMPTON, KENNETH WADE 7/8/2011 \$ 1,409.71 HAMPTON, KENNETH WADE 7/8/2011 \$ 1,409.71 HAMPTON, KENNETH WADE 8/5/2011 \$ 1,422.19 HAMPTON, KENNETH WADE 8/19/2011 \$ 1,451.69 HAMPTON, KENNETH WADE 9/2/2011 \$ 1,451.69 HAMPTON, KENNETH WADE 9/16/2011 \$ 1,451.69 HAMPTON, KENNETH WADE 9/30/2011 \$ 1,451.69 HAMPTON, KENNETH WADE 9/30/2011 \$ 1,451.69 HAMPTON, KENNETH WADE 9/16/2011 \$ 1,451.69 HAMPTON, KENNETH WADE 9/21/2011 \$ 705.55 HANNA'S PHARMACEUTICAL SUPPLY CO. 7/14/2011 \$ 705.55 HANNA'S PHARMACEUTICAL SU				-	
HALLIDAY, SEAN B 7/22/2011 \$ 2,901.60 HALLIDAY, SEAN B 8/5/2011 \$ 2,901.58 HALLIDAY, SEAN B 8/19/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/2/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.59 HALLIDAY, SEAN B 9/30/2011 \$ 2,901.60 HAMPTON, KENNETH WADE 7/8/2011 \$ 1,409.71 HAMPTON, KENNETH WADE 7/22/2011 \$ 1,520.03 HAMPTON, KENNETH WADE 8/5/2011 \$ 1,422.19 HAMPTON, KENNETH WADE 8/19/2011 \$ 1,451.69 HAMPTON, KENNETH WADE 9/16/2011 \$ 1,454.80 HAMPTON, KENNETH WADE 9/16/2011 \$ 1,454.80 HAMPTON, KENNETH WADE 9/30/2011 \$ 1,454.80 HANNA'S PHARMACEUTICAL SUPPLY CO. 7/14/2011 \$ 705.55 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/4/2011 \$ 80.74 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/25/2011 \$ 25.69 HANNA'S PHARMACEUTICAL SUPPLY CO. 9/21/2011 \$ 490.89 HARMON'S LAWN CARE 7/14/2011 \$ 440.00 HARMON'S LAWN CARE 8/42011 \$ 220.00	HALE, KATHERINE				
HALLIDAY, SEAN B 7/22/2011 \$ 2,901.60 HALLIDAY, SEAN B 8/5/2011 \$ 2,901.58 HALLIDAY, SEAN B 8/19/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/2/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/30/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/30/2011 \$ 2,901.60 HAMPTON, KENNETH WADE 7/8/2011 \$ 1,409.71 HAMPTON, KENNETH WADE 7/22/2011 \$ 1,520.03 HAMPTON, KENNETH WADE 8/5/2011 \$ 1,422.19 HAMPTON, KENNETH WADE 8/19/2011 \$ 1,451.69 HAMPTON, KENNETH WADE 9/16/2011 \$ 1,454.80 HAMPTON, KENNETH WADE 9/30/2011 \$ 1,454.80 HAMPTON, KENNETH WADE 9/30/2011 \$ 1,454.80 HAMPTON, KENNETH WADE 9/30/2011 \$ 705.55 HANNA'S PHARMACEUTICAL SUPPLY CO. 7/14/2011 \$ 705.55 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/4/2011 \$ 80.74 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/25/2011 \$ 25.69 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/25/2011 \$ 490.89 <	HALLIDAY, SEAN B		7/8/2011	\$	2,901.59
HALLIDAY, SEAN B 8/5/2011 \$ 2,901.68 HALLIDAY, SEAN B 8/19/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/2/2011 \$ 2,901.69 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.59 HALLIDAY, SEAN B 9/30/2011 \$ 2,901.60 HAMPTON, KENNETH WADE 7/8/2011 \$ 1,409.71 HAMPTON, KENNETH WADE 8/5/2011 \$ 1,520.03 HAMPTON, KENNETH WADE 8/5/2011 \$ 1,422.19 HAMPTON, KENNETH WADE 8/19/2011 \$ 1,451.69 HAMPTON, KENNETH WADE 9/2/2011 \$ 1,451.69 HAMPTON, KENNETH WADE 9/16/2011 \$ 1,454.80 HAMPTON, KENNETH WADE 9/30/2011 \$ 1,454.80 HAMPTON, KENNETH WADE 9/30/2011 \$ 1,454.80 HANPTON, FENNETH WADE 9/30/2011 \$ 1,454.80 HANPTON, FENNETH WADE 9/30/2011 \$ 1,454.80 HANNA'S PHARMACEUTICAL SUPPLY CO. 7/14/2011 \$ 705.55 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/4/2011 \$ 80.74 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/25/2011 \$ 490.89	HALLIDAY, SEAN B				
HALLIDAY, SEAN B 8/19/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/2/2011 \$ 2,901.60 HALLIDAY, SEAN B 9/16/2011 \$ 2,901.59 HALLIDAY, SEAN B 9/30/2011 \$ 2,901.60 HAMPTON, KENNETH WADE 7/8/2011 \$ 1,409.71 HAMPTON, KENNETH WADE 8/5/2011 \$ 1,520.03 HAMPTON, KENNETH WADE 8/19/2011 \$ 1,193.86 HAMPTON, KENNETH WADE 9/2/2011 \$ 1,451.69 HAMPTON, KENNETH WADE 9/16/2011 \$ 1,451.69 HAMPTON, KENNETH WADE 9/16/2011 \$ 1,451.69 HAMPTON, KENNETH WADE 9/30/2011 \$ 1,451.69 HAMPTON, KENNETH WADE 9/30/2011 \$ 1,451.69 HANDA'S PHARMACEUTICAL SUPPLY CO. 7/14/2011 \$ 705.55 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/4/2011 \$ 80.74 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/25/2011 25.69 HANNA'S PHARMACEUTICAL SUPPLY CO. 9/21/2011 490.89 HARMON'S LAWN CARE 7/14/2011 440.00 HARMON'S LAWN CARE 8/4/2011 220.00 HARMON'S LAWN CARE 8/18/2011 220.00 <td>HALLIDAY, SEAN B</td> <td></td> <td></td> <td></td> <td></td>	HALLIDAY, SEAN B				
HALLIDAY, SEAN B 9/16/2011 \$ 2,901.59 HALLIDAY, SEAN B 9/30/2011 \$ 2,901.60 HAMPTON, KENNETH WADE 7/8/2011 \$ 1,409.71 HAMPTON, KENNETH WADE 8/5/2011 \$ 1,520.03 HAMPTON, KENNETH WADE 8/5/2011 \$ 1,422.19 HAMPTON, KENNETH WADE 8/19/2011 \$ 1,193.86 HAMPTON, KENNETH WADE 9/2/2011 \$ 1,451.69 HAMPTON, KENNETH WADE 9/16/2011 \$ 1,454.80 HAMPTON, KENNETH WADE 9/30/2011 \$ 1,161.24 HANNA'S PHARMACEUTICAL SUPPLY CO. 7/14/2011 \$ 705.55 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/4/2011 \$ 80.74 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/25/2011 \$ 80.74 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/25/2011 \$ 25.69 HANNA'S PHARMACEUTICAL SUPPLY CO. 9/21/2011 \$ 490.89 HARMON'S LAWN CARE 7/14/2011 \$ 440.00 HARMON'S LAWN CARE 8/4/2011 \$ 220.00 HARMON'S LAWN CARE 8/18/2011 \$ 220.00	HALLIDAY, SEAN B		8/19/2011	\$	
HALLIDAY, SEAN B 9/16/2011 \$ 2,901.59 HALLIDAY, SEAN B 9/30/2011 \$ 2,901.60 HAMPTON, KENNETH WADE 7/8/2011 \$ 1,409.71 HAMPTON, KENNETH WADE 8/5/2011 \$ 1,520.03 HAMPTON, KENNETH WADE 8/5/2011 \$ 1,422.19 HAMPTON, KENNETH WADE 8/19/2011 \$ 1,193.86 HAMPTON, KENNETH WADE 9/2/2011 \$ 1,451.69 HAMPTON, KENNETH WADE 9/16/2011 \$ 1,454.80 HAMPTON, KENNETH WADE 9/30/2011 \$ 1,161.24 HANNA'S PHARMACEUTICAL SUPPLY CO. 7/14/2011 \$ 705.55 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/4/2011 \$ 80.74 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/25/2011 \$ 80.74 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/25/2011 \$ 25.69 HANNA'S PHARMACEUTICAL SUPPLY CO. 9/21/2011 \$ 490.89 HARMON'S LAWN CARE 7/14/2011 \$ 440.00 HARMON'S LAWN CARE 8/4/2011 \$ 220.00 HARMON'S LAWN CARE 8/18/2011 \$ 220.00	HALLIDAY, SEAN B				2,901.60
HAMPTON, KENNETH WADE 7/8/2011 \$ 1,409.71 HAMPTON, KENNETH WADE 7/22/2011 \$ 1,520.03 HAMPTON, KENNETH WADE 8/5/2011 \$ 1,422.19 HAMPTON, KENNETH WADE 8/19/2011 \$ 1,193.86 HAMPTON, KENNETH WADE 9/2/2011 \$ 1,451.69 HAMPTON, KENNETH WADE 9/16/2011 \$ 1,454.80 HAMPTON, KENNETH WADE 9/30/2011 \$ 1,161.24 HANNA'S PHARMACEUTICAL SUPPLY CO. 7/14/2011 \$ 705.55 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/4/2011 \$ 80.74 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/25/2011 \$ 25.69 HANNA'S PHARMACEUTICAL SUPPLY CO. 9/21/2011 \$ 490.89 HARMON'S LAWN CARE 7/14/2011 \$ 440.00 HARMON'S LAWN CARE 8/4/2011 \$ 220.00 HARMON'S LAWN CARE 8/18/2011 \$ 220.00	HALLIDAY, SEAN B	•			
HAMPTON, KENNETH WADE 7/22/2011 \$ 1,520.03 HAMPTON, KENNETH WADE 8/5/2011 \$ 1,422.19 HAMPTON, KENNETH WADE 8/19/2011 \$ 1,193.86 HAMPTON, KENNETH WADE 9/2/2011 \$ 1,451.69 HAMPTON, KENNETH WADE 9/16/2011 \$ 1,454.80 HAMPTON, KENNETH WADE 9/30/2011 \$ 1,161.24 HANNA'S PHARMACEUTICAL SUPPLY CO. 7/14/2011 \$ 705.55 HANNA'S PHARMACEUTICAL SUPPLY CO. 7/21/2011 \$ 717.34 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/4/2011 \$ 80.74 HANNA'S PHARMACEUTICAL SUPPLY CO. 9/21/2011 \$ 490.89 HARMON'S LAWN CARE 7/14/2011 \$ 490.89 HARMON'S LAWN CARE 8/4/2011 \$ 220.00 HARMON'S LAWN CARE 8/18/2011 \$ 220.00	HALLIDAY, SEAN B		9/30/2011	\$	2,901.60
HAMPTON, KENNETH WADE 8/5/2011 \$ 1,422.19 HAMPTON, KENNETH WADE 8/19/2011 \$ 1,193.86 HAMPTON, KENNETH WADE 9/2/2011 \$ 1,451.69 HAMPTON, KENNETH WADE 9/16/2011 \$ 1,454.80 HAMPTON, KENNETH WADE 9/30/2011 \$ 1,161.24 HANNA'S PHARMACEUTICAL SUPPLY CO. 7/14/2011 \$ 705.55 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/4/2011 \$ 80.74 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/25/2011 \$ 25.69 HANNA'S PHARMACEUTICAL SUPPLY CO. 9/21/2011 \$ 490.89 HARMON'S LAWN CARE 7/14/2011 \$ 440.00 HARMON'S LAWN CARE 8/4/2011 \$ 220.00 HARMON'S LAWN CARE 8/18/2011 \$ 220.00	HAMPTON, KENNETH WADE		7/8/2011	\$	1,409.71
HAMPTON, KENNETH WADE 8/19/2011 \$ 1,193.86 HAMPTON, KENNETH WADE 9/2/2011 \$ 1,451.69 HAMPTON, KENNETH WADE 9/16/2011 \$ 1,454.80 HAMPTON, KENNETH WADE 9/30/2011 \$ 1,161.24 HANNA'S PHARMACEUTICAL SUPPLY CO. 7/14/2011 \$ 705.55 HANNA'S PHARMACEUTICAL SUPPLY CO. 7/21/2011 \$ 80.74 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/25/2011 \$ 80.74 HANNA'S PHARMACEUTICAL SUPPLY CO. 9/21/2011 \$ 490.89 HARMON'S LAWN CARE 7/14/2011 \$ 490.89 HARMON'S LAWN CARE 8/4/2011 \$ 220.00 HARMON'S LAWN CARE 8/4/2011 \$ 220.00 HARMON'S LAWN CARE 8/18/2011 \$ 220.00	HAMPTON, KENNETH WADE				
HAMPTON, KENNETH WADE 8/19/2011 \$ 1,193.86 HAMPTON, KENNETH WADE 9/2/2011 \$ 1,451.69 HAMPTON, KENNETH WADE 9/16/2011 \$ 1,454.80 HAMPTON, KENNETH WADE 9/30/2011 \$ 1,161.24 HANNA'S PHARMACEUTICAL SUPPLY CO. 7/14/2011 \$ 705.55 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/4/2011 \$ 80.74 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/25/2011 \$ 25.69 HANNA'S PHARMACEUTICAL SUPPLY CO. 9/21/2011 \$ 490.89 HARMON'S LAWN CARE 7/14/2011 \$ 440.00 HARMON'S LAWN CARE 8/4/2011 \$ 220.00 HARMON'S LAWN CARE 8/18/2011 \$ 220.00	HAMPTON, KENNETH WADE		8/5/2011	\$	1,422.19
HAMPTON, KENNETH WADE 9/2/2011 \$ 1,451.69 HAMPTON, KENNETH WADE 9/16/2011 \$ 1,454.80 HAMPTON, KENNETH WADE 9/30/2011 \$ 1,161.24 HANNA'S PHARMACEUTICAL SUPPLY CO. 7/14/2011 \$ 705.55 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/4/2011 \$ 80.74 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/25/2011 \$ 25.69 HANNA'S PHARMACEUTICAL SUPPLY CO. 9/21/2011 \$ 490.89 HARMON'S LAWN CARE 7/14/2011 \$ 440.00 HARMON'S LAWN CARE 8/4/2011 \$ 220.00 HARMON'S LAWN CARE 8/18/2011 \$ 220.00	HAMPTON, KENNETH WADE				
HAMPTON, KENNETH WADE 9/30/2011 \$ 1,161.24 HANNA'S PHARMACEUTICAL SUPPLY CO. 7/14/2011 \$ 705.55 HANNA'S PHARMACEUTICAL SUPPLY CO. 7/21/2011 \$ 80.74 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/25/2011 \$ 25.69 HANNA'S PHARMACEUTICAL SUPPLY CO. 9/21/2011 \$ 490.89 HARMON'S LAWN CARE 7/14/2011 \$ 440.00 HARMON'S LAWN CARE 8/4/2011 \$ 220.00 HARMON'S LAWN CARE 8/18/2011 \$ 220.00	HAMPTON, KENNETH WADE		9/2/2011	\$	
HAMPTON, KENNETH WADE 9/30/2011 \$ 1,161.24 HANNA'S PHARMACEUTICAL SUPPLY CO. 7/14/2011 \$ 705.55 HANNA'S PHARMACEUTICAL SUPPLY CO. 7/21/2011 \$ 80.74 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/25/2011 \$ 25.69 HANNA'S PHARMACEUTICAL SUPPLY CO. 9/21/2011 \$ 490.89 HARMON'S LAWN CARE 7/14/2011 \$ 440.00 HARMON'S LAWN CARE 8/4/2011 \$ 220.00 HARMON'S LAWN CARE 8/18/2011 \$ 220.00	HAMPTON, KENNETH WADE		9/16/2011	\$	
HANNA'S PHARMACEUTICAL SUPPLY CO. 7/21/2011 \$ 717.34 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/4/2011 \$ 80.74 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/25/2011 \$ 25.69 HANNA'S PHARMACEUTICAL SUPPLY CO. 9/21/2011 \$ 490.89 HARMON'S LAWN CARE 7/14/2011 \$ 440.00 HARMON'S LAWN CARE 8/4/2011 \$ 220.00 HARMON'S LAWN CARE 8/18/2011 \$ 220.00	HAMPTON, KENNETH WADE	•	9/30/2011	\$	
HANNA'S PHARMACEUTICAL SUPPLY CO. 8/4/2011 \$ 80.74 HANNA'S PHARMACEUTICAL SUPPLY CO. 8/25/2011 \$ 25.69 HANNA'S PHARMACEUTICAL SUPPLY CO. 9/21/2011 \$ 490.89 HARMON'S LAWN CARE 7/14/2011 \$ 440.00 HARMON'S LAWN CARE 8/4/2011 \$ 220.00 HARMON'S LAWN CARE 8/18/2011 \$ 220.00	HANNA'S PHARMACEUTICAL SUPPLY CO.		7/14/2011	\$	
HANNA'S PHARMACEUTICAL SUPPLY CO. 8/25/2011 \$ 25.69 HANNA'S PHARMACEUTICAL SUPPLY CO. 9/21/2011 \$ 490.89 HARMON'S LAWN CARE 7/14/2011 \$ 440.00 HARMON'S LAWN CARE 8/4/2011 \$ 220.00 HARMON'S LAWN CARE 8/18/2011 \$ 220.00	HANNA'S PHARMACEUTICAL SUPPLY CO.		7/21/2011	\$	717.34
HANNA'S PHARMACEUTICAL SUPPLY CO. 9/21/2011 \$ 490.89 HARMON'S LAWN CARE 7/14/2011 \$ 440.00 HARMON'S LAWN CARE 8/4/2011 \$ 220.00 HARMON'S LAWN CARE 8/18/2011 \$ 220.00	HANNA'S PHARMACEUTICAL SUPPLY CO.		8/4/2011	\$	80.74
HANNA'S PHARMACEUTICAL SUPPLY CO. 9/21/2011 \$ 490.89 HARMON'S LAWN CARE 7/14/2011 \$ 440.00 HARMON'S LAWN CARE 8/4/2011 \$ 220.00 HARMON'S LAWN CARE 8/18/2011 \$ 220.00	HANNA'S PHARMACEUTICAL SUPPLY CO.		8/25/2011	. \$	25.69
HARMON'S LAWN CARE 7/14/2011 \$ 440.00 HARMON'S LAWN CARE 8/4/2011 \$ 220.00 HARMON'S LAWN CARE 8/18/2011 \$ 220.00	HANNA'S PHARMACEUTICAL SUPPLY CO.				490.89
HARMON'S LAWN CARE 8/4/2011 \$ 220.00 HARMON'S LAWN CARE 8/18/2011 \$ 220.00	HARMON'S LAWN CARE				
-,,,	HARMON'S LAWN CARE				
	HARMON'S LAWN CARE		8/18/2011	. \$	220.00
	HARMON'S LAWN CARE				220.00

Name of creditor	Payment date	Amount paid
HARMON'S LAWN CARE	9/15/2011 \$	220.00
HARMONY LABS	8/18/2011 \$	
HARTMAN, JASON	7/14/2011 \$	
HARTMAN, JASON	9/1/2011 \$	
HARTMAN, JASON	9/15/2011	· ·
HARVARD PILGRIM HEALTH CARE	9/15/2011	
HATFIELD, CHRISTOPHER	7/5/2011	
HATFIELD, CHRISTOPHER	7/8/2011	
HATFIELD, CHRISTOPHER	7/22/2011	,
HATFIELD, CHRISTOPHER	7/25/2011	17.34
HATFIELD, CHRISTOPHER	8/5/2011	2,383.18
HATFIELD, CHRISTOPHER	8/8/2011	14.00
HATFIELD, CHRISTOPHER	8/15/2011	1,181.23
HATFIELD, CHRISTOPHER	8/19/2011	2,383.19
HATFIELD, CHRISTOPHER	8/25/2011	4.00
HATFIELD, CHRISTOPHER	9/2/2011	2,545.89
HATFIELD, CHRISTOPHER	9/15/2011	127.36
HATFIELD, CHRISTOPHER	9/16/2011	2,545.89
HATFIELD, CHRISTOPHER	9/30/2011	4,697.09
HAYMARKET MEDIA, INC.	9/8/2011	\$ 8,321.50
HAYMARKET MEDIA, INC.	9/21/2011	\$ 8,653.00
HEALTH & HUMAN SERVICES COMMISSION	8/11/2011	\$ 11,110.91
HEALTH PARTNERS, INC.	9/1/2011	\$ 188.86
HEALTH RESEARCH, INC.	7/21/2011	\$ 3,954.80
HEANEY, JOSEPH	7/5/2011	\$ 37.00
HEANEY, JOSEPH	7/8/2011	\$ 1,856.33
HEANEY, JOSEPH	7/11/2011	\$ 31.00
HEANEY, JOSEPH	7/22/2011	\$ 1,856.33
HEANEY, JOSEPH	8/5/2011	\$ 1,856.33
HEANEY, JOSEPH	8/8/2011	\$ 126.24
HEANEY, JOSEPH	8/15/2011	\$ 3,585.95
HEANEY, JOSEPH	8/19/2011	\$ 1,856.33
HEANEY, JOSEPH	9/2/2011	\$ 1,863.99
HEANEY, JOSEPH	9/8/2011	\$ 6.25
HEANEY, JOSEPH	9/16/2011	\$ 1,891.77
HEANEY, JOSEPH	9/21/2011	\$ 24.70
HEANEY, JOSEPH	9/26/2011	\$ 12.50
HEANEY, JOSEPH	9/30/2011	\$ 2,214.76
HEDGETRACKERS	9/8/2011	\$ 75.00
HEINTZ, DEBORAH	7/8/2011	\$ 2,132.80
HEINTZ, DEBORAH	7/22/2011	\$ 2,132.80
HEINTZ, DEBORAH	8/5/2011	\$ 2,132.80
HEINTZ, DEBORAH	8/15/2011	\$ 15,213.66
HEINTZ, DEBORAH	8/19/2011	\$ 2,273.28
HEINTZ, DEBORAH	9/2/2011	\$ 2,273.28
HEINTZ, DEBORAH	9/16/2011	\$ 2,273.29
HEINTZ, DEBORAH	9/30/2011	
HEINTZ, MS DEBORAH	7/5/2011	
HEINTZ, MS DEBORAH	7/11/2011	\$ 11.00
HEINTZ, MS DEBORAH	8/8/2011	
HEINTZ, MS DEBORAH	8/15/2011	
HEINTZ, MS DEBORAH	9/8/2011	
HEINTZ, MS DEBORAH	9/26/2011	
HENTHORN, LINSEY	8/15/2011	\$ 599.05

Name of creditor	Payment date	Amount paid
HERLOVICH, CHERYLEN	7/8/2011	\$ 1,753.91
HERLOVICH, CHERYLEN	7/22/2011	·
HERLOVICH, CHERYLEN	8/5/2011	•
HERLOVICH, CHERYLEN	8/15/2011	
HERLOVICH, CHERYLEN	8/19/2011	•
HERLOVICH, CHERYLEN	9/2/2011	
HERLOVICH, CHERYLEN	9/16/2011	•
HERLOVICH, CHERYLEN	9/30/2011	
HERLOVICH, MS CHERYLEN	7/5/2011	
HERLOVICH, MS CHERYLEN	7/11/2011	*
HERLOVICH, MS CHERYLEN	8/8/2011	
HERLOVICH, MS CHERYLEN	9/8/2011	
HERLOVICH, MS CHERYLEN	9/21/2011	
HOGAN LOVELLS US LLP	7/14/2011	
HOGAN LOVELLS US LLP	8/5/2011	-
HOGAN LOVELLS US LLP	8/11/2011	"
HOGAN LOVELLS US LLP	8/25/2011	•
HOGAN LOVELLS US LLP	9/1/2011	·
HOGAN LOVELLS US LLP	9/15/2011	
HOGAN LOVELLS US LLP	9/26/2011	-
HOGAN LOVELLS US LLP	9/27/2011	
HOGUE, MARY	8/19/2011	
HOGUE, MARY	9/2/2011	
HOGUE, MARY	9/16/2011	•
HOGUE, MARY	9/30/2011	
HOLLAND, PHILIP	8/19/2011	·
HOLLAND, PHILIP	9/2/2011	
HOLLAND, PHILIP	9/16/2011	•
HOLLAND, PHILIP	9/30/2011	·
HOLLAND, PHILIP G.	9/21/2011	
HORSFALL, SALANAI	7/8/2011	•
HORSFALL, SALANAI	7/22/2011	
HORSFALL, SALANAI	8/5/2011	•
HORSFALL, SALANAI	8/15/2011	
HORSFALL, SALANAI	8/19/2011	•
HORSFALL, SALANAI	9/2/2011	•
HORSFALL, SALANAI	9/16/2011	
HORSFALL, SALANAI	9/30/2011	•
HOT SPOTS IN DERMATOLOGY	7/7/2011	
HUMANA, INC.		
HUMANA, INC.	7/15/2011	· .
HUNT, PATRICIA	8/18/2011	
HUNT, PATRICIA	7/8/2011	
HUNT, PATRICIA	7/22/2011 8/5/2011	
HUNT, PATRICIA		•
HUNT, PATRICIA	8/19/2011	•
	9/2/2011	" ,
HUNT, PATRICIA HUNT, PATRICIA	9/16/2011	
HUNTER SMITH & DAVIS LLP	9/30/2011	•
HUTCHINSON, JOHN ALAN	9/28/2011	•
HUTCHINSON, JOHN ALAN	7/8/2011	
-	7/22/2011	
HUTCHINSON, JOHN ALAN	8/5/2011	
HVAC, INC.	7/14/2011	-
HVAC. INC.	7/21/2011	\$ 1,544.67

Name of creditor	Payment date	Amount paid
HVAC. INC.	8/4/2011 \$	357.51
HVAC. INC.	8/11/2011 \$	1,597.67
HVAC. INC.	8/25/2011 \$	499.67
HVAC. INC.	9/1/2011	242.00
HVAC. INC.	9/8/2011	499.67
ILLUMITI CONSULTING LLC	7/14/2011	11,395.00
INFECTIOUS DISEASES SOCIETY FOR OB/GYN	8/11/2011	7,500.00
INFOTRIEVE, INC.	9/1/2011	47.81
INNOVATION PRINTING & COMMUNICATION	7/5/2011	28,125.40
INNOVATION PRINTING & COMMUNICATION	8/11/2011	22,455.89
INNOVATION PRINTING & COMMUNICATION	8/18/2011	75,231.52
INNOVATION PRINTING & COMMUNICATION	8/25/2011	3,362.86
INNOVATION PRINTING & COMMUNICATION	9/1/2011	8,053.12
INOVA PHARMACEUTICALS	9/21/2011	102,113.35
INTERMOUNTAIN DERM SOCIETY	7/7/2011	1,000.00
INTERNAL REVENUE SERVICE	9/22/2011	\$ 283,179.41
IRON MOUNTAIN	7/14/2011	\$ 293.20
IRON MOUNTAIN	8/4/2011	\$ 95.91
IRON MOUNTAIN	8/11/2011	198.34
IRON MOUNTAIN	9/1/2011	\$ 82.81
IRON MOUNTAIN	9/8/2011	\$ 198.34
IRON MOUNTAIN	9/15/2011	\$ 660.90
J RAYGOZA	7/5/2011	\$ 84.09
J RAYGOZA	7/25/2011	\$ 88.53
J RAYGOZA	8/25/2011	
JRAYGOZA	9/15/2011	\$ 10.00
J RAYGOZA	9/26/2011	
JACKSON LEWIS, LLP	8/11/2011	
JOHN, MALONE	7/8/2011	
JOHNSON CITY UTILITY SYSTEM	7/7/2011	
JOHNSON CITY UTILITY SYSTEM	7/21/2011	
JOHNSON CITY UTILITY SYSTEM	8/25/2011	\$ 72.12
JONES, SHANNON	7/8/2011	
JONES, SHANNON	7/22/2011	\$ 2,103.43
JONES, SHANNON	8/5/2011	\$ 2,103.42
JONES, SHANNON	8/15/2011	
JONES, SHANNON	8/19/2011	
JONES, SHANNON	9/2/2011	\$ 2,103.42
JONES, SHANNON	9/16/2011	\$ 2,098.70
JONES, SHANNON	9/30/2011	
JONES, SHANNON B.	7/11/2011	
JONES, SHANNON B.	7/25/2011	
JONES, SHANNON B.	8/8/2011	
JONES, SHANNON B.	9/21/2011	
JONES, TERRI	9/2/2011	
JONES, TERRI	9/6/2011	
JONES, TERRI	9/16/2011	
JONES, TERRI	9/30/2011	
JOURNAL OF DRUGS IN DERMATOLOGY	7/14/2011	
JOURNAL OF DRUGS IN DERMATOLOGY	9/1/2011	· · · · · · · · · · · · · · · · · · ·
JOURNAL OF DRUGS IN DERMATOLOGY	9/21/2011	
KASER, JAMES	9/2/2011	
KASER, JAMES	9/6/2011	
KASER, JAMES	9/16/2011	
	5/10/2011	. 01.11

Name of creditor	Payment date	Amount paid
KASER, JAMES	9/30/2011 \$	862.05
KELLEHER, KIMBERLE	7/8/2011 \$	
KELLEHER, KIMBERLE	7/22/2011 \$	•
KELLEHER, KIMBERLE	8/5/2011 \$	·
KELLEHER, KIMBERLE	8/15/2011 \$	
KELLEHER, KIMBERLE	8/19/2011 \$	-
KELLEHER, KIMBERLE	9/2/2011 \$,
KELLEHER, KIMBERLE	9/16/2011 \$	
KELLEHER, KIMBERLE	9/30/2011 \$	
KELLEHER, KIMBERLE A.	7/5/2011	·
KELLEHER, KIMBERLE A.	7/25/2011	
KELLEHER, KIMBERLE A.	8/8/2011 \$	
KELLEHER, KIMBERLE A.	8/31/2011	
KELLEHER, KIMBERLE A.	9/15/2011	
KELLER, NOEL	7/5/2011	· ·
KELLER, NOEL	7/8/2011	
KELLER, NOEL	7/22/2011	· ·
KELLER, NOEL	7/25/2011	•
KELLER, NOEL	8/5/2011	
KELLER, NOEL	8/15/2011	•
KELLER, NOEL	8/19/2011	•
KELLER, NOEL	8/25/2011	
KELLER, NOEL	9/2/2011	
KELLER, NOEL	9/16/2011	
KELLER, NOEL	9/26/2011	•
KELLER, NOEL	9/30/2011	
KENTUCKY BOARD OF PHARMACY	8/4/2011	•
KENTUCKY STATE TREASURER	9/15/2011	
KING BENEVOLENT FUND, INC.	8/25/2011	
KLINE, PAUL	7/8/2011	•
KLINE, PAUL	7/11/2011	·
KLINE, PAUL	7/22/2011	
KLINE, PAUL	8/5/2011	
KLINE, PAUL	8/15/2011	8,409.38
KLINE, PAUL	8/19/2011	
KLINE, PAUL	9/2/2011	
KLINE, PAUL	9/16/2011	\$ 4,021.77
KLINE, PAUL	9/30/2011	\$ 6,886.91
KONICA MINOLTA PREMIER FINANCE	7/14/2011	\$ 730.60
KONICA MINOLTA PREMIER FINANCE	7/21/2011	\$ 6,883.06
KONICA MINOLTA PREMIER FINANCE	8/22/2011	730.60
KONICA MINOLTA PREMIER FINANCE	8/25/2011	\$ 6,883.06
KONICA MINOLTA PREMIER FINANCE	9/8/2011	\$ 999.84
KOVARIK, CHRISTOPHER	7/5/2011	\$ 101.22
KOVARIK, CHRISTOPHER	7/11/2011	\$ 68.90
KOVARIK, CHRISTOPHER	7/25/2011	\$ 12.32
KOVARIK, CHRISTOPHER	8/8/2011	\$ 89.06
KOVARIK, CHRISTOPHER	8/15/2011	\$ 48.95
KOVARIK, CHRISTOPHER	8/31/2011	\$ 178.25
KOVARIK, CHRISTOPHER	9/8/2011	\$ 14.50
KOVARIK, CHRISTOPHER	9/15/2011	\$ 134.62
KOVARIK, CHRISTOPHER	9/26/2011	
KOVARIK, CHRISTOPHER B	7/8/2011	\$ 2,746.13
KOVARIK, CHRISTOPHER B	7/22/2011	\$ 2,746.14

Name of creditor	Payment date	Amount paid
KOVARIK, CHRISTOPHER B	8/5/2011	\$ 2,746.13
KOVARIK, CHRISTOPHER B	8/15/2011	_
KOVARIK, CHRISTOPHER B	8/19/2011	
KOVARIK, CHRISTOPHER B	9/2/2011	•
KOVARIK, CHRISTOPHER B	9/16/2011	
KOVARIK, CHRISTOPHER B	9/30/2011	
KRAYNYAK, HELEN	7/25/2011	· · · · · · · · · · · · · · · · · · ·
KRAYNYAK, HELEN	8/8/2011	
KRAYNYAK, HELEN	8/15/2011	
KRAYNYAK, HELEN	8/25/2011	
KRAYNYAK, HELEN	8/31/2011	
KRAYNYAK, HELEN	9/15/2011	
KRAYNYAK, HELEN	9/26/2011	
KRAYNYAK, HELEN H	7/8/2011	
KRAYNYAK, HELEN H	7/22/2011	·
KRAYNYAK, HELEN H	8/5/2011	-
KRAYNYAK, HELEN H	8/19/2011	=
KRAYNYAK, HELEN H	9/2/2011	•
KRAYNYAK, HELEN H	9/16/2011	· · · · · · · · · · · · · · · · · · ·
KRAYNYAK, HELEN H	9/30/2011	
KUPSKI, STEPHANIE L.	7/8/2011	•
KUPSKI, STEPHANIE L.	7/22/2011	" "
KUPSKI, STEPHANIE L.	8/5/2011	·
KUPSKI, STEPHANIE L.	8/19/2011	•
KUPSKI, STEPHANIE L.	9/2/2011	•
KUPSKI, STEPHANIE L.	9/16/2011	•
KUPSKI, STEPHANIE L.	9/30/2011	
KVL PRINTING	8/11/2011	
KVL PRINTING	9/21/2011	•
LACHMAN CONSULTANT SERVICES, INC	9/28/2011	•
LARSON, KELLY	7/8/2011	
LARSON, KELLY	7/3/2011	·
LARSON, KELLY	7/25/2011	·
LARSON, KELLY	8/5/2011	
LARSON, KELLY	8/15/2011	
LARSON, KELLY		•
LARSON, KELLY	8/19/2011	
LARSON, KELLY	8/31/2011	
LARSON, KELLY	9/2/2011	
LARSON, KELLY	9/15/2011	
LARSON, KELLY	9/16/2011	· ·
LATHAM & WATKINS LLP	9/30/2011	•
LATHAM & WATKINS LLP	7/6/2011	- ,
LATHAM & WATKINS LLP	8/10/2011	
LATHAM & WATKINS LLP	8/26/2011	
LATHAM & WATKINS LLP	9/16/2011	
	9/26/2011	
LAZARD FRERES & CO. LLC LAZARD FRERES & CO. LLC	7/15/2011	
LAZARD FRERES & CO. LLC LAZARD FRERES & CO. LLC	8/5/2011	•
LAZARD FRERES & CO. LLC LAZARD FRERES & CO. LLC	8/10/2011	
	9/16/2011	•
LEDER, RON	7/5/2011	
LEDER, RON	7/8/2011	
LEDER, RON	7/22/2011	•
LEDER, RON	8/5/2011	\$ 1,705.66

LEDER, RON	Name of creditor	Payment date	Amount paid
LEDER, RON 8/15/2011 \$ 1,708.65 LEDER, RON 9/2/2011 \$ 1,708.65 LEDER, RON 9/2/2011 \$ 1,708.65 LEDER, RON 9/8/2011 \$ 1,708.65 LEDER, RON 9/16/2011 \$ 1,708.65 LEDER, RON 9/16/2011 \$ 2,935.50 LEDER, RON 9/30/2011 \$ 2,935.50 LEVINSON, MICHAEL 7/5/2011 \$ 101.00 LEVINSON, MICHAEL 8/15/2011 \$ 28.00 LEVINSON, MICHAEL 9/15/2011 \$ 10.00 LEVINSON, MICHAEL 9/21/2011 \$ 10.00 LEVINSON, MICHAEL 9/21/2011 \$ 10.00 LEVINSON, MICHAEL O 7/8/2011 \$ 1,884.55 LEVINSON, MICHAEL O 8/15/2011 \$ 3,922.80 LEVINSON, MICHAEL O 8/15/2011 \$ 1,884.55 LEVINSON, MICHAEL O 8/15/2011 \$ 1,884.54 LEVINSON, MICHAEL O 9/2/2011 \$ 1,884.55 LEVINSON, MICHAEL O 9/2/2011 \$ 1,884.55 LEVINSON, MICHAEL O 9/16/2011 \$ 1,297.12 LEVINSON, MICHAEL O 9/16/2011 \$	LEDER, RON	8/8/2011	\$ 958.13
LEDER, RON \$/19/2011 \$ 1,708.66 LEDER, RON 9/8/2011 \$ 1,008.66 LEDER, RON 9/8/2011 \$ 1,708.66 LEDER, RON 9/16/2011 \$ 1,708.66 LEDER, RON 9/96/2011 \$ 1,2955.50 LEDER, RON 9/90/2011 \$ 1,2955.50 LEVINSON, MICHAEL 7/5/2011 \$ 101.00 LEVINSON, MICHAEL 8/15/2011 \$ 101.00 LEVINSON, MICHAEL 9/15/2011 \$ 144.40 LEVINSON, MICHAEL 9/21/2011 \$ 10.00 LEVINSON, MICHAEL 9/21/2011 \$ 1.00 LEVINSON, MICHAEL O. 7/8/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 8/15/2011 \$ 3,392.80 LEVINSON, MICHAEL O. 8/15/2011 \$ 3,392.80 LEVINSON, MICHAEL O. 9/16/2011 \$ 1,884.54 LEVINSON, MICHAEL O. <td></td> <td></td> <td></td>			
LEDER, RON 9/2/2011 \$ 3.00 LEDER, RON 9/8/2011 \$ 0.00 LEDER, RON 9/16/2011 \$ 1,705.65 LEDER, RON 9/16/2011 \$ 2,935.25 LEDER, RON 19/30/2011 \$ 2,935.25 LEVINSON, MICHAEL 7/25/2011 \$ 101.00 LEVINSON, MICHAEL 8/15/2011 \$ 28.00 LEVINSON, MICHAEL 9/15/2011 \$ 128.00 LEVINSON, MICHAEL 9/21/2011 \$ 14.40 LEVINSON, MICHAEL 9/21/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 7/22/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 8/15/2011 \$ 3,992.60 LEVINSON, MICHAEL O. 8/15/2011 \$ 3,992.60 LEVINSON, MICHAEL O. 8/15/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 8/15/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 9/2/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 9/2/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 9/16/2011 \$ 1,884.54 LEVINSON, MICHAEL O			the state of the s
LEDER, RON 9/8/2011 \$ 1,705.65 LEDER, RON 9/16/2011 \$ 1,705.65 LEDER, RON 9/16/2011 \$ 135.55 LEVINSON, MICHAEL 7/5/2011 \$ 101.00 LEVINSON, MICHAEL 8/15/2011 \$ 101.00 LEVINSON, MICHAEL 9/15/2011 \$ 104.40 LEVINSON, MICHAEL 9/15/2011 \$ 10.00 LEVINSON, MICHAEL 9/21/2011 \$ 10.00 LEVINSON, MICHAEL 9/21/2011 \$ 10.00 LEVINSON, MICHAEL O. 7/8/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 8/5/2011 \$ 3,894.54 LEVINSON, MICHAEL O. 8/5/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 8/19/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 8/19/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 9/2/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 9/16/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 9/16/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 9/16/2011 \$ 1,884.55 LEVI			
LEDER, RON 9/16/2011 \$ 2,035.50 LEDIER, RON 9/30/2011 \$ 2,035.50 LEVINSON, MICHAEL 7/52/2011 \$ 133.25 LEVINSON, MICHAEL 8/15/2011 \$ 28.00 LEVINSON, MICHAEL 8/15/2011 \$ 28.00 LEVINSON, MICHAEL 9/15/2011 \$ 10.00 LEVINSON, MICHAEL 9/21/2011 \$ 10.00 LEVINSON, MICHAEL O 7/8/2011 \$ 1,884.55 LEVINSON, MICHAEL O 8/15/2011 \$ 3,928.65 LEVINSON, MICHAEL O 9/2/2011 \$ 1,884.55 LEVINSON, MICHAEL O 9/2/2011 \$ 1,884.55 LEVINSON, MICHAEL O 9/2/2011 \$ 1,884.55 LEVINSON, MICHAEL O 9/30/2011 \$ 1,884.54			
LEDER, RON 9/30/2011 \$ 2935.50 LEVINSON, MICHAEL 7/5/2011 \$ 133.25 LEVINSON, MICHAEL 7/5/2011 \$ 101.00 LEVINSON, MICHAEL 8/15/2011 \$ 10.00 LEVINSON, MICHAEL 9/15/2011 \$ 10.00 LEVINSON, MICHAEL 9/21/2011 \$ 10.00 LEVINSON, MICHAEL O. 7/8/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 8/15/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 8/15/2011 \$ 3,392.80 LEVINSON, MICHAEL O. 8/15/2011 \$ 3,392.80 LEVINSON, MICHAEL O. 8/15/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 8/15/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 9/16/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 9/16/2011 \$ 1,884.55 </td <td></td> <td>•</td> <td></td>		•	
LEVINSON, MICHAEL	LEDER, RON		•
LEVINSON, MICHAEL 8/15/2011 \$ 28.00 LEVINSON, MICHAEL 9/15/2011 \$ 144.40 LEVINSON, MICHAEL 9/21/2011 \$ 18.84.51 LEVINSON, MICHAEL O. 7/8/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 7/22/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 8/15/2011 \$ 3,202.80 LEVINSON, MICHAEL O. 8/15/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 9/16/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 9/30/2011 \$ 1,297.12 LEVINSON, MICHAEL O. 9/30/2011 \$ 1,654.13 LEVINSON, MICHAEL O. 9/30/2011 \$ 1,654.13 LEVINSON, MICHAEL O. 9/30/2011 \$ 1,654.13 LIAPIS, MARIA 7/11/2011 \$ 1,654.13 LIAPIS, MARIA 7/22/2011 \$ 1,654.19 LIAPIS, MARIA 8/8/2011 \$ 38.00 LIAPIS, MARIA 8/15/2011 \$ 1,654.19 L	LEVINSON, MICHAEL		
LEVINSON, MICHAEL 9/15/2011 \$ 10.00 LEVINSON, MICHAEL O. 7/8/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 7/22/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 8/15/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 8/15/2011 \$ 3,392.80 LEVINSON, MICHAEL O. 8/19/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 8/19/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 9/2/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 9/30/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 9/30/2011 \$ 1,654.18 LEVINSON, MICHAEL O. 9/30/2011 \$ 1,654.18 LIAPIS, MARIA 7/11/2011 \$ 1,654.18 LIAPIS, MARIA 7/11/2011 \$ 1,654.19 LIAPIS, MARIA 7/22/2011 \$ 1,654.19 LIAPIS, MARIA 8/5/2011 \$ 1,654.19 LIAPIS, MARIA 8/5/2011 \$ 1,654.19 LIAPIS, MARIA 8/15/2011 \$ 1,654.19 LIAPIS, MARIA 9/30/2011 \$ 1,654.19 LIAPIS, MARIA 9/30/2011 \$ 1,654.19 LIAPIS, MARIA	LEVINSON, MICHAEL	7/25/2011	\$ 101.00
LEVINSON, MICHAEL 9/21/2011 \$ 1,000 LEVINSON, MICHAEL O. 7/8/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 8/5/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 8/15/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 8/19/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 9/2/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 9/2/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 9/30/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 9/30/2011 \$ 1,654.18 LIAPIS, MARIA 7/8/2011 \$ 1,654.18 LIAPIS, MARIA 7/11/2011 \$ 1,654.18 LIAPIS, MARIA 7/22/2011 \$ 1,654.19 LIAPIS, MARIA 8/5/2011 \$ 1,654.19 LIAPIS, MARIA 8/5/2011 \$ 1,654.19 LIAPIS, MARIA 8/5/2011 \$ 38.00 LIAPIS, MARIA 8/5/2011 \$ 36.00 LIAPIS, MARIA 8/5/2011 \$ 36.00 LIAPIS, MARIA 8/15/2011 \$ 37.09 LIAPIS, MARIA 8/15/2011 \$ 36.00 LIAPIS, MARIA 8/15/2011 \$ 36.00 LIAPIS, MARIA 9/2/2011 \$ 1,654.19 LIAPIS, MARIA 9/2/2011 \$ 1,654.19 LIAPIS, MARIA 9/3/2011 \$ 2.00 <td>LEVINSON, MICHAEL</td> <td>8/15/2011</td> <td>\$ 28.00</td>	LEVINSON, MICHAEL	8/15/2011	\$ 28.00
LEVINSON, MICHAEL 9/21/2011 \$ 1,000 LEVINSON, MICHAEL O. 7/8/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 8/5/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 8/15/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 8/19/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 9/2/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 9/2/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 9/30/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 9/30/2011 \$ 1,654.18 LIAPIS, MARIA 7/8/2011 \$ 1,654.18 LIAPIS, MARIA 7/11/2011 \$ 1,654.18 LIAPIS, MARIA 7/22/2011 \$ 1,654.19 LIAPIS, MARIA 8/5/2011 \$ 1,654.19 LIAPIS, MARIA 8/5/2011 \$ 1,654.19 LIAPIS, MARIA 8/5/2011 \$ 38.00 LIAPIS, MARIA 8/5/2011 \$ 36.00 LIAPIS, MARIA 8/5/2011 \$ 36.00 LIAPIS, MARIA 8/15/2011 \$ 37.09 LIAPIS, MARIA 8/15/2011 \$ 36.00 LIAPIS, MARIA 8/15/2011 \$ 36.00 LIAPIS, MARIA 9/2/2011 \$ 1,654.19 LIAPIS, MARIA 9/2/2011 \$ 1,654.19 LIAPIS, MARIA 9/3/2011 \$ 2.00 <td>LEVINSON, MICHAEL</td> <td>9/15/2011</td> <td>\$ 144.40</td>	LEVINSON, MICHAEL	9/15/2011	\$ 144.40
LEVINSON, MICHAEL O. 7/22/2011 \$ 1,884.54 LEVINSON, MICHAEL O. 8/15/2011 \$ 3,392.80 LEVINSON, MICHAEL O. 8/15/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 9/2/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 9/16/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 9/30/2011 \$ 1,297.12 LEVINSON, MICHAEL O. 9/30/2011 \$ 1,297.12 LAPIS, MARIA 7/8/2011 \$ 1,654.18 LIAPIS, MARIA 7/22/2011 \$ 1,654.18 LIAPIS, MARIA 7/22/2011 \$ 1,654.19 LIAPIS, MARIA 8/5/2011 \$ 1654.19 LIAPIS, MARIA 8/15/2011 \$ 1654.19 LIAPIS, MARIA 8/15/2011 \$ 1,654.19 LIAPIS, MARIA 9/3/2011 \$ 1,654.19 <td></td> <td>9/21/2011</td> <td>\$ 10.00</td>		9/21/2011	\$ 10.00
LEVINSON, MICHAEL O.	LEVINSON, MICHAEL O.		
LEVINSON, MICHAEL O. 8/5/2011 \$ 3,392.80 LEVINSON, MICHAEL O. 8/15/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 9/16/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 9/16/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 9/36/2011 \$ 1,297.12 LEVINSON, MICHAEL O. 9/36/2011 \$ 1,297.12 LIAPIS, MARIA 7/8/2011 \$ 1,654.18 LIAPIS, MARIA 7/12/2011 \$ 1,654.19 LIAPIS, MARIA 7/25/2011 \$ 17.00 LIAPIS, MARIA 8/5/2011 \$ 154.19 LIAPIS, MARIA 8/5/2011 \$ 158.00 LIAPIS, MARIA 8/5/2011 \$ 158.00 LIAPIS, MARIA 8/15/2011 \$ 157.09 LIAPIS, MARIA 8/15/2011 \$ 1,654.19 LIAPIS, MARIA 9/2/2011 \$ 1,654.19	LEVINSON, MICHAEL O.	7/22/2011	
LEVINSON, MICHAEL O. 8/15/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 9/2/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 9/16/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 9/30/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 9/30/2011 \$ 1,654.18 LIAPIS, MARIA 7/81/2011 \$ 35.00 LIAPIS, MARIA 7/11/2011 \$ 35.00 LIAPIS, MARIA 7/22/2011 \$ 1,654.19 LIAPIS, MARIA 8/5/2011 \$ 1,654.19 LIAPIS, MARIA 8/5/2011 \$ 1,654.19 LIAPIS, MARIA 8/5/2011 \$ 1,654.19 LIAPIS, MARIA 8/8/2011 \$ 1,654.19 LIAPIS, MARIA 8/8/2011 \$ 1,654.19 LIAPIS, MARIA 8/15/2011 \$ 5,770.92 LIAPIS, MARIA 8/15/2011 \$ 1,654.19 LIAPIS, MARIA 8/15/2011 \$ 1,654.19 LIAPIS, MARIA 9/2/2011 \$ 1,654.19 LIAPIS, MARIA 9/2/2011 \$ 1,654.19 LIAPIS, MARIA 9/2/2011 \$ 1,654.19 LIAPIS, MARIA 9/3/2011 \$ 1,654.19	LEVINSON, MICHAEL O.		· ·
LEVINSON, MICHAEL O. 8/19/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 9/2/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 9/30/2011 \$ 1,297.12 LEVINSON, MICHAEL O. 9/30/2011 \$ 1,297.12 LIAPIS, MARIA 7/8/2011 \$ 1,654.18 LIAPIS, MARIA 7/12/2011 \$ 1,654.19 LIAPIS, MARIA 7/22/2011 \$ 1,654.19 LIAPIS, MARIA 8/5/2011 \$ 1,654.19 LIAPIS, MARIA 8/5/2011 \$ 1,654.19 LIAPIS, MARIA 8/15/2011 \$ 1,654.19 LIAPIS, MARIA 8/15/2011 \$ 1,654.19 LIAPIS, MARIA 8/15/2011 \$ 5,770.92 LIAPIS, MARIA 8/15/2011 \$ 1,654.19 LIAPIS, MARIA 9/8/2011 \$ 1,654.19 LIAPIS, MARIA 9/16/2011 \$ 1,654.19	LEVINSON, MICHAEL O.		
LEVINSON, MICHAEL O. 9/2/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 9/16/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 9/30/2011 \$ 1,297.12 LIAPIS, MARIA 7/8/2011 \$ 1,654.18 LIAPIS, MARIA 7/11/2011 \$ 1,654.19 LIAPIS, MARIA 7/25/2011 \$ 1,654.19 LIAPIS, MARIA 7/25/2011 \$ 1,700 LIAPIS, MARIA 8/5/2011 \$ 38.00 LIAPIS, MARIA 8/5/2011 \$ 38.00 LIAPIS, MARIA 8/5/2011 \$ 38.00 LIAPIS, MARIA 8/15/2011 \$ 5770.92 LIAPIS, MARIA 8/15/2011 \$ 5770.92 LIAPIS, MARIA 8/15/2011 \$ 1,654.19 LIAPIS, MARIA 8/19/2011 \$ 1,654.19 LIAPIS, MARIA 8/19/2011 \$ 1,654.19 LIAPIS, MARIA 9/8/2011 \$ 1,654.19 LIAPI			
LEVINSON, MICHAEL O. 9/16/2011 \$ 1,884.55 LEVINSON, MICHAEL O. 9/30/2011 \$ 1,297.12 LIAPIS, MARIA 7/8/2011 \$ 1,654.18 LIAPIS, MARIA 7/22/2011 \$ 1,654.19 LIAPIS, MARIA 7/22/2011 \$ 1,654.19 LIAPIS, MARIA 7/22/2011 \$ 1,654.19 LIAPIS, MARIA 8/5/2011 \$ 38.00 LIAPIS, MARIA 8/15/2011 \$ 87.00 LIAPIS, MARIA 8/15/2011 \$ 87.00 LIAPIS, MARIA 8/15/2011 \$ 5,770.92 LIAPIS, MARIA 8/15/2011 \$ 5,770.92 LIAPIS, MARIA 8/15/2011 \$ 30.00 LIAPIS, MARIA 8/15/2011 \$ 30.00 LIAPIS, MARIA 8/15/2011 \$ 30.00 LIAPIS, MARIA 9/8/2011 \$ 30.00 LIAPIS, MARIA 9/8/2011 \$ 30.00 LIAPIS, MARIA 9/8/2011 \$ 1,654.19 LIAPIS, MARIA 9/12/2011 \$ 25.00 LIAPIS, MARIA 9/12/2011 \$ 25.00 LIAPIS, MARIA 9/12/2011 \$ 25.00 LIAPIS, MARIA	LEVINSON, MICHAEL O.		
LEVINSON, MICHAEL O. 9/30/2011 \$ 1,297.12 LIAPIS, MARIA 7/8/2011 \$ 1,654.18 LIAPIS, MARIA 7/11/2011 \$ 33.00 LIAPIS, MARIA 7/22/2011 \$ 1,654.19 LIAPIS, MARIA 7/25/2011 \$ 17.00 LIAPIS, MARIA 8/5/2011 \$ 38.00 LIAPIS, MARIA 8/15/2011 \$ 87.00 LIAPIS, MARIA 8/15/2011 \$ 77.09 LIAPIS, MARIA 8/15/2011 \$ 77.00 LIAPIS, MARIA 8/15/2011 \$ 77.00 LIAPIS, MARIA 8/15/2011 \$ 1,654.19 LIAPIS, MARIA 8/12/2011 \$ 1,654.19 LIAPIS, MARIA 9/2/2011 \$ 1,654.19 LIAPIS, MARIA 9/16/2011		· · · · · · · · · · · · · · · · · · ·	
LIAPIS, MARIA 7/8/2011 \$ 1,654.18 LIAPIS, MARIA 7/11/2011 \$ 33.00 LIAPIS, MARIA 7/22/2011 \$ 1,654.19 LIAPIS, MARIA 8/5/2011 \$ 1,654.19 LIAPIS, MARIA 8/5/2011 \$ 1,654.19 LIAPIS, MARIA 8/15/2011 \$ 38.00 LIAPIS, MARIA 8/15/2011 \$ 70.02 LIAPIS, MARIA 8/15/2011 \$ 5,770.92 LIAPIS, MARIA 8/15/2011 \$ 1,654.19 LIAPIS, MARIA 8/15/2011 \$ 1,654.19 LIAPIS, MARIA 8/15/2011 \$ 30.00 LIAPIS, MARIA 9/2/2011 \$ 1,654.19 LIAPIS, MARIA 9/8/2011 \$ 1,654.19 LIAPIS, MARIA 9/8/2011 \$ 12.00 LIAPIS, MARIA 9/16/2011 \$ 25.00 LIAPIS, MARIA 9/30/2011 \$ 25.00 LOGERIS, MARIA 9/30/2011 \$ 25.			•
LIAPIS, MARIA			
LLAPIS, MARIA	,		
LIAPIS, MARIA 7/25/2011 \$ 17.00 LIAPIS, MARIA 8/5/2011 \$ 1,654.19 LIAPIS, MARIA 8/8/2011 \$ 38.00 LIAPIS, MARIA 8/15/2011 \$ 87.00 LIAPIS, MARIA 8/15/2011 \$ 5,770.92 LIAPIS, MARIA 8/19/2011 \$ 1,654.19 LIAPIS, MARIA 8/25/2011 \$ 30.00 LIAPIS, MARIA 9/2/2011 \$ 1,654.19 LIAPIS, MARIA 9/8/2011 \$ 21.00 LIAPIS, MARIA 9/8/2011 \$ 21.00 LIAPIS, MARIA 9/16/2011 \$ 1,654.18 LIAPIS, MARIA 9/12/2011 \$ 25.00 LIAPIS, MARIA 9/12/2011 \$ 25.00 LIAPIS, MARIA 9/30/2011 \$ 243.55 LIQUENT, INC 9/8/2011 \$ 1,000.00 LOGCHER, CHRISTINA 7/5/2011 \$ 60.00 LOSCHER, CHRISTINA 8/8/2011 \$ 60.00 LOSCHER, CHRISTINA 8/8/2011 \$ 60.20 LOSCHER, CHRISTINA 9/15/2011 \$ 49.103.6 LOSCHER, CHRISTINA 9/15/2011 \$ 49.103.6 LOVAS, MR RANDALL 7/5/2011 <t< td=""><td></td><td></td><td></td></t<>			
LIAPIS, MARIA 8/5/2011 \$ 38.00 LIAPIS, MARIA 8/8/2011 \$ 38.00 LIAPIS, MARIA 8/15/2011 \$ 70.00 LIAPIS, MARIA 8/15/2011 \$ 5,770.92 LIAPIS, MARIA 8/19/2011 \$ 1,654.19 LIAPIS, MARIA 8/25/2011 \$ 30.00 LIAPIS, MARIA 9/2/2011 \$ 1,654.19 LIAPIS, MARIA 9/8/2011 \$ 21.00 LIAPIS, MARIA 9/16/2011 \$ 1,654.18 LIAPIS, MARIA 9/16/2011 \$ 25.00 LIAPIS, MARIA 9/21/2011 \$ 25.00 LIAPIS, MARIA 9/30/2011 \$ 25.00 LIAPIS, MARIA 9/30/2011 \$ 20.00 LIAPIS, MARIA 9/16/2011 \$ 20.00 LIAPIS, MARIA 9/17/2011 \$ 20.00 LIAPIS, MARIA 9/16/2011 \$ 20.00 LIAPIS, MARIA 9/16/2011 \$ 20.00 LOSCHER, CHRISTINA 7/5/2011 \$ 60.00 LOSCHER, CHRISTINA 8/8/2011 \$ 32.40 LOUSCHER, CHRISTINA 9/15/2011 \$ 49.			·
LIAPIS, MARIA 8/8/2011 \$ 87.00 LIAPIS, MARIA 8/15/2011 \$ 577.092 LIAPIS, MARIA 8/15/2011 \$ 5,770.92 LIAPIS, MARIA 8/15/2011 \$ 1,654.19 LIAPIS, MARIA 8/25/2011 \$ 30.00 LIAPIS, MARIA 9/2/2011 \$ 1,654.19 LIAPIS, MARIA 9/8/2011 \$ 21.00 LIAPIS, MARIA 9/16/2011 \$ 25.00 LIAPIS, MARIA 9/16/2011 \$ 25.00 LIAPIS, MARIA 9/30/2011 \$ 25.00 LIAPIS, MARIA 9/30/2011 \$ 25.00 LIAPIS, MARIA 9/30/2011 \$ 1,000.00 LOGMEIN, INC 9/8/2011 \$ 1,000.00 LOSCHER, CHRISTINA 7/14/2011 \$ 60.00 LOSCHER, CHRISTINA 7/25/2011 \$ 57.00 LOSCHER, CHRISTINA 8/8/2011 \$ 32.40 LOSCHER, CHRISTINA 8/15/2011 \$ 34.33 LOSCHER, CHRISTINA 9/15/2011 \$ 34.33 LOSCHER, CHRISTINA 9/26/2011 \$ 49.00.36 LOVAS, MR RANDALL 7/5/2011 \$ 52.25 LOVA			
LIAPIS, MARIA 8/15/2011 \$ 87.00 LIAPIS, MARIA 8/15/2011 \$ 5,770.92 LIAPIS, MARIA 8/19/2011 \$ 1,654.19 LIAPIS, MARIA 8/25/2011 \$ 30.00 LIAPIS, MARIA 9/2/2011 \$ 1,654.19 LIAPIS, MARIA 9/8/2011 \$ 21.00 LIAPIS, MARIA 9/16/2011 \$ 25.00 LIAPIS, MARIA 9/30/2011 \$ 25.00 LOUSCHER, CHRISTINA 9/8/2011 \$ 2,250.00 LOSCHER, CHRISTINA 8/8/2011 \$ 32.40 LOSCHER, CHRISTINA 8/8/2011 \$ 34.33 LOSCHER, CHRISTINA 9/15/2011 \$ 34.33			
LIAPIS, MARIA 8/15/2011 \$ 5,770.92 LIAPIS, MARIA 8/19/2011 \$ 1,654.19 LIAPIS, MARIA 9/2/2011 \$ 30.00 LIAPIS, MARIA 9/2/2011 \$ 1,654.19 LIAPIS, MARIA 9/8/2011 \$ 21.00 LIAPIS, MARIA 9/16/2011 \$ 21.00 LIAPIS, MARIA 9/16/2011 \$ 25.00 LIAPIS, MARIA 9/21/2011 \$ 25.00 LIQUENT, INC 9/8/2011 \$ 1,000.00 LOGHER, CHRISTINA 7/5/2011 \$ 60.00 LOSCHER, CHRISTINA 7/5/2011 \$ 57.00 LOSCHER, CHRISTINA 8/8/2011 \$ 32.40 LOSCHER, CHRISTINA 8/8/2011 \$ 60.20 LOSCHER, CHRISTINA 8/15/2011 \$ 60.20 LOSCHER, CHRISTINA 9/15/2011 \$ 49.103.6 LOSCHER, CHRISTINA 9/15/2011 \$ 49.103.6 LOSCHER, CHRISTINA 9/15/2011 \$ 49.100.36 LOSCHER, CHRISTINA 9/15/2011 \$ 49.100.36 LOUISIANA DEPT. OF HEALTH& HOSPITALS 7/5/2011 \$ 49.100.36 LOVAS, MR RANDALL 7/5/2011 \$ 48.654.10 L			
LIAPIS, MARIA 8/19/2011 \$ 1,654.19 LIAPIS, MARIA 8/25/2011 \$ 30.00 LIAPIS, MARIA 9/2/2011 \$ 1,654.19 LIAPIS, MARIA 9/8/2011 \$ 21.00 LIAPIS, MARIA 9/16/2011 \$ 25.00 LIAPIS, MARIA 9/21/2011 \$ 25.00 LIAPIS, MARIA 9/30/2011 \$ 4,243.55 LIQUENT, INC 9/8/2011 \$ 1,000.00 LOSCHER, CHRISTINA 7/5/2011 \$ 60.00 LOSCHER, CHRISTINA 7/25/2011 \$ 57.00 LOSCHER, CHRISTINA 8/8/2011 \$ 32.40 LOSCHER, CHRISTINA 8/8/2011 \$ 34.33 LOSCHER, CHRISTINA 8/15/2011 \$ 60.20 LOSCHER, CHRISTINA 9/15/2011 \$ 49.100.36 LOUSIANA DEPT. OF HEALTH& HOSPITALS 7/5/2011 \$ 49.100.36 LOUISIANA DEPT. OF HEALTH& HOSPITALS 7/5/2011 \$ 5.225 LOVAS, MR RANDALL 7/25/2011 \$ 1,058.55 LOVAS, MR RANDALL 8/8/2011 \$ 1,058.55 LOVAS, MR RANDALL 9/26/2011 \$ 45.25 LOVAS, RANDALL 7/8/2011 \$ 2,899.38			
LIAPIS, MARIA 8/25/2011 \$ 30.00 LIAPIS, MARIA 9/2/2011 \$ 1,654.19 LIAPIS, MARIA 9/8/2011 \$ 21.00 LIAPIS, MARIA 9/16/2011 \$ 25.00 LIAPIS, MARIA 9/21/2011 \$ 25.00 LIAPIS, MARIA 9/30/2011 \$ 4,243.55 LIQUENT, INC 9/8/2011 \$ 1,000.00 LOGMEIN, INC. 7/14/2011 \$ 2,250.00 LOSCHER, CHRISTINA 7/5/2011 \$ 60.00 LOSCHER, CHRISTINA 7/25/2011 \$ 60.00 LOSCHER, CHRISTINA 8/8/2011 \$ 32.40 LOSCHER, CHRISTINA 8/15/2011 \$ 60.20 LOSCHER, CHRISTINA 9/15/2011 \$ 34.33 LOSCHER, CHRISTINA 9/15/2011 \$ 49.100.36 LOUISIANA DEPT. OF HEALTH& HOSPITALS 7/5/2011 \$ 49.100.36 LOUSIANA DEPT. OF HEALTH& HOSPITALS 9/15/2011 \$ 49.50.20 LOVAS, MR RANDALL 7/5/2011 \$ 52.25 LOVAS, MR RANDALL 9			·
LIAPIS, MARIA 9/2/2011 \$ 1,654.19 LIAPIS, MARIA 9/8/2011 \$ 21.00 LIAPIS, MARIA 9/16/2011 \$ 1,654.18 LIAPIS, MARIA 9/21/2011 \$ 25.00 LIAPIS, MARIA 9/30/2011 \$ 4,243.55 LIQUENT, INC 9/8/2011 \$ 1,000.00 LOGMEIN, INC. 7/14/2011 \$ 2,250.00 LOSCHER, CHRISTINA 7/5/2011 \$ 60.00 LOSCHER, CHRISTINA 7/5/2011 \$ 57.00 LOSCHER, CHRISTINA 8/8/2011 \$ 32.40 LOSCHER, CHRISTINA 8/15/2011 \$ 60.20 LOSCHER, CHRISTINA 9/15/2011 \$ 34.33 LOSCHER, CHRISTINA 9/15/2011 \$ 49,100.36 LOUISIANA DEPT. OF HEALTH& HOSPITALS 7/5/2011 \$ 49,100.36 LOUISIANA DEPT. OF HEALTH& HOSPITALS 7/5/2011 \$ 1,058.55 LOVAS, MR RANDALL 7/25/2011 \$ 1,058.55 LOVAS, MR RANDALL 9/26/2011 \$ 45.25 LOVAS, MR RANDALL 9/26/2011 \$ 45.25 LOVAS, RANDALL 9/26/2011 \$ 49.24 LOVAS, RANDALL 7/8/2011 \$ 2,899.37	LIAPIS, MARIA	8/25/2011	
LIAPIS, MARIA 9/8/2011 \$ 21.00 LIAPIS, MARIA 9/16/2011 \$ 1,654.18 LIAPIS, MARIA 9/21/2011 \$ 25.00 LIAPIS, MARIA 9/30/2011 \$ 4,243.50 LIQUENT, INC 9/8/2011 \$ 1,000.00 LOGMEIN, INC. 7/14/2011 \$ 2,250.00 LOSCHER, CHRISTINA 7/5/2011 \$ 57.00 LOSCHER, CHRISTINA 8/8/2011 \$ 32.40 LOSCHER, CHRISTINA 8/15/2011 \$ 60.20 LOSCHER, CHRISTINA 9/15/2011 \$ 60.20 LOSCHER, CHRISTINA 9/15/2011 \$ 140.20 LOSCHER, CHRISTINA 9/15/2011 \$ 140.20 LOUSIANA DEPT. OF HEALTH& HOSPITALS 7/5/2011 \$ 49,100.36 LOUISIANA DEPT. OF HEALTH& HOSPITALS 7/5/2011 \$ 1,058.55 LOVAS, MR RANDALL 7/25/2011 \$ 1,058.55 LOVAS, MR RANDALL 7/25/2011 \$ 1,058.55 LOVAS, RANDALL 7/8/2011 \$ 2,899.37 LOVAS, RANDALL 7/22/2011 \$ 2,899.37 LOVAS, RANDALL 7/22/2011 \$ 2,899.38 LOVAS, RANDALL 8/5/2011 \$ 5,492.59 <td>LIAPIS, MARIA</td> <td></td> <td></td>	LIAPIS, MARIA		
LIAPIS, MARIA 9/16/2011 \$ 1,654.18 LIAPIS, MARIA 9/21/2011 \$ 25.00 LIAPIS, MARIA 9/30/2011 \$ 4,243.55 LIQUENT, INC 9/8/2011 \$ 1,000.00 LOGMEIN, INC. 7/14/2011 \$ 2,250.00 LOSCHER, CHRISTINA 7/5/2011 \$ 60.00 LOSCHER, CHRISTINA 8/8/2011 \$ 32.40 LOSCHER, CHRISTINA 8/15/2011 \$ 60.20 LOSCHER, CHRISTINA 8/15/2011 \$ 60.20 LOSCHER, CHRISTINA 9/15/2011 \$ 49,100.36 LOUISIANA DEPT. OF HEALTH& HOSPITALS 7/5/2011 \$ 49,100.36 LOUISIANA DEPT. OF HEALTH& HOSPITALS 7/5/2011 \$ 52.25 LOVAS, MR RANDALL 7/5/2011 \$ 1,058.55 LOVAS, MR RANDALL 8/8/2011 \$ 1,058.55 LOVAS, MR RANDALL 9/26/2011 \$ 45.25 LOVAS, RANDALL 7/8/2011 \$ 2,899.37 LOVAS, RANDALL 7/22/2011 \$ 2,899.37 LOVAS, RANDALL 8/5/2011 \$ 2,899.35 LOVAS, RANDALL 8/5/2011 \$ 2,899.35 LOVAS, RANDALL 8/5/2011 \$ 5,492.59 <td>LIAPIS, MARIA</td> <td></td> <td></td>	LIAPIS, MARIA		
LIAPIS, MARIA 9/30/2011 \$ 4,243.55 LIQUENT, INC 9/8/2011 \$ 1,000.00 LOGMEIN, INC. 7/14/2011 \$ 2,250.00 LOSCHER, CHRISTINA 7/5/2011 \$ 60.00 LOSCHER, CHRISTINA 7/25/2011 \$ 57.00 LOSCHER, CHRISTINA 8/8/2011 \$ 32.40 LOSCHER, CHRISTINA 8/15/2011 \$ 60.20 LOSCHER, CHRISTINA 9/15/2011 \$ 34.33 LOSCHER, CHRISTINA 9/26/2011 \$ 140.20 LOUISIANA DEPT. OF HEALTH& HOSPITALS 7/5/2011 \$ 49,100.36 LOUISIANA DEPT. OF HEALTH& HOSPITALS 9/15/2011 \$ 652.25 LOVAS, MR RANDALL 7/5/2011 \$ 52.25 LOVAS, MR RANDALL 7/5/2011 \$ 1,058.55 LOVAS, MR RANDALL 8/8/2011 \$ 217.41 LOVAS, RANDALL 9/26/2011 45.25 LOVAS, RANDALL 7/8/2011 \$ 2,899.37 LOVAS, RANDALL 7/22/2011 \$ 2,899.38 LOVAS, RANDALL 8/5/2011 \$ 2,993.65 LOVAS, RANDALL 8/5/2011 \$ 5,492.59	LIAPIS, MARIA	9/16/2011	\$ 1,654.18
LIQUENT, INC 9/8/2011 \$ 1,000.00 LOGMEIN, INC. 7/14/2011 \$ 2,250.00 LOSCHER, CHRISTINA 7/5/2011 \$ 60.00 LOSCHER, CHRISTINA 7/25/2011 \$ 57.00 LOSCHER, CHRISTINA 8/8/2011 \$ 32.40 LOSCHER, CHRISTINA 8/15/2011 \$ 60.20 LOSCHER, CHRISTINA 9/15/2011 \$ 34.33 LOSCHER, CHRISTINA 9/26/2011 \$ 140.20 LOUISIANA DEPT. OF HEALTH& HOSPITALS 7/5/2011 \$ 49,100.36 LOUISIANA DEPT. OF HEALTH& HOSPITALS 9/15/2011 \$ 48,654.10 LOVAS, MR RANDALL 7/5/2011 \$ 52.25 LOVAS, MR RANDALL 7/25/2011 \$ 1,058.55 LOVAS, MR RANDALL 8/8/2011 \$ 217.41 LOVAS, RANDALL 9/26/2011 \$ 45.25 LOVAS, RANDALL 7/8/2011 \$ 2,899.37 LOVAS, RANDALL 7/22/2011 \$ 2,899.38 LOVAS, RANDALL 8/5/2011 \$ 2,932.65 LOVAS, RANDALL 8/5/2011 \$ 5,492.59	LIAPIS, MARIA	9/21/2011	\$ 25.00
LOGMEIN, INC. 7/14/2011 \$ 2,250.00 LOSCHER, CHRISTINA 7/5/2011 \$ 60.00 LOSCHER, CHRISTINA 7/25/2011 \$ 57.00 LOSCHER, CHRISTINA 8/8/2011 \$ 32.40 LOSCHER, CHRISTINA 8/15/2011 \$ 60.20 LOSCHER, CHRISTINA 9/15/2011 \$ 34.33 LOSCHER, CHRISTINA 9/26/2011 \$ 140.20 LOUISIANA DEPT. OF HEALTH& HOSPITALS 7/5/2011 \$ 49,100.36 LOUISIANA DEPT. OF HEALTH& HOSPITALS 9/15/2011 \$ 52.25 LOVAS, MR RANDALL 7/5/2011 \$ 1,058.55 LOVAS, MR RANDALL 7/25/2011 \$ 1,058.55 LOVAS, MR RANDALL 8/8/2011 \$ 217.41 LOVAS, RANDALL 7/8/2011 \$ 2,899.37 LOVAS, RANDALL 7/22/2011 \$ 2,899.37 LOVAS, RANDALL 8/5/2011 \$ 2,932.65 LOVAS, RANDALL 8/5/2011 \$ 5,492.59	LIAPIS, MARIA	9/30/2011	\$ 4,243.55
LOGMEIN, INC. 7/14/2011 \$ 2,250.00 LOSCHER, CHRISTINA 7/5/2011 \$ 60.00 LOSCHER, CHRISTINA 7/25/2011 \$ 57.00 LOSCHER, CHRISTINA 8/8/2011 \$ 32.40 LOSCHER, CHRISTINA 8/15/2011 \$ 60.20 LOSCHER, CHRISTINA 9/15/2011 \$ 34.33 LOSCHER, CHRISTINA 9/26/2011 \$ 140.20 LOUISIANA DEPT. OF HEALTH& HOSPITALS 7/5/2011 \$ 49,100.36 LOUISIANA DEPT. OF HEALTH& HOSPITALS 9/15/2011 \$ 52.25 LOVAS, MR RANDALL 7/5/2011 \$ 1,058.55 LOVAS, MR RANDALL 7/25/2011 \$ 1,058.55 LOVAS, MR RANDALL 8/8/2011 \$ 217.41 LOVAS, RANDALL 7/8/2011 \$ 2,899.37 LOVAS, RANDALL 7/22/2011 \$ 2,899.37 LOVAS, RANDALL 8/5/2011 \$ 2,932.65 LOVAS, RANDALL 8/5/2011 \$ 5,492.59		9/8/2011	\$ 1,000.00
LOSCHER, CHRISTINA 7/5/2011 \$ 60.00 LOSCHER, CHRISTINA 7/25/2011 \$ 57.00 LOSCHER, CHRISTINA 8/8/2011 \$ 32.40 LOSCHER, CHRISTINA 8/15/2011 \$ 60.20 LOSCHER, CHRISTINA 9/15/2011 \$ 34.33 LOSCHER, CHRISTINA 9/26/2011 \$ 140.20 LOUISLANA DEPT. OF HEALTH& HOSPITALS 7/5/2011 \$ 49,100.36 LOUISLANA DEPT. OF HEALTH& HOSPITALS 9/15/2011 \$ 52.25 LOVAS, MR RANDALL 7/5/2011 \$ 52.25 LOVAS, MR RANDALL 8/8/2011 \$ 217.41 LOVAS, MR RANDALL 9/26/2011 \$ 45.25 LOVAS, RANDALL 7/8/2011 \$ 2,899.37 LOVAS, RANDALL 7/22/2011 \$ 2,899.38 LOVAS, RANDALL 8/5/2011 \$ 2,932.65 LOVAS, RANDALL 8/5/2011 \$ 5,492.59	LOGMEIN, INC.	7/14/2011	
LOSCHER, CHRISTINA 8/8/2011 \$ 32.40 LOSCHER, CHRISTINA 8/15/2011 \$ 60.20 LOSCHER, CHRISTINA 9/15/2011 \$ 34.33 LOSCHER, CHRISTINA 9/26/2011 \$ 140.20 LOUISIANA DEPT. OF HEALTH& HOSPITALS 7/5/2011 \$ 49,100.36 LOUISIANA DEPT. OF HEALTH& HOSPITALS 9/15/2011 \$ 52.25 LOVAS, MR RANDALL 7/25/2011 \$ 1,058.55 LOVAS, MR RANDALL 7/25/2011 \$ 217.41 LOVAS, MR RANDALL 9/26/2011 \$ 45.25 LOVAS, RANDALL 7/8/2011 \$ 2,899.37 LOVAS, RANDALL 7/22/2011 \$ 2,899.38 LOVAS, RANDALL 8/5/2011 \$ 2,932.65 LOVAS, RANDALL 8/5/2011 \$ 5,492.59	LOSCHER, CHRISTINA	7/5/2011	\$ 60.00
LOSCHER, CHRISTINA 8/15/2011 \$ 60.20 LOSCHER, CHRISTINA 9/15/2011 \$ 34.33 LOSCHER, CHRISTINA 9/26/2011 \$ 140.20 LOUISIANA DEPT. OF HEALTH& HOSPITALS 7/5/2011 \$ 49,100.36 LOUISIANA DEPT. OF HEALTH& HOSPITALS 9/15/2011 \$ 52.25 LOVAS, MR RANDALL 7/25/2011 \$ 1,058.55 LOVAS, MR RANDALL 8/8/2011 \$ 217.41 LOVAS, MR RANDALL 9/26/2011 \$ 45.25 LOVAS, RANDALL 7/8/2011 \$ 2,899.37 LOVAS, RANDALL 7/22/2011 \$ 2,899.38 LOVAS, RANDALL 8/5/2011 \$ 2,932.65 LOVAS, RANDALL 8/15/2011 \$ 5,492.59	LOSCHER, CHRISTINA	7/25/2011	\$ 57.00
LOSCHER, CHRISTINA 9/15/2011 \$ 34.33 LOSCHER, CHRISTINA 9/26/2011 \$ 140.20 LOUISIANA DEPT. OF HEALTH& HOSPITALS 7/5/2011 \$ 49,100.36 LOUISIANA DEPT. OF HEALTH& HOSPITALS 9/15/2011 \$ 52.25 LOVAS, MR RANDALL 7/5/2011 \$ 1,058.55 LOVAS, MR RANDALL 8/8/2011 \$ 217.41 LOVAS, MR RANDALL 9/26/2011 \$ 45.25 LOVAS, RANDALL 7/8/2011 \$ 2,899.37 LOVAS, RANDALL 7/22/2011 \$ 2,899.38 LOVAS, RANDALL 8/5/2011 \$ 2,932.65 LOVAS, RANDALL 8/15/2011 \$ 5,492.59	LOSCHER, CHRISTINA	8/8/2011	\$ 32.40
LOSCHER, CHRISTINA 9/26/2011 \$ 140.20 LOUISIANA DEPT. OF HEALTH& HOSPITALS 7/5/2011 \$ 49,100.36 LOUISIANA DEPT. OF HEALTH& HOSPITALS 9/15/2011 \$ 48,654.10 LOVAS, MR RANDALL 7/5/2011 \$ 52.25 LOVAS, MR RANDALL 7/25/2011 \$ 1,058.55 LOVAS, MR RANDALL 8/8/2011 \$ 217.41 LOVAS, MR RANDALL 9/26/2011 \$ 45.25 LOVAS, RANDALL 7/8/2011 \$ 2,899.37 LOVAS, RANDALL 7/22/2011 \$ 2,899.38 LOVAS, RANDALL 8/5/2011 \$ 2,932.65 LOVAS, RANDALL 8/15/2011 \$ 5,492.59	LOSCHER, CHRISTINA	8/15/2011	\$ 60.20
LOUISIANA DEPT. OF HEALTH& HOSPITALS 7/5/2011 \$ 49,100.36 LOUISIANA DEPT. OF HEALTH& HOSPITALS 9/15/2011 \$ 48,654.10 LOVAS, MR RANDALL 7/5/2011 \$ 52.25 LOVAS, MR RANDALL 7/25/2011 \$ 1,058.55 LOVAS, MR RANDALL 8/8/2011 \$ 217.41 LOVAS, MR RANDALL 9/26/2011 \$ 45.25 LOVAS, RANDALL 7/8/2011 \$ 2,899.37 LOVAS, RANDALL 7/22/2011 \$ 2,899.38 LOVAS, RANDALL 8/5/2011 \$ 2,932.65 LOVAS, RANDALL 8/15/2011 \$ 5,492.59	LOSCHER, CHRISTINA	9/15/2011	\$ 34.33
LOUISIANA DEPT. OF HEALTH& HOSPITALS 9/15/2011 \$ 48,654.10 LOVAS, MR RANDALL 7/5/2011 \$ 52.25 LOVAS, MR RANDALL 7/25/2011 \$ 1,058.55 LOVAS, MR RANDALL 8/8/2011 \$ 217.41 LOVAS, MR RANDALL 9/26/2011 \$ 45.25 LOVAS, RANDALL 7/8/2011 \$ 2,899.37 LOVAS, RANDALL 7/22/2011 \$ 2,899.38 LOVAS, RANDALL 8/5/2011 \$ 2,932.65 LOVAS, RANDALL 8/15/2011 \$ 5,492.59	LOSCHER, CHRISTINA		
LOUISIANA DEPT. OF HEALTH& HOSPITALS 9/15/2011 \$ 48,654.10 LOVAS, MR RANDALL 7/5/2011 \$ 52.25 LOVAS, MR RANDALL 7/25/2011 \$ 1,058.55 LOVAS, MR RANDALL 8/8/2011 \$ 217.41 LOVAS, MR RANDALL 9/26/2011 \$ 45.25 LOVAS, RANDALL 7/8/2011 \$ 2,899.37 LOVAS, RANDALL 7/22/2011 \$ 2,899.38 LOVAS, RANDALL 8/5/2011 \$ 2,932.65 LOVAS, RANDALL 8/15/2011 \$ 5,492.59	LOUISIANA DEPT. OF HEALTH& HOSPITALS	7/5/2011	\$ 49,100.36
LOVAS, MR RANDALL 7/5/2011 \$ 52.25 LOVAS, MR RANDALL 7/25/2011 \$ 1,058.55 LOVAS, MR RANDALL 8/8/2011 \$ 217.41 LOVAS, MR RANDALL 9/26/2011 \$ 45.25 LOVAS, RANDALL 7/8/2011 \$ 2,899.37 LOVAS, RANDALL 7/22/2011 \$ 2,899.38 LOVAS, RANDALL 8/5/2011 \$ 2,932.65 LOVAS, RANDALL 8/15/2011 \$ 5,492.59	LOUISIANA DEPT. OF HEALTH& HOSPITALS		
LOVAS, MR RANDALL 8/8/2011 \$ 217.41 LOVAS, MR RANDALL 9/26/2011 \$ 45.25 LOVAS, RANDALL 7/8/2011 \$ 2,899.37 LOVAS, RANDALL 7/22/2011 \$ 2,899.38 LOVAS, RANDALL 8/5/2011 \$ 2,932.65 LOVAS, RANDALL 8/15/2011 \$ 5,492.59	LOVAS, MR RANDALL	7/5/2011	
LOVAS, MR RANDALL 8/8/2011 \$ 217.41 LOVAS, MR RANDALL 9/26/2011 \$ 45.25 LOVAS, RANDALL 7/8/2011 \$ 2,899.37 LOVAS, RANDALL 7/22/2011 \$ 2,899.38 LOVAS, RANDALL 8/5/2011 \$ 2,932.65 LOVAS, RANDALL 8/15/2011 \$ 5,492.59	LOVAS, MR RANDALL	7/25/2011	\$ 1,058.55
LOVAS, MR RANDALL 9/26/2011 \$ 45.25 LOVAS, RANDALL 7/8/2011 \$ 2,899.37 LOVAS, RANDALL 7/22/2011 \$ 2,899.38 LOVAS, RANDALL 8/5/2011 \$ 2,932.65 LOVAS, RANDALL 8/15/2011 \$ 5,492.59	LOVAS, MR RANDALL		
LOVAS, RANDALL 7/8/2011 \$ 2,899.37 LOVAS, RANDALL 7/22/2011 \$ 2,899.38 LOVAS, RANDALL 8/5/2011 \$ 2,932.65 LOVAS, RANDALL 8/15/2011 \$ 5,492.59		· ·	
LOVAS, RANDALL 7/22/2011 \$ 2,899.38 LOVAS, RANDALL 8/5/2011 \$ 2,932.65 LOVAS, RANDALL 8/15/2011 \$ 5,492.59	•		
LOVAS, RANDALL 8/5/2011 \$ 2,932.65 LOVAS, RANDALL 8/15/2011 \$ 5,492.59	·	· '.	
LOVAS, RANDALL 8/15/2011 \$ 5,492.59			•
·			
			•

Name of creditor	Payment date	Amount paid
LOVAS, RANDALL	9/2/2011	3,162.95
LOVAS, RANDALL	9/16/2011	·
LOVAS, RANDALL	9/30/2011	•
LOVELACE HEALTH PLAN	8/4/2011	-
LOVELACE HEALTH PLAN	8/4/2011	
LUKRAFKA, ANDREA	7/5/2011	
LUKRAFKA, ANDREA	7/8/2011	
LUKRAFKA, ANDREA	7/22/2011	
LUKRAFKA, ANDREA	8/5/2011	
LUKRAFKA, ANDREA	8/8/2011	
LUKRAFKA, ANDREA	8/15/2011	"
LUKRAFKA, ANDREA	8/15/2011	
LUKRAFKA, ANDREA	8/19/2011	
LUKRAFKA, ANDREA	8/25/2011	* * * * * * * * * * * * * * * * * * * *
LUKRAFKA, ANDREA	9/2/2011	
LUKRAFKA, ANDREA	9/16/2011	
LUKRAFKA, ANDREA	9/30/2011	•
MACPHERSON, MATTHEW	7/8/2011	
MACPHERSON, MATTHEW	7/22/2011	•
MACPHERSON, MATTHEW	8/5/2011	
MACPHERSON, MATTHEW	8/15/2011	
MACPHERSON, MATTHEW	8/15/2011	*
MACPHERSON, MATTHEW	8/19/2011	. ,
MACPHERSON, MATTHEW	9/2/2011	· ·
MACPHERSON, MATTHEW	9/16/2011	
MACPHERSON, MATTHEW	9/30/2011	
MAGELLAN BEHAVIORAL HEALTH	8/4/2011	
MALONE, JOHN	7/22/2011	·
MALONE, JOHN	7/25/2011	
MALONE, JOHN	8/5/2011	
MALONE, JOHN	8/15/2011	· ·
MALONE, JOHN	8/15/2011	
MALONE, JOHN	8/19/2011	
MALONE, JOHN	9/2/2011	
MALONE, JOHN	9/16/2011	•
MALONE, JOHN	9/26/2011	· · · · · · · · · · · · · · · · · · ·
MALONE, JOHN	9/30/2011	
MANDELLO, GERALD	7/8/2011	•
MANDELLO, GERALD	7/22/2011	
	8/5/2011	
MANDELLO, GERALD MANDELLO, GERALD	8/15/2011	
MANDELLO, GERALD		•
MANDELLO, GERALD MANDELLO, GERALD	8/19/2011	•
MANDELLO, GERALD MANDELLO, GERALD	9/2/2011	
	9/16/2011	•
MANDELLO, GERALD MANDELLO, MR GERALD	9/30/2011	
MARK, DETTBARN	8/15/2011	
	7/8/2011	
MARQUETTE TRANSPORTATION FIN.,INC	7/14/2011	
MARQUETTE TRANSPORTATION FIN.,INC	7/21/2011	
MARQUETTE TRANSPORTATION FIN.,INC	8/4/2011	
MARQUETTE TRANSPORTATION FIN.,INC	8/11/2011	
MARQUETTE TRANSPORTATION FIN., INC	8/18/2011	•
MARQUETTE TRANSPORTATION FIN., INC	9/1/2011	
MARQUETTE TRANSPORTATION FIN.,INC	9/21/2011	\$ 3,191.82

Name of creditor	Payment date	Amount paid
MARSH USA INC.	7/14/2011	\$ 24,157.00
MARSH USA INC.	7/21/2011	
MARSH USA INC.	8/8/2011	\$ 17,595.00
MARSH USA INC.	8/25/2011	· ·
MARSH USA INC.	9/27/2011	\$ 240,990.00
MARSHALL-MCFARLAND, CYNTHIA	8/5/2011	
MARSHALL-MCFARLAND, CYNTHIA	8/19/2011	\$ 2,014.06
MARSHALL-MCFARLAND, CYNTHIA	9/2/2011	
MARSHALL-MCFARLAND, CYNTHIA	9/16/2011	
MARSHALL-MCFARLAND, CYNTHIA	9/30/2011	
MARTINEZ, EMILIO	7/5/2011	
MARTINEZ, EMILIO	7/25/2011	
MARTINEZ, EMILIO	8/15/2011	
MARTINEZ, EMILIO	8/25/2011	
MARTINEZ, EMILIO	9/15/2011	\$ 46.07
MARTINEZ, EMILIO	9/21/2011	
MARTINEZ, EMILIO J	7/8/2011	
MARTINEZ, EMILIO J	7/22/2011	•
MARTINEZ, EMILIO J	8/5/2011	
MARTINEZ, EMILIO J	8/15/2011	
MARTINEZ, EMILIO	8/19/2011	•
MARTINEZ, EMILIO J	9/2/2011	•
MARTINEZ, EMILIO J	9/16/2011	· ·
MARTINEZ, EMILIO J	9/30/2011	•
MASERGY COMMUNICATIONS, INC.	7/14/2011	
MASERGY COMMUNICATIONS, INC.	9/1/2011	
MASSACHUSETTS ACADEMY OF DERM.	7/14/2011	
MASTERS DERMATOLOGIC ASSOCIATION	8/18/2011	
MASTERS DERMATOLOGIC ASSOCIATION	9/1/2011	
MATRIX MEDICAL COMMUNICATIONS	8/25/2011	
MCCLELLAN, JOE	8/25/2011	•
MCCLELLAN, JOE K	7/8/2011	
MCCLELLAN, JOE K	7/22/2011	
MCCLELLAN, JOE K	8/5/2011	
MCCLELLAN, JOE K	8/19/2011	
MCCLELLAN, JOE K	9/2/2011	
MCCLELLAN, JOE K	9/16/2011	
MCCLELLAN, JOE K	9/30/2011	
MCCULLOUGH, CYNTHIA	7/8/2011	
MCCULLOUGH, CYNTHIA	7/22/2011	
MCCULLOUGH, CYNTHIA	8/5/2011	
MCCULLOUGH, CYNTHIA	8/19/2011	
MCCULLOUGH, CYNTHIA	9/2/2011	
MCCULLOUGH, CYNTHIA	9/16/2011	,
MCCULLOUGH, CYNTHIA	9/30/2011	-
MCDERMOTT WILL & EMERY LLP	7/5/2011	
MCDERMOTT WILL & EMERY LLP	8/11/2011	
MCDERMOTT WILL & EMERY LLP	9/1/2011	
MCDERMOTT WILL & EMERY LLP	9/28/2011	
MCDERMOTT WILL & EMERY LLP	9/28/2011	
MCDERMOTT, STEVEN	7/8/2011	
MCKESSON PATIENT RELATIONSHIP SOL	7/14/2011	
MCKESSON PATIENT RELATIONSHIP SOL	7/14/2011	·
MCKESSON PATIENT RELATIONSHIP SOL	7/21/2011	
MORESOUTH TATIENT RELATIONSHIP SOL	1/21/2011	₽ 307,431.0i

Name of creditor	Payment date	Amount paid
MCKESSON PATIENT RELATIONSHIP SOL	8/11/2011	\$ 297,180.00
MCKESSON PATIENT RELATIONSHIP SOL	8/18/2011	•
MCKESSON PATIENT RELATIONSHIP SOL	8/18/2011	
MCKESSON PATIENT RELATIONSHIP SOL	8/25/2011	·
MCKESSON PATTENT RELATIONSHIP SOL	9/1/2011	•
MCKESSON PATIENT RELATIONSHIP SOL	9/8/2011	·
MCWHORTER, CHRISTOPHER	7/8/2011	•
MCWHORTER, CHRISTOPHER	7/22/2011	•
MCWHORTER, CHRISTOPHER	8/5/2011	•
MCWHORTER, CHRISTOPHER	8/15/2011	•
MCWHORTER, CHRISTOPHER	8/19/2011	•
MCWHORTER, CHRISTOPHER	9/2/2011	
MCWHORTER, CHRISTOPHER	9/16/2011	*
MCWHORTER, CHRISTOPHER	9/30/2011	
MDSCONNECT, LLC	7/14/2011	·
MEADOWS, LISA	7/8/2011	·
MEADOWS, LISA	7/22/2011	
MEADOWS, LISA	8/5/2011	•
MEADOWS, LISA	8/19/2011	•
MEADOWS, LISA	9/2/2011	·
MEADOWS, LISA	9/16/2011	· ·
MEADOWS, LISA	9/30/2011	•
MED I BANK	7/25/2011	
MED I BANK	7/5/2011	
MED I BANK	7/14/2011	
MED I BANK	8/10/2011	
MED I BANK	9/1/2011	
MED I BANK	9/2/2011	
MED I BANK	9/6/2011	
MED I BANK	9/8/2011	
MED I BANK	9/28/2011	
MEDCO HEALTH SOLUTIONS, INC.	7/14/2011	
MEDCO HEALTH SOLUTIONS, INC.	7/14/2011	•
MEDIMPACT HEALTHCARE SYSTEMS, INC.	9/1/2011	· ·
MEDPHARM LTD	8/16/2011	•
MEDPHARM LTD	9/21/2011	•
MEDPRO SYSTEMS	8/11/2011	•
MENDELL, RONALD	7/5/2011	
MENDELL, RONALD	7/25/2011	
MENDELL, RONALD	8/8/2011	
MENDELL, RONALD	9/26/2011	
MENDELL, RONALD E	7/8/2011	•
MENDELL, RONALD E	7/22/2011	·
MENDELL, RONALD E	8/5/2011	•
MENDELL, RONALD E	8/15/2011	
MENDELL, RONALD E	8/19/2011	
MENDELL, RONALD E	9/2/2011	
MENDELL, RONALD E	9/16/2011	•
MENDELL, RONALD E	9/30/2011	•
META PHARMACEUTICAL SERVICES	8/11/2011	•
METAPHOR	7/14/2011	
METLIFE	7/5/2011	
METLIFE	8/4/2011	
METLIFE	9/1/2011	•
	>, 1, 2011	. 10,010.07

Name of creditor	Payment date	Amount paid
METILER TOLEDO INC.	9/8/2011	\$ 26,240.60
MEYERSON, BORIS	7/8/2011	\$ 5,188.17
MEYERSON, BORIS	7/22/2011	\$ 5,188.17
MEYERSON, BORIS	8/5/2011	\$ 5,188.18
MEYERSON, BORIS	8/19/2011	\$ 5,188.17
MEYERSON, BORIS	9/2/2011	\$ 5,188.18
MEYERSON, BORIS	9/16/2011	\$ 5,188.17
MEYERSON, BORIS	9/30/2011	\$ 5,616.30
MIDDLETON, FRED	7/8/2011	\$ 802.07
MIDDLETON, FRED	7/22/2011	\$ 827.03
MIDDLETON, MARTHA	7/8/2011	\$ 1,522.86
MIDDLETON, MARTHA	7/22/2011	
MIDDLETON, MARTHA	8/5/2011	
MIDDLETON, MARTHA	8/19/2011	
MIDDLETON, MARTHA	9/2/2011	
MIDDLETON, MARTHA	9/16/2011	•
MIDDLETON, MARTHA	9/30/2011	
MILE HIGH UNITED WAY	7/21/2011	
MILHORN, DENNIS	7/8/2011	
MILHORN, DENNIS	7/22/2011	
MILHORN, DENNIS	8/5/2011	
MILLER, JASON	7/8/2011	
MILLER, JASON	7/22/2011	•
MILLER, JASON	8/5/2011	. ,
MILLER, JASON	8/15/2011	
MILLER, JASON	8/19/2011	
MILLER, JASON	9/2/2011	
MILLER, SCOTT	7/5/2011	•
MILLER, SCOTT	7/8/2011	
MILLER, SCOTT	7/11/2011	•
MILLER, SCOTT	7/22/2011	•
MILLER, SCOTT	8/5/2011	·
MILLER, SCOTT	8/8/2011	
MILLER, SCOTT	8/15/2011	
MILLER, SCOTT	8/19/2011	
MILLER, SCOTT	9/2/2011	
MILLER, SCOTT	9/8/2011	4
MILLER, SCOTT	9/15/2011	
MILLER, SCOTT	9/16/2011	
MILLER, SCOTT	9/26/2011	
MILLER, SCOTT	9/30/2011	
MILLS, SARA	9/2/2011	
MILLS, SARA	9/6/2011	
MILLS, SARA	9/16/2011	
MILLS, SARA	9/30/2011	
MISSOURI DERMATOLOGICAL SOCIETY	7/7/2011	
MOODY SPRINKLER CO, INC	9/1/2011	
MOTTAGHIAN, MITRA	7/8/2011	· · · · · · · · · · · · · · · · · · ·
MOTTAGHIAN, MITRA	7/22/2011	
MOTTAGHIAN, MITRA	8/5/2011	
MOTTAGHIAN, MITRA	8/15/2011	•
MOTTAGHIAN, MITRA	8/19/2011	
MOTTAGHIAN, MITRA	9/2/2011	
MOTTAGHIAN, MITRA	9/16/2011	
MOTTVQHIVIN' MITKV	9/16/2011	\$ 2,180.84

Name of creditor	Payment date	Amount paid
MOTTAGHIAN, MITRA	9/30/2011 \$	2,684.87
MULLER, LINDSAY	7/5/2011 \$	
MULLER, LINDSAY	7/8/2011 \$	
MULLER, LINDSAY	7/22/2011 \$	•
MULLER, LINDSAY	8/5/2011 \$	
MULLER, LINDSAY	8/19/2011 \$	•
MULLER, LINDSAY	8/31/2011 \$	•
MULLER, LINDSAY	9/2/2011 \$	
MULLER, LINDSAY	9/16/2011 \$	•
MULLER, LINDSAY	9/30/2011 \$	
MUSICK, JOHN	7/8/2011 \$	·
MUSICK, JOHN	7/22/2011	
MUSICK, JOHN	8/5/2011	
MUSICK, JOHN	8/19/2011	
MUSICK, JOHN	9/2/2011	· .
MUSICK, JOHN	9/16/2011	
MUSICK, JOHN	9/30/2011	
NEEDLE, MARK E	8/15/2011	•
NEOUCOM	9/1/2011	•
NEVADA DEPARTMENT OF TAXATION	7/14/2011	
NEWION, JOEL T	7/8/2011	
NEWTON, JOEL T	7/22/2011	
NEWTON, JOEL T	8/5/2011	
NEWTON, JOEL T	8/19/2011	•
NEWTON, JOEL T	9/2/2011	•
NEWION, JOEL T	9/16/2011	=
NEWTON, JOEL T	9/30/2011	•
NEWTOWN OFFICE & COMPUTER SUPPLY	8/4/2011	
NEWTOWN OFFICE & COMPUTER SUPPLY	8/25/2011	
NEWTOWN OFFICE & COMPUTER SUPPLY	9/21/2011	
NJ MEDICAID DRUG REBATE PROGRAM	8/11/2011	
NJ MEDICAID DRUG REBATE PROGRAM	9/15/2011	· · · · · · · · · · · · · · · · · · ·
NOETIC SPECIALTY INSURANCE COMPANY	8/11/2011	
NOETIC SPECIALTY INSURANCE COMPANY	9/15/2011	•
NOLAN, MICHAEL	7/8/2011	
NOLAN, MICHAEL	7/22/2011	•
NOLAN, MICHAEL	8/5/2011	·
NOLAN, MICHAEL	8/15/2011	
NOLAN, MICHAEL	8/19/2011	
NOLAN, MICHAEL	9/2/2011	
NOLAN, MICHAEL	9/16/2011	
NOLAN, MICHAEL	9/30/2011	
NOLAN, MICHAEL P.	7/5/2011	\$ 104.01
NOLAN, MICHAEL P.	7/25/2011	\$ 21.00
NOLAN, MICHAEL P.	8/15/2011	\$ 17.00
NORTH CAROLINA DERMATOLOGY ASSOC.	7/22/2011	\$ 80.00
NORTH DAKOTA TAX COMMISSIONER	7/14/2011	\$ 4.90
NORTHERN CALIFORNIA DERM SOCIETY	8/8/2011	\$ 2,000.00
NORTHROP GRUMMAN INFO. TECHNOLOGY	8/25/2011	·
NOSEK, MAUREEN	7/5/2011	
NOSEK, MAUREEN	9/15/2011	
NOSEK, MAUREEN	9/26/2011	
NOSEK, MAUREEN A	7/8/2011	
NOSEK, MAUREEN A	7/22/2011	•

Name of creditor	Payment date	Amount paid
NOSEK, MAUREEN A	8/5/2011 \$	3,435.06
NOSEK, MAUREEN A	8/19/2011 \$	
NOSEK, MAUREEN A	9/2/2011 \$	·
NOSEK, MAUREEN A	9/16/2011 \$	· ·
NOSEK, MAUREEN A	9/30/2011 \$	
OCCUPATIONAL & TRAVEL HEALTH	8/11/2011 \$	
OCCUPATIONAL & TRAVEL HEALTH	9/8/2011 \$	
ODUM, SANDRA	7/8/2011 \$	
ODUM, SANDRA	7/22/2011 \$	
ODUM, SANDRA	8/5/2011	
ODUM, SANDRA	9/2/2011	
ODUM, SANDRA	9/6/2011	
ODUM, SANDRA	9/16/2011	
ODUM, SANDRA	9/30/2011	
OFFICE MACHINES & SUPPLY CO.	8/25/2011	
OKLAHOMA HEALTH CARE AUTHORITY	7/5/2011	
OKLAHOMA HEALTH CARE AUTHORITY	7/5/2011	
OKLAHOMA HEALTH CARE AUTHORITY	8/11/2011	•
ONE SOURCE OFF REFRESHMENT SERVICE	7/5/2011	'
ONE SOURCE OFF REFRESHMENT SERVICE	7/14/2011	
ONE SOURCE OFF REFRESHMENT SERVICE	7/21/2011	
ONE SOURCE OFF REFRESHMENT SERVICE	8/4/2011	
ONE SOURCE OFF REFRESHMENT SERVICE	8/11/2011	
ONE SOURCE OFF REFRESHMENT SERVICE	9/15/2011	
ONLINE BUSINESS APPLICATIONS, INC.	7/14/2011	
ONLINE BUSINESS APPLICATIONS, INC.	7/21/2011	
ONLINE BUSINESS APPLICATIONS, INC.	9/21/2011	
OPTUM HEALTH	7/11/2011	-
OPTUM HEALTH	9/16/2011	•
OPTUM HEALTH	7/22/2011	•
OPTUM HEALTH	8/5/2011	•
OPTUM HEALTH	8/19/2011	
OPTUM HEALTH	9/2/2011	
OREGON BOARD OF PHARMACY	7/14/2011	
OREGON BOARD OF PHARMACY	7/14/2011	
OREGON DERMATOLOGY SOCIETY	7/7/2011	
ORKIN PEST CONTROL	8/25/2011	•
OSBORNE, DONNA	9/2/2011	
OSBORNE, DONNA	9/6/2011	
OSBORNE, DONNA	9/16/2011	
OSBORNE, DONNA	9/30/2011	
OVID TECHNOLOGIES, INC	8/4/2011	
PACIFIC DERMATOLOGIC ASSOCIATION	8/11/2011	
PAETEC COMMUNICATIONS, INC.	7/5/2011	
PAETEC COMMUNICATIONS, INC.	8/4/2011	•
PAETEC COMMUNICATIONS, INC.	9/15/2011	-
PAGANO, COSMO	7/8/2011	· ·
PAGANO, COSMO	7/22/2011	" ·
PAGANO, COSMO	8/5/2011	" ,
PAGANO, COSMO	8/15/2011	
PAGANO, COSMO	8/19/2011	
PAGANO, COSMO	9/2/2011	-
PAGANO, COSMO	9/16/2011	•
PAGANO, COSMO	9/30/2011	•
· - ,	9/30/2011	\$ 4,211.80

Name of creditor		Payment date	Amount	paid
PANALPINA INC.		7/22/2011	\$ 2	21,054.56
PANALPINA INC.		8/4/2011		6,506.22
PANALPINA INC.		8/11/2011		6,201.55
PANALPINA INC.		8/25/2011		7,306.28
PANALPINA INC.		9/21/2011		324.84
PARIS-MCBRIDE, CHRISTINA		7/8/2011		1,473.36
PARIS-MCBRIDE, CHRISTINA		7/22/2011		1,473.35
PARIS-MCBRIDE, CHRISTINA		8/5/2011		1,473.35
PARIS-MCBRIDE, CHRISTINA		8/15/2011		5,208.49
PARIS-MCBRIDE, CHRISTINA		8/19/2011		1,473.36
PARIS-MCBRIDE, CHRISTINA		8/31/2011	\$	50.00
PARIS-MCBRIDE, CHRISTINA		9/2/2011		1,473.36
PARIS-MCBRIDE, CHRISTINA	•	9/16/2011		1,473.35
PARIS-MCBRIDE, CHRISTINA		9/30/2011		2,519.97
PASTRANA, LISA		7/8/2011		1,310.38
PASTRANA, LISA		7/22/2011		1,310.37
PASTRANA, LISA		8/5/2011		1,310.39
PASTRANA, LISA		8/15/2011		1,413.00
PASTRANA, LISA		8/19/2011		1,310.37
PASTRANA, LISA		9/2/2011		1,310.39
PASTRANA, LISA		9/16/2011		1,310.37
PASTRANA, LISA		9/30/2011		1,734.57
PASTRANA, MS LISA		7/5/2011		580.10
PASTRANA, MS LISA		7/11/2011		32.00
PASTRANA, MS LISA		7/25/2011		486.58
PASTRANA, MS LISA		9/21/2011		348.20
PERFORMRX		8/18/2011		499.84
PERFORMRX		8/18/2011		1,560.22
PETTY-OWENS, MONAE L		8/15/2011		2,609.78
PHARMACEUTICAL REPRESENTATIVE MAG		8/18/2011		3,550.00
PHARMAKON INC.		8/11/2011		8,227.00
PHILLIPS JR, JIMMY H.		7/8/2011		1,413.90
PHILLIPS JR, JIMMY H.		7/22/2011		1,413.89
PHILLIPS JR, JIMMY H.		8/5/2011		1,979.31
PIETROWSKI, CYNTHIA		7/5/2011		43.48
PIETROWSKI, CYNTHIA		7/8/2011		1,871.70
PIETROWSKI, CYNTHIA		7/22/2011		1,871.69
PIETROWSKI, CYNTHIA		7/25/2011		111.76
PIETROWSKI, CYNTHIA		8/5/2011		1,871.70
PIETROWSKI, CYNTHIA		8/15/2011		10.65
PIETROWSKI, CYNTHIA		8/15/2011		2,104.58
PIETROWSKI, CYNTHIA		8/19/2011		1,871.71
PIETROWSKI, CYNTHIA		9/2/2011		1,871.69
PIETROWSKI, CYNTHIA		9/8/2011		38.08
PIETROWSKI, CYNTHIA		9/15/2011	-	22,30
PIETROWSKI, CYNTHIA		9/16/2011		1,871.70
PIETROWSKI, CYNTHIA		9/30/2011		2,431.65
PINNACLE RESEARCH GROUP LLC		7/5/2011		56,381.80
PINSON, ELIZABETH		7/5/2011		200.00
PINSON, ELIZABETH		7/25/2011		40.00
PINSON, ELIZABETH		8/8/2011		247.02
PINSON, ELIZABETH		8/15/2011		30.00
PINSON, ELIZABETH		9/15/2011	-	30.00
PINSON, ELIZABETH		9/26/2011		30.00
t at 10 Ot 1, Differ 10 Diff.		9/20/2011	r 47	30.00

Name of creditor	Payment date	Amount paid
PINSON, ELIZABETH A.	7/8/2011	\$ 1,911.71
PINSON, ELIZABETH A.	7/22/2011	\$ 1,911.70
PINSON, ELIZABETH A.	8/5/2011	\$ 1,911.71
PINSON, ELIZABETH A.	8/15/2011	\$ 2,926.57
PINSON, ELIZABETH A.	8/19/2011	\$ 1,911.70
PINSON, ELIZABETH A.	9/2/2011	\$ 1,911.71
PINSON, ELIZABETH A.	9/16/2011	\$ 1,911.70
PINSON, ELIZABETH A.	9/30/2011	\$ 1,911.71
PITNEY BOWES	7/7/2011	\$ 177.97
PITNEY BOWES	8/4/2011	\$ 802.99
PITNEY BOWES	8/4/2011	\$ 1,469.12
PITNEY BOWES	8/25/2011	
PLESCIA, JOSEPH	9/30/2011	
POEDTKE, DOUGLAS	7/25/2011	•
POEDTKE, DOUGLAS A.	7/8/2011	
POEDTKE, DOUGLAS A.	7/22/2011	·
POEDTKE, DOUGLAS A.	8/5/2011	
POEDTKE, DOUGLAS A.	8/15/2011	
POEDTKE, DOUGLAS A.	8/19/2011	
POOL, JULIE	7/5/2011	•
POOL, JULIE	7/8/2011	
POOL, JULIE	7/22/2011	
POOL, JULIE	7/25/2011	
POOL, JULIE	8/5/2011	
POOL, JULIE	8/15/2011	
POOL, JULIE	8/15/2011	
POOL, JULIE	8/19/2011	
POOL, JULIE	9/2/2011	\$ 1,660.60
POOL, JULIE	9/16/2011	•
POOL, JULIE	9/30/2011	
PORETTA & ORR INC.	7/5/2011	•
PORETTA & ORR INC.	7/7/2011	•
PORETTA & ORR INC.	8/11/2011	
PORETTA & ORR INC.	9/8/2011	
PPD MEDICAL COMMUNICATION	7/5/2011	
PPD MEDICAL COMMUNICATION	7/21/2011	·
PPD MEDICAL COMMUNICATION	8/25/2011	
PPD MEDICAL COMMUNICATION	9/21/2011	
PREHM, AMBER	7/8/2011	
PREHM, AMBER	7/22/2011	
PREHM, AMBER	8/5/2011	
PREHM, AMBER	8/15/2011	
PREHM, AMBER	8/19/2011	
PREHM, AMBER	9/2/2011	· · · · · · · · · · · · · · · · · · ·
PREHM, AMBER	9/16/2011	•
PREHM, AMBER	9/30/2011	
PRICE, MOLLY	7/8/2011	
PRICE, MOLLY	7/22/2011	
PRICE, MOLLY	8/5/2011	
PRICE, MOLLY	8/15/2011	·
PRICE, MOLLY	8/19/2011	
PRICE, MOLLY	9/2/2011	
PRICE, MOLLY	9/16/2011	
PRICE, MOLLY	9/30/2011	

Name of creditor	Payment date	Amount paid
PRICEWATERHOUSECOOPERS, LLP	8/11/2011	\$ 67,455.00
PRICEWATERHOUSECOOPERS, LLP	8/12/2011	
PRICEWATERHOUSECOOPERS, LLP	9/15/2011	·
PRICEWATERHOUSECOOPERS, LLP	9/26/2011	
PRIME THERAPEUTICS	8/18/2011	· ·
PRIME THERAPEUTICS	8/18/2011	
PRISTINE SPRINGS	7/14/2011	
PRISTINE SPRINGS	7/21/2011	
PRISTINE SPRINGS	8/25/2011	
PRISTINE SPRINGS	9/8/2011	
PROSOFT SOFTWARE, INC.	7/14/2011	
PROSOFT SOFTWARE, INC.	8/11/2011	· · · · · · · · · · · · · · · · · · ·
PROSOFT' SOFTWARE, INC.	8/18/2011	•
PROSOFT SOFTWARE, INC.	9/1/2011	
PROSOFT SOFTWARE, INC.	9/8/2011	
PROSOFT SOFTWARE, INC.	9/21/2011	
PRUDENTIAL	7/21/2011	
PRUDENTIAL	8/11/2011	•
PRUDENTIAL	9/15/2011	
QUADRANT HEALTHCOM, INC.	7/5/2011	•
QUADRANT HEALTHCOM, INC.	8/25/2011	•
QUADRANT HEALTHCOM, INC.	9/1/2011	
RAY, MOLLY	7/5/2011	
RAY, MOLLY	7/25/2011	
RAY, MOLLY	8/8/2011	
RAY, MOLLY	9/8/2011	
RAY, MOLLY	9/15/2011	"
RAY, MOLLY M	7/8/2011	
RAY, MOLLY M	7/22/2011	
RAY, MOLLY M	8/5/2011	
RAY, MOLLY M	8/15/2011	
RAY, MOLLY M	8/19/2011	· ·
RAY, MOLLY M	9/2/2011	•
RAY, MOLLY M	9/16/2011	
RAY, MOLLY M	9/30/2011	
RAYGOZA, J CARLOS	7/8/2011	•
RAYGOZA, J CARLOS	7/22/2011	
RAYGOZA, J CARLOS	8/5/2011	
	8/15/2011	
RAYGOZA, J CARLOS	8/19/2011	and the second s
RAYGOZA, J CARLOS RAYGOZA, J CARLOS		
RAYGOZA, J CARLOS	9/2/2011	
RAYGOZA, J CARLOS	9/16/2011	
	9/30/2011	· ·
REDMOND, JOHN	7/11/2011	
REDMOND, JOHN	7/25/2011	
REDMOND, JOHN	8/8/2011	
REDMOND, JOHN	8/25/2011	
REDMOND, JOHN	9/8/2011	
REDMOND, JOHN	9/26/2011	
REDMOND, JOHN C	7/8/2011	·
REDMOND, JOHN C	7/22/2011	•
REDMOND, JOHN C	8/5/2011	
REDMOND, JOHN C	8/19/2011	
REDMOND, JOHN C	9/2/2011	\$ 3,093.00

Name of creditor	Payment date	Amount paid
REDMOND, JOHN C	9/16/2011	3,093.00
REDMOND, JOHN C	9/30/2011	
RI GENERAL TREASURER	8/25/2011	
RICCHINI JR, JOSEPH P.	7/8/2011	
RICCHINI JR, JOSEPH P.	7/22/2011	
RICCHINI JR, JOSEPH P.	8/5/2011	-
RICCHINI JR, JOSEPH P.	8/19/2011	
RICCHINI JR, JOSEPH P.	9/2/2011	
RICCHINI JR, JOSEPH P.	9/16/2011	
RICCHINI JR, JOSEPH P.	9/30/2011	
RING, CHRISTINA	7/8/2011	
RING, CHRISTINA	7/22/2011	
RING, CHRISTINA	8/5/2011	
RING, CHRISTINA	8/15/2011	•
RING, CHRISTINA	8/19/2011	
RING, CHRISTINA	9/2/2011	
RING, CHRISTINA	9/16/2011	•
RING, CHRISTINA	9/30/2011	·
ROBB, JEFFREY	7/5/2011	
ROBB, JEFFREY	7/8/2011	\$ 1,793.38
ROBB, JEFFREY	7/22/2011	
ROBB, JEFFREY	8/5/2011	\$ 1,793.38
ROBB, JEFFREY	8/15/2011	\$ 1,840.58
ROBB, JEFFREY	8/19/2011	1, 793.37
ROBB, JEFFREY	9/2/2011	\$ 1,793.38
ROBB, JEFFREY	9/16/2011	\$ 1,793.38
ROBB, JEFFREY	9/30/2011	\$ 3,662.01
ROBERTS, KARA	7/8/2011	\$ 852.21
ROBERTS, KARA	7/22/2011	\$ 956.76
ROBERTS, KARA	8/5/2011	\$ 982.90
ROBERTS, KARA	8/19/2011	\$ 1,009.03
ROBERTS, KARA	9/2/2011	\$ 939.35
ROBERTS, KARA	9/16/2011	\$ 1,061.30
ROBERTS, KARA	9/30/2011	
RODRIGUES, STEVE	7/8/2011	•
RODRIGUES, STEVE	7/22/2011	
RODRIGUES, STEVE	8/5/2011	
RODRIGUES, STEVE	8/19/2011	
RODRIGUES, STEVE	9/2/2011	
RODRIGUES, STEVE	9/16/2011	
RODRIGUES, STEVE	9/30/2011	
ROLLER, TINA	7/5/20 1 1	
ROLLER, TINA B	7/8/2011	
ROLLER, TINA B	7/22/2011	
ROLLER, TINA B	8/5/2011	
ROLLER, TINA B	8/19/2011	
ROLLER, TINA B	9/2/2011	
ROLLER, TINA B	9/16/2011	
ROLLER, TINA B	9/30/2011	•
ROOSEVELT, GARY A	7/8/2011	
ROOSEVELT, GARY A	7/22/2011	
ROOSEVELT, GARY A	8/5/2011	
ROOSEVELT, GARY A	8/19/2011	·
ROOSEVELT, GARY A	9/2/2011	\$ 5,165.80

Name of creditor	Payment date	Amount paid
ROOSEVELT, GARY A	9/16/2011	\$ 5,165.81
ROOSEVELT, GARY A	9/30/2011	\$ 5,165.81
ROPES & GRAY LLP	7/5/2011	\$ 228,595.16
ROPES & GRAY LLP	9/1/2011	\$ 263,353.09
ROSENBERG, CARL	7/5/2011	\$ 99.30
ROSENBERG, CARL	7/8/2011	\$ 2,119.21
ROSENBERG, CARL	7/11/2011	\$ 42.43
ROSENBERG, CARL	7/22/2011	\$ 2,119.20
ROSENBERG, CARL	7/25/2011	\$ 6.55
ROSENBERG, CARL	8/5/2011	\$ 2,119.21
ROSENBERG, CARL	8/8/2011	\$ 16.80
ROSENBERG, CARL	8/15/2011	\$ 7,596.52
ROSENBERG, CARL	8/19/2011	\$ 1,719.20
ROSENBERG, CARL	8/25/2011	\$ 15.20
ROSENBERG, CARL	9/2/2011	\$ 1,719.21
ROSENBERG, CARL	9/15/2011	\$ 38.27
ROSENBERG, CARL	9/16/2011	\$ 1,719.20
ROSENBERG, CARL	9/30/2011	\$ 4,442.51
ROSS, KENNETH	7/5/2011	
ROSS, KENNETH	7/8/2011	\$ 2,013.49
ROSS, KENNETH	7/22/2011	
ROSS, KENNETH	7/25/2011	\$ 109.05
ROSS, KENNETH	8/5/2011	
ROSS, KENNETH	8/15/2011	
ROSS, KENNETH	8/19/2011	\$ 2,013.50
ROSS, KENNETH	9/2/2011	\$ 2,013.49
ross, kenneth	9/15/2011	
ROSS, KENNETH	9/16/2011	\$ 2,013.49
ROSS, KENNETH	9/30/2011	
RUBIN, JUDY	7/5/2011	\$ 155.69
RUBIN, JUDY	7/8/2011	\$ 2,200.98
RUBIN, JUDY	7/22/2011	\$ 2,200.99
RUBIN, JUDY	7/25/2011	\$ 15.00
RUBIN, JUDY	8/5/2011	
RUBIN, JUDY	8/8/2011	\$ 47.00
RUBIN, JUDY	8/15/2011	\$ 5.00
RUBIN, JUDY	8/15/2011	\$ 3,667.22
RUBIN, JUDY	8/19/2011	\$ 2,200.99
RUBIN, JUDY	8/25/2011	\$ 5.00
RUBIN, JUDY	9/2/2011	
RUBIN, JUDY	9/15/2011	•
RUBIN, JUDY	9/16/2011	
RUBIN, JUDY	9/30/2011	
RUBIO, JANICE	7/8/2011	
RUBIO, JANICE	7/22/2011	
RUBIO, JANICE	8/5/2011	
RUBIO, JANICE	9/2/2011	
RUBIO, JANICE	9/6/2011	
RUBIO, JANICE	9/16/2011	
RUBIO, JANICE	9/30/2011	
RURAL RETREAT WINERY & VINEYARD LLC	8/4/2011	
RUSSO, ZACHARY M.	8/15/2011	
RUTLEDGE, DANIEL	7/8/2011	•
RUTLEDGE, DANIEL	7/22/2011	

Name of creditor	Payment date	Amount paid	
RUTLEDGE, DANIEL	8/5/2011	\$ 5,080.0	01
RUTLEDGE, DANIEL	8/19/2011		
RUTLEDGE, DANIEL	9/2/2011		
RUILEDGE, DANIEL	9/16/2011		
RUI'LEDGE, DANIEL	9/30/2011	•	
RX PROVISIONS, INC.	7/14/2011		
RX PROVISIONS, INC.	8/25/2011		
RX PROVISIONS, INC.	9/15/2011	•	
RYDER, APRIL D	7/8/2011		
RYDER, APRIL D	7/22/2011		
RYDER, APRIL D	8/5/2011		35
RYDER, APRIL D	8/19/2011		
RYDER, APRIL D	9/2/2011		
RYDER, APRIL D	9/16/2011		
RYDER, APRIL D	9/30/2011		
SALANAI HORSFALL	7/5/2011		
SALANAI HORSFALL	7/11/2011		
SALANAI HORSFALL	7/25/2011		
SALANAI HORSFALL	8/15/2011		
SALANAI HORSFALL	9/8/2011		
SALANAI HORSFALL	9/21/2011		
SALANAI HORSFALL	9/26/2011		.50
SASSAMAN III, MR. ROBERT	7/25/2011		.00
SASSAMAN III, ROBERT	7/8/2011	"	
SASSAMAN III, ROBERT	7/22/2011		
SASSAMAN III, ROBERT	8/5/2011	•	
SASSAMAN III, ROBERT	8/15/2011		
SASSAMAN III, ROBERT	8/19/2011		
SASSAMAN III, ROBERT	9/2/2011	•	
SASSAMAN III, ROBERT	9/16/2011	•	
SASSAMAN III, ROBERT	9/30/2011		
SAVINO, JENNIFER	7/8/2011		
SAVINO, JENNIFER	7/22/2011		
SAVINO, JENNIFER	8/5/2011		
SAVINO, JENNIFER	8/15/2011		
SAVINO, JENNIFER	8/19/2011		
SAVINO, JENNIFER	9/2/2011		
SAVINO, JENNIFER	9/16/2011	" "	
SAVINO, JENNIFER	9/30/2011		
SCHENKER	7/5/2011		.48
SCHENKER	7/14/2011		
SCHENKER	7/21/2011		
SCHENKER	8/4/2011		
SCHENKER	8/18/2011		
SCHENKER	8/25/2011		
SCHENKER	9/8/2011		
SDI HEALTH LLC	7/14/2011		
SELECT HEALTH	9/1/2011	·	5.16
SH MEDIA, LLC	7/14/2011		
SH MEDIA, LLC	9/15/2011	•	5.25
SHAW, BRUCE	7/8/2011		
SHAW, BRUCE	7/22/2011		
SHAW, BRUCE	8/5/2011	-	
SHAW, BRUCE	8/19/2011		
	0/17/2011	7 (ال ال	-40

Name of creditor	Payment date	Amount paid
SHAW, BRUCE	8/25/2011 \$	13.55
SHAW, BRUCE	9/2/2011 \$	
SHAW, BRUCE	9/16/2011	•
SHAW, BRUCE	9/30/2011 \$	•
SHEPHERD, LEIGH D.	7/14/2011	•
SHEPHERD, LEIGH D.	7/21/2011	
SHEPHERD, LEIGH D.	8/4/2011 \$	
SHEPHERD, LEIGH D.	8/11/2011	•
SHEPHERD, LEIGH D.	8/25/2011	
SHEPHERD, LEIGH D.	9/1/2011	
SHEPHERD, LEIGH D.	9/15/2011	
SHERATON GREAT VALLEY HOTEL	8/11/2011	·
SHEWEY, JAMIE	7/8/2011	•
SHEWEY, JAMIE	7/22/2011	
SHEWEY, JAMIE	8/5/2011	
SHEWEY, JAMIE	8/19/2011	
SHEWEY, JAMIE	9/2/2011	
SHEWEY, JAMIE	9/16/2011	
SHEWEY, JAMIE	9/30/2011	
SHURTLEFF, WILLIAM	7/5/2011	
SHURTLEFF, WILLIAM	8/25/2011	
SHURTLEFF, WILLIAM S	7/8/2011	
SHURTLEFF, WILLIAM S	7/22/2011	·
SHURTLEFF, WILLIAM S	8/5/2011	•
SHURTLEFF, WILLIAM S	8/15/2011	
SHURTLEFF, WILLIAM S	8/19/2011	
SHURTLEFF, WILLIAM S	9/2/2011	
SHURTLEFF, WILLIAM S	9/16/2011	
SHURTLEFF, WILLIAM S	9/30/2011	
SIEGEL, DAVID	7/8/2011	· ·
SIEGEL, DAVID	7/11/2011	·
SIEGEL, DAVID	7/22/2011	
SIEGEL, DAVID	7/25/2011	
SIEGEL, DAVID	8/5/2011	
SIEGEL, DAVID	8/8/2011	
SIEGEL, DAVID	8/15/2011	
SIEGEL, DAVID	8/19/2011	·
SIEGEL, DAVID	8/25/2011	
SIEGEL, DAVID		\$ 4,266.02
SIEGEL, DAVID	9/15/2011	
SIEGEL, DAVID	9/16/2011	
SIEGEL, DAVID	9/26/2011	
SIEGEL, DAVID	9/30/2011	
SMELSTOR, JEREMY	7/8/2011	
SMELSTOR, JEREMY	7/22/2011	
SMELSTOR, JEREMY	8/5/2011	
SMELSTOR, JEREMY	8/8/2011	
SMELSTOR, JEREMY		
SMELSTOR, JEREMY	8/15/2011 8/19/2011	
SOCIETY OF DERMATOLOGY PHY. ASSIST.		
SOKOLOW, TERRI L.	8/12/2011 7/8/2011	
SOKOLOW, TERRI L.	7/8/2011	
SOKOLOW, TERRI L.	7/22/2011	•
SOKOLOW, TERRI L.	8/5/2011	
DOROLOW, IEIGH L.	8/19/2011	\$ 1,525.69

Name of creditor	Payment date	Amount paid
SOKOLOW, TERRI L.	9/2/2011	\$ 1,525.70
SOKOLOW, TERRI L.	9/16/2011	•
SOKOLOW, TERRI L.	9/30/2011	•
SOLOMON, MARK LOUIS	8/18/2011	•
SOURCE HEALTHCARE ANALYTICS, INC.	7/21/2011	- ,
SOUTH DAKOTA DEPT, OF SOCIAL SERVIC	7/5/2011	· · · · · · · · · · · · · · · · · · ·
SPARKLE CLEANING SERVICES	7/14/2011	•
SPARKLE CLEANING SERVICES	8/4/2011	· · · · · · · · · · · · · · · · · · ·
SPARKLE CLEANING SERVICES	9/1/2011	·
SPAULDING CLINICAL RESEARCH, LLC	7/14/2011	•
SPAULDING CLINICAL RESEARCH, LLC	9/15/2011	· ·
SPRINT	7/21/2011	
SPRINT	7/21/2011	, ,
SPRINT	8/25/2011	•
STACY'S CARPET AND SMOKEBUSTERS	7/21/2011	
STACY'S CARPET AND SMOKEBUSTERS	8/25/2011	
STACY'S CARPET AND SMOKEBUSTERS	9/1/2011	
STANLEY JR, HAROLD WAYNE	7/8/2011	
STANLEY JR, HAROLD WAYNE	7/0/2011	
STANLEY JR, HAROLD WAYNE	8/5/2011	
STANLEY JR, HAROLD WAYNE	8/19/2011	
STANLEY JR, HAROLD WAYNE	9/2/2011	
STANLEY JR, HAROLD WAYNE	9/2/2011	
STANLEY JR, HAROLD WAYNE	9/30/2011	•
STANLEY, MARK	7/5/2011	•
STANLEY, MARK		
STANLEY, MARK	7/8/2011	
STANLEY, MARK	7/22/2011	
STANLEY, MARK	7/25/2011	
	8/5/2011	
STANLEY, MARK STANLEY, MARK	8/8/2011	
STANLEY, MARK	8/15/2011	
STANLEY, MARK STANLEY, MARK	8/19/2011	· ·
STANLEY, MARK	8/25/2011	
STANLEY, MARK	9/2/2011	·
	9/16/2011	
STANLEY, MARK	9/21/2011	
STANLEY, MARK STATE OF ALASKA DEPT. OF HEALTH & SOCIAL SERVICES	9/30/2011	
STATE OF ALASKA DEFT. OF HEALTH & SOCIAL SERVICES STATE OF DELAWARE	9/15/2011	
	7/22/2011	•
STATE OF HAWAII, DEPT OF HEALTH STATE OF MICHIGAN	7/21/2011	
	7/14/2011	•
STATE OF MISSOURI	9/1/2011	· ·
STATE OF MISSOURI	8/18/2011	
STATE OF MN COOP PURCH VENT	7/21/2011	
STATE OF MN COOP PURCH VENT	8/18/2011	=
STATE OF OHIO	8/23/2011	
STATE OF VERMONT	7/21/2011	•
STAY IN FRONT	7/5/2011	
STAY IN FRONT	7/14/2011	
STAY IN FRONT	7/21/2011	•
STAY IN FRONT	8/4/2011	=
STAY IN FRONT	8/11/2011	· ·
STAY IN FRONT	9/8/2011	
STAY IN FRONT	9/15/2011	\$ 22,313.22

Name of creditor	Payment date	Amount paid
STEEPROCK	7/14/2011 \$	15,600.00
STERLING COMMERCE	7/5/2011 \$	•
STERLING COMMERCE	8/18/2011	
STERLING COMMERCE	9/8/2011	
STUART, ANGELA R	7/8/2011	•
STUART, ANGELA R	7/22/2011	•
STUART, ANGELA R	8/5/2011	
STUART, ANGELA R	8/19/2011	-
STUART, ANGELA R	9/2/2011	
STUART, ANGELA R	9/16/2011	
STUART, ANGELA R	9/30/2011	•
SUCCESSFACTORS	7/21/2011	
SULLIVAN, JEFFREY P	7/8/2011	•
SULLIVAN, JEFFREY P	7/22/2011	•
SULLIVAN, JEFFREY P	8/5/2011	
SULLIVAN, JEFFREY P	8/15/2011	
SULLIVAN, JEFFREY P	8/19/2011	•
SULLIVAN, JEFFREY P	9/2/2011	·
SUN LIFE AND HEALTH INSURANCE COMPA	7/5/2011	•
SUN LIFE AND HEALTH INSURANCE COMPA	7/21/2011	
SUN LIFE AND HEALTH INSURANCE COMPA	8/25/2011	
SUN LIFE AND HEALTH INSURANCE COMPA	9/15/2011	
SUNSET LOGISTICS, INC.	8/11/2011	
SUNSET LOGISTICS, INC.	9/1/2011	
SUNSET LOGISTICS, INC.	9/8/2011	
SWETNAM, THOMAS	7/8/2011	•
SWETNAM, THOMAS	7/22/2011	
SWETNAM, THOMAS	8/5/2011	
SWETNAM, THOMAS	8/19/2011	·
SWETNAM, THOMAS	9/2/2011	*
SWETNAM, THOMAS	9/16/2011	, and the second se
SWETNAM, THOMAS	9/30/2011	- /
TALARICO, AMBER	7/5/2011	•
TALARICO, AMBER	7/8/2011	
TALARICO, AMBER	7/22/2011	
TALARICO, AMBER	8/5/2011	·
TALARICO, AMBER	8/15/2011	•
TALARICO, AMBER	8/19/2011	•
TALARICO, AMBER	9/2/2011	
TALARICO, AMBER	9/8/2011	
TALARICO, AMBER	9/16/2011	
TALARICO, AMBER	9/30/2011	
TALBOT, JON	7/5/2011	
TALBOT, JON	7/11/2011	
TALBOT, JON	7/25/2011	
TALBOT, JON	8/25/2011	
TALBOT, JON	9/26/2011	
TALBOT, JON P.	7/8/2011	
TALBOT, JON P.	7/22/2011	•
TALBOT, JON P.	8/5/2011	
TALBOT, JON P.	8/15/2011	
TALBOT, JON P.	8/19/2011	
TALBOT, JON P.	9/2/2011	·
TALBOT, JON P.	9/16/2011	·
	7, 10, 2011	. 2,075.07

Name of creditor	Payment date	Amount paid
TALBOT, JON P.	9/30/2011	\$ 2,247.10
TAPE RENTAL LIBRARY, INC.	7/5/2011	
TAPE RENTAL LIBRARY, INC.	7/21/2011	
TAPE RENTAL LIBRARY, INC.	9/8/2011	
TAPE RENTAL LIBRARY, INC.	9/21/2011	
TAYLOR, MR ROBERT	7/5/2011	\$ 66.90
TAYLOR, MR ROBERT	7/25/2011	
TAYLOR, MR ROBERT	8/8/2011	
TAYLOR, MR ROBERT	8/25/2011	
TAYLOR, MR ROBERT	9/8/2011	
TAYLOR, ROBERT	7/8/2011	\$ 2,766.22
TAYLOR, ROBERT	7/22/20 1 1	\$ 2,766.21
TAYLOR, ROBERT	8/5/2011	\$ 2,766.22
TAYLOR, ROBERT	8/19/2011	
TAYLOR, ROBERT	9/2/2011	
TAYLOR, ROBERT	9/16/2011	
TAYLOR, ROBERT	9/30/2011	\$ 2,766.22
TENNESSEE CLINICAL RESEARCH CENTER	7/14/2011	
TENNESSEE DEPT. OF HEALTH	7/5/2011	\$ 124,006.87
TENNESSEE DERMATOLOGY SOCIETY	8/18/2011	•
TEXAS DERMATOLOGICAL SOCIETY	7/7/2011	\$ 1,800.00
THERMOCOPY	8/25/2011	
THIN SPRING	7/14/2011	\$ 200.00
THIN SPRING	8/18/2011	
THIN SPRING	9/21/2011	
THOMAS, KEITH	8/4/2011	\$ 570.00
THOMAS, KEITH	8/18/2011	\$ 2,190.00
THOMAS, KEITH	9/1/2011	
THOMAS, KEITH	9/21/2011	-
THOMSON FINANCIAL, LLC	7/21/2011	•
TRAYNHAM, LAURA	7/5/2011	· ·
TRAYNHAM, LAURA	8/15/2011	
TRAYNHAM, LAURA S.	7/8/2011	
TRAYNHAM, LAURA S.	7/22/2011	\$ 2,045.79
TRAYNHAM, LAURA S.	8/5/2011	
TRAYNHAM, LAURA S.	8/15/2011	
TRAYNHAM, LAURA S.	8/19/2011	\$ 2,045.80
TRAYNHAM, LAURA S.	9/30/2011	\$ 2,117.65
TRC VALLEY CREEK ASSOCIATES-C, L.P.	7/7/2011	\$ 54,098.98
TRC VALLEY CREEK ASSOCIATES-C, L.P.	7/14/2011	\$ 5,425.34
TRC VALLEY CREEK ASSOCIATES-C, L.P.	8/4/2011	\$ 54,098.98
TRC VALLEY CREEK ASSOCIATES-C, L.P.	8/25/2011	\$ 5,965.94
TRC VALLEY CREEK ASSOCIATES-C, L.P.	9/1/2011	\$ 54,098.98
TREASURER, STATE OF NEW HAMPSHIRE	9/15/2011	\$ 3,548.40
TREASURER, STATE OF NEW JERSEY	9/15/2011	\$ 120.77
TRIAD PACKAGING INC.	9/21/2011	\$ 1,087.43
TRIALCARD, INC.	8/11/2011	\$ 10,859.98
TRICARE	9/27/2011	\$ 197,481.16
TRI-CITIES INFORMATION MANAGEMENT,	8/25/2011	\$ 840.00
TRI-CITIES INFORMATION MANAGEMENT,	9/8/2011	\$ 403.50
TWO LABS MARKETING LLC	9/15/2011	\$ 30,000.00
ULINE	8/4/2011	
ULINE	9/21/2011	
UMPLEBY, SARAH	7/5/2011	\$ 130.35

LWPLEBY, SARAH 7/22/2011 \$ 1,895 LWPLEBY, SARAH 7/25/2011 \$ 1,895 LWPLEBY, SARAH 8/5/2011 \$ 1,895 LWPLEBY, SARAH 8/15/2011 \$ 1,595 LWPLEBY, SARAH 8/15/2011 \$ 1,955 LWPLEBY, SARAH 8/25/2011 \$ 1,905 LWPLEBY, SARAH 8/25/2011 \$ 1,905 LWPLEBY, SARAH 9/15/2011 \$ 1,895 LWPLEBY, SARAH 9/15/2011 \$ 1,895 LWPLEBY, SARAH 9/16/2011 \$ 1,905 LWPLEBY, SARAH 9/16/2011 \$ 1,905 VALEXANDER & CO. INC. 8/4/2011 \$ 1,905 VALENTINE, LESLIE 7/8/2011 \$ 1,955 VALENTINE, LESLIE 7/8/2011 \$ 1,955 VALENTINE, LESLIE 8/15/2011 \$ 1,955 VALENTINE, LESLIE 8/15/2011 \$ 1,955 VALENTINE, LESLIE 9/2/2011 \$ 1,955 VALENTINE, LESLIE 9/16/2011 \$ 1,955 VALENTINE, LESLIE 9/16/2011 \$ 1,955 VALENTINE, MS LESLIE 9/16/2011 \$ 1,955 VALENTINE, MS LE	Name of creditor	Payment date	Amount paid
LMMELBY, SARAH 7/22/2011 \$ 1,95 LMMPLEBY, SARAH 7/25/2011 \$ 1,95 LMPLEBY, SARAH 8/5/2011 \$ 1,85 LMPLEBY, SARAH 8/5/2011 \$ 1,55 LMPLEBY, SARAH 8/15/2011 \$ 1,254 LMPLEBY, SARAH 8/15/2011 \$ 1,254 LMPLEBY, SARAH 8/25/2011 \$ 1,105 LMPLEBY, SARAH 9/2/2011 \$ 1,895 LMPLEBY, SARAH 9/15/2011 \$ 1,895 LMPLEBY, SARAH 9/15/2011 \$ 1,895 LMPLEBY, SARAH 9/30/2011 \$ 4,002 LMPLEBY, SARAH 9/30/2011 \$ 4,002 LMPLEBY, SARAH 9/30/2011 \$ 4,002 LMPLEBY, SARAH 9/30/2011 \$ 1,955 LMPLEBY, SARAH 9/50/2011 \$ 1,955 VALENTINE, LESLIE 7/8/2011 \$ 1,955 VALENTINE, LESLIE 8/5/2011 \$ 1,955 VALENTINE, LESLIE 8/5/2011 \$ 1,955 VALENTINE, LESLIE 9/2/2011 \$ 1,955 VALENTINE, LESLIE 9/2/2011 \$ 1,955 VALENTINE, LESLIE 9/30/2011 \$ 1,955 VALENTINE, LESLIE 9/30/2011 \$ 1,955 VALENTINE, MS LESLIE 9/15/2011 \$ 1,955 VALENTINE, MS LESLIE	UMPLEBY, SARAH	7/8/2011	1,895.62
LMMLEBY, SARAH 8/8/2011 \$ 3,95 UMPLEBY, SARAH 8/15/2011 \$ 1,754 UMPLEBY, SARAH 8/15/2011 \$ 1,754 UMPLEBY, SARAH 8/15/2011 \$ 1,105 UMPLEBY, SARAH 8/25/2011 \$ 1,109 UMPLEBY, SARAH 9/15/2011 \$ 1,895 UMPLEBY, SARAH 9/15/2011 \$ 1,895 UMPLEBY, SARAH 9/16/2011 \$ 1,895 UMPLEBY, SARAH 9/30/2011 \$ 4,000 VALENTINER, SARAH 9/30/2011 \$ 4,000 VALENTINER, LESLIE 7/8/2011 \$ 1,957 VALENTINE, LESLIE 7/8/2011 \$ 1,957 VALENTINE, LESLIE 8/15/2011 \$ 1,957 VALENTINE, LESLIE 8/15/2011 \$ 1,957 VALENTINE, LESLIE 8/15/2011 \$ 1,957 VALENTINE, LESLIE 9/16/2011 \$ 1,957 VALENTINE, ME LESLIE 9/16/2011 \$ 1,957 VALENTINE, ME LESLIE 9/16/2011 \$ 1,957 VALENTINE, ME LESLIE 9/16/2011 \$ 1,000 <	UMPLEBY, SARAH	7/22/2011	1,895.63
LMPLEBY, SARAH 8/15/2011 \$ 1,75 UMPLEBY, SARAH 8/15/2011 \$ 1,75 UMPLEBY, SARAH 8/15/2011 \$ 1,895 UMPLEBY, SARAH 8/25/2011 \$ 1,895 UMPLEBY, SARAH 9/15/2011 \$ 4,000 WALENTINE, LESLIE 7/8/2011 \$ 195 VALENTINE, LESLIE 7/8/2011 \$ 195 VALENTINE, LESLIE 8/4/2011 \$ 195 VALENTINE, LESLIE 8/5/2011 \$ 195 VALENTINE, LESLIE 8/15/2011 \$ 195 VALENTINE, LESLIE 8/15/2011 \$ 195 VALENTINE, LESLIE 9/2/2011 \$ 195 VALENTINE, LESLIE 9/2/2011 \$ 195 VALENTINE, LESLIE 9/16/2011 \$ 195 VALENTINE, LESLIE 9/16/2011 \$ 195 VALENTINE, LESLIE 9/16/2011 \$ 195 VALENTINE, MS LESLIE 9/16/2011 \$ 195 VALENTINE, MS LESLIE 9/16/2011 \$ 105 VALENTINE, MS LESLIE 9/16/2011 \$ 105 VALENTINE, MS LESLIE 9/16/2011 \$ 105 VALENTINE, MS LESLIE	UMPLEBY, SARAH	7/25/2011	\$ 101.82
UMPLEBY, SARAH 8/15/2011 \$ 1,395 UMPLEBY, SARAH 8/19/2011 \$ 1,895 UMPLEBY, SARAH 9/2/2011 \$ 1,895 UMPLEBY, SARAH 9/15/2011 \$ 1,895 UMPLEBY, SARAH 9/15/2011 \$ 1,895 UMPLEBY, SARAH 9/15/2011 \$ 1,895 UMPLEBY, SARAH 9/30/2011 \$ 1,895 UMPLEBY, SARAH 9/30/2011 \$ 1,995 VALENTORE, LESLIE 7/8/2011 \$ 1,957 VALENTINE, LESLIE 7/8/2011 \$ 1,957 VALENTINE, LESLIE 8/5/2011 \$ 1,957 VALENTINE, LESLIE 8/15/2011 \$ 1,957 VALENTINE, LESLIE 8/15/2011 \$ 1,957 VALENTINE, LESLIE 9/16/2011 \$ 1,957 VALENTINE, MS LESLIE 9/16/2011 \$ 1,957 VALENTINE, MS LESLIE 9/16/2011 \$ 1,957 VALENTINE, MS LESLIE 9/16/2011 \$ 1,957	UMPLEBY, SARAH		
UMPLEBY, SARAH 8/19/2011 \$ 1,895 UMPLEBY, SARAH 9/2/2011 \$ 1,895 UMPLEBY, SARAH 9/15/2011 \$ 1,895 UMPLEBY, SARAH 9/15/2011 \$ 1,895 UMPLEBY, SARAH 9/16/2011 \$ 1,895 UMPLEBY, SARAH 9/16/2011 \$ 1,895 UMPLEBY, SARAH 9/30/2011 \$ 4,002 V. ALEXANDER & CO. INC. 8/4/2011 \$ 1957 VALENTINE, LESLIE 7/8/2011 \$ 1,957 VALENTINE, LESLIE 7/2/2/2011 \$ 1,957 VALENTINE, LESLIE 8/15/2011 \$ 1,957 VALENTINE, LESLIE 8/15/2011 \$ 1,957 VALENTINE, LESLIE 8/15/2011 \$ 1,957 VALENTINE, LESLIE 9/16/2011 \$ 1,957 VALENTINE, ME LESLIE 7/11/2011 \$ 1,057 VALENTINE, ME LESLIE 7/11/2011 \$ 1,057 VALENTINE, ME LESLIE 9/16/2011 \$ 1,057 VALENTINE, ME LESLIE 9/16/2011 \$ 2,07	UMPLEBY, SARAH	8/8/2011	\$ 36.86
LMPLEBY, SARAH 8/25/2011 \$ 1.95 LMPLEBY, SARAH 9/2/2011 \$ 1.895 LMPLEBY, SARAH 9/16/2011 \$ 1.895 LMPLEBY, SARAH 9/16/2011 \$ 1.895 LMPLEBY, SARAH 9/30/2011 \$ 4,000 V. ALEXANDER & CO. INC. 8/4/2011 \$ 1957 VALENTINE, LESLIE 7/8/2011 \$ 1957 VALENTINE, LESLIE 7/8/2011 \$ 1.957 VALENTINE, LESLIE 8/15/2011 \$ 1.957 VALENTINE, LESLIE 8/15/2011 \$ 1.957 VALENTINE, LESLIE 9/2/2011 \$ 1.957 VALENTINE, LESLIE 9/2/2011 \$ 1.957 VALENTINE, LESLIE 9/16/2011 \$ 1.957 VALENTINE, LESLIE 9/16/2011 \$ 1.957 VALENTINE, LESLIE 9/16/2011 \$ 1.957 VALENTINE, ME LESLIE 9/16/2011 \$ 1.957 VALENTINE, ME LESLIE 7/11/2011 \$ 1.21 VALENTINE, ME LESLIE 7/11/2011 \$ 1.21 VALENTINE, ME LESLIE 9/16/2011 \$ 2.00 VALENTINE, ME LESLIE 9/8/2011 \$ 2.00 VALENTINE, ME LESLIE 9/15/2011 \$ 2.00	UMPLEBY, SARAH	8/15/2011	1, 754.81
LMPLEBY, SARAH 9/2/2011 \$ 1,895 LMPLEBY, SARAH 9/15/2011 \$ 1,895 LMPLEBY, SARAH 9/16/2011 \$ 1,895 LMPLEBY, SARAH 9/30/2011 \$ 4,002 V. ALEXANDER & CO. INC. 8/4/2011 \$ 1,957 VALENTINE, LESLIE 7/8/2011 \$ 1,957 VALENTINE, LESLIE 8/15/2011 \$ 1,957 VALENTINE, LESLIE 8/15/2011 \$ 1,957 VALENTINE, LESLIE 9/12/2011 \$ 1,957 VALENTINE, LESLIE 9/12/2011 \$ 1,957 VALENTINE, LESLIE 9/16/2011 \$ 1,957 VALENTINE, LESLIE 9/16/2011 \$ 1,957 VALENTINE, MS LESLIE 7/15/2011 \$ 1,957 VALENTINE, MS LESLIE 7/15/2011 \$ 1,957 VALENTINE, MS LESLIE 7/15/2011 \$ 1,002 VALENTINE, MS LESLIE 9/15/2011 \$ 1,002 VALENTINE, MS LESLIE	UMPLEBY, SARAH	8/19/2011	\$ 1,895.64
LMPLEBY, SARAH 9/2/2011 \$ 1,895 LMPLEBY, SARAH 9/15/2011 \$ 1,895 LMPLEBY, SARAH 9/16/2011 \$ 1,895 LMPLEBY, SARAH 9/30/2011 \$ 4,002 V. ALEXANDER & CO. INC. 8/4/2011 \$ 1,957 VALENTINE, LESLIE 7/8/2011 \$ 1,957 VALENTINE, LESLIE 8/15/2011 \$ 1,957 VALENTINE, LESLIE 8/15/2011 \$ 1,957 VALENTINE, LESLIE 9/12/2011 \$ 1,957 VALENTINE, LESLIE 9/12/2011 \$ 1,957 VALENTINE, LESLIE 9/16/2011 \$ 1,957 VALENTINE, LESLIE 9/16/2011 \$ 1,957 VALENTINE, MS LESLIE 7/15/2011 \$ 1,957 VALENTINE, MS LESLIE 7/15/2011 \$ 1,957 VALENTINE, MS LESLIE 7/15/2011 \$ 1,002 VALENTINE, MS LESLIE 9/15/2011 \$ 1,002 VALENTINE, MS LESLIE	UMPLEBY, SARAH	8/25/2011	\$ 110.07
UMPLEBY, SARAH 9/16/2011 \$ 1,895 UMPLEBY, SARAH 9/30/2011 \$ 4,002 VALEXANDER & CO. INC. 8/4/2011 \$ 195 VALENTINE, LESLIE 7/8/2011 \$ 1,957 VALENTINE, LESLIE 7/22/2011 \$ 1,957 VALENTINE, LESLIE 8/15/2011 \$ 1,957 VALENTINE, LESLIE 8/15/2011 \$ 1,957 VALENTINE, LESLIE 8/15/2011 \$ 1,957 VALENTINE, LESLIE 9/2(2011 \$ 1,957 VALENTINE, LESLIE 9/30/2011 \$ 1,957 VALENTINE, LESLIE 9/30/2011 \$ 1,957 VALENTINE, LESLIE 9/30/2011 \$ 1,957 VALENTINE, MS LESLIE 9/30/2011 \$ 1,957 VALENTINE, MS LESLIE 9/30/2011 \$ 1,957 VALENTINE, MS LESLIE 7/11/2011 \$ 1,957 VALENTINE, MS LESLIE 7/11/2011 \$ 1,007 VALENTINE, MS LESLIE 7/11/2011 \$ 1,007 VALENTINE, MS LESLIE 9/8/2011 \$ 1,007 VALENTINE, MS LESLIE 9/8/2011 \$ 2,007 VALENTINE, MS LESLIE 9/8/2011 \$ 2,00 VALENTINE, MS LESLIE 9/8/2011 \$ 2,00 VALERIO, SENON 7/8/2011 \$ 2,00 VALERIO, SENON 8/5/2011 \$ 2,00	UMPLEBY, SARAH	· · · · · · · · · · · · · · · · · · ·	
UMPLEBY, SARAH 9/30/2011 \$ 4,002 V. ALEXANDER & CO. INC. 8/4/2011 \$ 95 VALENTINE, LESLIE 7/8/2011 \$ 1,957 VALENTINE, LESLIE 7/22/2011 \$ 1,957 VALENTINE, LESLIE 8/5/2011 \$ 1,957 VALENTINE, LESLIE 8/15/2011 \$ 1,957 VALENTINE, LESLIE 8/15/2011 \$ 1,957 VALENTINE, LESLIE 9/2/2011 \$ 1,957 VALENTINE, LESLIE 9/30/2011 \$ 1,957 VALENTINE, MS LESLIE 9/30/2011 \$ 1,957 VALENTINE, MS LESLIE 7/5/2011 \$ 20 VALENTINE, MS LESLIE 7/5/2011 \$ 1,000 VALENTINE, MS LESLIE 9/30/2011 \$ 1,000 VALENTINE, MS LESLIE 9/36/2011 \$ 1,000 VALERIO, SENON 7/32/2011 \$ 2,000 VALERIO, SENON 9/36/2011 \$ 2,000<	UMPLEBY, SARAH	9/15/2011	
UMPLEBY, SARAH 9/30/2011 \$ 4002 V. ALEXANDER & CO. INC. 8/4/2011 \$ 1952 VALENTINE, LESLIE 7/8/2011 \$ 1957 VALENTINE, LESLIE 8/5/2011 \$ 1957 VALENTINE, LESLIE 8/5/2011 \$ 1957 VALENTINE, LESLIE 8/15/2011 \$ 1957 VALENTINE, LESLIE 8/15/2011 \$ 1957 VALENTINE, LESLIE 9/2/2011 \$ 1957 VALENTINE, LESLIE 9/30/2011 \$ 1957 VALENTINE, MISCHIE 9/30/2011 \$ 1957 VALENTINE, MISCHIE 9/30/2011 \$ 120 VALENTINE, MISCHIE 9/30/2011 \$ 120 VALENTINE, MISCHIE 9/30/2011 \$ 120 VALENTINE, MISCHIE 7/11/2011 \$ 14 VALENTINE, MISCHIE 7/25/2011 \$ 20 VALENTINE, MISCHIE 9/8/2011 \$ 20 VALENTINE, MISCHIE		9/16/2011	\$ 1,895.62
V. ALEXANDER & CO. INC. 8/4/2011 \$ 1,95 VALENTINE, LESLIE 7/8/2011 \$ 1,95 VALENTINE, LESLIE 8/5/2011 \$ 1,95 VALENTINE, LESLIE 8/15/2011 \$ 1,95 VALENTINE, LESLIE 8/15/2011 \$ 1,95 VALENTINE, LESLIE 8/15/2011 \$ 1,95 VALENTINE, LESLIE 9/2/2011 \$ 1,95 VALENTINE, LESLIE 9/16/2011 \$ 1,95 VALENTINE, LESLIE 9/16/2011 \$ 1,95 VALENTINE, MS LESLIE 9/16/2011 \$ 1,95 VALENTINE, MS LESLIE 9/16/2011 \$ 1,95 VALENTINE, MS LESLIE 9/16/2011 \$ 20 VALENTINE, MS LESLIE 7/5/2011 \$ 20 VALENTINE, MS LESLIE 9/8/2011 \$ 3 VALENTINE, MS LESLIE 9/8/2011 \$ 3 VALENTINE, MS LESLIE 9/8/2011 \$ 2 VALENTINE, MS LESLIE 9/8/2011 \$ 2 VALENTINE, MS LESLIE 9/15/2011 \$ 2 VALERIO, SENON 7/8/2011 \$ 2,00 VALERIO, SENON 9/2/2011 \$ 2,00 VA	UMPLEBY, SARAH	9/30/2011	
VALENTINE, LESLIE 7/22/2011 \$ 1,957 VALENTINE, LESLIE 8/15/2011 \$ 3,992 VALENTINE, LESLIE 8/15/2011 \$ 3,992 VALENTINE, LESLIE 8/19/2011 \$ 1,955 VALENTINE, LESLIE 9/2/2011 \$ 1,955 VALENTINE, LESLIE 9/30/2011 \$ 4,122 VALENTINE, LESLIE 9/30/2011 \$ 4,122 VALENTINE, MS LESLIE 9/30/2011 \$ 4,122 VALENTINE, MS LESLIE 7/5/2011 \$ 200 VALENTINE, MS LESLIE 7/25/2011 \$ 4 VALENTINE, MS LESLIE 9/25/2011 \$ 3 VALENTINE, MS LESLIE 9/15/2011 \$ 2 VALENTINE, MS LESLIE 9/26/2011 \$ 2,00 VALERIO, SENON 7/8/2011 \$ 2,00 VALERIO, SENON 7/22/2011 \$ 2,00 VALERIO, SENON 8/15/2011 \$ 2,00 VALERIO, SENON			
VALENTINE, LESLIE 7/22/2011 \$ 1,957 VALENTINE, LESLIE 8/15/2011 \$ 3,992 VALENTINE, LESLIE 8/15/2011 \$ 3,992 VALENTINE, LESLIE 8/19/2011 \$ 1,955 VALENTINE, LESLIE 9/2/2011 \$ 1,955 VALENTINE, LESLIE 9/30/2011 \$ 4,122 VALENTINE, LESLIE 9/30/2011 \$ 4,122 VALENTINE, MS LESLIE 9/30/2011 \$ 4,122 VALENTINE, MS LESLIE 7/5/2011 \$ 200 VALENTINE, MS LESLIE 7/25/2011 \$ 4 VALENTINE, MS LESLIE 9/25/2011 \$ 3 VALENTINE, MS LESLIE 9/15/2011 \$ 2 VALENTINE, MS LESLIE 9/26/2011 \$ 2,00 VALERIO, SENON 7/8/2011 \$ 2,00 VALERIO, SENON 7/22/2011 \$ 2,00 VALERIO, SENON 8/15/2011 \$ 2,00 VALERIO, SENON	VALENTINE, LESLIE	7/8/2011	\$ 1,957.54
VALENTINE, LESLIE 8/5/2011 \$ 3,95 VALENTINE, LESLIE 8/15/2011 \$ 3,95 VALENTINE, LESLIE 8/19/2011 \$ 1,95 VALENTINE, LESLIE 9/2/2011 \$ 1,95 VALENTINE, LESLIE 9/16/2011 \$ 1,95 VALENTINE, LESLIE 9/16/2011 \$ 1,95 VALENTINE, MS LESLIE 9/30/2011 \$ 4,122 VALENTINE, MS LESLIE 7/5/2011 \$ 10 VALENTINE, MS LESLIE 7/11/2011 \$ 14 VALENTINE, MS LESLIE 7/25/2011 \$ 12 VALENTINE, MS LESLIE 9/8/2011 \$ 2 VALERIO, SENON 7/8/2011 \$ 2,07 VALERIO, SENON 7/8/2011 \$ 2,07 VALERIO, SENON 8/5/2011 \$ 2,10 VALERIO, SENON 8/15/2011 \$ 2,10 VALERIO, SENON 9/16/2011 \$ 2,10 VALERIO, SENON 9/16/2011	VALENTINE, LESLIE		· ·
VALENTINE, LESLIE 8/15/2011 \$ 3,995 VALENTINE, LESLIE 8/19/2011 \$ 1,957 VALENTINE, LESLIE 9/2/2011 \$ 1,957 VALENTINE, LESLIE 9/16/2011 \$ 1,957 VALENTINE, LESLIE 9/30/2011 \$ 4,127 VALENTINE, MESLIE 7/5/2011 \$ 200 VALENTINE, MS LESLIE 7/11/2011 \$ 100 VALENTINE, MS LESLIE 7/25/2011 \$ 44 VALENTINE, MS LESLIE 8/8/2011 \$ 30 VALENTINE, MS LESLIE 9/15/2011 \$ 22 VALENTINE, MS LESLIE 9/15/2011 \$ 22 VALENTINE, MS LESLIE 9/26/2011 \$ 30 VALERIO, SENON 7/8/2011 \$ 2,00 VALERIO, SENON 7/8/2011 \$ 2,10 VALERIO, SENON 8/5/2011 \$ 2,10 VALERIO, SENON 8/15/2011 \$ 2,20 VALERIO, SENON 8/15/2011 \$ 2,20 VALERIO, SENON 9/2/2011 \$ 2,40 VALERIO, SENON 9/16/2011 \$ 2,40 VALERIO, SENON 9/3/2011	•		
VALENTINE, LESLIE 8/19/2011 \$ 1,95 VALENTINE, LESLIE 9/2/2011 \$ 1,95 VALENTINE, LESLIE 9/30/2011 \$ 1,95 VALENTINE, LESLIE 9/30/2011 \$ 4,12 VALENTINE, MS LESLIE 7/5/2011 \$ 20 VALENTINE, MS LESLIE 7/11/201 \$ 13 VALENTINE, MS LESLIE 7/25/2011 \$ 3 VALENTINE, MS LESLIE 8/8/2011 \$ 3 VALENTINE, MS LESLIE 9/8/2011 \$ 2 VALENTINE, MS LESLIE 9/26/2011 \$ 3 VALENTINE, MS LESLIE 9/26/2011 \$ 2 VALERIO, SENON 7/8/2011 \$ 2,07 VALERIO, SENON 7/8/2011 \$ 2,10 VALERIO, SENON 8/5/2011 \$ 2,10 VALERIO, SENON 8/15/2011 \$ 2,10 VALERIO, SENON 8/15/2011 \$ 2,10 VALERIO, SENON 9/2/2011 \$ 2,10 VALERIO, SENON 9/2/2011 \$ 2,10 VALERIO, SENON 9/16/2011 \$ 2,10 VALERIO, SENON 9/2/2011 \$	VALENTINE, LESLIE		
VALENTINE, LESLIE 9/2/2011 \$ 1,95 VALENTINE, LESLIE 9/16/2011 \$ 1,95 VALENTINE, LESLIE 9/30/2011 \$ 4,12 VALENTINE, MS LESLIE 7/5/2011 \$ 20 VALENTINE, MS LESLIE 7/11/2011 \$ 11 VALENTINE, MS LESLIE 7/11/2011 \$ 14 VALENTINE, MS LESLIE 7/25/2011 \$ 3 VALENTINE, MS LESLIE 9/8/2011 \$ 3 VALENTINE, MS LESLIE 9/8/2011 \$ 2 VALENTINE, MS LESLIE 9/15/2011 \$ 2 VALENTINE, MS LESLIE 9/26/2011 \$ 3 VALENTINE, MS LESLIE 9/26/2011 \$ 2 VALENTINE, MS LESLIE 9/26/2011 \$ 2 VALERIO, SENON 7/8/2011 \$ 2,00 VALERIO, SENON 8/5/2011 \$ 2,10 VALERIO, SENON 8/5/2011 \$ 2,10 VALERIO, SENON 8/5/2011 \$ 2,10 VALERIO, SENON 9/2/2011 \$ 2,10 VALERIO, SENON 9/2/2011 \$ 2,10 VALERIO, SENON 9/3/2011 \$ 2,10 VALERIO, SENON 9/16/2011 \$ 2,10 VALERIO, SENON 9/3/2011 \$ 2,10 VALERIO, SENON 9/16/2011 \$ 2,10 VANABDALEN, DARIN 7/8	·	8/19/2011	•
VALENTINE, LESLIE 9/16/2011 \$ 1,95 VALENTINE, LESLIE 9/30/2011 \$ 4,12° VALENTINE, MS LESLIE 7/5/2011 \$ 200 VALENTINE, MS LESLIE 7/11/2011 \$ 18 VALENTINE, MS LESLIE 7/25/2011 \$ 4 VALENTINE, MS LESLIE 8/8/2011 \$ 30 VALENTINE, MS LESLIE 9/8/2011 \$ 22 VALENTINE, MS LESLIE 9/15/2011 \$ 2 VALENTINE, MS LESLIE 9/26/2011 \$ 30 VALERIO, SENON 7/8/2011 \$ 2,07 VALERIO, SENON 7/8/2011 \$ 2,07 VALERIO, SENON 7/8/2011 \$ 2,10 VALERIO, SENON 8/5/2011 \$ 2,10 VALERIO, SENON 8/15/2011 \$ 2,10 VALERIO, SENON 8/19/2011 \$ 2,10 VALERIO, SENON 9/16/2011 \$ 2,10 VALERIO, SENON 9/16/2011 \$ 2,10 VALERIO, SENON 9/16/2011 \$ 2,10 VALERIO, SENON 9/30/2011 \$ 2,60 VAN ARSDALEN, DARIN 7/8/2011			•
VALENTINE, LESLIE 9/30/2011 \$ 4,12° VALENTINE, MS LESLIE 7/5/2011 \$ 20° VALENTINE, MS LESLIE 7/11/2011 \$ 12° VALENTINE, MS LESLIE 7/25/2011 \$ 4 VALENTINE, MS LESLIE 8/8/2011 \$ 3° VALENTINE, MS LESLIE 9/8/2011 \$ 2° VALENTINE, MS LESLIE 9/26/2011 \$ 3° VALENTINE, MS LESLIE 9/26/2011 \$ 3° VALERIO, SENON 7/8/2011 \$ 2,0° VALERIO, SENON 7/8/2011 \$ 2,0° VALERIO, SENON 8/5/2011 \$ 2,10° VALERIO, SENON 8/15/2011 \$ 2,10° VALERIO, SENON 8/15/2011 \$ 2,10° VALERIO, SENON 9/2/2011 \$ 2,10° VALERIO, SENON 9/30/2011 \$ 2,10° VALERIO, SENON 9/30/2011 \$ 2,10° VAN ARSDALEN, DARIN 7/8/2011			
VALENTINE, MS LESLIE 7/5/2011 \$ 200 VALENTINE, MS LESLIE 7/11/2011 \$ 14 VALENTINE, MS LESLIE 7/25/2011 \$ 4 VALENTINE, MS LESLIE 8/8/2011 \$ 30 VALENTINE, MS LESLIE 9/8/2011 \$ 22 VALENTINE, MS LESLIE 9/15/2011 \$ 22 VALENTINE, MS LESLIE 9/15/2011 \$ 2,07 VALERIO, SENON 7/8/2011 \$ 2,07 VALERIO, SENON 7/22/2011 \$ 2,10 VALERIO, SENON 8/5/2011 \$ 2,10 VALERIO, SENON 8/5/2011 \$ 2,10 VALERIO, SENON 8/15/2011 \$ 2,10 VALERIO, SENON 8/19/2011 \$ 2,10 VALERIO, SENON 9/2/2011 \$ 2,10 VALERIO, SENON 9/16/2011 \$ 2,10 VAN ARSDALEN, DARIN 7/8/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/2/2011 \$ 2,67 VAN ARSDALEN, DARIN	VALENTINE, LESLIE		•
VALENTINE, MS LESLIE 7/11/2011 \$ 18 VALENTINE, MS LESLIE 7/25/2011 \$ 44 VALENTINE, MS LESLIE 8/8/2011 \$ 36 VALENTINE, MS LESLIE 9/8/2011 \$ 22 VALENTINE, MS LESLIE 9/15/2011 \$ 22 VALENTINE, MS LESLIE 9/26/2011 \$ 36 VALERIO, SENON 7/8/2011 \$ 2,07 VALERIO, SENON 7/22/2011 \$ 2,10 VALERIO, SENON 8/5/2011 \$ 2,10 VALERIO, SENON 8/15/2011 \$ 2,10 VALERIO, SENON 8/15/2011 \$ 2,10 VALERIO, SENON 8/19/2011 \$ 2,10 VALERIO, SENON 9/16/2011 \$ 2,10 VALERIO, SENON J. 8/8/2011 \$ 32 VAN ARSDALEN, DARIN 7/8/2011 \$ 2,67 VAN ARSDALEN, DARIN 8/5/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/16/2011	•		
VALENTINE, MS LESLIE 7/25/2011 \$ 4 VALENTINE, MS LESLIE 8/8/2011 \$ 3 VALENTINE, MS LESLIE 9/8/2011 \$ 2 VALENTINE, MS LESLIE 9/15/2011 \$ 2 VALENTINE, MS LESLIE 9/26/2011 \$ 3 VALERIO, SENON 7/8/2011 \$ 2,07 VALERIO, SENON 7/22/2011 \$ 2,10 VALERIO, SENON 8/15/2011 \$ 2,10 VALERIO, SENON 8/15/2011 \$ 2,10 VALERIO, SENON 8/19/2011 \$ 2,10 VALERIO, SENON 8/19/2011 \$ 2,10 VALERIO, SENON 9/2/2011 \$ 2,10 VALERIO, SENON 9/16/2011 \$ 2,10 VALERIO, SENON 9/16/2011 \$ 2,10 VALERIO, SENON 9/30/2011 \$ 2,10 VALERIO, SENON 9/30/2011 \$ 2,0 VAN ARSDALEN, DARIN 7/8/2011 \$ 2,67 <tr< td=""><td>•</td><td></td><td></td></tr<>	•		
VALENTINE, MS LESLIE 8/8/2011 \$ 36 VALENTINE, MS LESLIE 9/8/2011 \$ 22 VALENTINE, MS LESLIE 9/15/2011 \$ 22 VALENTINE, MS LESLIE 9/26/2011 \$ 36 VALERIO, SENON 7/8/2011 \$ 2,07 VALERIO, SENON 7/22/2011 \$ 2,10 VALERIO, SENON 8/5/2011 \$ 2,10 VALERIO, SENON 8/15/2011 \$ 2,10 VALERIO, SENON 8/19/2011 \$ 2,10 VALERIO, SENON 9/2/2011 \$ 2,10 VALERIO, SENON 9/16/2011 \$ 2,10 VALERIO, SENON 9/30/2011 \$ 2,10 VALERIO, SENON 9/30/2011 \$ 2,10 VALERIO, SENON 9/16/2011 \$ 2,10 VALERIO, SENON 9/16/2011 \$ 2,60 VAN ARSDALEN, DARIN 7/22/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/16/2011 \$ 2,6	VALENTINE, MS LESLIE		
VALENTINE, MS LESLIE 9/8/2011 \$ 2.2 VALENTINE, MS LESLIE 9/15/2011 \$ 2.2 VALENTINE, MS LESLIE 9/26/2011 \$ 3.9 VALERIO, SENON 7/8/2011 \$ 2,07 VALERIO, SENON 7/22/2011 \$ 2,10 VALERIO, SENON 8/15/2011 \$ 2,10 VALERIO, SENON 8/15/2011 \$ 2,10 VALERIO, SENON 8/15/2011 \$ 2,10 VALERIO, SENON 9/2/2011 \$ 2,10 VALERIO, SENON 9/16/2011 \$ 2,10 VALERIO, SENON 9/16/2011 \$ 2,10 VALERIO, SENON 9/16/2011 \$ 2,10 VALERIO, SENON 9/30/2011 \$ 2,10 VALERIO, SENON 9/30/2011 \$ 2,0 VAN ARSDALEN, DARIN 7/22/2011 \$ 2,67 VAN ARSDALEN, DARIN 8/5/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/16/2011 \$ 2,67 </td <td>•</td> <td></td> <td></td>	•		
VALENTINE, MS LESLIE 9/15/2011 \$ 2 VALENTINE, MS LESLIE 9/26/2011 \$ 3 VALERIO, SENON 7/8/2011 \$ 2,07 VALERIO, SENON 7/22/2011 \$ 2,10 VALERIO, SENON 8/5/2011 \$ 2,10 VALERIO, SENON 8/15/2011 \$ 7,23 VALERIO, SENON 8/19/2011 \$ 2,10 VALERIO, SENON 9/2/2011 \$ 2,10 VALERIO, SENON 9/16/2011 \$ 2,10 VALERIO, SENON 9/30/2011 \$ 2,00 VALERIO, SENON 9/30/2011 \$ 2,67 VAN ARSDALEN, DARIN 7/8/2011 \$ 2,67 VAN ARSDALEN, DARIN 8/5/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/16/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/16/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/30/2011 \$ 2,67 VANNARATH, CHRISTINA 7/5/2011			
VALENTINE, MS LESLIE 9/26/2011 \$ 36 VALERIO, SENON 7/8/2011 \$ 2,07 VALERIO, SENON 7/22/2011 \$ 2,10 VALERIO, SENON 8/5/2011 \$ 2,10 VALERIO, SENON 8/15/2011 \$ 7,23 VALERIO, SENON 8/15/2011 \$ 2,10 VALERIO, SENON 9/2/2011 \$ 2,10 VALERIO, SENON 9/16/2011 \$ 2,10 VALERIO, SENON 9/16/2011 \$ 2,10 VALERIO, SENON 9/30/2011 \$ 2,00 VALERIO, SENON 9/30/2011 \$ 2,00 VALERIO, SENON J. 8/8/2011 \$ 32 VAN ARSDALEN, DARIN 7/8/2011 \$ 2,67 VAN ARSDALEN, DARIN 7/8/2011 \$ 2,67 VAN ARSDALEN, DARIN 8/19/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/2/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/2/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/30/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/30/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/30/2011 \$ 2,67 VANNARATH, CHRISTINA 7/5/2011 \$ 1,76 VANNARATH, CHRISTINA 7/22/2011 \$ 1,76 VANNARATH, CHRISTINA 8/5/2011 \$ 1,76 VANNARATH, CHRI			
VALERIO, SENON 7/8/2011 \$ 2,07 VALERIO, SENON 7/22/2011 \$ 2,10 VALERIO, SENON 8/5/2011 \$ 2,10 VALERIO, SENON 8/15/2011 \$ 7,23 VALERIO, SENON 8/19/2011 \$ 2,10 VALERIO, SENON 9/2/2011 \$ 2,10 VALERIO, SENON 9/16/2011 \$ 2,10 VALERIO, SENON 9/16/2011 \$ 2,10 VALERIO, SENON 9/30/2011 \$ 2,00 VALERIO, SENON 9/30/2011 \$ 2,67 VALERIO, SENONJ 7/8/2011 \$ 2,67 VAN ARSDALEN, DARIN 7/22/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/16/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/16/2011 \$ 2,67 VANNARATH, CHRISTINA 7/5/2011 \$ 4 VANNARATH, CHRISTINA 7/8/2011 \$ 1,76 VANNARATH, CHRISTINA 8/5/2011 \$ 1,26			
VALERIO, SENON 7/22/2011 \$ 2,10 VALERIO, SENON 8/5/2011 \$ 2,10 VALERIO, SENON 8/15/2011 \$ 7,23 VALERIO, SENON 8/19/2011 \$ 2,10 VALERIO, SENON 9/2/2011 \$ 2,10 VALERIO, SENON 9/16/2011 \$ 2,10 VALERIO, SENON 9/30/2011 \$ 2,10 VAN ARSDALEN, DARIN 7/8/2011 \$ 2,67 VAN ARSDALEN, DARIN 8/8/2011 \$ 2,67 VAN ARSDALEN, DARIN 8/19/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/2/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/16/2011 \$ 2,67 VANNARATH, CHRISTINA 7/5/2011 \$ 1,76 VANNARATH, CHRISTINA 7/22/2011 \$ 1,76 VANNARATH, CHRI	VALERIO, SENON	7/8/2011	\$ 2,076.59
VALERIO, SENON 8/5/2011 \$ 2,10 VALERIO, SENON 8/15/2011 \$ 7,23 VALERIO, SENON 8/19/2011 \$ 2,10 VALERIO, SENON 9/2/2011 \$ 2,10 VALERIO, SENON 9/16/2011 \$ 2,10 VALERIO, SENON 9/30/2011 \$ 5,70 VALERIO, SENON J. 8/8/2011 \$ 32 VAN ARSDALEN, DARIN 7/8/2011 \$ 2,67 VAN ARSDALEN, DARIN 7/22/2011 \$ 2,67 VAN ARSDALEN, DARIN 8/19/2011 \$ 2,67 VAN ARSDALEN, DARIN 8/19/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/2/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/16/2011 \$ 2,67 VANNARATH, CHRISTINA 7/8/2011 \$ 1,76 VANNARATH, CHRISTINA 7/8/2011 \$ 1,76 VANNARATH, CHRISTINA 7/22/2011 \$ 1,76 VANNARATH, CHRISTINA 8/15/2011 \$ 1,76 VANNARATH, CHRISTINA		7/22/2011	
VALERIO, SENON 8/15/2011 \$ 7,23 VALERIO, SENON 8/19/2011 \$ 2,10 VALERIO, SENON 9/2/2011 \$ 2,10 VALERIO, SENON 9/16/2011 \$ 2,10 VALERIO, SENON 9/30/2011 \$ 5,70 VALERIO, SENON J. 8/8/2011 \$ 32 VAN ARSDALEN, DARIN 7/8/2011 \$ 2,67 VAN ARSDALEN, DARIN 7/22/2011 \$ 2,67 VAN ARSDALEN, DARIN 8/5/2011 \$ 2,67 VAN ARSDALEN, DARIN 8/19/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/2/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/16/2011 \$ 2,67 VANNARATH, CHRISTINA 7/5/2011 \$ 2,67 VANNARATH, CHRISTINA 7/8/2011 \$ 1,76 VANNARATH, CHRISTINA 7/22/2011 \$ 1,76 VANNARATH, CHRISTINA 8/5/2011 \$ 1,76 VANNARATH, CHRISTINA 8/15/2011 \$ 1,76 VANNARATH, CHRISTINA	VALERIO, SENON		
VALERIO, SENON 8/19/2011 \$ 2,10 VALERIO, SENON 9/2/2011 \$ 2,10 VALERIO, SENON 9/16/2011 \$ 2,10 VALERIO, SENON 9/30/2011 \$ 5,70 VALERIO, SENON J. 8/8/2011 \$ 32 VAN ARSDALEN, DARIN 7/8/2011 \$ 2,67 VAN ARSDALEN, DARIN 8/5/2011 \$ 2,67 VAN ARSDALEN, DARIN 8/19/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/2/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/16/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/16/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/16/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/30/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/30/2011 \$ 2,67 VAN ARSDALEN, DARIN 7/5/2011 \$ 4 VANNARATH, CHRISTINA 7/5/2011 \$ 1,76 VANNARATH, CHRISTINA 7/22/2011 \$ 1,76 VANNARATH, CHRISTINA 8/5/2011 \$ 1,76 VANNARATH, CHRISTINA 8/15/2011 \$ 1,58 VANNARATH, CHRISTINA 8/15/2011 \$ 1,58 VANNARATH, CHRISTINA			
VALERIO, SENON 9/2/2011 \$ 2,10 VALERIO, SENON 9/16/2011 \$ 2,10 VALERIO, SENON 9/30/2011 \$ 5,70 VALERIO, SENON J. 8/8/2011 \$ 32 VAN ARSDALEN, DARIN 7/8/2011 \$ 2,67 VAN ARSDALEN, DARIN 8/5/2011 \$ 2,67 VAN ARSDALEN, DARIN 8/19/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/2/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/2/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/16/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/16/2011 \$ 2,67 VANNARATH, CHRISTINA 7/5/2011 \$ 4 VANNARATH, CHRISTINA 7/8/2011 \$ 1,76 VANNARATH, CHRISTINA 7/25/2011 \$ 1,26 VANNARATH, CHRISTINA 8/5/2011 \$ 1,26 VANNARATH, CHRISTINA </td <td></td> <td></td> <td></td>			
VALERIO, SENON 9/16/2011 \$ 2,10 VALERIO, SENON 9/30/2011 \$ 5,70 VALERIO, SENON J. 8/8/2011 \$ 32 VAN ARSDALEN, DARIN 7/8/2011 \$ 2,67 VAN ARSDALEN, DARIN 8/5/2011 \$ 2,67 VAN ARSDALEN, DARIN 8/19/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/2/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/16/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/16/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/30/2011 \$ 2,67 VANNARATH, CHRISTINA 7/5/2011 \$ 4 VANNARATH, CHRISTINA 7/8/2011 \$ 1,76 VANNARATH, CHRISTINA 7/22/2011 \$ 1,76 VANNARATH, CHRISTINA 8/5/2011 \$ 1,76 VANNARATH, CHRISTINA 8/15/2011 \$ 1,76 VANNARATH			
VALERIO, SENON 9/30/2011 \$ 5,70 VALERIO, SENON J. 8/8/2011 \$ 32 VAN ARSDALEN, DARIN 7/8/2011 \$ 2,67 VAN ARSDALEN, DARIN 8/5/2011 \$ 2,67 VAN ARSDALEN, DARIN 8/19/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/2/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/16/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/30/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/30/2011 \$ 2,67 VANNARATH, CHRISTINA 7/5/2011 4 VANNARATH, CHRISTINA 7/8/2011 \$ 1,76 VANNARATH, CHRISTINA 7/22/2011 1,76 VANNARATH, CHRISTINA 8/5/2011 1,76 VANNARATH, CHRISTINA 8/15/2011 1,76 VANNARATH, CHRISTINA 8/15/2011 1,58 VANNARATH, CHRISTINA 8/15/2011 1,58 VANNARATH, CHRISTINA 8/15/2011 1,58 VANNARATH, CHRISTINA 8/25/2011 1,58		• •	
VALERIO, SENON J. 8/8/2011 \$ 32 VAN ARSDALEN, DARIN 7/8/2011 \$ 2,67 VAN ARSDALEN, DARIN 8/5/2011 \$ 2,67 VAN ARSDALEN, DARIN 8/19/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/2/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/16/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/30/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/30/2011 \$ 2,67 VANNARATH, CHRISTINA 7/5/2011 \$ 4 VANNARATH, CHRISTINA 7/8/2011 \$ 1,76 VANNARATH, CHRISTINA 7/22/2011 \$ 1,76 VANNARATH, CHRISTINA 8/5/2011 \$ 1,76 VANNARATH, CHRISTINA 8/5/2011 \$ 1,76 VANNARATH, CHRISTINA 8/5/2011 \$ 1,76 VANNARATH, CHRISTINA 8/15/2011 \$ 1,58 VANNARATH, CHRISTINA 8/15/2011 \$ 1,58 VANNARATH, CHRISTINA 8/25/2011 \$ 1,58			
VAN ARSDALEN, DARIN 7/8/2011 \$ 2,67 VAN ARSDALEN, DARIN 8/5/2011 \$ 2,67 VAN ARSDALEN, DARIN 8/19/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/2/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/16/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/16/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/30/2011 \$ 2,67 VANNARATH, CHRISTINA 7/5/2011 \$ 4 VANNARATH, CHRISTINA 7/8/2011 \$ 1,76 VANNARATH, CHRISTINA 7/22/2011 \$ 1,76 VANNARATH, CHRISTINA 8/5/2011 \$ 1,76 VANNARATH, CHRISTINA 8/15/2011 \$ 1,58 VANNARATH, CHRISTINA 8/15/2011 \$ 1,58 VANNARATH, CHRISTINA 8/25/2011 \$ 1,58 VANNARATH, CHRISTINA 8/25/2011 \$ 1,58			•
VAN ARSDALEN, DARIN 7/22/2011 \$ 2,67 VAN ARSDALEN, DARIN 8/5/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/2/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/16/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/30/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/30/2011 \$ 2,67 VANNARATH, CHRISTINA 7/5/2011 \$ 4 VANNARATH, CHRISTINA 7/8/2011 \$ 1,76 VANNARATH, CHRISTINA 7/22/2011 \$ 1,76 VANNARATH, CHRISTINA 8/5/2011 \$ 1,76 VANNARATH, CHRISTINA 8/15/2011 \$ 1,76 VANNARATH, CHRISTINA 8/15/2011 \$ 1,58 VANNARATH, CHRISTINA 8/15/2011 \$ 1,58 VANNARATH, CHRISTINA 8/25/2011 \$ 1,58			
VAN ARSDALEN, DARIN 8/5/2011 \$ 2,67 VAN ARSDALEN, DARIN 8/19/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/2/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/30/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/30/2011 \$ 2,67 VANNARATH, CHRISTINA 7/5/2011 \$ 4 VANNARATH, CHRISTINA 7/8/2011 \$ 1,76 VANNARATH, CHRISTINA 7/22/2011 \$ 1,76 VANNARATH, CHRISTINA 8/5/2011 \$ 1,76 VANNARATH, CHRISTINA 8/15/2011 \$ 1,58 VANNARATH, CHRISTINA 8/15/2011 \$ 1,58 VANNARATH, CHRISTINA 8/25/2011 \$ 2,67			
VAN ARSDALEN, DARIN 8/19/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/2/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/16/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/30/2011 \$ 2,67 VANNARATH, CHRISTINA 7/5/2011 \$ 4 VANNARATH, CHRISTINA 7/8/2011 \$ 1,76 VANNARATH, CHRISTINA 7/22/2011 \$ 1,76 VANNARATH, CHRISTINA 8/5/2011 \$ 1,76 VANNARATH, CHRISTINA 8/15/2011 \$ 1,58 VANNARATH, CHRISTINA 8/25/2011 \$ 3 VANNARATH, CHRISTINA 8/25/2011 \$ 3			
VAN ARSDALEN, DARIN 9/2/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/16/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/30/2011 \$ 2,67 VANNARATH, CHRISTINA 7/5/2011 \$ 4 VANNARATH, CHRISTINA 7/8/2011 \$ 1,76 VANNARATH, CHRISTINA 7/22/2011 \$ 1,76 VANNARATH, CHRISTINA 8/5/2011 \$ 1,76 VANNARATH, CHRISTINA 8/15/2011 \$ 1,76 VANNARATH, CHRISTINA 8/15/2011 \$ 1,58 VANNARATH, CHRISTINA 8/25/2011 \$ 1,58 VANNARATH, CHRISTINA 8/25/2011 \$ 1,58	·	•	•
VAN ARSDALEN, DARIN 9/16/2011 \$ 2,67 VAN ARSDALEN, DARIN 9/30/2011 \$ 2,67 VANNARATH, CHRISTINA 7/5/2011 \$ 4 VANNARATH, CHRISTINA 7/8/2011 \$ 1,76 VANNARATH, CHRISTINA 7/22/2011 \$ 12 VANNARATH, CHRISTINA 8/5/2011 \$ 1,76 VANNARATH, CHRISTINA 8/15/2011 \$ 1,58 VANNARATH, CHRISTINA 8/25/2011 \$ 1,58 VANNARATH, CHRISTINA 8/25/2011 \$ 3	•		
VAN ARSDALEN, DARIN 9/30/2011 \$ 2,67 VANNARATH, CHRISTINA 7/5/2011 \$ 4 VANNARATH, CHRISTINA 7/8/2011 \$ 1,76 VANNARATH, CHRISTINA 7/22/2011 \$ 12 VANNARATH, CHRISTINA 8/5/2011 \$ 1,76 VANNARATH, CHRISTINA 8/15/2011 \$ 1,58 VANNARATH, CHRISTINA 8/25/2011 \$ 1,58 VANNARATH, CHRISTINA 8/25/2011 \$ 3	•		
VANNARATH, CHRISTINA 7/5/2011 \$ 4 VANNARATH, CHRISTINA 7/8/2011 \$ 1,76 VANNARATH, CHRISTINA 7/22/2011 \$ 12 VANNARATH, CHRISTINA 8/5/2011 \$ 1,76 VANNARATH, CHRISTINA 8/15/2011 \$ 1,58 VANNARATH, CHRISTINA 8/25/2011 \$ 1,58 VANNARATH, CHRISTINA 8/25/2011 \$ 3	•		
VANNARATH, CHRISTINA 7/8/2011 \$ 1,76 VANNARATH, CHRISTINA 7/22/2011 \$ 1,76 VANNARATH, CHRISTINA 7/25/2011 \$ 12 VANNARATH, CHRISTINA 8/5/2011 \$ 1,76 VANNARATH, CHRISTINA 8/15/2011 \$ 1,58 VANNARATH, CHRISTINA 8/25/2011 \$ 3	•		
VANNARATH, CHRISTINA 7/22/2011 \$ 1,76 VANNARATH, CHRISTINA 7/25/2011 \$ 12 VANNARATH, CHRISTINA 8/5/2011 \$ 1,76 VANNARATH, CHRISTINA 8/15/2011 \$ 1,58 VANNARATH, CHRISTINA 8/25/2011 \$ 3		the state of the s	
VANNARATH, CHRISTINA 7/25/2011 \$ 12 VANNARATH, CHRISTINA 8/5/2011 \$ 1,76 VANNARATH, CHRISTINA 8/15/2011 \$ 1,58 VANNARATH, CHRISTINA 8/25/2011 \$ 3	·		
VANNARATH, CHRISTINA 8/5/2011 \$ 1,76 VANNARATH, CHRISTINA 8/15/2011 \$ 1,58 VANNARATH, CHRISTINA 8/25/2011 \$ 3			•
VANNARATH, CHRISTINA 8/15/2011 \$ 1,58 VANNARATH, CHRISTINA 8/25/2011 \$ 3	·		
VANNARATH, CHRISTINA 8/25/2011 \$ 3			
	•		
	VANNARATH, CHRISTINA		

Name of creditor	Payment date	Amount paid
VB CONSULTING LLC	9/8/2011	1,150.00
VENEZIANO, KEITH	7/5/2011	129,49
VENEZIANO, KEITH	7/8/2011	2,997.05
VENEZIANO, KEITH	7/11/2011	4.00
VENEZIANO, KEITH	7/22/2011	2,997.06
VENEZIANO, KEITH	7/25/2011	82.54
VENEZIANO, KEITH	8/5/2011	2,997.06
VENEZIANO, KEITH	8/8/2011	86.00
VENEZIANO, KEITH	8/15/2011	8,187.17
VENEZIANO, KEITH	8/19/2011	3,193.65
VENEZIANO, KEITH	8/31/2011	81.25
VENEZIANO, KEITH	9/2/2011	3,193.66
VENEZIANO, KEITH	9/16/2011	\$ 3,501.40
VENEZIANO, KEITH	9/30/2011	5,613.20
VENTIV COMMERCIAL SERVICE,LLC	7/5/2011	\$ 22,392.89
VENTIV COMMERCIAL SERVICE, LLC	7/14/2011	\$ 15,746.05
VENTIV COMMERCIAL SERVICE,LLC	7/21/2011	\$ 5,108.18
VENTIV COMMERCIAL SERVICE,LLC	9/1/2011	\$ 10,248.29
VENTIV COMMERCIAL SERVICE,LLC	9/21/2011	\$ 12,280.78
VERIZON	7/22/2011	\$ 171.80
VERIZON	8/4/2011	\$ 80.28
VERIZON	8/25/2011	\$ 86.06
VERIZON CABS	7/14/2011	\$ 294.26
VERIZON CABS	9/15/2011	\$ 294.92
VERIZON CABS	9/15/2011	\$ 294.92
VERIZON WIRELESS	7/21/2011	\$ 60.29
VERIZON WIRELESS	8/25/2011	\$ 60.07
VISION SERVICE PLAN	7/5/2011	\$ 2,236.68
VISION SERVICE PLAN	7/21/2011	\$ 2,312.97
VISION SERVICE PLAN	8/18/2011	\$ 2,153.77
VIVACARE, INC	7/5/2011	\$ 24,000.00
VIVACARE, INC	7/14/2011	\$ 24,000.00
WACHOVIA	7/12/2011	\$ 267.91
WACHOVIA	7/12/2011	\$ 2,725.70
WACHOVIA	8/10/2011	\$ 175.34
WACHOVIA	8/10/2011	\$ 2,948.84
WACHOVIA	9/12/2011	\$ 180.43
WACHOVIA	9/12/2011	\$ 2,853.01
WACHTELL, LIPTON, ROSEN & KATZ	7/15/2011	\$ 16,167.99
WACHTELL, LIPTON, ROSEN & KATZ	8/10/2011	\$ 49,261.56
WACHTELL, LIPTON, ROSEN & KATZ	8/30/2011	\$ 94,211.98
WACHTELL, LIPTON, ROSEN & KATZ	9/21/2011	\$ 430,322.12
WACHTELL, LIPTON, ROSEN & KATZ	9/27/2011	\$ 285,774.47
WARMELINK, DAVID	7/11/2011	\$ 117.40
WARMELINK, DAVID	8/15/2011	\$ 82.58
WARMELINK, DAVID	8/31/2011	\$ 60.95
WARMELINK, DAVID	9/15/2011	\$ 12.00
WARMELINK, DAVID E	7/8/2011	
WARMELINK, DAVID E	7/22/2011	•
WARMELINK, DAVID E	8/5/2011	
WARMELINK, DAVID E	8/15/2011	
WARMELINK, DAVID E	8/19/2011	•
WARMELINK, DAVID E	9/2/2011	
11 I I I I I I I I I I I I I I I I I I		

WARMLINK, DAVID E 9/30/2011 \$ 5000 WASHINGTON DEPARTMENT OF HEALTH 8/8/2011 \$ 700.56 WASHINGTON STATE DEPT, OF REVENUE 8/2/2011 \$ 700.56 WASHINGTON STATE DERNATOLOGY ASSOC 7/7/2011 \$ 1,500.00 WASKO, PETER 7/11/2011 \$ 360.94 WASKO, PETER 7/12/2011 \$ 3,724.31 WASKO, PETER 8/5/2011 \$ 3,724.31 WASKO, PETER 8/5/2011 \$ 3,724.31 WASKO, PETER 8/15/2011 \$ 727.64 WASKO, PETER 8/15/2011 \$ 77.64 WASKO, PETER 8/15/2011 \$ 77.64 WASKO, PETER 8/15/2011 \$ 77.64 WASKO, PETER 8/15/2011 \$ 100.47 WASKO, PETER 8/19/2011 \$ 100.47 WASKO, PETER 9/2/2011 \$ 100.47 WASKO, PETER 9/12/2011 \$ 10.61 WASKO, PETER	Name of creditor	Payment date	Amount paid
WASHINGTON DEPARTMENT OF HEALTH 8/18/2011 \$ 500.00 WASHINGTON STATE DEFT OF REVENUE 1/2011 \$ 1,500.00 WASHINGTON STATE DEFT OF REVENUE 7/7/2011 \$ 1,500.00 WASKO, PETER 7/18/2011 \$ 3,504.11 WASKO, PETER 7/12/2011 \$ 3,00.91 WASKO, PETER 8/5/2011 \$ 3,724.31 WASKO, PETER 8/5/2011 \$ 7,724.31 WASKO, PETER 8/8/2011 \$ 226.82 WASKO, PETER 8/15/2011 \$ 7,748.165 WASKO, PETER 8/15/2011 \$ 7,481.65 WASKO, PETER 8/15/2011 \$ 7,764.15 WASKO, PETER 8/15/2011 \$ 160.47 WASKO, PETER 8/15/2011 \$ 4,266.79 WASKO, PETER 9/2/2011 \$ 4,266.79 WASKO, PETER 9/16/2011 \$ 4,266.79 WASKO, PETER 9/16/2011 \$ 160.10 WASKO, PETER 9/16/2011 \$ 12.00	WARMELINK, DAVID E	9/30/2011	\$ 6,607.93
WASHINGTON STATE DERMATOLOGY ASSOC 7/7/2011 \$ 1,000.0 WASKO, PETER 7/8/2011 \$ 3,584.12 WASKO, PETER 7/8/2011 \$ 3,584.12 WASKO, PETER 7/11/2011 \$ 3,584.12 WASKO, PETER 8/5/2011 \$ 3,724.31 WASKO, PETER 8/5/2011 \$ 3,724.31 WASKO, PETER 8/15/2011 \$ 277.44 WASKO, PETER 8/15/2011 \$ 7,781.65 WASKO, PETER 8/15/2011 \$ 3,776.15 WASKO, PETER 8/15/2011 \$ 3,776.15 WASKO, PETER 8/15/2011 \$ 100.47 WASKO, PETER 9/8/2011 \$ 106.47 WASKO, PETER 9/14/2011 \$ 28.11 WASKO, PETER 9/14/2011 \$ 28.11 WASKO, PETER 9/14/2011 \$ 28.07 WASKO, PETER 9/14/2011 \$ 10.60 WASKO, PETER 9/14/2011 \$			
WASRINGTON STATE DERMATOLOGY ASSOC 7/7/2011 \$ 3,584.12 WASKO, PETER 7/81,2011 \$ 3,60.94 WASKO, PETER 7/11/2011 \$ 3,60.94 WASKO, PETER 8/5/2011 \$ 3,724.31 WASKO, PETER 8/8/2011 \$ 236.82 WASKO, PETER 8/15/2011 \$ 7,748.16 WASKO, PETER 8/15/2011 \$ 7,748.16 WASKO, PETER 8/15/2011 \$ 7,748.16 WASKO, PETER 8/15/2011 \$ 10.47 WASKO, PETER 8/15/2011 \$ 10.47 WASKO, PETER 9/2/2011 \$ 4286.79 WASKO, PETER 9/16/2011 \$ 4286.79 WASKO, PETER 9/16/2011 \$ 4286.79 WASKO, PETER 9/30/2011 \$ 7,155.50 WASKO, PETER 9/30/2011 \$ 10.610 WASKO, PETER 9/30/2011 \$ 10.610 WASKO, PETER 9/30/2011 \$ 281 WASKO, PETER 9/30/2011 \$ 10.610 WASKO, PETER 9/30/2011 \$ 10.610 WASKO, PETER 9/30/2011 <t< td=""><td>WASHINGTON STATE DEPT. OF REVENUE</td><td></td><td></td></t<>	WASHINGTON STATE DEPT. OF REVENUE		
WASKO, PETER 7/18/2011 \$ 360.94 WASKO, PETER 7/11/2011 \$ 3,724.31 WASKO, PETER 7/22/2011 \$ 3,724.31 WASKO, PETER 8/5/2011 \$ 226.82 WASKO, PETER 8/8/2011 \$ 226.82 WASKO, PETER 8/15/2011 \$ 7,481.65 WASKO, PETER 8/15/2011 \$ 7,481.65 WASKO, PETER 8/15/2011 \$ 160.47 WASKO, PETER 8/15/2011 \$ 160.47 WASKO, PETER 9/2/2011 \$ 4286.79 WASKO, PETER 9/16/2011 \$ 126.67 WASKO, PETER 9/16/2011 \$ 126.67 <td>WASHINGTON STATE DERMATOLOGY ASSOC</td> <td></td> <td></td>	WASHINGTON STATE DERMATOLOGY ASSOC		
WASKO, PETER 7/12/2011 \$ 3,724,31 WASKO, PETER 8/5/2011 \$ 3,724,31 WASKO, PETER 8/5/2011 \$ 25,682 WASKO, PETER 8/5/2011 \$ 277,64 WASKO, PETER 8/15/2011 \$ 7,716,15 WASKO, PETER 8/15/2011 \$ 7,716,15 WASKO, PETER 8/15/2011 \$ 7,716,15 WASKO, PETER 8/15/2011 \$ 106,47 WASKO, PETER 9/2/2011 \$ 4,286,79 WASKO, PETER 9/16/2011 \$ 4,286,79 WASKO, PETER 9/16/2011 \$ 4,286,79 WASKO, PETER 9/16/2011 \$ 106,10 WASKO, PETER 9/16/2011 \$ 7,155,50 WASKO, PETER 9/30/2011 \$ 7,155,50 WASKO, PETER 9/30	WASKO, PETER		•
WASKO, PETER 3,724.31 WASKO, PETER 8/5/2011 \$ 3,724.31 WASKO, PETER 8/5/2011 \$ 226.82 WASKO, PETER 8/15/2011 \$ 7.74.41 WASKO, PETER 8/15/2011 \$ 7,74.15 WASKO, PETER 8/11/2011 \$ 3,776.15 WASKO, PETER 8/31/2011 \$ 4,286.79 WASKO, PETER 9/2/2011 \$ 4,286.79 WASKO, PETER 9/16/2011 \$ 106.10 WASKO, PETER 9/16/2011 \$ 228.12 WASKO, PETER 9/2/12/211 \$ 228.12 WASKO, PETER 9/2/12/2011 \$ 228.12 WASKO, PETER 9/3/2011 \$ 106.10 WASKO, PETER 9/2/12/2011 \$ 228.12 WASKO, PETER 9/2/12/2011 \$ 228.12 WASKO, PETER 9/3/2011 \$ 106.10 WASKO, PETER 9/12/2011 \$ 281.02 WASKO, PETER 9/12/2011 \$ 281.02 WASKO, PETER 9/12/2011 \$ 281.02 WASKO, PETER 9/12/2011 \$ 281.02 <t< td=""><td>·</td><td></td><td></td></t<>	·		
WASKO, PETER 8/5/2011 \$ 3,724.31 WASKO, PETER 8/8/2011 \$ 236.82 WASKO, PETER 8/15/2011 \$ 7,481.65 WASKO, PETER 8/15/2011 \$ 7,481.65 WASKO, PETER 8/19/2011 \$ 106.10 WASKO, PETER 8/31/2011 \$ 106.47 WASKO, PETER 9/2/2011 \$ 4,286.79 WASKO, PETER 9/16/2011 \$ 4,286.79 WASKO, PETER 9/30/2011 \$ 7,155.50 WASKO, PETER 9		• •	-
WASKO, PETER 8/8/2011 \$ 236.82 WASKO, PETER 8/15/2011 \$ 7,481.65 WASKO, PETER 8/15/2011 \$ 7,481.65 WASKO, PETER 8/19/2011 \$ 3,776.15 WASKO, PETER 8/31/2011 \$ 160.47 WASKO, PETER 9/2/2011 \$ 4,286.79 WASKO, PETER 9/8/2011 \$ 106.10 WASKO, PETER 9/16/2011 \$ 7,155.50 WASKO, PETER 9/30/2011 \$ 1,155.50 WASKO, PETER 9/30/2011 \$ 1,960.31 WASKO, PETER 9/30/2011 \$ 1,960.31 WASKO, PETER 9/30/2011 \$ 1,960.31 WEBB, MIRLAM 7			
WASKO, PETER 8/15/2011 \$ 7,481.65 WASKO, PETER 8/15/2011 \$ 3,776.15 WASKO, PETER 8/15/2011 \$ 1,60.47 WASKO, PETER 8/31/2011 \$ 4,28.679 WASKO, PETER 9/8/2011 \$ 160.47 WASKO, PETER 9/8/2011 \$ 106.10 WASKO, PETER 9/8/2011 \$ 92.81 WASKO, PETER 9/31/2011 \$ 92.81 WASKO, PETER 9/30/2011 \$ 71,55.50 WASKO, PETER 9/30/2011 \$ 42.86.79 WASKO, PETER 9/30/2011 \$ 42.86.79 WASKO, PETER 9/30/2011 \$ 45.33 WASTE MANAGEMENT OF TRI CITIES 8/18/2011 \$ 56.30 WEBB, MIRIAM 7/8/2011 \$ 58.65 WEBB, MIRIAM 7/8/2011 \$ 58.65 WEBB, MIRIAM 8/5/2011 \$ 85.07 WEBBE, COMMUNICATIONS, INC 7/14/2011 \$ 49.00 WEBBEX COMMUNICATIONS, INC 8/25/2011 \$ 300.00 WEBBEX COMMUNICATIONS, INC 8/25/2011 \$ 300.00 WEBEX COMMUNICATIONS, INC 8/25/2011 \$ 196.91 WEBEX COMMUNICATIONS, INC 8/25/2011 \$ 196.91 WEBEX, KIRK 7/8/2011 \$ 1,969.19 WEBEX, KIRK 9/15/2011 \$ 1,969.19			
WASKO, PETER 8/15/2011 \$ 7,481.65 WASKO, PETER 8/19/2011 \$ 160.47 WASKO, PETER 8/31/2011 \$ 160.47 WASKO, PETER 9/2/2011 \$ 428.679 WASKO, PETER 9/16/2011 \$ 428.679 WASKO, PETER 9/16/2011 \$ 42.86.79 WASKO, PETER 9/16/2011 \$ 42.86.79 WASKO, PETER 9/30/2011 \$ 7,155.50 WASKO, PETER 9/30/2011 \$ 463.38 WASKO, PETER 9/30/2011 \$ 56.07 WASKO, PETER 9/30/2011 \$ 463.38 WASKO, PETER 9/30/2011 \$ 40.00 WEBB, MIRLAM 7/12/2011 \$ 19.00			
WASKO, PETER 8/19/2011 \$ 160.47 WASKO, PETER 8/31/2011 \$ 160.47 WASKO, PETER 9/2/2011 \$ 42.86.79 WASKO, PETER 9/8/2011 \$ 106.10 WASKO, PETER 9/16/2011 \$ 42.86.79 WASKO, PETER 9/31/2011 \$ 22.81 WASKO, PETER 9/30/2011 \$ 71,55.50 WASKO, PETER 9/30/2011 \$ 43.38 WASTE MANAGEMENT OF TRI CITIES 8/18/2011 \$ 85.60 WEBB, MIRIAM 7/8/2011 \$ 85.60 WEBB, MIRIAM 7/8/2011 \$ 99.00 WEBB, MIRIAM 8/5/2011 \$ 99.00 WEBBE, COMMUNICATIONS, INC 8/4/2011 \$ 49.00 WEBBEX COMMUNICATIONS, INC 8/4/2011 \$ 49.00 WEBEX COMMUNICATIONS, INC 8/4/2011 \$ 49.00 WEBEX COMMUNICATIONS, INC 8/4/2011 \$ 19.00 WEBEX COMMUNICATIONS, INC 8/4/2011 \$ 19.00 WEBEX COMMUNICATIONS, INC 8/15/2011 \$ 19.00 WEEMS, KIRK 7/8/2011 \$ 19.00 WEEMS, KIRK 7/8/2011 \$ 19.00 WEEMS, KIRK 8/15/2011 \$ 19.00 WEEMS, KIRK 9/10/2011 \$ 19.00 WEEMS, KIRK 9/2/2011 \$ 19.00 <th< td=""><td></td><td></td><td></td></th<>			
WASKO, PETTER 8/31/2011 \$ 160.47 WASKO, PETTER 9/2/2011 \$ 106.10 WASKO, PETTER 9/8/2011 \$ 106.10 WASKO, PETTER 9/16/2011 \$ 42.86.79 WASKO, PETTER 9/16/2011 \$ 22.81 WASKO, PETTER 9/30/2011 \$ 7.155.50 WASKO, PETTER 9/30/2011 \$ 63.38 WASKO, PETTER 9/30/2011 \$ 63.38 WASTE MANAGEMENT OF TRI CITIES 8/18/2011 \$ 155.00 WEBB, MIRIAM 7/8/2011 \$ 98.65 WEBB, MIRIAM 7/22/2011 \$ 98.65 WEBB, MIRIAM 8/5/2011 \$ 98.65 WEBBX COMMUNICATIONS, INC 7/14/2011 \$ 49.00 WEBBX COMMUNICATIONS, INC 8/42/2011 \$ 49.00 WEBBX COMMUNICATIONS, INC 8/25/2011 \$ 49.00 WEBEX COMMUNICATIONS, INC 8/25/2011 \$ 49.00 WEEMS, KIRK 7/12/2011 \$ 19.69.17 WEEMS, KIRK 7/22/2011 \$ 19.69.17 WEEMS, KIRK 7/22/2011 \$ 1.969.17 WEEMS, KIRK 8/15/2011 \$ 1.969.17 WEEMS, KIRK 9/16/2011 \$ 1.969.18 WEEMS, KIRK 9/16/2011 \$ 1.969.18 WEEMS, KIRK 9/16/2011 \$ 1.969.18	WASKO, PETER		·
WASKO, PETER 9/8/2011 \$ 106.10 WASKO, PETER 9/8/2011 \$ 4,286.79 WASKO, PETER 9/16/2011 \$ 2,28.17 WASKO, PETER 9/30/2011 \$ 22.81 WASKO, PETER 9/30/2011 \$ 35.50 WASKE, PETER 9/30/2011 \$ 36.38 WASTE MANAGEMENT OF TRI CITIES 8/18/2011 \$ 515.00 WEBB, MIRIAM 7/8/2011 \$ 586.07 WEBB, MIRIAM 7/2/2011 \$ 598.65 WEBB, MIRIAM 8/5/2011 \$ 886.00 WEBB, COMMUNICATIONS, INC 8/4/2011 \$ 49.00 WEBEX COMMUNICATIONS, INC 8/4/2011 \$ 300.00 WEBEX COMMUNICATIONS, INC 8/25/2011 \$ 300.00 WEBEX COMMUNICATIONS, INC 8/15/2011 \$ 1969.17 WEBEX, KIRK 7/18/2011 \$ 1,969.17 WEBEX, KIRK 7/18/2011 \$ 1,969.18 WEBEX, KIRK 8/15/2011 \$ 1,969.18 WEEMS, KIRK 8/15/2011 \$ 1,969.18 WEEMS, KIRK 9/16/2011 \$ 1,969.19	WASKO, PETER		· ·
WASKO, PETER 9/8/2011 \$ 106.10 WASKO, PETER 9/16/2011 \$ 22.81 WASKO, PETER 9/21/2011 \$ 72.53 WASKO, PETER 9/30/2011 \$ 71,55.50 WASTE MANAGEMENT OF TRI CITIES 8/18/2011 \$ 515.00 WEBB, MIRIAM 7/8/2011 \$ 586.07 WEBB, MIRIAM 7/22/2011 \$ 598.65 WEBB, MIRIAM 8/5/2011 \$ 49.00 WEBEX COMMUNICATIONS, INC 7/14/2011 \$ 49.00 WEBEX COMMUNICATIONS, INC 8/25/2011 \$ 300.00 WEBEX COMMUNICATIONS, INC 9/15/2011 \$ 49.00 WEBEX COMMUNICATIONS, INC 9/15/2011 \$ 49.00 WEBEX COMMUNICATIONS, INC 9/15/2011 \$ 19.69.17 WEEMS, KIRK 7/8/2011 \$ 19.69.17 WEEMS, KIRK 7/8/2011 \$ 19.69.19 WEEMS, KIRK 8/5/2011 \$ 19.69.19 WEEMS, KIRK 9/16/2011 \$ 19.69.19	WASKO, PETER		
WASKO, PETER 9/16/2011 \$ 9.81 WASKO, PETER 9/21/2011 \$ 9.81 WASKO, PETER 9/30/2011 \$ 7,155.50 WASKE MANAGEMENT OF TRI CITIES 7/14/2011 \$ 463.38 WASTE MANAGEMENT OF TRI CITIES 8/18/2011 \$ 856.07 WEBB, MIRIAM 7/22/2011 \$ 98.65 WEBB, MIRIAM 7/22/2011 \$ 98.60 WEBB, COMMUNICATIONS, INC 8/4/2011 \$ 49.00 WEBEX COMMUNICATIONS, INC 8/25/2011 \$ 30.00 WEBEX COMMUNICATIONS, INC 8/25/2011 \$ 1.969.17 WEBEX COMMUNICATIONS, INC 8/25/2011 \$ 1.969.10 WEBEX, KIRK 7/22/2011 \$ 1.969.11 WEEMS, KIRK 7/22/2011 \$ 1.969.12 WEEMS, KIRK 8/15/2011 \$ 1.969.12 WEEMS, KIRK 8/15/2011 \$ 1.969.12 WEEMS, KIRK 9/16/2011 \$ 1.969.12	WASKO, PETER		·
WASKO, PETER 9/21/2011 \$ 7,155.50 WASKO, PETER 9/30/2011 \$ 7,155.50 WASTE MANAGEMENT OF TRI CITIES 71/4/2011 \$ 463.38 WASTE MANAGEMENT OF TRI CITIES 8/18/2011 \$ 515.00 WEBB, MIRIAM 7/8/2011 \$ 598.65 WEBB, MIRIAM 7/2/2011 \$ 598.65 WEBB, MIRIAM 8/5/2011 \$ 49.00 WEBEX COMMUNICATIONS, INC 8/4/2011 \$ 49.00 WEBEX COMMUNICATIONS, INC 8/25/2011 \$ 300.00 WEBEX COMMUNICATIONS, INC 8/25/2011 \$ 300.00 WEBEX COMMUNICATIONS, INC 8/25/2011 \$ 190.01 WEEMS, KIRK 7/8/2011 \$ 190.01 WEEMS, KIRK 7/22/2011 \$ 190.01 WEEMS, KIRK 7/22/2011 \$ 1,969.19 WEEMS, KIRK 8/5/2011 \$ 1,969.18 WEEMS, KIRK 8/19/2011 \$ 1,969.18 WEEMS, KIRK 9/2/2011 \$ 1,969.18 WEEMS, KIRK 9/10/2011 \$ 1,969.18 WEEMS, KIRK 9/10/2011 \$ 1,969.18 WEEMS, KIRK 9/2/2011 \$ 1,969.18 WEEMS, KIRK 9/30/2011 \$ 2,880.88 WEEMS, KIRK 9/30/2011 \$ 1,969.18 WEEMS, KIRK 9/30/2011 \$ 1,969.18	WASKO, PETER		
WASTE MANAGEMENT OF TRI CITIES 7/14/2011 \$ 515.00 WASTE MANAGEMENT OF TRI CITIES 8/18/2011 \$ 515.00 WEBB, MIRIAM 7/8/2011 \$ 598.65 WEBB, MIRIAM 7/22/2011 \$ 598.65 WEBB, MIRIAM 8/5/2011 \$ 598.65 WEBB, MIRIAM 8/5/2011 \$ 49.00 WEBEX COMMUNICATIONS, INC 7/14/2011 \$ 49.00 WEBEX COMMUNICATIONS, INC 8/25/2011 \$ 300.00 WEBEX COMMUNICATIONS, INC 9/15/2011 \$ 1969.17 WEEMS, KIRK 7/22/2011 \$ 1969.17 WEEMS, KIRK 7/22/2011 \$ 1969.19 WEEMS, KIRK 8/15/2011 \$ 1969.19 WEEMS, KIRK 9/16/2011 \$ 1969.19 WEEMS, KIRK 9/16/2011 \$ 1969.19 WEEMS, KIRK 9/16/2011 \$ 1969.19 WELLS FARGO 7/12/2011 \$ 1969.19 <	WASKO, PETER		•
WASTE MANAGEMENT OF TRI CITIES 7/14/2011 \$ 463.38 WASTE MANAGEMENT OF TRI CITIES 8/18/2011 \$ 556.07 WEBB, MIRIAM 7/22/2011 \$ 598.65 WEBB, MIRIAM 8/5/2011 \$ 598.65 WEBB, MIRIAM 8/5/2011 \$ 690.65 WEBB, MIRIAM 8/5/2011 \$ 49.00 WEBEX COMMUNICATIONS, INC 8/4/2011 \$ 49.00 WEBEX COMMUNICATIONS, INC 8/25/2011 \$ 300.00 WEEMS, KIRK 7/15/2011 \$ 49.00 WEEMS, KIRK 7/8/2011 \$ 1969.17 WEEMS, KIRK 7/22/2011 \$ 1,969.17 WEEMS, KIRK 8/15/2011 \$ 1,969.18 WEEMS, KIRK 8/15/2011 \$ 1,969.18 WEEMS, KIRK 8/15/2011 \$ 1,969.19 WEEMS, KIRK 8/15/2011 \$ 1,969.19 WEEMS, KIRK 9/2/2011 \$ 1,969.19 WEEMS, KIRK 9/16/2011 \$ 1,969.19 WEEMS, KIRK 9/16/2011 \$ 1,969.19 WEEMS, KIRK 9/16/2011 \$ 1,969.19 WEELS FARGO	WASKO, PETER	9/30/2011	\$ 7,155.50
WASTE MANAGEMENT OF TRI CITIES 8/18/2011 \$ 515.00 WEBB, MIRIAM 7/8/2011 \$ 596.07 WEBB, MIRIAM 7/22/2011 \$ 598.65 WEBB, MIRIAM 8/5/2011 \$ 886.00 WEBEX COMMUNICATIONS, INC 8/4/2011 \$ 49.00 WEBEX COMMUNICATIONS, INC 8/25/2011 \$ 300.00 WEBEX COMMUNICATIONS, INC 9/15/2011 \$ 49.00 WEEMS, KIRK 7/8/2011 \$ 1969.19 WEEMS, KIRK 7/8/2011 \$ 1969.19 WEEMS, KIRK 8/15/2011 \$ 1969.19 WEEMS, KIRK 8/15/2011 \$ 1969.19 WEEMS, KIRK 8/19/2011 \$ 1969.19 WEEMS, KIRK 9/2/2011 \$ 1969.19 WEEMS, KIRK 9/16/2011 \$ 1969.19 WELLS FARGO 7/11/2011 \$ 1969.19 WELLS FARGO 7/11/2011 \$ 106.36 WELLS FARGO 7/11/201	WASTE MANAGEMENT OF TRI CITIES		
WEBB, MIRIAM 7/22/2011 \$ 598.65 WEBB, MIRIAM 8/5/2011 \$ 88.00 WEBEX COMMUNICATIONS, INC 7/14/2011 \$ 49.00 WEBEX COMMUNICATIONS, INC 8/25/2011 \$ 300.00 WEBEX COMMUNICATIONS, INC 8/25/2011 \$ 300.00 WEEMS, KIRK 7/8/2011 \$ 1969.17 WEEMS, KIRK 7/22/2011 \$ 1,969.19 WEEMS, KIRK 8/5/2011 \$ 1,969.19 WEEMS, KIRK 8/15/2011 \$ 1,969.19 WEEMS, KIRK 8/15/2011 \$ 1,969.19 WEEMS, KIRK 8/19/2011 \$ 1,969.19 WEEMS, KIRK 9/2/2011 \$ 1,969.18 WEEMS, KIRK 9/16/2011 \$ 1,969.18 WELLS FARGO 7/18/2011<	WASTE MANAGEMENT OF TRI CITIES		
WEBB, MIRIAM 7/22/2011 \$ 598.65 WEBB, MIRIAM 8/5/2011 \$ 886.00 WEBEX COMMUNICATIONS, INC 7/14/2011 \$ 49.00 WEBEX COMMUNICATIONS, INC 8/25/2011 \$ 300.00 WEBEX COMMUNICATIONS, INC 8/25/2011 \$ 49.00 WEBEX COMMUNICATIONS, INC 9/15/2011 \$ 49.00 WEEMS, KIRK 7/8/2011 \$ 1,969.17 WEEMS, KIRK 8/5/2011 \$ 1,969.19 WEEMS, KIRK 8/5/2011 \$ 1,969.19 WEEMS, KIRK 8/19/2011 \$ 1,969.19 WEEMS, KIRK 9/2/2011 \$ 1,969.19 WEEMS, KIRK 9/16/2011 \$ 1,969.18 WELLS FARGO 7/18/2011 \$ 2,249.83 WELLS FARGO <td< td=""><td>WEBB, MIRIAM</td><td>7/8/2011</td><td>\$ 856.07</td></td<>	WEBB, MIRIAM	7/8/2011	\$ 856.07
WEBB, MIRIAM 8/5/2011 \$ 886.00 WEBEX COMMUNICATIONS, INC 7/14/2011 \$ 49.00 WEBEX COMMUNICATIONS, INC 8/4/2011 \$ 300.00 WEBEX COMMUNICATIONS, INC 9/15/2011 \$ 49.00 WEBEX COMMUNICATIONS, INC 9/15/2011 \$ 49.00 WEEMS, KIRK 7/22/2011 \$ 1,969.19 WEEMS, KIRK 8/5/2011 \$ 1,969.19 WEEMS, KIRK 8/15/2011 \$ 2,880.88 WEEMS, KIRK 8/19/2011 \$ 1,969.18 WEEMS, KIRK 8/19/2011 \$ 1,969.18 WEEMS, KIRK 9/2/2011 \$ 1,969.19 WEEMS, KIRK 9/16/2011 \$ 1,969.19 WEEMS, KIRK 9/16/2011 \$ 1,969.19 WELLS FARGO 9/16/2011 \$ 1,969.19 WELLS FARGO 8/10/2011 \$ 2,692.15 WELLS FARGO 7/11/2011 \$ 2,692.15 WELLS FARGO 7/11/2011 \$ 1,036 WELLS FARGO 7/12/2011 \$ 10.36 WELLS FARGO 8/19/2011 \$ 725.00 WELLS FARGO 8/16/	WEBB, MIRIAM		
WEBEX COMMUNICATIONS, INC 7/14/2011 \$ 49.00 WEBEX COMMUNICATIONS, INC 8/4/2011 \$ 300.00 WEBEX COMMUNICATIONS, INC 8/25/2011 \$ 300.00 WEBEX COMMUNICATIONS, INC 9/15/2011 \$ 49.00 WEEMS, KIRK 7/8/2011 \$ 1.969.17 WEEMS, KIRK 7/8/2011 \$ 1.969.19 WEEMS, KIRK 8/5/2011 \$ 1.969.19 WEEMS, KIRK 8/15/2011 \$ 1.969.19 WEEMS, KIRK 8/15/2011 \$ 1.969.17 WEEMS, KIRK 8/15/2011 \$ 1.969.17 WEEMS, KIRK 9/2/2011 \$ 1.969.19 WEEMS, KIRK 9/16/2011 \$ 1.969.19 WELLS FARGO 8/10/2011 \$ 1.969.19 WELLS FARGO 8/10/2011 \$ 1.969.19 WELLS FARGO 7/11/2011 \$ 1.969.19 WELL	WEBB, MIRIAM		
WEBEX COMMUNICATIONS, INC 8/4/2011 \$ 300.00 WEBEX COMMUNICATIONS, INC 8/25/2011 \$ 300.00 WEBEX COMMUNICATIONS, INC 9/15/2011 \$ 19.00 WEEMS, KIRK 7/8/2011 \$ 1969.17 WEEMS, KIRK 8/5/2011 \$ 1969.18 WEEMS, KIRK 8/5/2011 \$ 1969.18 WEEMS, KIRK 8/15/2011 \$ 1969.17 WEEMS, KIRK 8/19/2011 \$ 1969.17 WEEMS, KIRK 9/2/2011 \$ 1969.17 WEEMS, KIRK 9/16/2011 \$ 1969.17 WEEMS, KIRK 9/16/2011 \$ 1969.18 WEEMS, KIRK 9/16/2011 \$ 1969.18 WELLS FARGO 8/10/2011 \$ 1969.18 WELLS FARGO 7/8/2011 \$ 1969.18 WELLS FARGO 7/8/2011 \$ 1969.18 WELLS FARGO 7/11/2011 \$ 1969.15 WELLS FARGO 7/11/2011 \$ 1969.15 WELLS FARGO 7/11/2011 \$ 1969.15 WELLS FARGO 7/11/2011 \$ 37,192.38 WELLS FARGO 7/11/2011 \$ 37,192.38 WELLS FARGO 8/10/2011 \$ 36,914.91 WELLS FARGO 8/4/2011 \$ 725.00 WELLS FARGO 8/16/2011 \$ 35,958.30 WELLS FARGO 9/	WEBEX COMMUNICATIONS, INC		
WEBEX COMMUNICATIONS, INC 8/25/2011 \$ 49.00 WEBEX COMMUNICATIONS, INC 9/15/2011 \$ 1,969.10 WEEMS, KIRK 7/28/2011 \$ 1,969.17 WEEMS, KIRK 8/5/2011 \$ 1,969.18 WEEMS, KIRK 8/5/2011 \$ 1,969.18 WEEMS, KIRK 8/15/2011 \$ 2,880.88 WEEMS, KIRK 8/19/2011 \$ 1,969.17 WEEMS, KIRK 9/2/2011 \$ 1,969.17 WEEMS, KIRK 9/2/2011 \$ 1,969.17 WEEMS, KIRK 9/16/2011 \$ 1,969.17 WEEMS, KIRK 9/16/2011 \$ 1,969.17 WELLS FARGO 8/10/2011 \$ 2,692.15 WELLS FARGO 8/10/2011 \$ 410.36 WELLS FARGO 7/18/2011 \$ 37,192.38 WELLS FARGO 7/12/2011 \$ 37,192.38 WELLS FARGO 7/21/2011 \$ 725.00 WELLS FARGO 8/4/2011 \$ 720.00 WELLS FARGO 8/5/2011 \$ 35,958.30 WELLS FARGO 8/16/2011 \$ 35,958.30 WELLS FARGO 9/6/2011 \$ 30,583.48 WELLS FARGO 9/16/2011 \$ 30,583.48 WELLS FARGO 9/16/2011 \$ 30,583.48 WELLS FARGO 9/16/2011 \$ 30,583.48 WELLS FARGO	WEBEX COMMUNICATIONS, INC		
WEBEX COMMUNICATIONS, INC 9/15/2011 \$ 1,969.17 WEEMS, KIRK 7/8/2011 \$ 1,969.17 WEEMS, KIRK 8/5/2011 \$ 1,969.18 WEEMS, KIRK 8/5/2011 \$ 2,880.88 WEEMS, KIRK 8/19/2011 \$ 1,969.17 WEEMS, KIRK 8/19/2011 \$ 1,969.17 WEEMS, KIRK 9/16/2011 \$ 1,969.17 WEEMS, KIRK 9/16/2011 \$ 1,969.18 WELLS FARGO 8/10/2011 \$ 2,692.15 WELLS FARGO 7/8/2011 \$ 3,7192.38 WELLS FARGO 7/11/2011 \$ 725.00 WELLS FARGO 7/21/2011 \$ 725.00 WELLS FARGO 8/4/2011 \$ 37,873.26 WELLS FARGO 8/19/2011 \$ 33,983.26 WELLS FARGO 9/6/2011 \$ 3	WEBEX COMMUNICATIONS, INC		
WEEMS, KIRK 7/22/2011 \$ 1,969.19 WEEMS, KIRK 8/5/2011 \$ 1,969.18 WEEMS, KIRK 8/15/2011 \$ 2,880.88 WEEMS, KIRK 8/19/2011 \$ 1,969.17 WEEMS, KIRK 9/2/2011 \$ 1,969.19 WEEMS, KIRK 9/16/2011 \$ 1,969.19 WEEMS, KIRK 9/30/2011 \$ 2,692.15 WELLS FARGO 8/10/2011 \$ 410.36 WELLS FARGO 7/8/2011 \$ 2,249.83 WELLS FARGO 7/11/2011 \$ 37,192.38 WELLS FARGO 7/12/2011 \$ 410.36 WELLS FARGO 7/12/2011 \$ 410.36 WELLS FARGO 7/12/2011 \$ 725.00 WELLS FARGO 7/21/2011 \$ 725.00 WELLS FARGO 8/4/2011 \$ 720.00 WELLS FARGO 8/5/2011 \$ 36,914.91 WELLS FARGO 8/16/2011 \$ 37,873.26 WELLS FARGO 8/19/2011 \$ 33,468.35 WELLS FARGO 9/6/2011 \$ 30,844.22 WELLS FARGO 9/15/2011 \$ 30,834.8 WELLS FARGO 9/15/2011 \$ 30,834.8 WELLS FARGO 9/21/2011 \$ 30,834.8 WELLS FARGO 9/21/2011 \$ 30,834.8 WELLS FARGO 9/21/2011 \$ 15.35 WELLS FARGO BANK 9/21/2011 \$ 1,720.00 <	WEBEX COMMUNICATIONS, INC		
WEEMS, KIRK 8/5/2011 \$ 1,969.18 WEEMS, KIRK 8/15/2011 \$ 2,880.88 WEEMS, KIRK 8/19/2011 \$ 1,969.17 WEEMS, KIRK 9/2/2011 \$ 1,969.19 WEEMS, KIRK 9/16/2011 \$ 1,969.18 WEEMS, KIRK 9/30/2011 \$ 2,692.15 WELLS FARGO 8/10/2011 \$ 410.36 WELLS FARGO 7/8/2011 \$ 2,249.83 WELLS FARGO 7/11/2011 \$ 37,192.38 WELLS FARGO 7/11/2011 \$ 410.36 WELLS FARGO 7/12/2011 \$ 410.36 WELLS FARGO 7/12/2011 \$ 410.36 WELLS FARGO 7/21/2011 \$ 725.00 WELLS FARGO 8/4/2011 \$ 720.00 WELLS FARGO 8/16/2011 \$ 35,958.30 WELLS FARGO 8/19/2011 \$ 33,468.35 WELLS FARGO 9/6/2011 \$ 30,844.22 WELLS FARGO 9/15/2011 \$ 30,844.22 WELLS FARGO 9/15/2011 \$ 30,844.23 WELLS FARGO 9/15/2011 \$ 30,834.83	WEEMS, KIRK	7/8/2011	\$ 1,969.17
WEEMS, KIRK 8/15/2011 \$ 2,880.88 WEEMS, KIRK 8/19/2011 \$ 1,969.17 WEEMS, KIRK 9/2/2011 \$ 1,969.19 WEEMS, KIRK 9/16/2011 \$ 1,969.19 WEEMS, KIRK 9/30/2011 \$ 2,692.15 WELLS FARGO 8/10/2011 \$ 410.36 WELLS FARGO 7/8/2011 \$ 2,249.83 WELLS FARGO 7/11/2011 \$ 37,192.38 WELLS FARGO 7/21/2011 \$ 410.36 WELLS FARGO 7/21/2011 \$ 725.00 WELLS FARGO 7/21/2011 \$ 725.00 WELLS FARGO 8/4/2011 \$ 720.00 WELLS FARGO 8/5/2011 \$ 37,873.26 WELLS FARGO 8/16/2011 \$ 35,958.30 WELLS FARGO 9/6/2011 \$ 30,844.22 WELLS FARGO 9/15/2011 \$ 706.67 WELLS FARGO 9/16/2011 \$ 30,583.48 WELLS FARGO 9/16/2011 \$ 1,520.00 WELLS FARGO 9/16/2011 \$ 1,525 WELLS FARGO 9/27/2011 \$ 1,720.00 WELLS FARGO BANK 9/21/2011 \$ 1,720.00 <tr< td=""><td>WEEMS, KIRK</td><td></td><td>·</td></tr<>	WEEMS, KIRK		·
WEEMS, KIRK 8/15/2011 \$ 2,880.88 WEEMS, KIRK 8/19/2011 \$ 1,969.17 WEEMS, KIRK 9/2/2011 \$ 1,969.19 WEEMS, KIRK 9/16/2011 \$ 1,969.18 WEEMS, KIRK 9/30/2011 \$ 2,692.15 WELLS FARGO 8/10/2011 \$ 410.36 WELLS FARGO 7/8/2011 \$ 2,249.83 WELLS FARGO 7/11/2011 \$ 37,192.38 WELLS FARGO 7/12/2011 \$ 410.36 WELLS FARGO 7/21/2011 \$ 725.00 WELLS FARGO 8/4/2011 \$ 720.00 WELLS FARGO 8/4/2011 \$ 37,873.26 WELLS FARGO 8/16/2011 \$ 35,958.30 WELLS FARGO 8/16/2011 \$ 30,844.22 WELLS FARGO 9/15/2011 \$ 706.67 WELLS FARGO 9/15/2011 \$ 706.67 WELLS FARGO 9/16/2011 \$ 1,535 WELLS FARGO BANK 9/21/2011 1,720.00 WEST PAYMENT CENTER 7/14/2011 1,720.00 WEST PAYMENT CENTER 7/14/2011 1,720.00	WEEMS, KIRK	8/5/2011	\$ 1,969.18
WEEMS, KIRK 9/2/2011 \$ 1,969.19 WEEMS, KIRK 9/16/2011 \$ 1,969.18 WEEMS, KIRK 9/30/2011 \$ 2,692.15 WELLS FARGO 8/10/2011 \$ 410.36 WELLS FARGO 7/8/2011 \$ 2,249.83 WELLS FARGO 7/11/2011 \$ 410.36 WELLS FARGO 7/12/2011 \$ 410.36 WELLS FARGO 7/21/2011 \$ 725.00 WELLS FARGO 7/22/2011 \$ 36,914.91 WELLS FARGO 8/4/2011 \$ 720.00 WELLS FARGO 8/5/2011 \$ 37,873.26 WELLS FARGO 8/16/2011 \$ 33,468.35 WELLS FARGO 9/6/2011 \$ 33,468.35 WELLS FARGO 9/6/2011 \$ 30,844.22 WELLS FARGO 9/15/2011 \$ 706.67 WELLS FARGO 9/15/2011 \$ 30,834.8 WELLS FARGO 9/27/2011 \$ 36,023.43 WELLS FARGO BANK 9/21/2011 \$ 15.35 WEST PAYMENT CENTER 7/5/2011 \$ 1,720.00 WEST PAYMENT CENTER 7/14/2011 \$ 1,720.00 WEST PAYMENT CENTER 7/14/2011 \$ 1,720.00	WEEMS, KIRK	8/15/2011	
WEEMS, KIRK 9/2/2011 \$ 1,969.19 WEEMS, KIRK 9/16/2011 \$ 1,969.18 WEEMS, KIRK 9/30/2011 \$ 2,692.15 WELLS FARGO 8/10/2011 \$ 410.36 WELLS FARGO 7/8/2011 \$ 2,249.83 WELLS FARGO 7/11/2011 \$ 37,192.38 WELLS FARGO 7/21/2011 \$ 410.36 WELLS FARGO 7/21/2011 \$ 725.00 WELLS FARGO 7/22/2011 \$ 36,914.91 WELLS FARGO 8/4/2011 \$ 720.00 WELLS FARGO 8/5/2011 \$ 37,873.26 WELLS FARGO 8/16/2011 \$ 35,958.30 WELLS FARGO 8/19/2011 \$ 33,468.35 WELLS FARGO 9/6/2011 \$ 30,844.22 WELLS FARGO 9/15/2011 \$ 706.67 WELLS FARGO 9/15/2011 \$ 30,834.8 WELLS FARGO 9/27/2011 \$ 36,023.43 WELLS FARGO BANK 9/21/2011 \$ 15.35 WEST PAYMENT CENTER 7/5/2011 \$ 1,720.00 WEST PAYMENT CENTER 7/14/2011 \$ 1,720.00 WEST PAYMENT CENTER 7/14/2011 \$ 1,720.00	WEEMS, KIRK	•	•
WEEMS, KIRK 9/30/2011 \$ 2,692.15 WELLS FARGO 8/10/2011 \$ 410.36 WELLS FARGO 7/8/2011 \$ 2,249.83 WELLS FARGO 7/11/2011 \$ 37,192.38 WELLS FARGO 7/12/2011 \$ 410.36 WELLS FARGO 7/21/2011 \$ 725.00 WELLS FARGO 7/22/2011 \$ 36,914.91 WELLS FARGO 8/4/2011 \$ 720.00 WELLS FARGO 8/5/2011 \$ 37,873.26 WELLS FARGO 8/16/2011 \$ 35,958.30 WELLS FARGO 8/19/2011 \$ 33,468.35 WELLS FARGO 9/6/2011 \$ 30,844.22 WELLS FARGO 9/15/2011 \$ 706.67 WELLS FARGO 9/16/2011 \$ 30,583.48 WELLS FARGO BANK 9/21/2011 \$ 15.35 WEST PAYMENT CENTER 7/5/2011 \$ 1,720.00 WEST PAYMENT CENTER 7/14/2011 \$ 1,720.00 WEST PAYMENT CENTER 8/25/2011 \$ 1,720.00	WEEMS, KIRK	9/2/2011	\$ 1,969.19
WELLS FARGO 8/10/2011 \$ 410.36 WELLS FARGO 7/8/2011 \$ 2,249.83 WELLS FARGO 7/11/2011 \$ 37,192.38 WELLS FARGO 7/12/2011 \$ 410.36 WELLS FARGO 7/21/2011 \$ 725.00 WELLS FARGO 7/22/2011 \$ 36,914.91 WELLS FARGO 8/4/2011 \$ 720.00 WELLS FARGO 8/5/2011 \$ 37,873.26 WELLS FARGO 8/16/2011 \$ 35,958.30 WELLS FARGO 8/19/2011 \$ 33,468.35 WELLS FARGO 9/6/2011 \$ 30,844.22 WELLS FARGO 9/15/2011 \$ 706.67 WELLS FARGO 9/16/2011 \$ 30,583.48 WELLS FARGO 9/27/2011 \$ 36,023.43 WELLS FARGO BANK 9/21/2011 \$ 15.35 WEST PAYMENT CENTER 7/5/2011 \$ 1,720.00 WEST PAYMENT CENTER 7/14/2011 \$ 1,720.00 WEST PAYMENT CENTER 8/25/2011 \$ 1,720.00	WEEMS, KIRK	9/16/2011	\$ 1,969.18
WELLS FARGO 7/8/2011 \$ 2,249.83 WELLS FARGO 7/11/2011 \$ 37,192.38 WELLS FARGO 7/12/2011 \$ 410.36 WELLS FARGO 7/21/2011 \$ 725.00 WELLS FARGO 7/22/2011 \$ 36,914.91 WELLS FARGO 8/4/2011 \$ 720.00 WELLS FARGO 8/5/2011 \$ 37,873.26 WELLS FARGO 8/16/2011 \$ 35,958.30 WELLS FARGO 8/19/2011 \$ 33,468.35 WELLS FARGO 9/6/2011 \$ 30,844.22 WELLS FARGO 9/15/2011 \$ 706.67 WELLS FARGO 9/16/2011 \$ 30,583.48 WELLS FARGO 9/27/2011 \$ 36,023.43 WELLS FARGO BANK 9/21/2011 \$ 15.35 WEST PAYMENT CENTER 7/5/2011 \$ 1,720.00 WEST PAYMENT CENTER 7/14/2011 \$ 1,720.00 WEST PAYMENT CENTER 8/25/2011 \$ 1,720.00	WEEMS, KIRK	9/30/2011	\$ 2,692.15
WELLS FARGO 7/11/2011 \$ 37,192.38 WELLS FARGO 7/12/2011 \$ 410.36 WELLS FARGO 7/21/2011 \$ 725.00 WELLS FARGO 7/22/2011 \$ 36,914.91 WELLS FARGO 8/4/2011 \$ 720.00 WELLS FARGO 8/5/2011 \$ 37,873.26 WELLS FARGO 8/16/2011 \$ 35,958.30 WELLS FARGO 8/19/2011 \$ 33,468.35 WELLS FARGO 9/6/2011 \$ 30,844.22 WELLS FARGO 9/15/2011 \$ 706.67 WELLS FARGO 9/16/2011 \$ 30,583.48 WELLS FARGO BANK 9/27/2011 \$ 15.35 WEST PAYMENT CENTER 7/5/2011 \$ 1,720.00 WEST PAYMENT CENTER 7/14/2011 \$ 1,720.00 WEST PAYMENT CENTER 8/25/2011 \$ 1,720.00	WELLS FARGO	8/10/2011	\$ 410.36
WELLS FARGO 7/12/2011 \$ 410.36 WELLS FARGO 7/21/2011 \$ 725.00 WELLS FARGO 7/22/2011 \$ 36,914.91 WELLS FARGO 8/4/2011 \$ 720.00 WELLS FARGO 8/5/2011 \$ 37,873.26 WELLS FARGO 8/16/2011 \$ 35,958.30 WELLS FARGO 8/19/2011 \$ 33,468.35 WELLS FARGO 9/6/2011 \$ 30,844.22 WELLS FARGO 9/15/2011 \$ 706.67 WELLS FARGO 9/16/2011 \$ 30,583.48 WELLS FARGO BANK 9/21/2011 \$ 15.35 WEST PAYMENT CENTER 7/5/2011 \$ 1,720.00 WEST PAYMENT CENTER 7/14/2011 \$ 1,720.00 WEST PAYMENT CENTER 8/25/2011 \$ 1,720.00	WELLS FARGO	7/8/2011	\$ 2,249.83
WELLS FARGO 7/21/2011 \$ 725.00 WELLS FARGO 7/22/2011 \$ 36,914.91 WELLS FARGO 8/4/2011 \$ 720.00 WELLS FARGO 8/5/2011 \$ 37,873.26 WELLS FARGO 8/16/2011 \$ 35,958.30 WELLS FARGO 8/19/2011 \$ 33,468.35 WELLS FARGO 9/6/2011 \$ 30,844.22 WELLS FARGO 9/15/2011 \$ 706.67 WELLS FARGO 9/16/2011 \$ 30,583.48 WELLS FARGO BANK 9/27/2011 \$ 36,023.43 WELS FARGO BANK 9/21/2011 \$ 1,720.00 WEST PAYMENT CENTER 7/5/2011 \$ 1,720.00 WEST PAYMENT CENTER 7/14/2011 \$ 1,720.00 WEST PAYMENT CENTER 8/25/2011 \$ 1,720.00	WELLS FARGO	7/11/2011	\$ 37,192.38
WELLS FARGO 7/22/2011 \$ 36,914.91 WELLS FARGO 8/4/2011 \$ 720.00 WELLS FARGO 8/5/2011 \$ 37,873.26 WELLS FARGO 8/16/2011 \$ 35,958.30 WELLS FARGO 8/19/2011 \$ 33,468.35 WELLS FARGO 9/6/2011 \$ 30,844.22 WELLS FARGO 9/15/2011 \$ 706.67 WELLS FARGO 9/16/2011 \$ 30,583.48 WELLS FARGO 9/27/2011 \$ 36,023.43 WELLS FARGO BANK 9/21/2011 \$ 15.35 WEST PAYMENT CENTER 7/5/2011 \$ 1,720.00 WEST PAYMENT CENTER 7/14/2011 \$ 1,720.00 WEST PAYMENT CENTER 8/25/2011 \$ 1,720.00	WELLS FARGO	7/12/2011	\$ 410.36
WELLS FARGO 8/4/2011 \$ 720.00 WELLS FARGO 8/5/2011 \$ 37,873.26 WELLS FARGO 8/16/2011 \$ 35,958.30 WELLS FARGO 8/19/2011 \$ 33,468.35 WELLS FARGO 9/6/2011 \$ 706.67 WELLS FARGO 9/15/2011 \$ 706.67 WELLS FARGO 9/16/2011 \$ 30,583.48 WELLS FARGO 9/27/2011 \$ 36,023.43 WELLS FARGO BANK 9/21/2011 \$ 15.35 WEST PAYMENT CENTER 7/5/2011 \$ 1,720.00 WEST PAYMENT CENTER 7/14/2011 \$ 1,720.00 WEST PAYMENT CENTER 8/25/2011 \$ 1,720.00		7/21/2011	\$ 725.00
WELLS FARGO 8/5/2011 \$ 37,873.26 WELLS FARGO 8/16/2011 \$ 35,958.30 WELLS FARGO 8/19/2011 \$ 33,468.35 WELLS FARGO 9/6/2011 \$ 706.67 WELLS FARGO 9/15/2011 \$ 706.67 WELLS FARGO 9/16/2011 \$ 30,583.48 WELLS FARGO 9/27/2011 \$ 36,023.43 WELLS FARGO BANK 9/21/2011 \$ 15.35 WEST PAYMENT CENTER 7/5/2011 \$ 1,720.00 WEST PAYMENT CENTER 7/14/2011 \$ 1,720.00 WEST PAYMENT CENTER 8/25/2011 \$ 1,720.00	WELLS FARGO	7/22/2011	\$ 36,914.91
WELLS FARGO 8/16/2011 \$ 35,958.30 WELLS FARGO 8/19/2011 \$ 33,468.35 WELLS FARGO 9/6/2011 \$ 30,844.22 WELLS FARGO 9/15/2011 \$ 706.67 WELLS FARGO 9/16/2011 \$ 30,583.48 WELLS FARGO 9/27/2011 \$ 36,023.43 WELLS FARGO BANK 9/21/2011 \$ 15.35 WEST PAYMENT CENTER 7/5/2011 \$ 1,720.00 WEST PAYMENT CENTER 7/14/2011 \$ 1,720.00 WEST PAYMENT CENTER 8/25/2011 \$ 1,720.00	WELLS FARGO	8/4/2011	\$ 720.00
WELLS FARGO 8/19/2011 \$ 33,468.35 WELLS FARGO 9/6/2011 \$ 30,844.22 WELLS FARGO 9/15/2011 \$ 706.67 WELLS FARGO 9/16/2011 \$ 30,583.48 WELLS FARGO 9/27/2011 \$ 36,023.43 WELLS FARGO BANK 9/21/2011 \$ 15.35 WEST PAYMENT CENTER 7/5/2011 \$ 1,720.00 WEST PAYMENT CENTER 7/14/2011 \$ 1,720.00 WEST PAYMENT CENTER 8/25/2011 \$ 1,720.00	WELLS FARGO	8/5/2011	\$ 37,873.26
WELLS FARGO 9/6/2011 \$ 30,844.22 WELLS FARGO 9/15/2011 \$ 706.67 WELLS FARGO 9/16/2011 \$ 30,583.48 WELLS FARGO 9/27/2011 \$ 36,023.43 WELLS FARGO BANK 9/21/2011 \$ 15.35 WEST PAYMENT CENTER 7/5/2011 \$ 1,720.00 WEST PAYMENT CENTER 7/14/2011 \$ 1,720.00 WEST PAYMENT CENTER 8/25/2011 \$ 1,720.00	WELLS FARGO	8/16/2011	\$ 35,958.30
WELLS FARGO 9/15/2011 \$ 706.67 WELLS FARGO 9/16/2011 \$ 30,583.48 WELLS FARGO 9/27/2011 \$ 36,023.43 WELLS FARGO BANK 9/21/2011 \$ 15.35 WEST PAYMENT CENTER 7/5/2011 \$ 1,720.00 WEST PAYMENT CENTER 7/14/2011 \$ 1,720.00 WEST PAYMENT CENTER 8/25/2011 \$ 1,720.00	WELLS FARGO	8/19/2011	\$ 33,468.35
WELLS FARGO 9/16/2011 \$ 30,583.48 WELLS FARGO 9/27/2011 \$ 36,023.43 WELLS FARGO BANK 9/21/2011 \$ 15.35 WEST PAYMENT CENTER 7/5/2011 \$ 1,720.00 WEST PAYMENT CENTER 7/14/2011 \$ 1,720.00 WEST PAYMENT CENTER 8/25/2011 \$ 1,720.00	WELLS FARGO	9/6/2011	\$ 30,844.22
WELLS FARGO 9/27/2011 \$ 36,023.43 WELLS FARGO BANK 9/21/2011 \$ 15.35 WEST PAYMENT CENTER 7/5/2011 \$ 1,720.00 WEST PAYMENT CENTER 7/14/2011 \$ 1,720.00 WEST PAYMENT CENTER 8/25/2011 \$ 1,720.00	WELLS FARGO	9/15/2011	\$ 706.67
WELLS FARGO BANK 9/21/2011 \$ 15.35 WEST PAYMENT CENTER 7/5/2011 \$ 1,720.00 WEST PAYMENT CENTER 7/14/2011 \$ 1,720.00 WEST PAYMENT CENTER 8/25/2011 \$ 1,720.00	WELLS FARGO	9/16/2011	\$ 30,583.48
WEST PAYMENT CENTER 7/5/2011 \$ 1,720.00 WEST PAYMENT CENTER 7/14/2011 \$ 1,720.00 WEST PAYMENT CENTER 8/25/2011 \$ 1,720.00	WELLS FARGO	9/27/2011	\$ 36,023.43
WEST PAYMENT CENTER 7/14/2011 \$ 1,720.00 WEST PAYMENT CENTER 8/25/2011 \$ 1,720.00	WELLS FARGO BANK	9/21/2011	\$ 15.35
WEST PAYMENT CENTER 8/25/2011 \$ 1,720.00	WEST PAYMENT CENTER	7/5/2011	\$ 1,720.00
	·	7/14/2011	\$ 1,720.00
WEST WHITELAND TOWN SHIP 7/21/2011 \$ 372.00		8/25/2011	\$ 1,720.00
	WEST WHITELAND TOWN SHIP	7/21/2011	\$ 372.00

Name of creditor	Payment date	Amount paid
WEST WHITELAND TOWN SHIP	9/21/2011	\$ 82.00
WETZEL, SUZANNE	7/8/2011	
WETZEL, SUZANNE	7/22/2011	•
WETZEL, SUZANNE	8/5/2011	
WETZEL, SUZANNE	8/15/2011	· · · · · · · · · · · · · · · · · · ·
WETZEL, SUZANNE	8/19/2011	
WETZEL, SUZANNE	9/2/2011	
WETZEL, SUZANNE	9/16/2011	•
WHITE, GREGORY	7/5/2011	
WHITE, GREGORY	7/8/2011	i i
WHITE, GREGORY	7/22/2011	· ·
WHITE, GREGORY	8/5/2011	- ·
WHITE, GREGORY	8/8/2011	·
WHITE, GREGORY	8/15/2011	
WHITE, GREGORY	8/19/2011	•
WHITE, GREGORY	8/25/2011	- '
WHITE, GREGORY	9/2/2011	
WHITE, GREGORY	9/16/2011	
WHITE, GREGORY	9/26/2011	•
WHITE, GREGORY	9/30/2011	"
WHITSON, CHERIE	7/8/2011	
WHITSON, CHERIE	7/22/2011	· ·
WHITSON, CHERIE	8/5/2011	
WHITSON, CHERIE	8/19/2011	
WHITSON, CHERIE	9/2/2011	
WHITSON, CHERIE	9/16/2011	
WHITSON, CHERIE	9/30/2011	
WILLIAMS, BRADY	7/8/2011	·
WILLIAMS, BRADY	7/22/2011	·
WILLIAMS, BRADY	8/5/2011	
WILLIAMS, BRADY	8/15/2011	
WILLIAMS, BRADY	8/19/2011	•
WILLIAMS, JEREMY	7/8/2011	•
WILLIAMS, JEREMY	7/22/2011	
WILLIAMS, JEREMY	8/5/2011	·
WILLIAMS, JEREMY	8/19/2011	
WILLIAMS, JEREMY	9/2/2011	
WILLIAMS, JEREMY	9/16/2011	
WILLIAMS, JEREMY	9/30/2011	
WITTE, JULIE E	7/8/2011	
WITTE, JULIE E	8/15/2011	-
WITTE, MS JULIE	7/5/2011	
WOLF, SHAWN	7/8/2011	
WOLF, SHAWN	7/11/2011	
WOLF, SHAWN	7/22/2011	
WOLF, SHAWN	7/25/2011	•
WOLF, SHAWN	8/5/2011	
WOLF, SHAWN	8/8/2011	· ·
WOLF, SHAWN	8/15/2011	
WOLF, SHAWN	8/19/2011	
WOLF, SHAWN	9/2/2011	•
WOLF, SHAWN	9/8/2011	•
WOLF, SHAWN	9/16/2011	
WOLF, SHAWN		•
WOLK, OLLIWIN	9/21/2011	\$ 118.00

Name of creditor	Payment date		Amount paid
WOLF, SHAWN	9/30/2011	\$	1,936.20
WOLFE, JEFFREY	7/5/2011		117.66
WOLFE, JEFFREY	7/25/2011		147.53
WOLFE, JEFFREY	8/15/2011		149.31
WOLFE, JEFFREY	9/8/2011		43.75
WOLFE, JEFFREY	9/15/2011	\$	98.58
WOLFE, JEFFREY J	7/8/2011	\$	2,494.91
WOLFE, JEFFREY J	7/22/2011		2,494.91
WOLFE, JEFFREY J	8/5/2011		2,494.90
WOLFE, JEFFREY J	8/15/2011	\$	1,906.76
WOLFE, JEFFREY J	8/19/2011		2,494.91
WOLFE, JEFFREY J	9/2/2011	. \$	2,494.91
WOLFE, JEFFREY J	9/16/2011		2,494.90
WOLFE, JEFFREY J	9/30/2011	. \$	3,605.33
WOLKOV LAW	9/16/2011	\$	750.00
WOLTERS KLUWER HEALTH	7/21/2011	\$	10,187.25
WOLTERS KLUWER HEALTH	8/25/2011		5,000.00
WOOTEN, SHANNON	7/8/2013	\$	730.16
WOOTEN, SHANNON	7/22/2011	L \$	808.12
WOOTEN, SHANNON	8/5/2011	L \$	642.45
WRIGHT EXPRESS	7/6/2011	l \$	42,909.52
WRIGHT EXPRESS	8/8/201	\$	40,838.92
WRIGHT EXPRESS	9/7/201	1 \$	42,660.84
WV DEPT OF HEALTH & HUMAN RESOURCES	7/5/201	1 \$	1,041.87
WV DEPT OF HEALTH & HUMAN RESOURCES	7/5/201	1 \$	55,418.97
YALE CAROLINAS, INC.	9/8/201		154.86
YOUNG CONAWAY STARGATT & TAYLOR LLP	9/26/201	1 \$	50,000.00
YOUNG CONAWAY STARGATT & TAYLOR LLP	9/28/201	1 \$	25,000.00
ZAIDI, TARIQ	7/8/201	1 \$	4,055.24
ZAIDI, TARIQ	7/22/201	1 \$	4,055.25
ZAIDI, TARIQ	8/5/201	1 \$	4,055.24
ZAIDI, TARIQ	8/19/201	1 \$	4,055.24
ZAIDI, TARIQ	9/2/201	1 \$	4,288.11
ZAIDI, TARIQ	9/16/201	1 \$	4,295.73
ZAIDI, TARIQ	9/30/201	1 \$	4,295.73
ZOURDOS, WILLIAM	8/19/201	1 \$	962.03
ZOURDOS, WILLIAM	9/2/201	1 \$	2,090.09
ZOURDOS, WILLIAM	9/16/201	1 \$	2,426.91
ZOURDOS, WILLIAM	9/30/201	1 \$	2,894.32
	TOTAL	\$	34,511,501.06

EXHIBIT F

Insider Transfers Subject to Preference Review

Exhibit F

NAME	1	TAL AMOUNT ANSFERRED
BABILON, ROBERT VP, PRODUCT DEVELOPMENT PROJECTS	\$	17,656.55
BACKES, KEITH D VP, TRADE & MANAGED MKTS	\$	21,598.11
BAKER, KENNETH P. VP, TRADE & MANAGED MKTS	\$	17,495.83
BELLAMY, JOHN A EVP, LEGAL & GEN COUNSEL	\$	382,643.97
BLESSING, HERSCHEL SR VP, LOGISTICS	\$	214,945.39
BOWLES, JOHN VP, CORPORATE ATTORNEY	\$	15,164.49
BRENNAN, SEAN SR VP, REGULATORY AFFAIRS	\$	28,129.81
CORNERSTONE INK, LLC	\$	1,219.93
COWGILL, BRENT VP, MARKETING	\$	20,573.43
CURTIN, CHRISTOPHER EVP, TECH OPS	\$	269,218.37
CURTIN, JONATHAN DISTRIBUTION TEMPORARY/ FAMILY MEMBER	\$	3,827.04
DER, TOMMY WAYNE VP, REGULATORY AFFAIRS	\$	9,322.05
DUNN, KENNETH C. BOARD MEMBER	\$	28,021.00
FABRY, JOEL VP, SALES	\$	19,175.91
FIORENTINO, EDWARD J. BOARD MEMBER	\$	40,000.00
GREGORY, JEFFERSON J CHAIRMAN & CEO	\$	511,773.35
GTCR GOLDER RAUNER II, LLC	\$	8,389.01

Exhibit F

NAME	AL AMOUNT NSFERRED
JONES, GREGORY C EXEC VP STRATEGIC DEVELOPMENT	\$ 310,629.66
KUPSKI, STEPHANIE L. VP, PRICING PROGS & MANAGED CARE CONTRACTS	\$ 15,271.97
LEE, JAMES CHIEF MEDICAL OFFICER	\$ 247,584.67
LEVY, SHARON SR VP, PRODUCT DEVELOPMENT CLINICAL	\$ 162,293.03
MEADOWS, LISA SR VP, HUMAN RESOURCES	\$ 12,856.80
MEYERSON, BORIS SR VP, BUSINESS OPERATIONS	\$ 22,306.14
MOCCIA, CHRISTOPHER SALES, PROFESSIONAL/ FAMILY MEMBER	\$ 11,932.97
MOCCIA, ROBERT J COO & PRESIDENT	\$ 484,362.67
MUSICK, JOHN SR VP HUMAN RESOURCES	\$ 21,717.73
NEWTON, JOEL T SR DIRECTOR, LOGISTICS/ FAMILY MEMBER	\$ 12,253.75
NORDSIEK, MICHAEL T CHIEF SCIENCE OFFICER	\$ 376,841.89
NORDSIEK, SCOTT EXTON TEMPORARY/ FAMILY MEMBER	\$ 1,776.25
ROOSEVELT, GARY A VICE PRESIDENT, MEDICAL EDUCATION	\$ 18,601.51
RUTLEDGE, DANIEL VP, QUALITY OPERATIONS	\$ 18,046.45
SHRADER, BRIAN CHIEF FINANCIAL OFFICER	\$ 473,452.66
SJ STRATEGIC INVESTMENTS, LLC	\$ 483,398.41

Exhibit F

NAME		TOTAL AMOUNT TRANSFERRED	
SWETNAM, THOMAS SR VP INFORMATION TECHNOLOGY		\$	15,438.08
WHITSON, CHERIE VP, CONTROLLER		\$	12,662.06
WITHROW JR, ROBERT B VP, TREASURER		\$	138,877.93
YON, XAVIER BOARD MEMBER		\$	92,130.10
ZAIDI, TARIQ VP, CORPORATE COUNSEL BUS DEVELOPMENT		\$	15,763.51
	TOTAL	\$	4,557,352.48

EXHIBIT G

Distribution to Insiders On March 18, 2010

EXHBIT G

NAME	AMOUNT TRANSFERRED
GTCR Fund VIII	\$2,431,239.00
Fund VIII/B Graceway Splitter	\$426,684.00
GTCR Co-invest II	\$12,977.00
GTCR Fund IX	\$4,279,698.00
GTCR Fund IX/B	\$713,585.00
GTCR Co-invest III	\$32,209.00
Xavier Yon	\$21,900.00
Gracetree Investments	\$533,933.00
2nd Jordon Gregory Trust	\$66,742.00
2nd Jamison Gregory Trust	\$66,742.00
Robert Moccia	\$236,748.00
Michael Nordsiek	\$95,082.00
John Bellamy	\$46,585.00
Chris Curtin	\$23,291.00
Gregory Jones	\$23,291.00
Brian Shrader	\$93,169.00
Dave Fadness	\$23,291.00